

# 2024 ELECTION BOOK FORMS/INSTRUCTIONS

# FOR CITY COUNCIL POSITIONS 1, 2 & 3

## **First Steps for Candidates Running for a City Office**

This quick-start guide for candidates seeking a city office is not intended to provide comprehensive information. For more details, including information on political advertising requirements, fundraising rules, and filing schedules, see the Texas Ethics Commission's (TEC) website at www.ethics.state.tx.us.

#### 1. All candidates must file a Campaign Treasurer Appointment (Form CTA).

All candidates must file Form CTA even if you do not intend to raise or spend any money. Form CTA is required to be filed before you file an application for a place on the ballot, raise or spend any money for your campaign, or announce your candidacy. File Form CTA with the city clerk or city secretary, as applicable.

#### 2. Opposed Candidates: Will you accept or spend more than <u>\$1,010</u>\* for the election?

- <u>YES:</u>
  - You do not qualify to file on the modified reporting schedule.
  - You are *required* to file pre-election campaign finance reports using Form C/OH if you have an opponent on the ballot. Find Form C/OH and its instructions on our "Local Filers Non-Judicial Candidate/Officeholder" webpage.
  - Pre-election reports are due 30 days and 8 days prior to each election. To be timely filed, pre-election reports must be <u>received</u> by the city clerk or city secretary no later than the due date.
- <u>NO:</u>
  - You can elect to file on the modified reporting schedule by completing the *Modified Reporting Declaration* on page two of Form CTA. File Form CTA with the city clerk or city secretary.
  - If you elect to file on the modified reporting schedule, you do not have to file preelection campaign finance reports due 30 days and 8 days prior to the election.
- <u>Exceed \$1,010</u>\*: If you elect to file on the modified reporting schedule but later exceed \$1,010 in either contributions or expenditures, what reports you will be required to file depends upon when you exceed \$1,010\*.
  - If you exceed \$1,010\* <u>on or before</u> the 30th day before the election, you are *required* to file pre-election campaign finance reports due 30 days and 8 days prior to an election using Form C/OH. To be timely filed, pre-election reports must be <u>received</u> by the city clerk or city secretary no later than the due date. Find Form C/OH and its instructions on our "Local Filers Non-Judicial Candidate/Officeholder" webpage.
  - If you exceed \$1,010\* <u>after</u> the 30th day before the election, you are *required* to file an Exceeded Modified Reporting Limit report using Form C/OH. To be timely filed, this report must be filed with the city clerk or city secretary within 48 hours of exceeding \$1,010.\* You must also file the pre-election report due 8 days prior to an election. Find Form C/OH and its instructions on our "Local Filers Non-Judicial Candidate/Officeholder" webpage. If you exceed \$1,010\* <u>on or before</u> the 8th day before the election, you are *required* to file a pre-election campaign finance report due 8 days prior to an election using Form C/OH. To be timely filed, the pre-election

report must be <u>received</u> by the city clerk or city secretary no later than the due date. Find Form C/OH and its instructions on our "Local Filers Non-Judicial Candidate/Officeholder" webpage.

#### 3. Unopposed Candidates.

If you do not have an opponent whose name will appear on the ballot in the election, you are an unopposed candidate and are not required to file pre-election campaign finance reports prior to that election.

#### 4. All candidates must file semiannual campaign finance reports (Form C/OH).

All candidates are *required* to file semiannual reports using Form C/OH even if you have no campaign activity or were unsuccessful in the election. Semiannual reports are due on January 15th and July 15th and must be filed with the city clerk or city secretary. To end your filing obligations, you must cease campaign activity and file a Final report using Form C/OH and attaching Form C/OH-FR (Designation of Final Report). Form C/OH-FR is found on the last page of Form C/OH. Find Form C/OH and its instructions on our "Local Filers Non-Judicial Candidate/Officeholder" webpage. For more information, see "Ending Your Campaign" for local filers.

# 5. All candidates can use the TEC's Filing Application to prepare campaign finance reports (Form C/OH).

You can use the TEC's Filing Application to prepare a PDF version of your campaign finance report (Form C/OH). Select "Local Authority" and follow the steps to set up an account and login to the application. Once you have completed your report, print out a copy, add your treasurer information, get it notarized, and file it with the city clerk or city secretary by the appropriate deadline.

#### 6. Need More Information?

See the Campaign Finance Guide for Candidates and Officeholders Who File With Local Filing Authorities, forms, instructions, examples on how to disclose contributions and expenditures, political advertising and fundraising guides, and other information you may find useful on our website at *www.ethics.state.tx.us* under the "Resources" and "Forms/Instructions" main menu items.

\***NOTE:** The \$1,010 threshold is specific to transactions made in 2023.



## **INDEX TO LOCAL FILING AUTHORITIES**

First Steps for Candidates Running for a City Office

### **1 –TEXAS ETHICS COMMISSION RULES**

2 – GUIDE TO LOCAL FILING AUTHORITY'S DUTIES UNDER THE CAMPAIGN FINANCE LAW

3 – CAMPAIGN FINANCE GUIDE FOR CANDIDATES AND OFFICEHOLDERS WHO FILE WITH LOCAL FILING AUTHORITIES

### 4 - POLITICAL ADVERTISING

**5 – CHAPTER 258, ELECTION CODE FAIR CAMPAIGN PRACTICES ACT** 

6 – APPOINTMENT OF A CAMPAIGN TREASURER BY A CANDIDATE

7 - AMENDMENT: APPOINTMENT OF A CAMPAIGN TREASURER BY A CANDIDATE

8- CANDIDATE/OFFICEHOLDER CAMPAIGN FINANCE REPORT

9 – TITLE 15, ELECTION CODE REGULATING POLITICAL FUNDS AND CAMPAIGNS

**10 – Chapter 302 – SPEAKER OF THE HOUSE OF REPRESENTATIVES** 

### 11- PERSONAL FINANCIAL DISCLOSURE, STANDARDS OF CONDUCT AND CONFLICT OF INTEREST

**12 – FILING SCHEDULE FOR REPORTS AND FORMS** 



# 1

# TEXAS ETHICS COMMISSION RULES

# **TEXAS ETHICS COMMISSION RULES**



Effective August 23, 2023

Texas Ethics Commission, P.O. Box 12070, Austin, Texas 78711 www.ethics.state.tx.us (512) 463-5800 • TDD (800) 735-2989 Promoting Public Confidence in Government

#### **TABLE OF CONTENTS**

| CHAPTER 6. ORGANIZATION AND ADMINISTRATION1               | 1      |
|---|--------|
| Subchapter A. GENERAL RULES                               | 1      |
| §6.1. Definitions1  | 1      |
| §6.3. Purpose and Construction of Rules.                  | 3      |
| §6.5. Authority To Adopt Rules                            | 3      |
| §6.7. Actions That Require Six Votes                      | 3      |
| §6.9. Computation of Time                                 |        |
| Subchapter B. OFFICERS AND EMPLOYEES OF THE COMMISSION    | 4      |
| §6.21. Officers of the Commission.                        | 4      |
| §6.23. Commission Staff                                   |        |
| §6.25. Appointment and Operation of Advisory Committees   | 5      |
| Subchapter C. COMMISSION MEETINGS                         |        |
| §6.31. Quorum   | 5      |
| §6.33. Frequency of Meetings                              |        |
| §6.35. Called Meetings                                    |        |
| §6.37. Open Meetings                                      |        |
| §6.39. Meeting Agenda                                     |        |
| §6.41. Public Hearing and Participation                   |        |
| §6.43. Speakers Addressing the Commission                 |        |
| §6.45. Order and Conduct of Commission Meeting.           |        |
| §6.47. Tape Recording of Meeting; Minutes.                |        |
| Subchapter D. RULEMAKING PROCEDURES                       |        |
| §6.61. Comments on Proposed Rules.                        |        |
| §6.63. Public Hearings on Proposed Rules.                 | 8      |
| Subchapter E. AGENCY FEES AND CHARGES                     |        |
| §6.81. Charges; Payment of Money; Refunds                 |        |
| §6.83. Copying Charges                                    |        |
| §6.87. Waiver of Fees for Copies or Publications.         |        |
| CHAPTER 8. ADVISORY OPINIONS                              |        |
| §8.1. Definitions.  |        |
| §8.3. Subject of an Advisory Opinion.                     |        |
| §8.5. Persons Eligible To Receive an Advisory Opinion     |        |
| §8.7. Request for an Advisory Opinion                     |        |
| §8.9. Commission Initiated Opinion                        |        |
| §8.11. Review and Processing of a Request                 |        |
| §8.15. Publication in Texas Register; Comments            |        |
| §8.17. Request Answered by Written Response               |        |
| §8.17. Request Answered by Written Response               |        |
| §8.21. Compilation of Advisory Opinions.                  | כ      |
| CHAPTER 10. ETHICS TRAINING PROGRAMS                      | 2<br>2 |
| \$10.1. Training Programs                                 |        |
| \$10.3. Tuition Charges to Attendees of Training Programs |        |
| CHAPTER 12. SWORN COMPLAINTS                              |        |
| Subchapter A. GENERAL PROVISIONS AND PROCEDURES           |        |

| §12.5. Deadline for Filing a Complaint.  | 14 |
|--|----|
| §12.6. File Date for Purposes of Commission Response Deadline                                  | 14 |
| §12.7. Confidentiality   |    |
| §12.9. Compliance with Open Meetings Law and Open Records Law.                                 | 14 |
| §12.11. Delegation to Executive Director.  |    |
| §12.13. Representation by Counsel  |    |
| §12.15. Appearance of Complainant at Hearing.  |    |
| §12.19. Agreements to be in Writing.   |    |
| §12.21. Notice.  |    |
| §12.23. Hearing in Respondent's Absence.   |    |
| §12.25. Waiver of Hearing  |    |
| §12.27. Deadline Extension.  |    |
| \$12.28. Production of Documents During Preliminary Review.                                    |    |
| §12.29. Subpoenas Issued by Commission.  |    |
| \$12.30. Subpoenas Issued by Counsel for the Respondent  |    |
| §12.30. Subpoends issued by Counsel for the Respondent   |    |
| \$12.33. Sanctioning Authority.  |    |
| §12.33. Salctioning Autority.<br>§12.34. Agreed Orders.  |    |
| §12.34. Agreed Orders. §12.35. Frivolous Complaint   |    |
|  |    |
| §12.36. Assessment of Civil Penalty Subchapter B. FILING AND INITIAL PROCESSING OF A COMPLAINT |    |
| \$12.51 Non-Complying Complaint  |    |
|  |    |
| \$12.52. Response to Notice of Complaint.  |    |
| \$12.53. Commission Initiated Complaint.   |    |
| \$12.59. Description of Violation.   |    |
| \$12.61. Statement of Facts.   |    |
| \$12.67. Copies and Documents Provided by the Commission                                       |    |
| Subchapter C. INVESTIGATION AND PRELIMINARY REVIEW   |    |
| \$12.81. Technical, Clerical, or <i>De Minimis</i> Violations                                  |    |
| §12.83. Preliminary Review.  | 27 |
| Subchapter D. PRELIMINARY REVIEW HEARING   |    |
| \$12.84. Notice of Preliminary Review Hearing.   |    |
| \$12.85. Preliminary Review Hearing.   |    |
| \$12.86. Motions for Continuance   |    |
| \$12.87. Resolution of Preliminary Review Hearing.   |    |
| Subchapter E. FORMAL HEARING   |    |
| Division 1. General Procedures   |    |
| \$12.101. Application and Construction.  |    |
| \$12.102. Order of Formal Hearing.   |    |
| \$12.103. Notice of Formal Hearing.  |    |
| \$12.117. Formal Hearing: Venue  |    |
| \$12.119. Resolution after a Formal Hearing.   |    |
| Division 2. Scheduling, Filing, and Service  |    |
| \$12.121. Prehearing Conferences.  |    |
| §12.123. Scheduling Orders   |    |
| \$12.125. Filing of Documents  |    |
| \$12.127. Service of Documents.  |    |
| Division 3. Powers and Duties of Commission and Presiding Officer                              |    |

| §12.131. Powers and Duties of the Presiding Officer                                | 34 |
|--|----|
| §12.133. Orders from the Commission.   | 35 |
| Division 5. Pleadings and Motions  | 35 |
| §12.151. Required Form of Pleadings  | 35 |
| §12.153. Motions, Generally  | 36 |
| §12.155. Motions for Continuance and to Extend Time                                | 37 |
| Division 6. Hearings and Prehearing Conferences                                    | 38 |
| §12.161. Time Allotted to Parties.   | 38 |
| §12.163. Presentation of Evidence  | 38 |
| §12.165. Rules of Evidence.  | 39 |
| §12.167. Numbering of Exhibits   | 39 |
| Division 7. Disposition of Formal Hearing  |    |
| §12.171. Standard of Proof   | 39 |
| §12.173. Default Proceedings.  | 39 |
| §12.174. Summary Disposition   | 40 |
| §12.175. Resolution of Formal Hearing  | 42 |
| Chapter 16. FACIAL COMPLIANCE REVIEWS AND AUDITS                                   | 43 |
| §16.1. Definitions   |    |
| §16.2. Random Selection  | 43 |
| §16.3. Corrected or Amended Report Filed During a Facial Compliance Review; Late   |    |
| \$16.4. Additional Documents and Information Submitted in Response to a Facial     | 43 |
| Compliance Review; Timeliness  | 11 |
| §16.5. Commission Initiated Preliminary Review or Audit Resulting from a Facial    |    |
| Compliance Review.   | 11 |
| §16.6. Notice of Audit of Report.  |    |
| §16.7. Supporting Documentation in Response to Audit; Timeliness                   |    |
| §16.8. Complete Audit Report.  |    |
| §16.9. Representation by Attorney  |    |
| §16.10. Extension of Deadlines.  |    |
| §16.11. Waiver of Delivery by Certified Mail.                                      |    |
| §16.12. Facial Review of Total Amount of Political Contributions Maintained        |    |
| CHAPTER 18. GENERAL RULES CONCERNING REPORTS                                       |    |
| §18.1. Forms   |    |
| §18.3. Provision of Forms by Local Filing Authority                                |    |
| §18.5. Specification of Office.  |    |
| §18.7. Timely Reports and Complete Reports   |    |
| §18.9. Corrected/Amended Reports.  |    |
| §18.10. Guidelines for Substantial Compliance for a Corrected/Amended 8-day Pre-el |    |
| Report.  |    |
| §18.11 Guidelines for Waiver or Reduction of a Late Fine for a Corrected/Amended 8 |    |
| Pre-election Report  | •  |
| §18.13. Fine for a Late Report   |    |
| §18.15. Additional Fine  |    |
| §18.17. Report Must be Filed   |    |
| §18.19. Affidavit of Timely Filing   |    |
| §18.21. Jurisdiction to Consider Waiver Request.                                   |    |
| §18.23. Administrative Waiver of Statutory Civil Penalties                         |    |
| 0 = 0 = 0  |    |

| §18.24. General Guidelines for Other Administrative Waiver or Reduction of Statutor                             | y Civil |
|---|---------|
| Penalties   |         |
| §18.25. Administrative Waiver or Reduction of Certain Statutory Civil Penalties                                 |         |
| §18.26. Administrative Waiver or Reduction of Other Statutory Civil Penalties in Exc                            | ess of  |
| \$500   |         |
| §18.31. Adjustments to Reporting Thresholds   |         |
| APTER 20. REPORTING POLITICAL CONTRIBUTIONS AND EXPENDITUR  |         |
| ubchapter A. GENERAL RULES  |         |
| §20.1. Definitions  |         |
| §20.3. Reports Filed with the Commission  |         |
| §20.5. Reports Filed with a County Filing Authority.  |         |
| §20.7. Reports Filed with Other Local Filing Authority  | 74      |
| §20.9. Filing Option for Certain Specific-Purpose Committees  | 74      |
| §20.11. Federal Candidates and Officeholders.   | 74      |
| §20.13. Out-of-State Committees.  | 74      |
| §20.15. Change of Address.  | 75      |
| §20.16. Notices by Electronic Mail  | 75      |
| §20.18. Recordkeeping Required  | 76      |
| §20.19. Reports Must Be Filed on Official Forms   | 76      |
| \$20.20. Timeliness of Action by Electronic Filing.   |         |
| §20.21. Due Dates on Holidays and Weekends.   |         |
| §20.23. Timeliness of Action by Mail.   |         |
| §20.29. Information About Out-of-State Committees.  |         |
| \$20.33. Termination of Campaign Treasurer Appointment By Commission  |         |
| §20.35. Notice of Proposed Termination of Campaign Treasurer Appointment  |         |
| bchapter B. GENERAL REPORTING RULES   |         |
| §20.50. Total Political Contributions Maintained.   |         |
| §20.51. Value of In-Kind Contribution   |         |
| §20.52. Description of In-Kind Contribution for Travel  |         |
| §20.53. Disclosure of True Source of Contribution or Expenditure.   |         |
| §20.54. Reporting a Pledge of a Contribution  |         |
| §20.55. Time of Accepting Contribution.   |         |
| \$20.56. Expenditures to Vendors.   |         |
| \$20.57. Time of Making Expenditure.  |         |
| \$20.58. Disclosure of Political Expenditure.   |         |
| \$20.59. Reporting Expenditure by Credit Card   |         |
| \$20.60. Reporting Political Expenditures for Processing Fees.  |         |
| \$20.61. Purpose of Expenditure.  |         |
| §20.62. Reporting Staff Reimbursement   |         |
| \$20.63. Reporting the Use and Reimbursement of Personal Funds  |         |
| \$20.64. Reporting the Forgiveness of a Loan or Settlement of a Debt  |         |
| \$20.65. Reporting No Activity  |         |
| \$20.65. Reporting No Activity  |         |
| §20.67. Reporting after the Death or Incapacity of a Filer  |         |
| g20.67. Reporting after the Death of Incapacity of a Filer. ubchapter C. REPORTING REQUIREMENTS FOR A CANDIDATE |         |
| \$20.201. Required Appointment of Campaign Treasurer.   |         |
| §20.201. Required Appointment of Campaign Treasurer   |         |
| §20.205. Contents of Candidate's Campaign Treasurer Appointment.  |         |
| x20.200. Contents of Canutuate S Campaign Heasure Appointment.  |         |

| §20.206. Transfer of Campaign Treasurer Appointment.                                | 90    |
|---|-------|
| §20.207. Termination of Campaign Treasurer Appointment                              |       |
| \$20.209. Reporting Obligations Imposed on Candidate, Not Campaign Treasurer        |       |
| §20.211. Semiannual Reports.  |       |
| §20.213. Pre-election Reports.  |       |
| §20.215. Runoff Report.   |       |
| §20.217. Modified Reporting.  |       |
| §20.219. Content of Candidate's Sworn Report of Contributions and Expenditures.     |       |
| §20.220. Additional Disclosure for the Texas Comptroller of Public Accounts         |       |
| \$20.221. Special Pre-Election Report by Certain Candidates                         |       |
| §20.223. Form and Contents of Special Pre-Election Report.                          |       |
| §20.225. Special Session Reports.   |       |
| §20.227. Contents of Special Session Report.  | 100   |
| §20.229. Final Report   |       |
| §20.231. Contents of Final Report.  | 101   |
| §20.233. Annual Report of Unexpended Contributions                                  |       |
| §20.235. Contents of Annual Report  | 103   |
| §20.237. Final Disposition of Unexpended Contributions                              | 103   |
| §20.239. Report of Final Disposition of Unexpended Contributions.                   | 104   |
| §20.241. Contents of Report of Final Disposition of Unexpended Contributions        |       |
| §20.243. Contribution of Unexpended Political Contributions to Candidate or Politic | cal   |
| Committee   |       |
| Subchapter D. REPORTING REQUIREMENTS FOR AN OFFICEHOLDER W                          | HO    |
| DOES NOT HAVE A CAMPAIGN TREASURER APPOINTMENT ON FILE                              | 106   |
| §20.271. Officeholders Covered.   |       |
| §20.273. Semiannual Reports of Contributions and Expenditures                       |       |
| §20.275. Exception from Filing Requirement for Certain Local Officeholders          | 107   |
| §20.277. Appointment by Officeholder of Campaign Treasurer.                         |       |
| §20.279. Contents of Officeholder's Sworn Report of Contributions and Expenditure   | es107 |
| §20.281. Special Session Report by Certain Officeholders.                           | 111   |
| \$20.283. Contents of Special Session Report  |       |
| §20.285. Annual Report of Unexpended Contributions by Former Officeholder           |       |
| §20.287. Contents of Annual Report  |       |
| \$20.289. Disposition of Unexpended Contributions                                   |       |
| §20.291. Report of Final Disposition of Unexpended Contributions                    |       |
| §20.293. Contents of Report of Final Disposition of Unexpended Contributions        |       |
| §20.295. Contribution of Unexpended Political Contributions to Candidate or Politic |       |
| Committee   |       |
| Subchapter E. REPORTS BY A SPECIFIC-PURPOSE COMMITTEE                               |       |
| \$20.301. Thresholds for Campaign Treasurer Appointment.                            |       |
| \$20.303. Appointment of Campaign Treasurer   |       |
| \$20.307. Name of Specific-Purpose Committee  |       |
| §20.309. Contents of Specific-Purpose Committee Campaign Treasurer Appointmen       |       |
| \$20.311. Updating Certain Information on the Campaign Treasurer Appointment        |       |
| \$20.313. Converting to a General-Purpose Committee.                                |       |
| \$20.315. Termination of Campaign Treasurer Appointment                             |       |
| \$20.317. Termination Report.   |       |
| §20.319. Notice to Candidate or Officeholder.                                       | 120   |

| 20       \$20.323. Semiannual Reports.       121         \$20.325. Pre-election Reports.       121         \$20.327. Runoff Report.       122         \$20.331. Contents of Specific-Purpose Committee Sworn Report of Contributions and<br>Expenditures.       124         \$20.331. Contents of Special Pre-Election Report by Certain Specific-Purpose Committees.       124         \$20.333. Special Pre-Election Report by Certain Candidates.       129         \$20.337. Special Session Reports by Specific-Purpose Committees.       129         \$20.337. Special Session Report.       130         \$20.338. Contents of the Special Session Report.       130         \$20.341. Dissolution Report.       130         \$20.342. Contents of Dissolution Report.       131         Subchapter F. REPORTING REQUIREMENT FOR A GENERAL PURPOSE       131         COMMITTEE       131         \$20.401. Thresholds for Appointment of Campaign Treasurer by a General-Purpose       131         \$20.403. Reporting Requirements for Certain General-Purpose Committee.       131         \$20.4043. Reporting Requirements for Certain General-Purpose Committee.       132         \$20.4040. Appointing an Assistant Campaign Treasurer Appointment.       134         \$20.410. Ontents of General-Purpose Committee Campaign Treasurer Appointment.       134         \$20.410. Appointing an Assistant Campaign Treasurer Appointment   | §20.321. Involvement in More Than One Election by Certain Specific-Purpose Comm   | ittees. |
|--|---|---------|
| §20.325. Pre-election Reports.       121         §20.327. Runoff Report.       122         §20.329. Modified Reporting.       123         §20.331. Contents of Specific-Purpose Committee Sworn Report of Contributions and<br>Expenditures.       124         §20.333. Special Pre-Election Report by Certain Specific-Purpose Committees.       128         §20.335. Form and Contents of Special Pre-Election Report by a Specific-Purpose<br>Committee Supporting or Opposing Certain Candidates.       129         §20.337. Special Session Report.       130         §20.334. Contents of Dissolution Report.       130         §20.335. Contents of Dissolution Report.       130         §20.343. Contents of Dissolution Report.       131         Subchapter F. REPORTING REQUIREMENT FOR A GENERAL PURPOSE       131         COMMITTEE       131         §20.401. Thresholds for Appointment of Campaign Treasurer by a General-Purpose       131         Subchapter F. REPORTING REQUIREMENT FOR A GENERAL PURPOSE       132         Committee.       132         §20.403. Reporting Requirements for Certain General-Purpose Committees.       132         §20.4045. Campaign Treasurer Appointment for a General-Purpose Committee.       133         §20.4040. Name of General-Purpose Committee Campaign Treasurer Appointment.       134         §20.413. Updating Information on the Campaign Treasurer Appointment.   |   | 120     |
| §20.327. Runoff Report.       122         §20.320. Modified Reporting       123         §20.311. Contents of Specific-Purpose Committee Sworn Report of Contributions and<br>Expenditures.       124         §20.333. Special Pre-Election Report by Certain Specific-Purpose Committees.       128         §20.337. Special Pre-Election Report by Specific-Purpose Committees.       129         §20.337. Special Session Reports by Specific-Purpose Committees.       129         §20.333. Contents of the Special Session Report.       130         §20.341. Dissolution Report.       130         §20.343. Contents of Dissolution Report.       131         Subchapter F. REPORTING REQUIREMENT FOR A GENERAL PURPOSE       131         §20.401. Thresholds for Appointment of Campaign Treasurer by a General-Purpose       131         §20.403. Reporting Requirements for Certain General-Purpose Committee.       132         §20.407. Appointing an Assistant Campaign Treasurer.       132         §20.408. Name of General-Purpose Committee       133         §20.411. Contents of General-Purpose Committee       133         §20.412. Notice to Candidate or Officeholder.       134         §20.413. Updating Information on the Campaign Treasurer Appointment.       134         §20.414. Notice to Candidate or Officeholder.       136         §20.415. Termination Report.       137  | §20.323. Semiannual Reports.  | 121     |
| §20.329. Modified Reporting.       [23         §20.331. Contents of Specific-Purpose Committee Sworn Report of Contributions and       [24         §20.333. Special Pre-Election Report by Certain Specific-Purpose Committees.       [28         §20.335. Form and Contents of Special Pre-Election Report by a Specific-Purpose Committee Supporting or Opposing Certain Candidates.       [29         §20.337. Special Session Reports by Specific-Purpose Committees.       [29         §20.337. Special Session Report.       [30         §20.337. Special Session Report.       [30         §20.341. Dissolution Report.       [31]         Subchapter F. REPORTING REQUIREMENT FOR A GENERAL PURPOSE       [31]         COMMITTEE       [31]         §20.403. Reporting Requirements for Certain General-Purpose Committees.       [32]         §20.404. Reporting Requirements for Certain General-Purpose Political Committee.       [32]         §20.407. Appointing an Assistant Campaign Treasurer.       [33]         §20.408. Name of General-Purpose Committee.       [33]         §20.411. Contents of General-Purpose Committee Campaign Treasurer Appointment.       [34]         §20.417. Termination of Campaign Treasurer Appointment.       [34]         §20.421. Notice to Candidate or Officeholder.       [36]         §20.422. Notice of General-Purpose Committee.       [36]         §20.423. Sentiannual Rep  | §20.325. Pre-election Reports   | 121     |
| §20.331. Contents of Špecific-Purpose Committee Sworn Report of Contributions and<br>Expenditures       124         §20.333. Special Pre-Election Report by Certain Specific-Purpose Committees       129         §20.335. Form and Contents of Special Pre-Election Report by a Specific-Purpose<br>Committee Supporting or Opposing Certain Candidates       129         §20.337. Special Session Reports by Specific-Purpose Committees       130         §20.330. Contents of the Special Session Report.       130         §20.341. Dissolution Report.       130         §20.342. Contents of Dissolution Report.       131         Subchapter F. REPORTING REQUIREMENT FOR A GENERAL PURPOSE       131         ©COMMITTEE       131         §20.401. Thresholds for Appointment of Campaign Treasurer by a General-Purpose Committee       132         §20.402. Campaign Treasurer Appointment for a General-Purpose Committee       132         §20.403. Reporting an Assistant Campaign Treasurer.       132         §20.4040. Name of General-Purpose Committee       133         §20.413. Updating Information on the Campaign Treasurer Appointment.       134         §20.415. Termination of Campaign Treasurer Appointment.       134         §20.416. Converting to a Specific-Purpose Committee       136         §20.417. Termination of General-Purpose Committee.       136         §20.421. Notice to Candidate or Officeholder.       136  |   |         |
| Expenditures.       124         \$20.333. Special Pre-Election Report by Certain Specific-Purpose Committees.       128         \$20.335. Form and Contents of Special Pre-Election Report by a Specific-Purpose Committee Supporting or Opposing Certain Candidates.       129         \$20.337. Special Session Reports by Specific-Purpose Committees.       129         \$20.339. Contents of the Special Session Report.       130         \$20.341. Dissolution Report.       131         Subchapter F. REPORTING REQUIREMENT FOR A GENERAL PURPOSE       131         \$20.401. Thresholds for Appointment of Campaign Treasurer by a General-Purpose Committee.       131         \$20.403. Reporting Requirements for Certain General-Purpose Committees.       132         \$20.405. Campaign Treasurer Appointment for a General-Purpose Political Committee.       132         \$20.405. Campaign Treasurer Appointment for a General-Purpose Political Committee.       133         \$20.410. Lottents of General-Purpose Committee Campaign Treasurer Appointment.       134         \$20.413. Updating Information on the Campaign Treasurer Appointment.       134         \$20.4141. Contents of General-Purpose Committee.       136         \$20.412. Notice to Candidate or Officeholder.       136         \$20.413. Updating Information on the Campaign Treasurer Appointment.       134         \$20.4141. Contents of General-Purpose Committee.       136   | 1 0   |         |
| \$20.333. Special Pre-Election Report by Certain Specific-Purpose Committees       128         \$20.335. Form and Contents of Special Pre-Election Report by a Specific-Purpose       129         \$20.337. Special Session Reports by Specific-Purpose Committees       129         \$20.337. Special Session Reports by Specific-Purpose Committees       130         \$20.337. Special Session Report.       130         \$20.341. Dissolution Report.       131         Subchapter F. REPORTING REQUIREMENT FOR A GENERAL PURPOSE       131         \$20.401. Thresholds for Appointment of Campaign Treasurer by a General-Purpose       131         \$20.403. Reporting Requirements for Certain General-Purpose Committees.       132         \$20.405. Campaign Treasurer Appointment for a General-Purpose Political Committee.       132         \$20.409. Name of General-Purpose Committee       133         \$20.409. Name of General-Purpose Committee       133         \$20.411. Contents of General-Purpose Committee       133         \$20.413. Updating Information on the Campaign Treasurer Appointment.       134         \$20.421. Notice to Candidate or Officeholder.       136         \$20.422. Pre-election Reports.       137         \$20.423. Semiannual Reports.       137         \$20.424. Notice to Gandiate or Officeholder.       136         \$20.425. Pre-election Reports.       137      <   | §20.331. Contents of Specific-Purpose Committee Sworn Report of Contributions and | •       |
| §20.335. Form and Contents of Special Pre-Election Report by a Specific-Purpose       129         §20.337. Special Session Reports by Specific-Purpose Committees       129         §20.339. Contents of the Special Session Report.       130         §20.341. Dissolution Report.       130         §20.342. Contents of Dissolution Report.       131         Subchapter F. REPORTING REQUIREMENT FOR A GENERAL PURPOSE       131         §20.401. Thresholds for Appointment of Campaign Treasurer by a General-Purpose       131         §20.403. Reporting Requirements for Certain General-Purpose Committee.       132         §20.405. Campaign Treasurer Appointment for a General-Purpose Political Committee.       132         §20.405. Campaign Treasurer Appointment for a General-Purpose Political Committee.       133         §20.405. Campaign Treasurer Appointment on the Campaign Treasurer Appointment.       134         §20.405. Contents of General-Purpose Committee       133         §20.410. Updating Information on the Campaign Treasurer Appointment.       134         §20.411. Updating Information on the Campaign Treasurer Appointment.       134         §20.421. Notice to Candidate or Officeholder.       136         §20.422. Notice to Candidate or Officeholder.       136         §20.423. Semiannual Reports.       137         §20.424. Nuchters de General-Purpose Committee Sworn Report of Contributions and Expenditure   |   |         |
| Committee Supporting or Opposing Certain Candidates.       129         \$20.337. Special Session Reports by Specific-Purpose Committees.       129         \$20.339. Contents of the Special Session Report.       130         \$20.341. Dissolution Report.       131         Subchapter F. REPORTING REQUIREMENT FOR A GENERAL PURPOSE       131         COMMITTEE       131         \$20.401. Thresholds for Appointment of Campaign Treasurer by a General-Purpose       131         \$20.403. Reporting Requirements for Certain General-Purpose Committees.       132         \$20.405. Campaign Treasurer Appointment for a General-Purpose Committee.       133         \$20.407. Appointing an Assistant Campaign Treasurer.       132         \$20.408. Name of General-Purpose Committee       133         \$20.411. Contents of General-Purpose Committee Campaign Treasurer Appointment.       134         \$20.413. Updating Information on the Campaign Treasurer Appointment.       134         \$20.419. Converting to a Specific-Purpose Committee.       136         \$20.421. Notice to Candidate or Officeholder.       136         \$20.422. Semiannual Reports.       137         \$20.423. Semiannual Reports.       137         \$20.4242. Notice to General-Purpose Committee Sworn Report of Contributions and Expenditures.       140         \$20.423. Contents of General-Purpose Committee Sworn Report of Contributions a  |   | 128     |
| §20.337. Special Session Reports by Specific-Purpose Committees.       129         §20.339. Contents of the Special Session Report.       130         §20.341. Dissolution Report.       131         Subchapter F. REPORTING REQUIREMENT FOR A GENERAL PURPOSE       131         Subchapter F. REPORTING REQUIREMENT FOR A GENERAL PURPOSE       131         \$20.401. Thresholds for Appointment of Campaign Treasurer by a General-Purpose       131         \$20.403. Reporting Requirements for Certain General-Purpose Committees.       132         \$20.405. Campaign Treasurer Appointment for a General-Purpose Committee.       132         \$20.407. Appointing an Assistant Campaign Treasurer       132         \$20.409. Name of General-Purpose Committee       133         \$20.411. Contents of General-Purpose Committee Campaign Treasurer Appointment.       134         \$20.413. Updating Information on the Campaign Treasurer Appointment.       134         \$20.414. Contents of General-Purpose Committee       136         \$20.415. Termination of Campaign Treasurer Appointment.       134         \$20.416. Converting to a Specific-Purpose Committee       136         \$20.421. Notice to Candidate or Officeholder.       136         \$20.422. Notice to Candidate or Officeholder.       138         \$20.423. Reminanual Reports.       137         \$20.424. Nothy Reporting.       138  |   |         |
| §20.339. Contents of the Special Session Report.       130         §20.341. Dissolution Report.       131         Subchapter F. REPORTING REQUIREMENT FOR A GENERAL PURPOSE       131         COMMITTEE       131         §20.401. Thresholds for Appointment of Campaign Treasurer by a General-Purpose       131         §20.402. Campaign Treasurer Appointment for a General-Purpose Committee.       132         §20.405. Campaign Treasurer Appointment for a General-Purpose Political Committee.       132         §20.405. Campaign Treasurer Appointment for a General-Purpose Political Committee.       133         §20.405. Campaign Treasurer Appointment for a General-Purpose Political Committee.       133         §20.405. Campaign Information on the Campaign Treasurer Appointment.       134         §20.411. Undenting Information on the Campaign Treasurer Appointment.       134         §20.415. Termination of Campaign Treasurer Appointment.       134         §20.416. Converting to a Specific-Purpose Committee.       136         §20.421. Notice to Candidate or Officeholder.       136         §20.422. Pre-election Reports.       137         §20.423. Gontents of General-Purpose Committee Sworn Report of Contributions and       144         §20.424. Notice to Candidate or Officeholder.       138         §20.423. Contents of General-Purpose Committee Sworn Report of Contributions and       144  |   |         |
| §20.341. Dissolution Report.       130         §20.343. Contents of Dissolution Report.       131         Subchapter F. REPORTING REQUIREMENT FOR A GENERAL PURPOSE       131         §20.401. Thresholds for Appointment of Campaign Treasurer by a General-Purpose       131         §20.403. Reporting Requirements for Certain General-Purpose Committees.       132         §20.405. Campaign Treasurer Appointment for a General-Purpose Political Committee.       132         §20.407. Appointing an Assistant Campaign Treasurer.       133         §20.407. Appointing an Assistant Campaign Treasurer.       133         §20.407. Appointing an Assistant Campaign Treasurer Appointment.       133         §20.407. Appointing of General-Purpose Committee       133         §20.411. Contents of General-Purpose Committee       133         §20.412. Termination of Campaign Treasurer Appointment.       134         §20.413. Updating Information on the Campaign Treasurer Appointment.       134         §20.417. Termination Report.       135         §20.421. Notice to Candidate or Officeholder.       136         §20.422. Nunoff Report.       137         §20.423. Semiannual Reports.       137         §20.424. Nunoff Report.       138         §20.425. Special Pre-Election Reports by Certain General-Purpose Committees.       144         §20.433. Monthly Reportin   |   |         |
| §20.343. Contents of Dissolution Report.       131         Subchapter F. REPORTING REQUIREMENT FOR A GENERAL PURPOSE       131         §20.401. Thresholds for Appointment of Campaign Treasurer by a General-Purpose       131         §20.403. Reporting Requirements for Certain General-Purpose Committees       132         §20.405. Campaign Treasurer Appointment for a General-Purpose Political Committee.       132         §20.405. Campaign Treasurer Appointment for a General-Purpose Political Committee.       133         §20.409. Name of General-Purpose Committee       133         §20.411. Contents of General-Purpose Committee Campaign Treasurer Appointment.       134         §20.413. Updating Information on the Campaign Treasurer Appointment.       134         §20.410. Converting to a Specific-Purpose Committee       136         §20.411. Contents of Specific-Purpose Committee.       136         §20.412. Notice to Candidate or Officeholder.       136         §20.423. Semiannual Report.       137         §20.4242. Nuoff Report.       138         §20.433. Contents of General-Purpose Committee Sworn Report of Contributions and Expenditures.       140         §20.433. Contents of General-Purpose Committee Sworn Report of Contributions and Expenditures.       144         §20.433. Contents of General-Purpose Committee Sworn Report of Contributions and Expenditures.       144         §20.434. Alternate Repor   |   |         |
| Subchapter F. REPORTING REQUIREMENT FOR A GENERAL PURPOSE       131         \$20.401. Thresholds for Appointment of Campaign Treasurer by a General-Purpose       131         \$20.401. Thresholds for Appointment of Cartain General-Purpose Committee.       131         \$20.403. Reporting Requirements for Certain General-Purpose Committees.       132         \$20.405. Campaign Treasurer Appointment for a General-Purpose Political Committee.       132         \$20.407. Appointing an Assistant Campaign Treasurer.       132         \$20.409. Name of General-Purpose Committee       133         \$20.411. Contents of General-Purpose Committee Campaign Treasurer Appointment.       134         \$20.413. Updating Information on the Campaign Treasurer Appointment.       134         \$20.417. Termination Report.       135         \$20.418. Notice to Candidate or Officeholder.       136         \$20.421. Notice to Candidate or Officeholder.       136         \$20.422. Pre-election Reports.       137         \$20.423. Semiannual Report.       138         \$20.424. Nothly Reporting.       139         \$20.425. Pre-election Reports.       137         \$20.426. Option To File Monthly.       138         \$20.427. Runoff Report.       138         \$20.433. Contents of General-Purpose Committee Sworn Report of Contributions and       140         \$20.434. Alternate Rep  |   |         |
| COMMITTEE       131         \$20.401. Thresholds for Appointment of Campaign Treasurer by a General-Purpose       131         Committee       131         \$20.403. Reporting Requirements for Certain General-Purpose Committees       132         \$20.405. Campaign Treasurer Appointment for a General-Purpose Political Committee.       132         \$20.405. Campaign Treasurer Appointment for a General-Purpose Political Committee.       133         \$20.407. Appointing an Assistant Campaign Treasurer.       133         \$20.409. Name of General-Purpose Committee       133         \$20.411. Contents of General-Purpose Committee Campaign Treasurer Appointment.       134         \$20.413. Updating Information on the Campaign Treasurer Appointment.       134         \$20.417. Termination of Campaign Treasurer Appointment.       136         \$20.419. Converting to a Specific-Purpose Committee       136         \$20.421. Notice to Candidate or Officeholder.       136         \$20.422. Semiannual Reports.       137         \$20.423. Semiannual Reports.       137         \$20.4242. Notice to General-Purpose Committee Sworn Report of Contributions and       138         \$20.423. Contents of General-Purpose Committee Sworn Report of Contributions and       140         \$20.433. Contents of General-Purpose Committee Sworn Report of Contributions and       144         \$20.434. Alternate Reportin   |   | 131     |
| §20.401. Thresholds for Appointment of Campaign Treasurer by a General-Purpose       131         §20.403. Reporting Requirements for Certain General-Purpose Committees.       132         §20.405. Campaign Treasurer Appointment for a General-Purpose Political Committee.       132         §20.407. Appointing an Assistant Campaign Treasurer.       132         §20.409. Name of General-Purpose Committee.       133         §20.411. Contents of General-Purpose Committee Campaign Treasurer Appointment.       134         §20.413. Updating Information on the Campaign Treasurer Appointment.       134         §20.410. Termination of Campaign Treasurer Appointment.       134         §20.417. Termination Report.       135         §20.421. Notice to Candidate or Officeholder.       136         §20.423. Semiannual Reports.       137         §20.4247. Runoff Report.       138         §20.431. Monthly Reporting.       139         §20.432. Option To File Monthly.       138         §20.433. Contents of General-Purpose Committee Sworn Report of Contributions and       144         §20.434. Alternate Reporting Requirements for General-Purpose Committees.       144         §20.435. Special Pre-Election Report.       144         §20.436. Pre-telection Reports by Certain General-Purpose Committees.       144         §20.437. Form and Contents of Special Pre-Election Report. <td< td=""><td></td><td></td></td<>   |   |         |
| Committee131\$20.403. Reporting Requirements for Certain General-Purpose Committees132\$20.405. Campaign Treasurer Appointment for a General-Purpose Political Committee.132\$20.407. Appointing an Assistant Campaign Treasurer132\$20.409. Name of General-Purpose Committee133\$20.411. Contents of General-Purpose Committee133\$20.412. Updating Information on the Campaign Treasurer Appointment134\$20.413. Updating Information on the Campaign Treasurer Appointment134\$20.414. Converting to a Specific-Purpose Committee136\$20.421. Notice to Candidate or Officeholder136\$20.423. Semiannual Reports137\$20.424. Notice to Candidate or Officeholder138\$20.425. Pre-election Reports137\$20.426. Pre-election Reports137\$20.427. Runoff Report138\$20.431. Monthly Reporting138\$20.432. Ontents of General-Purpose Committee Sworn Report of Contributions and<br>Expenditures140\$20.433. Contents of General-Purpose Committee Sworn Report of Contributions and<br>Expenditures144\$20.434. Alternate Reporting Requirements for General-Purpose Committees144\$20.437. Form and Contents of Special Pre-Election Report145\$20.439. Dissolution Report146\$20.431. Contents of Dissolution Report146\$20.435. Special Pre-Election Reports by Certain General-Purpose Committees144\$20.437. Form and Contents of Special Pre-Election Report146\$20.439. Dissolution Report146\$20.431. Dotesting Distolut   |   | 131     |
| <ul> <li>\$20.403. Reporting Requirements for Certain General-Purpose Committees.</li> <li>\$20.405. Campaign Treasurer Appointment for a General-Purpose Political Committee.</li> <li>\$20.407. Appointing an Assistant Campaign Treasurer.</li> <li>\$20.409. Name of General-Purpose Committee.</li> <li>\$20.409. Name of General-Purpose Committee Campaign Treasurer Appointment.</li> <li>\$20.413. Updating Information on the Campaign Treasurer Appointment.</li> <li>\$20.413. Updating Information on the Campaign Treasurer Appointment.</li> <li>\$20.414. Termination of Campaign Treasurer Appointment.</li> <li>\$20.417. Termination Report.</li> <li>\$20.419. Converting to a Specific-Purpose Committee.</li> <li>\$20.421. Notice to Candidate or Officeholder.</li> <li>\$20.423. Semiannual Reports.</li> <li>\$20.427. Runoff Report.</li> <li>\$20.429. Option To File Monthly.</li> <li>\$20.431. Monthly Reporting.</li> <li>\$20.433. Contents of General-Purpose Committee Sworn Report of Contributions and Expenditures.</li> <li>\$20.434. Alternate Reporting Requirements for General-Purpose Committees.</li> <li>\$20.437. Form and Contents of Special Pre-Election Report.</li> <li>\$20.439. Dissolution Report.</li> <li>\$20.431. Contents of Dissolution Report.</li> <li>\$20.431. Contents of Special Pre-Election Report.</li> <li>\$20.432. Special Pre-Election Reports by Certain General-Purpose Committees.</li> <li>\$20.433. Contents of Special Pre-Election Report.</li> <li>\$20.434. Alternate Reporting Requirements for General-Purpose Committees.</li> <li>\$20.444. Special Pre-Election Report.</li> <li>\$20.445. Special Pre-Election Report.<!--</td--><td></td><td></td></li></ul> |   |         |
| <ul> <li>\$20.405. Campaign Treasurer Appointment for a General-Purpose Political Committee132</li> <li>\$20.407. Appointing an Assistant Campaign Treasurer</li></ul>   |   |         |
| \$20.407. Appointing an Assistant Campaign Treasurer.132\$20.409. Name of General-Purpose Committee.133\$20.411. Contents of General-Purpose Committee Campaign Treasurer Appointment.133\$20.413. Updating Information on the Campaign Treasurer Appointment.134\$20.415. Termination of Campaign Treasurer Appointment.134\$20.417. Termination Report.135\$20.419. Converting to a Specific-Purpose Committee.136\$20.421. Notice to Candidate or Officeholder.136\$20.422. Semiannual Reports.137\$20.425. Pre-election Reports.137\$20.427. Runoff Report.138\$20.428. Option To File Monthly.138\$20.429. Option To File Monthly.138\$20.433. Contents of General-Purpose Committee Sworn Report of Contributions andExpenditures.140\$20.434. Alternate Reporting Requirements for General-Purpose Committees.144\$20.437. Form and Contents of Special Pre-Election Report.145\$20.439. Dissolution Report.145\$20.431. Contents of Dissolution Report.146\$20.432. Special Pre-Election Reports by Certain General-Purpose Committees.144\$20.437. Form and Contents of Special Pre-Election Report.146\$20.441. Contents of Dissolution Report.146\$20.441. Contents of Dissolution Report.146\$20.441. Contents of Dissolution Report.146\$20.503. Exceptions from Certain Notice Requirements.146\$20.503. Exceptions from Certain Notice Requirements.146\$20.503. Exceptions from Certain Notice Requ  |   |         |
| §20.409. Name of General-Purpose Committee.133§20.411. Contents of General-Purpose Committee Campaign Treasurer Appointment.133§20.413. Updating Information on the Campaign Treasurer Appointment.134§20.415. Termination of Campaign Treasurer Appointment.134§20.417. Termination Report.135§20.419. Converting to a Specific-Purpose Committee.136§20.421. Notice to Candidate or Officeholder.136§20.423. Semiannual Reports.137§20.427. Runoff Report.137§20.427. Runoff Report.138§20.429. Option To File Monthly.138§20.431. Monthly Reporting.139§20.433. Contents of General-Purpose Committee Sworn Report of Contributions andExpenditures.140§20.434. Alternate Reporting Requirements for General-Purpose Committees.144§20.437. Form and Contents of Special Pre-Election Report.145§20.439. Dissolution Report.145§20.441. Contents of Dissolution Report.145§20.441. Contents of Dissolution Report.146§20.501. Designation of Principal Political Committee.146§20.501. Designation of Principal Political Committee.146§20.503. Exceptions from Certain Notice Requirements.146§20.503. Exceptions from Certain Notice Requirements.146§20.503. Exceptions from Certain Notice Requirements.146§20.503. Exceptions from Certain Notice Requirements.146   |   |         |
| §20.411. Contents of General-Purpose Committee Campaign Treasurer Appointment.133§20.413. Updating Information on the Campaign Treasurer Appointment.134§20.415. Termination of Campaign Treasurer Appointment.134§20.417. Termination Report.135§20.419. Converting to a Specific-Purpose Committee.136§20.421. Notice to Candidate or Officeholder.136§20.423. Semiannual Reports.137§20.425. Pre-election Reports.137§20.427. Runoff Report.138§20.431. Monthly Reporting.138§20.432. Option To File Monthly.138§20.433. Contents of General-Purpose Committee Sworn Report of Contributions and<br>Expenditures.140§20.434. Alternate Reporting Requirements for General-Purpose Committees.144§20.437. Form and Contents of Special Pre-Election Report.145§20.439. Dissolution Report.145§20.441. Contents of Dissolution Report.145§20.439. Dissolution Report.146Subchapter G. RULES APPLICABLE TO A PRINCIPAL POLITICAL COMMITTEE146Subchapter H. RULES APPLICABLE TO A POLITICAL PARTY ACCEPTING146  |   |         |
| §20.413. Updating Information on the Campaign Treasurer Appointment.134§20.415. Termination of Campaign Treasurer Appointment.134§20.415. Termination Report.135§20.417. Termination Report.136§20.421. Notice to Candidate or Officeholder.136§20.423. Semiannual Reports.137§20.425. Pre-election Reports.137§20.429. Option To File Monthly.138§20.431. Monthly Reporting.139§20.432. Contents of General-Purpose Committee Sworn Report of Contributions andExpenditures.144§20.433. Contents of General-Purpose Committee Sworn Report of Contributions andExpenditures.144§20.437. Form and Contents of Special Pre-Election Report.145§20.439. Dissolution Report.145§20.441. Contents of Dissolution Report.145§20.439. Dissolution Report.146Subchapter G. RULES APPLICABLE TO A PRINCIPAL POLITICAL COMMITTEE146OF A POLITICAL PARTY146§20.501. Designation of Principal Political Committee.146Subchapter H. RULES APPLICABLE TO A POLITICAL PARTY ACCEPTING146   |   |         |
| §20.415. Termination of Campaign Treasurer Appointment.134§20.417. Termination Report.135§20.417. Termination Report.136§20.419. Converting to a Specific-Purpose Committee.136§20.421. Notice to Candidate or Officeholder.136§20.423. Semiannual Reports.137§20.425. Pre-election Reports.137§20.427. Runoff Report.138§20.429. Option To File Monthly.138§20.431. Monthly Reporting.139§20.433. Contents of General-Purpose Committee Sworn Report of Contributions andExpenditures.140§20.434. Alternate Reporting Requirements for General-Purpose Committees.144§20.435. Special Pre-Election Reports by Certain General-Purpose Committees.144§20.437. Form and Contents of Special Pre-Election Report.145§20.441. Contents of Dissolution Report.145§20.501. Designation of Principal Political Committee.146§20.503. Exceptions from Certain Notice Requirements.146Subchapter H. RULES APPLICABLE TO A POLITICAL PARTY ACCEPTING146   |   |         |
| §20.417. Termination Report.135§20.419. Converting to a Specific-Purpose Committee.136§20.421. Notice to Candidate or Officeholder.136§20.423. Semiannual Reports.137§20.425. Pre-election Reports.137§20.427. Runoff Report.138§20.429. Option To File Monthly.138§20.433. Contents of General-Purpose Committee Sworn Report of Contributions andExpenditures.140§20.434. Alternate Reporting Requirements for General-Purpose Committees.144§20.437. Form and Contents of Special Pre-Election Report.145§20.439. Dissolution Report.145§20.441. Contents of Dissolution Report.146Subchapter G. RULES APPLICABLE TO A PRINCIPAL POLITICAL COMMITTEE146Subchapter H. RULES APPLICABLE TO A POLITICAL PARTY ACCEPTING146   |   |         |
| §20.419. Converting to a Specific-Purpose Committee.136§20.421. Notice to Candidate or Officeholder.136§20.423. Semiannual Reports.137§20.425. Pre-election Reports.137§20.427. Runoff Report.138§20.429. Option To File Monthly.138§20.431. Monthly Reporting.139§20.433. Contents of General-Purpose Committee Sworn Report of Contributions andExpenditures.140§20.434. Alternate Reporting Requirements for General-Purpose Committees.144§20.437. Form and Contents of Special Pre-Election Report.145§20.439. Dissolution Report.145§20.441. Contents of Dissolution Report.146Subchapter G. RULES APPLICABLE TO A PRINCIPAL POLITICAL COMMITTEE146Subchapter H. RULES APPLICABLE TO A POLITICAL PARTY ACCEPTING146  |   |         |
| §20.421. Notice to Candidate or Officeholder.136§20.423. Semiannual Reports.137§20.425. Pre-election Reports.137§20.426. Pre-election Reports.137§20.427. Runoff Report.138§20.429. Option To File Monthly.138§20.431. Monthly Reporting.139§20.433. Contents of General-Purpose Committee Sworn Report of Contributions andExpenditures.140§20.434. Alternate Reporting Requirements for General-Purpose Committees.144§20.435. Special Pre-Election Reports by Certain General-Purpose Committees.144§20.437. Form and Contents of Special Pre-Election Report.145§20.439. Dissolution Report.145§20.441. Contents of Dissolution Report.146Subchapter G. RULES APPLICABLE TO A PRINCIPAL POLITICAL COMMITTEE146§20.501. Designation of Principal Political Committee.146§20.503. Exceptions from Certain Notice Requirements.146Subchapter H. RULES APPLICABLE TO A POLITICAL PARTY ACCEPTING146  | •   |         |
| §20.423. Semiannual Reports.137§20.425. Pre-election Reports.137§20.427. Runoff Report.138§20.429. Option To File Monthly.138§20.431. Monthly Reporting.139§20.433. Contents of General-Purpose Committee Sworn Report of Contributions andExpenditures.140§20.434. Alternate Reporting Requirements for General-Purpose Committees.144§20.435. Special Pre-Election Reports by Certain General-Purpose Committees.144§20.437. Form and Contents of Special Pre-Election Report.145§20.439. Dissolution Report.145§20.441. Contents of Dissolution Report.146Subchapter G. RULES APPLICABLE TO A PRINCIPAL POLITICAL COMMITTEE146§20.501. Designation of Principal Political Committee.146§20.503. Exceptions from Certain Notice Requirements.146Subchapter H. RULES APPLICABLE TO A POLITICAL PARTY ACCEPTING146   |   |         |
| \$20.425. Pre-election Reports.137\$20.427. Runoff Report.138\$20.429. Option To File Monthly.138\$20.429. Option To File Monthly.139\$20.431. Monthly Reporting.139\$20.433. Contents of General-Purpose Committee Sworn Report of Contributions andExpenditures.140\$20.434. Alternate Reporting Requirements for General-Purpose Committees.144\$20.435. Special Pre-Election Reports by Certain General-Purpose Committees.144\$20.437. Form and Contents of Special Pre-Election Report.145\$20.439. Dissolution Report.145\$20.441. Contents of Dissolution Report.146Subchapter G. RULES APPLICABLE TO A PRINCIPAL POLITICAL COMMITTEE146\$20.501. Designation of Principal Political Committee.146\$20.503. Exceptions from Certain Notice Requirements.146Subchapter H. RULES APPLICABLE TO A POLITICAL PARTY ACCEPTING146  |   |         |
| §20.427. Runoff Report.138§20.429. Option To File Monthly.138§20.431. Monthly Reporting.139§20.433. Contents of General-Purpose Committee Sworn Report of Contributions andExpenditures.140§20.434. Alternate Reporting Requirements for General-Purpose Committees.144§20.435. Special Pre-Election Reports by Certain General-Purpose Committees.144§20.437. Form and Contents of Special Pre-Election Report.145§20.439. Dissolution Report.145§20.441. Contents of Dissolution Report.146Subchapter G. RULES APPLICABLE TO A PRINCIPAL POLITICAL COMMITTEE146§20.501. Designation of Principal Political Committee.146§20.503. Exceptions from Certain Notice Requirements.146Subchapter H. RULES APPLICABLE TO A POLITICAL PARTY ACCEPTING146   | - 1   |         |
| §20.429. Option To File Monthly.138§20.431. Monthly Reporting.139§20.433. Contents of General-Purpose Committee Sworn Report of Contributions and140Expenditures.140§20.434. Alternate Reporting Requirements for General-Purpose Committees.144§20.435. Special Pre-Election Reports by Certain General-Purpose Committees.144§20.437. Form and Contents of Special Pre-Election Report.145§20.439. Dissolution Report.145§20.441. Contents of Dissolution Report.146Subchapter G. RULES APPLICABLE TO A PRINCIPAL POLITICAL COMMITTEE146§20.501. Designation of Principal Political Committee.146§20.503. Exceptions from Certain Notice Requirements.146Subchapter H. RULES APPLICABLE TO A POLITICAL PARTY ACCEPTING146  | •   |         |
| §20.431. Monthly Reporting.139§20.433. Contents of General-Purpose Committee Sworn Report of Contributions and140Expenditures.140§20.434. Alternate Reporting Requirements for General-Purpose Committees.144§20.435. Special Pre-Election Reports by Certain General-Purpose Committees.144§20.437. Form and Contents of Special Pre-Election Report.145§20.439. Dissolution Report.145§20.441. Contents of Dissolution Report.146Subchapter G. RULES APPLICABLE TO A PRINCIPAL POLITICAL COMMITTEE146§20.501. Designation of Principal Political Committee.146§20.503. Exceptions from Certain Notice Requirements.146Subchapter H. RULES APPLICABLE TO A POLITICAL PARTY ACCEPTING146   |   |         |
| <ul> <li>\$20.433. Contents of General-Purpose Committee Sworn Report of Contributions and<br/>Expenditures</li></ul>  |   |         |
| Expenditures.140§20.434. Alternate Reporting Requirements for General-Purpose Committees.144§20.435. Special Pre-Election Reports by Certain General-Purpose Committees.144§20.437. Form and Contents of Special Pre-Election Report.145§20.439. Dissolution Report.145§20.441. Contents of Dissolution Report.146Subchapter G. RULES APPLICABLE TO A PRINCIPAL POLITICAL COMMITTEE146§20.501. Designation of Principal Political Committee.146§20.503. Exceptions from Certain Notice Requirements.146Subchapter H. RULES APPLICABLE TO A POLITICAL PARTY ACCEPTING146  |   |         |
| <ul> <li>§20.434. Alternate Reporting Requirements for General-Purpose Committees</li></ul>  | • •   |         |
| <ul> <li>§20.435. Special Pre-Election Reports by Certain General-Purpose Committees</li></ul>   |   |         |
| <ul> <li>§20.437. Form and Contents of Special Pre-Election Report.</li> <li>§20.439. Dissolution Report.</li> <li>§20.441. Contents of Dissolution Report.</li> <li>Subchapter G. RULES APPLICABLE TO A PRINCIPAL POLITICAL COMMITTEE</li> <li>OF A POLITICAL PARTY</li> <li>§20.501. Designation of Principal Political Committee.</li> <li>§20.503. Exceptions from Certain Notice Requirements.</li> <li>146</li> <li>Subchapter H. RULES APPLICABLE TO A POLITICAL PARTY ACCEPTING</li> </ul>   |   |         |
| <ul> <li>§20.439. Dissolution Report.</li> <li>§20.441. Contents of Dissolution Report.</li> <li>Subchapter G. RULES APPLICABLE TO A PRINCIPAL POLITICAL COMMITTEE</li> <li>OF A POLITICAL PARTY</li> <li>§20.501. Designation of Principal Political Committee.</li> <li>§20.503. Exceptions from Certain Notice Requirements.</li> <li>146</li> <li>Subchapter H. RULES APPLICABLE TO A POLITICAL PARTY ACCEPTING</li> </ul>   |   |         |
| <ul> <li>§20.441. Contents of Dissolution Report.</li> <li>Subchapter G. RULES APPLICABLE TO A PRINCIPAL POLITICAL COMMITTEE</li> <li>OF A POLITICAL PARTY</li> <li>\$20.501. Designation of Principal Political Committee.</li> <li>\$20.503. Exceptions from Certain Notice Requirements.</li> <li>\$146</li> <li>\$20.503. Exceptions from Certain Notice Requirements.</li> <li>\$146</li> <li>\$146</li> <li>\$20.503. Exceptions from Certain Notice Requirements.</li> </ul>  |   |         |
| Subchapter G. RULES APPLICABLE TO A PRINCIPAL POLITICAL COMMITTEEOF A POLITICAL PARTY146\$20.501. Designation of Principal Political Committee146\$20.503. Exceptions from Certain Notice Requirements146Subchapter H. RULES APPLICABLE TO A POLITICAL PARTY ACCEPTING   | •   |         |
| OF A POLITICAL PARTY   |   |         |
| <ul> <li>§20.501. Designation of Principal Political Committee</li></ul>   |   |         |
| \$20.503. Exceptions from Certain Notice Requirements  |   |         |
| Subchapter H. RULES APPLICABLE TO A POLITICAL PARTY ACCEPTING  |   |         |
|  |   |         |
|  |   | 147     |

| §20.521. Restrictions on Use of Contributions from Corporations or Labor Organi  | zations    |
|--|------------|
| 320.521. Resultions on ese of contributions from corporations of Earor organi  |            |
| §20.523. Separate Account Required   |            |
| §20.525. Record of Contributions and Expenditures and Contents of Report   |            |
| §20.527. Form of Report  |            |
| \$20.529. Reporting Schedule for Political Party Accepting Corporate or Labor Org  | ganization |
| Contributions.   |            |
| §20.531. Restrictions on Contributions before General Election.  | 151        |
| Subchapter I. RULES APPLICABLE TO A POLITICAL PARTY'S COUNTY   |            |
| EXECUTIVE COMMITTEE  | 151        |
| §20.551. Obligation To Maintain Records  |            |
| §20.553. Campaign Treasurer Appointment Not Required for County Executive C  |            |
| Accepting Contributions or Making Expenditures Under Certain Amount  |            |
| §20.555. County Executive Committee Accepting Contributions or Making Expen  |            |
| That Exceed Certain Amount   |            |
| §20.557. Exceptions from Certain Restrictions.   |            |
| §20.559. Exception from Notice Requirement.  |            |
| §20.561. County Executive Committee Accepting Contributions from Corporation   |            |
| Labor Organizations.   |            |
| Subchapter J. REPORTS BY A CANDIDATE FOR STATE OR COUNTY PAI   |            |
| CHAIR  |            |
| §20.571. Definitions   |            |
| <ul><li>§20.573. Rules Applicable to Candidate for State Chair of a Political Party.</li><li>§20.575. Contributions to and Expenditures by Candidate for State Chair of a Political Party.</li></ul> |            |
| §20.575. Contributions to and Expenditures by Candidate for State Chair of a Poin  | •          |
| \$20.577. Reporting Schedule for a Candidate for State Chair   |            |
| §20.579. Candidates for County Chair in Certain Counties   |            |
| Subchapter K. REPORTS BY POLITICAL COMMITTEES SUPPORTING O   | R          |
| OPPOSING A CANDIDATE FOR STATE OR COUNTY CHAIR OF A POLI   |            |
| PARTY  |            |
| §20.591. Appointment of Campaign Treasurer by Political Committee Supporting   | or         |
| Opposing Candidate for State Chair of a Political Party  |            |
| §20.593. Contributions and Expenditures by Political Committee Supporting or Op  | oposing    |
| Candidate for State Chair of a Political Party.  | 155        |
| §20.595. Reporting Schedule for a Political Committee Supporting or Opposing C   | andidate   |
| for State Chair of a Political Party.  |            |
| §20.597. Political Committees Supporting or Opposing Candidates for County Ch  |            |
| Certain Counties   |            |
| CHAPTER 22. RESTRICTIONS ON CONTRIBUTIONS AND EXPENDITURES   |            |
| §22.1. Certain Campaign Treasurer Appointments Required before Political Activ   | • •        |
|  |            |
| §22.3. Disclosure of True Source of Contribution or Expenditure.   |            |
| §22.6. Reporting Direct Campaign Expenditures.   |            |
| §22.7. Contribution from Out-of-State Committee.   |            |
| \$22.9. Cash Contributions Exceeding \$100 Prohibited.   |            |
| §22.11. Prohibition on Contributions during Regular Session.   | 101        |
| §22.13. Contributions in the Capitol Prohibited. §22.17. Prohibition on Personal Use of Political Contributions.   |            |
| 822.17. FIOIDIDUOID ON FEISONALUSE OF FONDCAL CONTIDUTIONS.  |            |

| §22.19. General Restrictions on Reimbursement of Personal Funds                    | .163         |
|--|--------------|
| §22.21. Additional Restrictions on Reimbursement of Personal Funds and Payments on |              |
| Certain Loans.   | .164         |
| §22.23. Restrictions on Certain Payments.  | .164         |
| §22.27. Time Limit on Retaining Unexpended Contributions                           |              |
| §22.29. Activity after Death or Incapacity of Candidate or Officeholder.           |              |
| §22.31. Restrictions on Foreign Nationals.   |              |
| §22.35. Corporate Contributions to Certain Political Committees.                   | .166         |
| §22.37. Virtual Currency Contributions   |              |
| CHAPTER 24. RESTRICTIONS ON CONTRIBUTIONS AND EXPENDITURES                         |              |
| APPLICABLE TO CORPORATIONS AND LABOR ORGANIZATIONS                                 | .167         |
| §24.1. Corporations and Certain Associations Covered.                              | .167         |
| §24.5. Corporate Loans.  | .168         |
| §24.15. Payments to a Corporation of the Candidate or Officeholder                 | .168         |
| §24.17. Corporate Expenditures for Get-Out-the-Vote Campaigns Permitted            | .168         |
| §24.18. Designation of Contribution for Administrative Purposes                    |              |
| §24.19. Affidavit Required by a Political Committee Making a Direct Campaign       |              |
| Expenditure from a Political Contribution Accepted from a Corporation or Labor     |              |
| Organization.  | .169         |
| CHAPTER 26. POLITICAL AND LEGISLATIVE ADVERTISING                                  | .170         |
| §26.1. Disclosure Statement  |              |
| §26.2. Newsletter of Public Officer of a Political Subdivision                     | .171         |
| §26.3. Legislative Advertising.  |              |
| §26.5. Code of Fair Campaign Practices   |              |
| §26.7. Use of the Term "Reelect" in Political Advertising                          |              |
| CHAPTER 28. REPORTS BY A CANDIDATE FOR SPEAKER OF THE HOUSE OF                     |              |
| REPRESENTATIVES  | .172         |
| §28.1. Definitions   | .172         |
| §28.5. Information To Report.  | .172         |
| §28.7. Permitted Expenditures.   |              |
| §28.9. Segregation of Campaign Funds.  |              |
| CHAPTER 34. REGULATION OF LOBBYISTS  |              |
| Subchapter A. GENERAL PROVISIONS   | .173         |
| §34.1. Definitions   |              |
| §34.3. Compensation for Preparation Time.  |              |
| §34.5. Certain Compensation Excluded   |              |
| §34.7. Reimbursement for Office Expenses.  |              |
| §34.9. Taxes and Tips.   |              |
| §34.11. Attribution of Expenditure to More Than One Person; Reimbursement of Lobby |              |
| Expenditure.   |              |
| §34.13. Incidental Expenditures for Transportation.                                | .175         |
| §34.14. Expenditures for Fact-Finding Trips  |              |
| §34.15. Reporting Subject Matter.  |              |
| §34.17. Satisfaction of Presence Requirement by Entity                             |              |
| §34.19. Courtesy Notices by Electronic Mail.                                       |              |
| Subchapter B. REGISTRATION REQUIRED  |              |
| §34.41. Expenditure Threshold.   |              |
| 354.41. Expenditure Threshold.   | .176         |
| §34.43. Compensation and Reimbursement Threshold.                                  | .176<br>.177 |

| §34.45. Entity Registration.   | 177 |
|--|-----|
| Subchapter C. COMPLETING THE REGISTRATION FORM                           | 177 |
| §34.63. Assistants   | 177 |
| §34.65. Compensation Reported by Lobby Firm Employee                     | 178 |
| §34.67. Paid, Earned, and Prospective Compensation.                      |     |
| §34.69. Subject Matter.  | 178 |
| §34.71. Amending a Registration Form.                                    | 179 |
| §34.75. Reporting of Commission or Fee Paid by State Agency              | 179 |
| §34.77. Disclosure of Registration under Foreign Agents Registration Act | 179 |
| Subchapter D. LOBBY ACTIVITY REPORTS                                     | 179 |
| §34.81. Election to File Annually.                                       | 179 |
| §34.82. Modified Reporting Threshold                                     | 180 |
| §34.83. Time of Expenditure.   | 180 |
| §34.85. Individual Reporting Expenditure by Entity.                      | 180 |
| Subchapter E. ELECTRONIC FILING  | 180 |
| §34.91. Exemptions from Electronic Filing                                |     |
| CHAPTER 40. FINANCIAL DISCLOSURE FOR PUBLIC OFFICERS                     |     |
| §40.1. Financial Statement   |     |
| §40.2. Disclosure of Financial Activity.                                 |     |
| §40.11. Publicly Traded Corporation as Source of Income over \$500       |     |
| CHAPTER 45. CONFLICTS OF INTEREST  |     |
| §45.1. Application   |     |
| §45.3. Definitions   |     |
| §45.5. Definitions   |     |
| §45.7. Rebates   | 184 |
| §45.9. Rebates   |     |
| CHAPTER 46. DISCLOSURE OF INTERESTED PARTIES                             |     |
| §46.1. Application   |     |
| §46.3. Definitions   | 185 |
| §46.4. Changes to Contracts  |     |
| §46.5. Disclosure of Interested Parties Form.                            |     |
| CHAPTER 50. LEGISLATIVE SALARIES AND PER DIEM                            |     |
| §50.1. Legislative Per Diem  | 188 |

#### CHAPTER 6. ORGANIZATION AND ADMINISTRATION

#### Subchapter A. GENERAL RULES

#### §6.1. Definitions.

The following words and terms, when used in this part, shall have the following meanings, unless the context clearly indicates otherwise.

(1) Act--The Government Code, Chapter 571 (concerning Texas Ethics Commission).

(2) Administrative Procedure Act--The Government Code, Chapter 2001 (concerning Administrative Procedure).

(3) Agency--The state agency governed by the commission, as it functions and operates through the administrative staff hired by the commission and its executive director.

(4) Commission--The Texas Ethics Commission, as constituted and described in the Texas Constitution, Article 3, §24a.

(5) Document--A report, complaint, response, letter, or any other written material.

(6) Executive director--The person employed by the commission to serve as the agency's chief administrative officer, or any other employee of the commission acting as the designee of the executive director.

(7) Family member or relative--An individual who is related within the second degree of affinity or consanguinity, as defined by the Government Code, Chapter 573, Subchapter B (concerning Relationships by Consanguinity or by Affinity).

(8) Filer--A person required to file a report with the commission or a local filing authority in accordance with this title.

(9) Individual--A human being who has been born and is alive.

(10) Local filing authority--A public servant other than the Texas Ethics Commission with whom a filer must file a report in accordance with this title, as identified in §20.5 of this title (relating to Reports Filed with a County Filing Authority) and §20.7 of this title (relating to Reports Filed with Other Local Filing Authority).

(11) Open Meetings Law--The Government Code, Chapter 551 (concerning Open Meetings).

(12) Open Records Law--The Government Code, Chapter 552 (concerning Open Records).

(13) Person--An individual, representative, corporation, association, or other entity, including any nonprofit corporation, or any agency or instrumentality of federal, state, or local government.

(14) Postmark--A postal cancellation by the United States Postal Service that contains the post office name, state, and zip code and the month, day, and year the canceling post office accepted custody of the material.

(15) Presiding officer--The person elected to serve as the commission's chairman or chairwoman under §6.21 of this title (relating to Officers of the Commission).

(16) Report--Any document or other information required to be filed under this title.

(17) Staff--Employees of the commission, hired by the commission or the executive director.

(18) Title 15--The Election Code, Title 15 (concerning Regulating Political Funds and Campaigns).

(19) First responder--An individual who is:

(A) a peace officer whose duties include responding rapidly to an emergency;

(B) fire protection personnel, as that term is defined by Section 419.021, Government Code;

(C) a volunteer firefighter who performs firefighting duties on behalf of a political subdivision;

(D) an ambulance driver; or

(E) an individual certified as emergency medical services personnel by the Department of State Health Services.

(20) Judicial office--The office of:

(A) chief justice or justice, supreme court;

(B) presiding judge or judge, court of criminal appeals;

(C) chief justice or justice, court of appeals;

(D) district judge;

(E) judge, statutory county court; or

(F) judge, statutory probate court.

(21) Non-judicial office--An elective public office and the secretary of state, but not including an office described by paragraph (20) of this section.

#### §6.3. Purpose and Construction of Rules.

(a) This title is adopted for the following purposes:

(1) to establish and explain the practice and procedures followed by the commission and its employees performing its duties under the law; and

(2) to provide specific guidance necessary to encourage and ensure full compliance with all laws administered and enforced by the commission.

(b) A person's obligation to comply with a requirement or prohibition established by statute exists even if this title is silent concerning a statutory requirement.

(c) This title shall always be construed in a manner consistent with all applicable constitutional and statutory requirements.

#### §6.5. Authority To Adopt Rules.

This title is adopted under the authority granted by the Act, the Administrative Procedure Act, and by any other law administered and enforced by the commission that establishes the commission's authority to adopt rules.

#### §6.7. Actions That Require Six Votes.

(a) The following actions require the affirmative vote of no less than six members of the commission:

(1) to adopt a rule to administer any law administered and enforced by the commission;

(2) to render any decision on a complaint or a report of a violation as provided by the Government Code, Chapter 571 (concerning Texas Ethics Commission), other than a final decision after a formal hearing that a violation has not occurred, which requires only five votes;

(3) to prohibit participation by a member of the commission in commission proceedings relating to the investigation, complaint, or motion;

(4) without a sworn complaint, to initiate a preliminary review of an alleged violation of a law administered or enforced by the commission;

(5) to subpoena and examine witnesses and documents that directly relate to a sworn complaint and issue a written request to a peace officer to serve a subpoena of the commission in the manner prescribed for service of a district court subpoena;

(6) to order and perform a complete audit at an informal or formal hearing of a sworn complaint or commission-initiated complaint; and

(7) to initiate civil enforcement actions and refer matters to the appropriate prosecuting attorney for criminal prosecution.

(b) Any action not listed in subsection (a) of this section that requires a vote of the commission requires the affirmative vote of no less than five members of the commission.

#### §6.9. Computation of Time.

(a) This section states how to compute a period of time prescribed or allowed by this title, by any order of the agency, or by any applicable statute. The day of the act, event, or default after which the designated period of time begins to run is not included. The last day of the period is included. However, if the last day of the time period would be a Saturday, a Sunday, or a legal holiday, the period is extended until the next day that is not a Saturday, a Sunday, or a legal holiday. A legal holiday, for purposes of this section, is any day other than a Saturday or Sunday that the agency is closed for a holiday established by state law.

(b) A time period described by statute or this title to be a certain number of business days is calculated under subsection (a) of this section without including any Saturday, Sunday, or legal holiday within that time period.

(c) A document required to be filed or served by a deadline established by statute or this title is filed or served when it is actually received. A document may be deemed to be filed or served when it is deposited with the United States Postal Service, properly addressed to the recipient, with all postage prepaid. The date of the postmark on the envelope for the document is presumed to be the date the document was deposited with the United States Postal Service.

(d) A document filed or served by delivery to the United States Postal Service is presumed to have been filed before 5:00 p.m. on the date indicated by the postmark.

#### Subchapter B. OFFICERS AND EMPLOYEES OF THE COMMISSION

#### §6.21. Officers of the Commission.

(a) The commission shall select a presiding officer and a vice-presiding officer.

(b) Commission officers are elected annually by majority vote of the commission. The election shall take place at the first commission meeting held after June 1 of each year. Each officer shall serve until his or her successor is selected.

(c) The presiding officer and the vice-presiding officer shall be elected from different political party caucus lists.

(d) The presiding officer may be re-elected; however, if a new presiding officer is elected it should be from a different political party caucus list than the former presiding officer.

(e) The presiding officer shall preside at all meetings of the commission. While presiding, the presiding officer shall direct the order of the meeting, appoint committees and persons to chair committees, recognize persons to be heard at hearings, set reasonable and necessary time limits for speakers, and take other actions to clarify issues and preserve order. When the presiding officer is absent, the vice-presiding officer shall perform all duties of the presiding officer.

(f) The presiding officer may perform the following actions of the commission:

(1) Sign previously approved subpoenas and orders;

(2) Schedule hearings and meetings;

(3) Timely respond to litigation matters on behalf of the commission when action is required before the next scheduled meeting and is within the scope of the authorization granted by the commission; and

(4) Respond to matters on behalf of the commission when action is required and is within the scope of the authorization granted by the commission.

(g) The presiding officer may appoint a commissioner as chair pro tem to preside over a hearing held by the commission.

#### §6.23. Commission Staff.

(a) The executive director is the chief administrative officer of the agency. The executive director shall attend commission meetings at the pleasure of the commission and serve as liaison between the commission and the public.

(b) The commission delegates to the executive director all powers conferred on the commission by the Act or other law, except for any power that requires a vote of the commission. Any action taken by the executive director shall conform with all applicable law, including this title and other policies that may be adopted from time to time by the commission.

(c) The executive director shall attend commission meetings unless specifically excused by the commission and shall perform any duties or assignments established by the commission.

(d) The general counsel shall attend commission meetings unless specifically excused by the commission, shall provide legal advice to the commission and executive director, and shall perform any duties delegated by the executive director.

#### §6.25. Appointment and Operation of Advisory Committees.

(a) The commission by resolution may establish one or more committees to obtain the viewpoints and advice of interested persons with respect to any contemplated rulemaking. The membership or method of appointment of members to a committee established under this section shall be specified in the resolution that creates the committee. A committee created under this section is advisory only.

(b) In addition to committees established under subsection (a) of this section, with the consent of other members of the commission the presiding officer may from time to time establish and appoint commission members and others to a special committee to exercise advisory duties specified by the presiding officer.

#### Subchapter C. COMMISSION MEETINGS

#### §6.31. Quorum.

Five commissioners must be present as a quorum to hold a commission meeting.

#### §6.33. Frequency of Meetings.

The commission shall meet at least once a quarter at the call of the presiding officer.

#### §6.35. Called Meetings.

The executive director shall give notice to each commissioner of the date and time of each meeting. Notice under this section shall be provided a reasonable amount of time in advance of the meeting, and may be by telephone, fax, or mail.

#### §6.37. Open Meetings.

(a) Except as provided by subsection (b) of this section, each meeting of the commission shall be conducted in accordance with the Open Meetings Law.

(b) A commission meeting limited to consideration and action on matters relating to sworn complaints is not subject to the Open Meetings Law.

#### §6.39. Meeting Agenda.

(a) The agenda shall consist of agenda items proposed by the executive director prior to the meetings for which the agenda is specified. At a reasonable time before filing a copy of the agenda as required by the Open Meetings Law, the executive director shall provide a copy of the proposed agenda to the presiding officer. If the presiding officer is not reasonably available, the executive director shall provide a copy of the proposed agenda to the vice-presiding officer. If the proposed agenda to the vice-presiding officer. If the proposed agenda to the vice-presiding officer. If the proposed agenda to the vice-presiding officer is not reasonably available, the executive director shall provide a copy of the proposed agenda to acopy of the proposed agenda to any two commissioners.

(b) The presiding officer, a commission member with the consent of the presiding officer, or any two commissioners may direct the executive director to include an item on the agenda if it complies with the posting requirements specified by law. The presiding officer may direct the executive director to remove an item included on a proposed agenda unless that item is requested by two commission members other than the presiding officer.

(c) A member of the public may ask the executive director to place an item on a proposed agenda. The executive director shall advise the commission of the request and may include the item on a proposed agenda.

#### **§6.41.** Public Hearing and Participation.

(a) A public hearing on an agenda item shall be conducted when required by law or requested by a commissioner.

(b) The executive director shall prepare and maintain a plan for providing special assistance (including without limitation translation of the English language) to persons who request such assistance for the purpose of attending, observing, or participating in a commission meeting.

#### §6.43. Speakers Addressing the Commission.

(a) The executive director shall prescribe a speaker registration form. Each person who wishes to speak at a commission meeting shall provide the following information:

- (1) the speaker's name;
- (2) the person or entity the speaker represents, if any;
- (3) the agenda item the speaker wishes to address; and
- (4) his or her mailing address and telephone number.

(b) Any person who addresses the commission shall state his or her name and the name of the person or entity the speaker represents, if any, for purposes of the tape recording under §6.47 of this title (relating to Tape Recording of Meeting; Minutes).

#### §6.45. Order and Conduct of Commission Meeting.

(a) The presiding officer shall preside at all meetings of the commission. The presiding officer shall direct the order of the meeting in accordance with its agenda, recognize persons to be heard, set reasonable and necessary time limits for speakers, maintain and enforce appropriate standards of conduct, and take any other action necessary in his or her discretion to clarify issues and preserve order. When the presiding officer is absent, the vice-presiding officer shall perform all duties under this subsection.

(b) Commission meetings shall be conducted in accordance with rules and procedures set forth in the most recently published edition of Robert's Rules of Order.

(c) With unanimous consent of all commissioners present, any provision or requirement of this section may be waived.

(d) No action of the commission that otherwise complies with law shall be void or invalid because the action was taken in violation of a rule or procedure established by this section.

#### §6.47. Tape Recording of Meeting; Minutes.

(a) All meetings of the commission shall be tape recorded. The tape recording shall be the official record of actions taken at the meeting.

(b) The presiding officer shall announce the names of each commissioner who makes or seconds a motion to be voted upon by the commission. After the vote has been taken, the presiding officer shall announce the vote in a manner that identifies how each commissioner voted, if a commissioner abstained, or if a commissioner was not present for the vote. (c) The executive director shall prepare minutes after each meeting that reflect all commission votes and other actions taken during the meeting. The minutes shall be approved by vote of the commission at a subsequent commission meeting.

#### Subchapter D. RULEMAKING PROCEDURES

#### §6.61. Comments on Proposed Rules.

(a) Written comments on a proposed rule received at the agency office shall be reviewed by the executive director and made available to each member of the commission before final action to adopt the rule.

(b) Oral or written comments on a proposed rule may also be offered at the public hearing required by §6.63 of this title (relating to Public Hearings on Proposed Rules).

#### §6.63. Public Hearings on Proposed Rules.

The commission will hold a public hearing on each proposed rule before it takes final action to adopt the rule. Unless otherwise scheduled, the public hearing will be held immediately before the commission votes on the proposed rule.

#### Subchapter E. AGENCY FEES AND CHARGES

#### §6.81. Charges; Payment of Money; Refunds.

(a) Any fee or charge payable to the agency shall be paid in advance, unless satisfactory arrangements for subsequent payment are approved by the executive director.

(b) Money paid by actual mistake or in excess, such as a payment not required by law, may be refunded. A mere change of purpose after the payment of money, as when a party desires to withdraw a filing, will not entitle a party to a refund.

#### §6.83. Copying Charges.

The charge for providing copies of documents shall be in accordance with rules established by the General Services Commission or other applicable law.

#### §6.87. Waiver of Fees for Copies or Publications.

The executive director may waive or reduce a charge established by this subchapter when, in his or her discretion, a waiver or reduction of the fee is in the public interest because furnishing the information primarily benefits the general public.

#### **CHAPTER 8. ADVISORY OPINIONS**

#### §8.1. Definitions.

The following words and terms, when used in this chapter, shall have the following meanings, unless the context clearly indicates otherwise: AOR number-An advisory opinion request file number assigned by the executive director to a pending advisory opinion request in accordance with this chapter.

#### §8.3. Subject of an Advisory Opinion.

(a) The commission may only issue a written advisory opinion on the application of any of the following laws:

(1) Government Code, Chapter 302 (concerning Speaker of the House of Representatives);

(2) Government Code, Chapter 303 (concerning Governor for a Day and Speaker's Reunion Day Ceremonies);

(3) Government Code, Chapter 305 (concerning Registration of Lobbyists);

(4) Government Code, Chapter 572 (concerning Personal Financial Disclosure, Standards of Conduct, and Conflict of Interest);

(5) Government Code, Chapter 2004 (concerning Representation Before State Agencies);

(6) Local Government Code, Chapter 159, Subchapter C, in connection with a county judicial officer, as defined by Section 159.051, Local Government Code, who elects to file a financial statement with the commission;

(7) Election Code, Title 15 (concerning Regulating Political Funds and Campaigns);

(8) Penal Code, Chapter 36 (concerning Bribery and Corrupt Influence);

(9) Penal Code, Chapter 39 (concerning Abuse of Office).

(10) Government Code, §2152.064 (concerning Conflict of Interest in Certain Transactions); and

(11) Government Code, §2155.003 (concerning Conflict of Interest).

(b) The commission may not issue an advisory opinion that concerns the subject matter of pending litigation known to the commission.

(c) For purposes of this section, the term litigation includes a sworn complaint proceeding before the commission only if the Government Code Subchapters C-H, Chapter 2001, applies to the proceeding.

(d) An advisory opinion cannot resolve a disputed question of fact.

#### §8.5. Persons Eligible To Receive an Advisory Opinion.

A person who is subject to one of the laws described in §8.3(a) of this chapter (relating to Subject of Advisory Opinions) may request an opinion that advises how the law applies to that person in a specific real or hypothetical factual situation.

#### §8.7. Request for an Advisory Opinion.

(a) A request for an advisory opinion shall describe a specified factual situation. The facts specified may be real or hypothetical. The request must provide sufficient detail to permit the commission to provide a response to the request, including the name of the person making the request and, if applicable, the name of the person on whose behalf the request is made.

(b) A request for an advisory opinion shall be:

- (1) in writing; and
- (2) mailed or hand-delivered to the commission at the agency office or emailed to the commission's email address designated for receiving requests.

#### **§8.9.** Commission Initiated Opinion.

When a majority of the commission determines that an opinion would be in the public interest or in the interest of any person or persons within the jurisdiction of the commission, the commission may on its own motion issue an advisory opinion.

#### **§8.11. Review and Processing of a Request.**

(a) Upon receipt of a written request for an advisory opinion, the executive director shall determine whether the request:

- (1) pertains to the application of a law specified under §8.3 of this chapter;
- (2) meets the standing requirements of §8.5 of this chapter;
- (3) meets the form requirements of §8.7 of this chapter; and
- (4) cannot be answered by written response under § 8.17 of this chapter by reference to the plain language of a statute, commission rule, or advisory opinion.

(b) If the executive director determines that a request for an opinion meets the requirements of this chapter as set forth in subsections (a)(1)-(3) of this section and that the request cannot be answered by written response under \$8.17 of this chapter, the executive director shall assign an AOR number to the request. The executive director shall notify the person making the request of the AOR number and of the proposed wording of the question to be answered by the commission.

(c) If the executive director determines that a request for an opinion does not meet the requirements of this chapter as set forth in subsections (a)(1)-(3) of this section or that the request can be answered by written response under §8.17 of this chapter, the executive director shall

notify the person making the request of the reason the person making the request is not entitled to an advisory opinion in response to the request.

(d) A person who requests an opinion may withdraw the request prior to its inclusion on a meeting agenda filed by the Commission pursuant to the Open Meetings Law. Once a request is included on such an agenda, it may not be withdrawn by the requestor.

#### §8.13. Time Period.

(a) The commission shall issue an advisory opinion in response to a request that meets the requirements of this chapter not later than the 60th day after the date the commission receives the request.

(b) The time available to issue an advisory opinion in response to a written request is automatically extended for 60 days pursuant to \$571.092(b), Government Code.

#### §8.15. Publication in Texas Register; Comments.

(a) Each request assigned an AOR number under this chapter shall be published in summary form in the Texas Register.

(b) Any person may submit written comments to the commission concerning an advisory opinion request. Comments submitted should reference the AOR number.

#### §8.17. Request Answered by Written Response.

If the executive director determines that a request can be answered by reference to the plain language of a statute, commission rule, or advisory opinion:

- (1) the executive director shall provide a written response to the person making the request that cites the language of the statute, rule, or advisory opinion, as applicable; and
- (2) the person making the request is not entitled to an advisory opinion in response to the request.

#### §8.18. No Defense to Prosecution or Civil Penalty.

A person who requests an advisory opinion does not obtain a defense to prosecution or to imposition of a civil penalty by requesting the opinion if any of the following apply:

(1) the commission is not authorized to answer the request because it does not pertain to the application of a law specified under §8.3 of this chapter;

(2) the request does not meet the standing requirements of §8.5 of this chapter;

(3) the request does not meet the form requirements of §8.7 of this chapter; or

(4) the executive director responds to the request by written response under §8.17 of this chapter.

#### §8.19. Confidentiality.

(a) The name of a person who requests an advisory opinion is confidential.

(b) The original request for an advisory opinion shall be placed in a confidential file.

(c) Confidentiality under subsection (a) of this section may be waived only if the person making the request for an advisory opinion provides a verified, written waiver of confidentiality to the executive director.

(d) If a request for a copy of an advisory opinion request is received, the executive director shall prepare a redacted version of the advisory opinion request by deleting any information that is likely to identify the person making the request. The redacted version of the request shall be provided to the person who requested a copy of the advisory opinion request.

#### **§8.21.** Compilation of Advisory Opinions.

The executive director shall number and categorize each advisory opinion issued and shall annually compile a summary of advisory opinions in a single reference document. The executive director may publish and provide copies of advisory opinions in other formats as may be in the public interest.

#### CHAPTER 10. ETHICS TRAINING PROGRAMS

#### §10.1. Training Programs.

Upon approval of the commission, the executive director shall establish a program to provide training relating to the laws administered and enforced by the commission and related laws for:

(1) members and members-elect of the legislature, to be held by January of each odd-numbered year;

(2) state employees, in cooperation with state agencies; and

(3) other persons and officials whose conduct is regulated by laws administered and enforced by the commission and related laws.

#### §10.3. Tuition Charges to Attendees of Training Programs.

Upon approval of the commission, the executive director may establish tuition charges for persons who attend training programs under §10.1(3) of this title (relating to Training Programs) to recover costs of the training.

#### CHAPTER 12. SWORN COMPLAINTS

#### Subchapter A. GENERAL PROVISIONS AND PROCEDURES

#### §12.5. Deadline for Filing a Complaint.

(a) The commission has no jurisdiction over an alleged violation:

(1) if the alleged violation is also a criminal offense, and if, at the time the complaint is filed or at the time the commission would vote to initiate a preliminary review of a matter, the allegation would be barred from criminal prosecution by operation of the applicable statute of limitations; or

(2) if the alleged violation is not also a criminal offense and if the allegation is based on facts that occurred more than three years before the date the complaint is filed or the date the commission would vote to initiate a preliminary review of a matter.

(b) For purposes of this section, a complaint is filed on the date it is hand-delivered to the commission or on the date that it is deposited in the mail or with a common or contract carrier, properly addressed, with postage prepaid.

(c) For purposes of this section, a complaint is not filed unless it complies with the requirements of section 571.122 of the Government Code.

#### §12.6. File Date for Purposes of Commission Response Deadline.

For purposes of section 571.123 of the Government Code, the file date for a complaint is the date the complaint is received at the agency office.

#### §12.7. Confidentiality.

(a) The commission and its employees shall not communicate any information about a sworn complaint, including whether or not a complaint has been filed, to any person other than the respondent, the complainant, and a witness or potential witness identified by the respondent, the complainant, or another witness or potential witness.

(b) Confidentiality under section 571.140 of the Government Code may be waived only if the complainant and each respondent named in the complaint provide a verified, written waiver of confidentiality to the executive director.

#### §12.9. Compliance with Open Meetings Law and Open Records Law.

(a) The Open Meetings Law, chapter 551 of the Government Code, does not apply to a meeting or decision of the commission in connection with a complaint until written notice of a formal hearing on the complaint is sent to the respondent and complainant.

(b) The Open Records Law, chapter 552 of the Government Code, does not apply to information relating to a complaint until written notice of a formal hearing on the complaint is sent to the respondent and complainant.

#### §12.11. Delegation to Executive Director.

Any duty or power of the commission relating to a complaint that does not require a commission vote is delegated to the executive director.

#### §12.13. Representation by Counsel.

(a) A respondent has the right to be represented by counsel retained by the respondent in any proceeding of a complaint.

(b) Counsel representing a respondent shall enter an appearance with the commission that contains the counsel's mailing address, email address, and telephone and fax numbers. If the respondent's counsel is not licensed to practice law in Texas, the representative must show authority to appear as the respondent's counsel.

(c) The commission may, through the approval of its presiding officer, admit an attorney who is a resident of and licensed to practice law in another state, and who is not an active member of the State Bar of Texas, to represent a respondent before the commission if the nonresident attorney complies with the requirements of Tex. Gov't Code §82.0361 and Rule XIX of the Rules Governing Admission to the Bar of Texas and files a motion, accompanied by proof of compliance with those provisions, with the commission requesting to be admitted to represent a respondent.

(d) This rule does not allow a person to engage in the unauthorized practice of law.

#### §12.15. Appearance of Complainant at Hearing.

The commission may grant a complainant the opportunity to be heard at a hearing.

#### **§12.19.** Agreements to be in Writing.

No stipulation or agreement with respect to any matter in a complaint shall be effective unless it has been:

(1) reduced to writing and signed by each person making the stipulation or agreement, or by that person's authorized representative; or

(2) entered into the record during the course of a hearing.

#### §12.21. Notice.

(a) A notice required to be sent to a complainant under chapter 571 of the Government Code shall be sent to the address most recently provided by the complainant.

(b) A notice required to be sent to a respondent under chapter 571 of the Government Code shall be sent to the address provided by the complainant or to the address most recently provided by the respondent.

(c) A person entitled to receive notice may waive that right by filing a written waiver with the executive director.

(d) A respondent or complainant in a complaint may waive the right under section 571.032 of the Government Code to receive written notices related to the complaint by registered or certified mail, restricted delivery, return receipt requested, and may agree to receive written notices related to the complaint by first class mail, electronic mail, or other means.

#### §12.23. Hearing in Respondent's Absence.

If a respondent fails to appear at a hearing, the commission may proceed in the respondent's absence and may find credible evidence of the violations alleged in the complaint and may issue a final order imposing a civil penalty.

#### §12.25. Waiver of Hearing.

A respondent may waive the right to a hearing.

#### §12.27. Deadline Extension.

The executive director may extend a deadline pursuant to §571.136 of the Government Code.

#### §12.28. Production of Documents During Preliminary Review.

(a) Before applying for the commission to issue a subpoena under §571.137(a-1) of the Government Code, commission staff must send to the person from whom records are sought a written request for the production or inspection of documents or other tangible things that:

(1) specifies the items to be produced or inspected, either by individual item or by category, and describes with reasonable particularity each item and category; and

(2) provides a reasonable amount of time, but not less than 30 days, to comply with the request.

(b) The person from whom records are sought must produce or allow the inspection of documents or other tangible things within the person's possession, custody or control within the time provided in the request, or submit in writing, as appropriate:

(1) objections to those records that are unreasonable, improper, or unnecessary to investigate the complaint; or

(2) that, after a diligent search, no items have been identified that are responsive to the request.

(c) Commission staff shall provide to the commission any response it receives to its request for production or inspection when applying for a subpoena under §571.137(a-1) of the Government Code.

#### §12.29. Subpoenas Issued by Commission.

(a) A subpoena issued under §571.137 of the Government Code shall specify the date, time, place, and manner for execution of the subpoena.

(b) A subpoena issued under section 571.137 of the Government Code that requires a person to provide testimony shall be served on that person at least 10 business days before the date the subpoena is to be executed.

#### §12.30. Subpoenas Issued by Counsel for the Respondent.

(a) This section applies only to subpoen issued by a respondent's counsel under section 571.125(f) (concerning the issuance of a subpoen a for a witness in a preliminary review hearing) or 571.130(f) (concerning the issuance of a subpoen a for a witness in a formal hearing) of the Government Code.

(b) A subpoena must be issued in the name of "The State of Texas" and must:

(1) state the sworn complaint numbers for the sworn complaints at issue in the hearing at which the witness is summoned to appear;

(2) state that the subpoena pertains to a sworn complaint proceeding before the Texas Ethics Commission;

(3) state the date on which the subpoena is issued;

(4) identify the person to whom the subpoena is directed;

(5) state the time and place of the preliminary review hearing or formal hearing at which the subpoena directs the person to appear;

(6) identify the respondent at whose instance the subpoena is issued and the respondent's attorney of record;

(7) specify with reasonable particularity any documents with which the person to whom the subpoena is directed shall appear;

(8) state the text of § 12.31(i) of this chapter; and

(9) be signed by the attorney issuing the subpoena.

(c) A subpoena must command the person to whom it is directed to appear and give testimony at:

(1) a preliminary review hearing; or

(2) a formal hearing.

(d) A subpoena may only direct a person to appear, with or without documents, and give testimony at a preliminary review hearing or formal hearing before the commission.

(e) A subpoena may be issued only by the counsel of record for a respondent in a sworn complaint proceeding before the commission against that respondent.

(f) Service.

(1) Manner of service. A subpoena may be served at any place within the State of Texas by any sheriff or constable of the State of Texas, or any person who is not a party and is 18 years of age or older. A subpoena must be served by delivering a copy to the witness and tendering to that person any fees required by law. If the witness is a party and is represented by an attorney of record in the sworn complaint proceeding, the subpoena may be served on the witness's attorney of record.

(2) Deadline for service. A subpoena must be served upon the person required to appear at least 21 days before the preliminary review hearing or formal hearing at which the person is required to appear. The subpoena and proof of service must be filed with the commission within three days of its service on the person required to appear.

(3) Proof of service. Proof of service must be made by filing either:

(A) the witness's signed written memorandum attached to the subpoena showing that the witness accepted the subpoena; or

(B) a statement by the person who made the service stating the date, time, and manner of service, and the name of the person served.

(g) Response.

(1) Except as provided in this subsection, a person served with a subpoena must comply with the command stated therein unless discharged by the commission or by the party summoning such witness. A person commanded to appear and give testimony must remain at the place of hearing from day to day until discharged by the commission or the party summoning the witness.

(2) If a subpoena commanding testimony is directed to a corporation, partnership, association, governmental agency, or other organization, and the matters on which examination is requested are described with reasonable particularity, the organization must designate one or more persons to testify on its behalf as to matters known or reasonably available to the organization.

(3) A person commanded to appear with documents must produce the documents as they are kept in the usual course of business or must organize and label them to correspond with the categories in the demand.

(4) A person commanded to appear at a hearing must file any motion to quash the subpoena or objection to a requirement to appear with certain documents with the commission no later than the 14th day before the hearing at which the person is directed to appear. Commission staff may move to quash a subpoena or object to appearance with certain documents in the same manner as the person commanded to appear by the subpoena. The filer of a motion to quash or objection to a requirement to appear with certain documents must serve the motion or objection on the proponent of the subpoena in person, by mail, by commercial delivery service, by fax, by email, or by other such manner as the presiding officer of the commission may direct, no later than the deadline for filing the motion to quash or objection to appearance with documents with the commission. After affording commission staff and the person commanded to appear an opportunity to move to quash the subpoena or object to appearance with certain documents, and affording the proponent of the subpoena an opportunity to respond to the motion to quash or objection to appearance with documents, the commission's presiding officer shall rule on a motion to quash or objection to appearance with documents.

(5) A person commanded to attend and give testimony, or to produce documents or things, at a preliminary review hearing or formal hearing may object to giving testimony or producing documents at the time and place specified for the hearing, rather than under subsection (g)(4) of this section.

(6) A party's appearance with a document in response to a subpoena directing the party to appear with the document authenticates the document for use against that party in any proceeding before the commission unless the party appearing with the document objects to the authenticity of the document, or any part of it, at the time of the party's appearance, stating the specific basis for objection. An objection must have a good faith factual and legal basis. An objection made to the authenticity of only part of a document does not affect the authenticity of the remainder. If objection is made, the party attempting to use the document should be given a reasonable opportunity to establish its authenticity. The requirement that the commission provide a reasonable opportunity to establish the document's authenticity may be satisfied by the opportunity to present a witness to authenticate the document at a subsequent hearing before the commission.

(h) A counsel for a respondent issuing a subpoena must take reasonable steps to avoid imposing undue burden or expense on the person served. In ruling on a motion to quash or objection to appearance with documents, the presiding officer must provide a person served with a subpoena an adequate time for compliance, protection from disclosure of privileged material or information, and protection from undue burden or expense. The presiding officer may impose reasonable conditions on compliance with a subpoena, including compensating the witness for undue hardship.

#### §12.31. Conduct and Decorum.

(a) Parties, representatives, and other participants at a hearing shall conduct themselves with dignity, show courtesy and respect for one another and for the commission, and follow any additional guidelines of decorum prescribed by the presiding officer, including adherence to the amount of time allotted for the hearing. Attorneys shall adhere to the standards of conduct in the Texas Lawyer's Creed promulgated by the Supreme Court of Texas and the Court of Criminal Appeals and the Texas Disciplinary Rules of Professional Conduct promulgated by the Supreme Court of Texas.

(b) Attorneys should advise their clients and witnesses of the applicable rules of conduct and decorum.

(c) All objections, arguments, and other comments by parties shall be directed to the commission and not to an opposing party.

(d) While a party is addressing the commission or questioning a witness, any other party shall not interrupt for any purpose except to make a valid objection.

(e) Parties shall not approach the dais without leave of the presiding officer and must not lean on the dais.

(f) Parties shall remain seated at the counsel table at all times except:

- (1) when addressing the commission; and
- (2) whenever it may be proper to handle documents, exhibits, or other evidence.

(g) Parties must question witnesses and deliver arguments to the commission while seated at the counsel table or standing at the lectern. If a party seeks to question or argue from another location, leave of the presiding officer must be requested and granted.

(h) Parties must request leave of the presiding officer to conduct a demonstration.

(i) The presiding officer may take appropriate action to maintain and enforce proper conduct and decorum, including:

- (1) issuing a warning;
- (2) sanctioning a party pursuant to §12.33 of this chapter;
- (3) excluding persons from the proceeding;
- (4) recessing the proceeding; and
- (5) clearing the hearing room of persons causing a disruption.

#### §12.33. Sanctioning Authority.

(a) The presiding officer has the authority to impose appropriate sanctions against a party or its representative for:

(1) filing a motion or pleading that is deemed by the presiding officer to be groundless and brought:

- (A) in bad faith;
- (B) for the purpose of harassment; or

(C) for any other improper purpose, such as to cause unnecessary delay or needless increase in the cost of the proceeding;

(2) abuse of the discovery process in seeking, making, or resisting discovery;

- (3) failure to comply with a commission order; or
- (4) violating §12.31 of this chapter.

(b) The presiding officer may issue an order imposing sanctions when justified by party or representative behavior described in subsection (a) of this section and after notice and opportunity for hearing. Sanctions may include:

(1) disallowing or limiting further discovery by the offending party;

(2) charging all or part of the expenses of discovery against the offending party or its representatives;

(3) deeming designated facts be admitted for purposes of the proceeding;

(4) refusing to allow the offending party to support or oppose a claim or defense or prohibiting the party from introducing designated matters into the record;

(5) disallowing in whole or in part requests for relief by the offending party and excluding evidence in support of those requests; and

(6) striking motions or testimony in whole or in part.

#### §12.34. Agreed Orders.

(a) The commission may enter into an agreed order with a respondent to resolve and settle a complaint filed against the respondent, including an assurance of voluntary compliance, a notice of reporting error, or an agreed order and resolution.

(b) An assurance of voluntary compliance:

(1) resolves a sworn complaint:

(A) with no determination that a violation within the jurisdiction of the commission has occurred, if entered into before a preliminary review hearing is completed; or

(B) with a determination that all violations within the jurisdiction of the commission, when viewed as a whole in consideration of any mitigating action taken by the respondent, are technical or de minimis; and

(2) may include a civil penalty.

(c) A notice of reporting error resolves a complaint with a determination that all violations within the jurisdiction of the commission are reporting errors that do not materially defeat the purpose of disclosure and may include a civil penalty in the form of an assessment fee.

(d) An agreed order and resolution resolves a sworn complaint with a determination that one or more violations within the jurisdiction of the commission occurred and may include a civil penalty.

#### §12.35. Frivolous Complaint.

In deciding if a complaint is frivolous, the commission will be guided by the Texas Rules of Civil Procedure, Rule 13, and interpretations of that rule, and may also consider:

(1) the timing of the complaint with respect to when the facts supporting the alleged violation became known or should have become known to the complainant, and with respect to the date of any pending election in which the respondent is a candidate or is involved with a candidacy, if any;

(2) the nature and type of any publicity surrounding the filing of the complaint, and the degree of participation by the complainant in publicizing the fact that a complaint was filed with the commission;

(3) the existence and nature of any relationship between the respondent and the complainant before the complaint was filed;

(4) if respondent is a candidate for election to office, the existence and nature of any relationship between the complainant and any candidate or group opposing the respondent;

(5) any evidence that the complainant knew or reasonably should have known that the allegations in the complaint were groundless; and

(6) any evidence of the complainant's motives in filing the complaint.

## §12.36. Assessment of Civil Penalty.

(a) The commission shall consider the factors listed in §571.177 of the Government Code when assessing a civil penalty against a respondent, including whether the respondent timely responds to written questions or subpoenas.

(b) The commission may consider the fine amounts established by chapter 18 of this title in determining the amount of a fine to be assessed in a sworn complaint proceeding.

(c) The commission may consider a late or corrected report or corrective action to be a mitigating factor in determining the amount of a fine, if any.

## Subchapter B. FILING AND INITIAL PROCESSING OF A COMPLAINT

## §12.51 Non-Complying Complaint.

(a) A complaint does not comply with the requirements of §571.122 of the Government Code if:

(1) the complaint includes an allegation of conduct that the commission has previously determined is not a violation of a statute or rule within the commission's jurisdiction; and

(2) before the complaint was filed the complainant was provided notice that such conduct is an invalid basis for an allegation.

(b) A complainant has been provided notice if, before a complaint is filed, the commission states the basis for rejecting an allegation in a written communication mailed to the complainant at the last address provided to the commission by the complainant, or in a written communication transmitted by electronic mail to the complainant at the last electronic mail address provided to the commission by the complainant.

(c) A complaint determined to be non-complying under subsection (a) of this section is presumed to be a frivolous or bad faith complaint.

#### §12.52. Response to Notice of Complaint.

(a) The response required by section 571.1242 of the Government Code must:

- (1) be in writing;
- (2) admit or deny the allegations set forth in the complaint; and
- (3) be signed by the respondent.

(b) If a respondent does not submit a response within the time period prescribed by section 571.1242 of the Government Code, the commission may issue an order imposing a civil penalty for failure to file a response.

(c) If a respondent does not submit a response that satisfies the requirements of subsection (a) of this section, the commission may issue an order imposing a penalty for failure to file a complete response.

### §12.53. Commission Initiated Complaint.

(a) Commission staff may gather or present documents or evidence, make recommendations, and otherwise communicate outside the presence of the respondent with commissioners in contemplation of, or in preparation for, a commission initiated preliminary review. Commissioners may request documents, evidence, or recommendations, and otherwise communicate with commission staff outside the presence of the respondent, in contemplation of, or in preparation for, a commission recommendations, and otherwise communicate with commission staff outside the presence of the respondent, in contemplation of, or in preparation for, a commission initiated preliminary review.

(b) A preliminary review initiated by the commission under section 571.124(b) of the Government Code is deemed to be a complaint for purposes of all further proceedings under chapter 571 of the Government Code and of this chapter.

(c) Documents or evidence gathered by the commission and commission staff in contemplation of, or in preparation for, a commission initiated preliminary review are related to the processing of a preliminary review or motion for the purposes of sections 571.139 and 571.140 of the Government Code.

(d) Discussions between the commission and commission staff regarding gathering documents or evidence in contemplation of, or in preparation for, a commission initiated preliminary review are related to the processing of a preliminary review or motion for the purposes of sections 571.139 and 571.140 of the Government Code.

#### §12.59. Description of Violation.

(a) If a complaint does not include the specific rule or provision of law alleged to have been violated, the complaint must clearly and concisely describe facts that, if true, would constitute a violation of a law administered and enforced by the commission.

(b) A complaint that erroneously cites a specific rule or provision of law is nonetheless sufficient if the correct citation can reasonably be ascertained by the commission. When a complaint erroneously cites a specific rule or provision of law, the commission shall cite the correct rule or provision of law in the notice provided to the respondent.

### §12.61. Statement of Facts.

(a) The alleged facts must provide sufficient detail to reasonably place the respondent on notice of the law violated and of the manner and means by which the violation allegedly occurred and to afford the respondent a basis on which to prepare a response.

(b) The facts alleged must disclose the specific date or dates on which the alleged violation occurred, if that date is known to the complainant. If the complainant is unable to provide a specific date for the violation, the complaint must disclose a specific period of time during which the alleged violation may have occurred.

(c) The facts alleged may adopt by reference the content of documents submitted with the complaint. However, the allegations must reasonably identify those portions of the document that are relevant to the alleged violation.

## §12.67. Copies and Documents Provided by the Commission.

(a) The executive director shall provide to a respondent copies of all documents submitted with a complaint unless the executive director determines that certain supporting documents are redundant.

(b) If the executive director determines that supporting documents are redundant, the notice of complaint shall describe the documents and inform the respondent that the documents are available for examination by the respondent at the agency office and that copies will be provided to the respondent on request at the respondent's expense.

(c) Whenever the executive director believes a complainant or respondent may agree a document is not needed, the executive director may ask if the complainant or respondent will waive the right to receive the document.

## Subchapter C. INVESTIGATION AND PRELIMINARY REVIEW

#### §12.81. Technical, Clerical, or *De Minimis* Violations.

(a) A technical, clerical, or de minimis violation for purposes of §571.0631 of the Government Code may include a first-time allegation against a respondent for:

(1) Typographical or incomplete information on a campaign finance report that is not misleading and does not substantially affect disclosure;

(2) Failure to include a disclosure statement on political advertising;

(3) Failure of a non-incumbent to use the word "for" in a campaign communication that is not otherwise misleading;

(4) Failure to include the highway right-of-way notice on political advertising;

(5) Using a representation of the state seal by a person who is not an officeholder in political advertising that is not otherwise misleading;

(6) Filing a late campaign finance report that is not a report due 30 or 8 days before an election or a special pre-election report, and the alleged violations do not substantially affect disclosure;

(7) Filing an incomplete or corrected campaign finance report that is not a report due 30 or 8 days before an election or a special pre-election report if:

(A) the total amount of incomplete or incorrectly reported political contributions does not exceed the lesser of 10% of the total amount of political contributions on the corrected report, or \$5,000;

(B) the total amount of incomplete or incorrectly reported political expenditures does not exceed the lesser of 10% of the total amount of political expenditures on the corrected report, or \$5,000; or

(C) the total amount of incomplete or incorrectly reported political contributions or political expenditures does not exceed the amount of the filing fee for a place on the ballot for the office sought or held by the respondent during the period covered by the report at issue, or, if there is not a set filing fee, \$500;

(8) Filing an incomplete or corrected campaign finance report if the incomplete or corrected information is not misleading and does not substantially affect disclosure, including:

(A) the filer's full name, address, office sought, or office held;

(B) the identity and date of the election for which the report is filed;

(C) the campaign treasurer's full name, address, or telephone number;

(D) the full name of each identified candidate or measure or classification by party of candidates supported or opposed by a political committee;

(E) the full name of each identified officeholder or classification by party of officeholders assisted by a political committee;

(F) the amount of total political contributions maintained as of the last day of the reporting period, if the error is a de minimis error as defined by §20.50 of this title;

(G) the purpose of a political expenditure; or

(H) the period covered by the report;

(9) Failure to timely file a campaign treasurer appointment if, before filing the campaign treasurer appointment, the total amount of political contributions accepted does not exceed \$2,500 and the total amount of political expenditures made or authorized does not exceed \$2,500;

(10) Failure to disclose information related to an out-of-state political committee required by \$20.29 or \$22.7 of this title if the total amount of political contributions accepted from the committee does not exceed \$10,000 and the contributions are otherwise properly disclosed;

(11) Failure to disclose the principal occupation, job title, or employer of a contributor if the total amount of political contributions accepted from the contributor does not exceed \$15,000 and the contributions are otherwise properly disclosed;

(12) As a general-purpose committee, making a political contribution to another generalpurpose committee without including in its campaign treasurer appointment the name of the recipient committee before making the contribution, if the contributing committee properly disclosed the contribution;

(13) Failure to file a termination report required by \$20.317 or \$20.417 of this title if the period covered by the termination report is included in a subsequently filed report;

(14) Filing a campaign finance report without using the form prescribed by the commission if the report:

(A) discloses all the information required by chapter 254 of the Election Code and this title;

(B) is substantially similar in size and format to the form prescribed by the commission; and

(C) is not misleading and does not substantially affect disclosure;

(15) Making a political contribution prohibited by §253.1611, Election Code, if the contribution does not exceed the limits by more than \$1,000 and the amount in excess is returned to the contributor; or

(16) Failure to timely respond to a sworn complaint if the response is no more than 30 days late and the respondent shows good cause for the late response.

(b) A technical, clerical, or de minimis violation for purposes of §571.0631 of the Government Code may include allegations against a respondent for:

(1) Typographical or incomplete information on a campaign finance report that is not misleading or does not substantially affect disclosure;

(2) Filing an incomplete or corrected campaign finance report if:

(A) the total amount of incomplete or incorrectly reported political contributions does not exceed the lesser of 5% of the total amount of political contributions on the corrected report, or \$2,500; or

(B) the total amount of incomplete or incorrectly reported political expenditures does not exceed the lesser of 5% of the total amount of political expenditures on the corrected report, or \$2,500; or

(3) Filing an incomplete or inaccurate campaign finance report by a general-purpose committee if, during the period covered by the report and during each of the two reporting periods preceding the period covered by the report, the committee did not:

(A) accept political contributions totaling \$3,000 or more;

(B) accept political contributions from a single person totaling \$1,000 or more; or

(C) make political expenditures totaling \$3,000 or more.

(c) During the review of a sworn complaint under Chapter 571, Subchapter E of the Government Code, if the executive director determines that all the alleged violations are technical, clerical, or de minimis under subsection (a) of this section, the executive director may enter into an assurance of voluntary compliance with the respondent. Before entering into an assurance of voluntary compliance, the executive director may require a respondent to correct the violations.

(d) During the review of a sworn complaint under Chapter 571, Subchapter E of the Government Code, if the executive director determines that all the alleged violations are technical, clerical, or de minimis under subsection (b) of this section, the executive director may enter into an agreed resolution with the respondent. Before entering into an agreed resolution, the executive director may require a respondent to correct the violations.

(e) An assurance of voluntary compliance or an agreed resolution entered into under this section is confidential under §571.140 of the Government Code.

(f) An assurance of voluntary compliance or an agreed resolution entered into under this section may include a penalty not to exceed \$500.

## §12.83. Preliminary Review.

(a) A complainant or respondent must respond to written questions not later than 15 business days after receiving the written questions. The executive director may grant an extension of the time period for good cause shown.

(b) If the commission staff submits written questions to a respondent, the 120-day deadline for the commission to propose an agreement to the respondent or dismiss the complaint (provided in section 571.1242(g) of the Government Code) is tolled beginning on the date the commission

sends the written questions and resets on the date the commission receives the respondent's written response.

(c) If the commission staff applies to the commission for the issuance of a subpoena pursuant to section 571.137(a-1) of the Government Code, the 120-day deadline for the commission to propose an agreement to the respondent or dismiss the complaint (provided in section 571.1242(g) of the Government Code) is tolled beginning on the date the staff applies to the commission for the subpoena and resets on either:

(1) the date the commission rejects the staff's application for a subpoena;

(2) the date the person to whom the subpoena is directed complies with the subpoena; or

(3) the date the commission receives a final ruling on a person's failure or refusal to comply with a subpoena that is reported to a district court pursuant to section 571.137(c) of the Government Code.

(d) If the commission staff proposes to a respondent an agreement to settle a complaint that would be effective upon approval by the commission and the respondent, the 120-day deadline for the commission to propose an agreement to the respondent or dismiss the complaint (provided in section 571.1242(g) of the Government Code) is met. If a respondent approves a proposed agreement, commission staff must submit the proposed agreement to the commission to seek final approval at the next scheduled commission meeting. If a respondent rejects a proposed agreement, the matter shall be set for a preliminary review hearing at the next commission meeting for which notice has not yet been posted. If a respondent rejects a proposed agreement within 45 days before the date of a commission meeting, the matter shall be set for a preliminary review hearing at the next commission meeting at the next commission meeting at the next commission meeting at the next scheduled.

(e) During a preliminary review, commission staff may present documents or evidence, make recommendations, or otherwise communicate with commissioners outside the presence of the respondent for the purpose of investigating and resolving a sworn complaint.

(f) Commission staff may not communicate with a commissioner outside the presence of the respondent for the purpose of influencing a decision on a pending sworn complaint after the complaint has been scheduled for a preliminary review hearing and notice of the hearing has been sent to the respondent.

## Subchapter D. PRELIMINARY REVIEW HEARING

## §12.84. Notice of Preliminary Review Hearing.

(a) Commission staff shall provide notice of a preliminary review hearing to a respondent and complainant at least 45 days before the date of the hearing and must include:

(1) the date, time, place, and nature of the hearing;

(2) a statement of the legal authority and jurisdiction under which the hearing is to be held;

(3) a reference to the particular sections of the statutes and rules involved; and

(4) a short and plain statement of the factual matters asserted.

(b) Commission staff shall provide to a respondent at least 30 days before the date of the hearing:

(1) a list of proposed witnesses to be called at the hearing and a brief statement as to the nature of the testimony expected to be given by each witness to be called at the hearing; and

(2) copies of all documents expected to be used or introduced as exhibits at the hearing.

(c) The respondent shall provide to commission staff the contents described by subsections (b)(1) and (b)(2) of this section at least 30 days before the date of the hearing. If a respondent or commission staff fail to comply with this section, the commission may reschedule the hearing or proceed with the hearing and exclude at the hearing evidence, documents, and testimony provided by the respondent or commission staff, as applicable, but such failure may be excused upon a showing of good cause.

### §12.85. Preliminary Review Hearing.

(a) Commission staff and the respondent may present any relevant evidence at a preliminary review hearing, including examination and cross-examination of witnesses.

(b) Commission staff and the respondent may present an opening and closing statement at a preliminary review hearing.

#### §12.86. Motions for Continuance.

(a) Contents of a motion for continuance. A request to postpone a preliminary review hearing must be in writing and include the specific reasons and supporting evidence for the continuance and the dates of any previous motions for continuance.

(b) Date of filing. Motions for continuance must be received by the Commission no later than 21 days before the date of the proceeding or must provide good cause with supporting evidence for presenting the motion after that time. If the presiding officer finds good cause has been demonstrated, the presiding officer may consider a motion that is not timely filed.

(c) Responses to motions for continuance. Responses to motions for continuance must be in writing and include the date the complaint was filed and the number of previous requests to postpone filed in the case. Unless otherwise ordered or allowed by the presiding officer, responses to motions for continuance must be made no later than seven business days after receipt of the motion.

(d) Rulings on motions for continuance. A motion for continuance is not granted until it has been ruled on by the presiding officer, even if the motion is uncontested or agreed. A case is subject to default under §12.23 of this chapter for a party's failure to appear at a scheduled hearing in which a motion for continuance has not been ruled on by the presiding officer.

### **§12.87. Resolution of Preliminary Review Hearing.**

(a) At the conclusion of a preliminary review hearing in which the commission finds credible evidence of a violation:

(1) commission staff shall send to the respondent a proposed resolution within 14 days; and

(2) not later than 30 days after the respondent receives the proposed resolution, or by a later date determined by the commission, commission staff must receive from the respondent:

(A) the proposed resolution signed by the respondent;

(B) a written counter offer; or

(C) a written request that the matter be set for a formal hearing.

(b) If the respondent does not comply with subsection (a)(2) of this section, commission staff may request that the commission order a formal hearing.

(c) Commission staff shall report to the commission any written counter offer, staff's recommendation to accept or reject a counter offer, if any, or any written request that a matter be set for a formal hearing received from the respondent under subsection (a)(2) of this section.

(d) After a written counter offer or a written request that a matter be set for a formal hearing is reported to the commission, the commission by record vote of at least six commissioners shall:

(1) accept the respondent's counter offer, if any; or

(2) determine the complaint cannot be resolved and settled and order a formal hearing.

(e) The executive director shall dismiss a complaint if the commission does not order a formal hearing within 180 days after the conclusion of a preliminary review hearing.

(f) This section may not be construed as limiting the commission's authority to agree to the settlement of a complaint under section 571.121 of the Government Code, including sending a revised proposed resolution to a respondent.

## Subchapter E. FORMAL HEARING

## **Division 1. General Procedures**

#### §12.101. Application and Construction.

The proceedings of a formal hearing shall be conducted in accordance with this subchapter, Chapter 571 of the Government Code, and Subchapters C-H, Chapter 2001, of the Government Code (the Administrative Procedure Act) only to the extent they are consistent with Chapter 571. In the event of a conflict, Chapter 571 controls.

## §12.102. Order of Formal Hearing.

As soon as practicable after the commission orders a formal hearing, the executive director shall provide to the parties to the complaint, and to the complainant, a copy of the commission's decision to order the hearing. The decision shall include the date, time, and place of the hearing and be signed by the presiding officer.

#### §12.103. Notice of Formal Hearing.

(a) Commission staff shall provide notice of a formal hearing to a respondent and complainant at least 60 days before the date of the hearing and must include, in addition to the contents required by section 571.126(b) of the Government Code:

(1) the date, time, place, and nature of the hearing;

(2) a statement of the legal authority and jurisdiction under which the hearing is to be held;

(3) a reference to the particular sections of the statutes and rules involved; and

(4) a short and plain statement of the factual matters asserted.

(b) Commission staff shall provide to a respondent and complainant at least 30 days before the date of the hearing:

(1) a list of proposed witnesses to be called at the hearing and a brief statement as to the nature of the testimony expected to be given by each witness to be called at the hearing; and

(2) copies of all documents expected to be used or introduced as exhibits at the hearing.

(c) The respondent shall provide to commission staff the contents described by subsections (b)(1) and (b)(2) of this section. The contents must be received by commission staff at least 14 days before the date of the hearing. If a respondent or commission staff fail to comply with this section, the commission may reschedule the hearing or proceed with the hearing and exclude at the hearing evidence, documents, and testimony provided by the respondent or commission staff, as applicable, but such failure may be excused upon a showing of good cause.

#### §12.117. Formal Hearing: Venue.

When the commission orders a formal hearing the commission shall decide whether the formal hearing will be held before the commission or before the State Office of Administrative Hearings.

#### §12.119. Resolution after a Formal Hearing.

The executive director shall dismiss a complaint if the commission fails to adopt a motion under section 571.132 of the Government Code. The dismissal shall state the complaint was dismissed because there were insufficient commission votes to find that there was or was not a violation of law.

#### **Division 2. Scheduling, Filing, and Service**

#### §12.121. Prehearing Conferences.

The presiding officer may order that one or more prehearing conferences be held to address any matters prior to the formal hearing, including motions for discovery or subpoenas, motions for sanction, or other matters related to the hearing. The commission shall provide such an order to the parties and the complainant within 5 business days after the decision is made. The order shall include the date, time, and place of the conference and a list of the matters to be addressed at the conference.

### §12.123. Scheduling Orders.

(a) The following deadlines apply to a prehearing conference or formal hearing, as applicable:

(1) All motions must be filed with the commission no later than 30 days before the date of the conference or hearing;

(2) All responses to motions must be filed with the commission no later than 14 days before the date of the conference or hearing; and

(3) All replies to responses must be filed with the commission no later than 7 days before the date of the conference or hearing.

(b) A scheduling order containing the deadlines under this section shall be included with the notice required by section 571.126 of the Government Code. The presiding officer may amend a scheduling order upon the request of a party for good cause shown. A decision by the presiding officer to amend a scheduling order or to deny a motion, response, or evidence shall be issued to the parties to a hearing within 5 business days after the decision is made.

(c) The presiding officer may deny a party's motions, responses, or replies or deny a party's evidence from being admitted into the record of the hearing if the party violates a scheduling order.

#### §12.125. Filing of Documents.

(a) Motions, responses, and other documents in a formal hearing must be filed with the Commission:

(1) by mail addressed to the commission at P.O. Box 12070, Austin, Texas 78711-2070;

(2) by hand-delivery to the commission at 201 East 14th Street, 10th Floor;

(3) by fax to the commission at (512) 463-5777; or

(4) by email to a dedicated filing address.

(b) All documents must clearly indicate the sworn complaint number and the name of the respondent for which it is filed.

(c) Time of filing. With respect to documents filed by mail, fax, or hand-delivery, the time and date of filing shall be determined by the file stamp affixed by the commission. With respect to documents filed by email, the time and date of filing is the electronic time stamp affixed by the commissions email system. Documents received when the commission is closed shall be deemed filed the next business day.

(d) Non-conforming documents. The commission may not refuse to file a document that fails to conform with this chapter. When a filed document fails to conform to this rule, the executive director or presiding officer may identify the errors to be corrected and state a deadline.

#### §12.127. Service of Documents.

(a) Service on all parties. On the same date a document is filed with the commission, a copy shall also be sent to each party or the party's authorized representative by hand-delivery; by regular, certified, or registered mail; by email, upon agreement of the parties; or by fax.

(b) Certificate of service. A person filing a document shall include a certificate of service that certifies compliance with this section.

(1) A certificate of service shall be sufficient if it substantially complies with the following example: "Certificate of Service: I certify that on {date}, a true and correct copy of this {name of document} has been sent to {name of opposing party or authorized representative for the opposing party} by {specify method of delivery, e.g., email, regular mail, fax, certified mail.} {Signature}"

(2) If a filing does not certify service, the commission may:

(A) return the filing;

(B) send a notice of noncompliance to all parties, stating the filing will not be considered until all parties have been served; or

(C) send a copy of the filing to all parties.

(c) Presumed time of receipt of served documents. The following rebuttable presumptions shall apply regarding a party's receipt of documents served by another party:

(1) If a document was hand-delivered to a party, the commission shall presume that the document was received on the date of filing at the commission.

(2) If a document was served by courier-receipted overnight delivery, the commission shall presume that the document was received no later than the next business day after filing at the commission.

(3) If a document was served by regular, certified, or registered mail, or non-overnight courier-receipted delivery, the commission shall presume that it was received no later than three days after mailing.

(4) If a document was served by fax or email before 5:00 p.m. on a business day, the commission shall presume that the document was received on that day; otherwise, the commission shall presume that the document was received on the next business day.

(d) Burden on sender. The sender has the burden of proving date and time of service.

## **Division 3. Powers and Duties of Commission and Presiding Officer**

### §12.131. Powers and Duties of the Presiding Officer.

(a) Presiding officer's authority and duties. The presiding officer shall have the authority and duty to:

- (1) conduct a full, fair, and efficient hearing;
- (2) take action to avoid unnecessary delay in the disposition of the proceeding;
- (3) maintain order; and
- (4) rule on prehearing matters.

(b) Presiding officer's powers. The presiding officer shall have the power to regulate the hearing, and the conduct of the parties and authorized representatives, including the power to:

(1) administer oaths;

(2) take testimony, including the power to question witnesses and to request the presence of a witness from a state agency;

- (3) rule on admissibility and other questions of evidence;
- (4) rule on discovery issues;
- (5) issue orders pursuant to this chapter;
- (6) exclude irrelevant, immaterial, or unduly repetitious testimony;
- (7) reasonably limit the time for presentations of evidence or argument;

(8) order parties to submit legal memoranda and proposed findings of fact and conclusions of law; and

(9) reopen the record when justice requires, if the commission has not issued a final order.

#### §12.133. Orders from the Commission.

(a) The presiding officer has authority to issue orders to control the conduct and scope of the proceeding, including orders to:

- (1) Rule on motions;
- (2) Impose sanctions;
- (3) Establish deadlines;
- (4) Schedule and conduct pre-hearing or post-hearing conferences;
- (5) Require the prefiling of exhibits and testimony;
- (6) Set out requirements for participation in the case; and
- (7) Take other steps conducive to a fair and efficient formal hearing.

(b) Record of rulings. Rulings not made orally at a recorded prehearing conference or hearing shall be in writing and issued to all parties of record.

(c) Consolidation or joinder for hearing. The presiding officer may order that cases be consolidated or joined for hearing if:

(1) there are common issues of law or fact; and

(2) consolidation or joint hearing will promote the fair and efficient handling of the matters.

(d) Severance of issues. The presiding officer may order severance of issues if separate hearings on the issues will promote the fair and efficient handling of the matters.

#### **Division 5. Pleadings and Motions**

#### §12.151. Required Form of Pleadings.

(a) Content generally. Written requests for action in a formal hearing shall be typewritten or printed legibly on  $8-1/2 \ge 11$ -inch paper and timely filed with the commission. Photocopies are acceptable if copies are clear and legible. All filings shall contain or be accompanied by the following:

- (1) the name of the party seeking action;
- (2) the sworn complaint number;
- (3) the parties to the case and their status as commission staff or respondent;

(4) a concise statement of the type of relief, action, or order desired and identification of the specific reasons for and facts to support the action requested;

(5) a certificate of service, as required by 12.127(b)(1) of this chapter;

- (6) any other matter required by statute or rule; and
- (7) the signature of the submitting party or the party's authorized representative.

(b) Amendment or supplementation of pleadings. A party may amend or supplement its pleadings as follows:

(1) If a notice of a hearing or other documents provided to the complainant or respondent under section 571.126(b)(2) of the Government Code contain a material defect, the commission may correct the notice or other document and deliver it to the complainant and respondent as soon as practicable and in the same manner as the original notice. If the respondent does not receive the correction at least 10 days before the date of the hearing, the presiding officer may by order reschedule the hearing. The executive director shall notify the parties and the complainant of the date, time, and place of the hearing as soon as practicable.

(2) As to all other matters in a pleading, an amendment or supplementation that includes information material to the substance of the hearing, requests for relief, changes to the scope of the hearing, or other matters that unfairly surprise other parties may not be filed later than seven days before the date of the hearing, except by agreement of all parties or by permission of the presiding officer.

#### §12.153. Motions, Generally.

(a) Purpose and effect of motions. To make a request, including to obtain a ruling, order, or any other procedural relief, a party shall file a written motion. The motion shall describe specifically the action requested and the basis for the requested action. Unless otherwise specified in this chapter, a motion is not granted until it has been ruled on by the presiding officer or commission, as applicable, even if the motion is uncontested or agreed.

(b) General requirements for motions. Except as provided in this chapter, or unless otherwise ordered by the presiding officer, all motions shall:

- (1) be filed in writing no later than the applicable deadline; except, for good cause demonstrated in the motion, the presiding officer may consider a motion filed after that time or presented orally at a hearing;
- (2) include a certificate of conference that complies substantially with one of the following examples:
  - (A) Example one: "Certificate of Conference: I certify that I conferred with {name of other party or other party's authorized representative} on {date} about this motion. {Succinct statement of other party's position on the action sought and/or a statement that the parties negotiated in good faith but were unable to resolve their dispute before submitting it to the commission for resolution.} Signature."; or

- (B) Example two: "Certificate of Conference: I certify that I made reasonable but unsuccessful attempts to confer with {name of other party or other party's authorized representative} on {date or dates} about this motion. {Succinctly describe these attempts.} Signature."
- (3) include a reference in the motion's title to a request for a hearing on the motion if the moving party seeks a hearing; and
- (4) include a proposed order sought by the moving party.

(c) Responses to motions. Except as otherwise provided in this chapter or as ordered or allowed by the commission, responses to motions shall be in writing and filed by the applicable deadline. However, if the presiding officer finds good cause has been shown, responses to written motions may be presented orally at hearing.

(d) Other motions. In addition, other types of motions are addressed in other sections of this chapter. If there is a conflict between this section and a requirement found in another section relating to a specific type of motion, the more specific provision applies.

### §12.155. Motions for Continuance and to Extend Time.

(a) Contents of a motion for continuance. A request to postpone or delay a hearing or prehearing conference shall include:

(1) a statement of the number of motions for continuance previously filed in the case by each party;

(2) the specific reason for the continuance;

(3) whether the movant is available if the hearing or prehearing conference is continued to the next tentatively scheduled commission meeting;

(4) a certificate of conference that complies substantially with one of the examples set out in 12.153(b)(2) of this subchapter.

(b) Contents of a motion to extend time. A request for more time to file a document or respond to discovery shall include:

(1) a statement of the number of extension requests previously sought in the case by the movant;

(2) the specific reason for the request;

(3) a proposed date for the deadline the movant seeks to extend; and

(4) a certificate of conference that complies substantially with one of the examples set out in 12.153(b)(2) of this subchapter.

(c) Date of filing. Motions for continuance or to extend time shall be filed no later than five days before the date of the proceeding or deadline at issue or shall state good cause for presenting the

motion after that time. If the presiding officer finds good cause has been demonstrated, the presiding officer may consider a motion filed after that time or presented orally at the proceeding.

(d) Date of service. Motions for continuance or extension shall be served in accordance with \$12.127 of this chapter. However, a motion for continuance that is filed five days or less before the date of the proceeding shall be served:

(1) by hand-delivery, fax, or email on the same day it is filed with the commission, if feasible; or

(2) if same-day service is not feasible, by overnight delivery on the next business day.

(e) Responses to motions for continuance. Responses to motions for continuance shall be in writing, except a response to a motion for continuance made on the date of the proceeding may be presented orally at the proceeding. Unless otherwise ordered or allowed by the presiding officer, responses to motions for continuance shall be made by the earlier of:

- (1) three days after receipt of the motion; or
- (2) the date and time of the proceeding.

(f) Responses to motions to extend time. Unless otherwise ordered by the presiding officer, responses to motions for extension of a deadline are due three days after receipt of the motion.

(g) A motion for continuance or extension of time is not granted until it has been ruled on by the presiding officer, even if the motion is uncontested or agreed. A case is subject to default or dismissal for a party's failure to appear at a scheduled hearing in which a motion for continuance has not been ruled on by the presiding officer, even when the motion is agreed or unopposed.

## **Division 6. Hearings and Prehearing Conferences**

#### §12.161. Time Allotted to Parties.

The presiding officer shall determine the amount of time allotted to each party for the presentation of its case to the commission. Upon request of a party, the presiding officer may adjust the time allotted for good cause shown or for a violation of a rule under this chapter. A determination shall be made by order or orally during the course of the proceeding.

#### §12.163. Presentation of Evidence.

(a) Presentation by parties. After the resolution of all prehearing matters, each party shall make its presentation during the formal hearing. Commission staff shall make the first opening statement. The respondent or respondent's authorized representative shall then make an opening statement, should the respondent wish to do so at that time. The respondent may reserve the opening statement until the presentation of the respondent's case.

(b) Following opening statements, commission staff may present evidence in its case. At the conclusion of the presentation of the evidence, commission staff may rest. The respondent or the respondent's authorized representative may then make an opening statement, or, if an opening

statement has already been made, present evidence in its defense of the allegations raised in the notice of formal hearing. At the conclusion of the presentation of evidence by the respondent, the respondent may rest.

(c) After both parties have rested their case, commission staff shall make a closing argument. The respondent may then make a closing argument. Commission staff may then make a reply.

(d) Unless otherwise ordered by the presiding officer, after closing arguments, evidence will be closed and the case will be turned over to the members of the commission for deliberation and decision.

### §12.165. Rules of Evidence.

(a) The Texas Rules of Evidence as applied in a nonjury civil case in district court govern a formal hearing only to the extent consistent with Chapter 571 of the Government Code.

(b) Evidence may be admitted if it meets the standards set out in section 2001.081 of the Government Code.

#### §12.167. Numbering of Exhibits.

(a) Each exhibit to be offered shall first be numbered by the offering party.

(b) Copies of the original exhibit shall be furnished by the party offering the exhibit to the commission and to each party present at the hearing unless otherwise ordered by the presiding officer.

(c) Excluded exhibits. An exhibit excluded from evidence will be considered withdrawn by the offering party and will be returned to the party.

(d) Pre-numbered exhibits may be filed with the commission prior to the formal hearing. Prenumbered exhibits that are not offered and admitted at the hearing will be deemed withdrawn.

#### **Division 7. Disposition of Formal Hearing**

#### §12.171. Standard of Proof.

At a formal hearing, the commission shall determine by preponderance of the evidence whether a violation within the jurisdiction of the commission has occurred.

#### §12.173. Default Proceedings.

(a) If a respondent to whom a notice of hearing with factual allegations is served or provided fails to appear for the hearing, the commission may proceed in the respondent's absence on a default basis.

(b) A default proceeding under this section requires adequate proof of the following:

(1) the notice of hearing to the respondent stated that the allegations listed in the notice could be deemed admitted and that the relief sought in the notice of hearing might be granted by default against the party that fails to appear at the hearing;

(2) the notice of hearing satisfies the requirements of sections 2001.051, 2001.052, and 571.126 of the Government Code; and

(3) the notice of hearing was:

(A) received by the defaulting party; or

(B) as authorized by Chapter 571 of the Government Code and this chapter, sent by regular mail or by certified mail, restricted delivery, return receipt requested, to the party's last known address as shown by the commission's records.

(c) In the absence of adequate proof to support a default, the presiding officer shall continue the hearing and direct commission staff to provide adequate notice of hearing. If adequate notice is unable to be provided, the commission may dismiss the complaint.

(d) Upon receiving the required showing of proof to support a default, the commission may by vote deem admitted the allegations in the notice of hearing and issue a default decision.

## §12.174. Summary Disposition.

(a) Granting of summary disposition. Summary disposition shall be granted on all or part of a complaint's allegations if the allegations, the motion for summary disposition, and the summary disposition evidence show that there is no genuine issue as to any material fact and that the moving party is entitled to a decision in its favor as a matter of law on all or some of the issues expressly set out in the motion. Summary disposition is not permitted based on the ground that there is no evidence of one or more essential elements of a claim or defense on which the opposing party would have the burden of proof at the formal hearing.

(b) Deadlines. Unless otherwise ordered by the presiding officer:

(1) A party may file a motion for summary disposition at any time after the commission orders a formal hearing, but the motion must be filed at least 45 days before a scheduled hearing on the merits.

(2) The response and opposing summary disposition evidence shall be filed no later than 15 days after the filing of the motion.

(c) Contents of Motion. A motion for summary disposition shall include the contents listed below. A motion may be denied for failure to comply with these requirements.

(1) The motion shall state the specific issues upon which summary disposition is sought and the specific grounds justifying summary disposition.

(2) The motion shall also separately state all material facts upon which the motion is based. Each material fact stated shall be followed by a clear and specific reference to the supporting summary disposition evidence.

(3) The first page of the motion shall contain the following statement in at least 12-point, bold-face type: "Notice to parties: This motion requests the commission to decide some or all of the issues in this case without holding an evidentiary hearing on the merits. You have 15 days after the filing of the motion to file a response. If you do not file a response, this case may be decided against you without an evidentiary hearing on the merits."

(d) Responses to motions.

(1) A party may file a response and summary disposition evidence to oppose a motion for summary disposition.

(2) The response shall include all arguments against the motion for summary disposition, any objections to the form of the motion, and any objections to the summary disposition evidence offered in support of the motion.

(e) Summary disposition evidence.

(1) Summary disposition evidence may include deposition transcripts; interrogatory answers and other discovery responses; pleadings; admissions; affidavits; materials obtained by discovery; matters officially noticed; stipulations; authenticated or certified public, business, or medical records; and other admissible evidence. No oral testimony shall be received at a hearing on a motion for summary disposition.

(2) Summary disposition may be based on uncontroverted written testimonial evidence of an interested witness, or of an expert witness as to subject matter concerning which the presiding officer must be guided solely by the opinion testimony of experts, if the evidence is clear, positive and direct, otherwise credible and free from contradictions and inconsistencies, and could have been readily controverted.

(3) All summary disposition evidence offered in support of or in opposition to a motion for summary disposition shall be filed with the motion or response. Copies of relevant portions of materials obtained by discovery that are relied upon to support or oppose a motion for summary disposition shall be included in the summary disposition evidence.

(f) Proceedings on motions.

(1) The presiding officer may order a hearing on a motion for summary disposition and the commission may rule on the motion without a hearing.

(2) The affirmative vote of six commissioners is necessary to grant summary disposition finding a violation by a preponderance of the evidence.

(3) If summary disposition is granted on all contested issues in a case, the record shall close on the date ordered by the presiding officer or on the later of the filing of the last summary disposition arguments or evidence, the date the summary disposition response was due, or the date a hearing was held on the motion. The commission shall issue a final decision and written report, including a statement of reasons, findings of fact, and conclusions of law in support of the summary disposition rendered. (4) If summary disposition is granted on some but not all of the contested issues in a case, the commission shall not take evidence or hear further argument upon the issues for which summary disposition has been granted. The commission shall issue an order:

(A) specifying the facts about which there is no genuine issue;

(B) specifying the issues for which summary disposition has been granted; and

(C) directing further proceedings as necessary. If an evidentiary hearing is held on the remaining issues, the facts and issues resolved by summary disposition shall be deemed established, and the hearing shall be conducted accordingly. After the evidentiary hearing is concluded, the commission shall include in the final decision a statement of reasons, findings of fact, and conclusions of law in support of the partial summary disposition rendered.

### §12.175. Resolution of Formal Hearing.

(a) As provided by section 571.139 of the Government Code, the commission may deliberate in private regarding the resolution of a sworn complaint or motion, including a dismissal of a complaint, a determination of whether a violation within the jurisdiction of the commission has occurred, and an appropriate penalty upon a finding of a violation. As provided by section 2001.061 of the Government Code, the presiding officer may permit the executive director, general counsel, or other employee of the commission who has not participated in a hearing in the complaint for the purpose of using the special skills or knowledge of the agency and its staff in evaluating the evidence.

(b) The presiding officer may announce the commission's decision on the resolution of a sworn complaint or motion after the conclusion of a formal hearing.

(c) The commission should issue a final order within 60 days after the conclusion of a formal hearing.

## Chapter 16. FACIAL COMPLIANCE REVIEWS AND AUDITS.

### §16.1. Definitions.

The following words and terms, when used in this chapter, shall have the following meanings, unless the context clearly indicates otherwise.

(1) Deficiency—An error, omission, inaccuracy, or violation of a law or rule administered and enforced by the commission that is apparent on the face of a statement or report filed with the commission.

(2) Compliance review report—A report sent to a filer detailing deficiencies in a report that is the subject of a facial compliance review.

(3) Facial compliance review—A review conducted under section 571.069 of the Government Code of the information disclosed on a report, randomly selected in accordance with §16.2 of this title, filed with the commission for facial completeness, accuracy, reliability, and compliance with the law.

(4) Report—A personal financial statement, lobby registration, lobby activities report, or campaign finance report filed with the commission.

#### §16.2. Random Selection.

The report subject to a facial compliance review must be randomly selected from a list of all reports filed by a particular filer type for a specific filing deadline.

# **§16.3.** Corrected or Amended Report Filed During a Facial Compliance Review; Late Fines.

(a) A correction filed for the report that is subject to the facial compliance review will not be subject to a late fine if:

(1) The correction is filed not later than the 30th day after the date the filer receives the compliance review report;

(2) The corrected information complies with the law; and

(3) The original report was filed in good faith and without an intent to mislead or misrepresent the information contained in the report.

(b) A late fine will not be assessed for corrections filed to correct reporting errors made in any report filed prior to the report that is subject to the facial compliance review if:

(1) The filer learned of the errors through the facial compliance review;

(2) The correction is filed not later than the 30th day after the date the filer receives the compliance review report;

(3) The corrections comply with the law; and

(4) The original report was filed in good faith and without an intent to mislead or misrepresent the information contained in the report.

(c) A correction filed in accordance with this section will not be considered a prior late offense for purposes of determining the waiver or reduction of a fine under chapter 18 of this title.

# §16.4. Additional Documents and Information Submitted in Response to a Facial Compliance Review; Timeliness.

(a) The commission may request from a filer documentation and other information used by the filer to compile a report that is subject to a facial compliance review.

(b) Documentation and other information requested by the commission is timely submitted if received by the commission not later than the 30th day after the date the filer receives the request for additional documentation.

# §16.5. Commission Initiated Preliminary Review or Audit Resulting from a Facial Compliance Review.

(a) The commission may initiate a preliminary review as authorized by §571.124 of the Government Code or perform a complete audit of a report that is subject to a facial compliance review under §571.069 of the Government Code if:

(1) a correction is not resubmitted to the commission in accordance with §16.3 of this title;

(2) documentation or other information requested by the commission during a facial compliance review is not submitted to the commission in accordance with §16.4 of this title; or

(3) the commission has determined by a vote of at least six commission members that the correction filed in response to a compliance review report, does not comply with the law.

## §16.6. Notice of Audit of Report.

The commission shall notify a filer that the commission will perform a complete audit of a report that is the subject of a facial compliance review not later than the seventh day after the date the commission votes to initiate the audit.

## §16.7. Supporting Documentation in Response to Audit; Timeliness.

(a) A filer must submit to the commission, upon request and where applicable, supporting documentation in the possession, custody, or control of the filer or filer's agents that contains information necessary for filing the report that is subject to the audit, such as:

- (1) bank statements;
- (2) cancelled checks;
- (3) receipts;

- (4) credit card statements;
- (5) invoices;
- (6) loan documents;
- (7) books or ledgers;
- (8) employee timesheets and payroll records;
- (9) certificates of formation or other business documents; and
- (10) real property records.

(b) A filer must submit to the commission the supporting documentation in response to an audit not later than the 30th business day from the date the filer receives notice of the audit.

#### §16.8. Complete Audit Report.

(a) Commission staff must complete a draft audit report not later than the 30th day after the commission receives from the filer the documentation requested under \$16.6 of this title.

(b) The filer must have an opportunity to confer and object in writing to any findings in the draft audit report before it is submitted to the commission for approval.

(c) Commission staff must consider the filer's objections before submitting the draft audit report to the commission for approval.

(d) Upon approval of an audit, the commission shall send to the filer a final audit report that includes:

(1) a notification that the commission has determined the report that was subject to the audit complies with the law; or

(2) required corrective actions that the filer must take to cure any deficiency found in the report that is subject to the audit.

(e) A filer must correct or amend a report to correct all deficiencies identified in a complete audit report not later than the 30th day from the date the filer receives the complete audit report.

#### §16.9. Representation by Attorney.

(a) A filer has the right to be represented by an attorney retained by the filer during a facial compliance review or an audit initiated by the commission as a result of a facial compliance review.

(b) A letter of representation must be submitted to the commission if the filer is represented by an attorney.

### §16.10. Extension of Deadlines.

The executive director may extend all deadlines related to this chapter except as provided by §571.069(a) of the Government Code (relating to when a corrected or amended report is considered filed as of the date the report was originally filed).

### §16.11. Waiver of Delivery by Certified Mail.

A filer may waive the right under §571.032 of the Government Code to receive written notices related to a facial compliance review or audit by registered or certified mail, restricted delivery, return receipt requested, and may agree to receive written notices by first class mail, electronic mail, or other means.

### §16.12. Facial Review of Total Amount of Political Contributions Maintained.

(a) In this section "expected total political contributions maintained" for a report subject to review is the total amount of political contributions maintained disclosed on the previous report and all monetary political contributions, loans, and credits, less all expenditures from political contributions disclosed on the report that is subject to review, excluding the purchase of investments that can be readily converted to cash.

(b) When there is a difference greater than the threshold set by 20.50(c) of this title (relating to Total Political Contributions Maintained) between the total amount of political contributions maintained disclosed in a report and the expected total political contributions maintained, the commission may request from the filer the bank statement showing the balance as of the last day of the reporting period for each account in which political contributions are maintained.

(c) Producing the requested bank statements that show the total amount of political contributions was accurately reported in the report that is subject to review is sufficient to end the review of the total amount of political contributions maintained as disclosed in the report.

#### **CHAPTER 18. GENERAL RULES CONCERNING REPORTS**

#### §18.1. Forms.

(a) The executive director shall prescribe forms for statements and reports required to be filed with the commission.

(b) The executive director may issue a certificate approving a form submitted to the commission for approval if the form:

- (1) provides for the reporting of all information required on the prescribed form;
- (2) is substantially similar in paper size and format to the prescribed form; and
- (3) will not be confusing to those who use the form.

(c) A filer whose form has been approved by the executive director under subsection (b) must submit a new form for approval if information required to be reported has changed since the original form was approved.

(d) A filer who files a report using computer software provided by the commission or using computer software that meets commission specifications for a standard file format must enter data for the report in accordance with the instructions provided for the software.

(e) A filer who files a report using computer software provided by the commission must use the most current version of the software.

#### §18.3. Provision of Forms by Local Filing Authority.

A local filing authority shall make the appropriate form available for use by persons required to file a report with that filing authority.

#### §18.5. Specification of Office.

When a filer is required to identify the office sought by a candidate or held by an officeholder, the filer shall list the title of the public office, including the district and, if the office is an office of a political subdivision, the name of the political subdivision.

#### §18.7. Timely Reports and Complete Reports.

(a) A report is timely if it is complete and is filed by the applicable deadline using the reporting method required by law.

(b) The deadline for any report filed electronically with the commission is midnight Central Time Zone on the last day for filing the report under the law requiring the filing of the report.

(c) A report is late if it is:

(1) incomplete;

(2) not filed by the applicable deadline; or

(3) not filed by computer diskette, modem, or other means of electronic transfer and the filer is required by law to file using one of these methods.

(d) A report filed electronically is not late if:

(1) the commission's office is closed on the deadline and the report is filed by midnight, Central Time Zone, on the next regular business day, excluding a legal holiday, when the commission's office is open; or

(2) the commission cannot accept reports on the deadline because the agency filing system is not accessible or the agency network is inoperable, and the report is filed by midnight, Central Time Zone, on the next regular business day, excluding a legal holiday, that the commission is able to accept reports.

### §18.9. Corrected/Amended Reports.

(a) A filer may correct/amend a report filed with the commission or a local filing authority at any time.

(b) A corrected/amended report must clearly identify how the corrected/amended report is different from the report being corrected/amended.

(c) A filer who files a corrected/amended report must submit an affidavit identifying the information that was corrected/amended.

(d) A corrected/amended report filed with the commission after the original report is due is subject to a late fine as provided by \$18.13 of this title (relating to Fine for a Late Report).

(e) Subsection (d) does not apply to:

(1) a lobby registration or report, other than an activities report, that is corrected/amended not later than the 14th business day after the date the filer became aware of the errors or omissions in the original registration or report;

(2) a semiannual report that is corrected/amended before the eighth day after the original report was filed;

(3) a semiannual report that is corrected/amended on or after the eighth day after the original report was filed if:

(A) the correction/amendment is made before a sworn complaint is filed with regard to the subject of the correction/amendment; and

(B) the original report was made in good faith and without an intent to mislead or misrepresent the information contained in the report;

(4) an 8-day pre-election report that is corrected/amended in accordance with §18.10 of this title (relating to Guidelines for Substantial Compliance for a Corrected/Amended 8-day Pre-election Report);

(5) a report other than an 8-day pre-election report that is corrected/amended not later than the 14th business day after the date the filer learns the report as originally filed is inaccurate or incomplete if:

(A) the errors or omissions were made in good faith; and

(B) the filer files an affidavit stating that the errors or omissions in the original report were made in good faith.

(f) In this section, "8-day pre-election report" has the same meaning assigned by §18.10(c) of this title.

(g) Except as provided by subsections (b) and (c), this section does not apply to a civil penalty assessed through the sworn complaint or facial compliance review process.

# §18.10. Guidelines for Substantial Compliance for a Corrected/Amended 8-day Pre-election Report.

(a) A corrected/amended 8-day pre-election report substantially complies with the applicable law and will not be assessed a late fine under §18.9 of this title (relating to Corrected/Amended Reports) if:

(1) The original report was filed in good faith and the corrected/amended report was filed not later than the 14th business day after the date the filer learned of the errors or omissions; and

(2) The only corrections/amendments needed were to correct the following types of errors or omissions:

(A) a technical, clerical, or de minimis error, including a typographical error, that is not misleading and does not substantially affect disclosure;

(B) an error in or omission of information that is solely required for the commission's administrative purposes, including a report type or filer identification number;

(C) an error that is minor in context and that, upon correction/amendment, does not result in changed monetary amounts or activity disclosed, including a descriptive change or a change to the period covered by the report;

(D) one or more errors in disclosing contributions that, in total:

(i) do not exceed \$2,000; or

(ii) do not exceed the lesser of 10% of the total contributions on the corrected/amended report or \$10,000;

(E) one or more errors in disclosing expenditures that, in total:

(i) do not exceed \$2,000; or

(ii) do not exceed the lesser of 10% of the total expenditures on the corrected/amended report or \$10,000;

(F) one or more errors in disclosing loans that, in total:

(i) do not exceed \$2,000; or

(ii) do not exceed the lesser of 10% of the amount originally disclosed or \$10,000; or

(G) an error in the amount of total contributions maintained that:

(i) does not exceed \$250; or

(ii) does not exceed the lesser of 10% of the amount originally disclosed or \$2,500.

(b) The executive director shall determine whether an 8-day pre-election report as originally filed substantially complies with applicable law by applying the criteria provided in this section.

(c) In this section, "8-day pre-election report" means a report due eight days before an election filed in accordance with the requirements of §20.213(d), 20.325(e), or 20.425(d) of this title (relating to a candidate, a specific-purpose committee, or a general-purpose committee, respectively) and §254.064(c), 254.124(c), or 254.154(c) of the Election Code (relating to a candidate, a specific-purpose committee, or a general-purpose committee, respectively).

## **§18.11** Guidelines for Waiver or Reduction of a Late Fine for a Corrected/Amended 8-day Pre-election Report.

(a) A filer who has filed a corrected/amended 8-day pre-election report may request the executive director to waive or reduce a late fine assessed under §18.9 of this title (relating to Corrected/Amended Reports) by submitting an affidavit to the executive director. The executive director shall waive a late fine if the report meets the criteria in subsection (b) of this section and shall reduce a late fine if the report meets the criteria in subsection.

(b) A late fine for a correction/amendment to an 8-day pre-election report shall be waived if:

(1) The corrected report was filed not later than the fourth day after the original report due date;

(2) The only correction/amendment by a candidate or officeholder was to add to or delete from the outstanding loans total an amount of loans made from personal funds;

(3) The only correction/amendment by a political committee was to add the name of each candidate supported or opposed by the committee, when each name was originally disclosed on the appropriate schedule for disclosing political expenditures; or

(4) The only correction/amendment was to disclose the actual amount of a contribution or expenditure, when:

(A) the amount originally disclosed was an overestimation;

(B) the difference between the originally disclosed amount and the actual amount did not vary by more than 10%; and

(C) the original report clearly included an explanation of the estimated amount disclosed and the filer's intention to file a correction/amendment as soon as the actual amount was known.

(c) A late fine for a correction/amendment to an 8-day pre-election report that does not meet the criteria for a waiver under subsection (b) of this section shall be reduced as follows:

(1) If the corrected/amended report was filed more than four days after the original report due date but was filed before the election day, the late fine is reduced to \$500;

(2) If the corrected/amended report was filed after the election and the amount of the incorrectly reported or unreported activity was more than 10% of the total amount disclosed on the corrected/amended report but did not exceed the lesser of 25% of the total amount of activity, or \$5,000, the late fine is reduced to \$1,000; or

(3) If the amount of the incorrectly reported or unreported activity was more than 40% of the total amount disclosed in the corrected/amended report and the corrected/amended report was filed over a year after the election, the late fine is reduced to 10% of the amount at issue.

(d) A late fine that is reduced under this section will revert to the full amount originally assessed if the reduced fine is not paid on or before the 30th calendar day after the date of the notice informing the filer of the reduction.

(e) A filer may appeal a determination made under this section by submitting a request in writing to the commission.

(1) The request for appeal should state the filer's reasons for requesting an appeal, provide any additional information needed to support the request, and state whether the filer would like the opportunity to appear before the commission and offer testimony regarding the appeal.

(2) After hearing a request for appeal, the commission may affirm the determination made under this section or make a new determination based on facts presented in the appeal.

(f) This section does not apply to a civil penalty assessed through the sworn complaint process or facial compliance review process.

## §18.13. Fine for a Late Report.

(a) Except as provided by subsection (b) or (c) of this section, the fine is \$500 for:

(1) a late report required to be filed with the commission under Election Code chapter 254 or 257 Government Code chapter 305, or Government Code chapter 572; or

(2) a late report filed with the commission under Local Government Code chapter 159, subchapter C.

(b) The fine for a report due eight days before an election is \$500 for the first day the report is late and \$100 for each day thereafter that the report is late, up to a maximum fine of \$10,000.

(c) The fine for the first semiannual report under Section 254.063, 254.123, or 254.153, Election Code, that is required to be filed by a candidate or political committee following the primary or general election is \$500 for the first day the report is late and \$100 for each day thereafter that the report is late, up to a maximum fine of \$10,000.

(d) A fine assessed under this chapter is in addition to any other sanction assessed under other law.

## §18.15. Additional Fine.

In addition to any other fine assessed under this chapter, the commission may vote to impose a fine of \$2,500 against a filer whose report is more than 30 days late and who has not paid the penalty related to that report within 10 days after receiving the commission notice of lateness.

#### §18.17. Report Must be Filed.

(a) The payment of a civil or criminal fine for failure to file a report, or for filing a report late, does not satisfy a filer's obligation to file the report. Late fines continue to accrue until the report is filed.

(b) A filer, other than the treasurer of a political committee, who dies or becomes incapacitated is considered to have filed the report on the date of the filer's death or the date the filer is determined to be incapacitated, as applicable, for purposes of this chapter. In this subsection, "incapacitated" means determined by a judgment of a court exercising probate jurisdiction to be either partially mentally incapacitated without the right to vote or totally mentally incapacitated.

## §18.19. Affidavit of Timely Filing.

A filer who has been notified by the commission that a report is late but who filed the report on or before the deadline may submit an affidavit to the executive director swearing that the report was timely filed.

#### §18.21. Jurisdiction to Consider Waiver Request.

(a) A filer must file a complete report before the executive director or commission will consider a request to waive or reduce a fine assessed for failure to file a timely report.

(b) A request to waive or reduce a fine assessed for failure to file a timely report or for filing a correction to a report will not be considered after an action to collect the fine is filed in court.

#### §18.23. Administrative Waiver of Statutory Civil Penalties.

(a) A filer may request the executive director to waive a civil penalty determined by §§305.033(b) or 572.033(b) of the Government Code or §254.042(b) of the Election Code by submitting an affidavit to the executive director.

(b) If, in the executive director's discretion, the affidavit establishes any of the following grounds for a waiver, the executive director shall waive the civil penalty, and the penalty waived is not a prior offense for purposes of §18.25 of this title (relating to Administrative Waiver or Reduction of Certain Statutory Civil Penalties) or §18.26 of this title (relating to Administrative Waiver or Reduction of Other Statutory Civil Penalties in Excess of \$500):

(1) the report was filed late because of an unforeseen serious medical emergency or condition or a death that involved the filer, a family member or relative of the filer, a member of the filer's household, or a person whose usual job duties include preparation of the report;

(2) the report was filed late as a result of verifiable severe weather at the filer's location that prevented the filer from filing the report by the applicable deadline and the report was filed within a reasonable time after the deadline;

(3) the report was filed late because the filer was a first responder, as defined in §6.1 of this title (relating to Definitions), deployed to an emergency situation at the time of the filing deadline or a member of the military deployed on active duty at the time of the filing deadline and the report was filed within a reasonable time after the deadline;

(4) the filer filed a timely report but accidentally selected the incorrect filing year or filing period in the agency's electronic filing system, and:

(A) the filer filed a corrected report amending the filing year or filing period no later than 30 days after the individual was notified that the report appeared to be late; and

(B) the corrected report is substantively identical to the originally-filed report;

(5) the filer reasonably relied on incorrect information given to the filer by the agency; or

(6) the report was filed late because of other administrative error by the agency.

(c) If, in the executive director's discretion, the affidavit establishes any of the following grounds for a waiver, the executive director shall waive the civil penalty, but the penalty waived is a prior offense for purposes of §18.25 or §18.26:

(1) the filer of the personal financial disclosure report is not an elected official, a candidate for election, or a salaried public servant, and the late report:

(A) was the first personal financial disclosure report filed late by the filer under Government Code chapter 572; and

(B) was filed no later than 30 days after the individual was notified that the report appeared to be late;

(2) the filer of the personal financial disclosure report was an unopposed candidate in a primary election, and the late report:

(A) was the first personal financial disclosure report filed late by the filer under Government Code chapter 572; and

(B) was filed before the primary election.

(3) the filer of the campaign finance report:

(A) had filed all previous reports by the applicable deadline;

(B) had no new contributions, expenditures, or loans to report during the filing period; and

(C) filed the report no later than 30 days after the filer first learned that the report was late;

# **§18.24.** General Guidelines for Other Administrative Waiver or Reduction of Statutory Civil Penalties.

(a) A filer who does not qualify for a waiver under §18.23 of this title (relating to Administrative Waiver of Statutory Civil Penalties) may request the executive director to waive a civil penalty determined by §§305.033(b) and 572.033(b) of the Government Code or §254.042(b) of the Election Code by submitting an affidavit to the executive director. The executive director may waive or reduce a civil penalty if the filer meets the criteria and the late report meets the qualifications set out in §18.25 of this title (relating to Administrative Waiver or Reduction of Certain Statutory Civil Penalties) and §18.26 of this title (relating to Administrative Waiver or Reduction of Other Statutory Civil Penalties in Excess of \$500).

(b) For purposes of determining a waiver or reduction of a civil penalty under §18.25 and §18.26 of this title, a filer requesting a waiver or reduction will be categorized as follows:

(1) Category A includes candidates for and officeholders of the following offices and specific-purpose committees supporting candidates for and officeholders of the following offices:

- (A) statewide office;
- (B) legislative office;
- (C) district judge;
- (D) state appellate court justice;
- (E) State Board of Education member; and

(F) Secretary of State.

(2) Category B includes all filers not categorized in Category A, as defined by paragraph(1) of this subsection, or Category C, as defined by paragraph(3) of this subsection.Examples of Category B filers include the following filer types:

(A) lobbyists;

(B) salaried non-elected officials;

(C) candidates for and officeholders of district attorney;

(D) candidates for and officeholders of political party chair;

(E) political committees with \$3,000 or more in annual activity in the calendar year in which the late report was due; and

(F) a legislative caucus.

(3) Category C includes:

(A) unsalaried appointed board members and officials; and

(B) political committees with less than \$3,000 in annual activity in the calendar year in which the late report was due.

(c) For purposes of a reduction of a civil penalty under §18.25 and §18.26 of this title, good cause includes, but is not limited to, the following:

(1) The report was filed no later than three days after the date it was due.

(2) The filer filed the report within five days after first learning the report was late from a late notice sent by the commission.

(3) The report was not a critical report and was prepared and placed in the mail on time but not postmarked by the deadline.

(4) The filer had technical difficulties after regular business hours, but the report was filed no later than the next business day after the commission's technical support staff fixed the technical difficulty.

(5) There are no funds in the filer's campaign or officeholder account and the filer is unemployed.

(6) A first-time filer that is required to file campaign finance reports with a county filing authority and personal financial statements with the commission, who mistakenly files the personal financial statement with the county on the filing deadline and then correctly files with the commission within seven days of realizing the mistake.

(d) For purposes of determining whether a filer is eligible for a waiver or reduction of a civil penalty under §18.25 or §18.26 of this title, a prior offense is any prior late report in which a civil penalty was assessed except:

- (1) the civil penalty for that prior late report was waived under Sections 18.23(b) of this title; or
- (2) no late notices were sent for that prior late report and the filer did not file a request that the civil penalty be waived or reduced for the prior late report.

(e) A civil penalty that is reduced under §18.25 or §18.26 of this title will revert to the full amount originally assessed if the reduced civil penalty is not paid within thirty (30) calendar days from the date of the letter informing the filer of the reduction.

(f) A filer may appeal a determination made under §18.25 or §18.26 of this title by submitting a request in writing to the commission.

(1) The request for appeal should state the filer's reasons for requesting an appeal, provide any additional information needed to support the request, and state whether the filer would like the opportunity to appear before the commission and offer testimony regarding the appeal.

(2) The Executive Director may review the appeal and reconsider the determination made under §18.25 or § 18.26 of this title or set the appeal for a hearing before the commission.

(3) After hearing a request for appeal, the commission may affirm the determination made under §18.25 or §18.26 of this title or make a new determination based on facts presented in the appeal.

## §18.25. Administrative Waiver or Reduction of Certain Statutory Civil Penalties

(a) The executive director shall apply this section to:

(1) a late report subject to a statutory civil penalty of not more than \$500; or

(2) a late report that:

(A) is subject to a statutory civil penalty in excess of \$500; and

(B) discloses less than \$3,000 in total political contributions and less than \$3,000 in total political expenditures for the reporting period.

(b) In order to qualify for a waiver or reduction of a civil penalty under this section, a filer must meet all of the following criteria:

(1) The filer has no more than two prior late offenses in the five (5) years preceding the filing deadline of the late report at issue;

(2) The filer filed the report within thirty (30) days of learning the report was late;

(3) The civil penalty for the report at issue has not been increased by the commission at a public meeting pursuant to §254.042(b), Election Code, or §305.033(c) or §572.033(b), Government Code; and

(4) The filer does not have an outstanding civil penalty for a prior late report.

(c) The executive director shall use the following chart to determine the level of waiver or reduction of a civil penalty under this section:

| LEVEL | # OF PRIORS<br>IN LAST 5<br>YEARS | CATEGORY<br>A | CATEGORY<br>B | CATEGORY<br>C | EXPLANATORY NOTE                            |
|-------|-----------------------------------|---------------|---------------|---------------|---|
| 1     | 0                                 | Waiver        | Waiver        | Waiver        |   |
| 1.5   | 1                                 | \$150         | \$100         | \$50          | Level 2 violation with good<br>cause shown* |
| 2     | 1                                 | \$300         | \$200         | \$100         |   |
| 2.5   | 2                                 | \$400         | \$300         | \$150         | Level 3 violation with good cause shown*    |
| 3     | 2                                 | \$500         | \$500         | \$250         |   |

\*The categorization shifts one-half level (from Level 2 to 1.5; from Level 3 to 2.5) if the filer's explanation qualifies as good cause under section 18.24(c) of this title.

# §18.26. Administrative Waiver or Reduction of Other Statutory Civil Penalties in Excess of \$500.

(a) The executive director shall apply this section to a late report that discloses more than \$3,000 in total political contributions or more than \$3,000 in total political expenditures during the reporting period and that is subject to a civil penalty in excess of \$500.

(b) In order to qualify for a waiver or reduction of a civil penalty under this section, a filer must meet all of the following criteria:

(1) The filer has no more than two prior late offenses in the five (5) years preceding the filing deadline of the late report at issue;

(2) The civil penalty for the report at issue has not been increased by the commission at a public meeting pursuant to §254.042(b), Election Code, or §305.033(c) or §572.033(b), Government Code; and

(3) The filer does not have an outstanding civil penalty for a prior late report.

(c) The executive director shall use the following chart to determine the level of waiver or reduction of a civil penalty under this section:

# Category A

| NO GOOD CAUSE                              | EXPLANATORY NOTES   |
|--|---|
| Starting Penalty = \$500                   | 1st day late  |
| + \$100 a day, up to \$1,000               | 2nd – 11th days late  |
| + \$500 for every full 30 days thereafter, | 12th day late – Filed Date: Take # of days divided by   |
| up to \$10,000                             | 30; drop remainder days that do not make a full 30-day segment  |
| GOOD CAUSE SHOWN                           | EXPLANATORY NOTES   |
| Starting Penalty = \$150 (0 priors); or    | 1st day late  |
| Starting Penalty = \$400 (1 or 2 priors)   |   |
| + \$100 a day, up to \$1,000               | 2nd – 11th days late  |
| + \$500 every full 30 days thereafter,     | 12th day late – Filed Date: Take # of days divided by<br>30: drop remainder days that do not make a full 30 day |
| up to \$10,000                             | 30; drop remainder days that do not make a full 30-day segment  |

# Category B

| NO GOOD CAUSE                            | EXPLANATORY NOTES  |
|--|--|
| Starting Penalty = \$500                 | 1st day late   |
| + \$100 a day, up to \$500               | 2nd – 6th days late  |
| + \$250 every full 30 days thereafter,   | 7th day late – Filed Date: Take # of days divided by 30;   |
| up to \$5,000                            | drop remainder days that do not make a full 30-day segment |
| GOOD CAUSE SHOWN                         | EXPLANATORY NOTES  |
| Starting Penalty = \$100 (0 priors); or  | 1st day late   |
| Starting Penalty = \$300 (1 or 2 priors) |  |
| + \$100 a day, up to \$500               | 2nd – 6th days late  |
| + \$250 every full 30 days thereafter,   | 7th day late – Filed Date: Take # of days divided by 30;   |
| up to \$5,000                            | drop remainder days that do not make a full 30-day segment |

# Category C

| NO GOOD CAUSE                            | EXPLANATORY NOTES  |
|--|--|
| Starting Penalty = \$500                 | 1st day late   |
| + \$100 a day, up to \$500               | 2nd – 6th days late  |
| + \$250 every full 30 days thereafter,   | 7th day late – Filed Date: Take # of days divided by 30;<br>drop remainder days that do not make a full 30-day |
| up to \$5,000                            | segment  |
| GOOD CAUSE SHOWN                         | EXPLANATORY NOTES  |
| Starting Penalty = \$50 (0 priors); or   | 1st day late   |
| Starting Penalty = \$150 (1 or 2 priors) |  |
| + \$100 a day, up to \$500               | 2nd – 6th days late  |
| + \$250 every full 30 days thereafter,   | 7th day late – Filed Date: Take # of days divided by 30;<br>drop remainder days that do not make a full 30-day |
| up to \$5,000                            | segment  |

# §18.31. Adjustments to Reporting Thresholds.

(a) Pursuant to section 571.064 of the Government Code, the reporting thresholds are adjusted as follows:

| Campaign Finance<br>Reports: Section of<br>Election Code | Threshold Description  | Original Threshold<br>Amount | Adjusted Amount |
|--|--|------------------------------|-----------------|
| 253.031(b)   | PAC: Amount of contributions or<br>expenditures permitted before TA is<br>required                         | \$500                        | \$980           |
| 253.031(d)(2)  | CEC: Amount of contributions or<br>expenditures permitted before TA is<br>required                         | \$25,000                     | \$36,630        |
| 253.032(a)   | Contribution by Out-of-state PAC:<br>Threshold above which certain<br>paperwork is required                | \$500                        | \$1,010         |
| 253.032(a)(1)  | Contribution to Out-of-state PAC:<br>Threshold above which certain<br>contribution information is required | \$100                        | \$200           |
| 253.032(e)   | Contribution by Out-of-state PAC:<br>Threshold at or below which certain<br>information is required        | \$500                        | \$1,010         |

| Campaign Finance<br>Reports: Section of<br>Election Code | Threshold Description  | Original Threshold<br>Amount | Adjusted Amount |
|--|--|------------------------------|-----------------|
| 254.031(a)(1)  | Contributions: Threshold over which more information is required   | \$50                         | \$100           |
| 254.031(a)(2)  | Loans: Threshold over which more information is required   | \$50                         | \$100           |
| 254.031(a)(3)  | Expenditures: Threshold over which more information is required  | \$100                        | \$200           |
| 254.031(a)(5)  | Contributions: Threshold at or below<br>which more information is not required   | \$50                         | \$100           |
| 254.031(a)(5)  | Expenditures: Threshold at or below<br>which more information is not required  | \$100                        | \$200           |
| 254.031(a)(9)  | Interest, credits, refunds: Threshold over<br>which more information is required   | \$100                        | \$130           |
| 254.031(a)(10)   | Sale of political assets: Threshold over which proceeds must be reported   | \$100                        | \$130           |
| 254.031(a)(11)   | Investment Gain: Threshold over which more information is required   | \$100                        | \$130           |
| 254.031(a)(12)   | Contribution Gain: Threshold over<br>which more information is required  | \$100                        | \$130           |
| 254.0311(b)(1)   | Caucus, contributions from non-caucus<br>members: Threshold over which more<br>information is required                                   | \$50                         | \$100           |
| 254.0311(b)(2)   | Caucus, loans: Threshold over which more information is required   | \$50                         | \$100           |
| 254.0311(b)(3)   | Caucus, expenditures: Threshold over which more information is required  | \$50                         | \$100           |
| 254.0311(b)(4)   | Caucus, contributions and expenditures:<br>Threshold at or below which more<br>information is not required                               | \$50                         | \$100           |
| 254.0312   | Contributions, Best Efforts: Threshold<br>under which filer is not required to<br>request contributor information to be in<br>compliance | \$500                        | \$770           |
| 254.036  | Electronic Filing Exemption: Threshold<br>at or below which a filer may qualify  | \$20,000                     | \$30,820        |
| 254.038(a)   | Daily Reports by certain candidates and<br>PACs: Contribution threshold triggering<br>report   | \$1,000                      | \$2,020         |
| 254.039  | Daily Reports by GPACs: Contribution<br>threshold triggering report  | \$5,000                      | \$6,910         |

| Campaign Finance<br>Reports: Section of<br>Election Code | Threshold Description  | Original Threshold<br>Amount | Adjusted Amount  |
|--|--|------------------------------|------------------|
| 254.039  | Daily reports by GPACs: DCE<br>expenditure thresholds (single<br>candidate/group of candidates)  | \$1,000/\$15,000             | \$2,020/\$30,330 |
| 254.0611(a)(2)   | Judicial candidates, contributions:<br>Threshold over which more information<br>is required  | \$50                         | \$100            |
| 254.0611(a)(3)   | Judicial candidates, asset purchase:<br>Threshold over which more information<br>is required   | \$500                        | \$1,010          |
| 254.0612   | Statewide executive and legislative<br>candidates, contributions: Threshold<br>over which more information is required                         | \$500                        | \$1,010          |
| 254.095  | Local officeholders, contributions:<br>Threshold under which reporting is not<br>required  | \$500                        | \$1,010          |
| 254.151(6)   | GPAC, contributions: Threshold over<br>which more information is required  | \$50                         | \$100            |
| 254.1541(a)  | GPAC, higher itemization threshold:<br>Threshold under which it applies  | \$20,000                     | \$29,300         |
| 254.1541(b)  | GPACs that meet higher itemization<br>threshold: Threshold over which more<br>contributor information is required                              | \$100                        | \$200            |
| 254.156(1)   | MPAC: Threshold over which<br>contribution, lender and expenditure<br>information is required  | \$10                         | \$20             |
| 254.156(2)   | MPACs that meet higher itemization<br>threshold: Threshold over which more<br>contributor information is required                              | \$20                         | \$40             |
| 254.181, 254.182,<br>254.183                             | Candidate or SPACs, modified<br>reporting: Contribution or expenditure<br>threshold at or below which filers may<br>avoid pre-election reports | \$500                        | \$1,010          |
| 254.261  | DCE filers: Threshold over which a report must be filed  | \$100                        | \$150            |

| Lobby Registrations<br>and Reports:<br>Section of<br>Government Code | Threshold Description  | Original Threshold<br>Amount                 | Adjusted Amount                       |
|--|--|--|---------------------------------------|
| 305.003(1)   | Lobbyist, expenditures: Threshold over<br>which registration is required                     | \$500, by 1 Tex.<br>Admin. Code<br>\$34.41   | \$880                                 |
| 305.003(2)   | Lobbyist, compensation: Threshold over<br>which registration is required                     | \$1,000, by 1 Tex.<br>Admin. Code<br>\$34.43 | \$1,760                               |
| 305.004(7)   | Lobbying for political party: Threshold<br>at or below which registration is not<br>required | \$5,000                                      | \$10,110                              |
| 305.005(g)(2)  | Lobbyist: Compensation threshold   | \$10,000                                     | Less than \$20,220                    |
| 305.005(g)(3)  | Lobbyist: Compensation threshold   | \$25,000                                     | \$20,220 to less<br>than \$50,540     |
| 305.005(g)(4)  | Lobbyist: Compensation threshold   | \$50,000                                     | \$50,540 to less<br>than \$101,090    |
| 305.005(g)(5)  | Lobbyist: Compensation threshold   | \$100,000                                    | \$101,090 to less<br>than \$202,180   |
| 305.005(g)(6)  | Lobbyist: Compensation threshold   | \$150,000                                    | \$202,180 to less<br>than \$303,270   |
| 305.005(g)(7)  | Lobbyist: Compensation threshold   | \$200,000                                    | \$303,270 to less<br>than \$404,350   |
| 305.005(g)(8)  | Lobbyist: Compensation threshold   | \$250,000                                    | \$404,350 to less<br>than \$505,440   |
| 305.005(g)(9)  | Lobbyist: Compensation threshold   | \$300,000                                    | \$505,440 to less<br>than \$606,530   |
| 305.005(g)(10)   | Lobbyist: Compensation threshold   | \$350,000                                    | \$606,530 to less<br>than \$707,620   |
| 305.005(g)(11)   | Lobbyist: Compensation threshold   | \$400,000                                    | \$707,620 to less<br>than \$808,710   |
| 305.005(g)(12)   | Lobbyist: Compensation threshold   | \$450,000                                    | \$808,710 to less<br>than \$909,800   |
| 305.005(g)(13)   | Lobbyist: Compensation threshold   | \$500,000                                    | \$909,800 to less<br>than \$1,010,880 |
| 305.005(g-1)   | Lobbyist: Compensation threshold   | \$500,000                                    | \$1,010,880 or more                   |

| Lobby Registrations<br>and Reports:<br>Section of<br>Government Code | Threshold Description  | Original Threshold<br>Amount | Adjusted Amount |
|--|--|------------------------------|-----------------|
| 305.0061(c)  | Lobbyist, legislative/executive branch<br>member: Threshold over which gifts,<br>awards and mementos must be disclosed | \$50                         | \$100           |
| 305.0061(e-1)  | Lobbyist, food and beverage: Threshold<br>at or below which it is considered a gift<br>and reported as such            | \$50                         | \$100           |
| 305.0063   | Lobbyist, annual filer: Expenditure<br>threshold at or below which filer may<br>file annually                          | \$1,000                      | \$2,020         |

| Personal Financial<br>Statements: Section<br>of Gov't Code | Threshold Description  | Original Threshold<br>Amount      | Adjusted Amount                   |
|--|--|-----------------------------------|-----------------------------------|
| 572.022(a)(1)  | PFS threshold  | less than \$5,000                 | less than \$10,110                |
| 572.022(a)(2)  | PFS threshold  | \$5,000 to less than<br>\$10,000  | \$10,110 to less<br>than \$20,220 |
| 572.022(a)(3)  | PFS threshold  | \$10,000 to less than<br>\$25,000 | \$20,220 to less<br>than \$50,540 |
| 572.022(a)(4)  | PFS threshold  | \$25,000 or more                  | \$50,540 or more                  |
| 572.005,<br>572.023(b)(1)                                  | PFS, retainer: Threshold over which<br>filer with a substantial interest in a<br>business entity must report more<br>information | \$25,000                          | \$50,540                          |
| 572.023(b)(4)  | PFS, interest, dividends, royalties and<br>rents: Threshold over which information<br>must be reported                           | \$500                             | \$1,010                           |
| 572.023(b)(5)  | PFS, loans: Threshold over which information must be reported  | \$1,000                           | \$2,020                           |
| 572.023(b)(7)  | PFS, gifts: Threshold over which information must be reported  | \$250                             | \$510                             |
| 572.023(b)(8)  | PFS, income from trust: Threshold over which information must be reported  | \$500                             | \$1,010                           |
| 572.023(b)(15)   | PFS, government contracts: Threshold<br>of aggregate over which more<br>information must be reported                             | Exceeds \$10,000                  | \$11,100                          |
| 572.023(b)(15)(A)  | PFS, government contracts: Itemization threshold   | \$2,500 or more                   | \$2,770 or more                   |
| 572.023(b)(16)(D)(i)                                       | PFS, bond counsel fees paid to legislator: Threshold   | less than \$5,000                 | less than \$5,550                 |

| Personal Financial<br>Statements: Section<br>of Gov't Code | Threshold Description  | Original Threshold<br>Amount                | Adjusted Amount                             |
|--|--|---|---|
| 572.023(b)(16)(D)(ii)                                      | PFS, bond counsel fees paid to legislator: Threshold           | at least \$5,000 but<br>less than \$10,000  | at least \$5,550 but<br>less than \$11,100  |
| 572.023(b)(16)(D)(iii)                                     | PFS, bond counsel fees paid to legislator: Threshold           | at least \$10,000 but<br>less than \$25,000 | at least \$11,100 but<br>less than \$27,740 |
| 572.023(b)(16)(D)(iv)                                      | PFS, bond counsel fees paid to legislator: Threshold           | \$25,000 or more                            | \$27,740 or more                            |
| 572.023(b)(16)(E)(i)                                       | PFS, bond counsel fees paid to<br>individual's firm: Threshold | less than \$5,000                           | less than \$5,550                           |
| 572.023(b)(16)(E)(ii)                                      | PFS, bond counsel fees paid to individual's firm: Threshold    | at least \$5,000 but<br>less than \$10,000  | at least \$5,550 but<br>less than \$11,100  |
| 572.023(b)(16)(E)(iii)                                     | PFS, bond counsel fees paid to individual's firm: Threshold    | at least \$10,000 but<br>less than \$25,000 | at least \$11,100 but<br>less than \$27,740 |
| 572.023(b)(16)(E)(iv)                                      | PFS, bond counsel fees paid to individual's firm: Threshold    | \$25,000 or more                            | \$27,740 or more                            |

| Speaker Election<br>and Certain<br>Ceremonial<br>Reports: Section of<br>Government Code | Threshold Description  | Original Threshold<br>Amount | Adjusted Amount |
|---|--|------------------------------|-----------------|
| 302.014(4)  | Speaker: Expenditures over which more information must be reported                             | \$10                         | \$20            |
| 303.005(a)(1) - (10)  | Governor for a Day/Speaker's Day:<br>Threshold over which more information<br>must be reported | \$50                         | \$100           |

(b) The changes made by this rule apply only to conduct occurring on or after the effective date of this rule.

(c) The effective date of this rule is January 1, 2023.

(d) In this section:

- (1) "CEC" means county executive committee;
- (2) "DCE" means direct campaign expenditure-only filer;
- (3) "GPAC" means general-purpose political committee;
- (4) "MPAC" means monthly-filing general-purpose political committee;
- (5) "PAC" means political committee;
- (6) "PFS" means personal financial statement;
- (7) "SPAC" means specific-purpose political committee; and

(8) "TA" means treasurer appointment.

### CHAPTER 20. REPORTING POLITICAL CONTRIBUTIONS AND EXPENDITURES

### Subchapter A. GENERAL RULES

#### §20.1. Definitions.

The following words and terms, when used in Title 15 of the Election Code, in this chapter, Chapter 22 of this title (relating to Restrictions on Contributions and Expenditures), and Chapter 24 of this title (relating to Restrictions on Contributions and Expenditures Applicable to Corporations and Labor Organizations), shall have the following meanings, unless the context clearly indicates otherwise.

(1) Campaign communication--The term does not include a communication made by e-mail.

(2) Campaign treasurer--Either the individual appointed by a candidate to be the campaign treasurer, or the individual responsible for filing campaign finance reports of a political committee under Texas law or the law of any other state.

(3) Contribution--The term does not include a transfer for consideration of any thing of value pursuant to a contract that reflects the usual and normal business practice of the vendor.

(4) Corporation--The term does not include professional corporations or professional associations.

(5) Election cycle--A single election and any related primary or runoff election.

(6) Identified measure--A question or proposal submitted in an election for an expression of the voters' will and includes the circulation and submission of a petition to determine whether a question or proposal is required to be submitted in an election for an expression of the voters' will.

(7) Non-political expenditure--An expenditure from political contributions that is not an officeholder expenditure or a campaign expenditure.

(8) Opposed candidate--A candidate who has an opponent whose name is to appear on the ballot. The name of a write-in candidate does not appear on the ballot.

(9) Out-of-state political committee--A political committee that makes political expenditures outside Texas and in the 12 months immediately preceding the making of a political expenditure by the committee inside Texas (other than an expenditure made in connection with a campaign for a federal office or made for a federal officeholder), makes 80% or more of the committee's total political expenditures in any combination of elections outside this state and federal offices not voted on in this state. Section 20.13 of this title (relating to Out-of-State Committees) explains the practical application of this definition.

(10) Pledge--A contribution in the form of an unfulfilled promise or unfulfilled agreement, whether enforceable or not, to provide a specified amount of money or

specific goods or services. The term does not include a contribution actually made in the form of a check.

(11) Political advertising:

(A) A communication that supports or opposes a political party, a public officer, a measure, or a candidate for nomination or election to a public office or office of a political party, and:

(i) is published in a newspaper, magazine, or other periodical in return for consideration;

(ii) is broadcast by radio or television in return for consideration;

(iii) appears in a pamphlet, circular, flier, billboard, or other sign, bumper sticker, or similar form of written communication; or

(iv) appears on an Internet website.

(B) The term does not include an individual communication made by e-mail or text message but does include mass e-mails and text messages involving an expenditure of funds beyond the basic cost of hardware messaging software and bandwidth.

(12) Political subdivision--A county, city, or school district or any other governmental entity that:

(A) embraces a geographic area with a defined boundary;

(B) exists for the purpose of discharging functions of government; and

(C) possesses authority for subordinate self-government through officers selected by it.

(13) Report--Any document required to be filed by this title, including an appointment of campaign treasurer, any type of report of contributions and expenditures, and any notice.

(14) Special pre-election report--A shorthand term for a report filed in accordance with the requirements of §§20.221, 20.333, or 20.435 of this title (relating to Special Pre-Election Report by Certain Candidates; Special Pre-Election Report by Certain Specific-Purpose Committees; Special Pre-Election Reports by Certain General-Purpose Committees) and §§254.038 and §254.039 of the Election Code (relating to Special Report Near Election by Certain Candidates and Political Committees and Special Report Near Election By Certain General-Purpose Committees).

(15) Specific-purpose committee--A political committee that does not meet the definition of general-purpose committee and that has among its principal purposes:

(A) supporting or opposing one or more:

(i) candidates, all of whom are identified and are seeking offices that are known; or

(ii) measures, all of which are identified;

(B) assisting one or more officeholders, all of whom are identified; or

(C) supporting or opposing only one candidate who is unidentified or who is seeking an office that is unknown.

(16) Unidentified measure--A question or proposal that is intended to be submitted in an election for an expression of the voters' will and that is not yet legally required to be submitted in an election, except that the term does not include the circulation or submission of a petition to determine whether a question or proposal is required to be submitted in an election for an expression of the voters' will. The circulation or submission of a petition to determine whether a question or proposal is required to be submitted in an election for an expression of the voters' will. The circulation or submission of a petition to determine whether a question or proposal is required to be submitted in an election for an expression of the voters' will is considered to be an identified measure.

(17) Principal purpose--A group has as a principal purpose of accepting political contributions or making political expenditures, including direct campaign expenditures, when that activity is an important or a main function of the group.

(A) A group may have more than one principal purpose.

(B) A group has as a principal purpose accepting political contributions if the proportion of the political contributions to the total contributions to the group is more than 25 percent within a calendar year. A contributor intends to make a political contribution if the solicitations that prompted the contribution or the statements made by the contributor about the contribution would lead to no other reasonable conclusion than that the contribution was intended to be a political contribution.

(C) The group may maintain specific evidence of contributions related only to political contributions or only to nonpolitical contributions. For example, the group may ask the contributor to make an indication when the contribution is made that the contribution is only a nonpolitical contribution.

(D) A group has as a principal purpose making political expenditures, including direct expenditures, if the group expends more than 25 percent of its annual expenses to make political expenditures within a calendar year. The following shall be included for purposes of calculating the threshold:

(i) the amount of money paid in compensation and benefits to the group's employees for work related to making political expenditures;

(ii) the amount of money spent on political expenditures; and

(iii) the amount of money attributable to the proportional share of administrative expenses related to political expenditures. The proportional share of administrative expenses is calculated by comparing the political expenditures in clause (ii) with nonpolitical expenditures. (For example, if the group sends three mailings a year and each costs \$10,000, if the first two are issue based newsletters and the third is a direct advocacy sample ballot, and there were no other outside expenditures, then the proportion of the administrative expenses attributable to political expenditures would be 33%.) Administrative expenses include:

(I) fees for services to non-employees;

(II) advertising and promotion;

(III) office expenses;

(IV) information technology;

(V) occupancy;

(VI) travel expenses;

(VII) interest; and

(VIII) insurance.

(E) The group may maintain specific evidence of administrative expenses related only to political expenditures or only to nonpolitical expenditures. Specifically identified administrative expenses shall not be included in the proportion established by subparagraph (D)(iii) but allocated by the actual amount of the expense.

(F) In this section, the term "political expenditures" includes direct campaign expenditures.

(18) In connection with a campaign:

(A) An expenditure is made in connection with a campaign for an elective office if it is:

(i) made for a communication that expressly advocates the election or defeat of a clearly identified candidate by:

(I) using such words as "vote for," "elect," "support," "vote against," "defeat," "reject," "cast your ballot for," or "Smith for city council;" or

(II) using such phrases as "elect the incumbent" or "reject the challenger," or such phrases as "vote pro-life" or "vote pro-choice"

accompanied by a listing of candidates described as "pro-life" or "pro-choice;"

(ii) made for a communication broadcast by radio, television, cable, or satellite or distributed by print or electronic media, including any print publication, mailing, Internet website, electronic mail, or automated phone bank, that:

(I) refers to a clearly identified candidate;

(II) is distributed within 30 days before a contested election for the office sought by the candidate;

(III) targets a mass audience or group in the geographical area the candidate seeks to represent; and

(IV) includes words, whether displayed, written, or spoken; images of the candidate or candidate's opponent; or sounds of the voice of the candidate or candidate's opponent that, without consideration of the intent of the person making the communication, are susceptible of no other reasonable interpretation than to urge the election or defeat of the candidate;

(iii) made by a candidate or political committee to support or oppose a candidate; or

- (iv) a campaign contribution to:
  - (I) a candidate; or

(II) a group that, at the time of the contribution, already qualifies as a political committee.

(B) An expenditure is made in connection with a campaign on a measure if it is:

(i) made for a communication that expressly advocates the passage or defeat of a clearly identified measure by using such words as "vote for," "support," "vote against," "defeat," "reject," or "cast your ballot for;"

(ii) made for a communication broadcast by radio, television, cable, or satellite or distributed by print or electronic media, including any print publication, mailing, Internet website, electronic mail, or automated phone bank, that:

(I) refers to a clearly identified measure;

(II) is distributed within 30 days before the election in which the measure is to appear on the ballot;

(III) targets a mass audience or group in the geographical area in

which the measure is to appear on the ballot; and

(IV) includes words, whether displayed, written, or spoken, that, without consideration of the intent of the person making the communication, are susceptible of no other reasonable interpretation than to urge the passage or defeat of the measure;

(iii) made by a political committee to support or oppose a measure; or

(iv) a campaign contribution to a group that, at the time of the contribution, already qualifies as a political committee.

(C) Any cost incurred for covering or carrying a news story, commentary, or editorial by a broadcasting station or cable television operator, Internet website, or newspaper, magazine, or other periodical publication, including an Internet or other electronic publication, is not a campaign expenditure if the cost for the news story, commentary, or editorial is not paid for by, and the medium is not owned or controlled by, a candidate or political committee.

(D) For purposes of this section:

(i) a candidate is clearly identified by a communication that includes the candidate's name, office sought, office held, likeness, photograph, or other apparent and unambiguous reference; and

(ii) a measure is clearly identified by a communication that includes the measure's name or ballot designation (such as "Proposition 1"), purposes, election date, or other apparent and unambiguous reference.

(19) Discount--The provision of any goods or services without charge or at a charge which is less than fair market value. A discount is an in-kind political contribution unless the terms of the transaction reflect the usual and normal practice of the industry and are typical of the terms that are offered to political and non-political persons alike, or unless the discount is given solely in order to comply with §253.041 of the Election Code. The value of an in-kind contribution in the form of a discount is the difference between the fair market value of the goods or services at the time of the contribution and the amount charged.

(20) School district--For purposes of §254.130 of the Election Code and §§20.3 (relating to Reports Filed with the Commission), 20.7 (relating to Reports Filed with Other Local Filing Authority), and 20.315 (relating to Termination of Campaign Treasurer Appointment) of this title, the term includes a junior college district or community college district.

(21) Vendor--Any person providing goods or services to a candidate, officeholder, political committee, or other filer under this chapter. The term does not include an employee of the candidate, officeholder, political committee, or other filer.

(22) Hybrid committee—A political committee that, as provided by section 252.003(a)(4) (relating to contents of a general-purpose committee's campaign treasurer appointment) or 252.0031(a)(2) (relating to a specific-purpose committee's campaign treasurer appointment) of the Election Code, as applicable, has filed a campaign treasurer appointment that includes an affidavit stating that:

(A) the committee is not established or controlled by a candidate or an officeholder; and

(B) the committee will not use any political contribution from a corporation or a labor organization to make a political contribution to:

(i) a candidate for elective office;

(ii) an officeholder; or

(iii) a political committee that has not filed an affidavit in accordance with this section.

(23) Direct campaign expenditure-only committee—A political committee, as authorized by section 253.105 of the Election Code (relating to political contributions to direct campaign expenditure-only committees) to accept political contributions from corporations or labor organizations, that:

- (A) is not established or controlled by a candidate or an officeholder;
- (B) makes or intends to make direct campaign expenditures;
- (C) does not make or intend to make political contributions to:
  - (i) a candidate;
  - (ii) an officeholder;

(iii) a specific-purpose committee established or controlled by a candidate or an officeholder; or

(iv) a political committee that makes or intends to make political contributions to a candidate, an officeholder, or a specific-purpose committee established or controlled by a candidate or an officeholder; and

(D) has filed an affidavit with the commission stating the committee's intention to operate as described by subparagraphs (B) and (C).

#### §20.3. Reports Filed with the Commission.

The Ethics Commission is the appropriate filing authority for reports filed by:

(1) a candidate for one of the following offices:

(A) a statewide office;

(B) a district office filled by voters in more than one county;

(C) a seat in the state legislature;

(D) a seat on the State Board of Education;

(E) an office of a political subdivision other than a county, if the governing body of the political subdivision has not been formed and if the political subdivision includes territory in more than one county; or

(F) a judicial district office filled by voters of only one county, subject to §20.5(b);

(2) a person holding an office listed in paragraph (1) of this section;

(3) the secretary of state;

(4) a specific-purpose committee supporting or opposing a candidate or officeholder required to file with the commission; or

(5) a specific-purpose committee supporting or opposing:

(A) a measure to be submitted to the voters of the entire state; or

(B) a measure that concerns a political subdivision other than a county, if the governing body of the political subdivision has not been formed and if the political subdivision includes territory in more than one county;

(6) a specific-purpose committee created to support or oppose a measure on the issuance of bonds by a school district; or

(7) a general-purpose committee.

#### §20.5. Reports Filed with a County Filing Authority.

The county clerk (or the county elections administrator or tax assessor-collector who is required to perform the functions of the county clerk as provided by §§31.043 or 31.071 of the Election Code) is the appropriate filing authority for reports filed by:

(1) a candidate for:

(A) a county office;

(B) a precinct office;

(C) a district office (except for an office in a multi-county district; or

(D) an office of a political subdivision other than a county if the political subdivision is within the boundaries of a single county and if the governing body of the political subdivision has not been formed;

(2) a person holding an office listed in paragraph (1) of this section;

(3) a specific-purpose committee supporting or opposing a candidate listed in paragraph

(1) of this section or an office holder listed in paragraph (2) of this section;

(4) a specific-purpose committee supporting or opposing:

(A) a measure to be submitted to the voters of a single county; or

(B) a measure concerning a political subdivision other than a county when the governing body for the political subdivision has not been formed and no boundary of the political subdivision crosses a boundary of a county.

### §20.7. Reports Filed with Other Local Filing Authority.

(a) Except as provided by \$20.3(6) of this title (relating to Reports Filed with the Commission), the secretary of a political subdivision (or the presiding officer if the political subdivision has no secretary) is the appropriate filing authority for reports filed by:

(1) a candidate for an office of a political subdivision other than a county;

(2) a person holding an office of a political subdivision other than a county; or

(3) a specific-purpose committee supporting or opposing a measure to be submitted at an election ordered by the authority of a political subdivision other than a county.

(b) The campaign treasurer of a specific-purpose committee created to support or oppose a measure on the issuance of bonds by a school district should file with the commission a file-stamped copy of any campaign treasurer appointment filed with the appropriate local filing authority.

## §20.9. Filing Option for Certain Specific-Purpose Committees.

A specific-purpose committee required to file reports with more than one authority may choose to file reports only with the commission.

#### §20.11. Federal Candidates and Officeholders.

The laws administered and enforced by the commission do not apply to a candidate for election to an office of the federal government or to a federal officeholder.

#### §20.13. Out-of-State Committees.

(a) An out-of-state political committee is required to file reports for each reporting period under Subchapter F, Chapter 254, Election Code, in which the out-of-state political committee accepts political contributions or makes political expenditures in connection with a state or local election in Texas. Section 254.1581, Election Code, applies to a report required to be filed under this section. An out-of-state political committee that files reports electronically in another jurisdiction may comply with §254.1581, Election Code, by sending a letter to the commission within the time prescribed by that section specifying in detail where the electronic report may be found on

the website of the agency with which the out-of-state political committee is required to file its reports. An out-of-state political committee that does not file reports electronically in another jurisdiction may comply with §254.1581, Election Code, by sending a copy of the cover sheets of the report and a copy of each page on which the committee reports a contribution or expenditure accepted or made in connection with a state or local election in Texas.

(b) An out-of-state political committee that files an appointment of campaign treasurer with a Texas filing authority is required to file reports under this title.

(c) A political committee must determine if it is an "out-of-state political committee" each time the political committee plans to make a political expenditure in Texas (other than an expenditure in connection with a campaign for a federal office or an expenditure for a federal officeholder). The determination is made as follows.

(1) Before making the expenditure (other than an expenditure in connection with a campaign for a federal office or an expenditure for a federal officeholder), the committee must calculate its total political expenditures made during the 12 months immediately preceding the date of the planned expenditure. This total does not include the planned political expenditure triggering the calculation requirement.

(2) If 80% or more of the total political expenditures are in connection with elections not voted on in Texas, the committee is an out-of-state committee.

(3) If less than 80% of the total political expenditures are in connection with elections not voted on in Texas, the committee is no longer an out-of-state committee.

(d) Section 20.29 (relating to Information About Out-of-State Committees) and §22.7 (relating to Contribution from Out-of-State Committee) of this title contain other provisions regarding requirements applicable to recipients of contributions from out-of-state political committees.

(e) An out-of-state political committee planning an expenditure in connection with a campaign for federal office voted on in Texas is not required to make the determination required under subsection (c) of this section. However, an expenditure in connection with a campaign for federal office voted on in Texas must be included in the calculation set out in subsection (c) of this section for an out-of-state committee making an expenditure in connection with a non-federal campaign voted on in Texas.

## §20.15. Change of Address.

The campaign treasurer of a political committee required to file reports with the commission shall provide written notice to the commission of any change in his or her mailing address no later than the 10th day after the date of the change.

## §20.16. Notices by Electronic Mail.

(a) A person required to file reports electronically with the commission shall provide to the commission an electronic mail address to which notices regarding filing requirements under Title 15 of the Election Code may be sent.

(b) A person required to file reports with the commission and who qualifies for an exemption from electronic filing may provide to the commission an electronic mail address to which notices regarding filing requirements under Title 15 of the Election Code may be sent.

(c) If the commission is twice unable to notify a person of a deadline at an electronic mail address provided under subsection (a) or (b) of this section, the commission is not required to make any further attempts to notify the person of that deadline or any future deadlines until the person has notified the commission of the person's current electronic mail address.

## §20.18. Recordkeeping Required.

(a) Records required to be maintained by §254.001 of the Election Code consist of records containing information needed to comply with reporting requirements, examples may include bank statements (front and back), deposit slips, cancelled checks (front and back), receipts, invoices, bills, and ledgers of contributions and expenditures.

(b) Candidates, officeholders, and campaign treasurers of a political committee comply with \$254.001 of the Election Code when they maintain the following:

- (1) Bank statements for all campaign activity;
- (2) Invoices or bills for campaign expenditures;
- (3) Copies of checks paid for campaign activity:

(4) Donation documentation for each person from whom a political contribution, loan, gain, or reimbursement is accepted;

(5) Receipts for reimbursed campaign expenses, which document the purpose of the reimbursement;

(6) Employee timesheets and payroll records;

(7) Extra care must be taken if cash is received or disbursed including: a separate receipt indicating the source of the donation or the person who received the disbursement, and the amount of the donation or expenditure.

(c) A person required to maintain a record under this section shall preserve the record for at least two years beginning on the filing deadline for the report containing the information in the record.

## §20.19. Reports Must Be Filed on Official Forms.

Except for special pre-election reports that are exempt from the electronic filing requirement, all reports required by Chapters 20 through 40 must be filed in a format prescribed by the commission or on forms approved by the executive director pursuant to §18.1 of this title (relating to Forms).

### §20.20. Timeliness of Action by Electronic Filing.

The filing deadline for any report filed electronically with the commission is midnight Central Time Zone on the last day for filing the report under the law requiring the filing of the report.

#### §20.21. Due Dates on Holidays and Weekends.

(a) Except as provided in subsection (b) of this section, if the deadline for a report falls on a Saturday, Sunday, or a legal state or national holiday, the report is due on the next regular business day.

(b) Subsection (a) of this section does not apply to a special pre-election report required by this title or by Title 15 of the Election Code. Special pre-election reports are due on the date assigned by the sections requiring those reports to be filed.

#### §20.23. Timeliness of Action by Mail.

When this chapter requires a notice, report, or other document or paper to be delivered, submitted, or filed within a specified period or before a specified deadline, a delivery, submission, or filing by first-class United States mail is timely except as otherwise provided by this chapter, if:

(1) it is properly addressed with postage prepaid; and

(2) it bears a post office cancellation mark indicating a time within the period or before the deadline, or if the person required to take the action furnishes satisfactory proof that it was deposited within in the mail within the period or before the deadline.

#### §20.29. Information About Out-of-State Committees.

(a) A person who files a report with the commission by electronic transfer and who accepts political contributions from an out-of-state political committee required to file its statement of organization with the Federal Election Commission shall either:

(1) enter the out-of-state committee's federal PAC identification number in the appropriate place on the report; or

(2) timely file a certified copy of the out-of-state committee's statement of organization that is filed with the Federal Election Commission.

(b) A person who files a report with the commission by electronic transfer and who accepts political contributions from an out-of-state political committee that is not required to file its statement of organization with the Federal Elections Commission shall either:

(1) enter the information required by 253.032(a)(1) or (e)(1), Election Code, as applicable, on the report filed by electronic transfer; or

(2) timely file a paper copy of the information required by 253.032(a)(1) or (e)(1), Election Code, as applicable.

(c) Except as provided by subsection (d) of this section, \$251.007, Election Code, applies to a document filed under subsection (a)(2) or (b)(2) of this section.

(d) A document filed under subsection (a)(2) or (b)(2) of this section for a pre-election report is timely filed if it is received by the commission no later than the report due date. A pre-election report includes reports due 30-days and 8-days before an election, reports due before a runoff election, and special reports due before an election.

## §20.33. Termination of Campaign Treasurer Appointment By Commission.

(a) The commission may terminate the campaign treasurer appointment of an inactive candidate or an inactive political committee.

(b) For purposes of subsection (a) of this section and §252.0131, Election Code, a candidate becomes "inactive" if the candidate files a campaign treasurer appointment with the commission and more than one year has lapsed since the candidate has filed any required campaign finance reports with the commission.

(c) For purposes of subsection (a) of this section and §252.0131, Election Code, a political committee becomes "inactive" if the political committee files a campaign treasurer appointment with the commission and more than one year has lapsed since the campaign treasurer of the political committee has filed any required campaign finance reports with the commission.

(d) This section does not apply to a candidate who holds an office specified by §252.005(1) or (5), Election Code.

#### §20.35. Notice of Proposed Termination of Campaign Treasurer Appointment.

(a) Before the commission may consider termination of a campaign treasurer appointment under \$20.33 of this title (relating to Termination of Campaign Treasurer Appointment by Commission) and \$252.0131, Election Code, the commission shall send written notice to the affected candidate or political committee.

(b) The written notice must be given at least 30 days before the date of the meeting at which the commission will consider the termination of campaign treasurer appointment and must include:

(1) The date, time, and place of the meeting;

(2) A statement of the commission's intention to consider termination of the campaign treasurer;

(3) A reference to the particular sections of the statutes and rules that give the commission the authority to consider the termination of the campaign treasurer; and

(4) The effect of termination of the campaign treasurer appointment.

## Subchapter B. GENERAL REPORTING RULES

#### §20.50. Total Political Contributions Maintained.

(a) For purposes of Election Code §254.031(a)(8) and §254.0611(a)(1), the total amount of political contributions maintained in one or more accounts includes the following:

(1) The balance on deposit in banks, savings and loan institutions and other depository institutions;

(2) The present value of any investments that can be readily converted to cash, such as certificates of deposit, money market accounts, stocks, bonds, treasury bills, etc.; and

(3) The balance of political contributions accepted and held in any online fundraising account over which the filer can exercise control by making a withdrawal, expenditure, or transfer.

(b) For purposes of Election Code §254.031(a)(8) and §254.0611(a)(1), the total amount of political contributions maintained includes personal funds that the filer intends to use for political expenditures only if the funds have been deposited in an account in which political contributions are held as permitted by Election Code §253.0351(c).

(c) For purposes of Election Code §254.031(a-1), the difference between the total amount of political contributions maintained that is disclosed in a report and the correct amount is a de minimis error if the difference does not exceed:

(1) \$250; or

(2) the lesser of 10% of the amount disclosed or \$2,500.

#### §20.51. Value of In-Kind Contribution.

(a) For reporting purposes, the value of an in-kind contribution is the fair market value.

(b) If an in-kind contribution is sold at a political fundraiser, the total amount received for the item at the fundraiser must be reported. This reporting requirement is in addition to the requirement that the fair market value of the in-kind contribution be reported.

(c) If political advertising supporting or opposing two or more candidates is an in-kind contribution, each person benefiting from the contribution shall report the amount determined by dividing the full value of the political advertising by the number of persons benefited by the political advertising.

#### §20.52. Description of In-Kind Contribution for Travel.

The description of an in-kind contribution for travel outside of the state of Texas must provide the following:

(1) The name of the person or persons traveling on whose behalf the travel was accepted;

(2) The means of transportation;

(3) The name of the departure city or the name of each departure location;

(4) The name of the destination city or the name of each destination location;

(5) The dates on which the travel occurred;

(6) The campaign or officeholder purpose of the travel, including the name of a conference, seminar, or other event.

### §20.53. Disclosure of True Source of Contribution or Expenditure.

A person may not knowingly make or authorize a political contribution or political expenditure in the name of or on behalf of another unless the person discloses the name and address of the person who is the true source of the contribution or expenditure.

### §20.54. Reporting a Pledge of a Contribution.

(a) The date of a pledge of a contribution is the date the pledge was accepted, regardless of when the pledge is actually received.

(b) Except as provided by subsection (c) of this section, a pledge of a contribution shall be reported on the appropriate pledge schedule for the reporting period in which the pledge was accepted and shall be reported on the appropriate receipts schedule for the reporting period in which the pledge is received.

(c) A pledge of a contribution that is actually received in the reporting period in which the pledge was accepted, shall be reported on the contribution schedule or the loan schedule, as applicable, and in accordance with subsection (a) of this section.

(d) The effective date of this rule is January 1, 2015.

#### §20.55. Time of Accepting Contribution.

(a) A candidate, office holder, or political committee shall make a determination to accept or refuse a political contribution not later than the end of the reporting period during which the contribution is received, except as provided by subsection (e) of this section.

(b) A determination to refuse a political contribution is a distinct act from returning a political contribution and may occur at a different time.

(c) If a determination to accept or refuse a political contribution is not made before the end of the reporting period during which the contribution is received, the contribution is considered to have been accepted on the last day of that reporting period.

(d) A political contribution that is received but not accepted shall be returned to the contributor not later than the 30th day after the deadline for filing a report for the reporting period during which the contribution is received. A contribution not returned within that time is considered to be accepted.

(e) A determination to accept or refuse a political contribution received during a special legislative session shall be made not later than the third day after the date the contribution is received.

## §20.56. Expenditures to Vendors.

(a) A political expenditure made by a vendor for a candidate, officeholder, political committee, or other filer, with the intent to seek reimbursement from the filer, shall be reported by the filer in accordance with this chapter as though the filer made the expenditure directly.

(b) A vendor of a candidate, officeholder, or specific-purpose committee may not, in providing goods or services for the candidate, officeholder, or committee, make an expenditure that, if made by the candidate, officeholder, or committee, would be prohibited by §§253.035, 253.038, or 253.041, Election Code.

(c) A candidate, officeholder, or specific-purpose committee may not use political contributions to pay or reimburse a vendor for an expenditure that, if made by the candidate, officeholder, or committee, would be prohibited by §§253.035, 253.038, or 253.041, Election Code.

### §20.57. Time of Making Expenditure.

(a) The date of a political expenditure is the date the amount is readily determinable by the person making the expenditure, except as provided by subsection (b) of this section.

(b) If under normal business practices, the amount of an expenditure is not known or readily ascertainable until receipt of a periodic bill, the date of the expenditure is the date the bill is received. Examples of expenditures to which this subsection is applicable are expenditures for use of electricity or for long-distance telephone calls.

(c) A political expenditure by credit card made during the period covered by a report required to be filed under <u>Section 254.064(b)</u> or (c), 254.124(b) or (c), or 254.154(b) or (c) of the Election Code, must be included in the report for the period during which the charge was made, not in the report for the period during which the statement from the credit card company was received.

(d) A political expenditure by credit card made during a period not covered by a report listed under subsection (c) of this section, must be included in the report for the period during which:

- (1) the charge was made; or
- (2) the person receives the credit card statement that includes the expenditure.

## §20.58. Disclosure of Political Expenditure.

(a) An expenditure that is not paid during the reporting period in which the obligation to pay the expenditure is incurred shall be reported on the Unpaid Incurred Obligations Schedule for the reporting period in which the obligation to pay is incurred.

(b) The use of political contributions to pay an expenditure previously disclosed on an Unpaid Incurred Obligations Schedule shall be reported on the appropriate disbursements schedule for the reporting period in which the payment is made. (c) The use of personal funds to pay an expenditure previously disclosed on an Unpaid Incurred Obligations Schedule shall be reported on the Political Expenditure Made from Personal Funds Schedule for the reporting period in which the payment is made.

(d) The effective date of this rule is January 1, 2015.

#### §20.59. Reporting Expenditure by Credit Card.

(a) A report of an expenditure charged to a credit card must be disclosed on the Expenditures Made to Credit Card Schedule and identify the vendor who receives payment from the credit card company.

(b) A report of a payment to a credit card company must be disclosed on the appropriate disbursements schedule and identify the credit card company receiving the payment.

#### §20.60. Reporting Political Expenditures for Processing Fees.

(a) Multiple political expenditures made to a single payee during a reporting period for fees to process political contributions may be itemized as a single expenditure, in an amount equal to the combined total amount of the expenditures, if all the expenditures are made to a single payee for the same purpose.

(b) The purpose of an expenditure reported under subsection (a) of this section must include the dates of the first and last of the multiple expenditures made to a single payee during the reporting period.

(c) For reporting purposes, the date of an expenditure reported under subsection (a) of this section is the date of the first expenditure made to the payee during the reporting period, as provided by \$20.57 (Time of Making Expenditure) of this title.

#### §20.61. Purpose of Expenditure.

(a) For reporting required under §254.031 of the Election Code, the purpose of an expenditure means:

(1) A description of the category of goods, services, or other thing of value for which an expenditure is made. Examples of acceptable categories include:

- (A) advertising expense;
- (B) accounting/banking;
- (C) consulting expense;
- (D) contributions/donations made by candidate/officeholder/political committee;
- (E) event expense;
- (F) fees;

- (G) food/beverage expense;
- (H) gifts/awards/memorials expense;
- (I) legal services;
- (J) loan repayment/reimbursement;
- (K) office overhead/rental expense;
- (L) polling expense;
- (M) printing expense;
- (N) salaries/wages/contract labor;
- (O) solicitation/fundraising expense;
- (P) transportation equipment and related expense;
- (Q) travel in district;
- (R) travel out of district;
- (S) other political expenditures; and

(2) A brief statement or description of the candidate, officeholder, or political committee activity that is conducted by making the expenditure and an additional indication if the expenditure is an officeholder expenditure for living in Austin, Texas. The brief statement or description must include the item or service purchased and must be sufficiently specific, when considered within the context of the description of the category, to make the reason for the expenditure clear. Merely disclosing the category of goods, services, or other thing of value for which the expenditure is made does not adequately describe the purpose of an expenditure.

(3) For purposes of this section, "consulting" means advice and strategy. "Consulting" does not include providing other goods or services, including without limitation media production, voter contact, or political advertising.

(b) An expenditure other than a reimbursement to a person, including a vendor, for more than one type of good or service must be reported by the filer as separate expenditures for each type of good or service provided by the person in accordance with this rule.

(c) The description of a political expenditure for travel outside of the state of Texas must provide the following:

(1) The name of the person or persons traveling on whose behalf the expenditure was made;

(2) The means of transportation;

(3) The name of the departure city or the name of each departure location;

(4) The name of the destination city or the name of each destination location;

(5) The dates on which the travel occurred; and

(6) The campaign or officeholder purpose of the travel, including the name of a conference, seminar, or other event.

(d) Except as provided by subsection (e) of this section, this rule applies to expenditures made on or after July 1, 2010.

(e) The requirement to include an additional indication if an expenditure is an officeholder expenditure for living in Austin, Texas, applies to an expenditure made on or after July 1, 2014.

(f) Comments:

The purpose of an expenditure must include both a description of the category of goods or services received in exchange for the expenditure and a brief statement or description of the candidate, officeholder, or political committee activity that is conducted by making the expenditure. A description of an expenditure that merely states the item or service purchased is not adequate because doing so does not allow a person reading the report to know the allowable activity for which an expenditure was made. The following is a list of examples that describe how the purpose of an expenditure may be reported under section 20.61. This list is for illustrative purposes only. It is intended to provide helpful information and to assist filers in reporting the purpose of an expenditure under this rule. However, it is not, and is not intended to be, an exhaustive or an exclusive list of how a filer may permissibly report the purpose of an expenditure under the candidate or officeholder to identify by name or affiliation an individual or group with whom the candidate or officeholder meets.

(1) Example: Candidate X is seeking the office of State Representative, District 2000. She purchases an airline ticket from ABC Airlines to attend a campaign rally within District 2000. The acceptable category for this expenditure is "travel in district." The candidate activity that is accomplished by making the expenditure is to attend a campaign rally. An acceptable brief statement is "airline ticket to attend campaign event."

(2) Example: Candidate X purchases an airline ticket to attend a campaign event outside of District 2000 but within Texas, the acceptable category is "travel out of district." The candidate activity that is accomplished by making the expenditure is to attend a campaign event. An acceptable brief statement is "airline ticket to attend campaign or officeholder event."

(3) Example: Candidate X purchases an airline ticket to attend an officeholder related seminar outside of Texas. The acceptable method for the purpose of this expenditure is by selecting the "travel out of district" category and completing the "Schedule T" (used to report travel outside of Texas).

(4) Example: Candidate X contracts with an individual to do various campaign related tasks such as work on a campaign phone bank, sign distribution, and staffing the office.

The acceptable category is "salaries/wages/contract labor." The candidate activity that is accomplished by making the expenditure is to compensate an individual working on the campaign. An acceptable brief statement is "contract labor for campaign services."

(5) Example: Officeholder X is seeking re-election and makes an expenditure to purchase a vehicle to use for campaign purposes and permissible officeholder purposes. The acceptable category is "transportation equipment and related expenses" and an acceptable brief description is "purchase of campaign/officeholder vehicle."

(6) Example: Candidate X makes an expenditure to repair a flat tire on a campaign vehicle purchased with political funds. The acceptable category is "transportation equipment and related expenses" and an acceptable brief description is "campaign vehicle repairs."

(7) Example: Officeholder X purchases flowers for a constituent. The acceptable category is "gifts/awards/memorials expense" and an acceptable brief description is "flowers for constituent."

(8) Example: Political Committee XYZ makes a political contribution to Candidate X. The acceptable category is "contributions/donations made by candidate/officeholder/political committee" and an acceptable brief description is "campaign contribution."

(9) Example: Candidate X makes an expenditure for a filing fee to get his name on the ballot. The acceptable category is "fees" and an acceptable brief description is "candidate filing fee."

(10) Example: Officeholder X makes an expenditure to attend a seminar related to performing a duty or engaging in an activity in connection with the office. The acceptable category is "fees" and an acceptable brief description is "attend officeholder seminar."

(11) Example: Candidate X makes an expenditure for political advertising to be broadcast by radio. The acceptable category is "advertising expense" and an acceptable brief description is "political advertising." Similarly, Candidate X makes an expenditure for political advertising to appear in a newspaper. The acceptable category is "advertising expense" and an acceptable brief description is "political advertising."

(12) Example: Officeholder X makes expenditures for printing and postage to mail a letter to all of her constituents, thanking them for their participation during the legislative session. Acceptable categories are "advertising expense" OR "printing expense" and an acceptable brief description is "letter to constituents."

(13) Example: Officeholder X makes an expenditure to pay the campaign office electric bill. The acceptable category is "office overhead/rental expense" and an acceptable brief description is "campaign office electric bill."

(14) Example: Officeholder X makes an expenditure to purchase paper, postage, and other supplies for the campaign office. The acceptable category is "office overhead/rental expense" and an acceptable brief description is "campaign office supplies."

(15) Example: Officeholder X makes an expenditure to pay the campaign office monthly rent. The acceptable category is "office overhead/rental expense" and an acceptable brief description is "campaign office rent."

(16) Example: Candidate X hires a consultant for fundraising services. The acceptable category is "consulting expense" and an acceptable brief description is "campaign services."

(17) Example: Candidate/Officeholder X pays his attorney for legal fees related to either campaign matters or officeholder matters. The acceptable category is "legal services" and an acceptable brief description is "legal fees for campaign" or "for officeholder matters."

(18) Example: Candidate/Officeholder X makes food and beverage expenditures for a meeting with her constituents. The acceptable category is "food/beverage expense" and an acceptable brief statement is "meeting with constituents."

(19) Example: Candidate X makes food and beverage expenditures for a meeting to discuss candidate issues. The acceptable category is "food/beverage expense" and an acceptable brief statement is "meeting to discuss campaign issues."

(20) Example: Officeholder X makes food and beverage expenditures for a meeting to discuss officeholder issues. The acceptable category is "food/beverage expense" and an acceptable brief statement is "meeting to discuss officeholder issues."

(21) Example: Candidate/Officeholder X makes food and beverage expenditures for a meeting to discuss campaign and officeholder issues. The acceptable category is "food/beverage expense" and an acceptable brief statement is "meeting to discuss campaign/officeholder issues."

## §20.62. Reporting Staff Reimbursement.

(a) Political expenditures made out of personal funds by a staff member of an officeholder, a candidate, or a political committee with the intent to seek reimbursement from the officeholder, candidate, or political committee that in the aggregate do not exceed \$6,910 during the reporting period may be reported as follows IF the reimbursement occurs during the same reporting period that the initial expenditure was made:

(1) the amount of political expenditures that in the aggregate exceed \$200 and that are made during the reporting period, the full names and addresses of the persons to whom the expenditures are made and the dates and purposes of the expenditures; and

(2) included with the total amount or a specific listing of the political expenditures of \$200 or less made during the reporting period.

(b) Except as provided by subsection (a) of this section, a political expenditure made out of personal funds by a staff member of an officeholder, a candidate, or a political committee with the intent to seek reimbursement from the officeholder, candidate, or political committee must be reported as follows:

(1) the aggregate amount of the expenditures made by the staff member as of the last day of the reporting period is reported as a loan to the officeholder, candidate, or political committee;

(2) the expenditure made by the staff member is reported as a political expenditure by the officeholder, candidate, or political committee; and

(3) the reimbursement to the staff member to repay the loan is reported as a political expenditure by the officeholder, candidate, or political committee.

## §20.63. Reporting the Use and Reimbursement of Personal Funds.

(a) A candidate is required to report a campaign expenditure from his or her personal funds.

(b) An officeholder is not required to report an officeholder expenditure from his or her personal funds unless he or she intends to be reimbursed from political contributions.

(c) A candidate or officeholder must report a political expenditure from his or her personal funds using one of the following methods:

(1) As a political expenditure made from personal funds reported on the political expenditure made from personal funds schedule;

(2) As a loan without depositing the personal funds in an account in which political contributions are held. The amount reported as a loan may not exceed the total amount actually spent in the reporting period. A political expenditure made from these funds must also be reported as a political expenditure made from political funds, not as made from personal funds; or

(3) If the candidate or officeholder deposits personal funds in an account in which political contributions are held, he or she must report that amount as a loan with an indication that personal funds were deposited in that account. A political expenditure made from an account in which political contributions are maintained must be reported as a political expenditure made from political funds, not as made from personal funds.

(d) A candidate or officeholder who makes political expenditures from his or her personal funds may reimburse those personal funds from political contributions only if:

(1) the expenditures were fully reported using one of the methods in subsection (c) of this section on the report covering the period during which the expenditures were made; and

(2) if the method in subsection (c)(1) of this section was used, the report disclosing the expenditures indicates that the expenditures are subject to reimbursement.

(e) A candidate's or officeholder's failure to comply with subsection (d) of this section may not be cured by filing a corrected report after the report deadline has passed.

(f) A candidate or officeholder who has complied with subsection (d) of this section and whose personal funds have been reimbursed from political contributions must report the amount of the

reimbursement as a political expenditure in the report covering the period during which the reimbursement was made.

(g) Section 22.21 of this title (relating to Additional Restrictions on Reimbursement of Personal Funds and Payments on Certain Loans) set limits on the amount of political expenditures from personal funds that a statewide officeholder may reimburse from political contributions.

## §20.64. Reporting the Forgiveness of a Loan or Settlement of a Debt.

(a) The forgiveness of a loan to a candidate, officeholder, or political committee is a reportable in-kind political contribution unless the loan does not constitute a contribution under §251.001(2) of the Election Code, and the forgiveness of the loan was made in the due course of business.

(b) The settlement of a debt owed by a candidate, officeholder, or political committee is a reportable in-kind political contribution unless the creditor is a commercial vendor that has treated the settlement in a commercially reasonable manner that reflects the usual and normal practice of the industry, and is typical of the terms the commercial vendor offers to political and non-political persons alike.

## §20.65. Reporting No Activity.

(a) As a general rule, a candidate or officeholder must file a report required by Subchapter C of this chapter (relating to Reporting Requirements for a Candidate) or Subchapter D of this chapter (relating to Reporting Requirements for an Officeholder Who Does Not Have a Campaign Treasurer Appointment on File), even if there has been no reportable activity during the period covered by the report.

(b) This general rule does not apply to:

- (1) special pre-election reports;
- (2) special session reports; or

(3) a local officeholder who does not have a campaign treasurer appointment on file and who does not accept more than \$1,010 in political contributions or make more than \$1,010 in political expenditures during the reporting period.

(c) If a required report will disclose that there has been no reportable activity during the reporting period, the filer shall submit only those pages of the report necessary to identify the filer and to swear to the lack of reportable activity.

#### §20.66. Discounts.

(a) A discount to a candidate, officeholder, or political committee is an in-kind political contribution unless the terms of the transaction reflect the usual and normal practice of the industry and are typical of the terms that are offered to political and non-political persons alike, or unless the discount is given solely in order to comply with §253.041 of the Election Code.

(b) The value of an in-kind contribution in the form of a discount is the difference between the fair market value of the goods or services at the time of the contribution and the amount charged.

#### §20.67. Reporting after the Death or Incapacity of a Filer.

(a) The responsibility to file reports required by this title survives the death or incapacity of a candidate or officeholder.

(b) The legal representative or the estate of a candidate or officeholder who has died, or the legal representative of a candidate who is incapacitated, shall file any reports due under Subchapter C of this chapter (relating to Reporting Requirements for a Candidate) or Subchapter D of this chapter (relating to Reporting Requirements for an Officeholder Who Does Not Have a Campaign Treasurer Appointment on File).

## Subchapter C. REPORTING REQUIREMENTS FOR A CANDIDATE

#### §20.201. Required Appointment of Campaign Treasurer.

A candidate must file a campaign treasurer appointment before accepting any campaign contributions or making or authorizing any campaign expenditures, including campaign expenditures from personal funds.

#### §20.203. Candidates for State Party Chair.

Because the state chair of a political party does not hold a public office, a candidate for state chair of a political party is not within the definition of "candidate" set out in §20.1(4) of this title (relating to Definitions). Nonetheless, a candidate for the state chair of a political party is subject to filing requirements as provided by Subchapter J of this chapter (relating to Reports by a Candidate for State Party Chair).

#### §20.205. Contents of Candidate's Campaign Treasurer Appointment.

Each candidate's campaign treasurer appointment shall include the following information:

- (1) the name of the candidate making the appointment;
- (2) the mailing address of the candidate making the appointment;
- (3) the office sought by the candidate making the appointment, if known;
- (4) the office held by the candidate, if any;
- (5) the name of the individual appointed campaign treasurer;
- (6) the campaign treasurer's residence or business street address;
- (7) the campaign treasurer's telephone number;

(8) a statement acknowledging awareness of the Government Code, Chapter 573, Subchapter C (concerning Nepotism Prohibitions); and

(9) the signature of the candidate making the appointment.

## §20.206. Transfer of Campaign Treasurer Appointment.

(a) If a candidate who has filed a campaign treasurer appointment decides to seek a different office that would require the appointment to be filed with another authority, a copy of the appointment certified by the authority with whom it was originally filed must be filed with the other authority in addition to the new campaign treasurer appointment.

(b) The original appointment terminates on the filing of the copy with the appropriate authority or on the 10th day after the date the decision to seek a different office is made, whichever is earlier.

#### §20.207. Termination of Campaign Treasurer Appointment.

(a) A candidate may terminate a campaign treasurer appointment by:

- (1) filing a campaign treasurer appointment for a successor campaign treasurer; or
- (2) filing a final report.

(b) A person may terminate his or her own status as campaign treasurer by immediately notifying both the appointing authority and the filing authority in writing.

(c) If a person terminates his or her own status as campaign treasurer, the termination is effective on the date the candidate receives the notice or on the date the filing authority actually receives the notice, whichever is later.

(d) Section 20.23 of this title (relating to Timeliness of Action by Mail) does not apply to subsection (c) of this section.

#### §20.209. Reporting Obligations Imposed on Candidate, Not Campaign Treasurer.

A candidate, not the candidate's campaign treasurer, is responsible for complying with this title.

#### §20.211. Semiannual Reports.

(a) A candidate shall file semiannual reports as provided by this section.

(b) One semiannual report is due no earlier than July 1 and no later than July 15.

(1) The period covered by a report under this subsection begins on the later of the following dates, as applicable:

(A) January 1;

(B) the first day after the period covered by the last report required by this subchapter (other than a special pre-election report or a special session report) or Subchapter D of this chapter (relating to Reporting Requirements for an Officeholder Who Does Not Have a Campaign Treasurer Appointment on File); or (C) the day the candidate's campaign treasurer appointment was filed, if this is the candidate's first report filed under this subchapter (other than a special preelection report or a special session report) or Subchapter D of this chapter.

(2) The period covered by a report under this subsection ends on June 30.

(c) One semiannual report is due no earlier than January 1 and no later than January 15.

(1) The period covered by a report under this subsection begins on the later of the following dates, as applicable:

(A) July 1;

(B) the first day after the period covered by the last report required by this subchapter (other than a special pre-election report or a special session report) or Subchapter D of this chapter; or

(C) the day the candidate's campaign treasurer appointment was filed, if this is the candidate's first report filed under this subchapter (other than a special preelection report or a special session report) or Subchapter D of this chapter.

(2) The period covered by a report under this subsection continues through December 31.

### §20.213. Pre-election Reports.

(a) A candidate who has an opponent on the ballot in an election must file two pre-election reports, except as provided by subsections (b), (e), and (f) of this section.

(b) A candidate who has declared the intention to file reports in accordance with \$20.217 of this title (relating to Modified Reporting) and who remains eligible to file under the modified schedule is not required to file pre-election reports.

(c) The first pre-election report must be received by the authority with whom the report is required to be filed not later than 30 days before election day. If this is the candidate's first report filed, the report covers a period that begins on the day the candidate's campaign treasurer appointment was filed. Otherwise the period begins on the first day after the period covered by the last report required by this subchapter (other than a special pre-election report or a special session report) or Subchapter D of this chapter (relating to Reporting Requirements for an Officeholder Who Does Not Have a Campaign Treasurer Appointment on File). The period covered by the report continues through the 40th day before the election.

(d) The second pre-election report must be received by the authority with whom the report is required to be filed not later than eight days before election day. The report covers the period that begins on the 39th day before the election and ends on the 10th day before the election.

(e) If a person becomes an opposed candidate during the period that begins on the 39th day before the election and ends on the 10th day before the election, the person shall file one preelection report. The report shall cover a period that begins on the day the candidate's campaign treasurer appointment was filed, if this is the candidate's first report filed, or on the first day after the period covered by the last report required by this subchapter (other than a special pre-election report or a special session report) or Subchapter D of this chapter. The period covered by the report ends on the 10th day before the election.

(f) If a person becomes an opposed candidate after the 10th day before the election, the person is not required to file pre-election reports. The person is required to file any special pre-election reports required by \$20.221 of this title (relating to Special Pre-Election Report by Certain Candidates).

## §20.215. Runoff Report.

(a) A candidate in a runoff election shall file a runoff report, except as provided by subsection (b) of this section.

(b) A candidate who has declared an intention to file reports in accordance with §20.217 of this title (relating to Modified Reporting) and who remains eligible to file under the modified schedule is not required to file a runoff report.

(c) A runoff report must be received by the authority with whom the report is required to be filed no later than the eighth day before the runoff election.

(d) A runoff report covers the period that begins on the ninth day before the date of the main election and ends on the 10th day before the runoff.

## §20.217. Modified Reporting.

(a) An opposed candidate who does not intend to accept more than \$1,010 in political contributions or make more than \$1,010 in political expenditures (excluding filing fees) in connection with any election in an election cycle may choose to file under the modified schedule.

(b) Under the modified schedule, an opposed candidate is not required to file pre-election reports or a runoff report.

(c) To select modified filing, a candidate must file a declaration of intent not to accept more than \$1,010 in political contributions or make more than \$1,010 in political expenditures (excluding filing fees) in connection with the election. The declaration must include a statement that the candidate understands that if either one of those limits is exceeded, the candidate will be required to file pre-election reports and, if necessary, a runoff report.

(d) A declaration under subsection (c) of this section is filed with the candidate's campaign treasurer appointment.

(e) To file under the modified schedule, a candidate must file the declaration required under subsection (c) of this section no later than the 30th day before the first election to which the declaration applies. A declaration filed under subsection (c) of this section is valid for one election cycle only.

(f) If an opposed candidate exceeds either of the \$1,010 limits, the candidate must file reports under \$20.213 of this title (relating to Pre-election Reports) and \$20.215 of this title (relating to Runoff Report).

(g) If an opposed candidate exceeds either of the \$1,010 limits after the 30th day before the election, the candidate must file a report not later than 48 hours after exceeding the limit. If this is the candidate's first report filed, the report covers a period that begins on the day the candidate's campaign treasurer appointment was filed. Otherwise, the period begins on the first day after the period covered by the last report required by this subchapter (other than a special pre-election report or a special session report) or Subchapter D of this chapter (relating to Reporting Requirements for an Officeholder Who Does Not Have a Campaign Treasurer Appointment on File). The period covered by the report continues through the day the candidate exceeded one of the limits for modified reporting.

## §20.219. Content of Candidate's Sworn Report of Contributions and Expenditures.

Semiannual reports, pre-election reports, and runoff reports must cover reportable activity during the reporting period and must include the following information:

- (1) the candidate's full name;
- (2) the candidate's address;
- (3) the office sought by the candidate, if known;
- (4) the identity and date of the election for which the report is filed, if known;
- (5) the campaign treasurer's name;
- (6) the campaign treasurer's telephone number;
- (7) the campaign treasurer's residence or business street address;

(8) for each political committee from which the candidate received notice under §20.319 of this title (relating to Notice to Candidate or Officeholder) or §20.421 of this title (relating to Notice to Candidate or Officeholder):

- (A) the committee's full name;
- (B) the committee's address;

(C) identification of the political committee as a general-purpose or a specificpurpose committee;

- (D) the full name of the committee's campaign treasurer; and
- (E) the address of the committee's campaign treasurer;

(9) on a separate page, the following information for each expenditure from political contributions made to a business in which the candidate has a participating interest of more than 10%, holds a position on the governing body of the business, or serves as an officer of the business:

(A) the full name of the business to which the expenditure was made;

(B) the address of the person to whom the expenditure was made;

(C) the date of the expenditure;

(D) the purpose of the expenditure; and

(E) the amount of the expenditure;

(10) for each person from whom the candidate accepted a political contribution (other than a pledge, loan, or a guarantee of a loan) of more than \$100 in value or political contributions (other than pledges, loans, or guarantees of loans) that total more than \$100 in value:

(A) the full name of the person making the contribution;

(B) the address of the person making the contribution;

(C) the total amount of contributions;

(D) the date each contribution was accepted; and

(E) a description of any in-kind contribution;

(11) for each person from whom the candidate accepted a pledge or pledges to provide more than \$100 in money or goods or services worth more than \$100:

(A) the full name of the person making the pledge;

(B) the address of the person making the pledge;

(C) the amount of each pledge;

(D) the date each pledge was accepted;

(E) a description of any goods or services pledged; and

(F) the total of all pledges accepted during the period for \$100 and less from a person, except those reported under subparagraphs (A)-(E) of this paragraph;

(12) for each person making a loan or loans to the candidate for campaign purposes, if the total amount loaned by the person during the period is more than \$100:

(A) the full name of the person or financial institution making the loan;

(B) the address of the person or financial institution making the loan;

(C) the amount of the loan;

(D) the date of the loan;

(E) the interest rate;

(F) the maturity date;

(G) the collateral for the loan, if any; and

(H) if the loan has guarantors:

(i) the full name of each guarantor;

- (ii) the address of each guarantor;
- (iii) the principal occupation of each guarantor;
- (iv) the name of the employer of each guarantor; and
- (v) the amount guaranteed by each guarantor;

(13) the total amount of loans accepted during the period for \$100 and less from persons other than financial institutions engaged in the business of making loans for more than one year, except for a loan reported under paragraph (12) of this section;

(14) for political expenditures made during the reporting period that total more than \$200 to a single payee, other than expenditures reported under paragraph (9) of this section:

(A) the full name of the person to whom each expenditure was made;

- (B) the address of the person to whom the expenditure was made;
- (C) the date of the expenditure;
- (D) the purpose of the expenditure; and
- (E) the amount of the expenditure;

(15) for each political expenditure of any amount made out of personal funds for which reimbursement from political contributions is intended:

(A) the full name of the person to whom each expenditure was made;

(B) the address of the person to whom the expenditure was made;

- (C) the date of the expenditure;
- (D) the purpose of the expenditure;
- (E) a declaration that the expenditure was made out of personal funds;
- (F) a declaration that reimbursement from political contributions is intended; and
- (G) the amount of the expenditure;

(16) for each non-political expenditure made from political contributions, other than expenditures reported under paragraph (9) of this section:

(A) the date of each expenditure;

(B) the full name of the person to whom the expenditure was made;

(C) the address of the person to whom the expenditure was made;

(D) the purpose of the expenditure; and

(E) the amount of the expenditure;

(17) for each other candidate or officeholder who benefits from a direct campaign expenditure made by the candidate during the reporting period:

(A) the name of the candidate or officeholder; and

(B) the office sought or held by the candidate or officeholder;

(18) for each political contribution from an out-of-state political committee, the information required by §22.7 of this title (relating to Contribution from Out-of-State Committee);

(19) any credit, interest, rebate, refund, reimbursement, or return of a deposit fee resulting from the use of a political contribution or an asset purchased with a political contribution that is received during the reporting period and the amount of which exceeds \$130;

(20) any proceeds of the sale of an asset purchased with a political contribution that is received during the reporting period and the amount of which exceeds \$130;

(21) any other gain from a political contribution that is received during the reporting period and the amount of which exceeds \$130;

(22) any investment purchased with a political contribution that is received during the reporting period and the amount of which exceeds \$130;

(23) the full name and address of each person from whom an amount described by paragraph (19), (20), (21), or (22) of this section is received, the date the amount is received, and the purpose for which the amount is received;

(24) the following total amounts:

(A) the total principal amount of all outstanding loans as of the last day of the reporting period;

(B) the total amount or an itemized listing of political contributions (other than pledges, loans, or guarantees of loans) of \$100 and less;

(C) the total amount of all political contributions (other than pledges, loans, or guarantees of loans);

(D) the total amount or an itemized listing of the political expenditures of \$200 and less; and

(E) the total amount of all political expenditures; and

(25) an affidavit, executed by the candidate, stating: "I swear, or affirm, that the accompanying report is true and correct and includes all information required to be reported by me under Title 15, Election Code."

# §20.220. Additional Disclosure for the Texas Comptroller of Public Accounts.

(a) For purposes of this section and §2155.003(e) of the Government Code, the term "vendor" means:

(1) a person, who during the comptroller's term of office, bids on or receives a contract under the comptroller's purchasing authority that was transferred to the comptroller by \$2151.004 of the Government Code; and

(2) an employee or agent of a person described by subsection (a)(1) of this section who communicates directly with the chief clerk, or an employee of the Texas Comptroller of Public Accounts who exercises discretion in connection with the vendor's bid or contract, about a bid or contract.

(b) Each report filed by the comptroller, or a specific-purpose committee created to support the comptroller, shall include:

(1) for each vendor whose aggregate campaign contributions equal or exceed \$650 during the reporting period, a notation that:

(A) the contributor was a vendor during the reporting period or during the 12month period preceding the last day covered by the report; and

(B) if the vendor is an individual, includes the name of the entity that employs or that is represented by the individual; and

(2) for each political committee directly established, administered, or controlled by a vendor whose aggregate campaign contributions equal or exceed \$650 during the reporting period, a notation that the contributor was a political committee directly established, administered, or controlled by a vendor during the reporting period or during the 12-month period preceding the last day covered by the report.

(c) The comptroller, or a specific-purpose committee created to support the comptroller, is considered to be in compliance with this section if:

(1) each written solicitation for a campaign contribution includes a request for the information required by subsection (b) of this section; and

(2) for each contribution that is accepted for which the information required by this section is not provided at least one oral or written request is made for the missing information. A request under this subsection:

(A) must be made not later than the 30th day after the date the contribution is received;

(B) must include a clear and conspicuous statement requesting the information required by subsection (b) of this section;

(C) if made orally, must be documented in writing; and

(D) may not be made in conjunction with a solicitation for an additional campaign contribution.

(d) The comptroller, or a specific-purpose committee created to support the comptroller, must report the information required by subsection (b) of this section that is not provided by the person making the political contribution and that is in the comptroller's or committee's records of political contributions or previous campaign finance reports required to be filed under Title 15 of the Election Code filed by the comptroller or committee.

(e) If the comptroller, or a specific-purpose committee created to support the comptroller, receives the information required by this section after the filing deadline for the report on which the contribution is reported the comptroller or committee must include the missing information on the next required campaign finance report.

(f) The disclosure required under subsection (b) of this section applies only to a contributor who was a vendor or a political committee directly established, administered, or controlled by a vendor on or after September 1, 2007.

# §20.221. Special Pre-Election Report by Certain Candidates.

(a) As provided by subsection (b) of this section, certain candidates must file reports about certain contributions accepted during the period that begins on the ninth day before an election and ends at noon on the day before an election. Reports under this section are known as "special pre-election" reports.

(b) An opposed candidate for an office specified by §252.005(1), Election Code, who, during the period described in subsection (a) of this section, accepts one or more political contributions from a person that in the aggregate exceed \$2,020 must file special pre-election reports.

(c) Except as provided in subsection (e) of this section, a candidate must file a special preelection report so that the report is received by the commission no later than the first business day after the candidate accepts a contribution from a person that triggers the requirement to file the special pre-election report.

(d) If, during the reporting period for special pre-election contributions, a candidate receives additional contributions from a person whose previous contribution or contributions have triggered the requirement to file a special pre-election report during that period, the candidate must file an additional special pre-election report for each such contribution. Except as provided

in subsection (e) of this section, each such special pre-election report must be filed so that it is received by the commission no later than the first business day after the candidate accepts the contribution.

(e) A candidate must file a special pre-election report that is exempt from electronic filing under \$254.036(c), Election Code, so that the report is received by the commission no later than 5 p.m. of the first business day after the candidate accepts a contribution from a person that triggers the requirement to file the special pre-election report.

(f) A candidate must file a special pre-election report for each person whose contribution or contributions made during the period for special pre-election reports exceed the threshold for special pre-election reports.

(g) A candidate must also report contributions reported on a special pre-election report on the next semiannual, pre-election, or runoff report filed, as applicable.

# §20.223. Form and Contents of Special Pre-Election Report.

(a) A special pre-election report shall be filed electronically as required by §254.036, Election Code, unless the report is exempt from electronic filing. A special pre-election report that is exempt from electronic filing under §254.036(c), Election Code, is not required to be on a form prescribed by the commission.

(b) A special pre-election report shall include the following information:

(1) the name of the candidate;

(2) the office sought by the candidate;

(3) the name of the person making a contribution or contributions that triggered the requirement to file a special pre-election report;

- (4) the address of the person making the contribution or contributions;
- (5) the amount of each contribution;
- (6) the date each contribution was accepted; and
- (7) a description of any in-kind contribution.

### §20.225. Special Session Reports.

(a) A candidate for a statewide office or for the legislature who accepts a political contribution during the period that begins on the date the governor signs a proclamation calling a special legislative session and ends on the date of final adjournment must file a special session report.

(b) A special session report must be filed with the commission no later than the 30th day after the date of final adjournment of the special session.

(c) A special session report is a report of contributions only, not expenditures. Expenditures made during the period covered by a special session report are required to be reported in the next applicable sworn report of contributions and expenditures.

(d) Contributions reported in a special session report are required to be reported in the next applicable sworn report of contributions and expenditures.

(e) A determination to accept or refuse a political contribution received during the period covered by a special session report shall be made no later than the third day after the date the contribution is received.

(f) A contribution that is refused under subsection (e) of this section must be returned no later than the 30th day after the date of final adjournment. A contribution not returned by that date will be deemed accepted.

(g) A candidate is not required to file a separate special session report if another report is due no later than the tenth day after the date a report required under this section would be due.

# §20.227. Contents of Special Session Report.

A special session report shall include the following information:

- (1) the candidate's name;
- (2) the candidate's address;
- (3) the office sought by the candidate;
- (4) the date each contribution was accepted;
- (5) the full name of each person making a contribution;
- (6) the address of each person making a contribution;
- (7) the amount of each contribution accepted during the period;
- (8) a description of any in-kind contribution accepted during the period; and

(9) an affidavit, executed by the candidate, stating: "I swear, or affirm, that the accompanying report is true and correct and includes all information required to be reported by me under Title 15, Election Code."

### §20.229. Final Report.

(a) A candidate who expects no further reportable activity in connection with his or her candidacy may file a final report at any time.

(b) The term "reportable activity" includes an expenditure to pay a campaign debt.

(c) Filing a final report terminates the candidate's campaign treasurer appointment and relieves the candidate of the responsibility for filing reports, except as provided by subsection (e) of this section.

(d) A former candidate may not accept campaign contributions or make campaign expenditures without a campaign treasurer appointment on file.

(e) A candidate who is not an officeholder when he or she files a final report under this section, and who retains unexpended political contributions, unexpended interest or other income from political contributions, assets purchased with political contributions or interest, or other income from political contributions is subject to the requirements of §§20.233, 20.235, 20.237, 20.239, 20.241, and 20.243 of this title (relating to Reporting Requirements for a Candidate).

(f) A candidate who is an officeholder when he or she files a final report under this section becomes subject to the reporting requirements set out in Subchapter D of this chapter (relating to Reporting Requirements for an Officeholder Who Does Not Have a Campaign Treasurer Appointment on File).

# §20.231. Contents of Final Report.

A final report must contain the following:

(1) the information listed in §20.219 of this title (relating to Content of Candidate's Sworn Report of Contributions and Expenditures);

(2) the following statement, signed by the candidate: "I do not expect any further political contributions or political expenditures in connection with my candidacy. I understand that designating a report as a final report terminates my campaign treasurer appointment. I also understand that I may not accept any campaign contributions or make any campaign expenditures without a campaign treasurer appointment on file.";

(3) if the candidate is not an officeholder, a statement that the candidate does or does not have unexpended contributions or unexpended interest or other income earned from political contributions;

(4) if the candidate is not an officeholder and has unexpended contributions or unexpended interest or income earned from political contributions, the following statement signed by the candidate: "I understand that I may not convert unexpended political contributions or unexpended interest or other income earned from political contributions to personal use. I also understand that I must file an annual report of unexpended contributions and that I may not retain unexpended contributions or unexpended interest or other income earned from political contributions longer than six years after filing this final report. Further, I understand that I must dispose of unexpended political contributions and unexpended interest or other income earned from political contributions in accordance with the requirements of Election Code, §254.204 (relating to Disposition of Unexpended Contributions)."; (5) if the candidate is not an officeholder, a statement that the candidate does or does not retain assets purchased with political contributions or interest or other income earned from political contributions;

(6) if the candidate is not an officeholder and retains assets purchased with political contributions or interest or other income from political contributions, the following statement signed by the candidate: "I understand that I may not convert assets purchased with political contributions or interest or other income earned from political contributions to personal use. I also understand that I must file an annual report of unexpended contributions and that I may not retain assets purchased with political contributions or interest or other income earned from political contributions or interest or other income earned from political contributions or interest or other income earned from political contributions longer than six years after filing this final report. I also understand that I must dispose of assets purchased with political contributions or interest or other income earned from political contributions in accordance with the requirements of Election Code, §254.204 (relating to Disposition of Unexpended Contributions)."; and

(7) if the candidate is an officeholder, a statement that the officeholder is aware that he or she remains subject to filing requirements applicable to an officeholder who does not have a campaign treasurer appointment on file.

# §20.233. Annual Report of Unexpended Contributions.

(a) A candidate who files a final report and is not an officeholder when he or she files a final report under §20.229 of this title (relating to Final Report) must file an annual report for each year that the former candidate retains unexpended political contributions, unexpended interest or other income earned from political contributions, or assets purchased with political contributions or interest or other income earned from political contributions.

(b) The report is due not earlier than January 1 and not later than January 15 of the year after a year in which the former candidate retained unexpended political contributions, unexpended interest or other income earned from political contributions, or assets purchased with political contributions or interest or other income earned from political contributions.

(c) The report is filed with the authority with whom the former candidate's campaign treasurer appointment was required to be filed.

(d) The requirement to file annual reports ends after:

(1) all unexpended political contributions, unexpended interest or other income earned from political contributions, or assets purchased with political contributions or interest or other income earned from political contributions have been disbursed and reported on an annual report; or

(2) the former candidate has complied with §20.237 of this title (relating to Disposition of Unexpended Contributions) and §20.239 of this title (relating to Report of Final Disposition of Unexpended Contributions).

## §20.235. Contents of Annual Report.

An annual report of unexpended contributions shall include the following information:

- (1) the candidate's full name;
- (2) the candidate's address;

(3) for each payment made by the candidate from unexpended political contributions, unexpended interest or other income earned from political contributions, or assets purchased with political contributions or interest or other income earned from political contributions during the previous year:

(A) the full name of each person to whom a payment was made;

(B) the address of each person to whom a payment was made;

- (C) the date of each payment;
- (D) the nature of the goods or services for which the payment was made; and
- (E) the amount of each payment;

(4) the total amount of unexpended political contributions as of December 31 of the previous year;

(5) the total amount of interest and other income earned on unexpended political contributions during the previous year; and

(6) an affidavit, executed by the candidate, stating, "I swear, or affirm, that the accompanying report is true and correct and includes all information required to be reported by me under Title 15, Election Code."

### §20.237. Final Disposition of Unexpended Contributions.

(a) A former candidate who was not an officeholder at the time he or she filed a final report may not retain unexpended political contributions, unexpended interest or other income earned from political contributions, or assets purchased with political contributions or interest or other income earned from political contributions for more than six years after the date of the final report, except as provided by subsection (f) of this section.

(b) During the six-year period after the final report is filed, a former candidate may disburse unexpended political contributions, unexpended interest or other income earned from political contributions, or assets purchased with political contributions or interest or other income earned from political contributions to one of the following:

(1) the political party with which the person was affiliated when the person's name last appeared on a ballot;

(2) a candidate or political committee, subject to the reporting requirements of §20.243 of this title (relating to Contribution of Unexpended Political Contributions to Candidate or Political Committee);

(3) the Comptroller of Public Accounts, for deposit in the state treasury for use in financing primary elections;

(4) one or more persons from whom political contributions were received, with contributions to a person not to exceed the aggregate amount the former candidate accepted from that person during the last two years that the candidate accepted political contributions;

(5) a recognized, tax-exempt charitable organization; or

(6) a public or private post-secondary educational institution or an institution of higher education, as defined by the Education Code, §61.003(8) (concerning Definitions), solely for the purpose of assisting or creating a scholarship program.

(c) A former candidate may not convert unexpended political contributions, unexpended interest or other income earned from political contributions, or assets purchased with political contributions or interest or other income earned from political contributions to personal use.

(d) At the end of the six-year period after the final report is filed, a former candidate must dispose of unexpended political contributions, unexpended interest or other income earned from political contributions, or assets purchased with political contributions or interest or other income earned from political contributions in one of the ways listed in subsection (b) of this section.

(e) A former candidate must make the disposition required by subsection (d) of this section by the 10th day after the end of the six-year period.

(f) The six-year period prescribed by subsection (a) of this section ceases to run if the former candidate files a new campaign treasurer appointment during the period.

# §20.239. Report of Final Disposition of Unexpended Contributions.

(a) A person required by §20.237 of this title (relating to Final Disposition of Unexpended Contributions) to dispose of unexpended contributions, unexpended interest or other income earned from political contributions, or assets purchased with political contributions or interest or other income earned from political contributions at the end of the period for retaining such funds is required to file a report of the disposition of such funds.

(b) The report must be filed no later than the 30th day after the end of the six-year period prescribed by 20.237(a) of this title (relating to Final Disposition of Unexpended Contributions).

(c) The report shall be filed with the authority with whom the person's campaign treasurer appointment was required to be filed.

(d) The report shall cover the period that begins on the first day after the period covered by the last annual report required through the day a report under this section is filed.

# §20.241. Contents of Report of Final Disposition of Unexpended Contributions.

A report of final disposition of unexpended contributions shall include the following information:

- (1) the candidate's full name;
- (2) the candidate's address;

(3) the full name of each person to whom a payment from unexpended political contributions, unexpended interest or other income earned from political contributions, or assets purchased with political contributions or interest or other income earned from political contributions was made;

(4) the address of each person to whom such a payment was made;

- (5) the date of each payment;
- (6) the nature of the goods or services for which the payment was made;
- (7) the amount of each payment; and

(8) an affidavit, executed by the candidate, stating: "I swear, or affirm, that the accompanying report is true and correct and includes all information required to be reported by me under Title 15, Election Code."

# **§20.243.** Contribution of Unexpended Political Contributions to Candidate or Political Committee.

(a) A former candidate who has filed a final report and who contributes unexpended political contributions, unexpended interest or other income earned from political contributions, or assets purchased with political contributions or interest or other income earned from political contributions to a candidate or political committee must report the contribution on an annual report of unexpended contributions or on a report of final disposition of unexpended contributions, as applicable. The former candidate must also report the contribution under subsection (b) of this section.

(b) A former candidate who has filed a final report and who contributes unexpended political contributions, unexpended interest or other income earned from political contributions, or assets purchased with political contributions or interest or other income earned from political contributions to a candidate or political committee must report each contribution to the filing authority with whom the candidate or political committee receiving the contributions files reports. The contribution must be reported on the form used for reports of contributions and expenditures by specific-purpose committees. The report should be filed by the due date for the report in which the candidate or political committee receiving the contribution must report the receipt of the contribution.

# Subchapter D. REPORTING REQUIREMENTS FOR AN OFFICEHOLDER WHO DOES NOT HAVE A CAMPAIGN TREASURER APPOINTMENT ON FILE

### §20.271. Officeholders Covered.

(a) The provisions in this subchapter that apply to an officeholder apply only to a person who holds an elective public office in the state and to the secretary of state.

(b) For purposes of this subchapter, a statewide officer-elect or a member-elect of the legislature is considered to be an officeholder beginning on the day after the date of the general or special election at which the officer-elect or member-elect was elected.

(c) An officeholder who has a campaign treasurer appointment on file is a candidate for filing purposes and shall file under Subchapter C of this chapter (relating to Reporting Requirements for a Candidate) rather than under this subchapter.

### §20.273. Semiannual Reports of Contributions and Expenditures.

(a) Except as provided by §20.275 of this title (relating to Exception from Filing Requirement for Certain Local Officeholders), an officeholder shall file semiannual reports of contributions and expenditures as provided by this section.

(b) One semiannual report is due no earlier than July 1 and no later than July 15.

(1) The period covered by a report under this subsection begins on the later of the following dates, as applicable:

(A) January 1;

(B) the first day after the period covered by the last report required by this chapter, whether that report was filed by the officeholder in his or her status as an officeholder or as a candidate; or

(C) the day the officeholder took office, if the report is the first report filed by the officeholder under this chapter.

(2) The period covered by a report under this subsection ends on June 30.

(c) One semiannual report is due no earlier than January 1 and no later than January 15.

(1) The period covered by a report under this subsection begins on the later of the following dates, as applicable:

(A) July 1;

(B) the first day after the period covered by the last report required by this chapter, whether that report was filed by the officeholder in his or her status as an officeholder or as a candidate; or

(C) the day the officeholder took office, if the report is the first report filed by the officeholder under this chapter.

(2) The period covered by a report under this subsection ends on December 31.

# §20.275. Exception from Filing Requirement for Certain Local Officeholders.

An officeholder is not required to file a semiannual report of contributions and expenditures if the officeholder:

(1) is required to file with an authority other than the commission;

(2) does not have a campaign treasurer appointment on file; and

(3) does not accept more than \$1,010 in political contributions or make more than \$1,010 in political expenditures during the reporting period.

# §20.277. Appointment by Officeholder of Campaign Treasurer.

(a) An officeholder who appoints a campaign treasurer after a period in which the officeholder did not have a campaign treasurer appointment on file must file a sworn report of contributions and expenditures no later than 15 days after the date the campaign treasurer appointment was filed.

(b) A report required by this section covers a period that begins on the later of the following dates, as applicable:

(1) the first day after the period covered by the last report filed under this chapter, whether that report was filed by the officeholder in his or her status as an officeholder or as a candidate; or

(2) the day the officeholder took office, if the report is the first report filed by the officeholder under this chapter.

(c) The period covered by a report required by this section ends on the day the campaign treasurer appointment was filed.

(d) After an officeholder files a campaign treasurer appointment, the officeholder is a candidate for filing purposes and shall file under Subchapter C of this chapter (relating to Reporting Requirements for a Candidate) rather than under this subchapter.

# §20.279. Contents of Officeholder's Sworn Report of Contributions and Expenditures.

An officeholder's semiannual report of contributions and expenditures required by this subchapter must cover reportable activity during the reporting period and must include the following information:

- (1) the officeholder's full name;
- (2) the officeholder's address;

(3) the office held by the officeholder;

(4) for each political committee from which the officeholder received notice under §20.319 of this title (relating to Notice to Candidate or Officeholder) or §20.421 of this title (relating to Notice to Candidate or Officeholder):

(A) the committee's full name;

(B) the committee's address;

(C) identification of the political committee as a general-purpose or a specificpurpose committee;

(D) the full name of the committee's campaign treasurer; and

(E) the address of the committee's campaign treasurer;

(5) on a separate page, the following information for each expenditure from political contributions made to a business in which the officeholder has a participating interest of more than 10%, holds a position on the governing body of the business, or serves as an officer of the business:

(A) the full name of the business to which the expenditure was made;

(B) the address of the business to which the expenditure was made;

(C) the date of the expenditure;

(D) the purpose of the expenditure; and

(E) the amount of the expenditure;

(6) for each person from whom the officeholder accepted a political contribution (other than a pledge, loan, or a guarantee of a loan) of more than \$100 in value or political contributions (other than pledges, loans, or guarantees of loans) that total more than \$100 in value:

(A) the full name of the person making the contribution;

(B) the address of the person making the contribution;

(C) the total amount of contributions;

(D) the date each contribution was accepted; and

(E) a description of any in-kind contribution;

(7) for each person from whom the officeholder accepted a pledge or pledges to provide more than \$100 in money or goods or services worth more than \$100:

(A) the full name of the person making the pledge;

(B) the address of the person making the pledge;

(C) the amount of each pledge;

- (D) the date each pledge was accepted; and
- (E) a description of any goods or services pledged;

(8) the total of all pledges accepted during the period for \$100 and less from a person, except those reported under paragraph (7) of this section;

(9) for each person making a loan or loans to the officeholder for officeholder purposes, if the total amount loaned by the person during the period is more than \$100:

- (A) the full name of the person or financial institution making the loan;
- (B) the address of the person or financial institution making the loan;
- (C) the amount of the loan;
- (D) the date of the loan;
- (E) the interest rate;
- (F) the maturity date;
- (G) the collateral for the loan, if any; and
- (H) if the loan has guarantors:
  - (i) the full name of each guarantor;
  - (ii) the address of each guarantor;
  - (iii) the principal occupation of each guarantor;
  - (iv) the name of the employer of each guarantor; and
  - (v) the amount guaranteed by each guarantor;

(10) the total amount of loans accepted during the period for \$100 and less from persons other than financial institutions engaged in the business of making loans for more than one year, except those reported under paragraph (9) of this section;

(11) for political expenditures made during the reporting period that total more than \$200 to a single payee, other than expenditures reported under paragraph (5) of this section:

(A) the full name of the person to whom each expenditure was made;

(B) the address of the person to whom the expenditure was made;

(C) the date of the expenditure;

- (D) the purpose of the expenditure; and
- (E) the amount of the expenditure;

(12) for each political expenditure of any amount made out of personal funds for which reimbursement from political contributions is intended:

(A) the full name of the person to whom each expenditure was made;

- (B) the address of the person to whom the expenditure was made;
- (C) the date of each expenditure;
- (D) the purpose of the expenditure;
- (E) a declaration that the expenditure was made from personal funds;
- (F) a declaration that reimbursement from political contributions is intended; and
- (G) the amount of the expenditure;

(13) for each non-political expenditure made from political contributions, other than expenditures reported under paragraph (5) of this section:

- (A) the date of each expenditure;
- (B) the full name of the person to whom the expenditure was made;
- (C) the address of the person to whom the expenditure was made;
- (D) the purpose of the expenditure; and
- (E) the amount of the expenditure;

(14) for each candidate or other officeholder who benefits from a direct campaign expenditure made by the officeholder during the reporting period:

(A) the name of the candidate or officeholder; and

(B) the office sought or held by the candidate or officeholder;

(15) for each political contribution from an out-of-state political committee, the information required by §22.7 of this title (relating to Contribution from Out-of-State Committee);

(16) any credit, interest, rebate, refund, reimbursement, or return of a deposit fee resulting from the use of a political contribution or an asset purchased with a political contribution that is received during the reporting period and the amount of which exceeds \$130;

(17) any proceeds of the sale of an asset purchased with a political contribution that is received during the reporting period and the amount of which exceeds \$130;

(18) any other gain from a political contribution that is received during the reporting period and the amount of which exceeds \$130;

(19) any investment purchased with a political contribution that is received during the reporting period and the amount of which exceeds \$130;

(20) the full name and address of each person from whom an amount described by paragraph (16), (17), (18), or (19) of the section is received, the date the amount is received, and the purpose for which the amount is received;

(21) the following total amounts:

(A) the total principal amount of all outstanding loans as of the last day of the reporting period;

(B) the total amount or an itemized listing of political contributions (other than pledges, loans, or guarantees of loans) of \$100 and less;

(C) the total amount of all political contributions (other than pledges, loans, or guarantees of loans);

(D) the total amount or an itemized listing of the political expenditures of \$200 and less; and

(E) the total amount of all political expenditures; and

(22) an affidavit, executed by the officeholder, stating: "I swear, or affirm, that the accompanying report is true and correct and includes all information required to be reported by me under Title 15, Election Code."

### §20.281. Special Session Report by Certain Officeholders.

(a) A statewide officeholder or member of the legislature who accepts a political contribution during the period that begins on the date the governor signs a proclamation calling a special legislative session and ends on the date of final adjournment must file a special session report.

(b) A special session report is a report of contributions only, not expenditures. Expenditures made during the period covered by a special session report are required to be reported in the next applicable sworn report of contributions and expenditures.

(c) Contributions reported in a special session report are required to be reported in the next applicable sworn report of contributions and expenditures.

(d) A special session report must be filed with the commission no later than the 30th day after the date of final adjournment of the special session.

(e) A determination to accept or refuse a political contribution received during the period covered by a special session report shall be made no later than the third day after the date the contribution is received.

(f) A contribution that is refused under subsection (e) of this section must be returned no later than the 30th day after the date of final adjournment. A contribution not returned by that date will be deemed accepted.

(g) An officeholder is not required to file a separate special session report if another report is due not later than the 10th day after the date a report required under this section would be due.

# §20.283. Contents of Special Session Report.

A report required by §20.281 of this title (relating to Special Session Report by Certain Officeholders) shall include the following information:

- (1) the officeholder's name;
- (2) the officeholder's address;
- (3) the office held;
- (4) the date each contribution was accepted;
- (5) the name of each person making a contribution;
- (6) the address of each person making a contribution;
- (7) the amount of each contribution accepted during the period;
- (8) a description of any in-kind contribution; and

(9) an affidavit, executed by the officeholder, stating: "I swear, or affirm, that the accompanying report is true and correct and includes all information required to be reported by me under Title 15, Election Code."

### §20.285. Annual Report of Unexpended Contributions by Former Officeholder.

(a) A person who ceases to be an officeholder at a time when he or she does not have a campaign treasurer appointment on file must file an annual report if he or she has unexpended political contributions, unexpended interest or other income earned from political contributions, or assets purchased with political contributions or interest or other income earned from political contributions after filing the last required report as an officeholder.

(b) A report under this section shall be filed not earlier than January 1 and not later than January 15 of each year following the year in which the former officeholder filed the last required report as an officeholder, unless the requirement to file annual reports has ended as provided by subsection (d) of this section.

(c) The report is filed with the authority with whom the former officeholder's campaign treasurer appointment was required to be filed.

(d) The requirement to file annual reports ends after:

(1) all political contributions, unexpended interest or other income earned from political contributions, or assets purchased with political contributions or interest or other income earned from political contributions have been disbursed and reported on an annual report; or

(2) the former officeholder has complied with §20.289 of this title (relating to Disposition of Unexpended Contributions) and §20.291 of this title (relating to Report of Final Disposition of Unexpended Contributions).

### §20.287. Contents of Annual Report.

(a) An annual report of unexpended contributions shall include the following information:

(1) the officeholder's full name;

(2) the officeholder's address;

(3) for each payment made by the officeholder from unexpended political contributions, unexpended interest or other income earned from political contributions, or assets purchased with political contributions or interest or other income earned from political contributions during the previous year:

- (A) the full name of each person to whom a payment was made;
- (B) the address of each person to whom a payment was made;
- (C) the date of each payment;

(D) the nature of the goods or services for which the payment was made; and

(E) the amount of the payment;

(4) the total amount of unexpended political contributions as of December 31 of the previous year;

(5) the total amount of interest and other income earned on unexpended political contributions during the previous year; and

(6) an affidavit, executed by the former officeholder, stating: "I swear, or affirm, that the accompanying report is true and correct and includes all information required to be reported by me under Title 15, Election Code."

# §20.289. Disposition of Unexpended Contributions.

(a) A former officeholder who did not have a campaign treasurer appointment on file at the time he or she ceased to be an officeholder may not retain unexpended political contributions, unexpended interest or other income earned from political contributions, or assets purchased with political contributions or interest or other income earned from political contributions for more than six years after the date he or she ceased to be an officeholder, except as provided by subsection (f) of this section.

(b) During the six-year period after the date a former officeholder ceased to be an officeholder, the former officeholder covered by subsection (a) of this section may disburse unexpended political contributions, unexpended interest or other income earned from political contributions, or assets purchased with political contributions or interest or other income earned from political contributions for a purpose listed in §20.237 of this title (relating to Final Disposition of Unexpended Contributions).

(c) A former officeholder may not convert political contributions, unexpended interest or other income earned from political contributions, or assets purchased with political contributions or interest or other income earned from political contributions to personal use.

(d) At the end of the six-year period, a former officeholder covered by subsection (a) of this section must dispose of unexpended political contributions, unexpended interest or other income earned from political contributions, and assets purchased with political contributions or interest or other income earned from political contributions in one of the ways listed in §20.237 of this title (relating to Final Disposition of Unexpended Contributions).

(e) A former officeholder must make the disposition required by subsection (c) of this section by the 10th day after the end of the six-year period.

(f) The six-year period prescribed by subsection (a) of this section ceases to run if the former officeholder files a campaign treasurer appointment during the period.

# §20.291. Report of Final Disposition of Unexpended Contributions.

(a) A former officeholder who disposes of unexpended contributions, unexpended interest or other income earned from political contributions, or assets purchased with political contributions or interest or other income earned from political contributions under §20.289 of this title (relating to Disposition of Unexpended Contributions) is required to file a report of the final disposition.

(b) A report of final disposition of unexpended contributions must be filed no later than the 30th day after the end of the six-year period prescribed by 20.289(a) of this title.

(c) The report shall be filed with the authority with whom the former officeholder's last required report as an officeholder was required to be filed.

(d) The report shall cover the period that begins on the first day after the period covered by the last report required through the day a report under this section is filed.

# §20.293. Contents of Report of Final Disposition of Unexpended Contributions.

(a) A report of final disposition of unexpended contributions shall include the following information:

(1) the officeholder's full name;

(2) the officeholder's address;

(3) the full name of each person to whom a payment from unexpended political contributions, unexpended interest or other income earned from political contributions, or assets purchased with political contributions or interest or other income earned from political contributions was made;

(4) the address of each person to whom such a payment was made;

(5) the date of each payment;

(6) the nature of the goods and services received for each payment;

(7) the amount of each payment; and

(8) an affidavit, executed by the former officeholder, stating: "I swear, or affirm, that the accompanying report is true and correct and includes all information required to be reported by me under Title 15, Election Code."

# **§20.295.** Contribution of Unexpended Political Contributions to Candidate or Political Committee.

(a) A former officeholder who contributes unexpended political contributions, unexpended interest or other income earned from political contributions, or assets purchased with political contributions or interest or other income earned from political contributions to a candidate or political committee must report the contribution on an annual report of unexpended contributions or on a report of final disposition of unexpended contributions, as applicable. The former officeholder must also report the contribution under subsection (b) of this section.

(b) A former officeholder who contributes unexpended political contributions, unexpended interest or other income earned from political contributions, or assets purchased with political contributions or interest or other income earned from political contributions to a candidate or political committee must report each contribution to the filing authority with whom the candidate or political committee receiving the contribution files reports.

(1) The former officeholder must report such contributions on the form used for reports of contributions and expenditures a specific-purpose committee.

(2) The former officeholder must file the report by the due date for the report in which the candidate or political committee receiving the contribution must report the receipt of the contribution.

# Subchapter E. REPORTS BY A SPECIFIC-PURPOSE COMMITTEE

#### §20.301. Thresholds for Campaign Treasurer Appointment.

(a) A specific-purpose committee may not accept political contributions exceeding \$980 and may not make or authorize political expenditures exceeding \$980 without filing a campaign treasurer appointment with the appropriate filing authority.

(b) A specific-purpose committee may not knowingly make or authorize campaign contributions or campaign expenditures exceeding \$980 to support or oppose a candidate in a primary or general election for an office listed below unless the committee's campaign treasurer appointment as filed not later than the 30th day before the appropriate election day:

- (1) a statewide office;
- (2) a seat in the state legislature;
- (3) a seat on the State Board of Education;
- (4) a multi-county district office; or
- (5) a judicial district office filled by voters of only one county.

#### §20.303. Appointment of Campaign Treasurer.

(a) A specific-purpose committee may appoint a campaign treasurer at any time before exceeding the thresholds described in §20.301(a) of this title (relating to Thresholds for Campaign Treasurer Appointment).

(b) After a specific-purpose committee appoints a campaign treasurer, the campaign treasurer must comply with all the requirements of this subchapter, even if the committee has not yet exceeded \$980 in political contributions or expenditures.

(c) With the exception of the campaign treasurer appointment, the individual named as a committee's campaign treasurer is legally responsible for filing all reports of the specific-purpose committee, including a report following the termination of his or her appointment as campaign treasurer.

#### §20.305. Appointing an Assistant Campaign Treasurer.

(a) A specific-purpose committee for supporting or opposing a candidate for an office specified by §252.005(1), Election Code or a statewide or district measure may appoint an assistant campaign treasurer by written appointment filed with the commission.

(b) A statewide measure is a measure to be voted on by all eligible voters in the state.

(c) A district measure is a measure to be voted on by the voters of a district.

(d) The assistant campaign treasurer has the same authority as the campaign treasurer. However, if the campaign treasurer appointment is terminated the assistant campaign treasurer no longer has authority to act as the campaign treasurer.

(e) The campaign treasurer, not the assistant campaign treasurer, is liable for any penalties assessed by the commission for late reports or incomplete reports or for failure to file a report.

(f) Section 20.315 of this title (relating to Termination of Campaign Treasurer Appointment) and §20.317 of this title (relating to Termination Report) apply to the appointment and removal of an assistant campaign treasurer.

# §20.307. Name of Specific-Purpose Committee.

The name of a specific-purpose committee that supports a candidate for or an officeholder of an office specified by §252.005(1), Election Code must include the full name of that candidate or officeholder.

# §20.309. Contents of Specific-Purpose Committee Campaign Treasurer Appointment.

A campaign treasurer appointment for a specific-purpose committee shall include the following information:

- (1) the full name of the specific-purpose committee;
- (2) the address of the specific-purpose committee;
- (3) the full name of the person appointing the campaign treasurer;

(4) the following information for the individual appointed campaign treasurer and, if an assistant campaign treasurer is appointed, for that individual also:

(A) the individual's full name;

(B) the individual's residence or business street address;

(C) if the individual's mailing address is different from the street address provided, the mailing address for the individual; and

(D) the individual's telephone number;

(5) for each candidate supported or opposed by the specific-purpose committee:

(A) the full name of the candidate;

(B) the office sought by the candidate; and

(C) an indication whether the specific-purpose committee supports or opposes the candidate;

(6) for each officeholder assisted by the specific-purpose committee:

(A) the full name of the officeholder;

(B) the office held by the officeholder; and

- (C) an indication that the specific-purpose committee assists the officeholder;
- (7) for each measure supported or opposed by the specific-purpose committee:

(A) a description of the measure; and

(B) an indication whether the specific-purpose committee supports or opposes the measure; and

(8) the signature of the individual appointed campaign treasurer.

### §20.311. Updating Certain Information on the Campaign Treasurer Appointment.

(a) Except as provided by subsection (b) of this section, if there is a change in any information that is required to be reported in a specific-purpose committee's campaign treasurer appointment, the campaign treasurer must notify the filing authority of the change no later than the 10th day after the date on which the change occurs.

(b) The campaign treasurer must report a change in the name of or office sought by a candidate whom the specific-purpose committee supports or opposes within 24 hours of the change.

### §20.313. Converting to a General-Purpose Committee.

(a) A specific-purpose committee that changes its operation and becomes a general-purpose committee is subject to the requirements applicable to a general-purpose committee as of the date it files its campaign treasurer appointment as a general-purpose committee with the commission.

(b) The campaign treasurer of a specific-purpose committee that becomes a general-purpose committee must deliver written notice of its change in status to the authority with whom the committee was required to file as a specific-purpose committee.

(c) The notice required under subsection (b) of this section is due no later than the next deadline for filing a report under this subchapter that:

(1) occurs after the committee's change in status; and

(2) would be applicable to the political committee if it were still a specific-purpose committee.

(d) The notice must state that future reports will be filed with the commission.

(e) The notice required under subsection (b) of this section is in addition to the requirement that the new general-purpose committee file a campaign treasurer appointment with the commission before it exceeds \$980 in political expenditures or \$980 in political contributions as a general-purpose committee.

# §20.315. Termination of Campaign Treasurer Appointment.

(a) A specific-purpose committee may terminate a campaign treasurer appointment at any time by:

- (1) notifying the filing authority in writing of the termination;
- (2) filing a campaign treasurer appointment for a successor campaign treasurer; or
- (3) filing a dissolution report.

(b) A committee's campaign treasurer may resign by immediately notifying both the appointing authority and the filing authority in writing.

(c) Except as provided by subsection (e) of this section, if the campaign treasurer resigns or otherwise leaves the position, the termination is effective on the date the committee actually receives the notice or on the date the filing authority actually receives the notice, whichever is later.

(d) Section 20.23 of this title (relating to Timeliness of Action by Mail) does not apply to subsection (c) of this section.

(e) For purposes of the termination report required by \$20.317 of this title (relating to Termination Report), a campaign treasurer's resignation is effective on the date the treasurer resigns as provided by subsection (b) of this section.

(f) Section 20.23 of this title (relating to Timeliness of Action by Mail) applies to subsection (e) of this section.

(g) A termination of a specific-purpose committee's campaign treasurer appointment and the filing of the termination report by themselves do not dissolve the specific-purpose committee. A specific-purpose committee can be dissolved only by filing a dissolution report.

(h) For purposes of this section, the appropriate filing authority for a campaign treasurer appointment of a specific-purpose committee created to support or oppose a measure on the issuance of bonds by a school district is the secretary of the school board (or the presiding officer if the school board has no secretary), except that the commission is the appropriate filing authority for a dissolution report.

### §20.317. Termination Report.

(a) If the campaign treasurer appointment of a specific-purpose committee is terminated, the campaign treasurer whose appointment was terminated shall file a termination report that contains the information listed in §20.331 of this title (relating to Contents of Specific-Purpose Committee Sworn Report of Contributions and Expenditures).

(b) A termination report is not required if the termination occurs on the last day of a reporting period under this subchapter and the campaign treasurer files a report for that period as provided by this subchapter.

(c) A termination report covers a period that begins on the day after the period covered by the last report of contributions and expenditures required to be filed under this subchapter (other than a special pre-election report or a special session report) or the day the campaign treasurer appointment was filed (if the committee has not yet filed a report of contributions and expenditures). The period covered by the report ends on the day the termination of the campaign treasurer appointment is effective.

(d) The report shall be filed not later than the 10th day after the date the termination of the campaign treasurer appointment is effective.

(e) Activity reported in a termination report is not required to be included in any subsequent report of the specific-purpose committee that is filed under this subchapter.

# §20.319. Notice to Candidate or Officeholder.

(a) The campaign treasurer of a specific-purpose committee that accepts political contributions or makes political expenditures for a candidate or officeholder shall notify the affected candidate or officeholder of that fact in accordance with this section.

(b) This section does not apply to a specific-purpose committee that has not appointed a campaign treasurer in accordance with §20.303(b) of this title (relating to Appointment of Campaign Treasurer).

(c) The notice required by this section shall be in writing and shall include:

(1) the full name of the specific-purpose committee;

- (2) the address of the specific-purpose committee;
- (3) the full name of the specific-purpose committee's campaign treasurer;
- (4) the address of the specific-purpose committee's campaign treasurer;
- (5) a statement that the committee is a specific-purpose committee; and
- (6) a statement that the specific-purpose committee has accepted political contributions or has made political expenditures on behalf of the candidate or officeholder.

(d) The notice required by this section shall be delivered no later than the end of the reporting period in which the reportable activity occurs.

### §20.321. Involvement in More Than One Election by Certain Specific-Purpose Committees.

A specific-purpose committee that supports or opposes more than one candidate or measure may be required to file reports covering overlapping periods. If so, the committee is only required to report activity occurring during the period of overlap on the first report on which the activity is required to be reported.

## §20.323. Semiannual Reports.

(a) The campaign treasurer of a specific-purpose committee shall file semiannual reports as provided by this section.

(b) One semiannual report is due no earlier than July 1 and no later than July 15.

(1) The period covered by a report under this subsection begins on the later of the following dates, as applicable:

(A) January 1;

(B) the day the committee's campaign treasurer appointment was filed, if this is the committee's first report filed under this subchapter (other than a special preelection report or a special session report); or

(C) the first day after the period covered by the last report required by this subchapter (other than a special pre-election report or a special session report).

(2) The period covered by a report under this subsection ends on June 30.

(c) One semiannual report is due no earlier than January 1 and no later than January 15.

(1) The period covered by a report under this subsection begins on the later of the following dates, as applicable:

(A) July 1;

(B) the day the committee's campaign treasurer appointment was filed, if this is the committee's first report filed under this subchapter (other than a special preelection report or a special session report); or

(C) the first day after the period covered by the last report required by this subchapter (other than a special pre-election report or a special session report).

(2) The period covered by a report under this subsection ends on December 31.

### §20.325. Pre-election Reports.

(a) The campaign treasurer of a specific-purpose committee that supports or opposes a candidate or a measure in an election shall file pre-election reports as provided by subsections (d) and (e) of this section.

(b) For purposes of this section, supporting or opposing a candidate or a measure in an election means accepting political contributions or making political expenditures to support or oppose the candidate or measure.

(c) The campaign treasurer of a specific-purpose committee that has declared an intention to file under the modified schedule in accordance with §20.329 of this title (relating to Modified

Reporting) and that remains eligible to file under the modified schedule is not required to file preelection reports.

(d) A specific-purpose committee that supports or opposes a candidate or measure in an election during the reporting period set out in the next sentence of this subsection must file a report under this subsection. The report required by this subsection covers a period that begins on either the day the committee's campaign treasurer appointment was filed or the first day after the period covered by the last report (other than a special pre-election report or special session report) filed under this subsection must be received by the authority with whom the report is required to be filed no later than the 30th day before the election.

(e) A specific-purpose committee that was required to file a pre-election report under subsection (d) of this section must file a report under this subsection by the eighth day before the election. The report shall cover a period that begins on the 39th day before the election and ends on the 10th day before the election. The report must be received by the authority with whom the report is required to be filed no later than the eighth day before the election.

(f) A committee that was not required to file a report under subsection (d) of this section is required to file a report by the eighth day before the election if the committee supports or opposes a candidate or measure during the period that begins on the 39th day before the election and ends on the 10th day before the election. A report required under this subsection shall cover a period that begins on either the day the committee's campaign treasurer appointment was filed or the first day after the period covered by the last report (other than a special pre-election report or special session report) filed under this subchapter, as applicable, and ends on the 10th day before the election.

# §20.327. Runoff Report.

(a) A specific-purpose committee that supports or opposes a candidate or measure in a runoff election shall file a runoff report, except as provided by subsection (b) of this section.

(b) A specific-purpose committee that has declared an intention to file under the modified schedule in accordance with §20.329 of this title (relating to Modified Reporting) and that remains eligible to file under the modified schedule is not required to file a runoff report.

(c) A runoff report must be received by the authority with whom the report is required to be filed no later than the eighth day before the runoff election.

(d) A runoff report covers the period that begins on the ninth day before the date of the main election and ends on the 10th day before the runoff.

(e) For purposes of this section, supporting or opposing a candidate or a measure in an election means accepting political contributions or making political expenditures to support or oppose the candidate or measure.

# §20.329. Modified Reporting.

(a) A specific-purpose committee that would otherwise be required to file pre-election reports and a runoff report, if necessary, may choose to file under the modified schedule if the committee does not intend to accept more than \$1,010 in political contributions or make more than \$1,010 in political expenditures (excluding filing fees) in connection with any election in an election cycle.

(b) Under the modified schedule, the campaign treasurer of a specific-purpose committee is not required to file pre-election reports or a runoff report.

(c) To select modified filing, a specific-purpose committee must file a declaration of the committee's intent not to accept more than \$1,010 in political contributions or make more than \$1,010 in political expenditures (excluding filing fees) in connection with the election. The declaration must include a statement that the committee understands that if either one of those limits is exceeded, the committee's campaign treasurer will be required to file pre-election reports and, if necessary, a runoff report.

(d) A declaration under subsection (c) of this section is filed with the committee's campaign treasurer appointment.

(e) To file under the modified schedule, a specific-purpose committee must file the declaration required under subsection (c) of this section no later than the 30th day before the first election to which the declaration applies. A declaration filed under subsection (c) of this section is valid for one election cycle only.

(f) Except as provided by subsection (g) of this section, a specific-purpose committee's campaign treasurer must file pre-election reports and, if necessary, a runoff report under the schedule set out in §20.325 of this title (relating to Pre-election Reports) and §20.327 of this title (relating to Runoff Report) if the committee exceeds either of the \$1,010 limits for modified reporting.

(g) If a specific-purpose committee exceeds either of the \$1,010 limits for modified reporting after the 30th day before the election, the committee's campaign treasurer must file a report not later than 48 hours after exceeding the limit.

(1) The period covered by a 48-hour report shall begin either on the day the committee's campaign treasurer appointment was filed (if it is the committee's first report of contributions and expenditures) or on the first day after the period covered by the last report (other than a special pre-election report or special session report) filed under this subchapter, as applicable.

(2) The period covered by a 48-hour report shall continue through the day the committee exceeded one of the limits for modified reporting.

(h) A specific-purpose committee that exceeds either of the \$1,010 limits for modified reporting after the 30th day before the election and on or before the 10th day before the election must file a report under \$20.325(f) of this title, in addition to any required special pre-election reports.

# **§20.331.** Contents of Specific-Purpose Committee Sworn Report of Contributions and Expenditures.

Semiannual reports, pre-election reports, and runoff reports must cover reportable activity during the reporting period and must include the following information:

- (1) the full name of the specific-purpose committee;
- (2) the address of the specific-purpose committee;
- (3) the full name of the specific-purpose committee's campaign treasurer;

(4) the residence or business street address of the specific-purpose committee's campaign treasurer;

- (5) the committee campaign treasurer's telephone number;
- (6) the identity and date of the election for which the report is filed, if applicable;
- (7) for each candidate supported or opposed by the specific-purpose committee:
  - (A) the full name of the candidate;
  - (B) the office sought by the candidate; and
  - (C) an indication of whether the committee supports or opposes the candidate;
- (8) for each officeholder assisted by the specific-purpose committee:
  - (A) the full name of the officeholder;
  - (B) the office held by the officeholder; and
  - (C) an indication of whether the committee supports or opposes the officeholder;
- (9) for each measure supported or opposed by the specific-purpose committee:
  - (A) a description of the measure; and
  - (B) an indication of whether the committee supports or opposes the measure;

(10) for each political expenditure by the committee that was made as a political contribution to a candidate, officeholder, or another political committee and that was returned to the specific-purpose committee during the reporting period:

- (A) the amount returned;
- (B) the full name of the person to whom the expenditure was originally made;
- (C) the address of the person to whom the expenditure was originally made; and

(D) the date the expenditure was returned to the specific-purpose committee;

(11) on a separate page, the following information for each expenditure from political contributions made to a business in which the candidate has a participating interest of more than 10%, holds a position on the governing body of the business, or serves as an officer of the business:

(A) the full name of the business to which the expenditure was made;

- (B) the address of the business to which the expenditure was made;
- (C) the date of the expenditure;
- (D) the purpose of the expenditure; and
- (E) the amount of the expenditure;

(12) if the specific-purpose committee supports or opposes measures exclusively, for each contribution accepted from a labor organization or corporation, as defined by §20.1 of this title (relating to Definitions):

- (A) the date each contribution was accepted;
- (B) the full name of the corporation or labor organization making the contribution;
- (C) the address of the corporation or labor organization making the contribution;
- (D) the amount of the contribution; and
- (E) a description of any in-kind contribution;

(13) for each person from whom the specific-purpose committee accepted a political contribution (other than a pledge, loan, or a guarantee of a loan) of more than \$100 in value or political contributions (other than pledges, loans, or guarantees of loans) that total more than \$100 in value:

- (A) the full name of the person;
- (B) the address of the person;
- (C) the total amount of contributions;
- (D) the date each contribution was accepted; and
- (E) a description of any in-kind contribution;

(14) for each person from whom the specific-purpose committee accepted a pledge or pledges to provide more than \$100 in money or to provide goods or services worth more than \$100:

(A) the full name of the person making a pledge;

(B) the address of the person making a pledge;

(C) the amount of the pledge;

- (D) the date each pledge was accepted; and
- (E) a description of any goods or services pledged;

(15) the total of all pledges accepted during the period for \$100 and less from a person, except those reported under paragraph (14) of this section;

(16) for each person making a loan or loans to the specific-purpose committee for campaign or officeholder purposes, if the total amount loaned by the person during the period is more than \$100:

- (A) the full name of the person or financial institution making the loan;
- (B) the address of the person or financial institution making the loan;
- (C) the amount of the loan;
- (D) the date of the loan;
- (E) the interest rate;
- (F) the maturity date;
- (G) the collateral for the loan, if any; and
- (H) if the loan has guarantors:
  - (i) the full name of each guarantor;
  - (ii) the address of each guarantor;
  - (iii) the principal occupation of each guarantor;
  - (iv) the name of the employer of each guarantor; and
  - (v) the amount guaranteed by each guarantor;

(17) the total amount of loans accepted during the period for \$100 and less from persons other than financial institutions engaged in the business of making loans for more than one year, except those reported under paragraph (16) of this section;

(18) for political expenditures made during the reporting period that total more than \$200 to a single payee:

(A) the full name of the person to whom each expenditure was made;

(B) the address of the person to whom the expenditure was made;

(C) the date of the expenditure;

(D) the purpose of the expenditure; and

(E) the amount of the expenditure;

(19) for each direct campaign expenditure benefiting a candidate or officeholder, except for a direct campaign expenditure made by a committee supporting only one candidate or officeholder:

(A) the name of the candidate or officeholder; and

(B) the office sought or held by the candidate or officeholder;

(20) for each non-political expenditure made from political contributions, other than expenditures reported under paragraph (11) of this section:

(A) the date of each expenditure;

(B) the full name of the person to whom the expenditure was made;

(C) the address of the person to whom the expenditure was made;

(D) the purpose of the expenditure; and

(E) the amount of the expenditure;

(21) for each political contribution accepted from an out-of-state political committee, the information required by §22.7 of this title (relating to Contribution from Out-of-State Committee);

(22) any credit, interest, rebate, refund, reimbursement, or return of a deposit fee resulting from the use of a political contribution or an asset purchased with a political contribution that is received during the reporting period and the amount of which exceeds \$130;

(23) any proceeds of the sale of an asset purchased with a political contribution that is received during the reporting period and the amount of which exceeds \$130;

(24) any other gain from a political contribution that is received during the reporting period and the amount of which exceeds \$130;

(25) any investment purchased with a political contribution that is received during the reporting period and the amount of which exceeds \$130;

(26) the full name and address of each person from whom an amount described by paragraph (22), (23), (24), or (25) of this section is received, the date the amount is received, and the purpose for which the amount is received;

(27) the following total amounts:

(A) the total principal amount of all outstanding loans as of the last day of the reporting period;

(B) the total amount or an itemized listing of political contributions (other than pledges, loans, or guarantees of loans) of \$100 and less;

(C) the total amount of all political contributions (other than pledges, loans, or guarantees of loans);

(D) the total amount or an itemized listing of the political expenditures of \$200 and less; and

(E) the total amount of all political expenditures; and

(28) an affidavit, executed by the campaign treasurer, stating: "I swear, or affirm, that the accompanying report is true and correct and includes all information required to be reported by me under Title 15, Election Code."

# §20.333. Special Pre-Election Report by Certain Specific-Purpose Committees.

(a) As provided by subsection (b) of this section, certain specific-purpose committees must file reports about certain contributions accepted during the period that begins on the ninth day before an election and ends at noon on the day before an election. Reports under this section are known as "special pre-election" reports.

(b) A campaign treasurer for a specific-purpose committee for supporting or opposing a candidate for an office specified by §252.005(1), Election Code, that, during the period described in subsection (a) of this section, accepts one or more political contributions from a person that in the aggregate exceed \$2,020 must file special pre-election reports.

(c) Except as provided in subsection (e) of this section, the campaign treasurer of a specificpurpose committee must file a report so that the report is received by the commission no later than the first business day after the committee accepts a contribution from a person that triggers the requirement to file the special pre-election report.

(d) If, during the reporting period for special pre-election contributions, a committee receives additional contributions from a person whose previous contribution or contributions have triggered the requirement to file a special pre-election report during the period, the campaign treasurer for the committee must file an additional special pre-election report for each such contribution. Except as provided in subsection (e) of this section, each such special pre-election report must be filed so that it is received by the commission no later than the first business day after the committee accepts the contribution.

(e) The campaign treasurer of a specific-purpose committee must file a special pre-election report for each person whose contribution or contributions made during the period for special preelection reports exceed the threshold for special pre-election reports.

(f) A campaign treasurer of a specific-purpose committee must also report contributions reported on a special pre-election report on the next semiannual, pre-election, or runoff report filed, as applicable.

# §20.335. Form and Contents of Special Pre-Election Report by a Specific-Purpose Committee Supporting or Opposing Certain Candidates.

(a) A special pre-election report shall be filed electronically as required by §254.036, Election Code, unless a report is exempt from electronic filing. A special pre-election report that is exempt from electronic filing under §254.036(c), Election Code, is not required to be on a form prescribed by the commission.

(b) A special pre-election report shall include the following information:

- (1) the full name of the specific-purpose committee;
- (2) the full name of the campaign treasurer;

(3) the name of the person making a contribution or contributions that triggered the requirement to file a special pre-election report;

- (4) the address of the person making the contribution or contributions;
- (5) the amount of each contribution;
- (6) the date each contribution was accepted; and
- (7) a description of any in-kind contribution.

## §20.337. Special Session Reports by Specific-Purpose Committees.

(a) A campaign treasurer of a specific-purpose committee for supporting, opposing, or assisting a candidate for or holder of a statewide office or the legislature that accepts a political contribution during the period that begins on the date the governor signs a proclamation calling a special legislative session and ends on the date of final adjournment must file a special session report.

(b) A special session report must be filed with the commission not later than the 30th day after the date of final adjournment of the special session.

(c) A special session report is a report of contributions only, not expenditures. Expenditures made during the period covered by a special session report are required to be reported in the next applicable sworn report of contributions and expenditures.

(d) Contributions reported in a special session report are required to be reported in the next applicable sworn report of contributions and expenditures.

(e) A determination to accept or refuse a political contribution received during the period covered by a special session report shall be made no later than the third day after the date the contribution is received.

(f) A contribution that is refused under subsection (e) of this section must be returned no later than the 30th day after the date of final adjournment. A contribution not returned by that date will be deemed accepted.

(g) A specific-purpose committee's campaign treasurer is not required to file a separate special session report if another report is due no later than the 10th day after the date a report required under this section would be due.

### §20.339. Contents of the Special Session Report.

A report required by §20.337 of this title (relating to Special Session Reports by Specific-Purpose Committees) shall include the following information:

- (1) the specific-purpose committee's full name;
- (2) the specific-purpose committee's address;
- (3) the committee campaign treasurer's full name;
- (4) the campaign treasurer's residence or business street address;
- (5) for each candidate supported or opposed by the specific-purpose committee:
  - (A) the full name of the candidate;
  - (B) the office sought by the candidate; and
  - (C) an indication of whether the committee supports or opposes the candidate;
- (6) for each officeholder supported or opposed by the committee:
  - (A) the full name of the officeholder;
  - (B) the office held by the officeholder; and
  - (C) an indication of whether the committee supports or opposes the officeholder;
- (7) the date each contribution was accepted;
- (8) the full name of each person making a contribution;
- (9) the address of each person making a contribution;
- (10) the amount of each contribution accepted during the period;
- (11) a description of any in-kind contribution; and

(12) an affidavit, executed by the campaign treasurer, stating: "I swear, or affirm, that the accompanying report is true and correct and includes all information required to be reported by me under Title 15, Election Code."

#### §20.341. Dissolution Report.

(a) The campaign treasurer of a specific-purpose committee may file a dissolution report at any time that the committee expects no further reportable activity to occur.

(b) A dissolution report does not have to be filed by a designated deadline.

(c) Filing a dissolution report:

(1) relieves the campaign treasurer of the duty to file additional reports under this subchapter; and

(2) terminates the specific-purpose committee's campaign treasurer appointment.

#### §20.343. Contents of Dissolution Report.

A dissolution report must contain:

(1) the information described in §20.331 of this title (relating to Contents of Specific-Purpose Committee Sworn Report of Contributions and Expenditures); and

(2) the following sworn statement, signed by the specific-purpose committee's campaign treasurer, and properly notarized: "I, the undersigned campaign treasurer, do not expect the occurrence of any further reportable activity by this specific-purpose committee for this or any other campaign or election for which reporting under the Election Code is required. I declare that all of the information required to be reported by me has been reported. I understand that designating a report as a dissolution report terminates the appointment of campaign treasurer. I further understand the circumstances in which the specific-purpose committee may not make or authorize political expenditures or accept political contributions without having an appointment of campaign treasurer on file."

## Subchapter F. REPORTING REQUIREMENT FOR A GENERAL PURPOSE COMMITTEE

## §20.401. Thresholds for Appointment of Campaign Treasurer by a General-Purpose Committee.

(a) A general-purpose committee may not accept political contributions exceeding \$980 and may not make or authorize political expenditures exceeding \$980 without filing a campaign treasurer appointment with the commission.

(b) Unless the committee's campaign treasurer appointment was filed not later than the 30th day before the appropriate election day, a general-purpose committee may not knowingly make or authorize campaign contributions or campaign expenditures exceeding \$980 to support or oppose a candidate in a primary or general election for the following:

- (1) a statewide office;
- (2) a seat in the state legislature;
- (3) a seat on the State Board of Education;

- (4) a multi-county district office; or
- (5) a judicial district office filled by voters of only one county.

## §20.403. Reporting Requirements for Certain General-Purpose Committees.

(a) A general-purpose committee that is established by a political party's county executive committee is subject to Subchapter I of this chapter (relating to Rules Applicable to a Political Party's County Executive Committee). Subchapter I of this chapter prevails over this subchapter in the case of conflict.

(b) A general-purpose committee that is the principal political committee of a political party is subject to Subchapter G of this chapter (relating to Rules Applicable to a Principal Political Committee of a Political Party). Subchapter G of this chapter prevails over this subchapter in the case of conflict.

(c) A general-purpose committee that supports or opposes a candidate for state chair of a political party is subject to Subchapter K of this chapter (relating to Reports by Political Committees Supporting or Opposing a Candidate for State Chair of a Political Party). Subchapter K of this chapter prevails over this subchapter in the case of conflict.

#### §20.405. Campaign Treasurer Appointment for a General-Purpose Political Committee.

(a) A general-purpose committee may appoint a campaign treasurer at any time before exceeding the thresholds described in §20.401(a) of this title (relating to Thresholds for Appointment of Campaign Treasurer by a General-Purpose Committee).

(b) After a general-purpose committee appoints a campaign treasurer, the campaign treasurer must comply with all the requirements of this subchapter, even if the committee has not yet exceeded \$980 in political contributions or expenditures.

(c) With the exception of the campaign treasurer appointment, the individual named as a committee's campaign treasurer is legally responsible for filing all reports of the general-purpose committee, including a report following the termination of his or her appointment as campaign treasurer.

#### §20.407. Appointing an Assistant Campaign Treasurer.

(a) A general-purpose committee may appoint an assistant campaign treasurer by written appointment filed with the commission.

(b) The assistant campaign treasurer has the same authority as the campaign treasurer. However, if the campaign treasurer appointment is terminated, the assistant campaign treasurer no longer has authority to act as the campaign treasurer.

(c) The campaign treasurer, not the assistant campaign treasurer, is liable for any penalties assessed by the commission for late reports or incomplete reports or for failure to file a report.

(d) Section 20.415 of this title (relating to Termination of Campaign Treasurer Appointment) and §20.417 of this title (relating to Termination Report) apply to the appointment and removal of an assistant campaign treasurer.

## §20.409. Name of General-Purpose Committee.

(a) The name of a general-purpose committee must include the full name of each corporation, labor organization, or other association or legal entity other than an individual that directly establishes, administers, or controls the general-purpose committee.

(b) A corporation, labor organization, or other association or legal entity that "directly establishes, administers, or controls" a general-purpose committee is one that has:

(1) the authority to actively participate in determining to whom the general-purpose committee makes political contributions or for what purposes the general-purpose committee makes political expenditures; or

(2) the authority to designate a person to a position of authority with the general-purpose committee, including that of an officer or director of the general-purpose committee.

(c) The name of an entity used in the name of a general-purpose committee may be a commonly recognized acronym by which the entity is known.

(d) The name of a general-purpose committee may not be the same as or deceptively similar to the name of any other general-purpose committee that has an active campaign treasurer appointment on file with the commission. The commission shall determine whether the name of a general-purpose political committee is in violation of this prohibition and shall immediately notify the campaign treasurer of the offending political committee of that determination. In determining whether the name of a general-purpose committee is the same as or deceptively similar to the name of any other general-purpose committee, the commission may be guided by Texas Administrative Code, Title 1, Part 4, Chapter 79. The campaign treasurer of the political committee must file a name change with the commission not later than the 14th day after the date of notification.

(e) For purposes of subsection (d) of this section, a general-purpose committee does not have an active campaign treasurer appointment on file with the commission if the committee files a dissolution report under §254.159, Election Code, or the campaign treasurer appointment for the committee is terminated and more than one year has lapsed since the committee has filed another campaign treasurer appointment with the commission.

#### §20.411. Contents of General-Purpose Committee Campaign Treasurer Appointment.

A campaign treasurer appointment for a general-purpose committee shall include the following information:

(1) the full name of the general-purpose committee, and, if the name is an acronym, the words the acronym represents;

(2) the address of the general-purpose committee;

(3) the full name of the person appointing the campaign treasurer;

(4) the following information for the individual appointed campaign treasurer and, if an assistant campaign treasurer is appointed, for that individual as well:

(A) the individual's full name;

(B) the individual's residence or business street address;

(C) if the individual's mailing address is different from the street address provided, the mailing address for the individual; and

(D) the individual's telephone number;

(5) one of the following:

(A) the full name and any acronym of the name that is used in the name of the general-purpose committee pursuant to \$20.409 of this title (relating to Name of General-Purpose Committee), if applicable; or

(B) the full name of each person who determines to whom the general-purpose committee makes contributions; or

(C) the full name of each person who determines for what purposes the generalpurpose committee makes expenditures;

(6) the name of each other general-purpose committee to which the general-purpose committee intends to make political contributions;

(7) an indication whether the general-purpose committee will file under the regular reporting schedule pursuant to §§20.423, 20.425, and 20.427 of this title (relating to Semiannual Reports; Pre-election Reports; Runoff Report) or under the monthly schedule pursuant to §20.429 of this title (relating to the Option To File Monthly); and

(8) the signature of the individual appointed campaign treasurer.

#### §20.413. Updating Information on the Campaign Treasurer Appointment.

(a) The campaign treasurer must notify the commission in writing of any change in the campaign treasurer's address no later than the 10th day after the date on which the change occurs.

(b) If any of the information required to be included in the general-purpose committee's appointment changes, excluding changes in the campaign treasurer's address, the campaign treasurer shall file a corrected appointment with the commission no later than the 30th day after the date the change occurs.

#### §20.415. Termination of Campaign Treasurer Appointment.

(a) A general-purpose committee may terminate a campaign treasurer appointment at any time by:

- (1) notifying the commission in writing of the termination;
- (2) filing a campaign treasurer appointment for a successor campaign treasurer; or
- (3) filing a dissolution report.

(b) A committee's campaign treasurer may resign by immediately notifying both the appointing authority and the filing authority in writing.

(c) If the campaign treasurer resigns or otherwise leaves the position, the termination is effective on the date the committee actually receives the notice or on the date the filing authority actually receives the notice, whichever is later. Section 20.23 of this title (relating to Timeliness of Action by Mail) does not apply to this subsection.

(d) For purposes of the termination report required by §20.417 of this title (relating to Termination Report), a campaign treasurer's resignation is effective on the date the treasurer resigns, as provided by subsection (b) of this section. Section 20.23 of this title (relating to Timeliness of Action by Mail) applies to this subsection.

(e) A termination of a general-purpose committee's campaign treasurer appointment and the filing of the termination report by themselves do not dissolve the general-purpose committee. A general-purpose committee can be dissolved only by filing a dissolution report with the commission.

#### §20.417. Termination Report.

(a) If the campaign treasurer appointment of a general-purpose committee is terminated, the campaign treasurer whose appointment was terminated shall file a termination report that contains the information listed in §20.433 of this title (relating to Contents of General-Purpose Committee Sworn Report of Contributions and Expenditures), except as provided by §20.434 of this title (relating to Alternate Reporting Requirements for Certain General-Purpose Committees).

(b) A termination report is not required if the termination occurs on the last day of a reporting period under this subchapter and the campaign treasurer files a report for that period as provided by this subchapter.

(c) A termination report covers a period that begins on either the day after the period covered by the last report of contributions and expenditures required to be filed under this subchapter (other than a special pre-election report) or the day the campaign treasurer appointment was filed (if the committee has not yet filed a report of contributions and expenditures). The period covered by the report continues through the day the termination of the campaign treasurer appointment is effective.

(d) The report shall be filed not later than the 10th day after the date the termination of the campaign treasurer appointment is effective.

(e) Activity reported in a termination report is not required to be included in any subsequent report of the general-purpose committee that is filed under this subchapter.

### §20.419. Converting to a Specific-Purpose Committee.

(a) A general-purpose committee that changes its operation and becomes a specific-purpose committee is subject to the requirements applicable to a specific-purpose committee as of the date it files its campaign treasurer appointment as a specific-purpose committee.

(b) The campaign treasurer of a general-purpose committee that changes its operation and becomes a specific-purpose committee shall deliver written notice of the change in status to the commission.

(c) The notice shall identify the filing authority with whom future filings by the committee are expected to be made.

(d) The notice required by this section is due not later than the next deadline for filing a report under this subchapter that:

- (1) occurs after the change in status; and
- (2) would be applicable to the committee if it were still a general-purpose committee.

(e) As provided by §20.301 of this title (relating to Thresholds for Campaign Treasurer Appointment), a new specific-purpose committee involved in an election supporting or opposing a candidate for a statewide office, the state legislature, the State Board of Education, or a multi-county district office in a primary or general election may not accept political contributions exceeding \$500 and may not make or authorize political expenditure exceeding \$500 unless the committee's campaign treasurer appointment as a specific-purpose committee has been on file at least 30 days before the applicable election day.

#### §20.421. Notice to Candidate or Officeholder.

(a) The campaign treasurer of a general-purpose committee that accepts political contributions or makes political expenditures for a candidate or officeholder shall notify the affected candidate or officeholder in accordance with this section.

(b) This section does not apply to a general-purpose committee that has not appointed a campaign treasurer in accordance with §20.405 of this title (relating to Campaign Treasurer Appointment for a General-Purpose Political Committee).

(c) The notice required by this section shall be in writing and shall include:

- (1) the full name of the general-purpose committee;
- (2) the address of the general-purpose committee;
- (3) the full name of the general-purpose committee's campaign treasurer;
- (4) the address of the general-purpose committee's campaign treasurer;
- (5) a statement that the committee is a general-purpose committee; and

(6) a statement that the general-purpose committee has accepted political contributions or has made political expenditures on behalf of the candidate or officeholder.

(d) The notice required by this section shall be delivered no later than the end of reporting period in which the reportable activity occurs.

## §20.423. Semiannual Reports.

(a) Except as provided by subsection (d) of this section, the campaign treasurer of a generalpurpose committee shall file semiannual reports as provided by this section.

(b) One semiannual report is due no earlier than July 1 and no later than July 15.

(1) The report due by July 15 shall cover a period that begins on either January 1, the day the committee's campaign treasurer appointment was filed, or the first day after the period covered by the last report required to be filed under this subchapter (other than a special pre-election report), as applicable.

(2) The period covered by the report due on July 15 ends on June 30.

(c) One semiannual report is due no earlier than January 1 and no later than January 15.

(1) The report due on January 15 shall cover a period that begins on either July 1, the day the committee's campaign treasurer appointment was filed, or the first day after the period covered by the last report required to be filed under this subchapter (other than a special pre-election report), as applicable.

(2) The period covered by the report due on January 15 ends on December 31.

(d) A general-purpose committee that files monthly reports under §20.429 of this title (relating to Option To File Monthly) does not file under this section.

#### §20.425. Pre-election Reports.

(a) A general-purpose committee that accepts political contributions or makes political expenditures in support of or in opposition to a candidate or measure to be voted on in an election shall file pre-election reports as provided by subsections (c) and (d) of this section.

(b) A general-purpose committee that files under §20.429 of this title (relating to Option To File Monthly) does not file under this section.

(c) The first pre-election report must be received by the authority with whom the report is required to be filed no later than the 30th day before the election.

(1) A general-purpose committee that accepts a political contribution or makes a political expenditure to support or oppose a candidate or measure in the election during the period set out in paragraph (2) of this subsection must file a report under this subsection.

(2) The report covers a period that begins on either the day the committee's campaign treasurer appointment was filed or the first day after the period covered by the last report

(other than a special pre-election report) filed under this subchapter, as applicable, and ends on the 40th day before the election.

(d) The second pre-election report must be received by the authority with whom the report is required to be filed no later than the eighth day before the election. The period covered by this report depends on whether the committee was required to file a report under subsection (c) of this section.

(1) A general-purpose committee that was required to file a pre-election report under subsection (c) of this section must file a report under this subsection by the eighth day before the election. The report shall cover a period that begins on the 39th day before the election and ends the 10th day before the election.

(2) A committee that was not required to file a report by the 30th day before the election is required to file a report by the eighth day before the election if the committee accepts a political contribution or makes a political expenditure to support or oppose a candidate or measure during the period that begins on the 39th day before the election and ends on the 10th day before the election.

(A) A report that is required to be filed under paragraph (2) of this subsection shall cover a period that begins on either the day the committee's campaign treasurer appointment was filed or the first day after the period covered by the last report (other than a special pre-election report) filed under this subchapter, as applicable.

(B) The period covered by a report under paragraph (2) of this subsection ends on the 10th day before the election.

## §20.427. Runoff Report.

(a) A general-purpose committee that accepts political contributions or makes political expenditures to support or oppose a candidate or measure in a runoff election shall file a runoff report, except as provided by §20.429 of this title (relating to Option To File Monthly).

(b) A runoff report must be received by the authority with whom the report is required to be filed no later than the eighth day before the runoff election.

(c) A runoff report covers the period that begins on the ninth day before the date of the main election and ends on the 10th day before the runoff.

## §20.429. Option To File Monthly.

(a) As an alternative to filing semiannual, pre-election, and runoff reports, a general-purpose committee may file monthly reports.

(b) A general-purpose committee that files on the monthly filing schedule must file special preelection reports required by §20.435 of this title (relating to Special Pre-Election Reports by Certain General-Purpose Committees).

(c) To be entitled to file monthly reports, the general-purpose committee must deliver written notice of its intent to file monthly to the commission.

(1) A general-purpose committee may file notice of its intent to file monthly at the time the committee files its campaign treasurer appointment.

(2) A general-purpose committee that does not file notice of its intent to file monthly at the time it files its campaign treasurer appointment may file the notice only during the period that begins on January 1 and ends on January 15.

(d) A general-purpose committee that files monthly reports may revert to the regular filing schedule prescribed by §20.423 of this title (relating to Semiannual Reports), §20.425 of this title (relating to Pre-Election Reports), and §20.427 of this title (relating to Runoff Report) by delivering notice to the commission of the general-purpose committee's intent to revert.

(1) The notice must be delivered in writing not earlier than January 1 or later than January 15 of the year for which the general-purpose committee intends to revert to the regular reporting schedule.

(2) The notice must include a report of all political contributions accepted and all political expenditures made that were not previously reported.

## §20.431. Monthly Reporting.

(a) A monthly report filed by a general-purpose committee shall include the information required by \$20.433 of this title (relating to Contents of General-Purpose Committee Sworn Report of Contributions and Expenditures), except that the threshold reporting amount of \$100 set out in \$20.433(11)-(16), and (20) of this title does not apply to a general-purpose committee reporting monthly. For a general-purpose committee reporting monthly, the threshold reporting amount under \$20.433(11)-(16) and (20) of this title is \$20, except as provided by \$20.434 of this title (relating to Alternate Reporting Requirements for Certain General-Purpose Committees).

(b) A monthly report is due not later than the fifth day of the month following the end of the period covered by the report. A monthly report covering the month preceding an election in which the committee is involved must be received by the authority with whom the report is required to be filed no later than the fifth day of the month following the end of the period covered by the report.

(c) Except for the first monthly report filed, a monthly report covers a period that begins on the 26th day of one month and ends on the 25th day of the next month.

(d) The beginning day for the first monthly report filed by a general-purpose committee shall be as follows.

(1) For a general-purpose committee that has been filing on the regular schedule and chooses monthly filing between January 1 and January 15 of a particular year, the first report will cover a period that begins on January 1 of that year.

(2) For a general-purpose committee that elected to file monthly at the time it filed its campaign treasurer appointment, the period covered by the first monthly report depends on the day of the month that the campaign treasurer was appointed.

(A) If the general-purpose committee filed its campaign treasurer appointment before the 25th of the month, the first report will cover a period that begins on the day the appointment was filed and ends on the 25th day of the same month.

(B) If the general-purpose committee filed its campaign treasurer appointment on or after the 25th of the month, the first report will cover the period that begins on the day the appointment is filed and ends on the 25th day of the next month.

## **§20.433.** Contents of General-Purpose Committee Sworn Report of Contributions and Expenditures.

Semiannual reports, pre-election reports, and runoff reports must cover reportable activity during the reporting period and must include the following information:

(1) the full name of the general-purpose committee;

(2) the address of the general-purpose committee;

(3) the full name of the general-purpose committee's campaign treasurer;

(4) the residence or business street address of the general-purpose committee's campaign treasurer;

(5) the committee campaign treasurer's telephone number;

(6) the identity and date of the election for which the report is filed, if applicable;

(7) the full name of each identified candidate or measure or classification by party of candidates supported or opposed by the general-purpose committee and an indication of whether the general-purpose committee supports or opposes each listed candidate, measure, or classification by party of candidates;

(8) the full name of each identified officeholder or classification by party of officeholders assisted by the general-purpose committee;

(9) if the general-purpose committee supports or opposes measures exclusively, for each contribution accepted from a corporation as defined by §20.1 of this title (relating to Definitions):

(A) the date each contribution was accepted;

(B) the full name of the corporation or labor organization making the contribution;

(C) the address of the corporation or labor organization making the contribution;

(D) the amount of the contribution; and

(E) a description of any in-kind contribution;

(10) for each political expenditure by the general-purpose committee that was made as a political contribution to a candidate, officeholder, or another political committee and that was returned to the general-purpose committee during the reporting period:

(A) the amount returned;

(B) the full name of the person to whom the expenditure was originally made;

(C) the address of the person to whom the expenditure was originally made; and

(D) the date the expenditure was returned to the general-purpose committee;

(11) for each person from whom the general-purpose committee accepted a political contribution other than a pledge or a loan of more than \$100 in value, or political contributions other than pledges or loans that total more than \$100 in value (or more than \$20 for a general-purpose committee reporting monthly):

- (A) the date each contribution was accepted;
- (B) the full name of the person making the contribution;

(C) the address of the person making the contribution;

(D) the principal occupation of the person making the contribution;

(E) the amount of the contribution; and

(F) a description of any in-kind contribution;

(12) for each person from whom the general-purpose committee accepted a pledge or pledges to provide more than \$100 in money or to provide goods or services worth more than \$100 (more than \$20 for a general-purpose committee reporting monthly):

(A) the full name of the person making the pledge;

(B) the address of the person making the pledge;

(C) the principal occupation of the person making the pledge;

- (D) the amount of each pledge;
- (E) the date each pledge was accepted; and
- (F) a description of any goods or services pledged;

(13) the total of all pledges accepted during the period for \$100 and less from a person, except for those reported under paragraph (12) of this subsection;

(14) for each person making a loan or loans to the general-purpose committee for campaign purposes, if the total amount loaned by the person during the period is more than \$100 (more than \$20 for a general-purpose committee reporting monthly):

- (A) the full name of the person or financial institution making the loan;
- (B) the address of the person or financial institution making the loan;
- (C) the amount of the loan;
- (D) the date of the loan;
- (E) the interest rate;
- (F) the maturity date;
- (G) the collateral for the loan, if any; and
- (H) if the loan has guarantors:
  - (i) the full name of each guarantor;
  - (ii) the address of each guarantor;
  - (iii) the principal occupation of each guarantor;
  - (iv) the name of the employer of each guarantor; and
  - (v) the amount guaranteed by each guarantor;

(15) the total amount of loans accepted during the period for \$100 and less from persons other than financial institutions engaged in the business of making loans for more than one year, except for those reported under paragraph (14) of this section;

(16) for political expenditures made during the reporting period that total more than \$200 (more than \$20 for a general-purpose committee reporting monthly) to a single payee:

- (A) the full name of the person to whom each expenditure was made;
- (B) the address of the person to whom the expenditure was made;
- (C) the date of the expenditure;
- (D) the purpose of the expenditure;
- (E) the amount of the expenditure; and

(F) indication for an expenditure paid in full or in part from corporations or labor organizations that it was paid from such sources.

- (17) for each non-political expenditure made from political contributions:
  - (A) the date of each expenditure;
  - (B) the full name of the person to whom the expenditure was made;

(C) the address of the person to whom the expenditure was made;

(D) the purpose of the expenditure;

(E) the amount of the expenditure; and

(F) indication for an expenditure paid in full or in part from corporations or labor organizations that it was paid from such sources.

(18) for each candidate or officeholder who benefits from a direct campaign expenditure made by the committee:

(A) the name of the candidate or officeholder; and

(B) the office sought or held by the candidate or officeholder;

(19) for each political contribution from an out-of-state political committee, the information required by §22.7 of this title (relating to Contribution from Out-of-State Committee);

(20) any credit, interest, rebate, refund, reimbursement, or return of a deposit fee resulting from the use of a political contribution or an asset purchased with a political contribution that is received during the reporting period and the amount of which exceeds \$130;

(21) any proceeds of the sale of an asset purchased with a political contribution that is received during the reporting period and the amount of which exceeds \$130;

(22) any other gain from a political contribution that is received during the reporting period and the amount of which exceeds \$130;

(23) any investment purchased with a political contribution that is received during the reporting period and the amount of which exceeds \$130;

(24) the full name and address of each person from whom an amount described by paragraph (20), (21), (22), or (23) of this section is received, the date the amount is received, and the purpose for which the amount is received;

(25) the following total amounts:

(A) the total principal amount of all outstanding loans as of the last day of the reporting period;

(B) the total amount or an itemized listing of political contributions (other than pledges, loans, or guarantees of loans) of \$100 and less (\$20 and less for a general-purpose committee reporting monthly);

(C) the total amount of all political contributions (other than pledges, loans, or guarantees of loans);

(D) the total amount or an itemized listing of the political expenditures of \$200 and less (\$20 and less for a general-purpose committee reporting monthly); and

(E) the total amount of all political expenditures; and

(26) an affidavit, executed by the campaign treasurer, stating: "I swear, or affirm, that the accompanying report is true and correct and includes all information required to be reported by me under Title 15, Election Code."

## §20.434. Alternate Reporting Requirements for General-Purpose Committees.

(a) This section and Election Code §254.1541 apply only to a general-purpose committee with less than \$29,300 in one or more accounts maintained by the committee in which political contributions are deposited, as of the last day of the preceding reporting period for which the committee was required to file a report.

(b) The alternative reporting requirement in Election Code §254.1541 applies only to contributions.

(c) A report by a campaign treasurer of a general-purpose committee to which this section and Election Code §254.1541 apply shall include the information required by §20.433 of this title (relating to Contents of General-Purpose Committee Sworn Report of Contributions and Expenditures) except that the campaign treasurer may choose a threshold reporting amount for political contributions of \$200 instead of the threshold reporting amount of *\$100* set out in §20.433(11) and 25(B) of this title.

(d) A monthly report by a campaign treasurer of a general-purpose committee to which this section and Election Code §254.1541 apply shall include the information required by §20.433 of this title, except that the campaign treasurer may choose a threshold reporting amount for political contributions of \$40 instead of the threshold reporting amount of \$20 set out in §20.433(11) and 25(B) of this title.

#### §20.435. Special Pre-Election Reports by Certain General-Purpose Committees.

(a) In addition to other reports required by this chapter, a general-purpose committee must file a special pre-election report if the committee is involved in an election and if it:

(1) makes direct campaign expenditures supporting or opposing a single candidate that in the aggregate exceed \$2,020 or a group of candidates that in the aggregate exceed \$30,330 during the reporting period for special pre-election reports; or

(2) accepts political contributions from a person that in the aggregate exceed \$6,910 during the reporting period for special pre-election reports.

(b) The period for special pre-election reports begins on the ninth day before election day and ends at noon on the day before election day.

(c) Except as provided by subsection (d) of this section, a report under this section must be received by the commission no later than the first business day after the contribution is accepted or the expenditure is made.

(d) A special pre-election report that is exempt from electronic filing under §254.036(c), Election Code, must be received by the commission no later than 5 p.m. of the first business day after the contribution is accepted or the expenditure is made.

(e) Expenditures and contributions reported under this section must be reported again in the next applicable sworn report of contributions and expenditures.

## §20.437. Form and Contents of Special Pre-Election Report.

(a) A special pre-election report shall be filed electronically as required by §254.036, Election Code, unless a report is exempt from electronic filing. A special pre-election report that is exempt from electronic filing under §254.036(c), Election Code, is not required to be on a form prescribed by the commission.

(b) A report required by §20.435 of this title (relating to Special Pre-Election Reports by Certain General-Purpose Committees) shall include the following information:

(1) the full name of the general-purpose committee;

- (2) the full name of the campaign treasurer;
- (3) the amount of each direct campaign expenditure;

(4) the full name and address of the person or persons to whom each direct campaign expenditure is made;

(5) the date of each direct campaign expenditure;

(6) a description of the goods or services for which each direct campaign expenditure was made;

(7) the identification of the candidates or group of candidates benefiting from the direct campaign expenditure;

(8) the name of the person making a contribution or contributions that triggered the requirement to file a special pre-election report;

(9) the address of the person making the contribution or contributions;

- (10) the amount of each contribution;
- (11) the date each contribution was accepted; and
- (12) a description of any in-kind contribution.

#### §20.439. Dissolution Report.

(a) The campaign treasurer of a general-purpose committee may file a dissolution report at any time that the committee expects no further reportable activity to occur.

(b) A dissolution report does not have to be filed by a designated deadline.

(c) Filing a dissolution report:

(1) relieves the campaign treasurer of the duty to file additional reports under this subchapter; and

(2) terminates the general-purpose committee's campaign treasurer appointment.

#### §20.441. Contents of Dissolution Report.

A dissolution report must contain:

(1) the information listed in §20.433 of this title (relating to Contents of General-Purpose Committee Sworn Report of Contributions and Expenditures), except as provided by §20.434 of this title (relating to Alternate Reporting Requirements for Certain General-Purpose Committees); and

(2) the following sworn statement, signed by the general-purpose committee's campaign treasurer, and properly notarized: "I, the undersigned campaign treasurer, do not expect the occurrence of any further reportable activity by this general-purpose committee for this or any other campaign or election for which reporting under the Election Code is required. I declare that all of the information required to be reported by me has been reported. I understand that designating a report as a dissolution report terminates the appointment of campaign treasurer. I further understand the circumstances in which the general-purpose committee may not make or authorize political expenditures or accept political contributions without having an appointment of campaign treasurer on file."

## Subchapter G. RULES APPLICABLE TO A PRINCIPAL POLITICAL COMMITTEE OF A POLITICAL PARTY

#### §20.501. Designation of Principal Political Committee.

The state or county executive committee of a political party may designate a general-purpose committee as the principal political committee for that state or county, as applicable.

#### §20.503. Exceptions from Certain Notice Requirements.

(a) The principal political committee for a political party in the state or in a county is exempted from complying with \$20.421 of this title (relating to Notice to Candidate or Officeholder).

(b) The principal political committee for a political party in the state or in a county is not required to report under §20.433(16) of this title (relating to Contents of General-Purpose Committee Sworn Report of Contributions and Expenditures) a direct campaign expenditure that it makes on behalf of a slate of two or more nominees of the party.

## Subchapter H. RULES APPLICABLE TO A POLITICAL PARTY ACCEPTING CONTRIBUTIONS FROM CORPORATIONS OR LABOR ORGANIZATIONS

#### §20.521. Restrictions on Use of Contributions from Corporations or Labor Organizations.

A political party that accepts a contribution authorized by §24.19 of this title (relating to Contribution to a Political Party) may use the contribution only for the following purposes:

(1) to defray normal overhead and administrative or operating costs incurred by the party; or

(2) to administer a primary election or convention held by the party.

#### §20.523. Separate Account Required.

(a) Contributions authorized by §24.19 of this title (relating to Contribution to a Political Party) must be maintained in an account separate from other contributions accepted by a political party.

(b) Interest and other income earned from contributions authorized by §24.19 of this title (relating to Contribution to a Political Party) must be maintained in the account required by subsection (a) of this section.

(c) Proceeds from the sale or rent of assets purchased either with contributions authorized by \$24.19 of this title (relating to Contribution to a Political Party) or with interest or other income earned from such contributions must be maintained in the account required by subsection (a) of this section.

#### §20.525. Record of Contributions and Expenditures and Contents of Report.

(a) The party chair of a political party is required to maintain a record of all contributions from corporations and labor organizations and all expenditures from such contributions.

(b) The party chair of a political party shall preserve the record required by subsection (a) of this section for at least two years after the filing deadline for the report containing the information on the record.

(c) The party chair of a political party that accepts contributions authorized by §24.19 of this title (relating to Contribution to a Political Party) shall report all contributions and expenditures made to and from the account required by §20.523 of this title (relating to Separate Account Required), in accordance with the reporting schedule in §20.529 of this title (relating to Reporting Schedule for Political Party Accepting Corporate or Labor Organization Contributions).

(d) The reports required by subsection (c) of this section shall contain the following information for the period covered by the report:

- (1) the full name of the political party;
- (2) the complete mailing address of the political party;
- (3) the full name of the political party's chair;

(4) the residence or business street address of the political party's chair;

(5) if the mailing address of the political party's chair is different from the street address provided, the mailing address for the political party's chair;

(6) the political party chair's telephone number;

(7) the identity and date of the election for which the report is filed, if applicable;

(8) for each corporation or labor organization from whom the political party accepted a contribution (other than a pledge, loan, or guarantee of a loan):

(A) the full name of the corporation or labor organization making the contribution;

(B) the address of the corporation or labor organization making the contribution;

(C) the amount of the contribution; and

(D) the date the contribution was accepted;

(E) a description of any in-kind contribution;

(9) for each corporation or labor organization from whom the political party accepted a pledge:

(A) the full name of the corporation or labor organization making the pledge;

(B) the address of the corporation or labor organization making the pledge;

(C) the amount of the pledge;

(D) the date the pledge was accepted; and

(E) a description of any goods or services pledged;

(10) for each corporation or labor organization making a loan or loans to the political party:

(A) the full name of the person or financial institution making the loan;

(B) the address of the person or financial institution making the loan;

(C) the amount of the loan;

(D) the date of the loan;

(E) the interest rate;

(F) the maturity date;

(G) the collateral for the loan, if any;

(H) if the loan has guarantors:

(i) the full name of each guarantor;

(ii) the address of each guarantor;

(iii) the principal occupation of each guarantor;

(iv) the name of the employer of each guarantor; and

(v) the amount guaranteed by each guarantor;

(11) for each expenditure made by the political party from the account required by \$20.523 of this title (relating to Separate Account Required):

(A) the date of the expenditure;

(B) the full name of the person to whom each expenditure was made;

(C) the address of the person to whom each expenditure was made;

(D) the purpose of the expenditure, for example, the nature of the goods or services for which the expenditure was made; and

(E) the amount of the expenditure;

(12) for each expenditure by the political party that was made as a political contribution to a candidate, officeholder, or another political committee and that was returned to the political party during the reporting period:

(A) the amount returned;

(B) the full name of the person to whom the expenditure was originally made;

(C) the address of the person to whom the expenditure was originally made; and

(D) the date the expenditure was returned to the general-purpose committee;

(13) the following total amounts:

(A) total amount of all contributions (other than pledges, loans, or guarantees of loans) accepted during the period from corporations or labor organizations;

(B) the total amount of all expenditures made during the period from the account required by §20.523 of this title (relating to Separate Account Required);

(14) if applicable, a statement that no reportable activity occurred during the reporting period; and

(15) an affidavit, executed by the political party's chair, stating: "I swear, or affirm, that the accompanying report is true and correct and includes all information required to be reported by me under Title 15, Election Code."

## §20.527. Form of Report.

(a) The report required by this subchapter is separate from any other report a political party is required to file under this title.

(b) The report is filed by the chair of the state party or county executive committee, as applicable, and not by the treasurer of a general-purpose committee. Contributions and expenditures required to be reported under this subchapter should not be included on a report filed in accordance with Subchapter F of this chapter (relating to Reporting Requirements for a General-Purpose Committee).

(c) Except as provided by Section 254.036(c), Election Code, each report filed with the commission under this subchapter and chapter 257 of the Election Code must be filed by computer diskette, modem, or other means of electronic transfer, using computer software provided by the commission or computer software that meets commission specifications for a standard file format.

## **§20.529. Reporting Schedule for Political Party Accepting Corporate or Labor Organization Contributions.**

A political party that has accepted a contribution from a corporation or labor organization shall file the following reports until the political party is no longer accepting corporate or labor organization contributions and the acceptance and expenditure of all such funds has been reported.

(1) A report shall be filed not earlier than July 1 and not later than July 15, covering the period that begins on either January 1 or the day after the last day included in a primary election report filed under paragraph (3) of this section, as applicable, and ends on June 30.

(2) A report shall be filed not earlier than January 1 and not later than January 15, covering the period that begins on either July 1 or the day after the last day included in a general-election report filed under paragraph (4) of this section, as applicable, and ends on December 31.

(3) A report shall be filed for each primary election held by the political party. The report shall be filed not later than the eighth day before the primary election, covering the period that begins on January 1 and ends on the 10th day before the primary election.

(4) A report shall be filed for the general election for state and county officers. The report shall be filed not later than the 50th day before the general election, covering the period that begins on July 1 and ends on the 61st day before the general election for state and county officers.

### §20.531. Restrictions on Contributions before General Election.

A political party may not knowingly accept a contribution authorized by §20.521 of this title (relating to Restrictions on Use of Contributions from Corporations or Labor Organizations) or make an expenditure from a separate account established pursuant to §20.523 of this title (relating to Separate Account Required) during the period that begins on the 60th day before the date of the general election for state and county officers and ends on the day of the election.

## Subchapter I. RULES APPLICABLE TO A POLITICAL PARTY'S COUNTY EXECUTIVE COMMITTEE

## §20.551. Obligation To Maintain Records.

A county executive committee that accepts political contributions or makes political expenditures shall maintain the records required by §20.17 of this title (relating to Retention of Records and Reports).

# **§20.553.** Campaign Treasurer Appointment Not Required for County Executive Committee Accepting Contributions or Making Expenditures Under Certain Amount.

(a) A county executive committee accepting political contributions or making political expenditures totaling \$36,630 or less in a calendar year is not required to:

(1) appoint a campaign treasurer before accepting political contributions or making political expenditures; or

(2) file the reports required by Subchapter F of this chapter (relating to Reporting Requirements for a General-Purpose Committee).

(b) A county executive committee described in subsection (a) of this section is required to comply with \$20.551 of this title (relating to Obligation To Maintain Records).

## §20.555. County Executive Committee Accepting Contributions or Making Expenditures That Exceed Certain Amount.

(a) A county executive committee described by subsection (b) of this section is subject to the requirements of Subchapter F of this chapter (relating to Reporting Requirements for a General-Purpose Committee), except where those rules conflict with this subchapter. In the case of conflict, this subchapter prevails over Subchapter F of this chapter.

(b) A county executive committee that accepts political contributions or that makes political expenditures that, in the aggregate, exceed \$36,630 in a calendar year shall file:

(1) a campaign treasurer appointment with the commission no later than the 15th day after the date that amount is exceeded; and

(2) the reports required by Subchapter F of this chapter (relating to Reporting Requirements for a General-Purpose Committee). The first report filed must include all

political contributions accepted and all political expenditures made before the county executive committee filed its campaign treasurer appointment.

(c) Contributions accepted from corporations and labor organizations under section 253.104 of the Election Code and reported under Subchapter H of this chapter (relating to Accepting and Reporting Contributions from Corporations and Labor Organizations) do not count against the \$36,630 thresholds described in subsection (b) of this section.

(d) A county executive committee that filed a campaign treasurer appointment may file a final report, which will notify the commission that the county executive committee does not intend to file future reports unless it exceeds one of the \$36,630 thresholds. The final report may be filed:

(1) beginning on January 1 and by the January 15 filing deadline if the committee has exceeded one of the \$36,630 thresholds in the previous calendar year; or

(2) at any time if the committee has not exceeded one of the \$36,630 thresholds in the calendar year.

## §20.557. Exceptions from Certain Restrictions.

A county executive committee is excepted from complying with §20.401(a)-(c) of this title (relating to Thresholds for Appointment of Campaign Treasurer by a General-Purpose Committee).

#### §20.559. Exception from Notice Requirement.

A county executive committee that accepts political contributions for or makes political expenditures on behalf of a candidate or officeholder is exempted from complying with §20.421 of this title (relating to Notice to Candidate or Officeholder).

# §20.561. County Executive Committee Accepting Contributions from Corporations and/or Labor Organizations.

(a) A county executive committee that accepts contributions from corporations or labor organizations authorized by §24.19 of this title (relating to Contribution to a Political Party) is subject to the provisions set out in Subchapter H of this chapter (relating to Rules Applicable to a Political Party Accepting Contributions from Corporations or Labor Organizations).

(b) The chair of a county executive committee that accepts contributions from a corporation or labor organization must file the report required by §20.525 of this title (relating to Record of Contributions and Expenditures).

## Subchapter J. REPORTS BY A CANDIDATE FOR STATE OR COUNTY PARTY CHAIR

#### §20.571. Definitions.

The following term, when used in this chapter, shall have the following meaning, unless the context clearly indicates otherwise: Candidate for state chair of a political party--A person who

seeks election to serve as the chair of the state executive committee of a political party with a nominee on the ballot in the most recent gubernatorial general election. Candidacy may be evidenced by any one or more of the following actions:

(1) declaring candidacy;

(2) soliciting or accepting a campaign contribution or making or authorizing a campaign expenditure; or

(3) appointing a campaign treasurer as a candidate for state chair.

#### §20.573. Rules Applicable to Candidate for State Chair of a Political Party.

Except as provided by this subchapter, a candidate for state chair of a political party is subject to the rules applicable to a candidate for a statewide public office.

## **§20.575.** Contributions to and Expenditures by Candidate for State Chair of a Political Party.

Except as provided by this subchapter, each contribution to and expenditure by a candidate for state chair of a political party is subject to the same rules as contributions to and expenditures by a candidate for statewide public office.

#### §20.577. Reporting Schedule for a Candidate for State Chair.

(a) A candidate for state chair of a political party is required to file only the reports listed in this section and is not required to file any other reports required by candidates for public office under Subchapter C of this chapter (relating to Reporting Requirements for a Candidate).

(b) A candidate for state chair of a political party is required to file semiannual reports as provided by this subsection.

(1) One semiannual report is due no earlier than July 1 and no later than July 15.

(A) The period covered by a report under this paragraph begins on the later of the following dates, as applicable:

(i) January 1;

(ii) the first day after the period covered by the last report required by this subchapter; or

(iii) the day the state chair's campaign treasurer appointment was filed, if this is the first report filed under this subchapter.

(B) The period covered by the report under this paragraph ends on June 30.

(2) One semiannual report is due no earlier than January 1 and no later than January 15. The period covered by a report under this paragraph begins on the later of the following dates, as applicable:

## (A) July 1;

(B) the first day after the period covered by the last report required by this subchapter; or

(C) the day the state chair's campaign treasurer appointment was filed, if this is the first report filed under this subchapter.

(c) A candidate for state chair of a political party shall also file the following reports.

(1) A candidate for state chair of a political party shall file a report not earlier than the 39th day before the convening of the state convention and not later than the 30th day before the convening of the state convention. The report shall cover the period that begins on either the day the candidate filed a campaign treasurer appointment with the commission or the first day after the period covered by the last report required to be filed, as applicable, and ends on the 40th day before the convening.

(2) A candidate for state chair of a political party shall file a report not earlier than the ninth day before the convening of the state convention and not later than the eighth day before the convening of the state convention. The report must cover the period that begins on either the day after the candidate filed a campaign treasurer appointment with the commission or the first day after the period covered by the last report required to be filed, as applicable, and ends on the 10th day before the convening.

(d) A candidate for state chair of a political party who expects no further reportable activity in connection with his or her candidacy may file a final report at any time in accordance with §20.229 of this title (relating to Final Report) and §20.231 of this title (relating to Contents of Final Report).

(e) A former candidate for state chair of a political party who retains unexpended political contributions, unexpended interest or other income from political contributions, or assets purchased with political contributions at the time of filing a final report is subject to the requirements of §§20.233, 20.235, 20.237, 20.239, 20.241, and 20.243 of this title (relating to Reporting Requirements for a Candidate).

(f) Except as provided by §254.036(c), Election Code, each report filed with the commission under this section must be filed by computer diskette, modem, or other means of electronic transfer, using computer software provided by the commission or computer software that meets commission specifications for a standard file format.

## §20.579. Candidates for County Chair in Certain Counties.

(a) This section applies to a candidate for election to the office of county chair of a political party with a nominee on the ballot in the most recent gubernatorial general election if the county has a population of 350,000 or more.

(b) The provisions of this subchapter that apply to a candidate for state party chair apply to a candidate for county chair covered by this section, except that a candidate for county chair is not

required to file the pre-convention reports that a state party chair is required to file under section 20.577(c) of this title (relating to Reporting Schedule for a Candidate for State Chair).

(c) In addition to the semiannual reports due to be filed with the commission by January 15 and July 15 under section 20.577(b) of this title, a candidate for county chair covered by this section who has an opponent on the ballot in an election shall file the following two reports with the commission for each primary election except as provided by subsection (d).

(1) The first report shall be filed not later than the 30th day before primary election day. The report covers the period beginning the day the candidate's campaign treasurer appointment is filed or the first day after the period covered by the last report required to be filed under this subchapter, as applicable, and continuing through the 40th day before primary election day.

(2) The second report shall be filed not later than the eighth day before election day. The report covers the period beginning the 39th day before primary election day and continuing through the 10th day before primary election day.

(d) A candidate who has declared the intention to file reports in accordance with section 20.217 of this title (relating to Modified Reporting) and who remains eligible to file under the modified schedule is not required to file pre-election reports.

(e) In addition to other required reports, a candidate for county chair covered by this section who is in a runoff election shall file one report with the commission for the runoff election. The runoff election report shall be filed not later than the eighth day before runoff election day. The report covers the period beginning the ninth day before the primary election day and continuing through the tenth day before runoff election day.

(f) Except as provided by Section 254.036(c), Election Code, each report filed with the commission under this section must be filed by computer diskette, modem, or other means of electronic transfer, using computer software provided by the commission or computer software that meets commission specifications for a standard file format.

## Subchapter K. REPORTS BY POLITICAL COMMITTEES SUPPORTING OR OPPOSING A CANDIDATE FOR STATE OR COUNTY CHAIR OF A POLITICAL PARTY

# §20.591. Appointment of Campaign Treasurer by Political Committee Supporting or Opposing Candidate for State Chair of a Political Party.

Except as provided by this subchapter, a political committee supporting or opposing a candidate for state chair of a political party is subject to the rules applicable to a specific-purpose committee supporting or opposing a candidate for a statewide public office.

# **§20.593.** Contributions and Expenditures by Political Committee Supporting or Opposing Candidate for State Chair of a Political Party.

Except as provided by this subchapter, each contribution to and expenditure by a political committee supporting or opposing a candidate for state chair of a political party is subject to the

same rules as a specific-purpose committee supporting or opposing a candidate for statewide public office.

# **§20.595.** Reporting Schedule for a Political Committee Supporting or Opposing Candidate for State Chair of a Political Party.

(a) A political committee supporting or opposing a candidate for state chair of a political party is required to file semiannual reports in accordance with this section.

(1) One semiannual report is due no earlier than July 1 and no later than July 15.

(A) The period covered by a report under this paragraph begins on the later of the following dates, as applicable:

(i) January 1;

(ii) the first day after the period covered by the last report required by this subchapter; or

(iii) the day the political committee's campaign treasurer appointment was filed, if this is the first report filed under this subchapter.

(B) The period covered by the report under this subparagraph ends on June 30.

(2) One semiannual report is due no earlier than January 1 and no later than January 15.

(A) The period covered by a report under this paragraph begins on the later of the following dates, as applicable:

(i) July 1;

(ii) the first day after the period covered by the last report required by this subchapter; or

(iii) the day the political committee's campaign treasurer appointment was filed, if this is the first report filed under this subchapter.

(B) The period covered by the report under this subparagraph ends on December 31.

(b) A political committee supporting or opposing a candidate for state chair of a political party shall also file the following reports.

(1) A political committee supporting or opposing a candidate for state chair of a political party shall file a report not earlier than the 39th day before the convening of the state convention and not later than the 30th day before the convening of the state convention. The report shall cover the period that begins on either the day the political committee filed a campaign treasurer appointment with the commission or the first day after the period covered by the last report required to be filed, as applicable, and ends on the 40th day before the convening.

(2) A political committee supporting or opposing a candidate for state chair of a political party shall file a report not earlier than the ninth day before the convening of the state convention and not later than the eighth day before the convening of the state convention. The report covers the period that begins on either the date the political committee filed a campaign treasurer appointment with the commission or the first day after the period covered by the last report required to be filed, as applicable, and ends on the 10th day before the convening.

(c) A political committee supporting or opposing a candidate for state chair of a political party may file a dissolution report in accordance with \$20.341 of this title (relating to Dissolution Report) and \$20.343 of this title (relating to Contents of Dissolution Report) at any time that the committee expects no further reportable activity to occur).

# **§20.597.** Political Committees Supporting or Opposing Candidates for County Chair in Certain Counties.

(a) This section applies to a political committee supporting or opposing a candidate for election to the office of county chair of a political party with a nominee on the ballot in the most recent gubernatorial general election if the county has a population of 350,000 or more.

(b) The provisions of this subchapter that apply to a political committee supporting or opposing a candidate for state party chair apply to a political committee covered by this section, except that a political committee covered by this section is not required to file the pre-convention reports under \$20.595(b) of this title (relating to Reporting Schedule for a Political Committee Supporting or Opposing Candidate for State Chair of a Political Party).

(c) In addition to the semiannual reports due to be filed with the commission by January 15 and July 15, a political committee covered by this section shall file the following two reports with the commission for each primary election.

(1) The first report shall be filed not later than the 30th day before primary election day. The report covers the period beginning the day the committee's campaign treasurer appointment is filed or the first day after the period covered by the last report required to be filed under this subchapter, as applicable, and continuing through the 40th day before primary election day.

(2) The second report shall be filed not later than the eighth day before election day. The report covers the period beginning the 39th day before primary election day and continuing through the 10th day before primary election day.

(d) In addition to other required reports, a political committee covered by this section shall file one report with the commission for a runoff election in which the candidate supported or opposed by the committee is involved. The runoff election report shall be filed not later than the eighth day before runoff election day. The report covers the period beginning the ninth day before the primary election day and continuing through the tenth day before runoff election day.

(e) Except as provided by Section 254.036(c), Election Code, each report filed with the commission under this section must be filed by computer diskette, modem, or other means of

electronic transfer, using computer software provided by the commission or computer software that meets commission specifications for a standard file format.

## CHAPTER 22. RESTRICTIONS ON CONTRIBUTIONS AND EXPENDITURES

# **§22.1.** Certain Campaign Treasurer Appointments Required before Political Activity Begins.

(a) An individual must file a campaign treasurer appointment with the proper authority before accepting a campaign contribution or making or authorizing a campaign expenditure.

(1) An officeholder may accept an officeholder contribution and make or authorize an officeholder expenditure without a campaign treasurer appointment on file.

(2) An officeholder who does not have a campaign treasurer appointment on file may not accept a campaign contribution or make or authorize a campaign expenditure.

(b) A political committee may not accept political contributions exceeding \$980 and may not make or authorize political expenditures exceeding \$980 without filing a campaign treasurer appointment with the appropriate filing authority.

(c) Unless the committee's campaign treasurer appointment was filed not later than the 30th day before the appropriate election day, a political committee may not knowingly make or authorize campaign contributions or campaign expenditures exceeding \$980 to support or oppose a candidate in a primary or general election for the following:

- (1) a statewide office;
- (2) a seat in the state legislature;
- (3) a seat on the State Board of Education;
- (4) a multi-county district office; or
- (5) a judicial district office filled by voters of only one county.

(d) This section does not apply to the county executive committee of a political party except as provided in Chapter 20, Subchapter I of this title (relating to Rules Applicable to a Political Party's County Executive Committee).

#### §22.3. Disclosure of True Source of Contribution or Expenditure.

A person may not knowingly make or authorize a political contribution or political expenditure in the name of or on behalf of another unless the person discloses the name and address of the person who is the true source of the contribution.

#### §22.6. Reporting Direct Campaign Expenditures.

(a) Section 254.261 of the Election Code applies to a person who, not acting in concert with another person, makes one or more direct campaign expenditures that exceed \$150 in an election from the person's own property.

#### §22.7. Contribution from Out-of-State Committee.

(a) For each reporting period during which a candidate, officeholder, or political committee accepts a contribution or contributions from an out-of-state political committee totaling more than \$1,010, the candidate, officeholder, or political committee must comply with subsections (b) and (c) of this section.

(b) The candidate, officeholder, or political committee covered by subsection (a) of this section must first obtain from the out-of-state committee one of the following documents before accepting the contribution that causes the total received from the out-of-state committee to exceed \$1,010 during the reporting period:

(1) a written statement, certified by an officer of the out-of-state political committee, listing the full name and address of each person who contributed more than \$200 to the out-of-state political committee during the 12 months immediately preceding the date of the contribution; or

(2) a copy of the out-of-state political committee's statement of organization filed as required by law with the Federal Election Commission and certified by an officer of the out-of-state committee.

(c) The document obtained pursuant to subsection (b) of this section shall be included as part of the report that covers the reporting period in which the candidate, officeholder, or political committee accepted the contribution that caused the total accepted from the out-of-state committee to exceed \$1,010.

(d) A candidate, officeholder, or political committee that:

(1) receives contributions covered by subsection (a) of this section from the same out-ofstate committee in successive reporting periods; and

(2) complies with subsection (b)(2) of this section before accepting the first contribution triggering subsection (a) of this section, may comply with subsection (c) of this section in successive reporting periods by submitting a copy of the certified document obtained before accepting the first contribution triggering subsection (a) of this section, rather than by obtaining and submitting an original certified document for each reporting period, provided the document has not been amended since the last submission.

(e) A candidate, officeholder, or political committee that accepts a contribution or contributions totaling \$1,010 or less from an out-of-state political committee shall include as part of the report covering the reporting period in which the contribution or contributions are accepted either:

(1) a copy of the out-of-state committee's statement of organization filed as required by law with the Federal Election Commission and certified by an officer of the out-of-state committee; or

(2) the following information:

(A) the full name of the committee, and, if the name is an acronym, the words the acronym represents;

- (B) the address of the committee;
- (C) the telephone number of the committee;
- (D) the name of the person appointing the campaign treasurer; and

(E) the following information for the individual appointed campaign treasurer and assistant campaign treasurer:

- (i) the individual's full name;
- (ii) the individual's residence or business street address; and
- (iii) the individual's telephone number.

(f) This section does not apply to a contribution from an out-of-state political committee if the committee filed a campaign treasurer appointment with the commission before making the contribution.

## §22.9. Cash Contributions Exceeding \$100 Prohibited.

(a) A candidate, officeholder, or specific-purpose committee may not knowingly accept political contributions in cash that in the aggregate exceed \$100 from a contributor in a reporting period.

(b) Checks are not considered cash for purposes of this section.

#### §22.11. Prohibition on Contributions during Regular Session.

(a) During the period that begins on the 30th day before the date a regular legislative session convenes and continuing through the 20th day after the date of final adjournment, a person may not knowingly make a political contribution to:

- (1) a statewide officeholder or officer-elect;
- (2) a member of the legislature or member-elect; or

(3) a specific-purpose committee for supporting, opposing, or assisting a statewide officeholder or officer-elect or member or member-elect of the legislature.

(b) An individual or committee described in subsection (a) of this section may not knowingly accept a political contribution, and shall refuse a political contribution that is received, during the period prescribed by subsection (a) of this section.

(c) A political contribution that is received and refused pursuant to this section shall be returned to the contributor not later than the 30th day after the date of receipt.

(d) A contribution made by United States mail or by common or contract carrier is not considered received during the period prescribed by subsection (a) of this section if it was deposited into an official repository of the United States Postal Service or delivered to a common or contract carrier with postage prepaid and properly addressed before the beginning of the period. The date

of the postmark or common or contact carrier documents is considered to be the date the contribution was placed in the mail or delivered to the common or contract carrier unless proven otherwise.

(e) This section does not apply to a political contribution that was made and accepted with the intent that it be used:

(1) in an election held or ordered during the period prescribed by subsection (a) of this section in which the person accepting the contribution is a candidate if the contribution was made after the person appointed a campaign treasurer with the appropriate authority and before the person was sworn in for that office;

(2) to defray expenses incurred in connection with an election contest; or

(3) by a person who holds a statewide office or a member of the legislature, if the person or member was defeated at the general election held immediately before the session is convened, or by a specific-purpose political committee that supports or assists only that person or member.

## §22.13. Contributions in the Capitol Prohibited.

In §253.039 of the Election Code, the term "Capitol" includes the Capitol Building and the Capitol Extension, and any office that is being used as the official capitol office for a member of the legislature, the governor, the lieutenant governor, or the secretary of state.

### §22.17. Prohibition on Personal Use of Political Contributions.

(a) A person who accepts a political contribution as a candidate or officeholder may not convert the contribution to personal use. This subsection applies only to political contributions accepted on or after September 1, 1983.

(b) A specific-purpose committee that accepts a political contribution may not convert the contribution to the personal use of a candidate, officeholder, or former candidate or officeholder. This prohibition applies only to political contributions accepted on or after September 1, 1987.

(c) The prohibitions set out in subsections (a) and (b) of this section apply to the use of an asset purchased with political contributions and to the use of any interest or other income earned on political contributions.

(d) "Personal use" means a use that primarily furthers individual or family purposes not connected with the performance of duties or activities as a candidate for or holder of a public office. It does not include:

(1) payments made to defray ordinary and necessary expenses incurred in connection with activities as a candidate or in connection with the performance of duties or activities as a public officeholder, including payment of rent, utility, and other reasonable housing or household expenses incurred in maintaining a residence in Travis County by members of the legislature who do not normally reside in Travis County, but excluding payments prohibited pursuant to §22.15 of this title (relating to Prohibition on Payments Made to Purchase Real Property);

(2) payments of federal income taxes due on interest and other income earned on political contributions;

(3) use of contributions for defending a criminal action or prosecuting or defending a civil action brought by or against the individual in his or her status as a candidate or officeholder;

(4) use of contributions for participating in an election contest or participating in a civil action to determine an individual's eligibility to be a candidate for, or elected or appointed to, a public office in this state;

(5) an expenditure for a purpose listed in §20.289 of this title (relating to Disposition of Unexpended Contributions);

(6) payment of travel expenses of a candidate's spouse or any other person if the spouse or other person is campaigning for candidate; or

(7) payment of travel expenses of an officeholder's spouse or any other person if the other person's travel is in connection with the performance of duties or activities as a public officeholder.

(e) An asset purchased with political contributions is not converted to personal use if the political contributions are fully reimbursed during the reporting period in which the use occurred in an amount that reasonably reflects the value of the use.

## §22.19. General Restrictions on Reimbursement of Personal Funds.

(a) If a candidate makes political expenditures from the candidate's personal funds, he or she may reimburse those personal funds from political contributions only if the expenditure is reported and the candidate states his or her intent to reimburse personal funds pursuant to §20.219(16) of this title (relating to Content of Candidate's Sworn Report of Contributions and Expenditures).

(b) If an officeholder who does not have a campaign treasurer appointment on file makes political expenditures from the officeholder's personal funds, he or she may reimburse those personal funds from political contributions only if the expenditure is reported and the officeholder states his or her intent to reimburse personal funds pursuant to §20.279(12) of this title (relating to Contents of Officeholder's Sworn Report of Contributions and Expenditures).

(c) A candidate or officeholder may reimburse personal funds from political contributions for the use of personal assets for political purposes provided that the reimbursement is reported as a political expenditure.

(d) A candidate or officeholder who makes political expenditures from his or her personal funds may reimburse those personal funds from political contributions only if:

(1) the expenditures were fully reported as political expenditures on the report covering the period during which the expenditures were made; and

(2) the report disclosing the expenditures indicates that the expenditures were made from the candidate's or officeholder's personal funds and are subject to reimbursement.

(e) A candidate's or officeholder's failure to comply with subsection (d) of this section may not be cured by filing a corrected report after the report deadline has passed.

(f) A candidate or officeholder who has complied with subsection (d) of this section and whose personal funds have been reimbursed from political contributions must report the amount of the reimbursement as a political expenditure in the report covering the period during which the reimbursement was made.

(g) Section 22.21 of this title (relating to Additional Restrictions on Reimbursement of Personal Funds and Payments on Certain Loans) sets limits on the amount of political expenditures from personal funds that a statewide officeholder may reimburse from political contributions.

# §22.21. Additional Restrictions on Reimbursement of Personal Funds and Payments on Certain Loans.

(a) A candidate or officeholder who makes political expenditures from personal funds may not reimburse his or her personal funds from political contributions in amounts that in the aggregate exceed the following amounts for each election in which the person's name appears on the ballot:

- (1) for a statewide office other than governor, \$250,000; or
- (2) for governor, \$500,000.

(b) A candidate or officeholder who accepts one or more political contributions in the form of loans, including an extension of credit or a guarantee of a loan or extension of credit, from one or more persons related to the candidate or officeholder within the second degree by affinity or consanguinity, as defined by the Government Code, Chapter 573, Subchapter B (concerning Relationships by Consanguinity or by Affinity), may not use political contributions to repay the loans in amounts that in the aggregate exceed the amount prescribed by subsection (a) of this section. Interest on loans from the personal funds of any person related to the candidate or officeholder within the second degree by affinity or consanguinity is included in the amount prescribed by subsection (a) of this section.

(c) The total amount of both reimbursements and repayments made by a candidate or officeholder under this section may not exceed the amount prescribed by subsection (a) of this section.

(d) An individual who is both a candidate and an officeholder covered by subsection (a) of this section may reimburse his or her personal funds or repay loans from political contributions only in one capacity.

## §22.23. Restrictions on Certain Payments.

(a) A candidate or officeholder, or a specific-purpose committee for supporting, opposing, or assisting the candidate or officeholder, may not knowingly make or authorize a payment from a political contribution if the payment is made for personal services rendered by the candidate or officeholder or by the spouse or dependent child of the candidate or officeholder to:

(1) a business in which the candidate or officeholder has a participating interest of more than 10%, holds a position on the governing body of the business, or serves as an officer of the business; or

(2) the candidate or officeholder or the spouse or dependent child of the candidate or officeholder.

(b) A payment made from a political contribution to a business described by subsection (a) of this section that is not prohibited by that subsection may not exceed the amount necessary to reimburse the business for actual expenditures made by the business.

(c) A discount given by a corporation to conform with subsection (b) of this section does not constitute a political contribution from the corporation.

## §22.27. Time Limit on Retaining Unexpended Contributions.

A former candidate may retain unexpended political contributions after he or she ceases to be a candidate or officeholder only in accordance with §20.237 of this title (relating to Final Disposition of Unexpended Contributions) or §20.289 of this title (relating to Disposition of Unexpended Contributions), as applicable.

## §22.29. Activity after Death or Incapacity of Candidate or Officeholder.

(a) The legal representative of a candidate or officeholder who has died or become incapacitated may accept political contributions and make or authorize expenditures only for the following purposes:

(1) payment of debts or expenses in connection with a campaign or in connection with officeholder duties and activities;

(2) payments to the political party with which the person was affiliated when the person's name last appeared on a ballot;

(3) political contributions to a candidate or political committee;

(4) donations to the Comptroller of Public Accounts for deposit in the state treasury;

(5) refunds of contributions to one or more persons from whom political contributions were received, not to exceed the total amount contributed by each person within the last two years;

(6) donations to a charity recognized by the Internal Revenue Service as tax-exempt;

(7) donations to a public or private post-secondary educational institution or an institution of higher education as defined by the Education Code, §61.003(8) (concerning Definitions), solely for the purpose of assisting or creating a scholarship program; or

(8) payment of federal income taxes due on interest and other income earned on political contributions.

(b) See §20.67 of this title (relating to Reporting after the Death or Incapacity of a Filer) in regard to reporting requirements after the death or incapacity of a candidate or officeholder.

#### §22.31. Restrictions on Foreign Nationals.

Federal law prohibits contributions from foreign nationals who have not been granted permanent residence in the United States. See United States Code, Title 2, §441(e).

#### §22.35. Corporate Contributions to Certain Political Committees.

(a) A political committee that accepts a monetary political contribution from a corporation or labor organization shall maintain the contribution in a separate account for political contributions from corporations and labor organizations.

(b) A political committee that accepts a political contribution from a corporation or labor organization shall not use the contribution to make a political contribution to:

- (1) a candidate for elective office;
- (2) an officeholder; or
- (3) a political committee other than a hybrid committee, a direct campaign expenditure-only committee, or a political committee that supports or opposes measures exclusively.

#### §22.37. Virtual Currency Contributions.

(a) Virtual currency contributions are considered "in-kind" contributions.

(b) A candidate, officeholder, or political committee must report a gain from the sale of virtual currency contributions on the appropriate schedule if the gain exceeds the reporting threshold set by Section 254.031(9) of the Election Code and amended by section 18.31 of this title (relating to Adjustments to Reporting Thresholds).

(c) The value of a virtual currency contribution shall be reported as the fair market value of the virtual currency upon receipt.

#### CHAPTER 24. RESTRICTIONS ON CONTRIBUTIONS AND EXPENDITURES APPLICABLE TO CORPORATIONS AND LABOR ORGANIZATIONS

#### §24.1. Corporations and Certain Associations Covered.

(a) This chapter applies to:

(1) labor organizations;

(2) corporations that are organized under the Texas Business Corporation Act, the Texas For-Profit Corporation Law, the Texas Non-Profit Corporation Act, the Texas Non-Profit Corporation Law, federal law, or the laws of another state or nation; and

(3) the following associations, whether incorporated or not, for purposes of this chapter are considered to be corporations covered by this chapter:

(A) banks;

(B) trust companies;

- (C) savings and loan associations or companies;
- (D) insurance companies;
- (E) reciprocal or interinsurance exchanges;
- (F) railroad companies;
- (G) cemetery companies;
- (H) government-regulated cooperatives;
- (I) stock companies; and
- (J) abstract and title insurance companies.

(b) For purposes of this chapter, members of a corporation that does not have stockholders and members of an association listed in subsection (a)(3) of this section are considered to be stockholders.

(c) This chapter does not apply to a political committee that incorporates for liability purposes only in accordance with subsection (d) of this section, provided that the sole principal purpose of the committee is accepting political contributions and making political expenditures.

(d) A political committee may incorporate to limit its liability by providing in its official incorporation documents that it is a political committee that is incorporating for liability purposes only, and that its only principal purpose is to accept political contributions and make political expenditures.

#### §24.5. Corporate Loans.

(a) A corporation may not make a loan to a candidate, officeholder, or political committee for campaign or officeholder purposes unless:

(1) the corporation has been legally and continuously engaged in the business of lending money for at least one year before the loan is made; and

(2) the loan is made in the due course of business.

(b) This section does not apply to a loan to a political committee that supports or opposes measures exclusively.

#### §24.15. Payments to a Corporation of the Candidate or Officeholder.

(a) If a corporation charges a candidate, officeholder, or specific-purpose committee for supporting or assisting a candidate or officeholder less than fair market value for goods or services in order to comply with §253.041(b) of the Election Code, the discount is not a prohibited corporate contribution.

(b) If the discount is greater than is necessary to comply with §253.041(b) of the Election Code, the discount is a prohibited corporate contribution if the discount is not otherwise authorized by this chapter.

#### §24.17. Corporate Expenditures for Get-Out-the-Vote Campaigns Permitted.

(a) An expenditure to finance a voter registration or get-out-the-vote drive is not a political expenditure if the drive encourages voting in general but does not encourage voting for or against a measure, candidate, officeholder, or political party.

(b) A corporation or labor organization is permitted to make an expenditure described in subsection (a) of this section.

(c) A corporate or labor organization expenditure described by subsection (a) of this section is not reportable.

#### §24.18. Designation of Contribution for Administrative Purposes.

(a) Any of the following will serve to designate a political expenditure in the form of a political contribution made by a corporation or labor organization as restricted to the establishment, administration, maintenance, or operation of a general-purpose committee:

(1) A contemporaneous written instruction that the contribution is restricted to the administration, maintenance, or operation of the committee accepting the contribution;

(2) The negotiable instrument conveying the contribution contains language indicating that the entity is a corporation, including but not limited to "Inc.," "Incorporated," "Corp.," or "Corporation;"

(3) The general-purpose committee accepting the contribution reports the contribution as monetary contribution or monetary support from a corporation or labor organization on the committee's campaign finance report; or

(4) The general-purpose committee accepting the contribution deposits the contribution into a separate segregated account for political contributions from corporations and labor organizations.

(b) Subsection (a) of this section shall not be read to restrict a hybrid committee, a direct campaign expenditure-only committee, or a political committee that supports or opposes measures exclusively from using a contribution from a corporation or labor organization to make a direct campaign expenditure.

#### §24.19. Affidavit Required by a Political Committee Making a Direct Campaign Expenditure from a Political Contribution Accepted from a Corporation or Labor Organization.

A political committee, including a direct campaign expenditure-only committee, must include in its campaign treasurer appointment the affidavit described by section 252.003(a)(4) (relating to contents of a general-purpose committee's campaign treasurer appointment) or 252.0031(a)(2) (relating to contents of a specific-purpose committee's campaign treasurer appointment) of the Election Code, as applicable, before using a political contribution from a corporation or labor organization to make a direct campaign expenditure in connection with a campaign for an elective office.

#### CHAPTER 26. POLITICAL AND LEGISLATIVE ADVERTISING

#### §26.1. Disclosure Statement.

(a) A disclosure statement that is required by §255.001, Election Code, must contain the words "political advertising" or any recognizable abbreviation, and must:

(1) appear on one line of text or on successive lines of text on the face of the political advertising; or

(2) be clearly spoken in the political advertising if the political advertising does not include written text.

(b) A disclosure statement is not required on political advertising printed on letterhead stationery if the letterhead contains the full name of one of the following:

(1) the person who paid for the political advertising;

(2) the political committee authorizing the political advertising; or

(3) the candidate authorizing the political advertising.

(c) A disclosure statement is not required on:

(1) campaign buttons, pins, or hats, or on objects whose size makes printing the disclosure impractical;

(2) political advertising posted or re-posted on an Internet website, as long as the person posting or re-posting the political advertising:

(i) is not an officeholder, candidate, or political committee; and

(ii) did not make an expenditure exceeding \$100 in a reporting period for political advertising beyond the basic cost of hardware messaging software and bandwidth;

(3) the Internet social media profile webpage of a candidate or officeholder, provided the webpage clearly and conspicuously displays the full name of the candidate or officeholder; or

(4) political advertising posted or re-posted by a person on an Internet website, provided the advertising is posted with a link to a publicly viewable Internet webpage that:

(i) contains the disclosure statement; or

(ii) is exempt from containing the disclosure statement under Subsection (c)(3).

(d) For the purposes of Subsection (c), an "Internet social media profile webpage" is an Internet webpage on a website where members of the public may, for no charge, connect electronically with other members of the public and share text, images, videos, and similar forms of communications.

#### §26.2. Newsletter of Public Officer of a Political Subdivision.

For purposes of §255.003 of the Election Code, a newsletter of a public officer of a political subdivision is not political advertising if:

(1) It includes no more than two pictures of a public officer per page and if the total amount of area covered by the pictures is no more than 20 percent of the page on which the pictures appear;

(2) It includes no more than eight personally phrased references (such as the public officer's name, "I", "me", "the city council member") on a page that is  $8 \frac{1}{2}$ " x 11" or larger, with a reasonable reduction in the number of such personally phrased references in pages smaller than  $8 \frac{1}{2}$ " x 11"; and

(3) When viewed as a whole and in the proper context:

(A) is informational rather than self-promotional;

(B) does not advocate passage or defeat of a measure; and

(C) does not support or oppose a candidate for nomination or election to a public office or office of political party, a political party, or a public officer.

#### §26.3. Legislative Advertising.

Political advertising as defined by the Election Code, Section 251.001(16) (concerning Definitions), does not constitute legislative advertising under the Government Code, Section 305.027 (concerning Required Disclosure on Legislative Advertising).

#### §26.5. Code of Fair Campaign Practices.

A candidate or political committee that has filed a copy of the Code of Fair Campaign Practices as provided by the Election Code, Chapter 258, may indicate that fact on political advertising by including the following or a substantially similar statement: (Name of the candidate or political committee, as appropriate) subscribes to the Code of Fair Campaign Practices.

#### §26.7. Use of the Term "Reelect" in Political Advertising.

A person or candidate may, in the event of redistricting, use the term "reelect" in a campaign for elective office only if the candidate is the elected incumbent of an office that represented any part of the new or renumbered district prior to the redistricting.

#### CHAPTER 28. REPORTS BY A CANDIDATE FOR SPEAKER OF THE HOUSE OF REPRESENTATIVES

#### §28.1. Definitions.

The following words and terms, when used in this Chapter 28, shall have the following meanings, unless the context clearly indicates otherwise:

Campaign funds--For purposes of the Government Code, Chapter 302 (concerning Speaker of the House of Representatives), "campaign funds" as defined in §302.011 (concerning Definitions) shall include "interest earned" and shall include "interest paid."

#### §28.5. Information To Report.

Each report required to be filed with the commission pursuant to the Government Code, §302.013 (concerning Filing of Statement of Contributions, Loans, and Expenditures), shall set forth the total amount of interest earned during the reporting period.

#### §28.7. Permitted Expenditures.

As required by the Government Code, §302.020 (concerning Permitted Expenditures), a speaker candidate shall not expend campaign funds for any purpose other than those permitted by §302.020 (concerning Permitted Expenditures), and then only if those expenditures are directly related to the speaker candidacy; provided, that this section is not intended to prohibit the payment from campaign funds of federal income taxes due on campaign funds.

#### §28.9. Segregation of Campaign Funds.

All contributed campaign funds shall be maintained in accounts separate and apart from any other accounts.

#### **CHAPTER 34. REGULATION OF LOBBYISTS**

#### Subchapter A. GENERAL PROVISIONS

#### §34.1. Definitions.

The following words and terms, when used in this chapter, shall have the following meanings, unless the context clearly indicates otherwise.

(1) Communicates directly with, or any variation of that phrase--In Government Code, Chapter 305, and in this chapter includes communication by facsimile transmission.

(2) Expenditure--In Government Code, Chapter 305, and in this chapter does not include a payment of less than \$200 that is fully reimbursed by the member of the legislative or executive branch who benefits from the expenditure if the member of the legislative or executive branch fully reimburses the person making the payment before the date the person would otherwise be required to report the payment.

(3) Lobby activity--Direct communication with and preparation for direct communication with a member of the legislative or executive branch to influence legislation or administrative action.

(4) Registrant--In Government Code, Chapter 305, and in this chapter means a person who is required to register as well as a person who has registered regardless of whether that person's registration was required.

#### §34.3. Compensation for Preparation Time.

Compensation a person receives for preparing to communicate directly with a member of the legislative or executive branch to influence legislation or administrative action is included in calculating compensation for purposes of the registration and reporting requirements in Government Code, Chapter 305, and this chapter. Examples of preparation for lobby communications include participation in strategy sessions, review and analysis of legislation or administrative matters, research and communication with the employer/client. A person who does not directly communicate with a member of the legislative or executive branch to influence legislation or administrative action is not required to register because of compensation received for preparing to do so.

#### §34.5. Certain Compensation Excluded.

(a) Compensation received for the following activities is not included for purposes of calculating the registration threshold under Government Code §305.003(a)(2) and this chapter:

(1) requesting a written opinion that interprets a law, regulation, rule, policy, practice, or procedure administered by a state office or agency;

(2) preparation or submission of an application or other written document that merely provides information required by law, statute, rule, regulation, order, or subpoena, or that responds to a document prepared by a state agency;

(3) communicating merely for the purpose of demonstrating compliance with an audit, inspection, examination of a financial institution, or government investigation to interpret and determine compliance with existing laws, rules, policies, and procedures;

(4) communicating for the purpose of achieving compliance with existing laws, rules, policies, and procedures, including communications to show qualification for an exception of general applicability that is available under existing laws, rules, policies, and procedures;

(5) providing to a member of the legislative or executive branch information consisting of facts or data that the member requested in writing regarding legislation or administrative action, when the request was not solicited by or on behalf of the person providing the information;

(6) communicating to an agency's legal counsel, an administrative law judge, or a hearings examiner concerning litigation or adjudicative proceedings to which the agency is a party, or concerning adjudicative proceedings of that agency;

(7) providing testimony, making an appearance, or any other type of communication documented as part of a public record in a proceeding of an adjudicative nature of the type authorized by or subject to the Administrative Procedure Act, Government Code, Chapter 2001, whether or not that proceeding is subject to the Open Meetings Law;

(8) providing oral or written comments, making an appearance, or any other type of communication, if documented as part of a public record in an agency's rule-making proceeding under the Administrative Procedure Act, Government Code, Chapter 2001, or in public records kept in connection with a legislative hearing; or

(9) providing only clerical assistance to another in connection with the other person's lobbying (for example, a person who merely types or delivers another person's letter to a member).

(b) Subsection (a) of this section does not apply to a registrant. A registrant's activity described by subsection (a) is subject to disclosure under Chapter 305 of the Government Code and this title.

#### §34.7. Reimbursement for Office Expenses.

Reimbursement received for the following office expenses is not included in calculating reimbursement for purposes of the registration and reporting requirements in Government Code, Chapter 305, and this chapter.

- (1) long distance telephone charges;
- (2) delivery charges;
- (3) photocopy expenses;
- (4) facsimile expenses;

- (5) office supplies;
- (6) postage; and
- (7) dues and subscriptions.

#### §34.9. Taxes and Tips.

Taxes and tips are not included in determining the amount of an expenditure for purposes of Government Code, Chapter 305, and this chapter.

### §34.11. Attribution of Expenditure to More Than One Person; Reimbursement of Lobby Expenditure.

(a) Except as provided by Government Code, §305.0021, a lobby expenditure made on a person's behalf and with the person's consent or ratification is an expenditure by that person for purposes of registration and reporting under Government Code, Chapter 305, and this chapter.

(b) Payment of reimbursement to a registrant is not included for purposes of calculation of the registration threshold under Government Code, §305.003(a)(1), and is not required to be reported if the registrant receiving the reimbursement reports the expenditure on a lobby activity report.

(c) A registrant is not required to report a lobby expenditure attributable to more than one person if another registrant has reported the expenditure.

#### §34.13. Incidental Expenditures for Transportation.

Government Code, §305.024(a)(3), does not prohibit an expenditure for transportation of incidental value such as transportation in the form of a ride of short duration in a personal car or taxi.

#### **§34.14.** Expenditures for Fact-Finding Trips.

(a) For purposes of §305.025(3), Government Code, an expenditure for transportation or lodging provided to a member of the legislative or executive branch is for a fact-finding trip only if:

(1) the expenditure is necessary for the member to obtain information that directly relates to the member's official duties;

(2) the member cannot reasonably obtain the information without the expenditure; and

(3) the expenditure is not for the member's attendance at a merely ceremonial event or pleasure trip.

(b) If an expenditure made for transportation or lodging for a fact-finding trip is required to be disclosed on a lobby activities report by §305.0061(a), Government Code, the purpose of the transportation or lodging must include a description of the information that the expenditure was necessary to obtain under subsection (a) of this section.

#### §34.15. Reporting Subject Matter.

(a) A registrant reporting subject matter under Government Code, 305.005(f)(4), (f)(5)(B), or 305.006(d), of this chapter, shall report subject matter by marking the appropriate subject matter categories.

(b) A registrant reporting the subject matter of communications to influence administrative action shall also report, if known or reasonably available to the registrant, the docket number or other administrative designation of any administrative action that is the subject of the registrant's direct communication with a member of the executive branch, and the name of the agency or department at which the administrative action is pending.

#### §34.17. Satisfaction of Presence Requirement by Entity.

An entity may satisfy the presence requirement in Government Code §305.006(f), and §305.024(a)(7), by the presence of:

(1) an individual registrant who represents the entity; or

(2) a person whose position, authority, or conduct on behalf of the entity could support an award of exemplary damages against the entity.

#### **§34.19.** Courtesy Notices by Electronic Mail.

(a) A person required to register as a lobbyist may provide to the commission an electronic mail address to which courtesy notices regarding filing requirements under Chapter 305 of the Government Code may be sent.

(b) The commission is not obligated to send notices regarding filing requirements to a person required to register as a lobbyist who does not provide to the commission an electronic mail address.

(c) Failure to receive a notice regarding filing requirements does not constitute an excuse for failing to comply with any filing deadline.

#### Subchapter B. REGISTRATION REQUIRED

#### §34.41. Expenditure Threshold.

(a) A person must register under Government Code, §305.003(a)(1), if the person makes total expenditures of more than \$880 in a calendar quarter, not including expenditures for the person's own travel, food, lodging, or membership dues, on activities described in Government Code §305.006(b) to communicate directly with one or more members of the legislative or executive branch to influence legislation or administrative action.

(b) An expenditure made by a member of the judicial, legislative, or executive branch of state government or an officer or employee of a political subdivision of the state acting in his or her official capacity is not included for purposes of determining whether a person is required to register under Government Code, §305.003(a)(1).

(c) An expenditure made in connection with an event to promote the interests of a designated geographic area or political subdivision is not included for purposes of determining whether a person has crossed the registration threshold in Government Code, §305.003(a)(1), if the expenditure is made by a group that exists for the limited purpose of sponsoring the event or by a person acting on behalf of such a group.

#### §34.43. Compensation and Reimbursement Threshold.

(a) A person must register under Government Code, §305.003(a)(2), if the person receives, or is entitled to receive under an agreement under which the person is retained or employed, more than \$1,760 in a calendar quarter in compensation and reimbursement, not including reimbursement for the person's own travel, food, lodging, or membership dues, from one or more other persons to communicate directly with a member of the legislative or executive branch to influence legislation or administrative action.

(b) For purposes of Government Code, §305.003(a)(2), and this chapter, a person is not required to register if the person spends not more than 40 hours for which the person is compensated or reimbursed during a calendar quarter engaging in lobby activity, including preparatory activity as described by §34.3 of this title.

(c) For purposes of Government Code, §305.003(a)(2), and this chapter, a person shall make a reasonable allocation of compensation between compensation for lobby activity and compensation for other activities.

#### §34.45. Entity Registration.

(a) An entity that is required to register under Government Code, §305.003, and this chapter may nonetheless avoid registration if all activity otherwise reportable by the entity is reported by one or more individual registrants in accordance with §34.65 and §34.85 of this title (relating to Compensation Reported by Lobby Firm Employee and Individual Reporting Expenditure by Entity).

(b) An entity that avoids registration under subsection (a) of this section becomes subject to Government Code, §305.024 on the earlier of the date the entity makes the expenditure that would have required the entity to register as a lobbyist or the date the entity receives, or is entitled to receive compensation or reimbursement that would have required the entity to register as a lobbyist.

(c) Registration by an entity does not relieve any individual of the requirement to register if that individual meets one of the registration thresholds in Government Code, §305.003.

#### Subchapter C. COMPLETING THE REGISTRATION FORM

#### §34.63. Assistants.

(a) For purposes of Government Code, 305.005(f)(5), a person "employed or retained by the registrant for the purpose of assisting in direct communication" includes any person who provides administrative or research assistance to the registrant but does not include a person who provides only clerical or secretarial help.

(b) An individual employed by the same employer as the registrant and who assists the registrant at the direction of the registrant is "employed or retained" by the registrant for purposes of Government Code, 305.005(f)(5).

(c) A person listed by a registrant as an assistant under Government Code, 305.005(f)(5), is required to register if the assistant meets one of the registration thresholds under Government Code, 305.003, and this chapter.

#### §34.65. Compensation Reported by Lobby Firm Employee.

(a) An individual registrant employed, reimbursed, or retained by a business entity that receives compensation and/or reimbursement for lobby activity is required to report all compensation and/or reimbursement paid to the entity for lobby activity by that individual.

(b) An individual registrant employed, reimbursed, or retained by a business entity that receives compensation and/or reimbursement for lobby activity may also report compensation and/or reimbursement paid to the entity for lobby activity by one or more other persons if the entity requests that the individual do so in order for the entity to avoid registration.

(c) The individual registrant shall report the compensation by the date on which the entity, if registered, would have been required to report it. The individual registrant shall indicate on a registration or amended registration, as applicable, that he has reported compensation and/or reimbursement paid to an entity for lobby activity by one or more persons other than the registrant.

#### §34.67. Paid, Earned, and Prospective Compensation.

(a) For purposes of Government Code, §305.005, and this chapter, compensation may be reported in any one of the following three ways:

(1) compensation actually paid for lobby activity during the year of registration as of the date the registration form or amended registration form is filed;

(2) compensation earned for lobby activity during the year of registration as of the date the registration form or amended registration form is filed, regardless of whether paid; or

(3) promised compensation for lobby activity during the year of registration, regardless of whether earned or paid on the date the registration form or amended registration form is filed.

(b) A registrant shall indicate on a registration form or amended registration form whether compensation is reported under subsection (a)(1), (2), or (3) of this section.

#### §34.69. Subject Matter.

A registrant shall report the subject matter of lobby activity, in accordance with Government Code, 305.005(f)(4), with respect to each person who reimburses, retains, or employs the registrant to engage in lobby activity.

#### §34.71. Amending a Registration Form.

(a) A change with respect to a docket number or other administrative designation is not required to be reported on an amended registration unless the docket number or other administrative designation is related to a subject matter category not previously reported by the registrant.

(b) Except as necessary to report changed information, a registrant shall not report information about subject matter on an amended registration form that the registrant reported on the registration form or on a previous amended registration form.

(c) A registrant is not required to report on an amended registration form reimbursement received for a lobby expenditure that the registrant will report on a lobby activity report.

#### §34.75. Reporting of Commission or Fee Paid by State Agency.

(a) In addition to the contents required by §305.005 of the Government Code and this chapter, a registration filed by a person who is paid a sales commission or such fee by a state agency must;

- (1) disclose the state agency as a client;
- (2) indicate that the client is a state agency;

(3) provide a description of the subject matter for which the person is paid a sales commission or such fee; and

(4) disclose the amount of the sales commission or such fee.

(b) If the amount of the sales commission or such fee is not known at the time of the reporting, the registration must disclose a reasonable estimate of the maximum amount of the sales commission or such fee and the method under which that amount will be computed.

#### §34.77. Disclosure of Registration under Foreign Agents Registration Act.

The registration of any person who has also filed an active registration statement under the Foreign Agents Registration Act of 1938, as amended (22 U.S.C. §611 et seq.), must include the registration number assigned to the registration statement by the United States Attorney General until the registration statement is terminated.

#### Subchapter D. LOBBY ACTIVITY REPORTS

#### §34.81. Election to File Annually.

A registrant who is eligible to file an annual lobby activity report under Government Code, \$305.0063, may elect to do so at any time during the registration year.

#### §34.82. Modified Reporting Threshold.

For purposes of section 305.0063(d) of the Texas Government Code, expenditures shall include all expenditures reportable under section 305.006, including all expenditures that are required to be reported under subsections 305.006(b) and 305.006(c).

#### §34.83. Time of Expenditure.

For reporting purposes, an expenditure is not made until the amount of the expenditure is readily determinable by the person making the expenditure. If the normal business practice of a vendor or service provider is to make the amount charged known by sending a bill after expenses are incurred, the date of the expenditure, for reporting purposes, is the date the person billed receives the bill.

#### §34.85. Individual Reporting Expenditure by Entity.

(a) An individual registrant may report an expenditure made by a lobby entity if the entity requests that the individual do so in order for the entity to avoid registration; and

(1) the entity makes the expenditure in order for the individual to act on the entity's behalf to communicate directly with a member of the legislative or executive branch to influence legislation or administrative action; or

(2) the entity compensates or reimburses the individual to act on behalf of the entity or on behalf of the entity's clients to communicate directly with a member of the legislative or executive branch to influence legislation or administrative action.

(b) The individual registrant shall report the expenditure by the date on which the entity, if registered, would have been required to report it. The individual registrant shall indicate on a lobby activity report that he or she has reported expenditures made by an entity and indicate the specific amount reported on behalf of the entity.

(c) For purposes of Government Code, §305.0021(b), an expenditure made by an entity under subsection (a) of this section, is not a joint expenditure for purposes of Government Code, §305.0021(b) if the entity makes the entirety of the expenditure at issue.

(d) In this provision "lobby entity" means a corporation, association, firm, partnership, committee, club, organization, or other group of persons voluntarily acting in concert that meets one of the registration thresholds in Government Code, §305.003.

#### Subchapter E. ELECTRONIC FILING

#### **§34.91.** Exemptions from Electronic Filing.

(a) A registrant is required to file each report electronically by using the Internet to transmit the report, by using the web-based filing application provided by the commission, unless the registrant files with the commission an affidavit stating that:

(1) the registrant does not use a computer or mobile device, including a tablet or smartphone with access to the Internet;

(2) no person acting as an agent or consultant of the registrant and no person with whom the registrant contracts uses a computer or mobile device, including a tablet or smartphone with access to the Internet;

(3) the registrant does not intend to be compensated or reimbursed for lobby activity in the calendar year covered by the registration;

(4) the registrant was not compensated for lobby activity in either of the previous two calendar years;

(5) the registrant does not intend to make lobby expenditures during the calendar year covered by the registration; and

(6) the registrant did not make lobby expenditures in either of the previous two calendar years.

(b) The commission has the discretion to exempt from the electronic filing requirement a registrant who is not eligible to file under subsection (a) of this section if a registrant submits an affidavit to the commission stating the basis for the inability to filing electronically.

(c) A registrant who is eligible to file under subsection (a) of this section must file an affidavit under subsection (a) of this section with each report filed under Chapter 305 of the Government Code and this chapter.

(d) A registrant who during a calendar year becomes ineligible to file on paper based on the criteria listed in subsection (a) of this section must file electronically beginning on the date on which the next report is due under \$305.007 of the Government Code.

(e) For purposes of this section, "lobby expenditure" means expenditures required to be reported under Chapter 305 of the Government Code and this chapter.

(f) For purposes of this section, a "report" includes any document required to be filed by a registrant under Chapter 305 of the Government Code and this chapter except that it does not include notices and statements required to be filed under §305.028 of the Government Code.

(g) For purposes of this section, a "report" includes the confidential social security information required to be filed by a lobbyist in compliance with 231.302(c)(1) of the Family Code.

#### CHAPTER 40. FINANCIAL DISCLOSURE FOR PUBLIC OFFICERS

#### §40.1. Financial Statement.

(a) The Texas Ethics Commission adopts by reference the financial statement form prescribed by the commission on January 13, 1992. This form is available from the Texas Ethics Commission, P.O. Box 12070, Austin, Texas 78711-2070.

(b) The form adopted under subsection (a) of this section may be revised by the executive director under §18.1 of this title (relating to Adoption and Revision of Forms), and if so revised shall be deemed to have been adopted by the commission under this section.

#### §40.2. Disclosure of Financial Activity.

For purposes of §572.023 of the Government Code, a filer's personal financial statement must include:

(1) the filer's financial activity in which the filer held an ownership interest, including but not limited to community property; and

(2) the financial activity of the filer's spouse and dependent children if the filer exercised or held the right to exercise any degree of legal or factual control over the activity, notwithstanding a partition agreement.

#### §40.11. Publicly Traded Corporation as Source of Income over \$500.

For purposes of §572.023(b)(4), Government Code, a publicly traded corporation is identified as a source of income by disclosing its full name in addition to the category of the amount of income.

#### CHAPTER 45. CONFLICTS OF INTEREST

#### §45.1. Application.

This chapter applies to §2152.064 and §2155.003 of the Government Code.

#### §45.3. Definitions.

(a) Section 2155.003 of the Government Code applies to:

(1) the chief clerk; and

(2) an employee who exercises discretion in connection with a contract, payment, claim, or other pecuniary transaction under the comptroller's purchasing authority.

(b) Under §2155.003 of the Government Code the following words and terms shall have the following meanings:

(1) "Chief clerk" and "employee" includes the spouse or dependent child of the chief clerk or employee.

(2) "Have an interest in" or "in any manner be connected with," is limited to the purchasing authority that was transferred to the comptroller by §2151.004 of the Government Code, and means a right, share, equitable or legal claim to, or pecuniary interest in, a contract or bid.

(3) "Value," "reward," and "compensation" includes anything with a monetary value of \$5 or more.

(c) Section 2155.003 of the Government Code does not apply to the ownership of stock the value of which does not exceed the lesser of \$25,000 or 5% in any one company, or ownership of shares in a publicly traded mutual fund or similar investment vehicle in which the person does not exercise any discretion regarding the investment of the assets of the fund or other investment vehicle.

#### §45.5. Definitions.

(a) Section 2152.064 of the Government Code applies to:

(1) a commission member and appointee; and

(2) to an employee who exercises discretion in connection with a contract, payment, claim, or other pecuniary transaction under §2152.064 of the Government Code, or in connection with state surplus or salvage property.

(b) Under §2152.064 of the Government Code the following words and terms shall have the following meanings:

(1) "Commission member," "appointee," and "employee" includes the spouse or dependent child of a commission member, appointee, or employee.

(2) "Have an interest in" or "in any manner be connected with," means a right, share, equitable or legal claim to, or pecuniary interest in, a contract or bid, or a recipient of state surplus or salvage property under control of the commission.

(3) "Value," "reward," and "compensation" includes anything with a monetary value of \$5 or more.

(c) Section 2152.064 of the Government Code does not apply to the ownership of stock the value of which does not exceed the lesser of \$25,000 or 5% in any one company, or ownership of shares in a publicly traded mutual fund or similar investment vehicle in which the person does not exercise any discretion regarding the investment of the assets of the fund or other investment vehicle.

#### §45.7. Rebates.

(a) The term "rebate" includes a discount, return, or refund of money.

(b) The chief clerk or an employee of the comptroller is not prohibited from accepting a rebate that is offered or given on the same terms to all state employees or to the general public.

#### §45.9. Rebates.

(a) The term "rebate" includes a discount, return, or refund of money.

(b) An employee, appointee, or commission member of the Texas Facilities Commission is not prohibited from accepting a rebate that is offered or given on the same terms to all state employees or to the general public.

#### **CHAPTER 46. DISCLOSURE OF INTERESTED PARTIES**

#### §46.1. Application.

(a) This chapter applies to §2252.908 of the Government Code.

(b) Section 2252.908 of the Government Code applies only to a contract of a governmental entity or state agency entered into after December 31, 2015, that meets either of the following conditions:

(1) the contract requires an action or vote by the governing body of the entity or agency; or

(2) The value of the contract is at least \$1 million.

(c) A contract does not require an action or vote by the governing body of a governmental entity or state agency if:

(1) The governing body has legal authority to delegate to its staff the authority to execute the contract;

(2) The governing body has delegated to its staff the authority to execute the contract; and

(3) The governing body does not participate in the selection of the business entity with which the contract is entered into.

#### §46.3. Definitions.

(a) "Contract" means a contract between a governmental entity or state agency and a business entity at the time it is voted on by the governing body or at the time it binds the governmental entity or state agency, whichever is earlier, and includes an amended, extended, or renewed contract.

(b) "Business entity" includes an entity through which business is conducted with a governmental entity or state agency, regardless of whether the entity is a for-profit or nonprofit entity. The term does not include a governmental entity or state agency.

(c) "Controlling interest" means:

(1) an ownership interest or participating interest in a business entity by virtue of units, percentage, shares, stock, or otherwise that exceeds 10 percent;

(2) membership on the board of directors or other governing body of a business entity of which the board or other governing body is composed of not more than 10 members; or

(3) service as an officer of a business entity that has four or fewer officers, or service as one of the four officers most highly compensated by a business entity that has more than four officers. This paragraph does not apply to an officer of a publicly held business entity or its wholly owned subsidiaries.

(d) "Interested party" means:

(1) a person who has a controlling interest in a business entity with whom a governmental entity or state agency contracts; or

(2) an intermediary.

(e) "Intermediary," for purposes of this rule, means, a person who actively participates in the facilitation of the contract or negotiating the contract, including a broker, adviser, attorney, or representative of or agent for the business entity who:

(1) receives compensation from the business entity for the person's participation;

(2) communicates directly with the governmental entity or state agency on behalf of the business entity regarding the contract; and

(3) is not an employee of the business entity or of an entity with a controlling interest in the business entity.

(f) "Signed" includes any symbol executed or adopted by a person with present intention to authenticate a writing, including an electronic signature.

(g) "Value" of a contract is based on the amount of consideration received or to be received by the business entity from the governmental entity or state agency under the contract.

#### §46.4. Changes to Contracts.

(a) Section 2252.908 of the Government Code does not apply to a change made to an existing contract, including an amendment, change order, or extension of a contract, except as provided by subsections (b) or (c) of this section.

(b) Section 2252.908 of the Government Code applies to a change made to an existing contract, including an amendment, change order, or extension of a contract, if a disclosure of interested parties form was not filed for the existing contract; and either:

(1) the changed contract requires an action or vote by the governing body of the entity or agency; or

(2) the value of the changed contract is at least \$1 million.

(c) Section 2252.908 of the Government Code applies to a change made to an existing contract, including an amendment, change order, or extension of a contract, if the business entity submitted a disclosure of interested parties form to the governmental entity or state agency that is a party to the existing contract; and either:

(1) there is a change to the disclosure of interested parties; or

(2) the changed contract requires an action or vote by the governing body of the entity or agency; or

(3) the value of the changed contract is at least \$1 million greater than the value of the existing contract.

#### §46.5. Disclosure of Interested Parties Form.

(a) A disclosure of interested parties form required by §2252.908 of the Government Code must be filed on an electronic form prescribed by the commission that contains the following:

(1) The name of the business entity filing the form and the city, state, and country of the business entity's place of business;

(2) The name of the governmental entity or state agency that is a party to the contract for which the form is being filed;

(3) The name of each interested party and the city, state, and country of the place of business of each interested party;

(4) The identification number used by the governmental entity or state agency to track or identify the contract for which the form is being filed and a short description of the services, goods, or other property used by the governmental entity or state agency provided under the contract; and

(5) An indication of whether each interested party has a controlling interest in the business entity, is an intermediary in the contract for which the disclosure is being filed, or both.

(b) The certification of filing and the completed disclosure of interested parties form generated by the commission's electronic filing application must be printed, signed by an authorized agency of the contracting business entity, and submitted to the governmental entity or state agency that is the party to the contract for which the form is being filed.

(c) A governmental entity or state agency that receives a completed disclosure of interested parties form and certification of filing shall notify the commission, in an electronic format prescribed by the commission, of the receipt of those documents not later than the 30<sup>th</sup> day after the date the governmental entity or state agency receives the disclosure.

(d) The commission shall make each disclosure of interested parties form filed with the commission under §2252.908(f) of the Government Code available to the public on the commission's Internet website not later than the seventh business day after the date the commission receives the notice required under subsection (c) of this section.

#### CHAPTER 50. LEGISLATIVE SALARIES AND PER DIEM

#### §50.1. Legislative Per Diem.

(a) The legislative per diem is \$221. The per diem is intended to be paid to each member of the legislature and the lieutenant governor for each day during the regular session and for each day during any special session.

(b) If necessary, this rule shall be applied retroactively to ensure payment of the \$221 per diem for 2019.

# GUIDE TO LOCAL FILING AUTHORITY'S DUTIES UNDER THE CAMPAIGN FINANCE LAW

2



### **TEXAS ETHICS COMMISSION**

### GUIDE TO A LOCAL FILING AUTHORITY'S DUTIES UNDER THE CAMPAIGN FINANCE LAW



This guide is intended for campaign finance filing authorities in cities, school districts, and other political subdivisions other than counties.

Revised January 1, 2023

Texas Ethics Commission, P.O. Box 12070, Austin, Texas 78711www.ethics.state.tx.us(512) 463-5800 • TDD (800) 735-2989Promoting Public Confidence in Government

### GUIDE TO A LOCAL FILING AUTHORITY'S DUTIES UNDER THE CAMPAIGN FINANCE LAW

#### TABLE OF CONTENTS

| GUIDE TO A LOCAL FILING AUTHORITY'S DUTIES UNDER THE CAMPAIGN      |   |
|--|---|
| FINANCE LAW  |   |
| IMPORTANT UPDATES  |   |
| PART I. THE ROLE OF THE FILING AUTHORITY                           | 3 |
| WHAT IS TITLE 15 ABOUT?  |   |
| WHAT ARE MY RESPONSIBILITIES UNDER THE CAMPAIGN FINANCE LAW?       | 4 |
| WHO FILES WITH ME?   |   |
| WHAT FORMS DO I NEED TO MAKE AVAILABLE?                            |   |
| WHY IS IT SO IMPORTANT TO PROVIDE THE INSTRUCTIONS WITH THE FORMS? | 7 |
| IS THERE OTHER INFORMATION I CAN MAKE AVAILABLE TO FILERS?         | 8 |
| HOW DO I KNOW THE FILING DEADLINES?                                | 8 |
| WHY SHOULD I DATE-STAMP FILINGS AND SAVE POSTMARKS?                | 8 |
| MAY FILERS USE COMPUTER PROGRAMS TO CREATE REPORTS?                | 9 |
| PART II. CANDIDATES  | 9 |
| WHAT MAKES SOMEONE A CANDIDATE?                                    |   |
| WHEN DOES A CANDIDATE HAVE TO FILE REPORTS? 1                      |   |
| WHEN MAY A CANDIDATE STOP FILING TITLE 15 REPORTS? 1               | 2 |
| PART III. OFFICEHOLDERS 1  |   |
| WHAT DOES TITLE 15 HAVE TO DO WITH OFFICEHOLDERS? 1                | 2 |
| WHEN DOES AN OFFICEHOLDER HAVE TO FILE REPORTS? 1                  | 2 |
| PART IV. SPECIFIC-PURPOSE POLITICAL COMMITTEES 1                   | 4 |
| WHAT IS A POLITICAL COMMITTEE?1                                    |   |
| WHAT IS A SPECIFIC-PURPOSE POLITICAL COMMITTEE? 1                  | 4 |
| WHAT POLITICAL COMMITTEES FILE WITH POLITICAL SUBDIVISIONS? 1      | 4 |
| WHEN MUST A POLITICAL COMMITTEE FILE A CAMPAIGN TREASURER          |   |
| APPOINTMENT? 1   |   |
| WHAT ARE THE RESPONSIBILITIES OF A COMMITTEE'S CAMPAIGN TREASURER  | ? |
|  |   |
| HOW DOES A COMMITTEE CHANGE A CAMPAIGN TREASURER? 1                | 5 |
| WHAT IF A SPECIFIC-PURPOSE COMMITTEE BECOMES A GENERAL-PURPOSE     |   |
| COMMITTEE? 1   | 5 |
| WHEN DOES THE TREASURER OF A SPECIFIC-PURPOSE COMMITTEE HAVE TO    |   |
| FILE REPORTS? 1  | 6 |
| PART V. PENALTIES 1  |   |
| PART VI. FREQUENTLY ASKED QUESTIONS 1                              |   |
| ENDNOTES1  | 9 |

### GUIDE TO A LOCAL FILING AUTHORITY'S DUTIES UNDER THE CAMPAIGN FINANCE LAW

This guide explains the responsibilities of local filing authorities (other than county filing authorities) under the Texas campaign finance law, which is set out in Title 15 of the Election Code. The Texas Ethics Commission is responsible for interpreting Title 15. Under Title 15, the campaign finance filing authority for a political subdivision other than a county is the clerk or secretary of the political subdivision's governing body. If the political subdivision does not have a clerk or secretary, the filing authority is the presiding officer of the political subdivision's governing body. You may direct questions about Title 15 to the Ethics Commission at (512) 463-5800. You should direct other questions about election law to the Secretary of State at (512) 463-5650 or (800) 252-8683.

Local filing authorities are not expected to be Title 15 experts. The Ethics Commission has prepared two filing guides for local filing authorities to distribute to filers: a CAMPAIGN FINANCE GUIDE FOR LOCAL CANDIDATES AND OFFICEHOLDERS WHO FILE WITH LOCAL FILING AUTHORITIES and a CAMPAIGN FINANCE GUIDE FOR POLITICAL COMMITTEES. For questions not answered in those guides or in the instructions to forms, filers should call the Ethics Commission. (If you are an elected officeholder, you need to understand your obligations as a *filer* in addition to your obligations as a filing authority.)

#### **IMPORTANT UPDATES**

As directed by section 571.064 of the Texas Government Code, the Commission is required to annually adjust certain reporting thresholds upward to the nearest multiple of \$10 in accordance with the percentage increase for the previous year in the Consumer Price Index for Urban Consumers published by the Bureau of Labor Statistics of the United States Department of Labor.

These changes will be made effective January 1<sup>st</sup> of each calendar year; the affected numbers and corresponding new thresholds are located in 1 T.A.C. §18.31, which can be found here: <u>https://www.ethics.state.tx.us/rules/</u>. The higher itemization thresholds will be reflected on the paper forms and in these instructions, as applicable.

Please verify that you are using the correct thresholds and forms that apply to your filing. For example, if you are filing a campaign finance report or lobby activities report that is due in January of 2021, you must use the forms and instructions that are applicable to the period ending December 31, 2020.

### PART I. THE ROLE OF THE FILING AUTHORITY

#### WHAT IS TITLE 15 ABOUT?

The campaign finance law, Title 15 of the Election Code, regulates the acceptance of, expenditure of, and reports regarding money in political campaigns by candidates, officeholders, and political committees. There are civil and criminal penalties for violations of Title 15. Title 15 has nothing to do with ballot access. A violation of Title 15 will not keep a person off the ballot nor will it invalidate an election.

#### WHAT ARE MY RESPONSIBILITIES UNDER THE CAMPAIGN FINANCE LAW?

**Forms:** You may print the various forms listed in this guide and the accompanying instructions from the Ethics Commission's website at *http://www.ethics.state.tx.us* on the Internet.

You are responsible for copying forms and making them available to filers. You may not charge filers for forms.

**Filings:** You are responsible for accepting documents that candidates, officeholders, and political committees file under Title 15. Always remember to date-stamp a filing and to save postmarks and receipt marks on envelopes.

**Code of Fair Campaign Practices:** Under chapter 258 of the Election Code, known as the Fair Campaign Practices Act, you must provide the following documents to each candidate or political committee that files a campaign treasurer appointment with you: (1) a copy of the Fair Campaign Practices Act and (2) a form on which the candidate or treasurer of the committee may subscribe to the Code of Fair Campaign Practices (FORM CFCP). The Ethics Commission makes these documents available. See "Forms" above. A candidate or committee treasurer may complete FORM CFCP and file it with you.

**No Filing Fees:** Tex. Elec. Code § 251.003 states that you may not charge a filing fee for filings required by Title 15.

Notices: You are not required to send filers notices to file.

**Penalties:** You have no responsibility for making sure that candidates, officeholders, or political committees file the required documents; nor do you have authority to penalize a filer for failure to submit a required filing. Filers should be aware, however, that there are penalties for violating Title 15. See "Part V. Penalties" in this guide.

**Public Access:** Forms filed under Title 15 are public records and must be made available for public inspection during regular business hours.

School Districts with a Student Enrollment of More Than 15,000 and Located Wholly or Partly in a Municipality with a Population of More Than 500,000. A school district that meets this enrollment and population criteria is required to post on the school district's Internet website the reports filed by school board trustees, candidates for school board trustee, and specificpurpose committees that support, oppose, or assist a candidate for or member of the board of trustees in a school district. A report must be posted on the school district's website not later than the fifth business day after the date the report is filed. Tex. Elec. Code § 254.04011.

**Retention:** You must keep a campaign treasurer appointment for two years *after the campaign treasurer appointment is terminated*. Also, if a candidate or committee treasurer files a form subscribing to the Code of Fair Campaign Practices, you must keep the form for the same period for which you keep the candidate's or committee treasurer's campaign treasurer appointment. You must keep other Title 15 records for at least two years after filing. If a criminal investigation or proceeding is pending in regard to the election to which Title 15 records pertain, you must keep the records until the investigation or proceeding is over. The Texas State Library and Archives Commission can provide general information about records retention and destruction. You may write the Library and Archives Commission at P.O. Box 12927, Austin, Texas 78711-2927. You may call the commission at (512) 463-5460.

**Questions from Filers:** You are not expected to be a Title 15 expert. Filers will find answers to most of their questions either in the instructions to the forms or in the applicable Ethics Commission campaign finance guide. For further information filers may call the Ethics Commission.

### WHO FILES WITH ME?

The following individuals and committees file with the filing authority for a political subdivision:

- 1. Candidates for and officeholders of elective offices of the political subdivision;
- 2. Specific-purpose committees supporting or opposing candidates for and officeholders of elective offices of the political subdivision; and
- 3. Specific-purpose committees supporting or opposing a measure to be submitted at an election ordered by an authority of the political subdivision. *See* "Part IV. Specific-Purpose Political Committees" in this guide.

**NOTE:** A specific-purpose committee that would be required to file with more than one local filing authority may instead file with the Ethics Commission.

### WHAT FORMS DO I NEED TO MAKE AVAILABLE?

When a candidate first comes in and says they want to run for office with your local filing authority, make only the following forms available (make sure to include the Instruction Guide!):

# 1. FORM CTA *and* FORM CTA Instruction Guide (Appointment of a Campaign Treasurer by a Candidate)

A person who takes action to gain nomination or election to a public office must file FORM CTA even if the person does not intend to accept campaign contributions or make campaign expenditures. Additionally, before a candidate may raise or spend money for his or her candidacy, the candidate must appoint a campaign treasurer by filing FORM CTA with the

appropriate filing authority. For example, a candidate must file a campaign treasurer appointment before paying a filing fee.

## 2. Campaign Finance Guide for Candidates and Officeholders Who File With Local Filing Authorities.

This guide gives general information on what is expected of a candidate or officeholder who files locally.

You can also direct new candidates to the video training presentation on Starting Your Campaign – Candidates found under the Training page of the Ethics Commission's website.

A candidate who brings back a completed CTA form is now considered a candidate for purposes of campaign finance laws. At that time, you should make the following forms available. For most forms, there is a separate instruction guide. <u>Remember</u>: always make the appropriate instruction guide available with a form!

## 1. FORM ACTA and FORM ACTA Instruction Guide (Amendment: Appointment of a Campaign Treasurer by a Candidate)

A candidate uses FORM ACTA to show changes in information on a campaign treasurer appointment.

# 2. FORM C/OH and FORM C/OH Instruction Guide (Candidate/Officeholder Campaign Finance Report)

Candidates and officeholders use FORM C/OH to file periodic reports of contributions and expenditures. A person who is both a candidate and an officeholder reports all activity on the same FORM C/OH.

# **3.** FORM C/OH-FR and FORM C/OH Instruction Guide (C/OH Report: Designation of Final Report)

A candidate uses FORM C/OH-FR to file a final report when he or she does not intend to accept further campaign contributions or make further campaign expenditures. The Ethics Commission makes FORM C/OH-FR available as the last page of FORM C/OH; thus the instructions for FORM C/OH-FR are included in the FORM C/OH Instruction Guide.

# 4. FORM C/OH-UC and FORM C/OH-UC Instruction Guide (Candidate/Officeholder Report of Unexpended Contributions)

Former candidates and officeholders use this form to report the disposition of unexpended contributions.

# 5. FORM STA *and* FORM STA Instruction Guide (Appointment of a Campaign Treasurer by a Specific-Purpose Committee)

A specific-purpose political committee uses FORM STA to appoint a campaign treasurer. Before a political committee may accept more than \$980 in political contributions or spend more than \$980 in political expenditures, the committee must appoint a campaign treasurer.

## 6. FORM ASTA *and* FORM ASTA Instruction Guide (Amendment: Appointment of a Campaign Treasurer by a Specific-Purpose Committee)

A specific-purpose political committee uses this form to show changes in information provided on a campaign treasurer appointment.

## 7. FORM SPAC and FORM SPAC Instruction Guide (Specific-Purpose Committee Campaign Finance Report)

The treasurer of a specific-purpose political committee uses this form to file periodic reports of contributions and expenditures.

# 8. FORM PAC-DR *and* FORM SPAC Instruction Guide (Political Committee Affidavit of Dissolution)

The treasurer of a political committee files FORM PAC-DR, along with a FORM SPAC designated as a dissolution report, to dissolve the committee. The Ethics Commission makes FORM PAC-DR available as the last page of FORM SPAC; thus the instructions for FORM PAC-DR are included in the FORM SPAC Instruction Guide.

#### 9. FORM CFCP and copy of the Fair Campaign Practices Act

You must provide a copy of FORM CFCP and a copy of the Fair Campaign Practices Act (Election Code chapter 258) to each candidate or political committee that files a campaign treasurer appointment with you.

#### **10. FORM PFS (Personal Financial Disclosure Statement)**

Certain local officials are required to file personal financial disclosure statements with local filing authorities. *See* Tex. Local Gov't. Code chs. 145, 159, 335; Tex. Educ. Code § 11.064; Tex. Water Code ch. 60, subch. O.

# WHY IS IT SO IMPORTANT TO PROVIDE THE INSTRUCTIONS WITH THE FORMS?

The forms consist mainly of blank spaces. The instructions explain the reporting requirements in detail.

#### IS THERE OTHER INFORMATION I CAN MAKE AVAILABLE TO FILERS?

The Ethics Commission makes the following guides available for filing authorities to provide to filers.

- 1. Campaign Finance Guide for Candidates and Officeholders Who File With Local Filing Authorities.
- 2. Campaign Finance Guide for Political Committees.
- 3. Campaign Finance Guide for Judicial Candidates and Officeholders.
- 4. Schedules of Filing Dates.
- 5. A Guide to Political Advertising: What You Need To Know.
- 6. A Guide To The Prohibition Against Using Political Subdivision Resources for Political Advertising.

#### HOW DO I KNOW THE FILING DEADLINES?

Title 15 of the Election Code prescribes the filing deadlines for candidates, officeholders, and treasurers of political committees. Filing schedules are also posted on the Ethics Commission's website at *http://www.ethics.state.tx.us* on the Internet. The filing schedules set out the filing deadlines for semiannual reports and pre-election reports for elections held on uniform election dates. Some elections ordered by a political subdivision may be held on dates other than uniform election dates. The Ethics Commission cannot know in advance the dates of all possible elections called by political subdivisions. If an election arises that is not covered on the schedule, please call the Ethics Commission for assistance in calculating the due dates for pre-election reports.

**Deadline on Weekend or Holiday.** If the due date for a report falls on a Saturday, Sunday, or legal holiday, the report is due on the next regular business day.

**5 p.m. Deadline.** The deadline for filing a report is 5 p.m. on the due date.

**Delivery by Mail or Other Carrier.** For most reporting deadlines, a document is filed on time if it is properly addressed with postage or handling charges prepaid and bears a postmark or receipt mark of a common or contract carrier indicating a time by the deadline.

**Pre-Election Reports.** A report due 30 days before an election and a report due 8 days before an election must be *received* by the appropriate filing authority no later than the report due date to be considered filed on time.

#### WHY SHOULD I DATE-STAMP FILINGS AND SAVE POSTMARKS?

Reports are due by certain deadlines, and members of the public or a prosecutor may be interested in knowing whether a particular report was filed on time. A file stamp will show

whether a hand-delivery was on time. For other filings, a postmark or receipt mark will show whether the filings were timely.

#### MAY FILERS USE COMPUTER PROGRAMS TO CREATE REPORTS?

The Ethics Commission makes available computer software for local candidates and committee treasurers to use in drafting reports required under Title 15. Local filers may use the software to draft a report, but they must print it, sign the unsworn declaration or get their signature signed in front of a notary, and then file the report with their local filing authority. Local filers who have questions about the software should call the Ethics Commission directly.

### PART II. CANDIDATES

#### WHAT MAKES SOMEONE A CANDIDATE?

Any action a person takes to seek nomination or election to public office makes the person a candidate for Title 15 filing purposes. A candidate must file a campaign treasurer appointment even if the candidate does not intend to accept campaign contributions or make campaign expenditures. Once a person has filed a campaign treasurer appointment, he or she must file periodic reports of contributions and expenditures as a candidate. The obligation to file reports as a candidate lasts until the person files a final report.

Additionally, a person may not accept a campaign contribution or make a campaign expenditure (even from personal funds) without a campaign treasurer appointment on file.

#### HOW DOES A CANDIDATE APPOINT A CAMPAIGN TREASURER?

To appoint a campaign treasurer, a candidate files FORM CTA with the appropriate filing authority.

**Qualifications of Campaign Treasurer.** A candidate may appoint himself, a relative, a friend, or anyone else as campaign treasurer. A person is ineligible for appointment as a campaign treasurer if the person is the campaign treasurer of a political committee that has outstanding filing obligations.

**Duties of a Candidate's Campaign Treasurer.** A candidate's campaign treasurer has no official duties. The candidate, not the campaign treasurer, is required to sign reports. (The treasurer of a political committee is required to file reports for the committee.)

**Effective Date of Appointment.** If delivered by hand, a campaign treasurer appointment takes effect on the date of delivery. If delivered by mail or common carrier, a campaign treasurer appointment takes effect on the date of the postmark or receipt mark.

**Transferring a Campaign Treasurer Appointment.** A person who has a campaign treasurer appointment on file with you may wish to begin raising or spending money in connection with a campaign for an office that requires filing with a different filing authority. In that case, the candidate transfers his or her campaign treasurer appointment by filing a FORM CTA-T with you,

then filing a file-stamped copy of the CTA-T, a certified copy of their old CTA and a new CTA with the new filing authority. This procedure can affect you in two ways: (1) You may receive a FORM CTA-T with a certified copy of an old campaign treasurer appointment attached, or (2) you may be asked to provide a certified copy of a campaign treasurer appointment on file with you.

**Candidate Who Files a Campaign Treasurer Appointment with a New Filing Authority.** When a candidate files a CTA-T with you to transfer their filing responsibilities to a different filing authority, this ends the person's obligation to file with you *even if he or she continues to hold an office of your political subdivision*.

*Example*: Jane Doe, a city council member with a campaign treasurer appointment on file with the city secretary, decides to run for the state legislature. She files a CTA-T with her city secretary, then files a file-stamped copy of the CTA-T, a certified copy of their old CTA and a new CTA with the Ethics Commission. She is no longer required to file Title 15 reports with the city secretary. Nonetheless, she may wish to file duplicate copies of reports with the city secretary as long as she holds a city office.

**Termination of a Campaign Treasurer Appointment.** It is important for you to know the termination date of a campaign treasurer appointment because you must keep a campaign treasurer appointment for two years *after it is terminated*. A campaign treasurer appointment may be terminated by the candidate or treasurer in three different ways: (1) the candidate files a final report; (2) the candidate appoints a new campaign treasurer; or (3) the campaign treasurer or the candidate notifies you that the appointment is terminated.

An "inactive" campaign treasurer appointment may be terminated by you if the governing body of your political subdivision adopts a process by ordinance or order allowing such termination. A candidate is inactive if the candidate: (1) has never filed or has ceased to file any required campaign finance reports, (2) has not been elected to an office which requires filing a campaign treasurer appointment with you, and (3) has not filed a final report. Before a campaign treasurer appointment may be terminated, the governing body of the political subdivision must consider the proposed termination in a regularly scheduled open meeting.

#### WHEN DOES A CANDIDATE HAVE TO FILE REPORTS?

**Officeholder Who Files a Campaign Treasurer Appointment.** An officeholder who files a campaign treasurer appointment *and who did not already have a campaign treasurer appointment on file* must file a report no later than 15 days after filing the campaign treasurer appointment. This requirement does not apply when an officeholder simply *changes* campaign treasurers. After filing a campaign treasurer appointment and the "15-day" report, the candidate/officeholder files according to the filing schedule for candidates.

**Exception.** The "15-day" reporting requirement does not apply if the candidate/officeholder had no more than \$1,010 in contributions or expenditures during the period covered by the report.

**Semiannual Reports.** A candidate is required to file semiannual reports on January 15 and July 15 of each year on FORM C/OH. A person who is both a candidate and an officeholder (that is,

an officeholder who has a campaign treasurer appointment on file) is not required to file two separate reports. Nor is a person required to distinguish between candidate activity and officeholder activity on the report.

**No Exception.** A person who has a campaign treasurer appointment on file must file semiannual reports, even for reporting periods during which there is no reportable activity and even if the person chose modified reporting.<sup>1</sup>

**Pre-Election Reports.** An opposed candidate in an upcoming election must file pre-election reports. Pre-election reports are due 30 days and 8 days before an election.<sup>2</sup> A pre-election report must be *received* by the appropriate filing authority no later than the report due date.

An "opposed candidate" is a candidate who has an opponent whose name is printed on the ballot. Pre-election reports are not required if a candidate's only opposition is a write-in candidate.<sup>3</sup>

**Modified Reporting.** An opposed candidate who selects "modified reporting" is not required to file pre-election reports (or runoff reports). (**The selection of modified reporting does not affect a candidate's obligation to file semiannual reports.**) A candidate is eligible to select modified reporting if he or she does not intend to exceed \$1,010 in contributions or expenditures in connection with an election. (A candidate has separate \$1,010 thresholds for a primary, a runoff, and a general election.) A candidate selects modified reporting by signing the appropriate blank on FORM CTA or FORM ACTA.

A "modified filer" who exceeds one of the \$1,010 thresholds in connection with an election is subject to the regular filing requirements for opposed candidates. If a modified filer exceeds one of the thresholds before the due date for the "30-day" pre-election report, the filer is not required to give special notice; the filer is simply required to file the pre-election reports by the scheduled due dates. A modified filer who exceeds one of the thresholds after the due date for the "30-day" pre-election report must file a report within 48 hours of exceeding the threshold. If the candidate files the "48-hour" report before the due date for the "8-day" pre-election report, he or she must also file an "8-day" pre-election report by the regular due date for that report.

Many filers at the local level select modified reporting. To understand the rules that apply to modified filing, candidates should consult the Ethics Commission's CAMPAIGN FINANCE GUIDE FOR CANDIDATES AND OFFICEHOLDERS WHO FILE WITH LOCAL FILING AUTHORITIES.

**Runoff Reports.** A candidate in a runoff must file a runoff report. A runoff report is due no later than the 8th day before the runoff election and must be *received* by the appropriate filing authority no later than the report due date.

**Modified Reporting**. A candidate who has selected modified reporting and who remains eligible for modified reporting is not required to file a runoff report. (The selection of modified reporting does not affect a candidate's obligation to file semiannual reports.)

A candidate who has selected modified reporting has \$1,010 thresholds in connection with the main election and new \$1,010 thresholds in connection with a runoff.

**Annual Reports of Unexpended Contributions.** A person who files a final report is no longer a candidate for Title 15 purposes. If the person has surplus funds or assets (and is not an officeholder), he or she must file annual reports of unexpended contributions.

Annual reports are due not earlier than January 1 and not later than January 15 of each year on FORM C/OH-UC. The obligation to file annual reports ends when the former candidate files a report of final disposition of unexpended contributions. Final disposition must be made within six years of filing a final report or leaving office, whichever is later.

**Report of Final Disposition of Unexpended Contributions.** After a former candidate disposes of all surplus funds and assets, he or she must file a report of final disposition of unexpended contributions, also on FORM C/OH-UC. The former candidate may file this report at any time during the year.

#### WHEN MAY A CANDIDATE STOP FILING TITLE 15 REPORTS?

Filing a final report<sup>4</sup> ends a person's obligation to file Title 15 reports unless (1) the filer has not yet filed past-due reports, (2) the person is an officeholder, or (3) the person has surplus funds at the time of filing a final report. An officeholder may be required to file Title 15 reports as an officeholder. See "Part III. Officeholders. When Does An Officeholder Have To File Reports?" in this guide. A former candidate who is not an officeholder but who has surplus campaign funds must file annual reports of contributions and expenditures. *See* "Part II. Candidates. Annual Reports of Unexpended Contributions" above.

### PART III. OFFICEHOLDERS

This section only applies to an officeholder who does not have a campaign treasurer appointment on file. Once an officeholder files a campaign treasurer appointment, he or she becomes a candidate for purposes of the Title 15 filing requirements and must comply with the Title 15 requirements applicable to candidates rather than the requirements applicable to officeholders. Note that only a person who has a campaign treasurer appointment on file may accept a campaign contribution or make a campaign expenditure (even from personal funds).<sup>5</sup>

#### WHAT DOES TITLE 15 HAVE TO DO WITH OFFICEHOLDERS?

Although Title 15 is commonly referred to as the *campaign* finance law, it also requires officeholders who are not candidates (in other words, who do not have a campaign treasurer appointment on file) to file reports of officeholder contributions and expenditures.

#### WHEN DOES AN OFFICEHOLDER HAVE TO FILE REPORTS?

**Semiannual Reports.** An officeholder is required to file semiannual reports of officeholder contributions and expenditures on January 15 and July 15 of each year. For this report officeholders use FORM C/OH.

**Exception for Certain Local Officeholders.** There is an exception to the requirement to file semiannual reports for a local officeholder (who does not have a campaign treasurer appointment on file) who did not exceed \$1,010 in either contributions or expenditures during the reporting period. An officeholder who has a campaign treasurer appointment on file must file semiannual reports even if there is no reportable activity during the reporting period.

**Report Required after Appointment of Campaign Treasurer.** An officeholder who files an appointment of campaign treasurer (and who did not already have a campaign treasurer appointment on file) must file a report of contributions and expenditures no later than 15 days after filing the appointment, using FORM C/OH. After filing the campaign treasurer appointment and the "15-day" report, the officeholder is subject to the filing requirements applicable to candidates.

**Exception.** The "15-day" reporting requirement does not apply if the candidate/officeholder had no more than \$1,010 in contributions or expenditures during the period covered by the report.

Annual Report of Unexpended Contributions. A former officeholder who did not have a campaign treasurer appointment on file at the time of leaving office may be required to file annual reports of unexpended contributions. This requirement applies if the former officeholder retained surplus political funds or assets at the time of filing his or her last required report of contributions and expenditures. The former officeholder may not retain surplus political funds or assets for more than 6 years after leaving office.

Annual reports of unexpended contributions are due not earlier than January 1 and not later than January 15 of each year. The reports, on FORM C/OH-UC, cover the preceding calendar year. A report is due regardless of whether there is any reportable activity. The obligation to file annual reports ends when the former officeholder files a report of final disposition of unexpended contributions.

**Report of Final Disposition of Unexpended Contributions.** Once a former officeholder disposes of surplus funds and assets, he or she must file a report of final disposition of unexpended contributions. A former officeholder may file this report at any time during the year. For this report former officeholders use FORM C/OH-UC.

**Officeholder Who Files a Campaign Treasurer Appointment with a Different Filing Authority.** An officeholder (who does not have a campaign treasurer appointment on file) may file a campaign treasurer appointment with a different filing authority. This ends the officeholder's obligation to file with you *even if he or she continues to hold an office of the political subdivision.* 

## PART IV. SPECIFIC-PURPOSE POLITICAL COMMITTEES

#### WHAT IS A POLITICAL COMMITTEE?

A political committee, commonly referred to as a "PAC," is any group that accepts political contributions or makes political expenditures. Although the term "PAC" may suggest a powerful "special interest" group, a political committee may also be a small group such as two people who get together to raise funds for an old friend who is a candidate for school board.

#### WHAT IS A SPECIFIC-PURPOSE POLITICAL COMMITTEE?

There are two main types of political committees: general-purpose political committees and specific-purpose political committees. In essence, a general-purpose political committee exists to support or promote a particular political point-of-view or the interests of a certain group, whereas a specific-purpose committee exists to support or oppose specific candidates, officeholders, or ballot measures. It is the filer's responsibility, not your responsibility, to determine whether a committee is a general-purpose committee or a specific-purpose committee. The Ethics Commission's Campaign Finance Guide for Political Committees explains the differences in detail.

#### WHAT POLITICAL COMMITTEES FILE WITH POLITICAL SUBDIVISIONS?

A specific-purpose committee files with the clerk, secretary, or presiding officer of a political subdivision other than a county if the committee supports or opposes either individual candidates or officeholders who file with the political subdivision or ballot measures on elections called by the political subdivision.<sup>6</sup> All general-purpose political committees file with the Ethics Commission. It is the filer's responsibility to determine where a political committee files campaign finance reports. The Ethics Commission's Campaign Finance Guide for Political Committees will help filers make this determination.

**Note for School Districts:** A specific-purpose committee created to support or oppose a measure on the issuance of bonds by a school district must file all of its *campaign finance reports* with the Ethics Commission. This does not affect the filing requirements for the committee's campaign treasurer appointment.

# WHEN MUST A POLITICAL COMMITTEE FILE A CAMPAIGN TREASURER APPOINTMENT?

**\$980 Thresholds.** A specific-purpose committee must file a campaign treasurer appointment, on FORM STA, before it exceeds \$980 in either political contributions or political expenditures. Once the committee has filed a campaign treasurer appointment, the treasurer must file periodic reports of contributions and expenditures.

**Effective Date.** If delivered by hand, a committee's campaign treasurer appointment takes effect on the day of delivery. If delivered by mail or common carrier, the appointment takes effect on the date of the postmark or receipt mark.

**Termination of a Committee's Campaign Treasurer Appointment.** It is important for you to know the termination date of a committee's campaign treasurer appointment because you must keep a campaign treasurer appointment for two years *after it is terminated*. A committee's campaign treasurer appointment may be terminated by the committee in three different ways: (1) the committee files a dissolution report; (2) the committee appoints a new campaign treasurer; or (3) you receive notification from the committee or the campaign treasurer that the appointment is terminated.

An "inactive" campaign treasurer appointment may be terminated by you if the governing body of your political subdivision adopts a process by ordinance or order allowing such termination. A political committee is inactive if the committee: (1) has never filed or has ceased to file any required campaign finance reports, and (2) has not filed a dissolution report. Before a campaign treasurer appointment may be terminated, the governing body of the political subdivision must consider the proposed termination in a regularly scheduled open meeting.

# WHAT ARE THE RESPONSIBILITIES OF A COMMITTEE'S CAMPAIGN TREASURER?

The treasurer of a political committee is responsible for filing the committee's reports of contributions and expenditures. (In contrast, a candidate or officeholder, not his or her campaign treasurer, is responsible for filing candidate and officeholder reports of contributions and expenditures.)

#### HOW DOES A COMMITTEE CHANGE A CAMPAIGN TREASURER?

A specific-purpose committee changes treasurers by filing an amended appointment of campaign treasurer on FORM ASTA. The new appointment terminates the old appointment. The outgoing treasurer is required to file a termination report on FORM SPAC not later than 10 days after the termination. (A separate termination report is not required if the termination occurs on the last day of a reporting period and the proper report for that period is filed.)

# WHAT IF A SPECIFIC-PURPOSE COMMITTEE BECOMES A GENERAL-PURPOSE COMMITTEE?

A change in political activity may mean that a political committee that has been filing with a local filing authority has become a general-purpose committee. This change will require the committee to file a new campaign treasurer appointment with the Ethics Commission. In addition to filing a new campaign treasurer appointment with the Ethics Commission, the committee is required to give notice to the local filing authority of the change in status. The committee should review the Ethics Commission's CAMPAIGN FINANCE GUIDE FOR POLITICAL COMMITTEES to become familiar with the contribution and expenditure limits that arise in connection with the transition.

# WHEN DOES THE TREASURER OF A SPECIFIC-PURPOSE COMMITTEE HAVE TO FILE REPORTS?

**Semiannual Reports.** The treasurer of a specific-purpose committee is required to file semiannual reports by January 15 and July 15 of each year. The treasurer must file semiannual reports even if there is no reportable activity.

**Pre-Election Reports.** A specific-purpose committee supporting or opposing an opposed candidate in an upcoming election must file pre-election reports. A specific-purpose committee supporting or opposing a ballot measure must also file pre-election reports. Filers use FORM SPAC for pre-election reports, which are due 30 days and 8 days before an election.<sup>7</sup> (If you are the filing authority for a school district, see the <u>Note under "What Political Committees File with Political Subdivisions?"</u> in this guide.) A pre-election report must be *received* by the appropriate filing authority no later than the report due date.

For purposes of filing pre-election reports, supporting or opposing a candidate or measure means accepting political contributions or making political expenditures to support or oppose the candidate or measure.

**Modified Reporting.** The treasurer of a specific-purpose committee that selects "modified reporting" is not required to file pre-election reports (or runoff reports). (**The selection of modified reporting does not affect the treasurer's obligation to file semiannual reports.)** A committee may select modified reporting if the committee does not intend to exceed \$1,010 in contributions or expenditures in connection with an election. (A committee has separate \$1,010 thresholds for a primary, a runoff, and a general election.) A committee selects modified reporting by signing the appropriate blank on FORM STA or FORM ASTA.

A committee that has selected modified reporting must file pre-election reports if the committee exceeds one of the \$1,010 thresholds. If the committee exceeds one of the thresholds before the due date for a "30-day" pre-election report, the committee is not required to give special notice of that fact; the treasurer is simply required to file the pre-election reports by the scheduled due dates. If the committee exceeds one of the thresholds after the due date for the "30-day" pre-election report, the treasurer must file a report within 48 hours of exceeding the threshold (on FORM SPAC) and then file any pre-election or runoff reports that come due.

**Runoff Reports.** A specific-purpose committee that supports or opposes a candidate in a runoff election must file a runoff report on FORM SPAC. A runoff report is due no later than the 8th day before the runoff and must be *received* by the appropriate filing authority no later than the report due date.

**Modified Reporting.** A specific-purpose committee that has selected modified reporting and remains eligible for modified reporting is not required to file a runoff report.

A committee that has selected modified reporting has \$1,010 thresholds in connection with the main election and additional \$1,010 thresholds in connection with a runoff.

**Termination Report.** After the treasurer of a political committee resigns or is replaced, the outgoing treasurer is required to file a termination report on FORM SPAC not later than 10 days after the termination. (A separate termination report is not required if the termination occurs on the last day of a reporting period and the proper report for that period is filed.)

**Dissolution Report.** A political committee that expects to receive no further political contributions or make no further political expenditures may file a dissolution report on FORM SPAC with FORM PAC-DR attached.

The dissolution report terminates the committee's campaign treasurer appointment and relieves the campaign treasurer of the duty to file additional reports. (In this case, the dissolution report serves as the treasurer's termination report.)

# PART V. PENALTIES

As a local filing authority, you have no authority to penalize filers in any way for violations of Title 15. Any individual may file a criminal complaint regarding a violation of Title 15 with the appropriate county or district attorney. Also, anyone who lives in Texas or owns property in Texas may file a sworn complaint with the Ethics Commission alleging a violation of Title 15.

# PART VI. FREQUENTLY ASKED QUESTIONS

If you have questions about your responsibilities as a filing authority, call the Ethics Commission. The following list contains questions that local filing authorities frequently ask.

# Q. What Title 15 documents should I give to a person who says he is interested in running for an office of my political subdivision?

A. In addition to information you provide about getting on the ballot, you should give the person a copy of the Ethics Commission's CAMPAIGN FINANCE GUIDE FOR CANDIDATES AND OFFICEHOLDERS WHO FILE WITH LOCAL FILING AUTHORITIES, FORM CTA and the FORM CTA Instruction Guide.

#### Q. What should I give to a person who files a campaign treasurer appointment?

A. The person, who is now a candidate for purposes of Title 15, will need FORM C/OH and the FORM C/OH Instruction Guide and a Filing Schedule. You are required to give the person a copy of the 1997 Fair Campaign Practices Act, Chapter 258 of the Election Code, and a copy of FORM CFCP. You should also make sure that the person has a copy of the Ethics Commission's CAMPAIGN FINANCE GUIDE FOR CANDIDATES AND OFFICEHOLDERS WHO FILE WITH LOCAL FILING AUTHORITIES.

#### Q. What should I give someone who is interested in forming a political committee?

A. A group that intends to accept political contributions or make political expenditures should get a copy of the Ethics Commission's CAMPAIGN FINANCE GUIDE FOR POLITICAL

COMMITTEES, FORM STA and the FORM STA Instruction Guide. The group should read the filing guide to determine whether it is a specific-purpose or general-purpose committee.

# Q. What should I give to someone who files a campaign treasurer appointment for a political committee?

A. The treasurer of the political committee will need FORM SPAC and the FORM SPAC Instruction Guide. You are required to give the committee a copy of the 1997 Fair Campaign Practices Act, Chapter 258 of the Election Code, and a copy of FORM CFCP. The treasurer should also have a copy of the Ethics Commission's CAMPAIGN FINANCE GUIDE FOR POLITICAL COMMITTEES.

#### Q. What should I tell a write-in candidate who asks about Title 15?

A. A person who declares himself or herself to be a write-in candidate must file a campaign treasurer appointment and reports of contributions and expenditures.

# Q. What do I tell a candidate who says he isn't going to file a campaign treasurer appointment because he is not going to accept campaign contributions?

- A. All candidates must file a campaign treasurer appointment even if a candidate does not intend to accept campaign contributions or make campaign expenditures. Also, the fact that a candidate doesn't accept campaign contributions does not mean that the candidate will have nothing to report. The candidate must report campaign expenditures from personal funds. A candidate who does not plan to *spend* over \$1,010 or *accept* more than \$1,010 in *total contributions* in connection with an election should take the following steps:
  - 1. The candidate should complete FORM CTA and sign the back for modified filing.
  - 2. If a July 15 or January 15 deadline occurs before the election, the candidate must file a report of contributions and expenditures on FORM C/OH.
  - 3. Whenever the candidate reaches a point where they do not intend to accept any further campaign contributions or make any further campaign expenditures (including payment of campaign debts), the candidate can file a report of contributions and expenditures on FORM C/OH and mark it as a Final Report in Box 9, page 1. The candidate should also complete FORM C/OH-FR and submit it along with FORM C/OH. The final report terminates the candidate's appointment of campaign treasurer and ends the person's obligation to file as a candidate. (If the person won the election, he or she will be subject to the filing requirements applicable to officeholders.)

# B. May a candidate refuse to list the office or seat sought on a campaign treasurer appointment?

A. Yes. A person may decide to start raising money to run for office before the person decides which office to run for.

#### Q. What do I do if someone files a report after the filing deadline?

A. You should accept the filing.

#### Q. What do I do if someone files a corrected or amended filing?

A. You should accept the filing. A filer may also wish to file a Correction/Amendment Affidavit for Candidate/Officeholder with the amendment or correction. Filers should call the Ethics Commission if they have questions about corrected reports.

#### Q. What should I tell a candidate who asks about disclosures on political advertising?

A. Give the filer a copy of the Ethics Commission's GUIDE TO POLITICAL ADVERTISING: WHAT YOU NEED TO KNOW, and A GUIDE TO THE PROHIBITION AGAINST USING POLITICAL SUBDIVISION RESOURCES FOR POLITICAL ADVERTISING. You can also direct them to the video training presentation on political advertising found under the Training page of the Ethics Commission's website.

#### Q. May I enter into a contract to perform the Title 15 functions of another entity?

A. No. An elections services contract may not change the authority with whom Title 15 documents are filed.

# **Q.** If a candidate forms a specific-purpose committee, must the candidate continue to file reports on FORM C/OH?

A. Yes. The candidate and the committee are subject to separate reporting requirements.

## **ENDNOTES**

- 2. An opposed candidate in a runoff election is only required to file a pre-election report 8 days before a *runoff* election; there is no "30-day" pre-election report due before a runoff.
- 3. A write-in candidate must file a campaign treasurer appointment before accepting campaign contributions or making campaign expenditures. Furthermore, a person who declares himself or herself to be a write-in candidate *is* required to file pre-election reports as long as the write-in candidate has an opponent whose name appears on the ballot.
- 4. A person terminates a campaign treasurer appointment by filing a final report on FORM C/OH with FORM C/OH-FR attached.
- 5. Once an officeholder files a campaign treasurer appointment, he or she may use contributions

<sup>1.</sup> See also Part III: Officeholders (Certain officeholders who do not have campaign treasurer appointment on file are excepted from filing semiannual reports).

received as an officeholder to make campaign expenditures.

- 6. A specific-purpose political committee that supports candidates, officeholders, or measures at the political subdivision level may file with the Ethics Commission if it also supports candidates, officeholders, or measures in a jurisdiction other than the political subdivision. For example, a specific-purpose committee that supports a particular candidate for county commissioner and a particular candidate for the city council of a city within the county has two choices of where to file campaign finance reports: (1) The committee may file both with the county election official and with the appropriate city filing authority, or (2) the committee may file with the Ethics Commission only.
- 7. A political committee is only required to file a report 8 days before a *runoff* election; there is no "30-day" pre-election report required before a runoff.

# CAMPAIGN FINANCE GUIDE FOR CANDIDATES AND OFFICEHOLDERS WHO FILE WITH LOCAL FILING AUTHORITIES

3



# **TEXAS ETHICS COMMISSION**

# CAMPAIGN FINANCE GUIDE FOR CANDIDATES AND OFFICEHOLDERS WHO FILE WITH LOCAL FILING AUTHORITIES



This guide is for candidates for and officeholders in the following positions:

- county offices;
- precinct offices;
- single-county district offices;
- city offices; and
- offices of other political subdivisions such as school districts

This guide applies to candidates for and officeholders of justice of the peace. This guide does not apply to candidates for and judges of statutory county courts, statutory probate courts, or district courts. For those candidates and officeholders, the Ethics Commission makes available a CAMPAIGN FINANCE GUIDE FOR JUDICIAL CANDIDATES AND OFFICEHOLDERS.

The Ethics Commission also makes available a CAMPAIGN FINANCE GUIDE FOR CANDIDATES AND OFFICEHOLDERS WHO FILE WITH THE ETHICS COMMISSION, a CAMPAIGN FINANCE GUIDE FOR POLITICAL COMMITTEES, and a CAMPAIGN FINANCE GUIDE FOR POLITICAL PARTIES.

#### **Revised January 1, 2023**

Texas Ethics Commission, P.O. Box 12070, Austin, Texas 78711
<u>www.ethics.state.tx.us</u>
(512) 463-5800 • TDD (800) 735-2989
Promoting Public Confidence in Government

## CAMPAIGN FINANCE GUIDE FOR CANDIDATES AND OFFICEHOLDERS WHO FILE WITH LOCAL FILING AUTHORITIES

#### **TABLE OF CONTENTS**

| INTRODUCTION   |     |
|--|-----|
| IMPORTANT UPDATES  | . 1 |
| OFFICEHOLDERS  | . 1 |
| JUDICIAL CANDIDATES AND OFFICEHOLDERS                        | . 1 |
| FEDERAL OFFICES  | . 2 |
| FILING AUTHORITIES   | . 2 |
| POLITICAL COMMITTEES (PACS)                                  | . 3 |
| FINANCIAL DISCLOSURE STATEMENTS                              |     |
| FEDERAL INCOME TAX   | . 3 |
| TEXAS ETHICS COMMISSION                                      | . 3 |
| APPOINTING A CAMPAIGN TREASURER                              | .3  |
| NO CAMPAIGN CONTRIBUTIONS OR EXPENDITURES WITHOUT TREASURER  |     |
| APPOINTMENT ON FILE  | . 4 |
| APPOINTING TREASURER TRIGGERS REPORTING DUTIES               | . 4 |
| QUALIFICATIONS OF CAMPAIGN TREASURER                         | . 4 |
| DUTIES OF CAMPAIGN TREASURER                                 | . 5 |
| EFFECTIVE DATE OF APPOINTMENT                                | 5   |
| CODE OF FAIR CAMPAIGN PRACTICES                              | . 5 |
| APPOINTMENT BY OFFICEHOLDER                                  | . 5 |
| FILING FOR A PLACE ON THE BALLOT                             | . 5 |
| CHANGING TREASURERS  | . 5 |
| TRANSFERRING TO A DIFFERENT FILING AUTHORITY                 | 5   |
| TERMINATING A CAMPAIGN TREASURER APPOINTMENT                 | 6   |
| DECIDING NOT TO RUN  | 6   |
| THINGS TO REMEMBER   | 6   |
| POLITICAL CONTRIBUTIONS AND EXPENDITURES                     | 7   |
| CAMPAIGN CONTRIBUTIONS                                       |     |
| CAMPAIGN EXPENDITURES  |     |
| OFFICEHOLDER CONTRIBUTIONS                                   | 8   |
| OFFICEHOLDER EXPENDITURES                                    | 8   |
| CAMPAIGN EXPENDITURES BY OFFICEHOLDER                        |     |
| PERMISSIBLE USE OF POLITICAL CONTRIBUTIONS                   |     |
| USE OF POLITICAL FUNDS TO RENT OR PURCHASE REAL PROPERTY     | 8   |
| ACCEPTING CONTRIBUTIONS                                      |     |
| REIMBURSEMENT FOR POLITICAL EXPENDITURES FROM PERSONAL FUNDS |     |
| SEPARATE ACCOUNT REQUIRED                                    | 9   |
| RESTRICTIONS INVOLVING LOBBYING                              |     |
| INFORMATION REQUIRED ON REPORTS 1                            |     |
| CONTRIBUTIONS  |     |
| PLEDGES  |     |
| LOANS  |     |
| CONTRIBUTIONS OF PERSONAL SERVICES                           | 12  |

| CONTRIBUTIONS OF PERSONAL TRAVEL                                    | 12 |
|---|----|
| CONTRIBUTIONS FROM OUT-OF-STATE POLITICAL COMMITTEES                | 12 |
| EXPENDITURES  |    |
| UNPAID INCURRED OBLIGATIONS   |    |
| EXPENDITURES MADE BY CREDIT CARD                                    |    |
| CAMPAIGN EXPENDITURES FROM PERSONAL FUNDS                           |    |
| OFFICEHOLDER EXPENDITURES FROM PERSONAL FUNDS                       | 14 |
| DIRECT EXPENDITURES   | 14 |
| SUPPORTING POLITICAL COMMITTEES                                     |    |
| PAYMENTS TO A BUSINESS OF THE CANDIDATE OR OFFICEHOLDER             | 15 |
| INTEREST EARNED AND OTHER CREDITS/GAINS/REFUNDS                     |    |
| PURCHASE OF INVESTMENTS   | 15 |
| TOTAL POLITICAL CONTRIBUTIONS MAINTAINED                            | 15 |
| TIME OF ACCEPTING CONTRIBUTION                                      | 16 |
| TIME OF MAKING EXPENDITURE  | 16 |
| PREPARING REPORTS   | 16 |
| FORMS   | 16 |
| SIGNATURE REQUIRED  | 17 |
| FILING DEADLINES  |    |
| PERIODS COVERED BY REPORTS  |    |
| DEADLINE ON WEEKEND OR HOLIDAY                                      | 17 |
| 5 P.M. DEADLINE   |    |
| DELIVERY BY MAIL OR OTHER CARRIER                                   |    |
| RETENTION OF RECORDS USED FOR REPORTS                               |    |
| REPORTS   |    |
| SEMIANNUAL REPORTS  |    |
| REPORT DUE 8 DAYS BEFORE A RUNOFF ELECTION                          |    |
| MODIFIED REPORTING  | 19 |
| "15 <sup>TH</sup> DAY AFTER APPOINTMENT OF CAMPAIGN TREASURER BY AN |    |
| OFFICEHOLDER" REPORT  |    |
| FINAL REPORT  | /  |
| ANNUAL REPORT OF UNEXPENDED CONTRIBUTIONS                           |    |
| FINAL DISPOSITION OF UNEXPENDED CONTRIBUTIONS REPORT                |    |
| THINGS TO REMEMBER  |    |
| ENDING FILING OBLIGATIONS   |    |
| FINAL REPORT  |    |
| ANNUAL REPORT OF UNEXPENDED CONTRIBUTIONS                           |    |
| REPORT OF FINAL DISPOSITION OF UNEXPENDED CONTRIBUTIONS             |    |
| THINGS TO REMEMBER  |    |
| PENALTIES FOR REPORTING VIOLATIONS                                  |    |
| CAMPAIGN FINANCE RESTRICTIONS                                       | 22 |

## **INTRODUCTION**

This guide is a summary of reporting requirements and other regulations set out in Title 15 of the Texas Election Code (Chs. 251-259) and in the rules adopted by the Texas Ethics Commission. This guide applies to candidates for and officeholders in most local offices in Texas.

This guide does not apply to candidates for or officeholders of statewide elective offices, the State Legislature, seats on the State Board of Education, or multi-county district offices. Nor does it apply to candidates for or judges of statutory county courts, statutory probate courts, or district courts.

#### **IMPORTANT UPDATES**

As directed by section 571.064 of the Texas Government Code, the Commission is required to annually adjust certain reporting thresholds upward to the nearest multiple of \$10 in accordance with the percentage increase for the previous year in the Consumer Price Index for Urban Consumers published by the Bureau of Labor Statistics of the United States Department of Labor.

These changes will be made effective January 1<sup>st</sup> of each calendar year; the affected numbers and corresponding new thresholds are located in 1 T.A.C. §18.31, which can be found here: <u>https://www.ethics.state.tx.us/rules/</u>. The higher itemization thresholds will be reflected on the paper forms and in these instructions, as applicable.

Verify that you are using the correct thresholds and forms that apply to your filing. For example, if you are filing a campaign finance report or lobby activities report that is due in January of 2021, you must use the forms and instructions that are applicable to the period ending December 31, 2020.

#### OFFICEHOLDERS

Officeholders as well as candidates are subject to regulation under Title 15. An officeholder who has a campaign treasurer appointment on file with a filing authority is a "candidate" for purposes of Title 15 and is subject to all the regulations applicable to candidates. An officeholder who does not have a campaign treasurer appointment on file is subject only to the regulations applicable to officeholders.

Most of the requirements discussed in this guide apply to both candidates (individuals who have a campaign treasurer appointment on file) and to officeholders who do not have a campaign treasurer appointment on file. The guide will indicate whether a particular requirement applies to individuals who have campaign treasurer appointments on file, to officeholders who do not have campaign treasurer appointments on file, or to both.

#### JUDICIAL CANDIDATES AND OFFICEHOLDERS

Candidates for and officeholders in most judicial offices are subject to various restrictions that do not apply to other candidates and officeholders. Those candidates and officeholders should review the CAMPAIGN FINANCE GUIDE FOR JUDICIAL CANDIDATES AND OFFICEHOLDERS and the POLITICAL ADVERTISING GUIDE which are available on the commission's website.

**Nonjudicial Officeholder Seeking Judicial Office.** Pursuant to Op. Tex. Ethics Comm'n No. 465 (2005), a nonjudicial officeholder who becomes a judicial candidate is required to file two campaign finance reports, one reporting nonjudicial activity and the other reporting judicial activity. Alternatively, a nonjudicial officeholder who becomes a judicial candidate may select to file a single report that includes both candidate and officeholder activity if the activity is clearly and properly reported. *See* the CAMPAIGN FINANCE GUIDE FOR JUDICIAL CANDIDATES AND OFFICEHOLDERS for more information.

#### FEDERAL OFFICES

This guide does not apply to candidates for federal offices. Candidates for federal offices should contact the Federal Election Commission. The FEC's toll-free number is (800) 424-9530.

#### FILING AUTHORITIES

Title 15 requires candidates and officeholders to file various documents and reports with the appropriate filing authority.

The filing authority for a local candidate or officeholder depends on the nature of the office sought or held.

**County Clerk.** The county clerk (or the county elections administrator if the county has an elections administrator, or tax assessor-collector if the county's commissioners court has transferred the filing authority function to the tax assessor-collector and the county clerk and tax assessor-collector have agreed to the transfer) is the appropriate filing authority for a candidate for:

- a county office;
- a precinct office;
- a district office (except for multi-county district offices); and
- an office of a political subdivision other than a county if the political subdivision is within the boundaries of a single county and if the governing body of the political subdivision has not been formed.

**Other local filing authority.** If a candidate is seeking an office of a political subdivision other than a county, the appropriate filing authority is the *clerk or secretary of the governing body* of the political subdivision. If the political subdivision has no clerk or secretary, the appropriate filing authority is the governing body's presiding officer.

**Texas Ethics Commission.** The Texas Ethics Commission is the appropriate filing authority for candidates for:

• Multi-county district offices. (Reminder: This guide does not apply to multicounty district offices.) • An office of a political subdivision other than a county if the political subdivision includes areas in more than one county and if the governing body of the political subdivision has not been formed.

#### POLITICAL COMMITTEES (PACS)

Often a candidate or officeholder chooses to establish a specific-purpose political committee. A political committee is subject to *separate* filing requirements. Establishing a specific-purpose political committee does not relieve a candidate or officeholder of the obligation to file as an individual. For more information about political committees, see the Ethics Commission's CAMPAIGN FINANCE GUIDE FOR POLITICAL COMMITTEES.

#### FINANCIAL DISCLOSURE STATEMENTS

Some local candidates and officeholders are required to file an annual personal financial statement in accordance with Government Code Chapter 572 or Local Government Code Chapter 159. This statement is not a campaign finance document, and is not addressed in this guide.

#### FEDERAL INCOME TAX

This pamphlet does not address the federal tax implications of campaign finance. Questions regarding federal tax law should be directed to the Internal Revenue Service.

#### **TEXAS ETHICS COMMISSION**

If you have a question about how Title 15 applies to you, you may call the Ethics Commission for assistance or you may request a written advisory opinion.

The Ethics Commission has authority to impose fines for violations of Title 15. If you have evidence that a person has violated Title 15, you may file a sworn complaint with the Ethics Commission.

The Ethics Commission's mailing address is P.O. Box 12070, Austin, Texas 78711. The phone number is (512) 463-5800. The Ethics Commission maintains a website at *www.ethics.state.tx.us*.

## **APPOINTING A CAMPAIGN TREASURER**

If you plan to run for a public office in Texas (except for a federal office), you must file an APPOINTMENT OF A CAMPAIGN TREASURER BY A CANDIDATE (FORM CTA) with the proper filing authority before you become a candidate, even if you do not intend to accept campaign contributions or make campaign expenditures. A "candidate" is a person who knowingly and willingly takes affirmative action for the purpose of gaining nomination or election to public office or for the purpose of satisfying financial obligations incurred by the person in connection with the campaign for nomination or election. Examples of affirmative action include:

(A) the filing of a campaign treasurer appointment, except that the filing does not constitute candidacy or an announcement of candidacy for purposes of the

automatic resignation provisions of Article XVI, Section 65, or Article XI, Section 11, of the Texas Constitution;

- (B) the filing of an application for a place on the ballot;
- (C) the filing of an application for nomination by convention;
- (D) the filing of a declaration of intent to become an independent candidate or a declaration of write-in candidacy;
- (E) the making of a public announcement of a definite intent to run for public office in a particular election, regardless of whether the specific office is mentioned in the announcement;
- (F) before a public announcement of intent, the making of a statement of definite intent to run for public office and the soliciting of support by letter or other mode of communication;
- (G) the soliciting or accepting of a campaign contribution or the making of a campaign expenditure; and
- (H) the seeking of the nomination of an executive committee of a political party to fill a vacancy.

# NO CAMPAIGN CONTRIBUTIONS OR EXPENDITURES WITHOUT TREASURER APPOINTMENT ON FILE

Additionally, the law provides that you must file a campaign treasurer appointment form with the proper filing authority before you may accept a campaign contribution or make or authorize a campaign expenditure, including an expenditure from your personal funds. A filing fee paid to a filing authority to qualify for a place on a ballot is a campaign expenditure that may not be made before filing a campaign treasurer appointment form with the proper filing authority.

## APPOINTING TREASURER TRIGGERS REPORTING DUTIES

After a candidate has filed a form appointing a campaign treasurer, the candidate is responsible for filing periodic reports of contributions and expenditures. Filing reports is the responsibility of the candidate, not the campaign treasurer. Even if a candidate loses an election, he or she must continue filing reports until he or she files a final report. *See* "Ending Filing Obligations" in this guide. (An officeholder who files a final report, and thereby terminates his or her campaign treasurer appointment, may still be required to file semiannual reports of contributions and expenditures as an officeholder.)

## QUALIFICATIONS OF CAMPAIGN TREASURER

A person is ineligible for appointment as a campaign treasurer if the person is the campaign treasurer of a political committee that has outstanding filing obligations (including outstanding penalties). This prohibition does not apply if the committee in connection with which the ineligibility arose has not accepted more than \$5,000 in political contributions or made more than \$5,000 in political expenditures in any semiannual reporting period. A person who violates

this prohibition is liable for a civil penalty not to exceed three times the amount of political contributions accepted or political expenditures made in violation of this provision.

#### **DUTIES OF CAMPAIGN TREASURER**

A candidate's campaign treasurer has no legal duties. (**Note:** The campaign treasurer of a *political committee* is legally responsible for filing reports.)

#### **EFFECTIVE DATE OF APPOINTMENT**

A campaign treasurer appointment is effective when filed. A hand-delivered appointment takes effect on the date of delivery. A mailed appointment takes effect on the date of the postmark.

#### CODE OF FAIR CAMPAIGN PRACTICES

A filing authority should provide to each individual who files a campaign treasurer appointment a form containing a Code of Fair Campaign Practices. A candidate may pledge to conduct his or her campaign in accordance with the principles and practices set out in the Code by signing the form and filing it with the appropriate filing authority.

#### **APPOINTMENT BY OFFICEHOLDER**

If an officeholder files an appointment of campaign treasurer after a period in which he or she did not have a campaign treasurer appointment on file, the officeholder may have to file a report of contributions and expenditures no later than 15 days after filing the appointment of campaign treasurer. *See* "15th Day After Appointment of Campaign Treasurer by Officeholder" in this guide. An officeholder who *changes* a campaign treasurer is not required to file this report.

**Note**: An officeholder who has a campaign treasurer appointment on file is a candidate for purposes of Title 15.

#### FILING FOR A PLACE ON THE BALLOT

Filing a campaign treasurer appointment and filing for a place on the ballot are two completely separate actions. The Secretary of State can provide information about filing for a place on the ballot. Call the Secretary of State at (512) 463-5650 or toll-free at (800) 252-8683.

#### CHANGING TREASURERS

A candidate may change campaign treasurers at any time by filing an amended appointment of campaign treasurer (FORM ACTA). Filing an appointment of a new treasurer automatically terminates the appointment of the old treasurer.

#### TRANSFERRING TO A DIFFERENT FILING AUTHORITY

If a candidate has a campaign treasurer appointment on file with one filing authority and wishes to accept campaign contributions or make campaign expenditures in connection with a candidacy for an office that would require reporting to a different filing authority, the candidate must file a new campaign treasurer appointment and a copy of the old campaign treasurer appointment (certified by original authority) with the second filing authority. The candidate should also

provide written notice to the original filing authority that future reports will be filed with another authority. In general, funds accepted in connection with one office may be used in connection with a campaign for a different office, as long as neither of the offices is a judicial office.

#### TERMINATING A CAMPAIGN TREASURER APPOINTMENT

A candidate may terminate a campaign treasurer appointment by filing an amended appointment of campaign treasurer or by filing a final report.

A campaign treasurer may terminate his or her own appointment by notifying both the candidate and the filing authority in writing. The termination is effective on the date the candidate receives the notice or on the date the filing authority receives the notice, whichever is later.

#### **DECIDING NOT TO RUN**

A campaign treasurer appointment does not simply expire. An individual who has a campaign treasurer appointment on file must file reports of contributions and expenditures until he or she files a final report with the filing authority. *See* "Ending Filing Obligations" in this guide.

#### THINGS TO REMEMBER

- If you plan to run for a public office in Texas (except for a federal office), you must file an APPOINTMENT OF A CAMPAIGN TREASURER BY A CANDIDATE (FORM CTA) with the proper filing authority before you become a candidate, even if you do not intend to accept campaign contributions or make campaign expenditures.
- A person may not accept a campaign contribution or make a campaign expenditure unless the person has a campaign treasurer appointment on file with the proper filing authority.
- Once a person files a form appointing a campaign treasurer, the person is a candidate for disclosure filing purposes and is responsible for filing periodic reports of contributions and expenditures with the proper filing authority until the person files a "final report."
- The candidate, not the campaign treasurer, is responsible for filing periodic reports of contributions and expenditures.
- Filing a campaign treasurer appointment does not automatically "sign you up" for a place on the ballot. The Secretary of State can provide information about getting on the ballot. Call (512) 463-5650 or (800) 252-8683.

## POLITICAL CONTRIBUTIONS AND EXPENDITURES

Title 15 regulates political contributions and political expenditures. There are two types of political contributions: campaign contributions and officeholder contributions. Similarly, there are two kinds of political expenditures: campaign expenditures and officeholder expenditures.

#### CAMPAIGN CONTRIBUTIONS

A person makes a campaign contribution to a candidate if the person provides or promises something of value with the intent that it be used in connection with a campaign. A contribution of goods or services is an "in-kind" campaign contribution. A loan is considered to be a contribution unless it is from an incorporated financial institution that has been in business for more than a year. Candidates must report all loans made for campaign purposes, including loans that are not "contributions."

- Donations to a candidate at a fund-raiser are campaign contributions.
- The provision of office space to a candidate is an "in-kind" campaign contribution.
- A promise to give a candidate money is a campaign contribution.
- An item donated to be auctioned at a fund-raiser is an "in-kind" campaign contribution. The purchase of the item at the auction is also a contribution.
- A campaign volunteer is making a contribution in the form of personal services. (Contributions of personal services are sometimes not required to be reported. See "Contributions of Personal Services" in this guide.)

**Note:** An individual may not accept a campaign contribution without an appointment of campaign treasurer on file with the proper filing authority.

#### **CAMPAIGN EXPENDITURES**

A campaign expenditure is a payment or an agreement to make a payment in connection with a campaign for an elective office.

- Paying a filing fee in connection with an application for a place on a ballot is a campaign expenditure.
- Purchasing stationery for fund-raising letters is a campaign expenditure.
- Renting a field to hold a campaign rally is a campaign expenditure.
- Paying people to put up yard signs in connection with an election is a campaign expenditure.

**Note:** An individual may not make a campaign expenditure unless he or she has a campaign treasurer appointment on file with the proper filing authority.

#### **OFFICEHOLDER CONTRIBUTIONS**

The provision of or a promise to provide goods or services to an officeholder that is intended to defray expenses in connection with an officeholder's duties or activities is an officeholder contribution if the expenses are not reimbursable with public money. A contribution of goods or services is an "in-kind" officeholder contribution.

A loan from an incorporated financial institution that has been in business for more than a year is not considered a contribution, but an officeholder must report any such loans made for officeholder purposes.

An officeholder is not required to have a campaign treasurer appointment on file to accept officeholder contributions. An officeholder who does not have a campaign treasurer on file may not accept *campaign* contributions.

#### **OFFICEHOLDER EXPENDITURES**

A payment or agreement to pay certain expenses in connection with an officeholder's duties or activities is an officeholder expenditure if the expenses are not reimbursable with public money.

An officeholder is not required to have a campaign treasurer appointment on file to make officeholder expenditures. An officeholder who does not have a campaign treasurer on file may not make *campaign* expenditures.

#### CAMPAIGN EXPENDITURES BY OFFICEHOLDER

An officeholder who has a campaign treasurer appointment on file may accept both campaign contributions and officeholder contributions and make both campaign expenditures and officeholder expenditures. On a report, there is no need for an officeholder who is a candidate to distinguish between campaign contributions and officeholder contributions or between campaign expenditures and officeholder expenditures. Both campaign contributions and officeholder contributions and officeholder contributions are reported as "political contributions" and both campaign expenditures and officeholder expenditures are reported as "political expenditures."

An officeholder who does not have a campaign treasurer on file may accept officeholder contributions and make officeholder expenditures but may not accept campaign contributions or make campaign expenditures.

#### PERMISSIBLE USE OF POLITICAL CONTRIBUTIONS

An officeholder may use officeholder contributions for campaign purposes if the officeholder has an appointment of campaign treasurer on file. Candidates and officeholders may not convert political contributions to personal use. *See* "Campaign Finance Restrictions" in this guide.

#### USE OF POLITICAL FUNDS TO RENT OR PURCHASE REAL PROPERTY

A candidate or officeholder is prohibited from using political funds to purchase real property or

to pay the interest on or principal of a note for the purchase of real property.

A candidate or officeholder may not knowingly make or authorize a payment from political funds for the rental or purchase of real property from: (1) a person related to the candidate or officeholder within the second degree of consanguinity or affinity as determined under Chapter 573, Government Code; or (2) a business in which the candidate or officeholder (or a person related to the candidate or officeholder within the second degree of consanguinity or affinity) has a participating interest of more than 10 percent, holds a position on the governing body, or serves as an officer. Tex. Elec. Code § 253.038 (a-1). This restriction applies to a payment made from political funds on or after September 1, 2007, without regard to whether the payment was made under a lease or other agreement entered into before that date.

#### ACCEPTING CONTRIBUTIONS

A candidate or officeholder must report contributions that he or she has *accepted*. Receipt is different from acceptance. A decision to *accept* a contribution must be made by the end of the reporting period during which the contribution is received.

**Failure to make a determination about acceptance or refusal.** If a candidate or officeholder fails to make a timely determination to accept or refuse a contribution by the deadline, the contribution is considered to have been accepted.

**Returning refused contributions.** If a candidate or officeholder receives a political contribution but does not accept it, he or she must return the contribution not later than the 30th day after the end of the reporting period in which the contribution was received. Otherwise, the contribution is considered to have been accepted.

#### REIMBURSEMENT FOR POLITICAL EXPENDITURES FROM PERSONAL FUNDS

If a candidate or officeholder makes political expenditures from personal funds, he or she may use political contributions to reimburse himself or herself if the expenditures are properly reported either on the reporting schedule for loans or on the reporting schedule for political expenditures from personal funds. In order for a candidate or officeholder to use political contributions to reimburse his or her personal funds, the political expenditure from personal funds must be properly reported on the report covering the period in which the expenditures are made. A filed report may not be later corrected to indicate an intention to reimburse personal funds from political contributions.

If a candidate or officeholder deposits personal funds in an account in which political contributions are held as permitted by section 253.0351(c) of the Election Code, the deposited amount must be reported as a loan and reimbursements to the candidate or officeholder may not exceed the amount reported as a loan. *See* "Campaign Expenditures from Personal Funds" in this guide for additional information.

#### SEPARATE ACCOUNT REQUIRED

A candidate or officeholder must keep political contributions in one or more accounts that are separate from any other account maintained by the candidate or officeholder. (There is no

requirement to keep campaign contributions in a separate account from officeholder contributions.)

#### **RESTRICTIONS INVOLVING LOBBYING**

The 2019 legislature passed House Bill 2677 to amend Chapter 305 of the Government Code and Chapter 253 of the Election Code to enact the following restrictions. Each prohibition begins on September 27, 2019. For the language of the bill, go to <a href="https://capitol.texas.gov/tlodocs/86R/billtext/html/HB02677F.htm">https://capitol.texas.gov/tlodocs/86R/billtext/html/HB02677F.htm</a>.

**Making Political Contributions and Direct Campaign Expenditures.** Unless expressly prohibited, a lobbyist may make political contributions and direct campaign expenditures. The campaign finance law, however, generally prohibits corporations and labor organizations from making political contributions. Tex. Elec. Code § 253.094.

Section 253.006 of the Election Code prohibits a person required to register as a lobbyist under Chapter 305 of the Government Code from making political contributions or direct campaign expenditures from certain sources of funds. A person required to register as a lobbyist is prohibited from making or authorizing a political contribution to another candidate, officeholder, or political committee, or making or authorizing a direct campaign expenditure, from political contributions accepted by:

- (1) the lobbyist as a candidate or officeholder;
- (2) a specific-purpose committee that supports the lobbyist as a candidate or assists the lobbyist as an officeholder; or
- (3) a political committee that accepted a political contribution from (1) or (2), described above, during the two years immediately before the contribution or expenditure was made.

**Two-Year Lobbying Prohibition After Making a Political Contribution or Direct Campaign Expenditure.** Section 253.007 of the Election Code prohibits lobbying by persons who make political contributions or direct campaign expenditures from certain sources of funds. A person who makes a political contribution to another candidate, officeholder, or political committee, or makes a direct campaign expenditure, from political contributions accepted by the person as a candidate or officeholder is prohibited from engaging in activities that require registration as a lobbyist under Chapter 305 of the Government Code for two years thereafter.

However, an exception to this prohibition allows a person who does not receive compensation other than reimbursement for actual expenses to lobby on behalf of a nonprofit organization, a group of low-income individuals, or a group of individuals with disabilities.

**Lobby Expenditures from Political Contributions.** Section 305.029 of the Government Code prohibits certain lobby expenditures made from political contributions. A lobbyist registered under Chapter 305 of the Government Code, or a person on behalf of the lobbyist and with the lobbyist's consent or ratification, is prohibited from making a reportable lobby expenditure from a political contribution accepted by:

(1) the lobbyist as a candidate or officeholder;

- (2) a specific-purpose committee that supports the lobbyist as a candidate or assists the lobbyist as an officeholder; or
- (3) a political committee that accepted a political contribution from (1) or (2), described above, during the two years immediately before the lobbyist made or authorized the expenditure.

# INFORMATION REQUIRED ON REPORTS

#### CONTRIBUTIONS

A report must disclose the amount of each contribution or the value and nature of any in-kind contribution, as well as the name and address of the individual or political committee making the contribution, and the date of the contribution. (Detailed information about a contributor is not required to be reported if the contributor contributed \$100 or less during the reporting period. However, all contributions made electronically must be itemized with this information.)

#### PLEDGES

Promises to transfer money, goods, services, or other things of value are contributions. If a filer accepts such a promise, he or she must report it (along with the information required for other contributions) on the reporting schedule for "pledges." Once a pledge has been received, it is reported on the appropriate receipts schedule for the reporting period in which the pledge is received. A pledge that is actually received in the same reporting period in which the pledge was accepted shall be reported only on the appropriate receipts schedule.

**Note:** A pledge is not a contribution unless it has been accepted.

*Example 1:* In June a supporter promises that he will give Juan Garcia \$1,000 in the last week before the November election. Juan accepts his promise. Juan must report the pledge on his July 15 report. Juan must also report a political contribution when the pledge is actually received. (**Note:** If Juan receives the pledge during the July semiannual reporting period then he does not report the pledge and only reports a political contribution. Also, if he never receives the \$1,000, he does not amend his report to delete the entry for the pledge.)

*Example 2*: At a party, an acquaintance says to Juan, "I'd like to give you some money; call me at my office." Juan agrees to call. At this point, Juan has accepted nothing and has nothing to report. Juan has not agreed to accept money; he has merely agreed to call.

#### LOANS

Loans made for campaign or officeholder purposes are reportable. A filer must report the amount of a loan, the date the loan is made, the interest rate, the maturity date, the type of collateral, and the name and address of the lender. The filer must also report the name, address, principal occupation, and employer of any guarantor and the amount guaranteed by the guarantor. (Detailed information is not required if a particular lender lent \$100 or less during a reporting period.) If a candidate or officeholder deposits personal funds in an account in which political contributions are held as permitted by section 253.0351(c) of the Election Code, the deposited amount must be reported as a loan. *See* "Campaign Expenditures from Personal

Funds" in this guide for additional information.

**Note:** A loan from an incorporated financial institution that has been in business for more than one year is not a contribution. Other loans are considered to be contributions. This distinction is important because of the prohibition on contributions from banks and certain other financial institutions. *See* "Campaign Finance Restrictions" in this guide. All loans are reported on the same schedule, regardless of whether they are contributions. Additionally, the forgiveness of a loan is a reportable in-kind contribution. *See* 1 Tex. Admin. Code § 20.64.

#### CONTRIBUTIONS OF PERSONAL SERVICES

A political contribution consisting of an individual's personal services is not required to be reported if the individual receives no compensation *from any source* for the services.

#### CONTRIBUTIONS OF PERSONAL TRAVEL

A political contribution consisting of personal travel expense incurred by an individual is not required to be reported if the individual receives no reimbursement for the expense.

#### CONTRIBUTIONS FROM OUT-OF-STATE POLITICAL COMMITTEES

There are restrictions on contributions from out-of-state political committees. The fact that a political committee has a mailing address outside of Texas does not mean that the committee is an out-of-state political committee for purposes of these restrictions. A political committee that has a campaign treasurer appointment on file in Texas is not an out-of-state political committee for purposes of these restrictions.

**Contributions over \$1,010 in a reporting period.** Before *accepting* more than \$1,010 in a reporting period from an out-of-state committee, a candidate or officeholder must obtain either (1) a written statement, certified by an officer of the out-of-state political committee, listing the full name and address of each person who contributed more than \$200 to the out-of-state political committee during the 12 months immediately preceding the contribution, *or* (2) a copy of the out-of-state political committee's statement of organization filed as required by law with the Federal Election Commission and certified by an officer of the out-of-state committee.

This documentation must be included with the report of contributions and expenditures for the period in which the contribution was received.

**Contributions of \$1,010 or less in a reporting period.** For a contribution of \$1,010 or less from an out-of-state committee in a reporting period, there is no requirement to obtain documentation *before accepting* the contribution. But there is a requirement to include certain documentation with the report of the contribution. The report must include *either* (1) a copy of the out-of-state political committee's statement of organization filed as required by law with the Federal Election Commission and certified by an officer of the out-of-state committee, *or* (2) the committee's name, address, and phone number; the name of the person appointing the committee's campaign treasurer; and the name, address, and phone number of the committee's campaign treasurer.

#### EXPENDITURES

A filer must report any campaign expenditure (regardless of whether it is made from political contributions or from personal funds) and any political expenditure (campaign or officeholder) from political contributions (regardless of whether the expenditure is a political expenditure). A filer must also report unpaid incurred obligations. *See* "Unpaid Incurred Obligations" in this guide. If the total expenditures to a particular payee do not exceed \$200 during the reporting period, a filer may report those expenditures as part of a lump sum. Otherwise, a filer must report the date of an expenditure, the name and address of the person to whom the expenditure is made, and the purpose of the expenditure.

#### **UNPAID INCURRED OBLIGATIONS**

An expenditure that is not paid during the reporting period in which the obligation to pay the expenditure is incurred shall be reported on the Unpaid Incurred Obligations Schedule for the reporting period in which the obligation to pay is incurred.

The use of political contributions to pay an expenditure previously disclosed on an Unpaid Incurred Obligations Schedule shall be reported on the appropriate disbursements schedule for the reporting period in which the payment is made.

The use of personal funds to pay an expenditure previously disclosed on an Unpaid Incurred Obligations Schedule shall be reported on the Political Expenditure Made from Personal Funds Schedule for the reporting period in which the payment is made.

#### EXPENDITURES MADE BY CREDIT CARD

An expenditure made by a credit card must be reported on the Expenditures Made to Credit Card Schedule for the reporting period in which the expenditure is made. The report must identify the vendor who receives the payment from the credit card company.

The use of political contributions to make a payment to a credit card company must be reported on the appropriate disbursements schedule for the reporting period in which the payment is made and identify the credit card company receiving the payment.

The use of personal funds to make a payment to a credit card company must be reported on the Political Expenditure Made from Personal Funds Schedule for the reporting period in which the payment is made and identify the credit card company receiving the payment.

#### CAMPAIGN EXPENDITURES FROM PERSONAL FUNDS

A candidate must report all campaign expenditures, whether made from political contributions or from personal funds. In order to use political contributions to reimburse himself or herself for campaign expenditures from personal funds, the candidate must properly report the expenditures either on the reporting schedule for loans or on the reporting schedule for political expenditures from personal funds. If the candidate does not indicate the intention to seek reimbursement on that report, he or she may not later correct the report to permit reimbursement.

If a candidate or officeholder deposits personal funds in an account in which political contributions are held as permitted by section 253.0351(c) of the Election Code, the deposited

amount must be reported as a loan on Schedule E. Political expenditures made from that loan, and any subsequent expenditures to reimburse the candidate or officeholder, must be reported on Schedule F1. The reimbursement may not exceed the amount reported as a loan. Any unexpended funds from such a loan are required to be included in the total amount of political contributions maintained as of the last day of the reporting period. Note: Personal funds deposited in an account in which political contributions are held are subject to the personal use restrictions.

#### OFFICEHOLDER EXPENDITURES FROM PERSONAL FUNDS

An officeholder is not required to report *officeholder expenditures* made from personal funds unless he or she intends to be reimbursed from political contributions. This rule applies regardless of whether an officeholder has an appointment of campaign treasurer on file.

In order for an officeholder to use political contributions to reimburse an officeholder expenditure from personal funds, the officeholder must properly report the expenditures either on the reporting schedule for loans or on the reporting schedule for political expenditures from personal funds. If the officeholder does not indicate the intention to seek reimbursement, he or she may not later correct the report to permit reimbursement.

If a candidate or officeholder deposits personal funds in an account in which political contributions are held as permitted by section 253.0351(c) of the Election Code, the deposited amount must be reported as a loan on Schedule E. Political expenditures made from that loan, and any subsequent expenditures to reimburse the candidate or officeholder, must be reported on Schedule F1. The reimbursement may not exceed the amount reported as a loan. Any unexpended funds from such a loan are required to be included in the total amount of political contributions maintained as of the last day of the reporting period. Note: Personal funds deposited in an account in which political contributions are held are subject to the personal use restrictions.

#### **DIRECT EXPENDITURES**

A direct campaign expenditure is "a campaign expenditure that does not constitute a campaign contribution by the person making the expenditure." As a practical matter, a direct campaign expenditure is an expenditure to support a candidate incurred without the candidate's prior consent or approval.

If a candidate or officeholder makes a direct campaign expenditure to support *another* candidate or officeholder, the expenditure must be included on the reporting schedule for political expenditures, and the report must indicate that the expenditure was a direct campaign expenditure.

#### SUPPORTING POLITICAL COMMITTEES

A political committee that accepts political contributions or makes political contributions on behalf of a candidate or officeholder is required to give the candidate or officeholder notice of that fact. The candidate or officeholder must report the receipt of such a notice on the report covering the period in which he or she receives the notice.

#### PAYMENTS TO A BUSINESS OF THE CANDIDATE OR OFFICEHOLDER

A candidate or officeholder is required to report payments from political funds to a business in which the candidate or officeholder has a participating interest of more than 10 percent; a position on the governing body of the business; *or* a position as an officer of a business.

A candidate or officeholder may not make a payment to such a business if the payment is for personal services rendered by the candidate or officeholder or by the spouse or dependent child of the candidate or officeholder. (Nor may a candidate or officeholder use political contributions to pay directly for such personal services.) Other payments to such a business are permissible only if the payment does not exceed the amount necessary to reimburse the business for actual expenditures made by the business. *See generally* Op. Tex. Ethics Comm'n No. 35 (1992).

A candidate or officeholder may not make or authorize a payment from political funds for the rental or purchase of real property from such a business. *See* "Use of Political Funds to Rent or Purchase Real Property" in this guide.

#### INTEREST EARNED AND OTHER CREDITS/GAINS/REFUNDS

A candidate or officeholder is required to disclose information regarding the following types of activity from political contributions:

- any credit, interest, rebate, refund, reimbursement, or return of a deposit fee resulting from the use of a political contribution or an asset purchased with a political contribution, the amount of which exceeds \$130;
- any proceeds of the sale of an asset purchased with a political contribution, the amount of which exceeds \$130; and
- any other gain from a political contribution, the amount of which exceeds \$130.

A candidate or officeholder must use Schedule K to report such information. Although you are not required to do so, you may also report any credit/gain/refund or interest that does not exceed \$130 in the period on this schedule. (Previously, this was an optional schedule because a candidate or officeholder was not required to report this information.) A candidate or officeholder may not use interest and other income from political contributions for personal purposes. Political expenditures made from such income must be reported on the expenditures schedule.

#### PURCHASE OF INVESTMENTS

A candidate or officeholder must report any investment purchased with a political contribution, the amount of which exceeds \$130. This information must be disclosed on Schedule F3 of the campaign finance report.

#### TOTAL POLITICAL CONTRIBUTIONS MAINTAINED

The law requires you to disclose the total amount of political contributions accepted, including interest or other income on those contributions, maintained in one or more accounts in which

political contributions are deposited as of the last day of the reporting period. The "total amount of political contributions maintained" includes: the total amount of political contributions maintained in one or more accounts, including the balance on deposit in banks, savings and loan institutions and other depository institutions; the present value of any investments that can be readily converted to cash, such as certificates of deposit, money market accounts, stocks, bonds, treasury bills, etc.; and the balance of political contributions accepted and held in any online fundraising account over which the filer can exercise control by making a withdrawal, expenditure, or transfer. 1 Tex. Admin. Code § 20.50.

The total amount of political contributions maintained does NOT include personal funds that the filer intends to use for political expenditures, *unless* the personal funds have been disclosed as a loan to your campaign and deposited into an account in which political contributions are held as permitted by section 253.0351(c) of the Election Code. Any unexpended funds from such a loan are required to be included in the total amount of political contributions maintained as of the last day of the reporting period. Note: Personal funds deposited in an account in which political contributions are held are subject to the personal use restrictions.

#### TIME OF ACCEPTING CONTRIBUTION

A filer must report the date he or she *accepts* a political contribution. The date of receipt may be different from the date of acceptance. *See* "Accepting Contributions" in this guide.

#### TIME OF MAKING EXPENDITURE

For reporting purposes, an expenditure is made when the amount of the expenditure is readily determinable. An expenditure that is not paid during the reporting period in which the obligation to pay is incurred must be reported on the reporting schedule for "Unpaid Incurred Obligations," and then reported again on the appropriate expenditure schedule when payment is actually made. If a filer cannot determine the amount of an expenditure until a periodic bill, the date of the expenditure is the date the bill is received.

**Credit Card Expenditures.** For purposes of 30 day and 8 day pre-election reports, the date of an expenditure made by a credit card is the date of the purchase, not the date of the credit card bill. For purposes of other reports, the date of an expenditure made by a credit card is the date of receipt of the credit card statement that includes the expenditure. For additional information regarding how to report expenditures made by credit card, *see* "Expenditures Made by Credit Card" in this guide.

## **PREPARING REPORTS**

#### FORMS

Reporting forms are available at *http://www.ethics.state.tx.us*. An individual who is both a candidate and an officeholder files one report for each reporting period and is not required to distinguish between campaign activity and officeholder activity.

#### SIGNATURE REQUIRED

The candidate or officeholder, not the campaign treasurer, must sign reports.

#### FILING DEADLINES

The next section of this guide explains the types of reports candidates and officeholders are required to file. Annual filing schedules are available at *http://www.ethics.state.tx.us*.

**Note**: Deadlines for filing reports for special elections or runoff elections will not be listed on the filing schedule. Call the Ethics Commission for specific information in these cases.

#### PERIODS COVERED BY REPORTS

Each report covers activity during a specific time period. Generally, a report begins where the last report ended. For a candidate's first report, the beginning date will be the date the campaign treasurer appointment was filed. For an officeholder who is appointed to an elective office and who did not have a campaign treasurer appointment on file at the time of the appointment, the beginning date for the first report will be the date the officeholder took office. Generally, there should not be gaps between the periods covered or overlapping time periods. See "Reports" below for information about filing deadlines and periods covered by reports.

#### DEADLINE ON WEEKEND OR HOLIDAY

If the due date for a report falls on a Saturday, Sunday, or legal holiday, the report is due on the next regular business day.

#### **5 P.M. DEADLINE**

The deadline for filing a report is 5 p.m. on the due date.

#### **DELIVERY BY MAIL OR OTHER CARRIER**

For most reporting deadlines, a document is considered timely filed if it is properly addressed with postage or handling charges prepaid and bears a postmark or receipt mark of a common or contract carrier indicating a time on or before the deadline.

**Pre-Election Reports.** A report due 30 days before an election and a report due 8 days before an election must be *received* by the appropriate filing authority no later than the report due date to be considered filed on time.

#### **RETENTION OF RECORDS USED FOR REPORTS**

A filer must keep records of all information used to prepare a report of contributions and expenditures, including, for example, receipts or ledgers of contributions and expenditures. A filer must maintain the records for two years after the deadline for the report.

## REPORTS

#### SEMIANNUAL REPORTS

Generally, candidates and officeholders are required to file reports of contributions and expenditures by January 15 and July 15 of each year. The reports filed on these dates are known as semiannual reports. These reports must be filed even if there is no activity to report for the period covered.

However, there is an exception to this requirement for officeholders who file with a local filing authority, do not have a campaign treasurer appointment on file, and do not accept more than \$1,010 in officeholder contributions or make more than \$1,010 in officeholder expenditures during the period covered by the report.

#### **REPORTS DUE 30 DAYS AND 8 DAYS BEFORE AN ELECTION**

An *opposed* candidate in an upcoming election must file reports of contributions and expenditures 30 days and 8 days before the election. Each of these pre-election reports must be *received* by the appropriate filing authority no later than the report due date. (A person who has elected modified reporting and who remains eligible for modified reporting is not required to file these reports. *See* "Modified Reporting" in this guide.)

An opposed candidate is a candidate who has an opponent whose name is printed on the ballot. If a candidate's only opposition is a write-in candidate, that candidate is considered unopposed for filing purposes. (**Note:** A write-in candidate who accepts political contributions or makes political expenditures is subject to the reporting requirements discussed in this guide.)

The report that is due 30 days before the election covers the period that begins on the first day after the period covered by the last required report and ends the 40th day before the election. If this is a filer's first required report, the period covered by the report begins on the day the filer filed a campaign treasurer appointment.

The report that is due 8 days before the election covers the period that begins on the first day after the period covered by the last required report and ends on the 10th day before the election.

#### **REPORT DUE 8 DAYS BEFORE A RUNOFF ELECTION**

A candidate in a runoff must file a report 8 days before the runoff election. A runoff report must be *received* by the appropriate filing authority no later than the report due date. (A candidate who has elected modified reporting and who remains eligible for modified reporting is not required to file this report. *See* "Modified Reporting" below.)

This report covers a period that begins either the first day after the period covered by the last required report or the day the filer filed a campaign treasurer appointment (if this is the filer's first report of contributions and expenditures). The period covered by the runoff report ends the 10th day before the runoff election.

#### **MODIFIED REPORTING**

On the campaign treasurer appointment form, there is an option to choose modified reporting for the next election cycle. Modified reporting excuses an opposed candidate from filing reports 30 days and 8 days before an election and 8 days before a runoff. An opposed candidate is eligible for modified reporting only if the candidate does not intend to exceed either \$1,010 in contributions or \$1,010 in expenditures (excluding filing fees) in connection with an election.

If an opposed candidate selects modified reporting but exceeds a threshold before the 30th day before the election, the candidate must file reports 30 days and 8 days before the election.

If an opposed candidate selects modified reporting but exceeds the \$1,010 threshold for contributions or expenditures after the 30th day before the election, the filer must file a report within 48 hours of exceeding the threshold. (The filer must meet this deadline even if it falls on a weekend or a holiday.) At that point, the filer is no longer eligible for modified reporting and must file according to the regular filing schedule.

A selection to file on the modified reporting schedule lasts for an entire election cycle. In other words, the selection is valid for a primary, a primary runoff, and a general election (as long as the candidate does not exceed one of the \$1,010 thresholds). A candidate must submit an amended campaign treasurer appointment (FORM ACTA) to select modified reporting for a different election cycle.

# **"15<sup>TH</sup> DAY AFTER APPOINTMENT OF CAMPAIGN TREASURER BY AN OFFICEHOLDER" REPORT**

An officeholder must file a report after filing a campaign treasurer appointment. (A report is not required after a *change* in campaign treasurers.) This report of contributions and expenditures is due no later than 15 days after the campaign treasurer appointment was filed. The report must cover the period that begins the day after the period covered by the last required report. The period ends on the day before the campaign treasurer appointment was filed. (**Note:** A person who is *appointed* to elective office may not have filed any previous reports. In that case, the beginning date for the report due 15 days after the campaign treasurer appointment is the date the officeholder took office.) The report is not required if the officeholder did not accept more than \$1,010 in contributions or make more than \$1,010 in expenditures by the end of the reporting period.

#### FINAL REPORT

See "Ending Filing Obligations" below.

#### ANNUAL REPORT OF UNEXPENDED CONTRIBUTIONS

See "Ending Filing Obligations" below.

#### FINAL DISPOSITION OF UNEXPENDED CONTRIBUTIONS REPORT

See "Ending Filing Obligations" below.

#### THINGS TO REMEMBER

- An officeholder must file semiannual reports for any period during which he or she is an officeholder. (There is an exception to this rule for officeholders who do not have a campaign treasurer appointment on file and who do not accept more than \$1,010 in political contributions or make more than \$1,010 in political expenditures during the period covered by the report.)
- An opposed candidate in an election must file reports of contributions and expenditures 30 days and 8 days before the election, unless the candidate has selected (and remains eligible for) modified reporting. An opposed candidate who has not selected modified reporting must also file a report 8 days before a runoff election. A report due 30 days before an election and a report due 8 days before an election must be received by the appropriate filing authority no later than the report due date.
- An unopposed candidate is not required to file reports 30 days before an election or 8 days before an election but is required to file semiannual reports.
- A candidate who selects modified reporting must file semiannual reports.

A filer who selects modified reporting for one election cycle will be required to file on the regular reporting schedule for the next election cycle unless the filer submits an amended campaign treasurer appointment selecting modified reporting for the next election cycle.

# **ENDING FILING OBLIGATIONS**

#### FINAL REPORT

If a filer expects to accept no further political contributions and to make no further political expenditures and if the filer expects to take no further action to get elected to a public office, the filer may file a final report. Filing a final report terminates a filer's campaign treasurer appointment and relieves the filer from any additional filing obligations *as a candidate*. (Note: A candidate who does not have a campaign treasurer appointment on file may still be required to file a personal financial statement in accordance with Chapter 572 of the Government Code or Chapter 159 of the Local Government Code.) If the filer is an officeholder, the filer will still be subject to the filing requirements applicable to officeholders. A filer who is not an officeholder at the time of filing a final report *and* who has surplus political funds or assets will be required to file annual reports of unexpended contributions and a report of final disposition of unexpended contributions. *See* "Annual Report of Unexpended Contributions" and "Report of Final Disposition of Unexpended Contributions" below.

A filer who intends to continue accepting contributions to pay campaign debts should *not* terminate his or her campaign treasurer appointment. An individual must have a campaign treasurer appointment on file to accept contributions to offset campaign debts or to pay campaign debts.

Terminating a campaign treasurer appointment does not relieve a filer of responsibility for any delinquent reports or outstanding civil penalties.

#### ANNUAL REPORT OF UNEXPENDED CONTRIBUTIONS

The following individuals must file annual reports of unexpended contributions:

- a former officeholder who did not have a campaign treasurer appointment on file at the time of leaving office and who retained any of the following after filing his or her last report: political contributions, interest or other income from political contributions, or assets purchased with political contributions or interest or other income from political contributions.
- a former candidate (a person who previously had a campaign treasurer appointment on file) who was not an officeholder at the time of filing a final report and who retained any of the following at the time of filing a final report: political contributions, interest or other income from political contributions, or assets purchased with political contributions.

Annual reports are due not earlier than January 1 and not later than January 15 of each year. An annual report (FORM C/OH-UC) must contain the following information: (1) information about expenditures from or disposition of surplus funds or assets; (2) the amount of interest or other income earned on surplus funds during the previous year; and (3) the total amount of surplus funds and assets at the end of the previous year.

The obligation to file annual reports ends when the former candidate or officeholder files a report of final disposition of unexpended contributions.

#### **REPORT OF FINAL DISPOSITION OF UNEXPENDED CONTRIBUTIONS**

A former candidate or former officeholder who has disposed of all surplus funds and assets must file a report of final disposition of unexpended contributions. This report may be filed as soon as all funds have been disposed of.

A former candidate or former officeholder has six years from the date of filing a final report or leaving office (whichever is later) to dispose of surplus funds and assets. The latest possible date for filing a report of unexpended contributions is 30 days after the end of that six-year period.

At the end of the six-year period, a former candidate or officeholder *must* dispose of surplus assets or funds in one of the following ways:

- The former candidate or officeholder may give them to the political party with which he or she was affiliated when last on the ballot;
- The former candidate or officeholder may contribute them to a candidate or a political committee. (This triggers a requirement to file a report of the contribution.);
- The former candidate or officeholder may give them to the comptroller for deposit in the state treasury to be used to finance primary elections;

- The former candidate or officeholder may give them to one or more contributors, but the total returned to any person may not exceed the aggregate amount accepted from that person during the last two years during which the former candidate or officeholder accepted political contributions;
- The former candidate or officeholder may give them to certain charitable organizations; or
- The former candidate or officeholder may give them to a public or private postsecondary educational institution or an institution of higher education as defined by section 61.003(8), Education Code, for the purpose of assisting or creating a scholarship program.

#### THINGS TO REMEMBER

- Anyone who has an appointment of campaign treasurer on file must file periodic reports of campaign contributions and expenditures.
- An individual who expects no further reportable activity in connection with his or her candidacy, files a final report and thereby terminates his or her campaign treasurer appointment. (Note: A candidate who does not have a campaign treasurer appointment on file may still be required to file a personal financial statement in accordance with Chapter 572 of the Government Code or Chapter 159 of the Local Government Code.)
- An officeholder may be required to file semiannual reports even if he or she does not have a campaign treasurer appointment on file. A local officeholder who has not accepted more than \$1,010 in contributions or made more than \$1,010 in expenditures in a semiannual period since terminating his or her campaign treasurer appointment is not required to file a semiannual report for that period.

## PENALTIES FOR REPORTING VIOLATIONS

Any citizen may file a criminal complaint with the district attorney, a civil complaint with the Ethics Commission, or a civil action against a candidate or officeholder for violations of Title 15. Any penalty stemming from such complaints would be assessed against *the candidate or officeholder*, not the campaign treasurer.

# **CAMPAIGN FINANCE RESTRICTIONS**

Chapter 253 of the Election Code contains a number of restrictions regarding the acceptance and use of political contributions, including the following:

1. An individual may not accept a campaign contribution or make a campaign expenditure (including a campaign expenditure from personal funds) without a campaign treasurer appointment on file. Tex. Elec. Code § 253.031. An officeholder may accept officeholder contributions and make officeholder

expenditures regardless of whether he or she has a campaign treasurer appointment on file.

- 2. Political contributions from labor organizations and from most corporations are prohibited. Tex. Elec. Code § 253.091, *et seq.* Partnerships that include one or more corporate partners are subject to the prohibition.
- 3. Certain documentation must be obtained in order to accept contributions from an out-of-state political committee. Tex. Elec. Code § 253.032. *See* "Contributions from Out-of-State Political Committees" in this guide.
- 4. Cash contributions of more than \$100 in the aggregate from one contributor in a reporting period are prohibited. (Here "cash" means coins and currency, not checks.) Tex. Elec. Code § 253.033.
- 5. The use of political contributions to purchase real property is prohibited. There is also a restriction on the use of political funds to rent or purchase real property from a person related to the candidate or officeholder within the second degree of consanguinity or affinity or from a business in which the candidate or officeholder or such a relative has a participating interest of more than 10 percent, holds a position on the governing body, or serves as an officer. Tex. Elec. Code § 253.038.
- 6. Texas law does not allow anonymous contributions. Also, reports must disclose the actual source of a contribution, not an intermediary. Tex. Elec. Code § 253.001.
- 7. Personal use of political contributions is prohibited. Tex. Elec. Code § 253.035.
- 8. A candidate or officeholder may not use political contributions to pay for personal services rendered by the candidate or officeholder or by the spouse, or dependent children of the candidate or officeholder. There are also restrictions of a candidate's or officeholder's use of political contributions to make payments to a business in which the candidate or officeholder holds a participating interest of more than 10 percent, a position on the governing body of the business, or a position as an officer of the business. *See* to Op. Tex. Ethics Comm'n No. 35 (1992) (regarding the combined effect of this prohibition and the prohibition on corporate contributions). Tex. Elec. Code § 253.041.

There are restrictions on the use of political contributions to reimburse political expenditures from personal funds. See "Reimbursement for Political Expenditures from Personal Funds," in this guide.

- 9. A candidate, officeholder, or political committee may not accept political contributions in the Capitol, the Capitol Extension, or a courthouse. "Courthouse" means any building owned by the state, a county, or a municipality, or an office or part of a building leased to the state, a county, or a municipality, in which a justice or judge sits to conduct court proceedings. Tex. Elec. Code § 253.039.
- 10. A person required to register as a lobbyist is prohibited from making or authorizing a political contribution to another candidate, officeholder, or political committee, or

making or authorizing a direct campaign expenditure, from political contributions accepted by: (1) the lobbyist as a candidate or officeholder; (2) a specific-purpose committee that supports or assists the lobbyist as a candidate or officeholder; or (3) a political committee that accepted a political contribution from (1) or (2), described above, during the two years immediately before the contribution or expenditure was made. Tex. Elec. Code § 253.006.

- 11. A person who makes a political contribution to another candidate, officeholder, or political committee, or makes a direct campaign expenditure, from political contributions accepted by the person as a candidate or officeholder is prohibited from engaging in activities that require registration as a lobbyist for two years thereafter. This does not apply to a person who does not receive compensation other than reimbursement for actual expenses to lobby on behalf of a nonprofit organization, a group of low-income individuals, or a group of individuals with disabilities. Tex. Elec. Code § 253.007.
- 12. A registered lobbyist, or a person on behalf of the lobbyist and with the lobbyist's consent or ratification, is prohibited from making a reportable lobby expenditure from a political contribution accepted by: (1) the lobbyist as a candidate or officeholder; (2) a specific-purpose committee that supports or assists the lobbyist as a candidate or officeholder; or (3) a political committee that accepted a political contribution from (1) or (2), described above, during the two years immediately before the lobbyist made or authorized the expenditure. Tex. Gov't Code § 305.029.
- 13. Federal law generally prohibits the acceptance of contributions from foreign sources. Contact the Federal Election Commission for more detailed information.



# 4

# **POLITICAL ADVERTISING**

# POLITICAL ADVERTISING What You Need to Know



The Texas Election Code requires certain disclosures and notices on political advertising. The law also prohibits certain types of misrepresentation in political advertising and campaign communications. This brochure explains what you need to know to insure that your political advertising and campaign communications comply with the law.

If you are not sure what the law requires, do the cautious thing. Use the political advertising disclosure statement whenever you think it might be necessary, and do not use any possibly misleading information in political advertising or a campaign communication. If you are using political advertising or campaign communications from a prior campaign, you should check to see if the law has changed since that campaign.

Candidates for federal office should check with the Federal Election Commission at (800) 424-9530 for information on federal political advertising laws.

**NOTICE:** This guide is intended only as a general overview of the disclosure statements that must appear on political advertising as required under <u>Chapter 255 of the Election Code</u>, which is distinct from political reporting requirements under <u>Chapter 254 of the Election Code</u>.

Texas Ethics Commission P.O. Box 12070 Austin, Texas 78711-2070

#### (512) 463-5800

TDD (800) 735-2989

Visit us at www.ethics.state.tx.us.

Revised July 16, 2019

# **REQUIRED DISCLOSURE ON POLITICAL ADVERTISING**

# I. What Is Political Advertising?

The disclosure statement and notice requirements discussed in this section apply to "political advertising." In the law, "political advertising" is a specifically defined term. Do not confuse this special term with your own common-sense understanding of advertising.

To figure out if a communication is political advertising, you must look at what it says and where it appears. If a communication fits in one of the categories listed in Part A (below) and if it fits in one of the categories listed in Part B (below), it is political advertising.

# Part A. What Does It Say?

- 1. Political advertising includes communications supporting or opposing a candidate for nomination or election to either a public office or an office of a political party (including county and precinct chairs).
- 2. Political advertising includes communications supporting or opposing an officeholder, a political party, or a measure (a ballot proposition).

# Part B. Where Does It Appear?

- 1. Political advertising includes communications that appear in pamphlets, circulars, fliers, billboards or other signs, bumper stickers, or similar forms of written communication.
- 2. Political advertising includes communications that are published in newspapers, magazines, or other periodicals in return for consideration.
- 3. Political advertising includes communications that are broadcast by radio or television in return for consideration.
- 4. Political advertising includes communications that appear on an Internet website.

# II. When Is a Disclosure Statement Required?

The law provides that political advertising that contains express advocacy is required to include a disclosure statement. The person who causes the political advertising to be published, distributed, or broadcast is responsible for including the disclosure statement.

The law does not define the term "express advocacy." However, the law does provide that political advertising is deemed to contain express advocacy if it is authorized by a candidate, an agent of a candidate, or a political committee filing campaign finance reports. Therefore, a disclosure statement is required any time a candidate, a candidate's agent, or a political committee authorizes political advertising.

The precise language of political advertising authorized by someone other than a candidate, the candidate's agent, or a political committee will determine if the advertising contains express advocacy and is therefore required to include a disclosure statement. Generally, the question is whether the communication expressly advocates the election or defeat of an identified candidate, or expressly advocates the passage or defeat of a measure, such as a bond election. The inclusion of words such as "vote for," "elect," "support," "defeat," "reject," or "Smith for Senate" would clearly constitute express advocacy, but express advocacy is not limited to communications that use those words. Similar phrases, such as "Cast your ballot for X," would also constitute express advocacy or its functional equivalent "if the ad is susceptible to no reasonable interpretation other than as an appeal to vote for or against a specific candidate." FEC v. Wis. Right to Life, Inc., 551 U.S. 449 (2007). It is a question of fact whether a particular communication constitutes express advocacy. If you are not sure whether political advertising contains express advocacy, do the cautious thing and include the disclosure statement. That way, there is no need to worry about whether you have violated the law.

Remember: The concept of "express advocacy" is relevant in determining whether political advertising is required to include a disclosure statement. However, the political advertising laws governing the right-of-way notice, misrepresentation, and use of public funds by political subdivisions will apply to political advertising regardless of whether the advertising contains express advocacy.

# **III. What Should the Disclosure Statement Say?**

A disclosure statement must include the following:

- 1. the words "political advertising" or a recognizable abbreviation such as "pol. adv."; and
- 2. the full name of one of the following: (a) the person who paid for the political advertising; (b) the political committee authorizing the political advertising; or (c) the candidate or specific-purpose committee supporting the candidate, if the political advertising is authorized by the candidate.

The disclosure statement must appear on the face of the political advertising or be clearly spoken if the political advertising is audio only and does not include written text.

The advertising should not be attributed to entities such as "Committee to Elect John Doe" unless a specific-purpose committee named "Committee to Elect John Doe" has filed a campaign treasurer appointment with the Ethics Commission or a local filing authority.

# **IV. Are There Any Exceptions to the Disclosure Statement Requirement?**

The following types of political advertising do not need the disclosure statement:

- 1. t-shirts, balloons, buttons, emery boards, hats, lapel stickers, small magnets, pencils, pens, pins, wooden nickels, candy wrappers, and similar materials;
- 2. invitations or tickets to political fundraising events or to events held to establish support for a candidate or officeholder;

- 3. an envelope that is used to transmit political advertising, provided that the political advertising in the envelope includes the disclosure statement;
- 4. circulars or fliers that cost in the aggregate less than \$500 to publish and distribute;
- 5. political advertising printed on letterhead stationery, if the letterhead includes the name of one of the following: (a) the person who paid for the advertising, (b) the political committee authorizing the advertising, or, (c) the candidate or specific-purpose committee supporting the candidate, if the political advertising is authorized by the candidate. (Note: There is also an exception for holiday greeting cards sent by an officeholder, provided that the officeholder's name and address appear on the card or the envelope.)
- postings or re-postings on an Internet website if the person posting or re-posting is not an officeholder, candidate, or political committee and did not make an expenditure exceeding \$100 in a reporting period for political advertising beyond the basic cost of hardware messaging software and bandwidth;
- 7. an Internet social media profile webpage of a candidate or officeholder, if the webpage clearly and conspicuously displays the full name of the candidate or officeholder; and
- 8. postings or re-postings on an Internet website if the advertising is posted with a link to a publicly viewable Internet webpage that either contains the disclosure statement or is an Internet social media profile webpage of a candidate or officeholder that clearly and conspicuously displays the candidate's or officeholder's full name.

# V. What Should I Do If I Discover That My Political Advertising Does Not Contain a Disclosure Statement?

The law prohibits a person from using, causing or permitting to be used, or continuing to use political advertising containing express advocacy if the person knows it does not include the disclosure statement. A person is presumed to know that the use is prohibited if the Texas Ethics Commission notifies the person in writing that the use is prohibited. If you receive notice from the Texas Ethics Commission that your political advertising does not comply with the law, you should stop using it immediately.

If you learn that a political advertising sign designed to be seen from the road does not contain a disclosure statement or contains an inaccurate disclosure statement, you should make a good faith attempt to remove or correct those signs that have been distributed. You are not required to attempt to recover other types of political advertising that have been distributed with a missing or inaccurate disclosure statement.

# VI. The Fair Campaign Practices Act.

The <u>Fair Campaign Practices Act</u> sets out basic rules of decency, honesty, and fair play to be followed by candidates and political committees during a campaign. A candidate or political committee may choose to subscribe to the voluntary code by signing a copy of the code and filing it with the authority with whom the candidate or committee is required to file its campaign

treasurer appointment. A person subscribing to the code may indicate that fact on political advertising by including the following or a substantially similar statement:

# (Name of the candidate or political committee, as appropriate) subscribes to the Code of Fair Campaign Practices.

### VII. Special Notice to Political Subdivisions and School Districts.

You may not use public funds or resources for political advertising. Please see our "Publications and Guides" section of our website for more information.

# **ROAD SIGNS**

### I. When Is the "Right-Of-Way" Notice Required?

All written political advertising that is meant to be seen from a road must carry a "right-of-way" notice. It is a criminal offense to omit the "right-of-way" notice in the following circumstances:

- 1. if you enter into a contract or agreement to print or make written political advertising meant to be seen from a road; or
- 2. if you instruct another person to place the written political advertising meant to be seen from a road.

#### II. What Should the "Right-Of-Way" Notice Say?

Section 259.001 of the Texas Election Code prescribes the exact language of the notice:

# NOTICE: IT IS A VIOLATION OF STATE LAW (CHAPTERS 392 AND 393, TRANSPORTATION CODE) TO PLACE THIS SIGN IN THE RIGHT-OF-WAY OF A HIGHWAY.

#### III. Do Yard Signs Have to Have the "Right-Of-Way" Notice?

Yes. The "right-of-way" notice requirement applies to signs meant to be seen from any road. The notice requirement assures that a person responsible for placing signs is aware of the restriction on placing the sign in the right-of-way of a highway.

#### **IV. What About Bumper Stickers?**

Bumper stickers do not need the "right-of-way" notice. They do, however, need a political advertising disclosure statement.

#### V. Where May I Place My Signs and How Long May Signs Be Posted?

For information about exactly where you may or may not place signs, or for information regarding the length of time your signs may be posted, check with your city or county government or your homeowner's association. The Texas Ethics Commission does not have

jurisdiction over matters involving the location of signs, and the length of time that they may be posted.

# MISREPRESENTATION

### I. Are There Restrictions on the Contents of Political Advertising?

Political advertising and campaign communications may not misrepresent a person's identity or official title, nor may they misrepresent the true source of the advertising or communication. The election law does not address other types of misrepresentation in political advertising or campaign communications.

Note that the misrepresentation rules apply to both political advertising and campaign communications. "Campaign communication" is a broader term than "political advertising."

A "campaign communication" means "a written or oral communication relating to a campaign for nomination or election to public office or office of a political party or to a campaign on a measure."

#### II. Misrepresentation of Office Title.

A candidate may not represent that he or she holds an office that he or she does not hold at the time of the representation. If you are not the incumbent in the office you are seeking, you must make it clear that you are seeking election rather than reelection by using the word "for" to clarify that you don't hold that office. The word "for" must be at least one-half the type size as the name of the office and should appear immediately before the name of the office. For example, a non-incumbent may use the following formats:

Vote John Doe for Attorney General John Doe For Attorney General

A non-incumbent may not be allowed to use the following verbiage:

Elect John Doe Attorney General John Doe Attorney General

#### **III.** Misrepresentation of Identity or Source.

A person violates the law if, with intent to injure a candidate or influence the result of an election, the person misrepresents the source of political advertising or a campaign communication or if the person misrepresents his or her own identity or the identity of his or her agent in political advertising or in a campaign communication. (If someone else is doing something for you, that person is your agent.) For example, you may not take out an ad in favor of your opponent that purports to be sponsored by a notoriously unpopular group.

#### IV. Use of State Seal.

Only current officeholders may use the state seal in political advertising.

### V. Criminal Offenses.

Be aware that many violations of the Election Code are criminal offenses. For example, unlawfully using public funds for political advertising can be a Class A misdemeanor. So can misrepresenting one's identity or office title in political advertising. For more details on these offenses and political advertising in general, see <u>Chapter 255 of the Election Code</u>.



# 5

# CHAPTER 258, ELECTION CODE FAIR CAMPAIGN PRACTICES ACT

# TEXAS ETHICS COMMISSION CHAPTER 258, ELECTION CODE FAIR CAMPAIGN PRACTICES



Effective September 1, 1997 (Revised 9/1/2023)

Texas Ethics Commission, P.O. Box 12070, Austin, Texas 78711 www.ethics.state.tx.us (512) 463-5800 • TDD (800) 735-2989 Promoting Public Confidence in Government

# **CHAPTER 258, ELECTION CODE**

# FAIR CAMPAIGN PRACTICES

# TABLE OF CONTENTS

| Chapter 258. Fair Campaign Practices                | 1 |
|---|---|
| Sec. 258.001. Short Title                           |   |
| Sec. 258.002. Purpose                               | 1 |
| Sec. 258.003. Delivery Of Copy Of Code              | 1 |
| Sec. 258.004. Text Of Code                          |   |
| Sec. 258.005. Forms                                 | 2 |
| Sec. 258.006. Acceptance And Preservation Of Copies | 2 |
| Sec. 258.007. Subscription To Code Voluntary        | 2 |
| Sec. 258.008. Indication On Political Advertising   | 3 |
| Sec. 258.009. Civil Cause Of Action                 | 3 |
|   |   |

# **ELECTION CODE**

# TITLE 15. REGULATING POLITICAL FUNDS AND CAMPAIGNS

# CHAPTER 258. FAIR CAMPAIGN PRACTICES

Sec. 258.001. SHORT TITLE. This chapter may be cited as the Fair Campaign Practices Act.

#### Sec. 258.002. PURPOSE.

(a) The purpose of this chapter is to encourage every candidate and political committee to subscribe to the Code of Fair Campaign Practices.

(b) It is the intent of the legislature that every candidate and political committee that subscribes to the Code of Fair Campaign Practices will follow the basic principles of decency, honesty, and fair play to encourage healthy competition and open discussion of issues and candidate qualifications and to discourage practices that cloud the issues or unfairly attack opponents.

### Sec. 258.003. DELIVERY OF COPY OF CODE.

(a) When a candidate or political committee files its campaign treasurer appointment, the authority with whom the appointment is filed shall give the candidate or political committee a blank form of the Code of Fair Campaign Practices and a copy of this chapter.

(b) The authority shall inform each candidate or political committee that the candidate or committee may subscribe to and file the code with the authority and that subscription to the code is voluntary.

Sec. 258.004. TEXT OF CODE. The Code of Fair Campaign Practices reads as follows:

# CODE OF FAIR CAMPAIGN PRACTICES

There are basic principles of decency, honesty, and fair play that every candidate and political committee in this state has a moral obligation to observe and uphold, in order that, after vigorously contested but fairly conducted campaigns, our citizens may exercise their constitutional rights to a free and untrammeled choice and the will of the people may be fully and clearly expressed on the issues.

# THEREFORE:

(1) I will conduct the campaign openly and publicly and limit attacks on my opponent to legitimate challenges to my opponent's record and stated positions on issues.

(2) I will not use or permit the use of character defamation, whispering campaigns, libel, slander, or scurrilous attacks on any candidate or the candidate's personal or family life.

(3) I will not use or permit any appeal to negative prejudice based on race, sex, religion, or national origin.

(4) I will not use campaign material of any sort that misrepresents, distorts, or otherwise falsifies the facts, nor will I use malicious or unfounded accusations that aim at creating or exploiting doubts, without justification, as to the personal integrity or patriotism of my opponent.

(5) I will not undertake or condone any dishonest or unethical practice that tends to corrupt or undermine our system of free elections or that hampers or prevents the full and free expression of the will of the voters, including any activity aimed at intimidating voters or discouraging them from voting.

(6) I will defend and uphold the right of every qualified voter to full and equal participation in the electoral process, and will not engage in any activity aimed at intimidating voters or discouraging them from voting.

(7) I will immediately and publicly repudiate methods and tactics that may come from others that I have pledged not to use or condone. I shall take firm action against any subordinate who violates any provision of this code or the laws governing elections.

I, the undersigned, candidate for election to public office in the State of Texas or campaign treasurer of a political committee, hereby voluntarily endorse, subscribe to, and solemnly pledge myself to conduct the campaign in accordance with the above principles and practices.

# $VOID - COPY ONLY - VOID^1$

Date

Signature

**Sec. 258.005. FORMS**. The commission shall print copies of the Code of Fair Campaign Practices and shall supply the forms to the authorities with whom copies of the code may be filed in quantities and at times requested by the authorities.

# Sec. 258.006. ACCEPTANCE AND PRESERVATION OF COPIES.

(a) An authority with whom a campaign treasurer appointment is filed shall accept each completed copy of the code submitted to the authority that is properly subscribed to by a candidate or the campaign treasurer of a political committee.

(b) Each copy of the code accepted under this section shall be preserved by the authority with whom it is filed for the period prescribed for the filer's campaign treasurer appointment.

**Sec. 258.007. SUBSCRIPTION TO CODE VOLUNTARY**. The subscription to the Code of Fair Campaign Practices by a candidate or a political committee is voluntary.

<sup>&</sup>lt;sup>1</sup> This document is a copy of chapter 258, Election Code. To subscribe to the Code of Fair Campaign Practices, a candidate or campaign treasurer of a political committee must submit Texas Ethics Commission FORM CFCP, not a signed copy of this document.

**Sec. 258.008. INDICATION ON POLITICAL ADVERTISING**. A candidate or a political committee that has filed a copy of the Code of Fair Campaign Practices may so indicate on political advertising in a form to be determined by the commission.

Sec. 258.009. CIVIL CAUSE OF ACTION. This chapter does not create a civil cause of action for recovery of damages or for enforcement of this chapter.



# 6

# APPOINTMENT OF A CAMPAIGN TREASURER BY A CANDIDATE

# **TEXAS ETHICS COMMISSION**

# APPOINTMENT OF A CAMPAIGN TREASURER BY A CANDIDATE

FORM CTA--INSTRUCTION GUIDE



Revised January 1, 2024

Texas Ethics Commission, P.O. Box 12070, Austin, Texas 78711 www.ethics.state.tx.us (512) 463-5800 • TDD (800) 735-2989 Promoting Public Confidence in Government

# FORM CTA—INSTRUCTION GUIDE

# TABLE OF CONTENTS

| GENERAL INSTRUCTIONS                            | 1 |
|---|---|
| DUTIES OF A CANDIDATE OR OFFICEHOLDER           | 1 |
| QUALIFICATIONS OF CAMPAIGN TREASURER            | 1 |
| DUTIES OF A CAMPAIGN TREASURER                  | 1 |
| REQUIREMENT TO FILE BEFORE BEGINNING A CAMPAIGN | 1 |
| WHERE TO FILE A CAMPAIGN TREASURER APPOINTMENT  | 2 |
| FILING WITH A DIFFERENT AUTHORITY               | 3 |
| FORMING A POLITICAL COMMITTEE                   | 4 |
| CHANGING A CAMPAIGN TREASURER                   | 4 |
| AMENDING A CAMPAIGN TREASURER APPOINTMENT       |   |
| REPORTING REQUIREMENT FOR CERTAIN OFFICEHOLDERS | 4 |
| TERMINATING A CAMPAIGN TREASURER APPOINTMENT    |   |
| FILING A FINAL REPORT                           | 5 |
| ELECTRONIC FILING                               | 5 |
| GUIDES  | 5 |
| SPECIFIC INSTRUCTIONS                           | 5 |
| PAGE 1  | 5 |
| PAGE 2  | 7 |
|   |   |

# APPOINTMENT OF A CAMPAIGN TREASURER BY A CANDIDATE

# **GENERAL INSTRUCTIONS**

These instructions are for the APPOINTMENT OF A CAMPAIGN TREASURER BY A CANDIDATE (Form CTA). Use Form CTA only for appointing your campaign treasurer. Use the AMENDMENT (Form ACTA) for changing information previously reported on Form CTA and for renewing your choice to report under the modified schedule. Note: Candidates for most judicial offices use Form JCTA to file a campaign treasurer appointment.

# DUTIES OF A CANDIDATE OR OFFICEHOLDER

As a candidate or officeholder, you alone, not the campaign treasurer, are responsible for filing this form and all candidate/officeholder reports of contributions, expenditures, and loans. Failing to file a report on time or filing an incomplete report may subject you to criminal or civil penalties.

# **QUALIFICATIONS OF CAMPAIGN TREASURER**

A person is ineligible for appointment as a campaign treasurer if the person is the campaign treasurer of a political committee that has outstanding filing obligations (including outstanding penalties). This prohibition does not apply if the committee in connection with which the ineligibility arose has not accepted more than \$5,000 in political contributions or made more than \$5,000 in political expenditures in any semiannual reporting period. A person who violates this prohibition is liable for a civil penalty not to exceed three times the amount of political contributions accepted or political expenditures made in violation of this provision. Note: A candidate may appoint himself or herself as his or her own campaign treasurer.

# **DUTIES OF A CAMPAIGN TREASURER**

State law does not impose any obligations on a candidate's campaign treasurer.

# **REQUIREMENT TO FILE BEFORE BEGINNING A CAMPAIGN**

If you plan to run for a public office in Texas (except for a federal office), you must file this form when you become a candidate even if you do not intend to accept campaign contributions or make campaign expenditures. A "candidate" is a person who knowingly and willingly takes affirmative action for the purpose of gaining nomination or election to public office or for the purpose of satisfying financial obligations incurred by the person in connection with the campaign for nomination or election. Examples of affirmative action include:

- (A) the filing of a campaign treasurer appointment, except that the filing does not constitute candidacy or an announcement of candidacy for purposes of the automatic resignation provisions of Article XVI, Section 65, or Article XI, Section 11, of the Texas Constitution;
- (B) the filing of an application for a place on the ballot;

- (C) the filing of an application for nomination by convention;
- (D) the filing of a declaration of intent to become an independent candidate or a declaration of write-in candidacy;
- (E) the making of a public announcement of a definite intent to run for public office in a particular election, regardless of whether the specific office is mentioned in the announcement;
- (F) before a public announcement of intent, the making of a statement of definite intent to run for public office and the soliciting of support by letter or other mode of communication;
- (G) the soliciting or accepting of a campaign contribution or the making of a campaign expenditure; and
- (H) the seeking of the nomination of an executive committee of a political party to fill a vacancy.

Additionally, the law provides that you must file this form before you may accept a campaign contribution or make or authorize a campaign expenditure, including an expenditure from your personal funds. A filing fee paid to a filing authority to qualify for a place on a ballot is a campaign expenditure that may not be made before filing a campaign treasurer appointment form with the proper filing authority.

If you are an officeholder, you may make officeholder expenditures and accept officeholder contributions without having a campaign treasurer appointment on file. If you do not have a campaign treasurer appointment on file and you wish to accept *campaign* contributions or make *campaign* expenditures in connection with your office or for a different office, you must file this form before doing so. In such a case, a sworn report of contributions, expenditures, and loans will be due no later than the 15th day after filing this form.

# WHERE TO FILE A CAMPAIGN TREASURER APPOINTMENT

The appropriate filing authority depends on the office sought or held.

- **a. Texas Ethics Commission.** The Texas Ethics Commission (Commission) is the appropriate filing authority for the Secretary of State and for candidates for or holders of the following offices:
  - Governor, Lieutenant Governor, Attorney General, Comptroller, Treasurer, Land Commissioner, Agriculture Commissioner, Railroad Commissioner.
  - State Senator or State Representative.
  - Supreme Court Justice, Court of Criminal Appeals Judge, and Court of Appeals Judge.\*

- State Board of Education.
- A multi-county district judge\* or multi-county district attorney.
- A single-county district judge.\*
- An office of a political subdivision other than a county if the political subdivision includes areas in more than one county and if the governing body of the political subdivision has not been formed.
- A chair of the state executive committee of a political party with a nominee on the ballot in the most recent gubernatorial election.
- A county chair of a political party with a nominee on the ballot in the most recent gubernatorial election if the county has a population of 350,000 or more.

\* Judicial candidates use FORM JCTA to appoint a campaign treasurer.

- **b.** County Clerk. The county clerk (or the county elections administrator or tax assessor, as applicable) is the appropriate local filing authority for a candidate for:
  - A county office.
  - A precinct office.
  - A district office (except for multi-county district offices).
  - An office of a political subdivision other than a county if the political subdivision is within the boundaries of a single county and if the governing body of the political subdivision has not been formed.
- **c.** Local Filing Authority. If a candidate is seeking an office of a political subdivision other than a county, the appropriate filing authority is the *clerk or secretary of the governing body* of the political subdivision. If the political subdivision has no clerk or secretary, the appropriate filing authority is the governing body's presiding officer. Basically, any political subdivision that is authorized by the laws of this state to hold an election is considered a local filing authority. Examples are cities, school districts, and municipal utility districts.

# FILING WITH A DIFFERENT AUTHORITY

If you have a campaign treasurer appointment on file with one authority, and you wish to accept campaign contributions or make or authorize campaign expenditures in connection with another office that would require filing with a different authority, you must file a new campaign treasurer appointment <u>and</u> a copy of your old campaign treasurer appointment (certified by the old authority) with the new filing authority before beginning your campaign. You should also provide written notice to the original filing authority that your future reports will be filed with another authority; use Form CTA-T for this purpose.

# FORMING A POLITICAL COMMITTEE

As a candidate, you must file an APPOINTMENT OF A CAMPAIGN TREASURER BY A CANDIDATE (FORM CTA). You may also form a specific-purpose committee to support your candidacy. Remember that filing a campaign treasurer appointment for a political committee does not eliminate the requirement that a candidate file his or her own campaign treasurer appointment (FORM CTA) and the related reports.

**NOTE:** See the Campaign Finance Guide for Political Committees for further information about specific-purpose committees.

# CHANGING A CAMPAIGN TREASURER

If you wish to change your campaign treasurer, simply file an amended campaign treasurer appointment (FORM ACTA). This will automatically terminate the outgoing campaign treasurer appointment.

# AMENDING A CAMPAIGN TREASURER APPOINTMENT

If *any* of the information reported on the campaign treasurer appointment (FORM CTA) changes, file an AMENDMENT: APPOINTMENT OF A CAMPAIGN TREASURER BY A CANDIDATE (FORM ACTA) to report the change.

# **REPORTING REQUIREMENT FOR CERTAIN OFFICEHOLDERS**

If you are an officeholder who appoints a campaign treasurer after a period of not having one, you must file a report of contributions, expenditures, and loans no later than the 15th day after your appointment is effective. This requirement is not applicable if you are a candidate or an officeholder who is merely changing campaign treasurers.

# TERMINATING A CAMPAIGN TREASURER APPOINTMENT

You may terminate your campaign treasurer appointment at any time by:

- 1) filing a campaign treasurer appointment for a successor campaign treasurer, or
- 2) filing a final report.

Remember that you may not accept any campaign contributions or make or authorize any campaign expenditures without a campaign treasurer appointment on file. You may, however, accept officeholder contributions and make or authorize officeholder expenditures.

If your campaign treasurer quits, he or she must give written notice to both you and your filing authority. The termination will be effective on the date you receive the notice or on the date your filing authority receives the notice, whichever is later.

# FILING A FINAL REPORT

For filing purposes, you are a "candidate" as long as you have an appointment of campaign treasurer on file. If you do not expect to accept any further campaign contributions or to make any further campaign expenditures, you may file a final report of contributions and expenditures. A final report terminates your appointment of campaign treasurer and relieves you of the obligation of filing further reports as a candidate. If you have surplus funds, or if you retain assets purchased with political funds, you will be required to file annual reports. (*See instructions for FORM C/OH - UC.*) If you are an officeholder at the time of filing a final report, you may be required to file semiannual reports of contributions, expenditures, and loans as an officeholder.

If you do not have an appointment of campaign treasurer on file, you may not accept *campaign* contributions or make *campaign* expenditures. A payment on a campaign debt is a campaign expenditure. An officeholder who does not have an appointment of campaign treasurer on file may accept *officeholder* contributions and make *officeholder* expenditures.

To file a final report, you must complete the CANDIDATE/OFFICEHOLDER CAMPAIGN FINANCE REPORT (FORM C/OH), check the "final" box on Page 1, Section 9, and complete and attach the DESIGNATION OF FINAL REPORT (FORM C/OH-FR).

# **ELECTRONIC FILING**

All persons filing campaign finance reports with the Commission are required to file those reports electronically unless the person is entitled to claim an exemption. Please check the Commission's website at *http://www.ethics.state.tx.us* for information about exemptions from the electronic filing requirements.

# GUIDES

All candidates should review the applicable Commission's campaign finance guide. Guides are available on the Commission's website at *http://www.ethics.state.tx.us*.

# **SPECIFIC INSTRUCTIONS**

Each numbered item in these instructions corresponds to the same numbered item on the form.

# <u>PAGE 1</u>

- 1. TOTAL PAGES FILED: After you have completed the form, enter the total number of pages of this form and any additional pages. A "page" is one side of a two-sided form. If you are not using a two-sided form, a "page" is a single sheet.
- **2. CANDIDATE NAME**: Enter your full name, including nicknames and suffixes (e.g., Sr., Jr., III), if applicable. Enter your name in the same way on Page 2, Section 11, of this form.
- **3. CANDIDATE MAILING ADDRESS**: Enter your complete mailing address, including zip code. This information will allow your filing authority to correspond with you. If this information changes, please notify your filing authority immediately.

- **4. CANDIDATE PHONE**: Enter your phone number, including the area code and extension, if applicable.
- **5. OFFICE HELD**: If you are an officeholder, please enter the office you currently hold. Include the district, precinct, or other designation for the office, if applicable.
- 6. OFFICE SOUGHT: If you are a candidate, please enter the office you seek, if known. Include the district, precinct, or other designation for the office, if applicable.
- 7. CAMPAIGN TREASURER NAME: Enter the full name of your campaign treasurer, including nicknames and suffixes (e.g., Sr., Jr., III), if applicable.
- 8. CAMPAIGN TREASURER STREET ADDRESS: Enter the complete street address of your campaign treasurer, including the zip code. You may enter either the treasurer's business or residential street address. If you are your own treasurer, you may enter either your business or residential street address.
- **9. CAMPAIGN TREASURER PHONE**: Enter the phone number of your campaign treasurer, including the area code and extension, if applicable.
- **10. CANDIDATE SIGNATURE**: Enter your signature after reading the summary. Your signature here indicates that you have read the following summary of the nepotism law; that you are aware of your responsibility to file timely reports; and that you are aware of the restrictions on contributions from corporations and labor organizations.
  - The Texas nepotism law (Government Code, chapter 573) imposes certain restrictions on both officeholders and candidates. You should consult the statute in regard to the restrictions applicable to officeholders.
  - A candidate may not take an affirmative action to influence an employee of the office to which the candidate seeks election in regard to the appointment, confirmation, employment or employment conditions of an individual who is related to the candidate within a prohibited degree.
  - A candidate for a multi-member governmental body may not take an affirmative action to influence an officer or employee of the governmental body to which the candidate seeks election in regard to the appointment, confirmation, or employment of an individual related to the candidate in a prohibited degree.
  - Two people are related within a prohibited degree if they are related within the third degree by consanguinity (blood) or the second degree by affinity (marriage). The degree of consanguinity is determined by the number of generations that separate them. If neither is descended from the other, the degree of consanguinity is determined by adding the number of generations that each is separated from a common ancestor. Examples: (1) first degree parent to child; (2) second degree grandparent to grandchild; or brother to sister; (3) third degree great-grandparent to great-grandchild; or aunt to niece who is child of individual's brother or sister. A husband and wife are related in the first degree by affinity. A wife has the same degree of relationship by affinity to her husband's relatives as her husband has by

consanguinity. For example, a wife is related to her husband's grandmother in the second degree by affinity.

# PAGE 2

- **11. CANDIDATE NAME**: Enter your name as you did on Page 1.
- **12. MODIFIED REPORTING DECLARATION**: Sign this option if you wish to report under the modified reporting schedule.

The modified reporting option is not available for candidates for the office of state chair of a political party and candidates for county chair of a political party.

To the left of your signature, enter the year of the election or election cycle to which your selection of modified reporting applies.

Your selection of modified reporting is valid for an entire election cycle. For example, if you choose modified reporting before a primary election, your selection remains in effect for any runoff and for the general election and any related runoff. You must make this selection at least 30 days before the first election to which your selection applies.

An opposed candidate in an election is eligible to report under the modified reporting schedule if he or she does not intend to accept more than \$1,080 in political contributions or make more than \$1,080 in political expenditures in connection with an election. The amount of a filing fee paid to qualify for a place on the ballot does not count against the \$1,080 expenditure limit. An opposed candidate who reports under the modified schedule is not required to file pre-election reports (due 30 days and 8 days before an election) or runoff reports (due 8 days before a runoff). (Note: An *unopposed* candidate is not required to file pre-election reports in the first place.) The obligations to file semiannual reports, special pre-election reports (formerly known as telegram reports), or special session reports, if applicable, are not affected by selecting the modified schedule.

The \$1,080 maximums apply to each election within the cycle. In other words, you are limited to \$1,080 in contributions and expenditures in connection with the primary, an additional \$1,080 in contributions and expenditures in connection with the general election, and an additional \$1,080 in contributions and expenditures in connection with a runoff.

**EXCEEDING \$1,080 IN CONTRIBUTIONS OR EXPENDITURES.** If you exceed \$1,080 in contributions or expenditures in connection with an election, you must file according to the regular filing schedule. In other words, you must file pre-election reports and a runoff report, if you are in a runoff.

If you exceed either of the \$1,080 limits *after the 30th day before the election*, you must file a sworn report of contributions and expenditures within 48 hours after exceeding the limit. After that, you must file any pre-election reports or runoff reports that are due under the regular filing schedule.

Your selection is not valid for other elections or election cycles. Use the AMENDMENT (FORM ACTA) to renew your option to file under the modified schedule for a different election year or election cycle.

For more information, see the Commission's campaign finance guide that applies to you.

# AMENDMENT: APPOINTMENT OF A CAMPAIGN TREASURER BY A CANDIDATE

7



# **TEXAS ETHICS COMMISSION**

# AMENDMENT: APPOINTMENT OF A CAMPAIGN TREASURER BY A CANDIDATE

# FORM ACTA-INSTRUCTION GUIDE



**Revised January 1, 2024** 

Texas Ethics Commission, P.O. Box 12070, Austin, Texas 78711
<a href="http://www.ethics.state.tx.us">www.ethics.state.tx.us</a>
<a href="http://www.ethics.state.tx.us">www.ethics.state.tx.us</a>
<a href="http://www.ethics.state.tx.us">www.ethics.state.tx.us</a>
<a href="http://www.ethics.state.tx.us">(512) 463-5800</a>
<a href="http://www.ethics.state.tx.us">TDD (800) 735-2989</a>
<a href="http://www.ethics.state.tx.us">Promoting Public Confidence in Government</a>

# FORM ACTA-AMENDMENT: APPOINTMENT OF A CAMPAIGN TREASURER BY A CANDIDATE

### **GENERAL INSTRUCTIONS**

These instructions are for the AMENDMENT: APPOINTMENT OF A CAMPAIGN TREASURER BY A CANDIDATE (Form ACTA). Use this form for changing information previously reported on Form CTA and for renewing your choice to report under the modified schedule. The information you enter on this form will replace the information from your previous APPOINTMENT OF A CAMPAIGN TREASURER BY A CANDIDATE (Form (CTA).

If any of the information required to be reported on your CAMPAIGN TREASURER APPOINTMENT changes, you should file an amendment. Use the AMENDMENT form (Form ACTA) to report the changes. Do not use the APPOINTMENT form (Form CTA).

You must also use the AMENDMENT form to renew your option to file under the modified schedule.

Except for your name at the top of the form (and your filer account number, if you file with the Texas Ethics Commission (Commission)), enter only the information that is <u>different</u> from what is on your current campaign treasurer appointment. Do not repeat information that has not changed. The "NEW" boxes emphasize that the information entered on this form should only be information that is different from what was previously reported. Any information entered in a space with a "NEW" box will replace the existing information.

# SPECIFIC INSTRUCTIONS

Each numbered item in these instructions corresponds to the same numbered item on the form.

# PAGE 1

- 1. CANDIDATE NAME: Enter your name as it is on your current campaign treasurer appointment. Enter your name in the same way on Page 2, Section 13, of this form. If you are reporting a name change, enter your new name under Section 4.
- 2. FILER ID #: If you are filing with the Commission, you were assigned a filer account number when you filed your initial campaign treasurer appointment. You should have received a letter acknowledging receipt of the form and informing you of your account number. Enter this number wherever you see "FILER ID #." If you do not file with the Ethics Commission, you are not required to enter an account number.
- **3. TOTAL PAGES FILED**: After you have completed the form, enter the total number of pages of this form and any additional pages. A "page" is one side of a two-sided form. If you are not using a two-sided form, a "page" is a single sheet.

- **4. CANDIDATE NAME**: Complete this section only if your name has *changed*. If your name has changed, enter your complete new name, including nicknames and suffixes (e.g., Sr., Jr., III) if applicable.
- **5. CANDIDATE MAILING ADDRESS**: Complete this section only if your mailing address has *changed*. If your mailing address has changed, enter your complete new address, including zip code. This information will allow your filing authority to correspond with you.
- 6. CANDIDATE PHONE: Complete this section only if your phone number has *changed*. If your phone number has changed, enter your new phone number, including the area code and extension, if applicable.
- **7. OFFICE HELD**: If you are an officeholder, complete this section only if your office has *changed*. If your office has changed, please enter the new office held. Include the district, precinct, or other designation for the office, if applicable.
- **8. OFFICE SOUGHT**: If you are a candidate, complete this section only if the office you seek has *changed*. If the office has changed, please enter the office you now seek, if known. Include the district, precinct, or other designation for the office, if applicable.

**Note:** Changing the office you are seeking may require you to file your reports with a different filing authority. See the Campaign Finance Guide for further information on filing with a different authority.

**9.** CAMPAIGN TREASURER NAME: Complete this section only if your campaign treasurer has *changed*. If your campaign treasurer has changed, enter the full name of your new campaign treasurer, including nicknames and suffixes (e.g., Sr., Jr., III), if applicable.

**Qualifications of Campaign Treasurer.** A person is ineligible for appointment as a campaign treasurer if the person is the campaign treasurer of a political committee that has outstanding filing obligations (including outstanding penalties). This prohibition does not apply if the committee in connection with which the ineligibility arose has not accepted more than \$5,000 in political contributions or made more than \$5,000 in political expenditures in any semiannual reporting period. A person who violates this prohibition is liable for a civil penalty not to exceed three times the amount of political contributions accepted or political expenditures made in violation of this provision.

- **10. CAMPAIGN TREASURER STREET ADDRESS**: Complete this section only if your campaign treasurer's street address has *changed*. If your campaign treasurer's street address has changed, enter the complete new address of your campaign treasurer, including the zip code. You may enter either the treasurer's new business or residential street address. If you are your own treasurer, you may enter either your business or residential street address.
- **11. CAMPAIGN TREASURER PHONE**: Complete this section only if your campaign treasurer's phone number has *changed*. If your campaign treasurer's phone number has

changed, enter the new phone number of your campaign treasurer, including the area code and extension, if applicable.

- **12. CANDIDATE SIGNATURE**: Enter your signature after reading the summary. Your signature here indicates that you have read the following summary of the nepotism law; that you are aware of your responsibility to file timely reports; and that you are aware of the restrictions on contributions from corporations and labor organizations.
  - The Texas nepotism law (Government Code, chapter 573) imposes certain restrictions on both officeholders and candidates. You should consult the statute in regard to the restrictions applicable to officeholders.
  - A candidate may not take an affirmative action to influence an employee of the office to which the candidate seeks election in regard to the appointment, confirmation, employment or employment conditions of an individual who is related to the candidate within a prohibited degree.
  - A candidate for a multi-member governmental body may not take an affirmative action to influence an officer or employee of the governmental body to which the candidate seeks election in regard to the appointment, confirmation, or employment of an individual related to the candidate in a prohibited degree.
  - Two people are related within a prohibited degree if they are related within the third degree by consanguinity (blood) or the second degree by affinity (marriage). The *degree of consanguinity* is determined by the number of generations that separate them. If neither is descended from the other, the degree of consanguinity is determined by adding the number of generations that each is separated from a common ancestor. **Examples**: (1) first degree parent to child; (2) second degree grandparent to grandchild; or brother to sister; (3) third degree great-grandparent to great-grandparent to niece who is child of individual's brother or sister. A husband and wife are related in the first degree by affinity. A wife has the same degree of relationship by affinity to her husband's relatives as her husband has by consanguinity. For example, a wife is related to her husband's grandmother in the second degree by affinity.

**Note:** The changes you have made on this form will replace the information on your previous APPOINTMENT form (Form CTA).

# <u>PAGE 2</u>

- **13. CANDIDATE NAME**: Enter your name as you did on Page 1, Section 1.
- **14. MODIFIED REPORTING DECLARATION**: Sign this option if you wish to report under the modified reporting schedule.

The modified reporting option is not available for candidates for the office of state chair of a political party.

To the left of your signature, enter the year of the election or election cycle to which your selection of modified reporting applies.

Your selection of modified reporting is valid for an entire election cycle. For example, if you choose modified reporting before a primary election, your selection remains in effect for any runoff and for the general election and any related runoff. You must make this selection at least 30 days before the first election to which your selection applies.

An opposed candidate in an election is eligible to report under the modified reporting schedule if he or she does not intend to accept more than \$1,080 in political contributions or make more than \$1,080 in political expenditures in connection with an election. The amount of a filing fee paid to qualify for a place on the ballot does not count against the \$1,080 expenditure limit. An opposed candidate who reports under the modified schedule is not required to file pre-election reports (due 30 days and 8 days before an election) or runoff reports (due 8 days before a runoff). (Note: An *unopposed* candidate is not required to file pre-election reports in the first place.) The obligations to file semi-annual reports, special pre-election reports, or special session reports, if applicable, are not affected by selecting the modified schedule.

The \$1,080 maximums apply to each election within the cycle. In other words, you are limited to \$1,080 in contributions and expenditures in connection with the primary, an additional \$1,080 in contributions and expenditures in connection with the general election, and an additional \$1,080 in contributions and expenditures in connection with a runoff.

**Exceeding \$1,080 in contributions or expenditures.** If you exceed \$1,080 in contributions or expenditures in connection with an election, you must file according to the regular schedule. In other words, you must file pre-election reports and a runoff report, if you are in a runoff.

If you exceed either of the \$1,080 limits *after the 30th day before the election*, you must file a sworn report of contributions and expenditures within 48 hours after exceeding the limit. After that, you must file any pre-election reports or runoff reports that are due under the regular filing schedule.

Your selection is not valid for other elections or election cycles. Use another amendment form (ACTA) to renew your option to file under the modified schedule.

For more information, see the Commission's campaign finance guide that applies to you.

# CANDIDATE/ OFFICEHOLDER CAMPAIGN FINANCE REPORT

8



# **TEXAS ETHICS COMMISSION**

# CANDIDATE/OFFICEHOLDER CAMPAIGN FINANCE REPORT

FORM C/OH – INSTRUCTION GUIDE

(PAPER FILERS ONLY)

# To Report Activity Occurring on or after January 1, 2022



Revised January 1, 2022

Texas Ethics Commission, P.O. Box 12070, Austin, Texas 78711 www.ethics.state.tx.us (512) 463-5800 • TDD (800) 735-2989 Promoting Public Confidence in Government

# FORM C/OH – INSTRUCTION GUIDE

# **TABLE OF CONTENTS**

These instructions are for the CANDIDATE/OFFICEHOLDER CAMPAIGN FINANCE REPORT (Form C/OH) and all schedules that are filed with it. FORM C/OH includes a three-page cover sheet and Schedules A1, A2, B, E, F1, F2, F3, F4, G, H, I, K, and T. Candidates or officeholders filing a Final Report should also attach Form C/OH-FR. All filers must submit the cover sheet, but only the schedules on which there is information to report need to be included.

| GENERAL INSTRUCTIONS   | 3    |
|--|------|
| IMPORTANT UPDATES  | 3    |
| ELECTRONIC FILING  | 3    |
| FILLING OUT THE FORMS  |      |
| TEXAS ETHICS COMMISSION GUIDES                                   | 4    |
| PHOTOCOPIES OF FORMS   | 4    |
| FILING DATE  |      |
| FORM C/OH: CANDIDATE/OFFICEHOLDER CAMPAIGN FINANCE REPORT        | 5    |
| GENERAL INFORMATION  |      |
| COMPLETING THE COVER SHEET                                       | 7    |
| PAGE 1   | 7    |
| PAGE 2   | 13   |
| PAGE 3   | 15   |
| SCHEDULE A1: MONETARY POLITICAL CONTRIBUTIONS                    |      |
| SCHEDULE A2: NON-MONETARY (IN-KIND) POLITICAL CONTRIBUTIONS      | . 19 |
| SCHEDULE B: PLEDGED CONTRIBUTIONS                                | 21   |
| SCHEDULE E: LOANS  |      |
| SCHEDULE F1: POLITICAL EXPENDITURES FROM POLITICAL CONTRIBUTIONS |      |
| SCHEDULE F2: UNPAID INCURRED OBLIGATIONS                         | . 30 |
| SCHEDULE F3: PURCHASE OF INVESTMENTS FROM POLITICAL              |      |
| CONTRIBUTIONS  |      |
| SCHEDULE F4: EXPENDITURES MADE BY CREDIT CARD                    |      |
| SCHEDULE G: POLITICAL EXPENDITURES MADE FROM PERSONAL FUNDS      |      |
| SCHEDULE H: PAYMENT FROM POLITICAL CONTRIBUTIONS TO A BUSINESS ( |      |
| С/ОН   | . 37 |
| SCHEDULE I: NON-POLITICAL EXPENDITURES MADE FROM POLITICAL       |      |
| CONTRIBUTIONS  | 38   |
| SCHEDULE K: INTEREST, CREDITS, GAINS, REFUNDS, AND CONTRIBUTIONS |      |
| RETURNED TO FILER  | . 39 |
| SCHEDULE T: IN-KIND CONTRIBUTIONS OR POLITICAL EXPENDITURES FOR  |      |
| TRAVEL OUTSIDE OF TEXAS  |      |
| FORM C/OH-FR: DESIGNATION OF FINAL REPORT                        |      |
| GENERAL INFORMATION  |      |
| COMPLETING THE FORM  |      |
| ADDITIONAL INFORMATION REGARDING EXPENDITURES                    |      |
| EXAMPLES: REPORTING EXPENDITURES MADE BY CREDIT CARD             |      |
| EXAMPLES: PURPOSE OF EXPENDITURES                                |      |
| EXAMPLES: REPORTING EXPENDITURES FROM PERSONAL FUNDS             |      |
| EXAMPLES: REPORTING STAFF REIMBURSEMENT                          | 50   |

# **GENERAL INSTRUCTIONS**

These general instructions apply to all C/OH forms required to be filed under title 15, Texas Election Code, for activity that occurs on or after January 1, 2022. For a report that includes activity occurring before January 1, 2022, you must use the instructions applicable before calendar year 2022, which are available on the Texas Ethics Commission's website at <a href="https://www.ethics.state.tx.us/forms/coh/cohfrm.php">https://www.ethics.state.tx.us/forms/coh/cohfrm.php</a>.

# **IMPORTANT UPDATES**

### **Increased Disclosure Thresholds**

On January 1, 2020, the Texas Ethics Commission began adjusting certain reporting thresholds to account for inflation. As directed by section 571.064 of the Texas Election Code, the Commission is required to annually adjust these thresholds upward to the nearest multiple of \$10 in accordance with the percentage increase for the previous year in the Consumer Price Index for Urban Consumers published by the Bureau of Labor Statistics of the United States Department of Labor. Accordingly, one or more thresholds will generally be adjusted each year, depending upon the figures in the index.

These changes will be made effective January 1<sup>st</sup> of each calendar year; the affected numbers and corresponding new thresholds are located in 1 T.A.C. §18.31, which can be found here: <u>https://www.ethics.state.tx.us/rules/</u>. The higher itemization thresholds will be reflected on the paper forms and in these instructions, as applicable.

Please verify that you are using the correct thresholds and forms that apply to your filing. For example, if you are filing a campaign finance report or lobby activities report that is due in January of 2021, you must use the forms and instructions that are applicable to the period ending December 31, 2020.

# **Contributions Made Electronically Must Be Itemized**

Beginning on September 1, 2019, all political contributions that are made electronically and accepted by a filer during the reporting period must be itemized in the filer's campaign finance report. This change is made by House Bill 2586, adopted by the 86th Texas Legislature.

# **ELECTRONIC FILING**

All persons filing campaign finance reports with the Texas Ethics Commission (Commission) are required to file those reports electronically unless the person is eligible to claim an exemption. Please check the Commission's website at *https://www.ethics.state.tx.us* for information about exemptions from the electronic filing requirement.

# FILLING OUT THE FORMS

All reports filed on paper must be either handwritten in ink or typewritten. If you complete the report by hand, please print everything other than your signature.

If you are filing with the Commission, and you are eligible to claim an exemption to electronic filing, *you may use your own computer-generated form* if it provides for disclosure of all the information required on the Commission's form and it is *substantially identical* in paper size, color, layout, and format. A substitute form that is substantially identical to the Commission's prescribed form must be submitted for pre-approval by the Commission's executive director.

Always file the cover sheet of the campaign finance report form. You need to file only those schedules on which you have information to report.

You must keep an exact copy of each report filed and all records necessary to complete the report for at least two (2) years after the deadline for filing the report.

If you have questions, please call our office at (512) 463-5800.

# **TEXAS ETHICS COMMISSION GUIDES**

The Commission publishes a Campaign Finance Guide for each type of filer. These guides are designed to explain your responsibilities as a filer. The Commission encourages you to read the appropriate guide before you begin accepting political contributions or making or authorizing political expenditures.

# **PHOTOCOPIES OF FORMS**

You may use photocopies of Commission forms. For example, if the space provided on Schedule A1 is insufficient, you may make copies of a blank Schedule A1 form and attach more pages as needed.

# FILING DATE

For most reporting deadlines, a document is considered timely filed if it is properly addressed with postage or handling charges prepaid and bears a postmark or receipt mark of a common or contract carrier indicating a time on or before the deadline.

**Pre-Election Reports:** A report due 30 days before an election and a report due 8 days before an election must be *received* by the appropriate filing authority no later than the report due date.

If you are filing with the Commission, please address your reports and correspondence to the Texas Ethics Commission, P.O. Box 12070, Austin, Texas 78711-2070. For hand-deliveries, the Commission's street address is 201 East 14th Street, Sam Houston Building, 10th Floor, Austin, Texas 78701.

If the due date for a report falls on a Saturday, Sunday, or legal holiday, the report is due on the next regular business day.

### FORM C/OH: CANDIDATE/OFFICEHOLDER CAMPAIGN FINANCE REPORT

These instructions are for the CANDIDATE/OFFICEHOLDER CAMPAIGN FINANCE REPORT (Form C/OH). A complete report includes the Form C/OH cover sheet, and any of the following schedules on which there is information to report: A1, A2, B, E, F1, F2, F3, F4, G, H, I, K, and T. A complete Final Report must also include Form C/OH-FR.

Note: Judicial candidates and officeholders must use a different form, Form JC/OH.

#### **GENERAL INFORMATION**

Use Form C/OH for filing the following reports:

- Semiannual reports (January 15 and July 15)
- Pre-election reports (30th day before election, 8th day before election)
- Runoff report (8th day before runoff election)
- Exceeded Modified Reporting Limit report
- 15th day after officeholder campaign treasurer appointment
- Final Report

See the instructions for sections 9 and 10 of the Cover Sheet for help in deciding which reports you are required to file.

#### **OFFICEHOLDER ACTIVITY**

An officeholder may make officeholder expenditures and accept officeholder contributions without having a campaign treasurer appointment on file. However, an officeholder must have a campaign treasurer appointment on file before the officeholder may make campaign expenditures or accept campaign contributions.

#### **DUTIES OF CANDIDATE OR OFFICEHOLDER**

As a candidate or officeholder, you alone, not the campaign treasurer, are responsible for filing this form. Failing to file a report on time or filing an incomplete report may subject you to criminal or civil penalties.

#### **DUTIES OF CAMPAIGN TREASURER**

State law does not impose any reporting or record-keeping obligations on a candidate's campaign treasurer.

#### WHERE TO FILE

This form is filed with the same filing authority with which you were required to file your Campaign Treasurer Appointment (Form CTA). If you are an officeholder who does not have a campaign treasurer appointment on file, file your reports with the same authority with which a candidate for your office must file the campaign treasurer appointment.

#### FILING A FINAL REPORT

For filing purposes, you are a "candidate" as long as you have an appointment of campaign treasurer on file. If you do not expect to accept any further campaign contributions or to make any further campaign expenditures, you may file a Final Report of contributions and expenditures. A Final Report terminates your appointment of campaign treasurer and relieves you of the obligation of filing further reports *as a candidate*. If you are an officeholder at the time of filing a Final Report, you may be required to file semiannual reports of contributions and expenditures as an officeholder. The only officeholders who are not required to file semiannual reports are officeholders who file locally, who do not have a campaign treasurer appointment on file, *and* who do not exceed \$940 in contributions or expenditures during the reporting period.

If you are not an officeholder at the time of filing a Final Report *and* if you have surplus funds or retain assets purchased with political funds, you will be required to file annual reports of Unexpended Contributions. (*See instructions for Form C/OH-UC*.)

To file a Final Report, you must complete the "C/OH CAMPAIGN FINANCE REPORT" (Form C/OH), check the "final" box in section 9 on the Cover Sheet, and complete and attach the "C/OH REPORT: DESIGNATION OF FINAL REPORT" (Form C/OH- FR).

## **COMPLETING THE COVER SHEET**

Each numbered item in these instructions corresponds to the same numbered item on the form.

#### <u>PAGE 1</u>

- 1. FILER ID: If you are filing with the Commission, you were assigned a filer identification number when you filed your initial campaign treasurer appointment. You should have received a letter acknowledging receipt of the form and informing you of your Filer ID. Enter this number wherever you see "FILER ID." If you do not file with the Commission, you are not required to enter a Filer ID.
- 2. TOTAL PAGES FILED: After you have completed the form, count the total number of pages of this form and any attached schedules. Enter that number where indicated on the top line of page 1 only. Each side of a two-sided form counts as one page.
- **3.** CANDIDATE/OFFICEHOLDER NAME: Enter your full name, including nicknames and suffixes (e.g., Sr., Jr., III), if applicable.
- 4. CANDIDATE/OFFICEHOLDER MAILING ADDRESS: Enter your complete mailing address. If your mailing address has changed since you last gave notice of your address, check the "Change of Address" box.
- **5. CANDIDATE/OFFICEHOLDER PHONE:** Enter your phone number including the area code, and your extension, if applicable.

Sections 6 - 8 pertain to a candidate's campaign treasurer. If you are an officeholder who does not have a campaign treasurer appointment on file, skip these sections.

- 6. CAMPAIGN TREASURER NAME: Enter the full name of your campaign treasurer, including nicknames and suffixes (e.g., Sr., Jr., III), if applicable.
- 7. CAMPAIGN TREASURER ADDRESS: Enter the complete address of your campaign treasurer.
- 8. CAMPAIGN TREASURER PHONE: Enter the phone number of your campaign treasurer including the area code, and the extension, if applicable.
- **9. REPORT TYPE:** Check the box that describes the type of report you are filing, according to the descriptions below. See the instructions for section 10 for the periods covered by each type of report.

**January 15 Report:** All candidates and most officeholders must file a semiannual report by January 15. The only officeholders who are not required to file this report are officeholders who file locally, who do not have a campaign treasurer appointment on file, *and* who do not exceed \$940 in contributions or expenditures during the reporting period.

All candidates and officeholders who file with the Commission must file this report by midnight Central Time on the January 15 report due date. All candidates and officeholders who file locally must file this report by 5 p.m. on the January 15 report due date.

Note: Anyone who has a campaign treasurer appointment (Form CTA) on file must file semiannual reports, even after an election has ended and even if the filer lost the election. To end this semiannual filing requirement, the filer must cease campaign activity and file a Final Report. (See "Final Report" below for more information.)

**July 15 Report:** All candidates and most officeholders must file a semiannual report by July 15. The only officeholders who are not required to file this report are officeholders who file locally, who do not have a campaign treasurer appointment on file, *and* who do not exceed \$940 in contributions or expenditures during the reporting period.

See "January 15 Report" above for more information on filing requirements and deadlines for semiannual reports.

**30th Day Before Election Report:** Opposed candidates in an election who did not choose the modified reporting schedule must file this pre-election report. If an opposed candidate chose modified reporting, but then exceeded a threshold before the 30th day before the election, the candidate must file this report.

The report is due no later than 30 days before the election. For all candidates and officeholders who file with the Commission, this report must be received by the Commission no later than midnight Central Time on the report due date. For all candidates and officeholders who file locally, this report must be received by the filing authority no later than 5 p.m. on the report due date.

You are an "opposed" candidate if you have an opponent, including a minor party candidate, whose name is printed on the ballot. If your only opposition is a write-in candidate, you are not considered opposed for filing purposes. If you are a write-in candidate, you are an "opposed" candidate subject to the reporting requirements if you accept political contributions or make political expenditures. Candidates who are unopposed in an election are not required to file pre-election reports for that election.

**8th Day Before Election Report:** Opposed candidates in an election who did not choose the modified reporting schedule must file this pre-election report. If an opposed candidate chose modified reporting but then exceeded a threshold before the 8th day before the election, the candidate must file this report.

The report is due no later than 8 days before the election. For all candidates and officeholders who file with the Commission, this report must be received by the Commission no later than midnight Central Time on the report due date. For all candidates and officeholders who file locally, this report must be received by the filing authority no later than 5 p.m. on the report due date.

See "30<sup>th</sup> Day Before Election Report" above for the definition of an opposed candidate.

**Runoff Report:** Opposed candidates who are participating in a runoff election and who did not choose the modified reporting schedule must file this runoff report. The report is due no later than 8 days before the runoff election. For all candidates and officeholders who file with Commission, this report must be received by the Commission no later than midnight Central Time on the report due date. For all candidates and officeholders who file

locally, this report must be received by the filing authority no later than 5 p.m. on the report due date.

See "30<sup>th</sup> Day Before Election Report" above for the definition of an opposed candidate.

**Exceeded Modified Reporting Limit Report:** Candidates who chose to file under the modified reporting schedule but then, after the 30th day before the election, exceeded \$940 in contributions or \$940 in expenditures in connection with the election must file this Exceeded Modified Reporting Limit report within 48 hours after exceeding the \$940 limit. The candidate must meet this deadline even if it falls on a weekend or a holiday.

**15th Day After Campaign Treasurer Appointment Report (Officeholders Only):** An officeholder must file this report if he or she appoints a campaign treasurer after a period of not having a campaign treasurer appointment (Form CTA) on file. For all officeholders who file with Commission, this report is due no later than midnight Central Time on the 15th day after an officeholder files Form CTA with the Commission. For all officeholders who file locally, this report is due no later than 5 p.m. on the 15th day after an officeholder files Form CTA with the filing authority. It is not required of officeholders who are merely changing their campaign treasurer. It is not required of an officeholder who files locally if the officeholder did not exceed \$940 in either contributions or expenditures during the period covered by the report. Candidates who are not officeholders do not file this report.

**Final Report:** A person who has a campaign treasurer appointment on file may file this report when he or she does not expect to accept any further campaign contributions or make or authorize any further campaign expenditures. There is not a fixed deadline for this report. This report must have a completed "C/OH REPORT: DESIGNATION OF FINAL REPORT" (Form C/OH-FR) attached.

A candidate must have a CTA on file to accept campaign contributions or make campaign expenditures, including contributions intended to offset campaign debts or expenditures made to pay campaign debts. A candidate who intends to continue campaign activity should not file a Final Report.

A Final Report terminates a candidate's CTA and relieves the candidate from any additional filing obligations as a candidate. Officeholders who file a Final Report will still be subject to the filing requirements applicable to officeholders. A person who is not an officeholder but who has surplus political funds or assets after filing a Final Report will be required to file annual Unexpended Contribution reports. (See "Form C/OH-FR: Designation of Final Report" for more information.) A candidate or officeholder who does not have a CTA on file may still be required to file a personal financial statement (PFS).

Filing a Final Report does not relieve a candidate of responsibility for any delinquent reports or outstanding civil penalties.

<u>Daily Pre-Election Report of Contributions</u>: A candidate or officeholder who files with the Commission may be required to file daily pre-election reports disclosing contributions during the period beginning the 9th day before an election and ending at 12 noon on the day before the election. This information can be disclosed on Form C/OH-T. For more information, please see the instructions for Form C/OH-T.

<u>Legislative Special Session Report</u>: A candidate or officeholder who files with the Commission and who accepts a political contribution during the period beginning on the date the governor signs the proclamation calling a special legislative session and continuing through the date of final adjournment is required to file a report after a special session of the legislature. This information can be disclosed on Form C/OH-SS. For more information, please see the instructions for Form C/OH-SS.

**10. PERIOD COVERED:** A reporting period includes the start date and the end date. The *due date* for filing will generally be *after* the end of the period. Generally, a report picks up where the last report left off, and there should be no gaps or overlapping periods. The exceptions are Daily Pre-election reports, which do create overlaps because you are required to report the activity twice.

<u>First Reports:</u> If this is the first report of contributions and expenditures that you have filed, the beginning date will depend on the date your campaign treasurer appointment (Form CTA) was filed or the date you took office.

- If you are a candidate (a person who has filed a Form CTA) and you are filing your first report, the start date will be the date your Form CTA was filed.
- If you are an officeholder who was appointed to an elective office and who did not have a Form CTA on file at the time of the appointment, the start date for your first report will be the date you took office.

**January 15th Semiannual Report:** The start date is July 1 of the previous year or the day after the last day covered by your last required report, whichever is later. If this is the first report you have filed, please see the "First Reports" section above. The end date is December 31 of the previous year.

**July 15th Semiannual Report:** The start date is January 1 or the day after the last day covered by your last required report, whichever is later. If this is the first report you have filed, please see the "First Reports" section above. The end date is June 30.

**30th Day Before Election Report:** The start date is the day after the last day covered by your last required report. If this is the first report you have filed, please see the "First Reports" section above. The end date is the 40th day before the election. This report is not required for unopposed candidates or candidates who are filing under the modified reporting schedule.

**8th Day Before Election Report:** The start date is the 39th day before the election if you filed a 30th Day Before Election Report. If you did not file the 30th Day Before Election Report, the day after the last day covered by your last required report is the start date. If this is the first report you have filed, please see the "First Reports" section above. The end date is the 10th day before the election. This report is not required for unopposed candidates or candidates who are filing under the modified reporting schedule.

**Runoff Report:** The start date is the 9th day before the main election if you filed an 8th Day Before Election Report. Otherwise, the start date is the day after the last day covered by your last required report or the day you appointed a campaign treasurer,

whichever is later. The end date is the 10th day before the runoff election. This report is not required for candidates who are filing under the modified reporting schedule.

**Exceeded Modified Reporting Limit Report:** The start date for the report is either the day you appointed your campaign treasurer or the day after the last day covered by your last required report, whichever is later. The end date is the day you exceeded the \$940 limit for contributions or expenditures.

**15th Day After Campaign Treasurer Appointment Report (Officeholders Only):** The start date is either the day after the last day covered by your last required report or the day you began serving an appointment to elective office. The end date is the day before the campaign treasurer appointment was filed. This report is due no later than 15 days after the campaign treasurer appointment was filed.

**Final Report:** The start date is the day after the last day covered by your last required report. The end date is the day the final report is filed.

If you are an officeholder without a campaign treasurer appointment on file, or if you have a campaign treasurer appointment on file but you are not a candidate in an upcoming election and were not a candidate in a recent election, you may skip Section 11.

**11. ELECTION:** If you are a candidate in an upcoming election or were a candidate in a recently held election, provide the following information concerning the upcoming or recent election.

**Election Date:** Enter the month, day, and year of the election for which this report is filed, if known.

<u>Candidate in an Upcoming Election:</u> If the political activity in the report primarily pertains to an upcoming election, provide the date of the upcoming election in which you intend to participate as a candidate that most immediately follows the deadline for this report.

<u>Candidate in a Recently Held Election:</u> If the political activity in this report primarily pertains to a recently held election, provide the date of the recently held election in which you participated as a candidate that most immediately precedes the deadline for this report.

**Election Type:** Check the box next to the type of election that most accurately describes the election for which this report is filed.

Primary: An election held by a political party to select its nominees for office.

**Runoff:** An election held if no candidate for a particular office receives the vote necessary to be elected in an election requiring a majority vote.

**General:** An election, other than a primary election, that regularly occurs at fixed dates.

**Special:** An election that is neither a general election nor a primary election nor a runoff election.

**Other:** If none of the listed election types apply, check "Other" and provide your own description of the election for which the report is filed.

- **12. OFFICE HELD:** If you are an officeholder, please enter the office you currently hold. Include the district, precinct, or other designation for the office, if applicable.
- **13. OFFICE SOUGHT:** If you are a candidate in an upcoming election, please enter the office you seek. If you were a candidate in a recently held election, but were unsuccessful or are not currently an officeholder, please enter the office you sought during the election that most immediately precedes the deadline for this report. Include the district, precinct, or other designation for the office, if applicable.
- 14. NOTICE FROM POLITICAL COMMITTEE(S): Complete this section if you received notice from a political committee that it accepted political contributions or made political expenditures on your behalf. You are required to disclose the receipt of such a notice in the report covering the period in which you receive the notice. If you have not received such notice, you may skip this section.

The political committee is required to include in the notice the full name and address of the committee, the full name and address of the committee's campaign treasurer, and a statement indicating whether the committee is a general-purpose committee or a specific-purpose committee. If the notice also describes the expenditure, do not include the description in this section.

**"Additional Pages" box:** If you received notice from more than one committee, check this box and attach an additional page listing the names and addresses of the other committees and of their campaign treasurers.

#### **Committee Type:**

**"General" box:** Check this box if the notice is from a general-purpose committee.

**"Specific" box:** Check this box if the notice is from a specific-purpose committee.

**Committee Name:** Enter the full name of the committee as reported in the notice.

Committee Address: Enter the address of the committee as reported in the notice.

**Committee Campaign Treasurer Name:** Enter the name of the committee's campaign treasurer as reported in the notice.

**Committee Campaign Treasurer Address:** Enter the address of the committee's campaign treasurer as reported in the notice.

#### <u>PAGE 2</u>

#### 15. C/OH (CANDIDATE/OFFICEHOLDER) NAME: Enter your full name.

**16. FILER ID:** See instructions for section 1.

17. TOTALS: Complete this section only after you have completed all applicable schedules.

Line 1- Total Unitemized Political Contributions: Enter the total of all unitemized contributions (other than pledges, loans, guarantees of loans, or contributions made electronically) of \$90 or less. Do not include any contributions itemized on Schedules A1 or A2 or any contribution made electronically. Enter a "0" if you did not receive any unitemized contributions during the period covered.

On Schedules A1 and A2, you are required to itemize political contributions that totaled more than \$90 from one person <u>and</u> any political contribution that is made electronically. You also may itemize contributions of \$90 or less from one person. Do not include any itemized contributions in the total entered on line 1, regardless of amount.

**Line 2- Total Political Contributions:** Add the total contributions listed on Schedules A1 and A2 to the amount you entered on line 1. Enter that total on line 2. Enter a "0" if you did not receive any contributions during the period covered.

**Line 3- Total Unitemized Political Expenditures:** Enter the total of all unitemized political expenditures of \$190 or less. Do not include any expenditures itemized on Schedules F1, F2, F3, F4, G, or H. Enter a "0" if you did not make any unitemized expenditures during the period covered.

On Schedule F1, you were required to itemize political expenditures that totaled more than \$190 to one payee. You also had the option of itemizing expenditures totaling \$190 or less to one payee. Do not include any expenditures itemized on Schedule F1 in the total entered on line 3, regardless of amount.

On Schedule F2, you were required to itemize incurred but not yet paid political expenditures that totaled more than \$190 to one payee. You also had the option of itemizing incurred political expenditures totaling \$190 or less to one payee. Do not include any political or non-political expenditures itemized on Schedule F2 in the total entered on line 3, regardless of amount.

On Schedule F4, you were required to itemize political expenditures made by a credit card that totaled more than \$190 to one payee. You also had the option of itemizing political expenditures totaling \$190 or less to one payee. Do not include any political or non-political expenditures itemized on Schedule F4 in the total entered on line 3, regardless of amount.

On Schedule G, you were required to itemize political expenditures from personal funds if you intend to seek reimbursement from political contributions. Do not include any expenditures itemized on Schedule G in the total entered on line 3, regardless of amount.

On Schedule H, you were required to itemize payments from political contributions made to certain businesses. Do not include any expenditures itemized on Schedule H in the total entered on line 3, regardless of amount.

Line 4- Total Political Expenditures: Add the following:

- (a) the total expenditures itemized on Schedule F1;
- (b) the total political expenditures itemized on Schedule F2;
- (c) the total political expenditures itemized on Schedule F4;
- (d) the total political expenditures itemized on Schedule G;
- (e) the total political expenditures itemized on Schedule H; and
- (f) the amount you entered on line 3.

Enter that total on line 4.

Enter a "0" if you did not make any expenditures during the period covered.

Line 5- Total Political Contributions Maintained: Enter the total amount of political contributions, including interest or other income on those contributions, maintained as of the last day of the reporting period. Enter "0" if you do not maintain political contributions, including interest or other income on those contributions, as of the last day of the reporting period. This is different from the total contributions reported on line 2. Only contributions accepted during the period covered by the report are entered on line 2.

The law requires you to disclose the total amount of political contributions accepted, including interest or other income on those contributions, maintained in one or more accounts in which political contributions are deposited as of the last day of the reporting period.

The "total amount of political contributions maintained" includes the total amount of political contributions maintained in one or more accounts, including the balance on deposit in banks, savings and loan institutions and other depository institutions; the present value of any investments that can be readily converted to cash, such as certificates of deposit, money market accounts, stocks, bonds, treasury bills, etc.; and the balance of political contributions accepted and held in any online fundraising account over which the filer can exercise control by making a withdrawal, expenditure, or transfer.

The total amount of political contributions maintained does *not* include personal funds that the filer intends to use for political expenditures, *unless* the personal funds have been disclosed as a loan to your campaign and deposited into an account in which political contributions are held as permitted by section 253.0351(c) of the Election Code. Any unexpended funds from such a loan are required to be included in the total amount of political contributions maintained as of the last day of the reporting period.

Note: Personal funds deposited in an account in which political contributions are held are subject to the personal use restrictions.

Line 6- Total Principal Amount of All Outstanding Loans: Enter the aggregate outstanding principal amount of all loans accepted for campaign or officeholder purposes as of the last day of the reporting period. Enter a "0" if you did not accept any loans during the period covered and have no outstanding loans as of the last day of the reporting period. This is different from the information reported on Schedule E. This line must include outstanding principal of loans made in this reporting period as well as outstanding principal of loans made previously.

**18. SIGNATURE:** Complete this section only after you have completed all applicable sections and schedules. You must always sign a report that you file. You must complete this section even if you have no schedules to attach. *Only the candidate or officeholder filing the report may sign the report*.

If you are using the paper form, fill this section out by hand after you finish the rest of this report. You have the option to either: (1) take the completed form to a notary public where you will sign above the first line that says "Signature of Candidate/Officeholder (Declarant)" (an electronic signature is not acceptable) and your signature will be notarized, or (2) sign above both lines that say "Signature of Candidate/Officeholder (Declarant)" (an electronic signature of Candidate/Officeholder (Declarant)" (an electronic signature of Candidate/Officeholder (Declarant)" (an electronic signature is not acceptable), and fill out the unsworn declaration section.

#### PAGE 3

- 19. C/OH (CANDIDATE/OFFICEHOLDER) NAME: Enter your full name.
- **20. FILER ID:** See instructions for section 1.
- **21. SCHEDULE SUBTOTALS:** Complete this section only after you have completed all applicable schedules.

Check the appropriate boxes to indicate which schedules are attached to your report. If a schedule is not included in the report, leave the check box blank.

**Line 1- Schedule A1:** Add the total amount of contributions itemized on Schedule A1 to the amount of unitemized monetary political contributions accepted during the period covered. Enter that total on line 1. Enter a "0" if you did not accept any contributions during the period covered.

**Line 2- Schedule A2:** Add the total amount of non-monetary in-kind contributions itemized on Schedule A2 to the amount of unitemized non-monetary in-kind contributions accepted during the period covered. Enter that total on line 2. Enter a "0" if you did not accept any non-monetary in-kind contributions during the period covered.

**Line 3- Schedule B:** Add the total amount of pledged contributions itemized on Schedule B to the amount of unitemized pledged contributions accepted during the

period covered. Enter that total on line 3. Enter a "0" if you did not accept any pledged contributions during the period covered.

**Line 4- Schedule E:** Add the total amount of loans itemized on Schedule E to the amount of unitemized loans accepted during the period covered. Enter that total on line 4. Enter a "0" if you did not accept any loans during the period covered.

**Line 5- Schedule F1:** Add the total amount of political expenditures from political contributions itemized on Schedule F1 to the amount of unitemized political expenditures from political contributions made during the period covered. Enter that total on line 5. Enter a "0" if you did not make any political expenditures from political contributions during the period covered.

**Line 6- Schedule F2:** Add the total amount of unpaid incurred obligations itemized on Schedule F2 to the amount of unitemized unpaid obligations incurred during the period covered. Enter that total on line 6. Enter a "0" if you did not incur any unpaid obligations during the period covered.

**Line 7- Schedule F3:** Enter the total amount of investments purchased from political contributions itemized on Schedule F3. Enter a "0" if you did not purchase any investments from political contributions during the period covered.

**Line 8- Schedule F4:** Add the total amount of expenditures made by a credit card itemized on Schedule F4 to the amount of unitemized expenditures made by a credit card during the period covered. Enter that total on line 8. Enter a "0" if you did not make any expenditures by credit card during the period covered.

**Line 9- Schedule G:** Add the total amount of political expenditures from personal funds itemized on Schedule G to the amount of unitemized political expenditures from personal funds made during the period covered. Enter that total on line 9. Enter a "0" if you did not make any political expenditures from personal funds during the period covered.

**Line 10- Schedule H:** Enter the total amount of payments from political contributions to a business of the candidate or officeholder itemized on Schedule H. Enter a "0" if you did not make any payments from political contributions to a business of the candidate or officeholder during the period covered.

**Line 11- Schedule I:** Enter the total amount of non-political expenditures from political contributions itemized on Schedule I. Enter a "0" if you did not make any non-political expenditures from political contributions during the period covered.

**Line 12- Schedule K:** Enter the total amount of interests, credits, gains, refunds, and contributions returned to the filer itemized on Schedule K. Enter a "0" if you did not have any such activity during the period covered.

## SCHEDULE A1: MONETARY POLITICAL CONTRIBUTIONS

# These instructions are for candidates and officeholders using SCHEDULE A1: MONETARY POLITICAL CONTRIBUTIONS.

Use this schedule to disclose information about monetary campaign and officeholder contributions accepted during the reporting period. Do not enter on this schedule information on non-monetary, in-kind contributions, pledges, loans, or guarantees of loans. Once you actually receive pledged money, it must be reported on Schedule A1. (Report non-monetary, in-kind contributions on Schedule A2; report pledges on Schedule B; report loans and guarantees of loans on Schedule E.)

**Itemization:** You must enter incoming monetary contributions that exceed \$90 from one person, and any monetary contribution made electronically, during a reporting period on this schedule. If you accepted two or more contributions from the same person, the total of which exceeds \$90, enter each contribution separately. Although you are not required to do so, you may also report contributions from one person that do not exceed \$90 in the period on this schedule. If you do not itemize contributions of \$90 and less on this schedule, you must total all such contributions and report them on the Cover Sheet, page 2, section 17, line 1.

Each numbered item in these instructions corresponds to the same numbered item on the form.

- 1. TOTAL PAGES SCHEDULE A1: After you have completed Schedule A1, count the total number of pages. Each side of a two-sided form counts as one page.
- 2. FILER NAME: Enter your full name.
- **3. FILER ID:** See instructions for Cover Sheet, page 1, section 1.
- **4. DATE:** Enter the date you *accepted* the contribution. Accepting a contribution is different from receiving a contribution. You accept a contribution when you decide to accept it rather than reject it. This may or may not be the same day that you receive the contribution.
- **5. FULL NAME OF CONTRIBUTOR:** Enter the full name of the contributor. If the contributor is an individual, enter the full first and last name, and suffix (Jr., III, etc.) if applicable. If the contributor is an entity, enter the full name of the entity.

**"Out-of-State PAC" box:** If the contributor is an out-of-state political committee, check the box. Certain restrictions apply to contributions from out-of-state PACS. The fact that a political committee has a mailing address outside of Texas does not mean that the committee is an out-of-state PAC for purposes of these restrictions. A political committee that has a campaign treasurer appointment on file in Texas is not an out-of-state PAC. A political committee that makes most of its political expenditures outside of Texas may be an out-of-state PAC. A political committee must determine if it is an out-of-state PAC.

If the contributor is an out-of-state political committee from which you accepted more than \$940 in the reporting period (including pledges or loans from sources other than financial institutions that have been in business for more than a year), you must include one of the following with your report:

- a written statement, certified by an officer of the out-of-state political committee, listing the full name and address of each person who contributed more than \$190 to the out-of-state political committee during the 12 months immediately preceding the contribution; *or*
- a copy of the out-of-state political committee's statement of organization filed as required by law with the FEC and certified by an officer of the out-of-state committee.

If the contributor is an out-of-state political committee from which you accepted \$940 or less (including pledges) during the reporting period, you must include one of the following with your report:

- a copy of the out-of-state political committee's statement of organization filed as required by law with the FEC and certified by an officer of the out-of-state committee; *or*
- a document listing the committee's name, address and phone number; the name of the person appointing the committee's campaign treasurer; and the name, address and phone number of the committee's campaign treasurer.

**"ID #" Line (Electronic Filing Only):** If you are filing your report electronically, you may enter in this field the out-of-state committee's Federal Election Commission (FEC) identification number. If you do not have an FEC # for the out-of-state PAC or are not filing electronically with the Commission, you must provide other documentation as explained above.

- 6. CONTRIBUTOR ADDRESS: Enter the complete address of the contributor.
- 7. AMOUNT OF CONTRIBUTION: Enter the amount of the contribution.
- 8. PRINCIPAL OCCUPATION OR JOB TITLE: Candidates for and holders of statewide offices in the executive branch and candidates for and holders of legislative offices must disclose the principal occupation or job title of an individual from whom the candidate or officeholder has accepted contributions (including pledges) of \$940 or more during the reporting period. In other circumstances, filers are not required to report this information but may do so.
- **9. EMPLOYER:** Candidates for and holders of statewide offices in the executive branch and candidates for and holders of legislative offices must disclose the employer of an individual from whom the candidate or officeholder has accepted contributions (including pledges) of \$940 or more during the reporting period. In other circumstances, filers are not required to report this information but may do so.

## SCHEDULE A2: NON-MONETARY (IN-KIND) POLITICAL CONTRIBUTIONS

These instructions are for candidates and officeholders using SCHEDULE A2: NON-MONETARY (IN-KIND) POLITICAL CONTRIBUTIONS.

Use this schedule to disclose information about non-monetary, in-kind campaign and officeholder contributions received during the reporting period. An in-kind contribution is a contribution of goods, services, or any other thing of value *other than money* that is given to your campaign. You are not required to include contributions of an individual's personal services or travel if the individual receives no compensation from any source for the services. Do not enter on this schedule information on monetary political contributions, pledges, loans, or guarantees of loans. Once you actually receive a pledged in-kind contribution, it must be reported on Schedule A2. (Report monetary contributions on Schedule A1; report pledges on Schedule B; report loans and guarantees of loans on Schedule E.)

**Itemization:** You must enter non-monetary (in-kind) contributions of goods, services, or other things of value that exceed \$90 from one person, and any non-monetary contribution made electronically, during a reporting period on this schedule. If you accepted two or more non-monetary contributions from the same person, the total of which exceeds \$90, enter each contribution separately. Although you are not required to do so, you may also report contributions from one person that do not exceed \$90 in the period on this schedule. If you do not itemize contributions of \$90 and less on this schedule, you must total all such contributions and report them on the Cover Sheet, page 2, section 17, line 1.

Each numbered item in these instructions corresponds to the same numbered item on the form.

- 1. TOTAL PAGES SCHEDULE A2: After you have completed Schedule A2, count the total number of pages. Each side of a two-sided form counts as one page.
- 2. FILER NAME: Enter your full name.
- **3. FILER ID:** See instructions for Cover Sheet, page 1, section 1.
- 4. TOTAL OF UNITEMIZED IN-KIND POLITICAL CONTRIBUTIONS: Enter the total amount of in-kind political contributions of \$90 or less that you accepted during the period covered that are not itemized on this schedule. If you choose to itemize an in-kind contribution of \$90 or less on this schedule, do not include it in this total. All contributions made electronically must be itemized.
- 5. DATE: See instructions for Schedule A1, section 4.
- 6. FULL NAME OF CONTRIBUTOR: See instructions for Schedule A1, section 5.

"Out-of-State PAC" box: See instructions for Schedule A1, section 5.

- 7. CONTRIBUTOR ADDRESS: Enter the complete address of the contributor.
- 8. AMOUNT OF CONTRIBUTION: Enter the fair market value of the in-kind contribution.

**9. IN-KIND CONTRIBUTION DESCRIPTION:** Enter a description of the contribution. The description should be sufficiently detailed to allow a person reviewing your report to understand what was contributed.

**"Travel Outside of Texas" box:** If the contribution was for travel outside of Texas, please check the box and *report this information on Schedule T.* 

- **10. PRINCIPAL OCCUPATION OR JOB TITLE:** See instructions for Schedule A1, section 8.
- 11. EMPLOYER: See instructions for Schedule A1, section 9.

Sections 12-16 pertain to judicial candidates and officeholders only. Do not complete these sections. If you are a judicial candidate or officeholder, please use form JC/OH and the corresponding instructions.

## SCHEDULE B: PLEDGED CONTRIBUTIONS

## These instructions are for candidates and officeholders using SCHEDULE B: PLEDGED CONTRIBUTIONS.

Use this schedule to disclose information about pledges accepted during the reporting period for campaign or officeholder purposes. You are not required to include pledges of an individual's personal services or travel if the individual receives no compensation from any source for the services. Do not enter on this schedule information on contributions actually received, loans, or guarantees of loans. (Report contributions actually received on Schedule A1 or Schedule A2, as applicable; report loans and guarantees of loans on Schedule E.)

If you accept a pledge from a person to give you money, goods, services, or anything of value, that pledge is a reportable contribution and you must include the pledge on this schedule for the report covering the period in which you accept the pledge.

**Itemization:** You must itemize pledges that exceed \$90 in the aggregate from one person during the reporting period. If you received pledges totaling more than \$90 from one person during the reporting period, you must itemize all of those pledges, even if individual pledges were for \$90 or less. Although you are not required to do so, you may also itemize pledges for \$90 or less from one person. You must also disclose the receipt of the pledged contribution on Schedule A1 (used for monetary contributions) or A2 (used for non-monetary contributions), as applicable, in the reporting period in which you actually receive the pledged money or thing of value. If the pledge is accepted and received in the same reporting period, it is not required to be reported on Schedule B.

Note: See the Campaign Finance Guide for more information on pledges.

Each numbered item in these instructions corresponds to the same numbered item on the form.

- 1. TOTAL PAGES SCHEDULE B: After you have completed Schedule B, count the total number of pages. Each side of a two-sided form counts as one page.
- 2. FILER NAME: Enter your full name.
- **3. FILER ID:** See instructions for Cover Sheet, page 1, section 1.
- 4. TOTAL OF UNITEMIZED PLEDGES: Enter the total amount of pledges that you accepted during the period that did not exceed \$90 in the aggregate per person. Although you are not required to do so, you may also itemize pledges of \$90 or less on this schedule. If you itemize some pledges of \$90 or less, do not include those pledges in the total entered here. If you choose to itemize all pledges of \$90 or less, do not enter a total amount here.
- **5. DATE:** Enter the date you *accepted* the pledge. Accepting a pledge is different from receiving a contribution. You accept a pledge when you decide to accept it rather than reject it.

<u>Pledge accepted and received in different reporting periods:</u> If you accept a pledge in one reporting period and then receive the pledged money or other thing of value in a later reporting period, you will disclose the pledge on this schedule in

the reporting period in which you accepted the pledge. You will also disclose the receipt of the pledged money or other thing of value on the appropriate incoming funds schedule (report monetary contributions on Schedule A1; report in-kind contributions on Schedule A2; report loans on Schedule E) in the reporting period in which you received the pledge.

<u>Pledge received in same reporting period as accepted:</u> If you receive a pledge in the same reporting period in which it was accepted, then you will not report the pledge on this schedule. You will only disclose the contribution on the appropriate incoming funds schedule (report monetary contributions on Schedule A1; report in-kind contributions on Schedule A2; report loans on Schedule E). The date of the contribution will be the date you accepted the pledged contribution, regardless of when the pledged contribution was actually received.

<u>Pledge accepted but never received:</u> You will disclose the pledge on this schedule in the reporting period in which you accepted the pledge. If you never actually receive the pledge, it is not necessary to correct your report to delete the pledge.

<u>Example:</u> In June a supporter promises that he will give Juan Garcia \$1,000 in the last week before the November election. Juan accepts his promise. Juan must disclose the pledge on his July 15 report covering the period in which he accepted the pledge. (Note: When he receives the \$1,000, he will disclose it as a monetary contribution on Schedule A1 of the report covering the period in which he received the money. Also, if he never receives the \$1,000, he does not correct/amend his report to delete the entry for the pledge.)

6. FULL NAME OF PLEDGOR: Enter the full name of the person who made the pledge.

"Out-of-State PAC" box: See instructions for Schedule A1, section 5.

- 7. **PLEDGOR ADDRESS:** Enter the complete address of the person who made the pledge.
- **8. AMOUNT OF PLEDGE:** Enter the amount of the pledge or the fair market value of any pledged goods or services or other thing of value, as applicable.
- **9. IN-KIND DESCRIPTION:** If the pledge was for goods or services or any other thing of value, enter a description of the pledged goods or services or other thing of value. The description should be sufficiently detailed to allow a person reviewing your report to understand what was pledged.

**"Travel Outside of Texas" box:** If the pledged contribution was an in-kind contribution for travel outside of Texas, please check the box and *report this information on Schedule T*.

- **10. PRINCIPAL OCCUPATION OR JOB TITLE:** See instructions for Schedule A1, section 8.
- **11. EMPLOYER:** See instructions for Schedule A1, section 9.

You do not need Schedules C1-4 and D. These schedules are for political committees to report contributions from corporations and labor organizations. Candidates and officeholders are generally prohibited from accepting such contributions.

## **SCHEDULE E: LOANS**

#### These instructions are for candidates and officeholders using SCHEDULE E: LOANS.

Use this schedule to disclose information about loans and guarantees of loans accepted during the reporting period for campaign or officeholder purposes. This schedule must also be used to disclose deposits of personal funds into an account in which political contributions are held as permitted by section 253.0351(c) of the Election Code. This schedule may also be used to disclose political expenditures from personal funds.

Loans to Your Campaign from Your Personal Funds: You may disclose political expenditures from personal funds as a loan to your campaign on Schedule E. Outgoing political expenditures made from that loan must then be disclosed as if they were made from political contributions. The amount you disclose as a loan from yourself in a reporting period may NOT exceed the amount you actually spent from personal funds in that reporting period. In other words, do not report a \$100,000 loan to your campaign if the amount actually spent from your personal funds in the reporting period was \$5,000. When you reimburse yourself, disclose the reimbursement as an outgoing political expenditure on Schedule F1. The reimbursement may not exceed the amount disclosed as a loan. (You may also disclose political expenditures from personal funds on Schedule G. See the Schedule G instructions below for more information.)

<u>Personal Funds Deposited into a Political Account:</u> If you deposit personal funds in an account in which political contributions are held, you must disclose the deposited amount as a loan on Schedule E and check the box indicating "Personal Funds Deposited into Political Account." Personal funds deposited in an account in which political contributions are held are subject to the personal use restriction. Disclose the outgoing political expenditures made from that loan as if they were made from political contributions. When you reimburse yourself, disclose the reimbursement as an outgoing political expenditure on Schedule F1. The reimbursement may not exceed the amount disclosed as a loan.

**Itemization:** You must itemize loans (including loans from personal funds) that exceed \$90 that you accepted during the period from one person. If you accepted two or more loans from the same person, the total of which exceeds \$90, itemize each loan separately. You must also itemize loans that are made electronically by a person other than a financial institution. Although you are not required to do so, you may also itemize any other loans that do not exceed \$90.

- 1. TOTAL PAGES SCHEDULE E: After you have completed Schedule E, count the total number of pages. Each side of a two-sided form counts as one page.
- 2. FILER NAME: Enter your full name.
- **3. FILER ID:** See instructions for Cover Sheet, page 1, section 1.

**4. TOTAL OF UNITEMIZED LOANS:** Enter the total amount of loans accepted during the reporting period that did not exceed \$90 in the aggregate per person and were not from financial institutions, unless the loans were made electronically.

Although you are not required to do so, you may itemize loans of \$90 or less from persons other than financial institutions on this schedule. If you itemize some loans of \$90 or less, do not include those loans in the total you enter here. If you choose to itemize all loans of \$90 or less, enter a "0" here.

- 5. DATE OF LOAN: Enter the date you *accepted* the loan.
- 6. IS LENDER A FINANCIAL INSTITUTION?: If you accepted the loan from a corporation that has been legally engaged in the business of making loans for more than one year, circle "Y" for yes. If you accepted the loan from any other source, circle "N" for no. A loan from a corporation that has not been legally engaged in the business of making loans for more than one year is a corporate contribution. Candidates and officeholders may not accept corporate contributions.
- 7. NAME OF LENDER: Enter the full name of the person or financial institution that made the loan. If the lender is an individual, enter the full first and last name and suffix (Jr., III, etc.) if applicable. If the lender is an entity, enter the full name of the entity.

"Out-of-State PAC" box: See instructions for Schedule A1, section 5.

Note: See the Campaign Finance Guide for detailed information on accepting and reporting contributions from out-of-state political committees.

- 8. LENDER ADDRESS: Enter the complete address of the person or financial institution that made the loan.
- 9. LOAN AMOUNT: Enter the principal amount of the loan.
- **10. INTEREST RATE:** Enter the interest rate.
- **11. MATURITY DATE:** Enter the maturity date.
- 12. PRINCIPAL OCCUPATION OR JOB TITLE: Candidates for and holders of statewide offices in the executive branch and candidates for and holders of legislative offices must disclose the principal occupation or job title of each individual from whom the candidate or officeholder has accepted a loan (including a pledge of a loan) of \$940 or more during the reporting period. Other types of filers are not required to report this information but may do so.
- **13. EMPLOYER:** Candidates for and holders of statewide offices in the executive branch and candidates for and holders of legislative offices must disclose the full name of the employer of an individual from whom the candidate or officeholder has accepted a loan (including a pledge of a loan) of \$940 or more during the reporting period. Other types of filers are not required to report this information but may do so.

- 14. **DESCRIPTION OF COLLATERAL:** If there is no collateral for the loan, check the "none" box and go to section 15. If there is collateral for the loan, enter a description of the collateral for the loan.
- **15.** "Check if personal funds were deposited into political account" box: Check this box *only if* the loan is a deposit of your personal funds into an account in which political contributions are held as permitted by section 253.0351(c) of the Election Code. Political expenditures made from that loan, and any subsequent expenditures to reimburse the candidate or officeholder, must be reported as if they were made from political contributions. The reimbursement may not exceed the amount reported as a loan. Personal funds deposited in an account in which political contributions are held are subject to the personal use restrictions.
- **16. GUARANTOR INFORMATION:** If there are no guarantors for the loan, check the "Not Applicable" box and go to the next loan. If you have no further loans to report, go to the next applicable schedule.

A person who guarantees all or part of a loan makes a reportable contribution in the amount of the guarantee. You must report such a contribution on this schedule, and not on the contributions schedule.

- **17.** NAME OF GUARANTOR: Enter the full name of the person guaranteeing the loan. If the guarantor is an individual, enter the full first and last name and suffix (Jr., III, etc.) if applicable. If the guarantor is an entity, enter the full name of the entity.
- **18. GUARANTOR ADDRESS:** Enter the complete address of the guarantor.
- **19. AMOUNT GUARANTEED:** Enter the dollar amount of the loan that the guarantor has agreed to guarantee.
- **20. PRINCIPAL OCCUPATION:** Enter the principal occupation of the guarantor.
- **21. EMPLOYER:** Enter the employer of the guarantor.

## SCHEDULE F1: POLITICAL EXPENDITURES FROM POLITICAL CONTRIBUTIONS

These instructions are for candidates and officeholders using SCHEDULE F1: POLITICAL EXPENDITURES FROM POLITICAL CONTRIBUTIONS.

Use this schedule to disclose information about political expenditures from political contributions that were made during the reporting period. Do not enter on this schedule unpaid incurred obligations, political expenditures made from personal funds, the purchase of investments from political contributions, expenditures made by credit card, or payments from political contributions made to a business that you own or control. (Report unpaid incurred obligations on Schedule F2; report expenditures from personal funds on Schedule G; report the purchase of investments from political contributions on Schedule F3; report expenditures made by credit card on Schedule F4; and report payments from political contributions made to a business that you own or control contributions made to a business that you own or schedule F3; report expenditures made by credit card on Schedule F4; and report payments from political contributions made to a business that you own or control contributions made to a business that you own or control contributions made to a business that you own or control payments from political contributions made to a business that you own or control on Schedule F3; report expenditures made by credit card on Schedule F4; and report payments from political contributions made to a business that you own or control on Schedule H.)

Expenditures Made by Credit Card: You must disclose expenditures charged to a credit card on Schedule F4 and *not* on this schedule. When you pay the credit card bill, you must disclose the payment to the credit card company on Schedule F1 (used for political payments from political contributions), Schedule G (used for political payments from personal funds), Schedule H (used for payments from political contributions made to a business that you own or control), or Schedule I (used for nonpolitical payments from political contributions), as applicable. See instructions for Schedule F4: Expenditures Made by Credit Card for more information.

See the *Campaign Finance Guide for Candidates and Officeholders* for important restrictions regarding the use of political funds to rent or purchase real property.

**Itemization:** You must enter expenditures paid to one individual or entity during a reporting period that in the aggregate exceed \$190 on this schedule. If you made more than one expenditure to the same payee, the total of which exceeded \$190, enter each expenditure separately. Although you are not required to do so, you may also report expenditures to one person that do not exceed \$190 in the period on this schedule. If you choose not to itemize expenditures of \$190 and less on this schedule, you must total all unitemized expenditures and report them on the Cover Sheet, page 2, section 17, line 3.

- 1. TOTAL PAGES SCHEDULE F1: After you have completed Schedule F1, count the total number of pages. Each side of a two-sided form counts as one page.
- 2. FILER NAME: Enter your full name.
- **3. FILER ID:** See instructions for Cover Sheet, page 1, section 1.
- **4. DATE:** Enter the date the expenditure payment was made. Remember: Expenditure obligations you incurred in this reporting period *but have not yet paid* are entered on Schedule F2. Expenditures made by credit card are entered on Schedule F4.
- 5. **PAYEE NAME:** Enter the full name of the person to whom the expenditure was made.

Note: If you make an expenditure for goods or services to benefit another candidate, officeholder, or committee, enter the name of the vendor who sold you the goods or services. Do not enter the name of the person for whose benefit you made the expenditure. Include that information under section 8, "Purpose of Expenditure."

- 6. AMOUNT: Enter the exact amount of the expenditure.
- **7. PAYEE ADDRESS:** Enter the complete address of the person to whom the expenditure was made.
- 8. PURPOSE OF EXPENDITURE: You must disclose the purpose of the expenditure in two parts: Category and Description. Merely disclosing the category of goods, services, or other thing of value for which the expenditure is made does not adequately describe the purpose of an expenditure.
  - (a) **Category:** Select a category of goods, services, or other thing of value for which an expenditure is made. If none of the listed categories apply, select "Other" and enter your own category. Examples of acceptable categories include:

Advertising Expense

Accounting/Banking

Consulting Expense

Contributions/Donations Made By Candidate/Officeholder/Political Committee

Credit Card Payment

Event Expense

Fees

Food/Beverage Expense

Gifts/Awards/Memorials Expense

Legal Services

Loan Repayment/Reimbursement

Office Overhead/Rental Expense

Polling Expense

Printing Expense

Salaries/Wages/Contract Labor

Solicitation/Fundraising Expense

Transportation Equipment and Related Expense

Travel In District

Travel Out Of District

Other

(b) **Description:** Enter a brief statement or description of the candidate or officeholder activity that is conducted by making the expenditure. The brief statement or description must include the item or service purchased and must be sufficiently specific, when considered within the context of the description of the category, to make the reason for the expenditure clear. Merely disclosing the category of goods, services, or other thing of value for which the expenditure is made does not adequately describe the purpose of an expenditure.

For examples of acceptable ways to disclose the purpose of an expenditure, please see the "Examples: Purpose of Expenditures" on page 46.

**"Check if travel outside of Texas" box:** Check this box if the expenditure is for travel outside of Texas. The description of a political expenditure for travel outside of the state of Texas must include detailed information. Please report this information on Schedule T.

"Check if Austin, TX, officeholder living expense" box: Check this box if the expenditure is an officeholder expense for living in Austin, Texas.

#### 9. DIRECT CAMPAIGN EXPENDITURE TO BENEFIT CANDIDATE/OFFICEHOLDER:

If you made a direct campaign expenditure to benefit another candidate or officeholder, enter the full name of the candidate or officeholder and the name of the office sought or held, including the district, precinct, or other designation of the office, as applicable. (Attach additional sheets to list multiple candidates.) Do not complete this section if the expenditure was not a direct campaign expenditure.

A "direct campaign expenditure" to benefit another candidate is not a "political contribution" to that other candidate. A direct campaign expenditure is a campaign expenditure that you make on someone else's behalf and without the prior consent or approval of that person. This is in contrast to a political contribution, which the person has the opportunity to accept or reject.

<u>Example:</u> If you made expenditures to prepare and distribute an endorsement letter in support of a candidate after first asking for and getting the candidate's approval, you made an *in-kind contribution*. However, if you did not get the candidate's approval *before* you made the expenditure, you made a *direct campaign expenditure*.

## SCHEDULE F2: UNPAID INCURRED OBLIGATIONS

# These instructions are for candidates and officeholders using SCHEDULE F2: UNPAID INCURRED OBLIGATIONS.

Use this schedule to disclose information about obligations to make an expenditure that you incurred during the reporting period but have not yet paid. Do not enter on this schedule obligations that were incurred and paid during the reporting period, or other outgoing funds. (Report obligations incurred and paid during the reporting period on Schedule F1, F3, G, H, or I as appropriate, and report expenditures made by credit card on Schedule F4.)

See the *Campaign Finance Guide for Candidates and Officeholders* for important restrictions regarding the use of political funds to rent or purchase real property.

**Itemization:** Itemization requirements differ depending on whether the unpaid incurred obligation is for a political or non-political expenditure.

<u>Unpaid Incurred Political Obligations:</u> You must enter political obligations incurred but not yet paid to one individual or entity during a reporting period that in the aggregate exceed \$190 on this schedule. If you incurred more than one obligation to the same payee, the total of which exceeded \$190, enter each expenditure separately. Although you are not required to do so, you may also report political obligations incurred to one person that do not exceed \$190 in the period on this schedule. If you choose not to itemize incurred political obligations of \$190 and less on this schedule, you must total all unitemized obligations and report them in section 4 of this Schedule. You must also include that amount in the total unitemized political expenditures of \$190 or less on C/OH Cover Sheet, page 2, section 17, line 3.

<u>Unpaid Incurred Non-Political Obligations:</u> You must enter non-political obligations incurred but not yet paid to one individual or entity during a reporting period on this schedule, regardless of the amount.

- 1. TOTAL PAGES SCHEDULE F2: After you have completed Schedule F2, count the total number of pages. Each side of a two-sided form counts as one page.
- 2. FILER NAME: Enter your full name.
- **3. FILER ID:** See instructions for Cover Sheet, page 1, section 1.
- 4. TOTAL OF UNITEMIZED UNPAID INCURRED OBLIGATIONS: Enter the total amount of political obligations incurred during the reporting period that do not exceed \$190 in the aggregate per person, unless itemized on this schedule. You are not required to itemize unpaid incurred political obligations of \$190 or less, but if you choose to do so, do not include those unpaid incurred obligations in the total you enter here.
- 5. DATE: Enter the date the obligation was incurred. Obligations you incurred *and* paid during the reporting period are not entered on this schedule.

6. PAYEE NAME: See instructions for Schedule F1, section 5.

Note: If you incurred an obligation for goods or services to benefit another candidate, officeholder, or committee, enter the name of the vendor of the goods or services. Do not enter the name of the person for whose benefit you incurred the obligation. Include that information under section 10, "Purpose of Expenditure."

- 7. AMOUNT: Enter the exact amount of the incurred obligation.
- **8. PAYEE ADDRESS:** Enter the complete address of the person to whom the obligation is owed.
- **9. TYPE OF EXPENDITURE:** Check only one box to indicate whether the incurred obligation was political or non-political.

A non-political expenditure is an expenditure that is neither a campaign expenditure nor an officeholder expenditure. As a practical matter, *very few* expenditures made from political contributions are non-political expenditures. For instance, expenditures for administrative expenses, banking fees, and professional dues are typically political expenditures.

- 10. PURPOSE OF EXPENDITURE: See instructions for Schedule F1, section 8.
- **11. DIRECT CAMPAIGN EXPENDITURE TO BENEFIT CANDIDATE/OFFICEHOLDER:** See instructions for Schedule F1, section 9.

## SCHEDULE F3: PURCHASE OF INVESTMENTS FROM POLITICAL CONTRIBUTIONS

These instructions are for candidates and officeholders using SCHEDULE F3: PURCHASE OF INVESTMENTS FROM POLITICAL CONTRIBUTIONS.

Use this schedule to disclose information about investments purchased from political contributions during the reporting period. Do not enter on this schedule political expenditures from political contributions, unpaid incurred obligations, expenditures made by credit card, political expenditures made from personal funds, or payments from political contributions made to a business that you own or control. (Report political expenditures from political contributions on Schedule F1; report unpaid incurred obligations on Schedule F2; report expenditures made by credit card on Schedule F4; report expenditures from personal funds on Schedule G; and report payments from political contributions made to a business that you own or control on Schedule H.)

See the *Campaign Finance Guide for Candidates and Officeholders* for important restrictions regarding the use of political funds to rent or purchase real property.

**Itemization:** You must enter investments purchased with political contributions during a reporting period that in the aggregate exceed \$120 on this schedule. Although you are not required to do so, you may also report investments purchased with political contributions that do not exceed \$120 in the period on this schedule.

- 1. TOTAL PAGES SCHEDULE F3: After you have completed Schedule F3, count the total number of pages. Each side of a two-sided form counts as one page.
- 2. FILER NAME: Enter your full name.
- **3. FILER ID:** See instructions for Cover Sheet, page 1, section 1.
- 4. DATE: Enter the date you purchased the investment.
- **5.** NAME OF PERSON FROM WHOM INVESTMENT IS PURCHASED: Enter the full name of the person or entity from whom you purchased the investment. If you purchased the investment from an individual, enter the full first and last name, and suffix (Jr., III, etc.) if applicable (title is optional). If you purchased the investment from an entity, enter the full name of the entity.
- **6. ADDRESS OF PERSON FROM WHOM INVESTMENT IS PURCHASED:** Enter the complete address of the person or entity from whom you purchased the investment.
- **7. DESCRIPTION OF INVESTMENT:** Enter a brief statement or description of the investment. For example, "Ten shares of stock in ABC company."
- 8. AMOUNT OF INVESTMENT: Enter the amount of the investment purchased.

## SCHEDULE F4: EXPENDITURES MADE BY CREDIT CARD

# These instructions are for candidates and officeholders using SCHEDULE F4: EXPENDITURES MADE BY CREDIT CARD.

Use this schedule to disclose information about expenditures made by a credit card. You must disclose expenditures charged to a credit card on this schedule and identify the individual, entity, or vendor who receives payment from the credit card company. When you pay the credit card bill, you must disclose the payment to the credit card company on Schedule F1 (used for political payments from political contributions), Schedule G (used for political payments from personal funds), Schedule H (used for payments from political contributions made to a business that you own or control), or Schedule I (used for nonpolitical payments from political contributions), as applicable.

Do not enter on this schedule political expenditures from political contributions, unpaid incurred obligations, political expenditures made from personal funds, or payments from political contributions made to a business that you own or control. (Report political expenditures from political contributions on Schedule F1; report unpaid incurred obligations on Schedule F2; report the purchase of investments from political contributions on Schedule G; and report payments from political contributions made to a business that you own or control from political contributions made to a business that political contributions on Schedule F3; report expenditures from personal funds on Schedule G; and report payments from political contributions made to a business that you own or control on Schedule H.)

For examples regarding the disclosure of expenditures made by credit card, please see "Examples: Reporting Expenditures Made by Credit Card" on page 43.

**Itemization:** Itemization requirements differ depending on whether the expenditure made by a credit card is for a political or non-political expenditure.

Political Expenditures Made by Credit Card: You must itemize political expenditures made by credit card that exceed \$190 (in the aggregate) to a single payee. If you made two or more expenditures to the same payee, the total of which exceeded \$190, enter each expenditure made by credit card separately. Although you are not required to do so, you may also report political expenditures made by credit card that do not exceed \$190 in the reporting period on this schedule. If you choose not to itemize political expenditures made by credit card of \$190 and less on this schedule, you must total all unitemized political expenditures and report them in section 4 of this Schedule. You must also include that amount in the total unitemized political expenditures of \$190 or less on C/OH Cover Sheet, page 2, section 17, line 3.

<u>Non-Political Expenditures Made by Credit Card:</u> You must itemize any non-political expenditure made by credit card, regardless of the amount.

- **1. TOTAL PAGES SCHEDULE F4:** After you have completed Schedule F4, count the total number of pages. Each side of a two-sided form counts as one page.
- 2. FILER NAME: Enter your full name.

- **3. FILER ID:** See instructions for Cover Sheet, page 1, section 1.
- 4. TOTAL OF UNITEMIZED EXPENDITURES CHARGED TO A CREDIT CARD: Enter the total amount of political expenditures charged to a credit card during the reporting period that do not exceed \$190 in the aggregate per person, unless itemized on this schedule. You are not required to itemize political expenditures made by credit card of \$190 or less, but if you choose to do so, do not include those political expenditures made by credit card in the total you enter here.
- 5. DATE: Enter the date you made the expenditure by credit card.

Note: There is a special reporting rule for expenditures made by credit card. For reports due 30 days and 8 days before an election (pre-election reports) and for runoff reports, the date of the credit card expenditure is the date the credit card is used. For other reports, the date of the credit card expenditure is either the date of the charge or the date the credit card statement is received. A *filer can never go wrong by disclosing the date of the expenditure as the date of the charge*.

6. PAYEE NAME: See instructions for Schedule F1, section 5. Disclose the name of the vendor who sold you the goods or services as the payee, NOT the credit card company. You do not report the name of the credit card company on this schedule.

Note: If you made an expenditure for goods or services to benefit another candidate, officeholder, or committee, enter the name of the vendor of the goods or services. Do not enter the name of the person for whose benefit you made the expenditure. Include that information under section 10, "Purpose of Expenditure."

- 7. AMOUNT: Enter the amount of the credit card expenditure.
- 8. PAYEE ADDRESS: Enter the complete address of the payee of the credit card expenditure.
- **9. TYPE OF EXPENDITURE:** Check only one box to indicate whether the credit card expenditure was political or non-political.

A non-political expenditure is an expenditure that is neither a campaign expenditure nor an officeholder expenditure. As a practical matter, *very few* expenditures made from political contributions are non-political expenditures. For instance, expenditures for administrative expenses, banking fees, and professional dues are typically political expenditures.

10. PURPOSE OF EXPENDITURE: See instructions for Schedule F1, section 8.

Note: Do not choose "Credit Card Payment" as the category for an expenditure made by credit card when an individual, entity, or vendor receives payment from the credit card company. Instead, choose the category that corresponds to the goods, services, or other thing of value purchased from the individual, entity, or vendor.

**11. DIRECT CAMPAIGN EXPENDITURE TO BENEFIT CANDIDATE/OFFICEHOLDER:** See instructions for Schedule F1, section 9.

### SCHEDULE G: POLITICAL EXPENDITURES MADE FROM PERSONAL FUNDS

# These instructions are for candidates and officeholders using SCHEDULE G: POLITICAL EXPENDITURES MADE FROM PERSONAL FUNDS.

You may use this schedule to disclose information about political expenditures from personal funds that were made during the reporting period. Alternatively, you may choose to disclose political expenditures from personal funds as a loan on Schedule E (see the Schedule E instructions above for more information). Do not enter on this schedule information about personal funds deposited in an account in which political contributions are held as permitted by section 253.0351(c) of the Election Code. (Report the deposit of personal funds into a political account as a loan on Schedule E.)

**Expenditures Made by Credit Card:** You must disclose expenditures charged to a credit card on Schedule F4 and *not* on this schedule. When you pay the credit card bill, you must disclose the payment to the credit card company on Schedule F1 (used for political payments from political contributions), Schedule G (used for political payments from personal funds), Schedule H (used for payments from political contributions made to a business that you own or control), or Schedule I (used for nonpolitical payments from political contributions), as applicable. See instructions for Schedule F4: Expenditures Made by Credit Card for more information.

If you intend to seek reimbursement *in any amount* from political contributions for a political expenditure made from personal funds, you must either report the expenditure on Schedule E or itemize the expenditure on this schedule and check the box in Section 6 to indicate that you intend to seek reimbursement from political contributions. *You may not correct a report to allow reimbursement*. When you reimburse yourself, disclose the reimbursement as an outgoing political expenditure on Schedule F1.

See the Campaign Finance Guide for important restrictions regarding the use of political funds to rent or purchase real property.

**Itemization:** If you choose to report political expenditures from personal funds on this schedule, you must itemize political expenditures paid to one individual or entity during a reporting period that in the aggregate exceed \$190 on this schedule. If you made more than one expenditure to the same payee, the total of which exceeded \$190, enter each expenditure separately. Although you are not required to do so, you may also report expenditures to one person that do not exceed \$190 in the period on this schedule. You must total all political expenditures from personal funds that you do not itemize on this schedule and include them in the total of unitemized political expenditures on the C/OH Cover Sheet, page 2, section 17, line 3.

Officeholder expenditures from personal funds for which you do not intend to seek reimbursement are not required to be reported on this schedule or included in the total of unitemized political expenditures.

Each numbered item in these instructions corresponds to the same numbered item on the form.

**1 TOTAL PAGES SCHEDULE G:** After you have completed Schedule G, count the total number of pages. Each side of a two-sided form counts as one page.

- 2. FILER NAME: Enter your full name.
- **3. FILER ID:** See instructions for Cover Sheet, page 1, section 1.
- **4. DATE:** Enter the date the expenditure was made.
- **5. PAYEE NAME:** See instructions for Schedule F1, section 7.
- 6. AMOUNT: Enter the exact amount of the expenditure.

**"Reimbursement from Political Contributions Intended" box:** Check this box if you intend to reimburse yourself for the expenditure. (In order to be reimbursed from political contributions in any amount for an expenditure made out of personal funds, you must itemize the expenditure on this schedule and check this box or you must report the expenditure as a loan to yourself on Schedule E.)

- **7. PAYEE ADDRESS:** Enter the complete address of the person to whom the expenditure was made.
- 8. PURPOSE OF EXPENDITURE: See instructions for Schedule F1, section 8.
- **9. DIRECT CAMPAIGN EXPENDITURE TO BENEFIT CANDIDATE/OFFICEHOLDER:** See instructions for Schedule F1, section 9.

## SCHEDULE H: PAYMENT FROM POLITICAL CONTRIBUTIONS TO A BUSINESS OF C/OH

These instructions are for candidates and officeholders using SCHEDULE H: PAYMENT FROM POLITICAL CONTRIBUTIONS TO A BUSINESS OF C/OH.

Use this schedule to disclose information about payments from political contributions that were made to a business in which you have an interest of more than 10%, a position on the governing body, or a position as an officer. Do not enter on this schedule other payments from political contributions made during the reporting period.

See the *Campaign Finance Guide for Candidates and Officeholders* for a discussion on the important restrictions on making and reporting payments from political contributions to a business in which you have an interest.

This schedule is for payments to a business in which you have one or more of the following interests or positions:

- 1) a participating interest of more than 10%;
- 2) a position on the governing body of the business; or
- 3) a position as an officer of the business.

**Itemization:** You must enter all payments from political contributions made to certain businesses (as defined above) of a candidate or officeholder made during the reporting period on this schedule, regardless of the amount.

- **1. TOTAL PAGES SCHEDULE H:** After you have completed Schedule H, count the total number of pages. Each side of a two-sided form counts as one page.
- 2. FILER NAME: Enter your full name.
- **3. FILER ID:** See instructions for Cover Sheet, page 1, section 1.
- 4. DATE: Enter the date you made the payment.
- 5. BUSINESS NAME: Enter the full name of the business to which you made the payment.
- 6. AMOUNT: Enter the dollar amount of the payment.
- **7. BUSINESS ADDRESS:** Enter the complete address of the business to which you made the payment.
- 8. PURPOSE OF EXPENDITURE: See instructions for Schedule F1, section 8.
- **9. DIRECT CAMPAIGN EXPENDITURE TO BENEFIT CANDIDATE/OFFICEHOLDER:** See instructions for Schedule F1, section 9.

## SCHEDULE I: NON-POLITICAL EXPENDITURES MADE FROM POLITICAL CONTRIBUTIONS

These instructions are for candidates and officeholders using SCHEDULE I: NON-POLITICAL EXPENDITURES MADE FROM POLITICAL CONTRIBUTIONS.

Use this schedule to disclose information about non-political expenditures from political contributions made during the reporting period. Do not enter political expenditures on this schedule. Also, do not enter non-political expenditure obligations you incurred in this reporting period but have not yet paid or non-political expenditures made by credit card. (Report unpaid incurred obligations on Schedule F2; report expenditures made by a credit card on Schedule F4.)

**Expenditures Made by Credit Card:** You must disclose non-political expenditures charged to a credit card on Schedule F4 and *not* on this schedule. When you pay the credit card bill, you must disclose the payment to the credit card company on Schedule F1 (used for political payments from political contributions), Schedule G (used for political payments from personal funds), Schedule H (used for payments from political contributions made to a business that you own or control), or Schedule I (used for nonpolitical payments from political contributions), as applicable. See instructions for Schedule F4: Expenditures Made by Credit Card for more information.

**Itemization:** You must enter all non-political expenditures from political contributions on this schedule, regardless of the amount. A non-political expenditure is an expenditure that is neither a campaign expenditure nor an officeholder expenditure. As a practical matter, *very few* expenditures made from political contributions are non-political expenditures. For instance, expenditures for administrative expenses, banking fees, and professional dues are typically political expenditures. You may not convert political contributions to personal use.

- 1. TOTAL PAGES SCHEDULE I: After you have completed Schedule I, count the total number of pages. Each side of a two-sided form counts as one page.
- 2. FILER NAME: Enter your full name.
- **3. FILER ID:** See instructions for Cover Sheet, page 1, section 1.
- 4. DATE: Enter the date the expenditure payment was made.
- **5. PAYEE NAME:** See instructions for Schedule F1, section 5.
- **6. AMOUNT:** Enter the exact amount of the expenditure payment.
- **7. PAYEE ADDRESS:** Enter the complete address of the person to whom the expenditure was made.
- 8. PURPOSE OF EXPENDITURE: See instructions for Schedule F1, section 8.

## SCHEDULE K: INTEREST, CREDITS, GAINS, REFUNDS, AND CONTRIBUTIONS RETURNED TO FILER

*These instructions are for candidates and officeholders using SCHEDULE K: INTEREST, CREDITS, GAINS, REFUNDS, AND CONTRIBUTIONS RETURNED TO FILER.* 

Use this schedule to report information regarding any credit, interest, rebate, refund, reimbursement, or return of a deposit fee resulting from the use of a political contribution or an asset purchased with a political contribution, any proceeds of the sale of an asset purchased with a political contribution, the amount of which exceeds \$120, and any other gain from a political contribution received during the reporting period.

**Itemization:** You must enter interest, credits, gains, refunds and returned contributions received during a reporting period that in the aggregate exceed \$120 on this schedule. Although you are not required to do so, you may also report any credit/gain/refund, or interest that does not exceed \$120 in the period on this schedule.

Each numbered item in these instructions corresponds to the same numbered item on the form.

- 1. TOTAL PAGES SCHEDULE K: After you have completed Schedule K, count the total number of pages. Each side of a two-sided form counts as one page.
- 2. FILER NAME: Enter your full name.
- **3. FILER ID:** See instructions for Cover Sheet, page 1, section 1.
- 4. DATE: Enter the date the credit/gain/refund was received or the interest was earned, as applicable.
- **5.** NAME OF PERSON FROM WHOM AMOUNT IS RECEIVED: Enter the full name of the person or business from whom the credit/gain/refund/returned contribution or interest was received.
- 6. ADDRESS OF PERSON FROM WHOM AMOUNT IS RECEIVED: Enter the complete address of the person or business from whom the credit/gain/refund/returned contribution or interest was received.
- 7. PURPOSE FOR WHICH AMOUNT IS RECEIVED: Enter a brief statement or description of the purpose for which the amount was received (for example, "phone service deposit return" "returned contribution" or "interest on savings account").

"Check if political contribution returned to filer" box: If the incoming credit/gain was originally made by you in the form of a political contribution to another candidate or political committee and was returned to you in this reporting period, check this box.

**8. AMOUNT:** Enter the exact dollar amount of the credit/gain/refund/returned contribution, or interest.

## SCHEDULE T: IN-KIND CONTRIBUTIONS OR POLITICAL EXPENDITURES FOR TRAVEL OUTSIDE OF TEXAS

# These instructions are for candidates and officeholders using SCHEDULE T: IN-KIND CONTRIBUTIONS OR POLITICAL EXPENDITURES FOR TRAVEL OUTSIDE OF TEXAS.

Use this schedule to disclose information about contributions accepted or expenditures made during the reporting period. In addition to completing this schedule, you must also report the actual contribution or expenditure on the appropriate schedule or form. The law requires detailed information regarding in-kind contributions or political expenditures for travel outside of the state of Texas.

- 1. TOTAL PAGES SCHEDULE T: After you have completed Schedule T, count the total number of pages. Each side of a two-sided form counts as one page.
- **2. FILER NAME:** Enter the full name of the candidate, committee, or party on whose report you are including this schedule.
- **3. FILER ID:** If you are filing with the Commission, enter your filer account number. If you do not file with the Commission, you are not required to enter a filer account number.
- 4. NAME OF CONTRIBUTOR / CORPORATION OR LABOR ORGANIZATION / PLEDGOR / PAYEE: Enter the full name of the contributor / corporation or labor organization / pledgor / payee as it appears on the schedule or form on which you reported the actual contribution or expenditure.
- 5. **CONTRIBUTION / EXPENDITURE REPORTED ON:** Check the appropriate box for the schedule or form on which you reported the actual contribution or expenditure.
- 6. DATES OF TRAVEL: Enter the dates on which the travel occurred.
- 7. NAME OF PERSON(S) TRAVELING: Enter the full name of the person or persons traveling on whose behalf the travel was accepted or on whose behalf the expenditure was made.
- **8. DEPARTURE CITY OR NAME OF DEPARTURE LOCATION:** Enter the name of the departure city or the name of each departure location.
- **9. DESTINATION CITY OR NAME OF DESTINATION LOCATION:** Enter the name of the destination city or the name of each destination location.
- **10. MEANS OF TRANSPORTATION:** Enter the method of travel (e.g., airplane, bus, boat, car, etc.)
- **11. PURPOSE OF TRAVEL:** Enter the campaign or officeholder purpose of the travel, including the name of a conference, seminar, or other event.

### FORM C/OH-FR: DESIGNATION OF FINAL REPORT

These instructions are for candidates and officeholders using Form C/OH-FR: C/OH REPORT: DESIGNATION OF FINAL REPORT. A final report must include this form (Form C/OH-FR) and the CAMPAIGN FINANCE REPORT (Form C/OH) with the "Final Report" box checked on page 1, section 9. It must also include Schedules A1, A2, B, E, F1, F2, F3, F4, G, H, I, K, and T, as applicable.

### **GENERAL INFORMATION**

For filing purposes, you are a "candidate" as long as you have an appointment of campaign treasurer on file. If you do not expect to accept any further campaign contributions or to make any further campaign expenditures, you may file a final report of contributions and expenditures. A final report terminates your appointment of campaign treasurer and relieves you of the obligation of filing further reports as a candidate.

If you do not have an appointment of campaign treasurer on file, you may not accept *campaign* contributions or make *campaign* expenditures. A payment on a campaign debt is a campaign expenditure. An officeholder who does not have an appointment of campaign treasurer on file may accept *officeholder* contributions and make *officeholder* expenditures.

The effect of filing a final report differs depending on whether you are an officeholder at the time you file a final report.

**Officeholders Filing a Final Report:** You will not have to worry about surplus political funds and assets until you cease to be an officeholder. You may still be required to file semiannual reports of contributions and expenditures as an officeholder. The only officeholders who are not required to file semiannual reports are local officeholders who do not exceed \$940 in contributions or \$940 in expenditures during the reporting period.

If you cease to be an officeholder at a time when you do not have a campaign treasurer appointment on file, and you retain political contributions, interest or other income from political contributions, or assets purchased with political contributions or interest or other income from political contributions after filing the last required report as an officeholder, you *must* file an annual report of unexpended contributions not earlier than January 1 and not later than January 15 of each year following the year in which you filed the last required report as an officeholder. You may not retain these unexpended funds longer than six years after the date you ceased to be an officeholder. For information about important restrictions regarding the use and reporting of unexpended contributions, see the Campaign Finance Guide.

**Non-Officeholders Filing a Final Report:** You will no longer be required to file reports *unless* you retain political contributions, interest or other income from political contributions, or assets purchased with political contributions or interest or other income from political contributions. If you retain any of those items, you must file an annual report of unexpended contributions not earlier than January 1 and not later than January 15 of each year after the year in which you filed your final report. You may not retain these unexpended funds longer than six years after the date of filing a final report. For information about important restrictions regarding the use and reporting of unexpended contributions, see the Campaign Finance Guide.

### **COMPLETING THE FORM**

Each numbered item in these instructions corresponds to the same numbered item on the form.

- **1. C/OH NAME:** Enter your full name.
- **2. FILER ID:** If you are filing with the Commission, enter your Filer ID. If you do not file with the Commission, you are not required to enter a Filer ID.
- **3. SIGNATURE:** You must sign this section to indicate that you understand the consequences of filing a final report.
- 4. FILER WHO IS NOT AN OFFICEHOLDER: Complete this section if you are <u>not</u> an officeholder at the time of filing your final report. Be sure to check the appropriate box in both sections A and B and sign on the "Signature" line.
- **5. OFFICEHOLDER:** Complete this section if you are an officeholder at the time of filing your final report. You must check the box to indicate awareness of further filing requirements.

### ADDITIONAL INFORMATION REGARDING EXPENDITURES

### **EXAMPLES: REPORTING EXPENDITURES MADE BY CREDIT CARD**

This list is for illustrative purposes only. It is intended to provide helpful information and to assist filers in reporting expenditures made by credit card and payments made to credit card companies.

# **Example #1:** Candidate Using Credit Card to Make Political Expenditures and Using Political Contributions to Pay the Credit Card Bill in the Same Reporting Period

A candidate for office uses her credit card to buy \$1,000 in campaign office supplies from an office store. During the same reporting period, the candidate uses her credit card to buy \$500 in political advertising signs from a sign company. During the same reporting period, the candidate makes a single payment from her political contributions account to pay the \$1,500 credit card bill.

To report that activity, the candidate would report all of the following on a campaign finance report (Form C/OH) covering the period in which she made the credit card charges and sent the payment to the credit card company:

- For the credit card charges: a \$1,000 expenditure on the "Expenditures Made by Credit Card" Schedule (F4). The schedule identifies the office store as the payee of the expenditure and includes the address, date, amount, a category of the expenditure as "Office Overhead/Rental Expense," and a description as "Campaign Office Supplies." In Section 9 of the schedule, the box for "Political" is also checked. The candidate also reports the \$500 expenditure on the "Expenditures Made by Credit Card" Schedule and identifies the sign company as the payee of the expenditure and includes the address, date, amount, a category of the expenditure as "Advertising Expense," and a description as "Political Advertising Signs." In Section 9 of the schedule, the box for "Political" is also checked.
- 2. For the payment to the credit card company: a \$1,500 expenditure on the "Political Expenditures from Political Contributions" Schedule (F1). The schedule identifies the credit card company as the payee of the expenditure and includes the address, date, amount, a category of the expenditure as "Credit Card Payment," and a description as "Payment of credit card bill for credit card expenditures."
- 3. Both \$1,500 amounts reported on each schedule will also be included in the appropriate totals sections of Cover Sheet Pages 2 and 3.

# **Example #2:** Candidate Using Credit Card to Make a Political Expenditure and Using Personal Funds to Pay the Credit Card Bill in the Same Reporting Period

A candidate for *non-judicial* office uses his credit card to purchase \$3,000 in political advertising materials from a print shop. During the same reporting period, the candidate makes a payment from his personal funds account to pay the \$3,000 credit card bill.

To report that activity, the candidate would report all of the following on a campaign finance report (Form C/OH) covering the period in which he made the credit card charge and sent the payment to the credit card company:

- For the credit card charge: a \$3,000 expenditure on the "Expenditures Made by Credit Card" Schedule (F4). The schedule identifies the print shop as the payee of the expenditure and includes the address, date, amount, a category of the expenditure as "Advertising Expense," and a description as "Political Advertising Materials." In Section 9 of the schedule, the box for "Political" is also checked.
- 2. For the payment to the credit card company: a \$3,000 expenditure on the "Political Expenditures Made from Personal Funds" Schedule (G). The schedule identifies the credit card company as the payee of the expenditure and includes the address, date, amount, a category of the expenditure as "Credit Card Payment," and a description as "Payment of credit card bill for political advertising materials." If the candidate intends to seek reimbursement from political contributions, the candidate may also check the appropriate box in Section 6.
- 3. Both \$3,000 amounts reported on each schedule will also be included in the appropriate sections of Cover Sheet Pages 2 and 3.

# **Example #3:** Political Committee Using Credit Card to Make a Political Expenditure and Using Political Contributions to Pay the Credit Card Bill in Different Reporting Periods

A general-purpose committee uses its credit card to buy \$500 in political advertising in a newspaper. The committee receives the statement from the credit card company but does not send a payment until after the reporting period ends. When the committee sends a payment to the credit card company, it makes a \$500 payment from its political contributions account.

To report the credit card charge, the committee's campaign treasurer would report all of the following on a campaign finance report (Form GPAC) covering the period in which it made the credit card charge:

- 1. A \$500 expenditure on the "Expenditures Made by Credit Card" Schedule (F4). The schedule identifies the newspaper as the payee of the expenditure and includes the address, date, amount, a category of the expenditure as "Advertising Expense," and a description as "Political Advertising." In Section 9 of the schedule, the box for "Political" is also checked.
- 2. The \$500 amount reported on the "Expenditures Made by Credit Card" Schedule (F4) will also be included in the appropriate sections of Cover Sheet Pages 2 and 3.

To report the payment to the credit card company, the committee's campaign treasurer would also report all of the following on a campaign finance report (Form GPAC) covering the period in which it made the payment to the credit card company:

1. A \$500 expenditure on the "Political Expenditures from Political Contributions" Schedule (F1). The schedule identifies the credit card company as the payee of the expenditure and includes the address, date, amount, a category of the expenditure as "Credit Card Payment," and a description as "Payment of credit card bill for political advertising."

2. The \$500 amount reported on the "Political Expenditures from Political Contributions" Schedule (F1) will also be included in the appropriate sections of Cover Sheet Pages 2 and 3.

# **Example #4: Candidate Using Credit Card to Make a Political Expenditure and Using Political Contributions to Pay the Credit Card Bill in Different Reporting Periods**

A candidate for *judicial* office uses her credit card to buy \$500 in political advertising in a newspaper. The candidate receives the statement from the credit card company but does not send a payment until after the reporting period ends. When the candidate sends a payment to the credit card company, she makes a \$500 payment from her political contributions account.

To report the credit card charge, the candidate would report all of the following on a campaign finance report (Form JC/OH) covering the period in which she made the credit card charge:

- 1. A \$500 expenditure on the "Expenditures Made by Credit Card" Schedule (F4). The schedule identifies the newspaper as the payee of the expenditure and includes the address, date, amount, a category of the expenditure as "Advertising Expense," and a description as "Political Advertising." In Section 9 of the schedule, the box for "Political" is also checked.
- 2. The \$500 amount reported on the "Expenditures Made by Credit Card" Schedule (F4) will also be included in the appropriate sections of Cover Sheet Pages 2 and 3.

To report the payment to the credit card company, the candidate would also report all of the following on a campaign finance report (Form JC/OH) covering the period in which the payment to the credit card company was made:

- 1. A \$500 expenditure on the "Political Expenditures from Political Contributions" Schedule (F1). The schedule identifies the credit card company as the payee of the expenditure and includes the address, date, amount, a category of the expenditure as "Credit Card Payment," and a description as "Payment of credit card bill for political advertising."
- The \$500 amount reported on the "Political Expenditures from Political Contributions" Schedule (F1) will also be included in the appropriate sections of Cover Sheet Pages 2 and 3.

### **EXAMPLES: PURPOSE OF EXPENDITURES**

This list is for illustrative purposes only. It is intended to provide helpful information and to assist filers in reporting the purpose of an expenditure. However, it is not, and is not intended to be, an exhaustive or an exclusive list of how a filer may permissibly report the purpose of an expenditure.

(1) Example: Candidate X is seeking the office of State Representative, District 2000. She purchases an airline ticket from ABC Airlines to attend a campaign rally within District 2000. The acceptable category for this expenditure is "travel in district." The candidate activity that is accomplished by making the expenditure is to attend a campaign rally. An acceptable brief statement is "airline ticket to attend campaign event."

(2) Example: Candidate X purchases an airline ticket to attend a campaign event outside of District 2000 but within Texas, the acceptable category is "travel out of district." The candidate activity that is accomplished by making the expenditure is to attend a campaign event. An acceptable brief statement is "airline ticket to attend campaign or officeholder event."

(3) Example: Candidate X purchases an airline ticket to attend an officeholder related seminar outside of Texas. The acceptable method for the purpose of this expenditure is by selecting the "travel out of district" category and completing the "Schedule T" (used to report travel outside of Texas).

(4) Example: Candidate X contracts with an individual to do various campaign related tasks such as work on a campaign phone bank, sign distribution, and staffing the office. The acceptable category is "salaries/wages/contract labor." The candidate activity that is accomplished by making the expenditure is to compensate an individual working on the campaign. An acceptable brief statement is "contract labor for campaign services."

(5) Example: Officeholder X is seeking re-election and makes an expenditure to purchase a vehicle to use for campaign purposes and permissible officeholder purposes. The acceptable category is "transportation equipment and related expenses" and an acceptable brief description is "purchase of campaign/officeholder vehicle."

(6) Example: Candidate X makes an expenditure to repair a flat tire on a campaign vehicle purchased with political funds. The acceptable category is "transportation equipment and related expenses" and an acceptable brief description is "campaign vehicle repairs."

(7) Example: Officeholder X purchases flowers for a constituent. The acceptable category is "gifts/awards/memorials expense" and an acceptable brief description is "flowers for constituent."

(8) Example: Political Committee XYZ makes a political contribution to Candidate X. The acceptable category is "contributions/donations made by candidate/officeholder/political committee" and an acceptable brief description is "campaign contribution."

(9) Example: Candidate X makes an expenditure for a filing fee to get his name on the ballot. The acceptable category is "fees" and an acceptable brief description is "candidate filing fee."

(10) Example: Officeholder X makes an expenditure to attend a seminar related to performing a duty or engaging in an activity in connection with the office. The acceptable category is "fees" and an acceptable brief description is "attend officeholder seminar."

(11) Example: Candidate X makes an expenditure for political advertising to be broadcast by radio. The acceptable category is "advertising expense" and an acceptable brief description is "political advertising." Similarly, Candidate X makes an expenditure for political advertising to appear in a newspaper. The acceptable category is "advertising expense" and an acceptable brief description is "political advertising."

(12) Example: Officeholder X makes expenditures for printing and postage to mail a letter to all of her constituents, thanking them for their participation during the legislative session. Acceptable categories are "advertising expense" OR "printing expense" and an acceptable brief description is "letter to constituents."

(13) Example: Officeholder X makes an expenditure to pay the campaign office electric bill. The acceptable category is "office overhead/rental expense" and an acceptable brief description is "campaign office electric bill."

(14) Example: Officeholder X makes an expenditure to purchase paper, postage, and other supplies for the campaign office. The acceptable category is "office overhead/rental expense" and an acceptable brief description is "campaign office supplies."

(15) Example: Officeholder X makes an expenditure to pay the campaign office monthly rent. The acceptable category is "office overhead/rental expense" and an acceptable brief description is "campaign office rent."

(16) Example: Candidate X hires a consultant for fundraising services. The acceptable category is "consulting expense" and an acceptable brief description is "campaign services."

(17) Example: Candidate/Officeholder X pays his attorney for legal fees related to either campaign matters or officeholder matters. The acceptable category is "legal services" and an acceptable brief description is "legal fees for campaign" or "for officeholder matters."

(18) Example: Candidate/Officeholder X makes food and beverage expenditures for a meeting with her constituents. The acceptable category is "food/beverage expense" and an acceptable brief statement is "meeting with constituents."

(19) Example: Candidate X makes food and beverage expenditures for a meeting to discuss candidate issues. The acceptable category is "food/beverage expense" and an acceptable brief statement is "meeting to discuss campaign issues."

(20) Example: Officeholder X makes food and beverage expenditures for a meeting to discuss officeholder issues. The acceptable category is "food/beverage expense" and an acceptable brief statement is "meeting to discuss officeholder issues."

(21) Example: Candidate/Officeholder X makes food and beverage expenditures for a meeting to discuss campaign and officeholder issues. The acceptable category is "food/beverage expense" and an acceptable brief statement is "meeting to discuss campaign/officeholder issues."

### **EXAMPLES: REPORTING EXPENDITURES FROM PERSONAL FUNDS**

*This list is for illustrative purposes only. It is intended to provide helpful information and to assist filers in reporting expenditures from personal funds.* 

If you intend to seek reimbursement of any amount from political contributions for a political expenditure made from your personal funds, you must report the expenditure in one of three ways. Keep in mind that this reporting system is not an accounting system and duplication of expenditures is not uncommon when reporting transactions related to expenditures made from personal funds.

**Method #1:** Itemize the expenditure on the "Political Expenditures Made from Personal Funds" schedule (Schedule G) and check the box to indicate that you intend to seek reimbursement from political contributions. You may not correct a report to allow reimbursement without subjecting yourself to a possible penalty. When you reimburse yourself, which could be months or years later, report the reimbursement on the "Political Expenditures" schedule (Schedule F1).

**Example:** On December 1, 2007, Candidate A spends \$500 of her own personal funds to purchase political advertising signs. She reports the expenditure to the vendor on Schedule G and checks the box to indicate that reimbursement is intended. One year later, Candidate A reimburses herself from political contributions. She reports the reimbursement on Schedule F1. Candidate A is the payee and the purpose of the expenditure is to reimburse herself for a political expenditure made from personal funds on December 1, 2007.

If you intend to seek reimbursement from political contributions for a political expenditure of any amount made from personal funds, you must itemize the expenditure on Schedule G.

**Method #2:** Report the political expenditures made from your personal funds as a loan to your campaign on the "Loans" schedule (Schedule E). Next, report the political expenditures made from that loan as if they were made from political funds (report on Schedules F1, F2, F3, F4, or H as appropriate). Do NOT report political expenditures made from the loan on Schedule G.

The amount you report as a loan in a reporting period may NOT exceed the amount you actually spent from personal funds in that reporting period. In other words, do not report a \$100,000 loan to your campaign if the amount actually spent from personal funds in the reporting period was \$5,000. When you reimburse yourself, which could be months or years later, report the reimbursement on the Schedule F1.

**Example:** In one reporting period, Candidate B spends \$5,000 of his own personal funds to purchase political advertising materials. He spends \$3,000 at Business One and \$2,000 at Business Two. He reports the expenditures as a \$5,000 loan on Schedule E and then itemizes each of the two expenditures as a political expenditure on Schedule F1. A year later, Candidate B reimburses himself from political contributions by disclosing the reimbursement on Schedule F1. He reports the reimbursement on Schedule F1. The payee in this instance is Candidate B, the category of the expenditure is "Loan Repayment/Reimbursement," and "political expenditure made from personal funds reported as a loan" is an acceptable brief description.

**Method #3:** Deposit personal funds in an account in which your political contributions are maintained and report that amount as a loan on the "Loans" schedule (Schedule E). Next, report the political expenditures made from that loan as if they were made from political funds (report on Schedules F1, F2, F3, or H as appropriate). When you reimburse yourself, which could be months or years later, report the reimbursement on the Schedule F1. The reimbursement may not exceed the amount reported as a loan. Personal funds deposited in an account in which political contributions are held are subject to the personal use restriction.)

**Example:** In one reporting period, Candidate C opens a campaign bank account and deposits \$5,000 of her own personal funds into the account. She makes one \$3,000 expenditure for political advertising. Candidate C has no other activity in the reporting period. She reports the \$5,000 as a loan on Schedule E, itemizes the \$3,000 expenditure for the political advertising on Schedule F1, and includes the remaining \$2,000 on her contributions maintained at the end of the reporting period total. A year later, Candidate C reimburses herself from political contributions by disclosing the reimbursement on Schedule F1. The payee in this instance is Candidate C, the category of expenditure is "Loan Repayment/Reimbursement," and "political expenditure made from personal funds reported as a loan" is an acceptable brief description.

### **EXAMPLES: REPORTING STAFF REIMBURSEMENT**

This list is for illustrative purposes only. It is intended to provide helpful information and to assist filers in reporting staff reimbursements.

When a staff member makes political payment(s) out of his or her personal funds, how you disclose the payment(s) depends on two things: 1) the aggregate total of those payments in the reporting period; and 2) whether or not you reimburse the staff worker in the same reporting period.

**Example #1:** The payment out of the staff worker's personal funds does not exceed \$5,000 in the reporting period *and* you reimburse the staff worker from political funds in the same reporting period – You will simply itemize the payment (if over the \$190 itemization threshold) on Schedule F1 as if you made the expenditure directly to the vendor out of your political funds, with the name of the vendor who sold the goods or services as the payee for the expenditure. *Do not* disclose as the payee the name of your staff worker.

**Example #2:** The payment(s) out of the staff worker's personal funds are over \$5,000 in the aggregate in the reporting period *and* you reimburse the staff worker from political funds in the same reporting period – You will use a 3-step process, disclosing everything on the same report: (1) On Schedule E, disclose the total amount paid from the staff worker's personal funds as a loan from the staff worker to your campaign; (2) On Schedule F1, itemize the payments made by your staff worker separately, with the names of the vendors who sold the goods or services to your staff worker; and (3) On Schedule F1, disclose the payment to your staff worker for the reimbursement of the loan.

**Example #3:** The payment(s) out of the staff worker's personal funds do not exceed \$5,000 in the aggregate in the reporting period *but* you reimburse the staff worker from political funds in a different reporting period – You will use a 3-step process, disclosing steps 1 and 2 on the same report and step 3 later, when the reimbursement occurs: (1) On Schedule E, disclose the total amount paid from the staff worker's personal funds as a loan from the staff worker to your campaign; (2) On Schedule F1, itemize the payments made by your staff worker separately, with the names of the vendors who sold the goods or services to your staff worker; and (3) When you reimburse your staff worker, if ever, disclose on Schedule F1 of the report covering the period in which the reimbursement occurs the payment to your staff worker for the reimbursement of the loan.

# TITLE 15, ELECTION CODE REGULATING POLITICAL FUNDS AND CAMPAIGNS

9



# TEXAS ETHICS COMMISSION TITLE 15, ELECTION CODE REGULATING POLITICAL FUNDS AND CAMPAIGNS



All Amendments Effective on September 1, 2023

(Revised 8/1/2023)

Texas Ethics Commission, P.O. Box 12070, Austin, Texas 78711 www.ethics.state.tx.us (512) 463-5800 • TDD (800) 735-2989 Promoting Public Confidence in Government

### TITLE 15, ELECTION CODE

### **REGULATING POLITICAL FUNDS AND CAMPAIGNS**

### TABLE OF CONTENTS

| CHAPTER 251. GENERAL PROVISIONS 1                                    |
|--|
| SUBCHAPTER A. GENERAL PROVISIONS1                                    |
| Sec. 251.001. DEFINITIONS  |
| Sec. 251.0015. COMMUNICATION WITH CANDIDATE                          |
| Sec. 251.0016. COMMON VENDOR   |
| Sec. 251.002. OFFICEHOLDERS COVERED                                  |
| Sec. 251.003. PROHIBITION OF DOCUMENT FILING FEE 4                   |
| Sec. 251.004. VENUE  |
| Sec. 251.005. OUT-OF-STATE COMMITTEES EXCLUDED                       |
| Sec. 251.006. FEDERAL OFFICE EXCLUDED 4                              |
| Sec. 251.007. TIMELINESS OF ACTION BY MAIL                           |
| Sec. 251.008. CERTAIN POLITICAL CLUB MEETINGS EXCLUDED               |
| Sec. 251.009. LEGISLATIVE CAUCUS CONTRIBUTION OR EXPENDITURE NOT     |
| CONSIDERED TO BE OFFICEHOLDER CONTRIBUTION OR EXPENDITURE            |
| SUBCHAPTER B. DUTIES OF COMMISSION5                                  |
| Sec. 251.032. FORMS  |
| Sec. 251.033. NOTIFICATION OF DEADLINE FOR FILING REPORTS            |
| CHAPTER 252. CAMPAIGN TREASURER                                      |
| Sec. 252.001. APPOINTMENT OF CAMPAIGN TREASURER REQUIRED7            |
| Sec. 252.0011. INELIGIBILITY FOR APPOINTMENT AS CAMPAIGN TREASURER 7 |
| Sec. 252.002. CONTENTS OF APPOINTMENT7                               |
| Sec. 252.003. CONTENTS OF APPOINTMENT BY GENERAL-PURPOSE             |
| COMMITTEE  |
| Sec. 252.0031. CONTENTS OF APPOINTMENT BY SPECIFIC-PURPOSE           |
| COMMITTEE9   |
| Sec. 252.0032. CONTENTS OF APPOINTMENT BY CANDIDATE                  |
| Sec. 252.004. DESIGNATION OF ONESELF                                 |
| Sec. 252.005. AUTHORITY WITH WHOM APPOINTMENT FILED: CANDIDATE 9     |
| Sec. 252.006. AUTHORITY WITH WHOM APPOINTMENT FILED: SPECIFIC-       |
| PURPOSE COMMITTEE FOR SUPPORTING OR OPPOSING CANDIDATE OR            |
| ASSISTING OFFICEHOLDER 10  |
| Sec. 252.007. AUTHORITY WITH WHOM APPOINTMENT FILED: SPECIFIC-       |
| PURPOSE COMMITTEE FOR SUPPORTING OR OPPOSING MEASURE 10              |
| Sec. 252.008. MULTIPLE FILINGS BY SPECIFIC-PURPOSE COMMITTEE NOT     |
| REQUIRED   |
| Sec. 252.009. AUTHORITY WITH WHOM APPOINTMENT FILED: GENERAL-        |
| PURPOSE COMMITTEE 11   |
| Sec. 252.010. TRANSFER OF APPOINTMENT 11                             |
| Sec. 252.011. TIME APPOINTMENT TAKES EFFECT; PERIOD OF EFFECTIVENESS |
|  |

| Sec. 252.012. REMOVAL OF CAMPAIGN TREASURER  | 11    |
|--|-------|
| Sec. 252.013. TERMINATION OF APPOINTMENT ON VACATING POSITION                              | 11    |
| Sec. 252.0131. TERMINATION OF CAMPAIGN TREASURER APPOINTMENT.                              | 12    |
| Sec. 252.014. PRESERVATION OF FILED APPOINTMENTS   | 12    |
| Sec. 252.015. ASSISTANT CAMPAIGN TREASURER   | 13    |
| CHAPTER 253. RESTRICTIONS ON CONTRIBUTIONS AND EXPENDITURES                                | 14    |
| SUBCHAPTER A. GENERAL RESTRICTIONS   | 14    |
| Sec. 253.001. CONTRIBUTION OR EXPENDITURE IN ANOTHER'S NAME                                |       |
| PROHIBITED   |       |
| Sec. 253.003. UNLAWFULLY MAKING OR ACCEPTING CONTRIBUTION                                  | 14    |
| Sec. 253.004. UNLAWFULLY MAKING EXPENDITURE  |       |
| Sec. 253.005. EXPENDITURE FROM UNLAWFUL CONTRIBUTION                                       | 14    |
| Sec. 253.006. CERTAIN CONTRIBUTIONS AND EXPENDITURES BY LOBBYING                           | STS   |
| RESTRICTED   | 15    |
| Sec. 253.007. PROHIBITION ON LOBBYING BY PERSON MAKING OR                                  |       |
| AUTHORIZING CERTAIN POLITICAL CONTRIBUTIONS AND DIRECT                                     |       |
| CAMPAIGN EXPENDITURES  |       |
| SUBCHAPTER B. CANDIDATES, OFFICEHOLDERS, AND POLIT   | ΓICAL |
| COMMITTEES   | 15    |
| Sec. 253.031. CONTRIBUTION AND EXPENDITURE WITHOUT CAMPAIGN                                |       |
| TREASURER PROHIBITED   |       |
| Sec. 253.032. LIMITATION ON CONTRIBUTION BY OUT-OF-STATE COMMI                             |       |
|  |       |
| Sec. 253.033. CASH CONTRIBUTIONS EXCEEDING \$100 PROHIBITED                                |       |
| Sec. 253.034. RESTRICTIONS ON CONTRIBUTIONS DURING AND FOLLOW                              |       |
| REGULAR LEGISLATIVE SESSION  | 17    |
| Sec. 253.0341. RESTRICTIONS ON CONTRIBUTIONS TO LEGISLATIVE                                |       |
| CAUCUSES DURING AND FOLLOWING REGULAR LEGISLATIVE SESSION.                                 |       |
| Sec. 253.035. RESTRICTIONS ON PERSONAL USE OF CONTRIBUTIONS                                |       |
| Sec. 253.0351. LOANS FROM PERSONAL FUNDS   |       |
| Sec. 253.036. OFFICEHOLDER CONTRIBUTIONS USED IN CONNECTION WI                             |       |
| CAMPAIGN   | 20    |
| Sec. 253.037. RESTRICTIONS ON CONTRIBUTION OR EXPENDITURE BY                               | 20    |
| GENERAL-PURPOSE COMMITTEE  |       |
| Sec. 253.038. PAYMENTS MADE TO PURCHASE REAL PROPERTY OR TO RI                             |       |
| CERTAIN REAL PROPERTY PROHIBITED   |       |
| Sec. 253.039. CONTRIBUTIONS IN CERTAIN PUBLIC BUILDINGS PROHIBIT                           |       |
| Sec. 253.040. SEPARATE ACCOUNTS  |       |
| Sec. 253.041. RESTRICTIONS ON CERTAIN PAYMENTS   |       |
| Sec. 253.042. RESTRICTIONS ON REIMBURSEMENT OF PERSONAL FUNDS                              |       |
| PAYMENTS ON CERTAIN LOANS<br>Sec. 253.043. POLITICAL CONTRIBUTIONS USED IN CONNECTION WITH |       |
|  | 22    |
| APPOINTIVE OFFICE  |       |
| SUBCHAPTER D. CORPORATIONS AND LABOR ORGANIZATIONS   |       |
| Sec. 253.091. CORPORATIONS COVERED   |       |
| Sec. 253.092. TREATMENT OF INCORPORATED POLITICAL COMMITTEE                                |       |

| Sec. 253.094. CONTRIBUTIONS PROHIBITED.       22         Sec. 253.095. PUNISHMENT OF AGENT       23         Sec. 253.096. CONTRIBUTION ON MEASURE.       23         Sec. 253.097. CONTRIBUTION FROM CORPORATION OR LABOR ORGANIZATION       23         Sec. 253.098. COMMUNICATION WITH STOCKHOLDERS OR MEMBERS.       23         Sec. 253.099. NONPARTISAN VOTER REGISTRATION AND GET-OUT-THE-VOTE       23         Sec. 253.100. EXPENDITURES FOR GENERAL-PURPOSE COMMITTEE       23         Sec. 253.101. UNLAWFUL CONTRIBUTION OR EXPENDITURE BY COMMITTEE24       25         Sec. 253.102. COERCION PROHIBITED       25         Sec. 253.103. CONTRIBUTION TO POLITICAL PARTY       25         Sec. 253.104. CONTRIBUTIONS TO DIRECT EXPENDITURE ONLY COMMITTEES       26         SUBCHAPTER E. CIVIL LIABILITY       26         Sec. 253.131. LIABILITY TO CANDIDATES       26         Sec. 253.132. LIABILITY TO STATE       26         Sec. 253.151. APPLICABILITY OF SUBCHAPTER.       27         Sec. 253.152. DEFINITIONS       27         Sec. 253.153. CONTRIBUTION PROHIBITED EXCEPT DURING ELECTION  | Sec. 253.093. CERTAIN ASSOCIATIONS COVERED                      | 22   |
|---|---|------|
| Sec. 253.096. CONTRIBUTION ON MEASURE       23         Sec. 253.097. CONTRIBUTION FROM CORPORATION OR LABOR ORGANIZATION       23         Sec. 253.098. COMMUNICATION WITH STOCKHOLDERS OR MEMBERS       23         Sec. 253.099. NONPARTISAN VOTER REGISTRATION AND GET-OUT-THE-VOTE       23         CAMPAIGNS       23         Sec. 253.100. EXPENDITURES FOR GENERAL-PURPOSE COMMITTEE       23         Sec. 253.101. UNLAWFUL CONTRIBUTION OR EXPENDITURE BY COMMITTEE       23         Sec. 253.102. COERCION PROHIBITED       25         Sec. 253.104. CONTRIBUTION TO POLITICAL PARTY       25         Sec. 253.105. CONTRIBUTIONS TO DIRECT EXPENDITURE ONLY COMMITTEES       26         SUBCHAPTER E. CIVIL LIABILITY       26         Subchapter E. CIVIL DABILITY       26         Subchapter E. CIVIL DABILITY       26         Subchapter E. CIVIL DABILITY TO CANDIDATES       26         Sec. 253.131. LIABILITY TO STATE       26         Sec. 253.132. LIABILITY TO STATE       26         SubCHAPTER F. JUDICIAL CAMPAIGN FAIRNESS ACT       27         Sec. 253.154. CONTRIBUTION PROHIBITED EXCEPT DURING ELECTION PERIOD       27         Sec. 253.155. CONTRIBUTION PROHIBITED EXCEPT DURING ELECTION PERIOD       27         Sec. 253.154. WRITE-IN CANDIDACY       28         Sec. 253.155. CONTRIBUTION PROHIBITED EXCEPT DURING ELECTION   | Sec. 253.094. CONTRIBUTIONS PROHIBITED                          | 22   |
| Sec. 253.097. CONTRIBUTION FROM CORPORATION OR LABOR ORGANIZATION<br>23<br>Sec. 253.098. COMMUNICATION WITH STOCKHOLDERS OR MEMBERS   | Sec. 253.095. PUNISHMENT OF AGENT                               | 23   |
| 23       Sec. 253.098. COMMUNICATION WITH STOCKHOLDERS OR MEMBERS.       23         Sec. 253.099. NONPARTISAN VOTER REGISTRATION AND GET-OUT-THE-VOTE       23         Sec. 253.100. EXPENDITURES FOR GENERAL-PURPOSE COMMITTEE.       23         Sec. 253.101. UNLAWFUL CONTRIBUTION OR EXPENDITURE BY COMMITTEE.4       25         Sec. 253.102. COERCION PROHIBITED.       25         Sec. 253.103. CORPORATE LOANS.       25         Sec. 253.104. CONTRIBUTION TO POLITICAL PARTY.       25         Sec. 253.105. CONTRIBUTIONS TO DIRECT EXPENDITURE ONLY COMMITTEES       26         Sec. 253.131. LIABILITY TO CANDIDATES       26         Sec. 253.132. LIABILITY TO CANDIDATES       26         Sec. 253.133. LIABILITY TO STATE       26         Sec. 253.134. CIVIL PENALTIES IMPOSED BY COMMISSION       26         SubchApter F, Judicial CAMPAIGN FAIRNESS ACT.       27         Sec. 253.151. APPLICABILITY OF SUBCHAPTER       27         Sec. 253.152. DEFINITIONS       27         Sec. 253.153. CONTRIBUTION PROHIBITED EXCEPT DURING ELECTION PERIOD       27         Sec. 253.154. WRITE-IN CANDIDACY       28         Sec. 253.155. CONTRIBUTION LIMITS       28         Sec. 253.155. CONTRIBUTION NEW SPOUSE OR CHILD.       30         Sec. 253.159. EXCEPTION TO CONTRIBUTION BY GENERAL-PURPOSE COMMITTEES       29         Se  | Sec. 253.096. CONTRIBUTION ON MEASURE                           | 23   |
| Sec. 253.098. COMMUNICATION WITH STOCKHOLDERS OR MEMBERS.       23         Sec. 253.099. NONPARTISAN VOTER REGISTRATION AND GET-OUT-THE-VOTE       23         CAMPAIGNS       23         Sec. 253.100. EXPENDITURES FOR GENERAL-PURPOSE COMMITTEE.       23         Sec. 253.101. UNLAWFUL CONTRIBUTION OR EXPENDITURE BY COMMITTEE24       25         Sec. 253.102. COERCION PROHIBITED.       25         Sec. 253.103. CORPORATE LOANS.       25         Sec. 253.104. CONTRIBUTION TO POLITICAL PARTY.       25         Subchapter E. CIVIL LIABILITY       26         Sec. 253.131. LIABILITY TO CANDIDATES.       26         Sec. 253.132. LIABILITY TO CANDIDATES.       26         Sec. 253.133. LIABILITY TO POLITICAL COMMITTEES       26         Sec. 253.134. CIVIL PENALTIES IMPOSED BY COMMISSION       26         SubCHAPTER F. JUDICIAL CAMPAIGN FAIRNESS ACT.       27         Sec. 253.151. APPLICABILITY OF SUBCHAPTER       27         Sec. 253.152. DEFINITIONS       27         Sec. 253.153. CONTRIBUTION PROHIBITED EXCEPT DURING ELECTION PERIOD       27         Sec. 253.154. WRITE-IN CANDIDACY.       28         Sec. 253.155. CONTRIBUTION PROHIBITED EXCEPT DURING ELECTION PERIOD       27         Sec. 253.154. WRITE-IN CANDIDACY.       28         Sec. 253.155. CONTRIBUTION NEW GENERAL-PURPOSE COMMITTEES       29 <td>Sec. 253.097. CONTRIBUTION FROM CORPORATION OR LABOR ORGANIZAT</td> <td>ΓΙΟΝ</td>   | Sec. 253.097. CONTRIBUTION FROM CORPORATION OR LABOR ORGANIZAT  | ΓΙΟΝ |
| Sec. 253.099. NONPARTISAN VOTER REGISTRATION AND GET-OUT-THE-VOTE       CAMPAIGNS       23         Sec. 253.100. EXPENDITURES FOR GENERAL-PURPOSE COMMITTEE       23         Sec. 253.101. UNLAWFUL CONTRIBUTION OR EXPENDITURE BY COMMITTEE44       25         Sec. 253.102. COERCION PROHIBITED       25         Sec. 253.103. CORPORATE LOANS       25         Sec. 253.104. CONTRIBUTION TO POLITICAL PARTY       25         Sec. 253.105. CONTRIBUTION TO POLITICAL PARTY       25         SubchApter E. CIVIL LIABILITY       26         Sec. 253.131. LIABILITY TO CANDIDATES       26         Sec. 253.132. LIABILITY TO CANDIDATES       26         Sec. 253.133. LIABILITY TO STATE       26         Sec. 253.134. CIVIL PENALTIES IMPOSED BY COMMISSION       26         SubCHAPTER F. JUDICIAL CAMPAIGN FAIRNESS ACT       27         Sec. 253.151. APPLICABILITY OF SUBCHAPTER       27         Sec. 253.152. DEFINITIONS       27         Sec. 253.154. WRITE-IN CANDIDACY       28         Sec. 253.154. WRITE-IN CANDIDACY       28         Sec. 253.154. WRITE-IN CANDIDACY       28         Sec. 253.155. CONTRIBUTION LIMITS       28         Sec. 253.154. WRITE-IN CANDIDACY       28         Sec. 253.155. CONTRIBUTION LIMITS       28         Sec. 253.155. CONTRIBUTION LIMITS   |   |      |
| CAMPAIGNS23Sec. 253.100. EXPENDITURES FOR GENERAL-PURPOSE COMMITTEE23Sec. 253.101. UNLAWFUL CONTRIBUTION OR EXPENDITURE BY COMMITTEE2425Sec. 253.102. COERCION PROHIBITED.25Sec. 253.103. CORPORATE LOANS25Sec. 253.104. CONTRIBUTION TO POLITICAL PARTY.25Sec. 253.105. CONTRIBUTIONS TO DIRECT EXPENDITURE ONLY COMMITTEES26Suberta Statistic Contrained and the cont |   |      |
| Sec. 253.100.       EXPENDITURES FOR GENERAL-PURPOSE COMMITTEE       23         Sec. 253.101.       UNLAWFUL CONTRIBUTION OR EXPENDITURE BY COMMITTEE24         Sec. 253.102.       COERCION PROHIBITED       25         Sec. 253.103.       CORPORATE LOANS       25         Sec. 253.104.       CONTRIBUTION TO POLITICAL PARTY       25         Sec. 253.105.       CONTRIBUTIONS TO DIRECT EXPENDITURE ONLY COMMITTEES       25         SUBCHAPTER E.       CIVIL LIABILITY       26         Sec. 253.131.       LIABILITY TO CANDIDATES       26         Sec. 253.132.       LIABILITY TO POLITICAL COMMITTEES       26         Sec. 253.133.       LIABILITY TO STATE       26         Sec. 253.151.       APPLICABILITY OF SUBCHAPTER       27         Sec. 253.151.       APPLICABILITY OF SUBCHAPTER       27         Sec. 253.152.       DEFINITIONS       27         Sec. 253.153.       CONTRIBUTION PROHIBITED EXCEPT DURING ELECTION PERIOD       27         Sec. 253.154.       WRITE-IN CANDIDACY       28         Sec. 253.154.       MURTE-IN CANDIDACY       28         Sec. 253.154.       ACCEPTANCE OF POLITICAL CONTRIBUTIONS BY PERSON       29         Sec. 253.157.       LIMIT ON CONTRIBUTION BY GENERAL-PURPOSE COMMITTEES       28         Sec. 25  |   |      |
| Sec. 253.101.       UNLAWFUL CONTRIBUTION OR EXPENDITURE BY COMMITTEE24         Sec. 253.102.       COERCION PROHIBITED.       25         Sec. 253.103.       CORPORATE LOANS.       25         Sec. 253.104.       CONTRIBUTION TO POLITICAL PARTY.       25         Sec. 253.105.       CONTRIBUTIONS TO DIRECT EXPENDITURE ONLY COMMITTEES       25         SUBCHAPTER E.       CIVIL LIABILITY       26         Sec. 253.131.       LIABILITY TO CANDIDATES.       26         Sec. 253.132.       LIABILITY TO POLITICAL COMMITTEES       26         Sec. 253.133.       LIABILITY TO POLITICAL COMMITTEES       26         Sec. 253.134.       CIVIL PENALTIES IMPOSED BY COMMISSION       26         SubCHAPTER F.       JUDICIAL CAMPAIGN FAIRNESS ACT       27         Sec. 253.151.       APPLICABILITY OF SUBCHAPTER       27         Sec. 253.152.       DEFINITIONS       27         Sec. 253.154.       ACTEPTANCE OF POLITICAL CONTRIBUTIONS BY PERSON       28         Sec. 253.155.       CONTRIBUTION LIMITS       28         Sec. 253.154.       WRITE-IN CANDIDACY       28         Sec. 253.155.       CONTRIBUTION LIMITS       28         Sec. 253.157.       LIMIT ON CONTRIBUTION BY GENERAL-PURPOSE COMMITTEES       29         Sec. 253.1601.   |   |      |
| Sec. 253.102. COERCION PROHIBITED.25Sec. 253.103. CORPORATE LOANS.25Sec. 253.104. CONTRIBUTION TO POLITICAL PARTY.25Sec. 253.105. CONTRIBUTIONS TO DIRECT EXPENDITURE ONLY COMMITTEES25SUBCHAPTER E. CIVIL LIABILITY.26Sec. 253.131. LIABILITY TO CANDIDATES.26Sec. 253.132. LIABILITY TO POLITICAL COMMITTEES26Sec. 253.133. LIABILITY TO POLITICAL COMMITTEES26Sec. 253.134. CIVIL PENALTIES IMPOSED BY COMMISSION26SUBCHAPTER F. JUDICIAL CAMPAIGN FAIRNESS ACT.27Sec. 253.151. APPLICABILITY OF SUBCHAPTER27Sec. 253.152. DEFINITIONS27Sec. 253.153. CONTRIBUTION PROHIBITED EXCEPT DURING ELECTION PERIOD27Sec. 253.154. WRITE-IN CANDIDACY28Sec. 253.155. CONTRIBUTION PROHIBITED EXCEPT DURING ELECTION PERIOD28Sec. 253.154. ACCEPTANCE OF POLITICAL CONTRIBUTIONS BY PERSON28APPOINTED TO FILL VACANCY28Sec. 253.155. CONTRIBUTION LIMITS28Sec. 253.154. ILIMIT ON CONTRIBUTION BY GENERAL-PURPOSE COMMITTEES2929Sec. 253.155. CONTRIBUTION BY SPOUSE OR CHILD.30Sec. 253.156. CONTRIBUTION NO CONTRIBUTION COMMITTEES CONSIDERED29CONTRIBUTION TO CANDIDATE OR OFFICEHOLDER30Sec. 253.161. CONTRIBUTION TO CERTAIN COMMITTEES CONSIDERED30CONTRIBUTION TO CANDIDATE OR OFFICEHOLDER30Sec. 253.161. USE OF CONTRIBUTION S BY JUDICIAL CANDIDATES,30OFFICE PROHIBITED30Sec. 253.161. CERTAIN CONTRIBUTIONS BY JUDICIAL CANDIDATES,31  |   |      |
| Sec. 253.103. CORPORATE LOANS25Sec. 253.104. CONTRIBUTION TO POLITICAL PARTY25Sec. 253.105. CONTRIBUTIONS TO DIRECT EXPENDITURE ONLY COMMITTEES26Subcrassing26Subcrassing26Sec. 253.131. LIABILITY TO CANDIDATES26Sec. 253.132. LIABILITY TO POLITICAL COMMITTEES26Sec. 253.133. LIABILITY TO POLITICAL COMMITTEES26Sec. 253.134. CIVIL PENALTIES IMPOSED BY COMMISSION26SUBCHAPTER F. JUDICIAL CAMPAIGN FAIRNESS ACT27Sec. 253.151. APPLICABILITY OF SUBCHAPTER27Sec. 253.152. DEFINITIONS27Sec. 253.153. CONTRIBUTION PROHIBITED EXCEPT DURING ELECTION PERIOD27Sec. 253.154. WRITE-IN CANDIDACY28Sec. 253.155. CONTRIBUTION PROHIBITED EXCEPT DURING BY PERSON28APPOINTED TO FILL VACANCY28Sec. 253.157. LIMIT ON CONTRIBUTION BY GENERAL-PURPOSE COMMITTEES29Sec. 253.158. CONTRIBUTION BY SPOUSE OR CHILD.30Sec. 253.159. EXCEPTION TO CONTRIBUTION LIMITS30Sec. 253.161. USE OF CONTRIBUTION BY SPOUSE OR CHILD.30Sec. 253.161. USE OF CONTRIBUTION FROM NONJUDICIAL OR JUDICIAL30Sec. 253.161. USE OF CONTRIBUTION FROM NONJUDICIAL CANDIDATES,30OFFICE PROHIBITED30Sec. 253.161. CERTAIN COMMITTEES RESTRICTED31Sec. 253.1612. CERTAIN CAMPAIGN ACTIVITIES AUTORIZED.31   |   |      |
| Sec. 253.104. CONTRIBUTION TO POLITICAL PARTY.25Sec. 253.105. CONTRIBUTIONS TO DIRECT EXPENDITURE ONLY COMMITTEES25SUBCHAPTER E. CIVIL LIABILITY26Sec. 253.131. LIABILITY TO CANDIDATES26Sec. 253.132. LIABILITY TO POLITICAL COMMITTEES26Sec. 253.133. LIABILITY TO POLITICAL COMMITTEES26Sec. 253.134. CIVIL PENALTIES IMPOSED BY COMMISSION26SUBCHAPTER F. JUDICIAL CAMPAIGN FAIRNESS ACT27Sec. 253.151. APPLICABILITY OF SUBCHAPTER27Sec. 253.152. DEFINITIONS27Sec. 253.153. CONTRIBUTION PROHIBITED EXCEPT DURING ELECTION PERIOD27Sec. 253.154. WRITE-IN CANDIDACY28Sec. 253.155. CONTRIBUTION PROHIBITED EXCEPT DURING BY PERSON29APPOINTED TO FILL VACANCY28Sec. 253.157. LIMIT ON CONTRIBUTION BY GENERAL-PURPOSE COMMITTEES20Sec. 253.158. CONTRIBUTION BY SPOUSE OR CHILD30Sec. 253.159. EXCEPTION TO CONTRIBUTION LIMITS30Sec. 253.1601. CONTRIBUTION BY SPOUSE OR CHILD30Sec. 253.1611. USE OF CONTRIBUTION FROM NONJUDICIAL OR JUDICIAL30Sec. 253.1611. USE OF CONTRIBUTION FROM NONJUDICIAL OR JUDICIAL30Sec. 253.1611. CERTAIN CONTRIBUTIONS BY JUDICIAL CANDIDATES,30OFFICEHOLDERS, AND COMMITTEES RESTRICTED31Sec. 253.1612. CERTAIN CAMPAIGN ACTIVITIES AUTORIZED31   |   |      |
| Sec. 253.105. CONTRIBUTIONS TO DIRECT EXPENDITURE ONLY COMMITTEES<br>25<br>SUBCHAPTER E. CIVIL LIABILITY  |   |      |
| 25SUBCHAPTER E. CIVIL LIABILITY26Sec. 253.131. LIABILITY TO CANDIDATES26Sec. 253.132. LIABILITY TO POLITICAL COMMITTEES26Sec. 253.133. LIABILITY TO STATE26Sec. 253.134. CIVIL PENALTIES IMPOSED BY COMMISSION26SUBCHAPTER F. JUDICIAL CAMPAIGN FAIRNESS ACT27Sec. 253.151. APPLICABILITY OF SUBCHAPTER27Sec. 253.152. DEFINITIONS27Sec. 253.153. CONTRIBUTION PROHIBITED EXCEPT DURING ELECTION PERIOD27Sec. 253.154. WRITE-IN CANDIDACYSec. 253.154. WRITE-IN CANDIDACY28Sec. 253.155. CONTRIBUTION PROHIBITED EXCEPT DURING ELECTION PERIOD27Sec. 253.154. WRITE-IN CANDIDACYSec. 253.155. CONTRIBUTION LIMITS28Sec. 253.157. LIMIT ON CONTRIBUTION BY GENERAL-PURPOSE COMMITTEES29Sec. 253.158. CONTRIBUTION LIMITS29Sec. 253.159. EXCEPTION TO CONTRIBUTION BY GENERAL-PURPOSE COMMITTEES20CONTRIBUTION TO CERTAIN COMMITTEES CONSIDEREDCONTRIBUTION TO CANDIDATE OR OFFICEHOLDER30Sec. 253.161. CONTRIBUTION TO CERTAIN COMMITTEES CONSIDEREDCONTRIBUTION TO CANDIDATE OR OFFICEHOLDER30Sec. 253.161. USE OF CONTRIBUTION FROM NONJUDICIAL OR JUDICIALOFFICE PROHIBITED30Sec. 253.1611. CERTAIN CONTRIBUTIONS BY JUDICIAL CANDIDATES,OFFICEHOLDERS, AND COMMITTEES RESTRICTED31Sec. 253.1612. CERTAIN CAMPAIGN ACTIVITIES AUTORIZED31   |   |      |
| SUBCHAPTER E. CIVIL LIABILITY26Sec. 253.131. LIABILITY TO CANDIDATES26Sec. 253.132. LIABILITY TO POLITICAL COMMITTEES26Sec. 253.133. LIABILITY TO STATE26Sec. 253.134. CIVIL PENALTIES IMPOSED BY COMMISSION26SUBCHAPTER F. JUDICIAL CAMPAIGN FAIRNESS ACT27Sec. 253.151. APPLICABILITY OF SUBCHAPTER27Sec. 253.152. DEFINITIONS27Sec. 253.153. CONTRIBUTION PROHIBITED EXCEPT DURING ELECTION PERIOD27Sec. 253.154. WRITE-IN CANDIDACY28Sec. 253.155. CONTRIBUTION PROHIBITED EXCEPT DURING BY PERSON28APPOINTED TO FILL VACANCY28Sec. 253.155. CONTRIBUTION LIMITS28Sec. 253.156. CONTRIBUTION LIMITS28Sec. 253.157. LIMIT ON CONTRIBUTION BY GENERAL-PURPOSE COMMITTEES29Sec. 253.158. CONTRIBUTION BY SPOUSE OR CHILD30Sec. 253.159. EXCEPTION TO CONTRIBUTION BY GENERAL-PURPOSE COMMITTEES30Sec. 253.1601. CONTRIBUTION BY SPOUSE OR CHILD30Sec. 253.1611. CERTAIN CONTRIBUTION FROM NONJUDICIAL OR JUDICIAL30Sec. 253.1611. CERTAIN CONTRIBUTIONS BY JUDICIAL CANDIDATES,30OFFICEHOLDERS, AND COMMITTEES RESTRICTED31Sec. 253.1612. CERTAIN CAMPAIGN ACTIVITIES AUTORIZED31  |   |      |
| Sec. 253.131. LIABILITY TO CANDIDATES26Sec. 253.132. LIABILITY TO POLITICAL COMMITTEES26Sec. 253.133. LIABILITY TO STATE26Sec. 253.134. CIVIL PENALTIES IMPOSED BY COMMISSION26SUBCHAPTER F. JUDICIAL CAMPAIGN FAIRNESS ACT27Sec. 253.151. APPLICABILITY OF SUBCHAPTER27Sec. 253.152. DEFINITIONS27Sec. 253.153. CONTRIBUTION PROHIBITED EXCEPT DURING ELECTION PERIOD27Sec. 253.154. WRITE-IN CANDIDACY28Sec. 253.155. CONTRIBUTION PROHIBITED EXCEPT DURING BY PERSONAPPOINTED TO FILL VACANCY28Sec. 253.155. CONTRIBUTION LIMITS28Sec. 253.157. LIMIT ON CONTRIBUTION BY GENERAL-PURPOSE COMMITTEESSec. 253.158. CONTRIBUTION BY SPOUSE OR CHILD30Sec. 253.159. EXCEPTION TO CONTRIBUTION LIMITS30Sec. 253.1601. CONTRIBUTION BY SPOUSE OR CHILD30Sec. 253.1611. USE OF CONTRIBUTION FROM NONJUDICIAL OR JUDICIAL30Sec. 253.1611. CERTAIN CONTRIBUTION FROM NONJUDICIAL OR JUDICIAL30Sec. 253.1611. CERTAIN CONTRIBUTIONS BY JUDICIAL CANDIDATES,31OFFICEHOLDERS, AND COMMITTEES RESTRICTED31Sec. 253.1612. CERTAIN CAMPAIGN ACTIVITIES AUTORIZED31  |   |      |
| Sec. 253.132.LIABILITY TO POLITICAL COMMITTEES26Sec. 253.133.LIABILITY TO STATE26Sec. 253.134.CIVIL PENALTIES IMPOSED BY COMMISSION26SUBCHAPTER F. JUDICIAL CAMPAIGN FAIRNESS ACT27Sec. 253.151.APPLICABILITY OF SUBCHAPTER27Sec. 253.152.DEFINITIONS27Sec. 253.153.CONTRIBUTION PROHIBITED EXCEPT DURING ELECTION PERIOD27Sec. 253.154.WRITE-IN CANDIDACY28Sec. 253.155.CONTRIBUTION EOF POLITICAL CONTRIBUTIONS BY PERSON28APPOINTED TO FILL VACANCY28Sec. 253.155.CONTRIBUTION LIMITS28Sec. 253.157.LIMIT ON CONTRIBUTION BY GENERAL-PURPOSE COMMITTEES29Sec. 253.158.CONTRIBUTION BY SPOUSE OR CHILD.30Sec. 253.1601.CONTRIBUTION TO CERTAIN COMMITTEES CONSIDERED30Sec. 253.1611.USE OF CONTRIBUTION FROM NONJUDICIAL OR JUDICIAL30Sec. 253.1611.CERTAIN CONTRIBUTIONS BY JUDICIAL CANDIDATES,30Sec. 253.1612.CERTAIN CAMPAIGN ACTIVITIES AUTORIZED31  |   |      |
| Sec. 253.133. LIABILITY TO STATE26Sec. 253.134. CIVIL PENALTIES IMPOSED BY COMMISSION26SUBCHAPTER F. JUDICIAL CAMPAIGN FAIRNESS ACT27Sec. 253.151. APPLICABILITY OF SUBCHAPTER27Sec. 253.152. DEFINITIONS27Sec. 253.153. CONTRIBUTION PROHIBITED EXCEPT DURING ELECTION PERIOD27Sec. 253.154. WRITE-IN CANDIDACY28Sec. 253.155. CONTRIBUTION PROHIBITED EXCEPT DURING BY PERSONAPPOINTED TO FILL VACANCY28Sec. 253.155. CONTRIBUTION LIMITS28Sec. 253.157. LIMIT ON CONTRIBUTION BY GENERAL-PURPOSE COMMITTEES29Sec. 253.158. CONTRIBUTION BY SPOUSE OR CHILD30Sec. 253.1601. CONTRIBUTION TO CONTRIBUTION LIMITS30Sec. 253.1611. CONTRIBUTION TO CERTAIN COMMITTEES CONSIDERED30Sec. 253.1611. USE OF CONTRIBUTION FROM NONJUDICIAL OR JUDICIAL30Sec. 253.1611. CERTAIN CONTRIBUTIONS BY JUDICIAL CANDIDATES,30OFFICE PROHIBITED31Sec. 253.1612. CERTAIN CAMPAIGN ACTIVITIES AUTORIZED31   |   |      |
| Sec. 253.134. CIVIL PENALTIES IMPOSED BY COMMISSION26SUBCHAPTER F. JUDICIAL CAMPAIGN FAIRNESS ACT27Sec. 253.151. APPLICABILITY OF SUBCHAPTER27Sec. 253.152. DEFINITIONS27Sec. 253.153. CONTRIBUTION PROHIBITED EXCEPT DURING ELECTION PERIOD27Sec. 253.154. WRITE-IN CANDIDACY28Sec. 253.154. WRITE-IN CANDIDACY28Sec. 253.154. ACCEPTANCE OF POLITICAL CONTRIBUTIONS BY PERSONAPPOINTED TO FILL VACANCY28Sec. 253.155. CONTRIBUTION LIMITS28Sec. 253.157. LIMIT ON CONTRIBUTION BY GENERAL-PURPOSE COMMITTEES29Sec. 253.158. CONTRIBUTION BY SPOUSE OR CHILD30Sec. 253.159. EXCEPTION TO CONTRIBUTION LIMITS30Sec. 253.1601. CONTRIBUTION BY SPOUSE OR CHILD30Sec. 253.161. USE OF CONTRIBUTION FROM NONJUDICIAL OR JUDICIAL30Sec. 253.161. USE OF CONTRIBUTION FROM NONJUDICIAL OR JUDICIAL30Sec. 253.161. CERTAIN CONTRIBUTIONS BY JUDICIAL CANDIDATES,30OFFICE PROHIBITED31Sec. 253.1612. CERTAIN CAMPAIGN ACTIVITIES AUTORIZED31   |   |      |
| SUBCHAPTER F. JUDICIAL CAMPAIGN FAIRNESS ACT  |   |      |
| Sec. 253.151. APPLICABILITY OF SUBCHAPTER.27Sec. 253.152. DEFINITIONS.27Sec. 253.153. CONTRIBUTION PROHIBITED EXCEPT DURING ELECTION PERIOD27Sec. 253.154. WRITE-IN CANDIDACY.28Sec. 253.154. ACCEPTANCE OF POLITICAL CONTRIBUTIONS BY PERSONAPPOINTED TO FILL VACANCY.28Sec. 253.155. CONTRIBUTION LIMITS28Sec. 253.157. LIMIT ON CONTRIBUTION BY GENERAL-PURPOSE COMMITTEES29Sec. 253.158. CONTRIBUTION BY SPOUSE OR CHILD.30Sec. 253.159. EXCEPTION TO CONTRIBUTION LIMITS.30Sec. 253.161. CONTRIBUTION TO CERTAIN COMMITTEES CONSIDERED30Sec. 253.161. USE OF CONTRIBUTION FROM NONJUDICIAL OR JUDICIAL30Sec. 253.161. USE OF CONTRIBUTION S BY JUDICIAL CANDIDATES,30Sec. 253.161. CERTAIN CONTRIBUTIONS BY JUDICIAL CANDIDATES,30Sec. 253.161. CERTAIN CONTRIBUTIONS BY JUDICIAL CANDIDATES,31Sec. 253.161. CERTAIN CAMPAIGN ACTIVITIES AUTORIZED31   |   |      |
| Sec. 253.152. DEFINITIONS27Sec. 253.153. CONTRIBUTION PROHIBITED EXCEPT DURING ELECTION PERIOD27Sec. 253.154. WRITE-IN CANDIDACY28Sec. 253.154. WRITE-IN CANDIDACY28Sec. 253.154. ACCEPTANCE OF POLITICAL CONTRIBUTIONS BY PERSONAPPOINTED TO FILL VACANCY28Sec. 253.155. CONTRIBUTION LIMITS28Sec. 253.157. LIMIT ON CONTRIBUTION BY GENERAL-PURPOSE COMMITTEES29Sec. 253.158. CONTRIBUTION BY SPOUSE OR CHILD30Sec. 253.1601. CONTRIBUTION TO CONTRIBUTION LIMITS30Sec. 253.1611. CONTRIBUTION TO CERTAIN COMMITTEES CONSIDERED30Sec. 253.1611. CERTAIN CONTRIBUTION FROM NONJUDICIAL OR JUDICIAL30Sec. 253.1611. CERTAIN CONTRIBUTIONS BY JUDICIAL CANDIDATES,31Sec. 253.1612. CERTAIN CAMPAIGN ACTIVITIES AUTORIZED31   |   |      |
| Sec. 253.153. CONTRIBUTION PROHIBITED EXCEPT DURING ELECTION PERIOD27Sec. 253.154. WRITE-IN CANDIDACY.28Sec. 253.154. ACCEPTANCE OF POLITICAL CONTRIBUTIONS BY PERSONAPPOINTED TO FILL VACANCY.28Sec. 253.155. CONTRIBUTION LIMITS28Sec. 253.157. LIMIT ON CONTRIBUTION BY GENERAL-PURPOSE COMMITTEES29Sec. 253.158. CONTRIBUTION BY SPOUSE OR CHILD.30Sec. 253.1601. CONTRIBUTION TO CONTRIBUTION LIMITS.30Sec. 253.1611. CONTRIBUTION TO CERTAIN COMMITTEES CONSIDEREDCONTRIBUTION TO CANDIDATE OR OFFICEHOLDER30Sec. 253.161. USE OF CONTRIBUTION FROM NONJUDICIAL OR JUDICIALOFFICE PROHIBITED30Sec. 253.1611. CERTAIN CONTRIBUTIONS BY JUDICIAL CANDIDATES,OFFICEHOLDERS, AND COMMITTEES RESTRICTED31Sec. 253.1612. CERTAIN CAMPAIGN ACTIVITIES AUTORIZED31  |   |      |
| 27Sec. 253.154. WRITE-IN CANDIDACY  |   |      |
| Sec. 253.154. WRITE-IN CANDIDACY.28Sec. 253.1541. ACCEPTANCE OF POLITICAL CONTRIBUTIONS BY PERSON28APPOINTED TO FILL VACANCY.28Sec. 253.155. CONTRIBUTION LIMITS28Sec. 253.157. LIMIT ON CONTRIBUTION BY GENERAL-PURPOSE COMMITTEES29Sec. 253.158. CONTRIBUTION BY SPOUSE OR CHILD.30Sec. 253.159. EXCEPTION TO CONTRIBUTION LIMITS30Sec. 253.1601. CONTRIBUTION TO CERTAIN COMMITTEES CONSIDERED30Sec. 253.1611. USE OF CONTRIBUTION FROM NONJUDICIAL OR JUDICIAL30Sec. 253.1611. USE OF CONTRIBUTION FROM NONJUDICIAL OR JUDICIAL30Sec. 253.1611. CERTAIN CONTRIBUTIONS BY JUDICIAL CANDIDATES,31Sec. 253.1612. CERTAIN CAMPAIGN ACTIVITIES AUTORIZED31   |   | -    |
| Sec. 253.1541. ACCEPTANCE OF POLITICAL CONTRIBUTIONS BY PERSON<br>APPOINTED TO FILL VACANCY   |   |      |
| APPOINTED TO FILL VACANCY28Sec. 253.155. CONTRIBUTION LIMITS28Sec. 253.157. LIMIT ON CONTRIBUTION BY GENERAL-PURPOSE COMMITTEES   |   | 20   |
| Sec. 253.155. CONTRIBUTION LIMITS28Sec. 253.157. LIMIT ON CONTRIBUTION BY GENERAL-PURPOSE COMMITTEES29Sec. 253.158. CONTRIBUTION BY SPOUSE OR CHILD.30Sec. 253.159. EXCEPTION TO CONTRIBUTION LIMITS30Sec. 253.1601. CONTRIBUTION TO CERTAIN COMMITTEES CONSIDERED30CONTRIBUTION TO CANDIDATE OR OFFICEHOLDER30Sec. 253.161. USE OF CONTRIBUTION FROM NONJUDICIAL OR JUDICIAL30OFFICE PROHIBITED30Sec. 253.161. CERTAIN CONTRIBUTIONS BY JUDICIAL CANDIDATES,30OFFICEHOLDERS, AND COMMITTEES RESTRICTED31Sec. 253.1612. CERTAIN CAMPAIGN ACTIVITIES AUTORIZED31   |   | 28   |
| Sec. 253.157. LIMIT ON CONTRIBUTION BY GENERAL-PURPOSE COMMITTEES29Sec. 253.158. CONTRIBUTION BY SPOUSE OR CHILD.30Sec. 253.159. EXCEPTION TO CONTRIBUTION LIMITS.30Sec. 253.1601. CONTRIBUTION TO CERTAIN COMMITTEES CONSIDEREDCONTRIBUTION TO CANDIDATE OR OFFICEHOLDER30Sec. 253.161. USE OF CONTRIBUTION FROM NONJUDICIAL OR JUDICIALOFFICE PROHIBITED30Sec. 253.1611. CERTAIN CONTRIBUTIONS BY JUDICIAL CANDIDATES,OFFICEHOLDERS, AND COMMITTEES RESTRICTED31Sec. 253.1612. CERTAIN CAMPAIGN ACTIVITIES AUTORIZED.31   |   |      |
| 29<br>Sec. 253.158. CONTRIBUTION BY SPOUSE OR CHILD   |   |      |
| Sec. 253.158. CONTRIBUTION BY SPOUSE OR CHILD.30Sec. 253.159. EXCEPTION TO CONTRIBUTION LIMITS30Sec. 253.1601. CONTRIBUTION TO CERTAIN COMMITTEES CONSIDERED30CONTRIBUTION TO CANDIDATE OR OFFICEHOLDER30Sec. 253.161. USE OF CONTRIBUTION FROM NONJUDICIAL OR JUDICIAL30Sec. 253.1611. CERTAIN CONTRIBUTIONS BY JUDICIAL CANDIDATES,30OFFICEHOLDERS, AND COMMITTEES RESTRICTED31Sec. 253.1612. CERTAIN CAMPAIGN ACTIVITIES AUTORIZED.31  |   |      |
| Sec. 253.159. EXCEPTION TO CONTRIBUTION LIMITS30Sec. 253.1601. CONTRIBUTION TO CERTAIN COMMITTEES CONSIDERED30CONTRIBUTION TO CANDIDATE OR OFFICEHOLDER30Sec. 253.161. USE OF CONTRIBUTION FROM NONJUDICIAL OR JUDICIAL30OFFICE PROHIBITED30Sec. 253.1611. CERTAIN CONTRIBUTIONS BY JUDICIAL CANDIDATES,31OFFICEHOLDERS, AND COMMITTEES RESTRICTED31Sec. 253.1612. CERTAIN CAMPAIGN ACTIVITIES AUTORIZED31  |   |      |
| Sec. 253.1601. CONTRIBUTION TO CERTAIN COMMITTEES CONSIDERED<br>CONTRIBUTION TO CANDIDATE OR OFFICEHOLDER30Sec. 253.161. USE OF CONTRIBUTION FROM NONJUDICIAL OR JUDICIAL<br>OFFICE PROHIBITED30Sec. 253.1611. CERTAIN CONTRIBUTIONS BY JUDICIAL CANDIDATES,<br>OFFICEHOLDERS, AND COMMITTEES RESTRICTED31Sec. 253.1612. CERTAIN CAMPAIGN ACTIVITIES AUTORIZED31  |   |      |
| Sec. 253.161. USE OF CONTRIBUTION FROM NONJUDICIAL OR JUDICIAL<br>OFFICE PROHIBITED30Sec. 253.1611. CERTAIN CONTRIBUTIONS BY JUDICIAL CANDIDATES,<br>OFFICEHOLDERS, AND COMMITTEES RESTRICTED31Sec. 253.1612. CERTAIN CAMPAIGN ACTIVITIES AUTORIZED31   |   |      |
| OFFICE PROHIBITED30Sec. 253.1611. CERTAIN CONTRIBUTIONS BY JUDICIAL CANDIDATES,OFFICEHOLDERS, AND COMMITTEES RESTRICTED31Sec. 253.1612. CERTAIN CAMPAIGN ACTIVITIES AUTORIZED31   | CONTRIBUTION TO CANDIDATE OR OFFICEHOLDER                       | 30   |
| OFFICE PROHIBITED30Sec. 253.1611. CERTAIN CONTRIBUTIONS BY JUDICIAL CANDIDATES,OFFICEHOLDERS, AND COMMITTEES RESTRICTED31Sec. 253.1612. CERTAIN CAMPAIGN ACTIVITIES AUTORIZED31   | Sec. 253.161. USE OF CONTRIBUTION FROM NONJUDICIAL OR JUDICIAL  |      |
| OFFICEHOLDERS, AND COMMITTEES RESTRICTED  |   | 30   |
| OFFICEHOLDERS, AND COMMITTEES RESTRICTED  | Sec. 253.1611. CERTAIN CONTRIBUTIONS BY JUDICIAL CANDIDATES,    |      |
| Sec. 253.1612. CERTAIN CAMPAIGN ACTIVITIES AUTORIZED  | OFFICEHOLDERS, AND COMMITTEES RESTRICTED                        | 31   |
|   | Sec. 253.1612. CERTAIN CAMPAIGN ACTIVITIES AUTORIZED            | 31   |
| Sec. 253.162. RESTRICTIONS ON REIMBURSEMENT OF PERSONAL FUNDS AND   | Sec. 253.162. RESTRICTIONS ON REIMBURSEMENT OF PERSONAL FUNDS A |      |
| PAYMENTS ON CERTAIN LOANS   |   |      |
| Sec. 253.1621. APPLICATION OF CONTRIBUTION AND REIMBURSEMENT  |   |      |
|   | LIMITS TO CERTAIN CANDIDATES                                    | 32   |
| T TRATTES TEATED TO A INTERNATION AT TASE   | LIMITS TO CERTAIN CANDIDATES                                    | 32   |

| Sec. 253.167. CERTIFICATION OF POPULATION; NOTICE OF CONTRIBUTION   |      |
|---|------|
| LIMITS  | . 32 |
| Sec. 253.171. CONTRIBUTION FROM OR DIRECT CAMPAIGN EXPENDITURE B    | SY   |
| POLITICAL PARTY   | . 32 |
| Sec. 253.176. CIVIL PENALTY   | . 33 |
| CHAPTER 254. POLITICAL REPORTING                                    | . 34 |
| SUBCHAPTER A. RECORDKEEPING   | . 34 |
| Sec. 254.001. RECORDKEEPING REQUIRED                                | . 34 |
| SUBCHAPTER B. POLITICAL REPORTING GENERALLY                         | . 34 |
| Sec. 254.031. GENERAL CONTENTS OF REPORTS                           | . 34 |
| Sec. 254.0311. REPORT BY LEGISLATIVE CAUCUS                         | . 35 |
| Sec. 254.0312. BEST EFFORTS   | . 36 |
| Sec. 254.0313. OMISSION OF ADDRESS FOR JUDGE AND                    | . 37 |
| Sec. 254.032. NONREPORTABLE PERSONAL TRAVEL EXPENSE                 |      |
| Sec. 254.033. NONREPORTABLE PERSONAL SERVICE                        | . 37 |
| Sec. 254.034. TIME OF ACCEPTING CONTRIBUTION                        |      |
| Sec. 254.035. TIME OF MAKING EXPENDITURE                            |      |
| Sec. 254.036. FORM OF REPORT; AFFIDAVIT; MAILING OF FORMS           | . 38 |
| Sec. 254.0362. USE OF PUBLICLY ACCESSIBLE COMPUTER TERMINAL FOR     |      |
| PREPARATION OF REPORTS  | . 40 |
| Sec. 254.037. FILING DEADLINE                                       |      |
| Sec. 254.038. SPECIAL REPORT NEAR ELECTION BY CERTAIN CANDIDATES    |      |
| AND POLITICAL COMMITTEES  | . 40 |
| Sec. 254.039. SPECIAL REPORT NEAR ELECTION BY CERTAIN GENERAL-      |      |
| PURPOSE COMMITTEES  | . 41 |
| Sec. 254.0391. REPORT DURING SPECIAL LEGISLATIVE SESSION            | . 42 |
| Sec. 254.040. PRESERVATION OF REPORTS; RECORD OF INSPECTION         |      |
| Sec. 254.0401. AVAILABILITY OF REPORTS ON INTERNET                  |      |
| Sec. 254.04011. Repealed  |      |
| Sec. 254.0402. PUBLIC INSPECTION OF REPORTS                         |      |
| Sec. 254.0405. AMENDMENT OF FILED REPORT                            |      |
| Sec. 254.041. CRIMINAL PENALTY FOR UNTIMELY OR INCOMPLETE REPORT    |      |
| Sec. 254.042. CIVIL PENALTY FOR LATE REPORT                         | . 44 |
| Sec. 254.043. ACTION TO REQUIRE COMPLIANCE                          |      |
| Sec. 254.044. REPORTING OF POLITICAL CONTRIBUTIONS AND EXPENDITURES | 5    |
| MADE USING CREDIT CARD  |      |
| SUBCHAPTER C. REPORTING BY CANDIDATE                                | . 45 |
| Sec. 254.061. ADDITIONAL CONTENTS OF REPORTS                        |      |
| Sec. 254.0611. ADDITIONAL CONTENTS OF REPORTS BY CERTAIN JUDICIAL   |      |
| CANDIDATES  | . 46 |
| Sec. 254.0612. ADDITIONAL CONTENTS OF REPORTS BY CANDIDATE FOR      |      |
| STATEWIDE EXECUTIVE OFFICE OR LEGISLATIVE OFFICE                    | . 46 |
| Sec. 254.062. CERTAIN OFFICEHOLDER ACTIVITY INCLUDED                |      |
| Sec. 254.063. SEMIANNUAL REPORTING SCHEDULE FOR CANDIDATE           |      |
| Sec. 254.064. ADDITIONAL REPORTS OF OPPOSED CANDIDATE               |      |
| Sec. 254.065. FINAL REPORT.   |      |

| Sec. 254.066. AUTHORITY WITH WHOM REPORTS FILED                    | . 48 |
|--|------|
| SUBCHAPTER D. REPORTING BY OFFICEHOLDER                            | 48   |
| Sec. 254.091. ADDITIONAL CONTENTS OF REPORTS                       | . 48 |
| Sec. 254.0911. ADDITIONAL CONTENTS OF REPORTS BY CERTAIN JUDICIAL  |      |
| OFFICEHOLDERS  | . 48 |
| Sec. 254.0912. ADDITIONAL CONTENTS OF REPORTS BY STATEWIDE         |      |
| EXECUTIVE OFFICEHOLDERS AND LEGISLATIVE OFFICEHOLDERS              | . 48 |
| Sec. 254.092. CERTAIN OFFICEHOLDER EXPENDITURES EXCLUDED           | . 48 |
| Sec. 254.093. SEMIANNUAL REPORTING SCHEDULE FOR OFFICEHOLDER       | . 48 |
| Sec. 254.094. REPORT FOLLOWING APPOINTMENT OF CAMPAIGN TREASURE    | R    |
|  |      |
| Sec. 254.095. REPORT NOT REQUIRED                                  |      |
| Sec. 254.096. OFFICEHOLDER WHO BECOMES CANDIDATE                   |      |
| Sec. 254.097. AUTHORITY WITH WHOM REPORTS FILED                    |      |
| SUBCHAPTER E. REPORTING BY SPECIFIC-PURPOSE COMMITTEE              |      |
| Sec. 254.121. ADDITIONAL CONTENTS OF REPORTS                       | . 49 |
| Sec. 254.1211. ADDITIONAL CONTENTS OF REPORTS OF CERTAIN           |      |
| COMMITTEES   | . 50 |
| Sec. 254.1212. ADDITIONAL CONTENTS OF REPORTS OF COMMITTEE         |      |
| SUPPORTING OR OPPOSING CANDIDATE FOR STATEWIDE EXECUTIVE           |      |
| OFFICEHOLDERS OR LEGISLATIVE OFFICEHOLDERS OR ASSISTING            |      |
| STATEWIDE EXECUTIVE OFFICEHOLDERS OR LEGISLATIVE OFFICEHOLDER      | RS   |
|  | . 50 |
| Sec. 254.122. INVOLVEMENT IN MORE THAN ONE ELECTION BY CERTAIN     |      |
| COMMITTEES   | . 50 |
| Sec. 254.123. SEMIANNUAL REPORTING SCHEDULE FOR COMMITTEE          | . 50 |
| Sec. 254.124. ADDITIONAL REPORTS OF COMMITTEE FOR SUPPORTING OR    |      |
| OPPOSING CANDIDATE OR MEASURE                                      |      |
| Sec. 254.125. FINAL REPORT OF COMMITTEE FOR SUPPORTING OR OPPOSING | 3    |
| CANDIDATE OR MEASURE   | . 51 |
| Sec. 254.126. DISSOLUTION REPORT OF COMMITTEE FOR ASSISTING        |      |
| OFFICEHOLDER   |      |
| Sec. 254.127. TERMINATION REPORT                                   | . 52 |
| Sec. 254.128. NOTICE TO CANDIDATE AND OFFICEHOLDER OF              |      |
| CONTRIBUTIONS AND EXPENDITURES                                     |      |
| Sec. 254.129. NOTICE OF CHANGE IN COMMITTEE STATUS                 | . 52 |
| Sec. 254.130. AUTHORITY WITH WHOM REPORTS FILED                    |      |
| SUBCHAPTER F. REPORTING BY GENERAL-PURPOSE COMMITTEE               | . 53 |
| Sec. 254.151. ADDITIONAL CONTENTS OF REPORTS                       |      |
| Sec. 254.152. TIME FOR REPORTING CERTAIN EXPENDITURES              |      |
| Sec. 254.153. SEMIANNUAL REPORTING SCHEDULE FOR COMMITTEE          |      |
| Sec. 254.154. ADDITIONAL REPORTS OF COMMITTEE INVOLVED IN ELECTIO  | Ν    |
|  | . 54 |
| Sec. 254.1541. ALTERNATE REPORTING REQUIREMENTS FOR CERTAIN        |      |
| COMMITTEES   |      |
| Sec. 254.155. OPTION TO FILE MONTHLY; NOTICE                       | . 55 |

| Sec. 254.156. CONTENTS OF MONTHLY REPORTS                            |      |
|--|------|
| Sec. 254.157. MONTHLY REPORTING SCHEDULE                             |      |
| Sec. 254.158. EXCEPTION TO MONTHLY REPORTING SCHEDULE                | . 56 |
| Sec. 254.1581. REPORTING BY OUT-OF-STATE POLITICAL COMMITTEE         | . 56 |
| Sec. 254.159. DISSOLUTION REPORT                                     |      |
| Sec. 254.160. TERMINATION REPORT                                     | . 56 |
| Sec. 254.161. NOTICE TO CANDIDATE AND OFFICEHOLDER OF                |      |
| CONTRIBUTIONS AND EXPENDITURES                                       | . 56 |
| Sec. 254.162. NOTICE OF CHANGE IN COMMITTEE STATUS                   | . 56 |
| Sec. 254.163. AUTHORITY WITH WHOM REPORTS FILED                      | . 56 |
| Sec. 254.164. CERTAIN COMMITTEES EXEMPT FROM CIVIL PENALTIES         | . 56 |
| SUBCHAPTER G. MODIFIED REPORTING PROCEDURES; \$500 MAXIMUM           | IN   |
| CONTRIBUTIONS OR EXPENDITURES  |      |
| Sec. 254.181. MODIFIED REPORTING AUTHORIZED                          | . 57 |
| Sec. 254.182. DECLARATION OF INTENT REQUIRED                         |      |
| Sec. 254.183. MAXIMUM EXCEEDED                                       |      |
| Sec. 254.184. APPLICABILITY OF REGULAR REPORTING REQUIREMENTS        | . 57 |
| SUBCHAPTER H. UNEXPENDED CONTRIBUTIONS                               |      |
| Sec. 254.201. ANNUAL REPORT OF UNEXPENDED CONTRIBUTIONS              | . 58 |
| Sec. 254.202. FILING OF REPORT; CONTENTS                             |      |
| Sec. 254.203. RETENTION OF CONTRIBUTIONS                             |      |
| Sec. 254.204. DISPOSITION OF UNEXPENDED CONTRIBUTIONS                |      |
| Sec. 254.205. REPORT OF DISPOSITION OF UNEXPENDED CONTRIBUTIONS      |      |
| SUBCHAPTER I. CIVIL LIABILITY  |      |
| Sec. 254.231. LIABILITY TO CANDIDATES                                |      |
| Sec. 254.232. LIABILITY TO STATE                                     |      |
| SUBCHAPTER J. REPORTING BY CERTAIN PERSONS MAKING DIRE               |      |
| CAMPAIGN EXPENDITURES  | . 60 |
| Sec. 254.261. DIRECT CAMPAIGN EXPENDITURE EXCEEDING \$100            |      |
| Sec. 254.262. TRAVEL EXPENSE   |      |
| CHAPTER 255. REGULATING POLITICAL ADVERTISING AND CAMPAIO            |      |
| COMMUNICATIONS   |      |
| Sec. 255.001. REQUIRED DISCLOSURE ON POLITICAL ADVERTISING           |      |
| Sec. 255.002. RATES FOR POLITICAL ADVERTISING                        | . 61 |
| Sec. 255.003. UNLAWFUL USE OF PUBLIC FUNDS FOR POLITICAL             |      |
| ADVERTISING  |      |
| Sec. 255.0031. UNLAWFUL USE OF INTERNAL MAIL SYSTEM FOR POLITICAL    |      |
| ADVERTISING.   |      |
| Sec. 255.004. TRUE SOURCE OF COMMUNICATION                           |      |
| Sec. 255.005. MISREPRESENTATION OF IDENTITY                          |      |
| Sec. 255.006. MISLEADING USE OF OFFICE TITLE                         |      |
| Sec. 255.008. DISCLOSURE ON POLITICAL ADVERTISING FOR JUDICIAL OFFIC |      |
|  |      |
| CHAPTER 257. POLITICAL PARTIES                                       |      |
| Sec. 257.001. PRINCIPAL POLITICAL COMMITTEE OF POLITICAL PARTY       | . 65 |
| Sec. 257.002. REQUIREMENTS RELATING TO CORPORATE OR LABOR UNION      |      |

| CONTRIBUTIONS  | 65 |
|--|----|
| Sec. 257.003. REPORT REQUIRED                                      | 65 |
| Sec. 257.004. RESTRICTIONS ON CONTRIBUTIONS BEFORE GENERAL         |    |
| ELECTION   | 65 |
| Sec. 257.005. CANDIDATE FOR STATE OR COUNTY CHAIR OF POLITICAL     |    |
| PARTY  | 65 |
| Sec. 257.006. CRIMINAL PENALTY FOR FAILURE TO COMPLY               | 66 |
| Sec. 257.007. RULES  | 66 |
| CHAPTER 258. FAIR CAMPAIGN PRACTICES                               | 67 |
| Sec. 258.001. SHORT TITLE  |    |
| Sec. 258.002. PURPOSE  | 67 |
| Sec. 258.003. DELIVERY OF COPY OF CODE                             | 67 |
| Sec. 258.004. TEXT OF CODE   | 67 |
| Sec. 258.005. FORMS  | 68 |
| Sec. 258.006. ACCEPTANCE AND PRESERVATION OF COPIES                | 68 |
| Sec. 258.007. SUBSCRIPTION TO CODE VOLUNTARY                       | 68 |
| Sec. 258.008. INDICATION ON POLITICAL ADVERTISING                  | 68 |
| Sec. 258.009. CIVIL CAUSE OF ACTION                                |    |
| CHAPTER 259. POLITICAL SIGNS                                       | 69 |
| Sec. 259.001. NOTICE REQUIREMENT ON POLITICAL ADVERTISING SIGNS    | 69 |
| Sec. 259.002. REGULATION OF DISPLAY OF POLITICAL SIGNS BY PROPERTY |    |
| OWNERS' ASSOCIATIONS   | 69 |
| Sec. 259.003. REGULATION OF POLITICAL SIGNS BY MUNICIPALITY        | 70 |

#### **ELECTION CODE**

#### TITLE 15. REGULATING POLITICAL FUNDS AND CAMPAIGNS

#### **CHAPTER 251. GENERAL PROVISIONS**

#### SUBCHAPTER A. GENERAL PROVISIONS

#### Sec. 251.001. DEFINITIONS. In this title:

(1) "Candidate" means a person who knowingly and willingly takes affirmative action for the purpose of gaining nomination or election to public office or for the purpose of satisfying financial obligations incurred by the person in connection with the campaign for nomination or election. Examples of affirmative action include:

(A) the filing of a campaign treasurer appointment, except that the filing does not constitute candidacy or an announcement of candidacy for purposes of the automatic resignation provisions of Article XVI, Section 65, or Article XI, Section 11, of the Texas Constitution;

(B) the filing of an application for a place on a ballot;

(C) the filing of an application for nomination by convention;

(D) the filing of a declaration of intent to become an independent candidate or a declaration of write-in candidacy;

(E) the making of a public announcement of a definite intent to run for public office in a particular election, regardless of whether the specific office is mentioned in the announcement;

(F) before a public announcement of intent, the making of a statement of definite intent to run for public office and the soliciting of support by letter or other mode of communication;

(G) the soliciting or accepting of a campaign contribution or the making of a campaign expenditure; and

(H) the seeking of the nomination of an executive committee of a political party to fill a vacancy.

(2) "Contribution" means a direct or indirect transfer of money, goods, services, or any other thing of value and includes an agreement made or other obligation incurred, whether legally enforceable or not, to make a transfer. The term includes a loan or extension of credit, other than those expressly excluded by this subdivision, and a guarantee of a loan or extension of credit, including a loan described by this subdivision. The term does not include:

(A) a loan made in the due course of business by a corporation that is legally engaged in the business of lending money and that has conducted the business continuously for more than one year before the loan is made; or

(B) an expenditure required to be reported under Section 305.006(b), Government Code.

(3) "Campaign contribution" means a contribution to a candidate or political committee that is offered or given with the intent that it be used in connection with a campaign for elective office or on a measure. Whether a contribution is made before, during, or after an election does not affect its status as a campaign contribution.

(4) "Officeholder contribution" means a contribution to an officeholder or political committee that is offered or given with the intent that it be used to defray expenses that:

(A) are incurred by the officeholder in performing a duty or engaging in an activity in connection with the office; and

(B) are not reimbursable with public money.

(5) "Political contribution" means a campaign contribution or an officeholder contribution.

(6) "Expenditure" means a payment of money or any other thing of value and includes an agreement made or other obligation incurred, whether legally enforceable or not, to make a payment.

(7) "Campaign expenditure" means an expenditure made by any person in connection with a campaign for an elective office or on a measure. Whether an expenditure is made before, during, or after an election does not affect its status as a campaign expenditure.

(8) "Direct campaign expenditure" means a campaign expenditure that does not constitute a campaign contribution by the person making the expenditure. A campaign expenditure does not constitute a contribution by the person making the expenditure to a candidate or officeholder if the expenditure is made without the prior consent or approval of the candidate or officeholder on whose behalf the expenditure is made. A campaign expenditure made in connection with a measure does not constitute a contribution by the person making the expenditure of the expenditure if it is not made as a political contribution to a political committee supporting or opposing the measure.

(9) "Officeholder expenditure" means an expenditure made by any person to defray expenses that:

(A) are incurred by an officeholder in performing a duty or engaging in an activity in connection with the office; and

(B) are not reimbursable with public money.

(10) "Political expenditure" means a campaign expenditure or an officeholder expenditure.

(11) "Reportable activity" means a political contribution, political expenditure, or other activity required to be reported under this title.

(12) "Political committee" means two or more persons acting in concert with a principal purpose of accepting political contributions or making political expenditures. The term does not include a group composed exclusively of two or more individual filers or political committees required to file reports under this title who make reportable expenditures for a joint activity.

(13) "Specific-purpose committee" means a political committee that does not have among its principal purposes those of a general-purpose committee but does have among its principal purposes:

(A) supporting or opposing one or more:

(i) candidates, all of whom are identified and are seeking offices

that are known; or

(ii) measures, all of which are identified;

(B) assisting one or more officeholders, all of whom are identified; or

(C) supporting or opposing only one candidate who is unidentified or who is seeking an office that is unknown.

(14) "General-purpose committee" means a political committee that has among its principal purposes:

(A) supporting or opposing:

offices that are unknown; or

- (i) two or more candidates who are unidentified or are seeking
- (ii) one or more measures that are unidentified; or

(B) assisting two or more officeholders who are unidentified.

(15) "Out-of-state political committee" means a political committee that:

(A) makes political expenditures outside this state; and

(B) in the 12 months immediately preceding the making of a political expenditure by the committee inside this state (other than an expenditure made in connection with a campaign for a federal office or made for a federal officeholder), makes 80 percent or more of the committee's total political expenditures in any combination of elections outside this state and federal offices not voted on in this state.

(16) "Political advertising" means a communication supporting or opposing a candidate for nomination or election to a public office or office of a political party, a political party, a public officer, or a measure that:

(A) in return for consideration, is published in a newspaper, magazine, or other periodical or is broadcast by radio or television; or

(B) appears:

(i) in a pamphlet, circular, flier, billboard or other sign, bumper sticker, or similar form of written communication; or

(ii) on an Internet website.

(17) "Campaign communication" means a written or oral communication relating to a campaign for nomination or election to public office or office of a political party or to a campaign on a measure.

(18) "Labor organization" means an agency, committee, or any other organization in which employees participate that exists for the purpose, in whole or in part, of dealing with employers concerning grievances, labor disputes, wages, rates of pay, hours of employment, or conditions of work.

(19) "Measure" means a question or proposal submitted in an election for an expression of the voters' will and includes the circulation and submission of a petition to determine whether a question or proposal is required to be submitted in an election for an expression of the voters' will.

(20) "Commission" means the Texas Ethics Commission.

(21) "In-kind contribution" means a contribution of goods, services, or any other thing of value that is not money, and includes an agreement made or other obligation incurred, whether legally enforceable or not, to make the contribution. The term does not include a direct campaign expenditure.

**Sec. 251.0015. COMMUNICATION WITH CANDIDATE**. For purposes of Section 251.001(8), communication between a person and a candidate, officeholder, or candidate's or officeholder's agent is not evidence that the person obtained the candidate's or officeholder's consent or approval for a campaign expenditure made after the communication by the person on behalf of the candidate or officeholder unless the communication establishes that:

(1) the expenditure is incurred at the request or suggestion of the candidate, officeholder, or candidate's or officeholder's agent;

(2) the candidate, officeholder, or candidate's or officeholder's agent is materially involved in decisions regarding the creation, production, or distribution of a campaign communication related to the expenditure; or (3) the candidate, officeholder, or candidate's or officeholder's agent shares information about the candidate's or officeholder's plans or needs that is:

(A) material to the creation, production, or distribution of a campaign communication related to the expenditure; and

(B) not available to the public.

Sec. 251.0016. COMMON VENDOR. A person using the same vendor as a candidate, officeholder, or political committee established or controlled by a candidate or officeholder is not acting in concert with the candidate, officeholder, or committee to make a campaign expenditure unless the person makes the expenditure using information from the vendor about the campaign plans or needs of the candidate, officeholder, or committee that is:

(1) material to the expenditure; and

(2) not available to the public.

**Sec. 251.002. OFFICEHOLDERS COVERED**. (a) The provisions of this title applicable to an officeholder apply only to a person who holds an elective public office and to the secretary of state.

(b) For purposes of this title, a state officer-elect or a member-elect of the legislature is considered an officeholder beginning on the day after the date of the general or special election at which the officer-elect or member-elect was elected. This subsection does not relieve a state officer-elect or member-elect of the legislature of any reporting requirements the person may have as a candidate under this title.

Sec. 251.003. PROHIBITION OF DOCUMENT FILING FEE. A charge may not be made for filing a document required to be filed under this title.

**Sec. 251.004. VENUE**. (a) Venue for a criminal offense prescribed by this title is in the county of residence of the defendant, unless the defendant is not a Texas resident, in which case venue is in Travis County.

(b) Venue for the recovery of delinquent civil penalties imposed by the commission under this title is in Travis County.

#### Sec. 251.005. OUT-OF-STATE COMMITTEES EXCLUDED.

(a) An out-of-state political committee is not subject to Chapter 252 or 254, except as provided by Subsection (b), (c), or (d).

(b) If an out-of-state committee decides to file a campaign treasurer appointment under Chapter 252, at the time the appointment is filed the committee becomes subject to this title to the same extent as a political committee that is not an out-of-state committee.

(c) If an out-of-state committee performs an activity that removes the committee from out-of-state status as defined by Section 251.001(15), the committee becomes subject to this title to the same extent as a political committee that is not an out-of-state committee.

(d) An out-of-state political committee that does not file a campaign treasurer appointment shall comply with Section 254.1581.

#### Sec. 251.006. FEDERAL OFFICE EXCLUDED.

(a) Except as provided by Subsection (b), this title does not apply to a candidate for an office of the federal government.

(b) A candidate for an elective office of the federal government shall file with the commission a copy of each document relating to the candidacy that is required to be filed under federal law. The document shall be filed within the same period in which it is required to be filed under the federal law.

Sec. 251.007. TIMELINESS OF ACTION BY MAIL. When this title requires a notice, report, or other document or paper to be delivered, submitted, or filed within a specified period or before a specified deadline, a delivery, submission, or filing by first-class United States mail or common or contract carrier is timely, except as otherwise provided by this title, if:

(1) it is properly addressed with postage or handling charges prepaid; and

(2) it bears a post office cancellation mark or a receipt mark of a common or contract carrier indicating a time within the period or before the deadline, or if the person required to take the action furnishes satisfactory proof that it was deposited in the mail or with a common or contract carrier within the period or before the deadline.

#### Sec. 251.008. CERTAIN POLITICAL CLUB MEETINGS EXCLUDED.

(a) An expense incurred in connection with the conduct of a meeting of an organization or club affiliated with a political party at which a candidate for an office regularly filled at the general election for state and county officers, or a person holding that office, appears before the members of the organization or club is not considered to be a political contribution or political expenditure if no political contributions are made to or solicited for the candidate or officeholder at the meeting.

(b) In this section, an organization or club is affiliated with a political party if it:

(1) supports the nominees of that political party but does not support any candidate seeking the party's nomination for an office over any other candidate seeking that nomination; and

(2) is recognized by the political party as an auxiliary of the party.

Sec. 251.009. LEGISLATIVE CAUCUS CONTRIBUTION OR EXPENDITURE NOT CONSIDERED TO BE OFFICEHOLDER CONTRIBUTION OR EXPENDITURE. A contribution to or expenditure by a legislative caucus, as defined by Section 253.0341, is not considered to be an officeholder contribution or officeholder expenditure for purposes of this title.

#### SUBCHAPTER B. DUTIES OF COMMISSION

Sec. 251.032. FORMS. In addition to furnishing samples of the appropriate forms to the authorities having administrative duties under this title, the commission shall furnish the forms to each political party's state executive committee and county chair of each county executive committee.

#### Sec. 251.033. NOTIFICATION OF DEADLINE FOR FILING REPORTS.

(a) The commission shall notify each person responsible for filing a report with the commission under Subchapters C through F, Chapter 254, of the deadline for filing a report, except that notice of the deadline is not required for a political committee involved in an election other than a primary election or the general election for state and county officers. Notification under this subsection may be sent by electronic mail.

(b) If the commission is unable to notify a person of a deadline after two attempts, the commission is not required to make any further attempts to notify the person of that deadline or any future deadlines until the person has notified the commission of the person's current address or electronic mail address.

(c) Chapter 552, Government Code, does not apply to a notification under this section sent by electronic mail.

#### CHAPTER 252. CAMPAIGN TREASURER

Sec. 252.001. APPOINTMENT OF CAMPAIGN TREASURER REQUIRED. Each candidate and each political committee shall appoint a campaign treasurer as provided by this chapter.

# Sec. 252.0011. INELIGIBILITY FOR APPOINTMENT AS CAMPAIGN TREASURER.

(a) Except as provided by Subsection (b) or (c), a person is ineligible for appointment as a campaign treasurer if the person is the campaign treasurer of a political committee that does not file a report required by Chapter 254.

(b) The period for which a person is ineligible under Subsection (a) for appointment as a campaign treasurer ends on the date on which the political committee in connection with which the person's ineligibility arose has filed each report required by Chapter 254 that was not timely filed or has paid all fines and penalties in connection with the failure to file the report.

(c) Subsection (a) does not apply to a person if, in any semiannual reporting period prescribed by Chapter 254:

(1) the political committee in connection with which the person's ineligibility arose did not accept political contributions that in the aggregate exceed \$5,000 or make political expenditures that in the aggregate exceed \$5,000; and

(2) the candidate who or political committee that subsequently appoints the person does not accept political contributions that in the aggregate exceed \$5,000 or make political expenditures that in the aggregate exceed \$5,000.

(d) Subsection (c) applies to a person who is the campaign treasurer of a generalpurpose committee regardless of whether the committee files monthly reports under Section 254.155. For purposes of this subsection, political contributions accepted and political expenditures made during a monthly reporting period are aggregated with political contributions accepted and political expenditures made in each other monthly reporting period that corresponds to the semiannual reporting period that contains those months.

(e) A candidate or political committee is considered to have not appointed a campaign treasurer if the candidate or committee appoints a person as campaign treasurer whose appointment is prohibited by Subsection (a).

(f) A person who violates this section is liable for a civil penalty not to exceed three times the amount of political contributions accepted or political expenditures made in violation of this section.

#### Sec. 252.002. CONTENTS OF APPOINTMENT.

(a) A campaign treasurer appointment must be in writing and include:

- (1) the campaign treasurer's name;
- (2) the campaign treasurer's residence or business street address;
- (3) the campaign treasurer's telephone number; and
- (4) the name of the person making the appointment.

(b) A political committee that files its campaign treasurer appointment with the commission must notify the commission in writing of any change in the campaign treasurer's address not later than the 10th day after the date on which the change occurs.

# Sec. 252.003. CONTENTS OF APPOINTMENT BY GENERAL-PURPOSE COMMITTEE.

(a) In addition to the information required by Section 252.002, a campaign treasurer appointment by a general-purpose committee must include:

(1) the full name, and any acronym of the name that will be used in the name of the committee as provided by Subsection (d), of each corporation, labor organization, or other association or legal entity that directly establishes, administers, or controls the committee, if applicable, or the name of each person who determines to whom the committee makes contributions or the name of each person who determines for what purposes the committee makes expenditures;

(2) the full name and address of each general-purpose committee to whom the committee intends to make political contributions;

(3) the name of the committee and, if the name is an acronym, the words the acronym represents; and

(4) before the committee may use a political contribution from a corporation or a labor organization to make a direct campaign expenditure in connection with a campaign for an elective office, an affidavit stating that:

(A) the committee is not established or controlled by a candidate or an officeholder; and

(B) the committee will not use any political contribution from a corporation or a labor organization to make a political contribution to:

(i) a candidate for elective office;

(ii) an officeholder; or

(iii) a political committee that has not filed an affidavit under this subdivision or Section 252.0031(a)(2).

(a-1) Filing an affidavit under Subsection (a)(4) does not create any additional reporting requirements under Section 254.261.

(b) If any of the information required to be included in a general-purpose committee's appointment changes, excluding changes reported under Section 252.002(b), the committee shall file an amended appointment with the commission not later than the 30th day after the date the change occurs.

(c) The name of a general-purpose committee may not be the same as or deceptively similar to the name of any other general-purpose committee whose campaign treasurer appointment is filed with the commission. The commission shall determine whether the name of a general-purpose political committee is in violation of this prohibition and shall immediately notify the campaign treasurer of the offending political committee of that determination. The campaign treasurer of the political committee must file a name change with the commission not later than the 14th day after the date of notification. A campaign treasurer who fails to file a name change as provided by this subsection or a political committee that continues to use a prohibited name after its campaign treasurer has been notified by the commission commits an offense. An offense under this subsection is a Class B misdemeanor.

(d) The name of a general-purpose committee must include the name of each corporation, labor organization, or other association or legal entity other than an individual that directly establishes, administers, or controls the committee. The name of an entity that is required to be included in the name of the committee may be a commonly recognized acronym by which the entity is known.

# Sec. 252.0031. CONTENTS OF APPOINTMENT BY SPECIFIC-PURPOSE COMMITTEE.

(a) In addition to the information required by Section 252.002, a campaign treasurer appointment by a specific-purpose committee for supporting or opposing a candidate for an office specified by Section 252.005(1) must include:

(1) the name of and the office sought by the candidate; and

(2) before the committee may use a political contribution from a corporation or a labor organization to make a direct campaign expenditure in connection with a campaign for an elective office, an affidavit stating that:

(A) the committee is not established or controlled by a candidate or an officeholder; and

(B) the committee will not use any political contribution from a corporation or a labor organization to make a political contribution to:

(i) a candidate for elective office;

(ii) an officeholder; or

(iii) a political committee that has not filed an affidavit under this subdivision or Section 252.003(a)(4).

(a-1) If the information required to be provided under Subsection (a) changes, the committee shall immediately file an amended appointment reflecting the change.

(a-2) Filing an affidavit under Subsection (a)(2) does not create any additional reporting requirements under Section 254.261.

(b) The name of a specific-purpose committee for supporting a candidate for an office specified by Section 252.005(1) must include the name of the candidate that the committee supports.

#### Sec. 252.0032. CONTENTS OF APPOINTMENT BY CANDIDATE.

(a) In addition to the information required by Section 252.002, a campaign treasurer appointment by a candidate must include:

(1) the candidate's telephone number; and

(2) a statement, signed by the candidate, that the candidate is aware of the nepotism law, Chapter 573, Government Code.

(b) A campaign treasurer appointment that is filed in a manner other than by use of an officially prescribed form is not invalid because it fails to comply with Subsection (a)(2).

Sec. 252.004. DESIGNATION OF ONESELF. An individual may appoint himself or herself as campaign treasurer.

Sec. 252.005. AUTHORITY WITH WHOM APPOINTMENT FILED: CANDIDATE. An individual must file a campaign treasurer appointment for the individual's own candidacy with:

(1) the commission, if the appointment is made for candidacy for:

- (A) a statewide office;
- (B) a district office filled by voters of more than one county;
- (C) a judicial district office filled by voters of only one county;
- (D) state senator;
- (E) state representative; or
- (F) the State Board of Education;

(2) the county clerk, if the appointment is made for candidacy for a county office, a precinct office, or a district office other than one included in Subdivision (1);

(3) the clerk or secretary of the governing body of the political subdivision or, if the political subdivision has no clerk or secretary, with the governing body's presiding officer, if the appointment is made for candidacy for an office of a political subdivision other than a county;

(4) the county clerk if:

(A) the appointment is made for candidacy for an office of a political subdivision other than a county;

(B) the governing body for the political subdivision has not been formed; and

(C) no boundary of the political subdivision crosses a boundary of the

county; or

formed; and

(5) the commission if:

(A) the appointment is made for candidacy for an office of a political subdivision other than a county;

(B) the governing body for the political subdivision has not been

(C) the political subdivision is situated in more than one county.

Sec. 252.006. AUTHORITY WITH WHOM APPOINTMENT FILED: SPECIFIC-PURPOSE COMMITTEE FOR SUPPORTING OR OPPOSING CANDIDATE OR ASSISTING OFFICEHOLDER. A specific-purpose committee for supporting or opposing a candidate or assisting an officeholder must file its campaign treasurer appointment with the same authority as the appointment for candidacy for the office.

Sec. 252.007. AUTHORITY WITH WHOM APPOINTMENT FILED: SPECIFIC-PURPOSE COMMITTEE FOR SUPPORTING OR OPPOSING MEASURE. A specific-purpose committee for supporting or opposing a measure must file its campaign treasurer appointment with:

(1) the commission, if the measure is to be submitted to voters of the entire state;

(2) the county clerk, if the measure is to be submitted to voters of a single county in an election ordered by a county authority;

(3) the secretary of the governing body of the political subdivision or, if the political subdivision has no secretary, with the governing body's presiding officer, if the measure is to be submitted at an election ordered by an authority of a political subdivision other than a county;

(4) the county clerk if:

(A) the measure concerns a political subdivision other than a county;

(B) the governing body for the political subdivision has not been

formed; and

(C) no boundary of the political subdivision crosses a boundary of a

county; or

(5) the commission if:

(A) the measure concerns a political subdivision other than a county;

formed; and

- (B) the governing body for the political subdivision has not been
- (C) the political subdivision is situated in more than one county.

Sec. 252.008. MULTIPLE FILINGS BY SPECIFIC-PURPOSE COMMITTEE NOT REQUIRED. If under this chapter a specific-purpose committee is required to file its campaign treasurer appointment with more than one authority, the appointment need only be filed with the commission and, if so filed, need not be filed with the other authorities.

Sec. 252.009. AUTHORITY WITH WHOM APPOINTMENT FILED: GENERAL-PURPOSE COMMITTEE. A general-purpose committee must file its campaign treasurer appointment with the commission.

#### Sec. 252.010. TRANSFER OF APPOINTMENT.

(a) If a candidate who has filed a campaign treasurer appointment decides to seek a different office that would require the appointment to be filed with another authority, a copy of the appointment certified by the authority with whom it was originally filed must be filed with the other authority in addition to the new campaign treasurer appointment.

(b) The original appointment terminates on the filing of the copy with the appropriate authority or on the 10th day after the date the decision to seek a different office is made, whichever is earlier.

# Sec. 252.011. TIME APPOINTMENT TAKES EFFECT; PERIOD OF EFFECTIVENESS.

(a) A campaign treasurer appointment takes effect at the time it is filed with the authority specified by this chapter.

(b) A campaign treasurer appointment continues in effect until terminated.

#### Sec. 252.012. REMOVAL OF CAMPAIGN TREASURER.

(a) A campaign treasurer appointed under this chapter may be removed at any time by the appointing authority by filing the written appointment of a successor in the same manner as the original appointment.

(b) The appointment of a successor terminates the appointment of the campaign treasurer who is removed.

(c) If the campaign treasurer of a specific-purpose political committee required to file its campaign treasurer appointment with the commission or of a general-purpose political committee is removed by the committee, the departing campaign treasurer shall immediately file written notification of the termination of appointment with the commission.

## Sec. 252.013. TERMINATION OF APPOINTMENT ON VACATING POSITION.

(a) If a campaign treasurer resigns or otherwise vacates the position, the appointment is terminated at the time the vacancy occurs.

(b) A campaign treasurer who vacates the treasurer's position shall immediately notify the appointing authority in writing of the vacancy.

(c) If the campaign treasurer of a specific-purpose political committee required to file its campaign treasurer appointment with the commission or of a general-purpose political committee resigns or otherwise vacates the position, the campaign treasurer shall immediately file written notification of the vacancy with the commission.

# Sec. 252.0131. TERMINATION OF CAMPAIGN TREASURER APPOINTMENT.

(a) The commission by rule shall adopt a process by which the commission may terminate the campaign treasurer appointment of an inactive candidate or political committee that is required to file a campaign treasurer appointment with the commission. The governing body of a political subdivision by ordinance or order may adopt a process by which the clerk or secretary, as applicable, of the political subdivision may terminate the campaign treasurer appointment of an inactive candidate or political committee that is required to file a campaign treasurer appointment with the clerk or secretary. For purposes of this section, a candidate or political committee is inactive if the candidate or committee:

(1) has never filed or has ceased to file reports under Chapter 254;

(2) in the case of a candidate, has not been elected to an office for which a candidate is required to file a campaign treasurer appointment with the authority who is seeking to terminate the candidate's campaign treasurer appointment; and

(3) has not filed:

(A) a final report under Section 254.065 or 254.125; or

(B) a dissolution report under Section 254.126 or 254.159.

(b) Before the commission may terminate a campaign treasurer appointment, the commission must consider the proposed termination in a regularly scheduled open meeting. Before the clerk or secretary of a political subdivision may terminate a campaign treasurer appointment, the governing body of the political subdivision must consider the proposed termination in a regularly scheduled open meeting.

(c) Rules or an ordinance or order adopted under this section must:

(1) define "inactive candidate or political committee" for purposes of terminating the candidate's or committee's campaign treasurer appointment; and

(2) require written notice to the affected candidate or committee of:

(A) the proposed termination of the candidate's or committee's campaign treasurer appointment;

(B) the date, time, and place of the meeting at which the commission or governing body of the political subdivision, as applicable, will consider the proposed termination; and

(C) the effect of termination of the candidate's or committee's campaign treasurer appointment.

(d) The termination of a campaign treasurer appointment under this section takes effect on the 30th day after the date of the meeting at which the commission or governing body, as applicable, votes to terminate the appointment. Following that meeting, the commission or the clerk or secretary of the political subdivision, as applicable, shall promptly notify the affected candidate or political committee that the appointment has been terminated. The notice must state the effective date of the termination.

Sec. 252.014. PRESERVATION OF FILED APPOINTMENTS. The authority with whom a campaign treasurer appointment is filed under this chapter shall preserve the appointment for two years after the date the appointment is terminated.

#### Sec. 252.015. ASSISTANT CAMPAIGN TREASURER.

(a) Each specific-purpose committee for supporting or opposing a candidate for an office specified by Section 252.005(1) or a statewide or district measure and each general-purpose committee may appoint an assistant campaign treasurer by written appointment filed with the commission.

(b) In the campaign treasurer's absence, the assistant campaign treasurer has the same authority as a campaign treasurer.

(c) Sections 252.011, 252.012, 252.013, and 252.014 apply to the appointment and removal of an assistant campaign treasurer.

#### CHAPTER 253. RESTRICTIONS ON CONTRIBUTIONS AND EXPENDITURES

#### SUBCHAPTER A. GENERAL RESTRICTIONS

## Sec. 253.001. CONTRIBUTION OR EXPENDITURE IN ANOTHER'S NAME PROHIBITED.

(a) A person may not knowingly make or authorize a political contribution in the name of or on behalf of another unless the person discloses in writing to the recipient the name and address of the person actually making the contribution in order for the recipient to make the proper disclosure.

(b) A person may not knowingly make or authorize a political expenditure in the name of or on behalf of another unless the person discloses in writing to the person on whose behalf the expenditure is made the name and address of the person actually making the expenditure in order for the person on whose behalf the expenditure is made to make the proper disclosure.

(c) A person who violates this section commits an offense. An offense under this section is a Class A misdemeanor.

#### Sec. 253.003. UNLAWFULLY MAKING OR ACCEPTING CONTRIBUTION.

(a) A person may not knowingly make a political contribution in violation of this chapter.

(b) A person may not knowingly accept a political contribution the person knows to have been made in violation of this chapter.

(c) This section does not apply to a political contribution made or accepted in violation of Subchapter F.

(d) Except as provided by Subsection (e), a person who violates this section commits an offense. An offense under this section is a Class A misdemeanor.

(e) A violation of Subsection (a) or (b) is a felony of the third degree if the contribution is made in violation of Subchapter D.

#### Sec. 253.004. UNLAWFULLY MAKING EXPENDITURE.

(a) A person may not knowingly make or authorize a political expenditure in violation of this chapter.

(b) This section does not apply to a political expenditure made or authorized in violation of Subchapter F.

(c) A person who violates this section commits an offense. An offense under this section is a Class A misdemeanor.

#### Sec. 253.005. EXPENDITURE FROM UNLAWFUL CONTRIBUTION.

(a) A person may not knowingly make or authorize a political expenditure wholly or partly from a political contribution the person knows to have been made in violation of this chapter.

(b) This section does not apply to a political expenditure that is:

(1) prohibited by Section 253.101; or

(2) made from a political contribution made in violation of Subchapter F.

(c) A person who violates this section commits an offense. An offense under this section is a Class A misdemeanor.

Sec. 253.006. CERTAIN CONTRIBUTIONS AND EXPENDITURES BY LOBBYISTS RESTRICTED. Notwithstanding any other provision of law, a person required to register under Chapter 305, Government Code, may not knowingly make or authorize a political contribution or political expenditure that is a political contribution to another candidate, officeholder, or political committee, or direct campaign expenditure, from political contributions accepted by:

(1) the person as a candidate or officeholder;

(2) a specific-purpose committee for the purpose of supporting the person as a candidate or assisting the person as an officeholder; or

(3) a political committee that accepted a political contribution from a source described by Subdivision (1) or (2) during the two-year period immediately before the date the political contribution or expenditure was made.

#### Sec. 253.007. PROHIBITION ON LOBBYING BY PERSON MAKING OR AUTHORIZING CERTAIN POLITICAL CONTRIBUTIONS AND DIRECT CAMPAIGN EXPENDITURES.

(a) In this section, "administrative action," "communicates directly with," "legislation," "member of the executive branch," and "member of the legislative branch" have the meanings assigned by Section 305.002, Government Code.

(b) Notwithstanding any other provision of law and except as provided by Subsection (c), a person who knowingly makes or authorizes a political contribution or political expenditure that is a political contribution to another candidate, officeholder, or political committee, or direct campaign expenditure, from political contributions accepted by the person as a candidate or officeholder may not engage in any activities that require the person to register under Chapter 305, Government Code, during the two-year period after the date the person makes or authorizes the political contribution or direct campaign expenditure.

(c) Subsection (b) does not apply to a person who:

(1) communicates directly with a member of the legislative or executive branch only to influence legislation or administrative action on behalf of:

(A) a nonprofit organization exempt from federal income taxation under Section 501(a), Internal Revenue Code of 1986, as an organization described by Section 501(c)(3) of that code;

(B) a group of low-income individuals; or

(C) a group of individuals with disabilities; and

(2) does not receive compensation other than reimbursement for actual expenses for engaging in communication described by Subdivision (1).

#### SUBCHAPTER B. CANDIDATES, OFFICEHOLDERS, AND POLITICAL COMMITTEES

# Sec. 253.031. CONTRIBUTION AND EXPENDITURE WITHOUT CAMPAIGN TREASURER PROHIBITED.

(a) A candidate may not knowingly accept a campaign contribution or make or authorize a campaign expenditure at a time when a campaign treasurer appointment for the candidate is not in effect. (b) A political committee may not knowingly accept political contributions totaling more than \$500 or make or authorize political expenditures totaling more than \$500 at a time when a campaign treasurer appointment for the committee is not in effect.

(c) A political committee may not knowingly make or authorize a campaign contribution or campaign expenditure supporting or opposing a candidate for an office specified by Section 252.005(1) in a primary or general election unless the committee's campaign treasurer appointment has been filed not later than the 30th day before the appropriate election day.

(d) This section does not apply to a political party's county executive committee that accepts political contributions or makes political expenditures, except that:

(1) a county executive committee that accepts political contributions or makes political expenditures shall maintain the records required by Section 254.001; and

(2) a county executive committee that accepts political contributions or makes political expenditures that, in the aggregate, exceed \$25,000 in a calendar year shall file:

(A) a campaign treasurer appointment as required by Section 252.001 not later than the 15th day after the date that amount is exceeded; and

(B) the reports required by Subchapter F, Chapter 254, including in the political committee's first report all political contributions accepted and all political expenditures made before the effective date of the campaign treasurer appointment.

(e) This section does not apply to an out-of-state political committee unless the committee is subject to Chapter 252 under Section 251.005.

(f) A person who violates this section commits an offense. An offense under this section is a Class A misdemeanor.

# Sec. 253.032. LIMITATION ON CONTRIBUTION BY OUT-OF-STATE COMMITTEE.

(a) In a reporting period, a candidate, officeholder, or political committee may not knowingly accept political contributions totaling more than \$500 from an out-of-state political committee unless, before accepting a contribution that would cause the total to exceed \$500, the candidate, officeholder, or political committee, as applicable, receives from the out-of-state committee:

(1) a written statement, certified by an officer of the out-of-state committee, listing the full name and address of each person who contributed more than \$100 to the out-ofstate committee during the 12 months immediately preceding the date of the contribution; or

(2) a copy of the out-of-state committee's statement of organization filed as required by law with the Federal Election Commission and certified by an officer of the out-of-state committee.

(b) This section does not apply to a contribution from an out-of-state political committee if the committee appointed a campaign treasurer under Chapter 252 before the contribution was made and is subject to the reporting requirements of Chapter 254.

(c) A person who violates Subsection (a) commits an offense. An offense under this section is a Class A misdemeanor.

(d) A candidate, officeholder, or political committee shall include the statement or copy required by Subsection (a) as a part of the report filed under Chapter 254 that covers the reporting period to which Subsection (a) applies.

(e) A candidate, officeholder, or political committee that accepts political contributions totaling \$500 or less from an out-of-state political committee shall include as part of the report filed under Chapter 254 that covers the reporting period in which the contribution is accepted:

(1) the same information for the out-of-state political committee required for general-purpose committees by Sections 252.002 and 252.003; or

(2) a copy of the out-of-state committee's statement of organization filed as required by law with the Federal Election Commission and certified by an officer of the out-of-state committee.

#### Sec. 253.033. CASH CONTRIBUTIONS EXCEEDING \$100 PROHIBITED.

(a) A candidate, officeholder, or specific-purpose committee may not knowingly accept from a contributor in a reporting period political contributions in cash that in the aggregate exceed \$100.

(b) A person who violates this section commits an offense. An offense under this section is a Class A misdemeanor.

# Sec. 253.034. RESTRICTIONS ON CONTRIBUTIONS DURING AND FOLLOWING REGULAR LEGISLATIVE SESSION.

(a) During the period beginning on the 30th day before the date a regular legislative session convenes and continuing through the 20th day after the date of final adjournment, a person may not knowingly make a political contribution to:

(1) a statewide officeholder;

(2) a member of the legislature; or

(3) a specific-purpose committee for supporting, opposing, or assisting a statewide officeholder or member of the legislature.

(b) A statewide officeholder, a member of the legislature, or a specific-purpose committee for supporting, opposing, or assisting a statewide officeholder or member of the legislature may not knowingly accept a political contribution, and shall refuse a political contribution that is received, during the period prescribed by Subsection (a). A political contribution that is received and refused during that period shall be returned to the contributor not later than the 30th day after the date of receipt. A contribution made by United States mail or by common or contract carrier is not considered received during that period if it was properly addressed and placed with postage or carrier charges prepaid or prearranged in the mail or delivered to the contribution mark or the common or contract carrier documents is considered to be the date the contribution was placed in the mail or delivered to the contract carrier unless proven otherwise.

(c) This section does not apply to a political contribution that was made and accepted with the intent that it be used:

(1) in an election held or ordered during the period prescribed by Subsection (a) in which the person accepting the contribution is a candidate if the contribution was made after the person appointed a campaign treasurer with the appropriate authority and before the person was sworn in for that office;

(2) to defray expenses incurred in connection with an election contest; or

(3) by a person who holds a state office or a member of the legislature if the person or member was defeated at the general election held immediately before the session is convened or by a specific-purpose political committee that supports or assists only that person or member.

(d) This section does not apply to a political contribution made to or accepted by a holder of an office to which Subchapter F applies.

(e) A person who violates this section commits an offense. An offense under this section is a Class A misdemeanor.

# Sec. 253.0341. RESTRICTIONS ON CONTRIBUTIONS TO LEGISLATIVE CAUCUSES DURING AND FOLLOWING REGULAR LEGISLATIVE SESSION.

(a) During the period beginning on the 30th day before the date a regular legislative session convenes and continuing through the 20th day after the date of final adjournment, a person not a member of the caucus may not knowingly make a contribution to a legislative caucus.

(b) A legislative caucus may not knowingly accept from a nonmember a contribution, and shall refuse a contribution from a nonmember that is received, during the period prescribed by Subsection (a). A contribution that is received and refused during that period shall be returned to the contributor not later than the 30th day after the date of receipt. A contribution made by United States mail or by common or contract carrier is not considered received during that period if it was properly addressed and placed with postage or carrier charges prepaid or prearranged in the mail or delivered to the contract carrier before the beginning of the period. The date indicated by the post office cancellation mark or the common or contract carrier documents is considered to be the date the contribution was placed in the mail or delivered to the contract or the mail or delivered to the contract or the mail or delivered to the contribution was placed in the mail or delivered to the contract carrier before the beginning of the period.

(c) A person who violates this section commits an offense. An offense under this section is a Class A misdemeanor.

(d) A person who knowingly makes or accepts a contribution in violation of this section is liable for damages to the state in the amount of triple the value of the unlawful contribution.

(e) In this section, "legislative caucus" means an organization that is composed exclusively of members of the legislature, that elects or appoints officers and recognizes identified legislators as members of the organization, and that exists for research and other support of policy development and interests that the membership hold in common. The term includes an entity established by or for a legislative caucus to conduct research, education, or any other caucus activity. An organization whose only nonlegislator members are the lieutenant governor or the governor remains a "legislative caucus" for purposes of this section.

#### Sec. 253.035. RESTRICTIONS ON PERSONAL USE OF CONTRIBUTIONS.

(a) A person who accepts a political contribution as a candidate or officeholder may not convert the contribution to personal use.

(b) A specific-purpose committee that accepts a political contribution may not convert the contribution to the personal use of a candidate, officeholder, or former candidate or officeholder.

(c) The prohibitions prescribed by Subsections (a) and (b) include the personal use of an asset purchased with the contribution and the personal use of any interest and other income earned on the contribution.

(d) In this section, "personal use" means a use that primarily furthers individual or family purposes not connected with the performance of duties or activities as a candidate for or holder of a public office. The term does not include:

(1) payments made to defray ordinary and necessary expenses incurred in connection with activities as a candidate or in connection with the performance of duties or activities as a public officeholder, including payment of rent, utility, and other reasonable housing or household expenses incurred in maintaining a residence in Travis County by members of the legislature who do not ordinarily reside in Travis County, but excluding payments prohibited under Section 253.038; or

(2) payments of federal income taxes due on interest and other income earned on political contributions.

(e) Subsection (a) applies only to political contributions accepted on or after September 1, 1983. Subsection (b) applies only to political contributions accepted on or after September 1, 1987.

(f) A person who converts a political contribution to the person's personal use in violation of this section is civilly liable to the state for an amount equal to the amount of the converted contribution plus reasonable court costs.

(g) A specific-purpose committee that converts a political contribution to the personal use of a candidate, officeholder, or former candidate or officeholder in violation of this section is civilly liable to the state for an amount equal to the amount of the converted contribution plus reasonable court costs.

(h) Except as provided by Section 253.0351 or 253.042, a candidate or officeholder who makes political expenditures from the candidate's or officeholder's personal funds may reimburse those personal funds from political contributions in the amount of those expenditures only if:

(1) the expenditures from personal funds were fully reported as political expenditures, including the payees, dates, purposes, and amounts of the expenditures, in the report required to be filed under this title that covers the period in which the expenditures from personal funds were made; and

(2) the report on which the expenditures from personal funds are disclosed clearly designates those expenditures as having been made from the person's personal funds and that the expenditures are subject to reimbursement.

(i) "Personal use" does not include the use of contributions for:

(1) defending a criminal action or prosecuting or defending a civil action brought by or against the person in the person's status as a candidate or officeholder; or

(2) participating in an election contest or participating in a civil action to determine a person's eligibility to be a candidate for, or elected or appointed to, a public office in this state.

(j), (k) Repealed by Acts 1991, 72nd Leg., ch. 304, Sec. 5.20, eff. Jan. 1, 1992.

## Sec. 253.0351. LOANS FROM PERSONAL FUNDS.

(a) A candidate or officeholder who makes political expenditures from the candidate's or officeholder's personal funds may report the amount expended as a loan and may reimburse those personal funds from political contributions in the amount of the reported loan.

(b) Section 253.035(h) applies if the person does not report an amount as a loan as authorized by Subsection (a).

(c) A candidate or officeholder who deposits personal funds in an account in which political contributions are held shall report the amount of personal funds deposited as a loan and may reimburse the amount deposited as a loan from political contributions or unexpended personal funds deposited in the account. The reimbursement may not exceed the amount reported as a loan. Personal funds deposited in an account in which political contributions are held are subject to Section 253.035 and must be included in the reports of the total amount of political contributions maintained required by Sections 254.031(a)(8) and 254.0611(a).

Sec. 253.036. OFFICEHOLDER CONTRIBUTIONS USED IN CONNECTION WITH CAMPAIGN. An officeholder who lawfully accepts officeholder contributions may use those contributions in connection with the officeholder's campaign for elective office after appointing a campaign treasurer.

## Sec. 253.037. RESTRICTIONS ON CONTRIBUTION OR EXPENDITURE BY GENERAL-PURPOSE COMMITTEE.

(a) Repealed by Acts 2019, 86th Leg., R.S., Ch. 209 (H.B. 3580), Sec. 5(1), eff. September 1, 2019.

(b) A general-purpose committee may not knowingly make a political contribution to another general-purpose committee unless the other committee is listed in the campaign treasurer appointment of the contributor committee.

(c) Repealed by Acts 2019, 86th Leg., R.S., Ch. 209 (H.B. 3580), Sec. 5(1), eff. September 1, 2019.

(d) A person who violates this section commits an offense. An offense under this section is a Class A misdemeanor.

## Sec. 253.038. PAYMENTS MADE TO PURCHASE REAL PROPERTY OR TO RENT CERTAIN REAL PROPERTY PROHIBITED.

(a) A candidate or officeholder or a specific-purpose committee for supporting, opposing, or assisting the candidate or officeholder may not knowingly make or authorize a payment from a political contribution to purchase real property or to pay the interest on or principal of a note for the purchase of real property.

(a-1) A candidate or officeholder or a specific-purpose committee for supporting, opposing, or assisting the candidate or officeholder may not knowingly make or authorize a payment from a political contribution for the rental or purchase of real property from:

(1) a person related within the second degree by consanguinity or affinity, as determined under Chapter 573, Government Code, to the candidate or officeholder; or

(2) a business in which the candidate or officeholder or a person described by Subdivision (1) has a participating interest of more than 10 percent, holds a position on the governing body, or serves as an officer.

(b) A person who violates this section commits an offense. An offense under this subsection is a Class A misdemeanor.

(c) This section does not apply to a payment made in connection with real property that was purchased before January 1, 1992.

# Sec. 253.039. CONTRIBUTIONS IN CERTAIN PUBLIC BUILDINGS PROHIBITED.

(a) A person may not knowingly make or authorize a political contribution while in the Capitol or a courthouse to:

- (1) a candidate or officeholder;
- (2) a political committee; or
- (3) a person acting on behalf of a candidate, officeholder, or political committee.

(b) A candidate, officeholder, or political committee or a person acting on behalf of a candidate, officeholder, or political committee may not knowingly accept a political contribution, and shall refuse a political contribution that is received, in the Capitol or a courthouse.

(c) This section does not prohibit contributions made in the Capitol or a courthouse through the United States postal service or a common or contract carrier.

(d) A person who violates this section commits an offense. An offense under this section is a Class A misdemeanor.

(h) In this section, "courthouse" means any building owned by the state, a county, or a municipality, or an office or part of a building leased to the state, a county, or a municipality, in which a justice or judge sits to conduct court proceedings.

#### Sec. 253.040. SEPARATE ACCOUNTS.

(a) Except as provided by Section 253.0351(c), each candidate or officeholder shall keep the person's campaign and officeholder contributions in one or more accounts that are separate from any other account maintained by the person.

(b) A person who violates this section commits an offense. An offense under this section is a Class B misdemeanor.

## Sec. 253.041. RESTRICTIONS ON CERTAIN PAYMENTS.

(a) A candidate or officeholder or a specific-purpose committee for supporting, opposing, or assisting the candidate or officeholder may not knowingly make or authorize a payment from a political contribution if the payment is made for personal services rendered by the candidate or officeholder or by the spouse or dependent child of the candidate or officeholder to:

(1) a business in which the candidate or officeholder has a participating interest of more than 10 percent, holds a position on the governing body of the business, or serves as an officer of the business; or

(2) the candidate or officeholder or the spouse or dependent child of the candidate or officeholder.

(b) A payment that is made from a political contribution to a business described by Subsection (a) and that is not prohibited by that subsection may not exceed the amount necessary to reimburse the business for actual expenditures made by the business.

(c) A person who violates this section commits an offense. An offense under this subsection is a Class A misdemeanor.

## Sec. 253.042. RESTRICTIONS ON REIMBURSEMENT OF PERSONAL FUNDS AND PAYMENTS ON CERTAIN LOANS.

(a) A candidate or officeholder who makes political expenditures from the candidate's or officeholder's personal funds may not reimburse those personal funds from political contributions in amounts that in the aggregate exceed the following amounts for each election in which the person's name appears on the ballot:

(1) for a statewide office other than governor, \$250,000; and

(2) for governor, \$500,000.

(b) A candidate or officeholder who accepts one or more political contributions in the form of loans, including an extension of credit or a guarantee of a loan or extension of credit, from one or more persons related to the candidate or officeholder within the second degree by affinity or consanguinity may not use political contributions to repay the loans in amounts that in the aggregate exceed the amount prescribed by Subsection (a).

(c) The total amount of both reimbursements and repayments made by a candidate or officeholder under this section may not exceed the amount prescribed by Subsection (a).

(d) A person who is both a candidate and an officeholder covered by Subsection (a) may reimburse the person's personal funds or repay loans from political contributions only in one capacity.

(e) This section does not prohibit the payment of interest on loans covered by this section at a commercially reasonable rate, except that interest on loans from a candidate's or officeholder's personal funds or on loans from the personal funds of any person related to the candidate or officeholder within the second degree by affinity or consanguinity is included in the amount prescribed by Subsection (a), (b), or (c).

(f) A person who violates this section commits an offense. An offense under this section is a Class A misdemeanor.

(g) The commission shall study possible restrictions on amounts of reimbursements under Subsection (a) in connection with the offices of state senator and state representative and shall make appropriate recommendations to the legislature on those matters.

Sec. 253.043. POLITICAL CONTRIBUTIONS USED IN CONNECTION WITH APPOINTIVE OFFICE. A former candidate or former officeholder who lawfully accepts political contributions may use those contributions to make an expenditure to defray expenses incurred by the person in performing a duty or engaging in an activity in connection with an appointive office of a state board or commission.

## SUBCHAPTER D. CORPORATIONS AND LABOR ORGANIZATIONS

Sec. 253.091. CORPORATIONS COVERED. This subchapter applies only to corporations that are organized under the Texas Business Corporation Act, the Texas For-Profit Corporation Law, the Texas Non-Profit Corporation Act, the Texas Nonprofit Corporation Law, federal law, or law of another state or nation.

**Sec. 253.092. TREATMENT OF INCORPORATED POLITICAL COMMITTEE.** If a political committee the only principal purpose of which is accepting political contributions and making political expenditures incorporates for liability purposes only, the committee is not considered to be a corporation for purposes of this subchapter.

## Sec. 253.093. CERTAIN ASSOCIATIONS COVERED.

(a) For purposes of this subchapter, the following associations, whether incorporated or not, are considered to be corporations covered by this subchapter: banks, trust companies, savings and loan associations or companies, insurance companies, reciprocal or interinsurance exchanges, railroad companies, cemetery companies, government-regulated cooperatives, stock companies, and abstract and title insurance companies.

(b) For purposes of this subchapter, the members of the associations specified by Subsection (a) are considered to be stockholders.

## Sec. 253.094. CONTRIBUTIONS PROHIBITED.

(a) A corporation or labor organization may not make a political contribution that is not authorized by this subchapter.

(b) A corporation or labor organization may not make a political contribution in connection with a recall election, including the circulation and submission of a petition to call an election.

(c) A person who violates this section commits an offense. An offense under this section is a felony of the third degree.

Sec. 253.095. PUNISHMENT OF AGENT. An officer, director, or other agent of a corporation or labor organization who commits an offense under this subchapter is punishable for the grade of offense applicable to the corporation or labor organization.

Sec. 253.096. CONTRIBUTION ON MEASURE. A corporation or labor organization may make campaign contributions from its own property in connection with an election on a measure only to a political committee for supporting or opposing measures exclusively.

Sec. 253.097. CONTRIBUTION FROM CORPORATION OR LABOR ORGANIZATION. A corporation or labor organization may make campaign contributions from its own property to a political committee that has filed an affidavit with the committee's campaign treasurer appointment in accordance with Section 252.003(a)(4) or 252.0031(a)(2).

#### Sec. 253.098. COMMUNICATION WITH STOCKHOLDERS OR MEMBERS.

(a) A corporation or labor organization may make one or more direct campaign expenditures from its own property for the purpose of communicating directly with its stockholders or members, as applicable, or with the families of its stockholders or members.

(b) An expenditure under this section is not reportable under Chapter 254.

### Sec. 253.099. NONPARTISAN VOTER REGISTRATION AND GET-OUT-THE-VOTE CAMPAIGNS.

(a) A corporation or labor organization may make one or more expenditures to finance nonpartisan voter registration and get-out-the-vote campaigns aimed at its stockholders or members, as applicable, or at the families of its stockholders or members.

(b) An expenditure under this section is not reportable under Chapter 254.

#### Sec. 253.100. EXPENDITURES FOR GENERAL-PURPOSE COMMITTEE.

(a) A corporation, acting alone or with one or more other corporations, may make one or more political expenditures to finance the establishment or administration of a general-purpose committee. In addition to any other expenditure that is considered permissible under this section, a corporation may make an expenditure for the maintenance and operation of a general-purpose committee, including an expenditure for:

(1) office space maintenance and repairs;

- (2) telephone and Internet services;
- (3) office equipment;
- (4) utilities;
- (5) general office and meeting supplies;

(6) salaries for routine clerical, data entry, and administrative assistance necessary for the proper administrative operation of the committee;

(7) legal and accounting fees for the committee's compliance with this title;

(8) routine administrative expenses incurred in establishing and administering a general-purpose political committee;

(9) management and supervision of the committee, including expenses incurred in holding meetings of the committee's governing body to interview candidates and make endorsements relating to the committee's support;

(10) the recording of committee decisions;

(11) expenses incurred in hosting candidate forums in which all candidates for a particular office in an election are invited to participate on the same terms;

(12) expenses incurred in preparing and delivering committee contributions;

(13) creation and maintenance of the committee 's public Internet web pages that do not contain political advertising.

(b) A corporation may make political expenditures, including fully or partially matching contributions to an organization that is exempt from federal income tax under Section 501(c)(3), Internal Revenue Code of 1986, to finance the solicitation of political contributions to a general-purpose committee assisted under Subsection (a) from the stockholders, employees, or families of stockholders or employees of one or more corporations.

(c) A labor organization may engage in activity authorized for a corporation by this section. For purposes of this section, the members of a labor organization are considered to be corporate stockholders.

(d) A corporation or labor organization may not make expenditures under this section for:

(1) political consulting to support or oppose a candidate;

(2) telephoning or telephone banks to communicate with the public;

(3) brochures and direct mail supporting or opposing a candidate;

(4) partisan voter registration and get-out-the-vote drives;

(5) political fund-raising other than from its stockholders or members, as applicable, or the families of its stockholders or members;

(6) voter identification efforts, voter lists, or voter databases that include persons other than its stockholders or members, as applicable, or the families of its stockholders or members;

(7) polling designed to support or oppose a candidate other than of its stockholders or members, as applicable, or the families of its stockholders or members; or

(8) recruiting candidates.

(e) Subsection (d) does not apply to a corporation or labor organization making a campaign contribution to a political committee under Section 253.097 or an expenditure to communicate with its stockholders or members, as applicable, or with the families of its stockholders or members as provided by Section 253.098.

## Sec. 253.101. UNLAWFUL CONTRIBUTION OR EXPENDITURE BY COMMITTEE.

(a) A political committee assisted by a corporation or labor organization under Section 253.100 may not make a political contribution or political expenditure in whole or part from money that is known by a member or officer of the political committee to be dues, fees, or other money required as a condition of employment or condition of membership in a labor organization.

(a-1) Subsection (a) does not prohibit a political committee from making a political contribution or political expenditure wholly or partly from a campaign contribution made by a corporation or labor organization to the political committee under Section 253.096 or 253.097.

(b) A person who violates this section commits an offense. An offense under this section is a felony of the third degree.

#### Sec. 253.102. COERCION PROHIBITED.

(a) A corporation or labor organization or a political committee assisted by a corporation or labor organization under Section 253.100 commits an offense if it uses or threatens to use physical force, job discrimination, or financial reprisal to obtain money or any other thing of value to be used to influence the result of an election or to assist an officeholder.

(b) A political committee assisted by a corporation or labor organization under Section 253.100 commits an offense if it accepts or uses money or any other thing of value that is known by a member or officer of the political committee to have been obtained in violation of Subsection (a).

(c) An offense under this section is a felony of the third degree.

#### Sec. 253.103. CORPORATE LOANS.

(a) A corporation may not make a loan to a candidate, officeholder, or political committee for campaign or officeholder purposes unless:

(1) the corporation has been legally and continuously engaged in the business of lending money for at least one year before the loan is made; and

(2) the loan is made in the due course of business.

(b) This section does not apply to a loan covered by Section 253.096.

(c) A person who violates this section commits an offense. An offense under this section is a felony of the third degree.

#### Sec. 253.104. CONTRIBUTION TO POLITICAL PARTY.

(a) A corporation or labor organization may make a contribution from its own property to a political party to be used as provided by Chapter 257.

(b) A corporation or labor organization may not knowingly make a contribution authorized by Subsection (a) during a period beginning on the 60th day before the date of a general election for state and county officers and continuing through the day of the election.

(c) A corporation or labor organization that knowingly makes a contribution in violation of this section commits an offense. An offense under this section is a felony of the third degree.

## Sec. 253.105. CONTRIBUTIONS TO DIRECT EXPENDITURE ONLY COMMITTEES.

(a) A corporation or labor organization may make a political contribution from its own property to a political committee that:

(1) is not established or controlled by a candidate or an officeholder;

(2) makes or intends to make direct campaign expenditures;

(3) does not make or intend to make political contributions to:

- (A) a candidate;
- (B) an officeholder;

(C) specific-purpose committee established or controlled by a candidate or an officeholder; or

(D) a political committee that makes or intends to make political contributions to a candidate, an officeholder, or a specific-purpose committee established or controlled by a candidate or an officeholder; and

(4) has filed an affidavit with the commission stating the committee's intention to operate as described by Subdivisions (2) and (3).

(b) A political contribution made by a corporation or labor organization under this section does not constitute a violation of Section 253.094(a) and the acceptance of the political contribution does not constitute a violation of Section 253.003(b).

## SUBCHAPTER E. CIVIL LIABILITY

#### Sec. 253.131. LIABILITY TO CANDIDATES.

(a) A person who knowingly makes or accepts a campaign contribution or makes a campaign expenditure in violation of this chapter is liable for damages as provided by this section.

(b) If the contribution or expenditure is in support of a candidate, each opposing candidate whose name appears on the ballot is entitled to recover damages under this section.

(c) If the contribution or expenditure is in opposition to a candidate, the candidate is entitled to recover damages under this section.

(d) In this section, "damages" means:

- (1) twice the value of the unlawful contribution or expenditure; and
- (2) reasonable attorney's fees incurred in the suit.

(e) Reasonable attorney's fees incurred in the suit may be awarded to the defendant if judgment is rendered in the defendant's favor.

#### Sec. 253.132. LIABILITY TO POLITICAL COMMITTEES.

(a) A corporation or labor organization that knowingly makes a campaign contribution to a political committee or a direct campaign expenditure in violation of Subchapter D is liable for damages as provided by this section to each political committee of opposing interest in the election in connection with which the contribution or expenditure is made.

- (b) In this section, "damages" means:
  - (1) twice the value of the unlawful contribution or expenditure; and
  - (2) reasonable attorney's fees incurred in the suit.

(c) Reasonable attorney's fees incurred in the suit may be awarded to the defendant if judgment is rendered in the defendant's favor.

**Sec. 253.133. LIABILITY TO STATE**. A person who knowingly makes or accepts a political contribution or makes a political expenditure in violation of this chapter is liable for damages to the state in the amount of triple the value of the unlawful contribution or expenditure.

Sec. 253.134. CIVIL PENALTIES IMPOSED BY COMMISSION. This title does not prohibit the imposition of civil penalties by the commission in addition to criminal penalties or other sanctions imposed by law.

## SUBCHAPTER F. JUDICIAL CAMPAIGN FAIRNESS ACT

**Sec. 253.151. APPLICABILITY OF SUBCHAPTER**. This subchapter applies only to a political contribution or political expenditure in connection with the office of:

(1) chief justice or justice, supreme court;

- (2) presiding judge or judge, court of criminal appeals;
- (3) chief justice or justice, court of appeals;
- (4) district judge;
- (5) judge, statutory county court; or
- (6) judge, statutory probate court.

#### Sec. 253.152. DEFINITIONS. In this subchapter:

(1) "Child" means a person under 18 years of age who is not and has not been married or who has not had the disabilities of minority removed for general purposes.

(2) "In connection with an election" means:

(A) with regard to a contribution that is designated in writing for a particular election, the election designated; or

(B) with regard to a contribution that is not designated in writing for a particular election, the next election for that office occurring after the contribution is made.

(3) "Judicial district" means the territory from which a judicial candidate is elected or appointed.

(4) "Law firm" means a partnership, limited liability partnership, limited liability company, professional corporation, or other entity organized for the practice of law.

(5) "Law firm group" means:

(A) a law firm;

(B) a general-purpose committee established or controlled by the law firm or a member of the law firm;

(C) a member of the law firm; and

(D) the spouse of a member of the law firm.

(6) "Member of a law firm" means:

- (A) a person designated "of counsel" or "of the firm";
- (B) a partner of the law firm, whether an individual or an entity;

(C) an associate of the law firm;

(D) a shareholder of the law firm, whether an individual or an entity; or

(E) an employee of the law firm

(7) "Statewide judicial office" means the office of chief justice or justice, supreme court, or presiding judge or judge, court of criminal appeals.

## Sec. 253.153. CONTRIBUTION PROHIBITED EXCEPT DURING ELECTION PERIOD.

(a) A judicial candidate or officeholder, a specific-purpose committee for supporting or opposing a judicial candidate, or a specific-purpose committee for assisting a judicial officeholder may not knowingly accept a political contribution except during the period:

(1) beginning on:

(A) the 210th day before the date an application for a place on the ballot or for nomination by convention for the office is required to be filed, if the election is for a full term; or (B) the later of the 210th day before the date an application for a place on the ballot or for nomination by convention for the office is required to be filed or the date a vacancy in the office occurs, if the election is for an unexpired term; and

(2) ending on the 120th day after the date of the election in which the candidate or officeholder last appeared on the ballot, regardless of whether the candidate or officeholder has an opponent in that election.

(b) Subsection (a)(2) does not apply to a political contribution that was made and accepted with the intent that it be used to defray expenses incurred in connection with an election, including the repayment of any debt that is:

(1) incurred directly by the making of a campaign expenditure during the period beginning on the date the application for a place on the ballot or for nomination by convention was required to be filed for the election in which the candidate last appeared on the ballot and ending on the date of that election; and

(2) subject to the restrictions prescribed by Sections 253.162 and 253.1621.

(c) Repealed by Acts 2009, 81st Leg., R.S., Ch. 1329, Sec. 2, eff. September 1, 2009.

(d) A person who violates this section is liable for a civil penalty not to exceed three times the amount of the political contributions accepted in violation of this section.

#### Sec. 253.154. WRITE-IN CANDIDACY.

(a) A write-in candidate for judicial office or a specific-purpose committee for supporting a write-in candidate for judicial office may not knowingly accept a political contribution before the candidate files a declaration of write-in candidacy.

(b) A person who violates this section is liable for a civil penalty not to exceed three times the amount of the political contributions accepted in violation of this section.

## Sec. 253.1541. ACCEPTANCE OF POLITICAL CONTRIBUTIONS BY PERSON APPOINTED TO FILL VACANCY.

(a) This section applies only to a person appointed to fill a vacancy in an office covered by this subchapter who, at the time of appointment, does not hold another office covered by this subchapter.

(b) Notwithstanding Section 253.153, a person to whom this section applies may accept political contributions beginning on the date the person assumes the duties of office and ending on the 60th day after that date.

## Sec. 253.155. CONTRIBUTION LIMITS.

(a) A judicial candidate or officeholder may not knowingly accept political contributions from a person that in the aggregate, exceed the contribution limits prescribed by Subsection (b) in connection with each election in which the judicial candidate's name appears on the ballot.

(b) The contribution limits under this section are:

- (1) for a statewide judicial office, \$5,000; or
- (2) for any other judicial office:
  - (A) \$1,000, if the population of the judicial district is less than 250,000;

(B) \$2,500, if the population of the judicial district is 250,000 to one

million; or

(C) \$5,000, if the population of the judicial district is more than one

million.

(c) This section does not apply to a political contribution made by a general-purpose committee.

(d) Repealed by Acts 2019, 86th Leg., R.S., Ch. 384 (H.B. 3233), Sec. 17(1), eff. June 2, 2019.

(d-1) In addition to the contribution limits imposed on each contributor under this section, a judicial candidate or officeholder may not accept a political contribution in excess of \$50 from a person if:

(1) the person is part of a law firm group; and

(2) the contribution, when aggregated with all political contributions accepted by the candidate or officeholder from the same law firm group in connection with the election, would exceed six times the applicable contribution limit under this section.

(e) A person who receives a political contribution that violates this section shall return the contribution to the contributor not later than the later of:

(1) the last day of the reporting period in which the contribution is received; or

(2) the fifth day after the date the contribution is received.

(f) A person who violates this section is liable for a civil penalty not to exceed three times the amount of the political contributions accepted in violation of this section.

# Sec. 253.157. LIMIT ON CONTRIBUTION BY GENERAL-PURPOSE COMMITTEES.

(a) Repealed by Acts 2019, 86th Leg., R.S., Ch. 384 (H.B. 3233), Sec. 17(2), eff. June 2, 2019.

(a-1) A judicial candidate or officeholder may not knowingly accept political contributions from a general-purpose committee that, in the aggregate, exceed the contribution limits prescribed by this subsection in connection with an election in which the judicial candidate's name appears on the ballot. The contribution limits under this subsection are:

(1) for a statewide judicial office, \$25,000; or

(2) for any other judicial office, \$5,000.

(a-2) In addition to the contribution limits imposed on each contribution in Subsection (a-1), a judicial candidate or officeholder may not accept a political contribution in excess of \$50 from a general-purpose committee if the contribution, when aggregated with all political contributions from all general-purpose committees in connection with an election, would exceed:

(1) for a statewide judicial office, \$300,000;

(2) for the office of chief justice or justice, court of appeals:

(A) \$75,000, if the population of the judicial district is more than one

million; or

(B) \$52,500, if the population of the judicial district is one million or less;

or

(3) for an office other than an office included under Subdivision (1) or (2):

(A) \$52,500, if the population of the judicial district is more than one

million;

(B) \$30,000, if the population of the judicial district is 250,000 to one or

million; or

(C) \$15,000, if the population of the judicial district is less than 250,000.

(b) A person who receives a political contribution that violates this section shall return the contributor not later than the later of:

(1) the last day of the reporting period in which the contribution is received; or

(2) the fifth day after the date the contribution is received.

(c) A person who violates this section is liable for a civil penalty not to exceed three times the amount of the political contributions accepted in violation of this section.

(d) Repealed by Acts 2019, 86th Leg., R.S., Ch. 384 (H.B. 3233), Sec. 17(2), eff. June 2, 2019.

(e) Repealed by Acts 2019, 86th Leg., R.S., Ch. 384 (H.B. 3233), Sec. 17(2), eff. June 2, 2019.

### Sec. 253.158. CONTRIBUTION BY SPOUSE OR CHILD.

(a) For purposes of this subchapter, a contribution by the spouse of an individual is not considered to be a contribution by the individual.

(b) For purposes of this subchapter, a contribution by a child of an individual is considered to be a contribution by the individual.

**Sec. 253.159. EXCEPTION TO CONTRIBUTION LIMITS**. Section 253.155 does not apply to an individual who is related to the candidate or officeholder within the second degree by consanguinity, as determined under Subchapter B, Chapter 573, Government Code.

Sec. 253.1601. CONTRIBUTION TO CERTAIN COMMITTEES CONSIDERED CONTRIBUTION TO CANDIDATE OR OFFICEHOLDER. For purposes of Sections 253.155 and 253.157, a contribution to a specific-purpose committee for the purpose of supporting a judicial candidate, opposing the candidate's opponent, or assisting a judicial officeholder is considered to be a contribution to the candidate or officeholder.

# Sec. 253.161. USE OF CONTRIBUTION FROM NONJUDICIAL OR JUDICIAL OFFICE PROHIBITED.

(a) A judicial candidate or officeholder, a specific-purpose committee for supporting or opposing a judicial candidate, or a specific-purpose committee for assisting a judicial officeholder may not use a political contribution to make a campaign expenditure for judicial office or to make an officeholder expenditure in connection with a judicial office if the contribution was accepted while the candidate or officeholder:

(1) was a candidate for an office other than a judicial office; or

(2) held an office other than a judicial office, unless the person had become a candidate for judicial office and the contribution was made in connection with an election for judicial office.

(b) A candidate, officeholder, or specific-purpose committee for supporting, opposing, or assisting the candidate or officeholder may not use a political contribution to make a campaign expenditure for an office other than a judicial office or to make an officeholder expenditure in connection with an office other than a judicial office if the contribution was accepted while the candidate or officeholder:

(1) was a candidate for a judicial office; or

(2) held a judicial office, unless the person had become a candidate for another office and the contribution was made in connection with an election for judicial office.

(c) Repealed by Acts 2019, 86th Leg., R.S., Ch. 384 (H.B. 3233), Sec. 17(4), eff. June 2, 2019.

(d) A person who violates this section is liable for a civil penalty not to exceed three times the amount of political contributions used in violation of this section.

# Sec. 253.1611. CERTAIN CONTRIBUTIONS BY JUDICIAL CANDIDATES, OFFICEHOLDERS, AND COMMITTEES RESTRICTED.

(a) A judicial candidate or officeholder or a specific-purpose committee for supporting or opposing a judicial candidate or assisting a judicial officeholder may not use a political contribution to knowingly make political contributions that in the aggregate exceed \$100 in a calendar year to a candidate or officeholder.

(b) A judicial candidate or a specific-purpose committee for supporting or opposing a judicial candidate may not use a political contribution to knowingly make political contributions to a political committee in connection with a primary election.

(c) A judicial candidate or a specific-purpose committee for supporting or opposing a judicial candidate may not use a political contribution to knowingly make a political contribution to a political committee that, when aggregated with each other political contribution to a political committee in connection with a general election, exceeds \$500.

(d) A judicial officeholder or a specific-purpose committee for assisting a judicial officeholder may not, in any calendar year in which the office held is not on the ballot, use a political contribution to knowingly make a political contribution to a political committee that, when aggregated with each other political contribution to a political committee in that calendar year, exceeds \$250.

(e) This section does not apply to a political contribution made to the principal political committee of the state executive committee or a county executive committee of a political party that *provides* goods or services, including political advertising or a campaign communication, to or for the benefit of judicial candidates.

(e-1) This subsection applies only to a political party required to nominate candidates by primary election. This section does not apply to a political contribution made, for the purpose of sponsoring or attending an event, to a political committee affiliated with:

(1) an organization that has been designated as an auxiliary, coalition, or county chair association of a political party as provided by political party rule or state executive committee bylaw; or

(2) a local chapter of an organization described by Subdivision (1).

(f) Repealed by Acts 2017, 85th Leg., R.S., Ch. 905 (H.B. 3903), Sec. 2, eff. June 15, 2017.

(g) A person who violates this section is liable for a civil penalty not to exceed three times the amount of political contributions used in violation of this section.

**Sec. 253.1612. CERTAIN CAMPAIGN ACTIVITIES AUTHORIZED.** The Code of Judicial Conduct may not prohibit, and a judicial candidate may not be penalized for, a joint campaign activity conducted by two or more judicial candidates.

# Sec. 253.162. RESTRICTIONS ON REIMBURSEMENT OF PERSONAL FUNDS AND PAYMENTS ON CERTAIN LOANS.

(a) A judicial candidate or officeholder who makes political expenditures from the person's personal funds or who accepts one or more political contributions in the form of a loan, including an extension of credit or guarantee of a loan or extension of credit, from one or more persons related to the candidate or officeholder within the second degree of affinity or

consanguinity, as determined under Subchapter B, Chapter 573, Government Code, may not reimburse those personal funds or repay those loans from political contributions in amounts that in the aggregate exceed, for each election in which the person's name appears on the ballot:

(1) for a statewide judicial office, \$100,000; or

(2) for an office other than a statewide judicial office, five times the applicable contribution limit under Section 253.155.

(b) Repealed by Acts 2019, 86th Leg., R.S., Ch. 384 (H.B. 3233), Sec. 17(5), eff. June 2, 2019.

(c) A person who is both a candidate and an officeholder may reimburse the person's personal funds in only one capacity.

(d) A person who violates this section is liable for a civil penalty not to exceed three times the amount by which the reimbursement made in violation of this section exceeds the applicable limit prescribed by Subsection (a).

## Sec. 253.1621. APPLICATION OF CONTRIBUTION AND REIMBURSEMENT LIMITS TO CERTAIN CANDIDATES.

(a) For purposes of the contribution limits prescribed by Section 253.155 or 253.157 and the limit on reimbursement of personal funds and repayment of certain loans prescribed by Section 253.162, the general and primary elections are considered separate elections for a candidate whose name appears on the ballot.

(b) For purposes of the contribution limits prescribed by Sections 253.155 and 253.157 and the limits on reimbursement of personal funds and repayment of certain loans prescribed by Section 253.162, a runoff election in which the candidate's name is on the ballot is considered a separate election.

# Sec. 253.167. CERTIFICATION OF POPULATION; NOTICE OF CONTRIBUTION LIMITS.

(a) For purposes of this subchapter only, not later than June 1 of each odd-numbered year, the commission shall:

(1) make a written certification of the population of each judicial district for which a candidate for judge or justice must file a campaign treasurer appointment with the commission; and

(2) deliver to the county clerk of each county a written certification of the county's population, if the county:

(A) comprises an entire judicial district under Chapter 26, Government Code; or

(B) has a statutory county court or statutory probate court, other than a multicounty statutory county court created under Subchapter D, Chapter 25, Government Code.

(b) Following certification of population under Subsection (a), the commission or county clerk, as appropriate, shall make available to each candidate for an office covered by this subchapter written notice of the contribution limits applicable to the office the candidate seeks.

(c) The commission shall post the written certification required by this section on the commission's Internet website.

Sec. 253.171. CONTRIBUTION FROM OR DIRECT CAMPAIGN EXPENDITURE BY POLITICAL PARTY. A political expenditure that is made by the principal political committee of the state executive committee or a county executive committee

of a political party for a generic get-out-the-vote campaign or to create and distribute a written list of two or more candidates is not considered a contribution to a judicial candidate who benefits from the get-out-the-vote campaign or is included in the written list and is not subject to the limits of Section 253.155 or 253.157 if the get-out-the-vote campaign or written list:

(1) identifies the party's candidates by name and office sought, office held, or photograph;

(2) does not include any reference to the judicial philosophy or positions on issues of the party's judicial candidates; and

(3) is not broadcast, cablecast, published in a newspaper or magazine, or placed on a billboard.

### Sec. 253.176. CIVIL PENALTY.

(a) The commission may impose a civil penalty against a person as provided by this subchapter only after a formal hearing as provided by Subchapter E, Chapter 571, Government Code.

(b) The commission shall base the amount of the penalty on:

(1) the seriousness of the violation;

(2) the history of previous violations;

(3) the amount necessary to deter future violations; and

(4) any other matter that justice may require.

(c) Repealed by Acts 2019, 86th Leg., R.S., Ch. 384 (H.B. 3233), Sec. 17(7), eff. June 2,

2019.

### **CHAPTER 254. POLITICAL REPORTING**

#### SUBCHAPTER A. RECORDKEEPING

**Sec. 254.001. RECORDKEEPING REQUIRED**. (a) Each candidate and each officeholder shall maintain a record of all reportable activity.

(b) Each campaign treasurer of a political committee shall maintain a record of all reportable activity.

(c) The record must contain the information that is necessary for filing the reports required by this chapter.

(d) A person required to maintain a record under this section shall preserve the record for at least two years beginning on the filing deadline for the report containing the information in the record.

(e) A person who violates this section commits an offense. An offense under this section is a Class B misdemeanor.

### SUBCHAPTER B. POLITICAL REPORTING GENERALLY

#### Sec. 254.031. GENERAL CONTENTS OF REPORTS.

(a) Except as otherwise provided by this chapter, each report filed under this chapter must include:

(1) the amount of political contributions, other than political contributions described by Subdivision (1-a), from each person that in the aggregate exceed \$50 and that are accepted during the reporting period by the person or committee required to file a report under this chapter, the full name and address of the person making the contributions, and the dates of the contributions;

(1-a) the amount of political contributions from each person that are made electronically and that are accepted during the reporting period by the person or committee required to file a report under this chapter, the full name and address of the person making the contributions, and the dates of the contributions;

(2) the amount of loans that are made during the reporting period for campaign or officeholder purposes to the person or committee required to file the report and that in the aggregate exceed \$50, the dates the loans are made, the interest rate, the maturity date, the type of collateral for the loans, if any, the full name and address of the person or financial institution making the loans, the full name and address, principal occupation, and name of the employer of each guarantor of the loans, the amount of the loans guaranteed by each guarantor, and the aggregate principal amount of all outstanding loans as of the last day of the reporting period;

(3) the amount of political expenditures that in the aggregate exceed \$100 and that are made during the reporting period, the full name and address of the persons to whom the expenditures are made, and the dates and purposes of the expenditures;

(4) the amount of each payment made during the reporting period from a political contribution if the payment is not a political expenditure, the full name and address of the person to whom the payment is made, and the date and purpose of the payment;

(5) the total amount or a specific listing of the political contributions of \$50 or less accepted and the total amount or a specific listing of the political expenditures of \$100 or less made during the reporting period;

(6) the total amount of all political contributions accepted and the total amount of all political expenditures made during the reporting period;

(7) the name of each candidate or officeholder who benefits from a direct campaign expenditure made during the reporting period by the person or committee required to file the report, and the office sought or held, excluding a direct campaign expenditure that is made by the principal political committee of a political party on behalf of a slate of two or more nominees of that party;

(8) as of the last day of a reporting period for which the person is required to file a report, the total amount of political contributions accepted, including interest or other income on those contributions, maintained in one or more accounts in which political contributions are deposited as of the last day of the reporting period;

(9) any credit, interest, rebate, refund, reimbursement, or return of a deposit fee resulting from the use of a political contribution or an asset purchased with a political contribution that is received during the reporting period and the amount of which exceeds \$100;

(10) any proceeds of the sale of an asset purchased with a political contribution that is received during the reporting period and the amount of which exceeds \$100;

(11) any investment purchased with a political contribution that is received during the reporting period and the amount of which exceeds \$100;

(12) any other gain from a political contribution that is received during the reporting period and the amount of which exceeds \$100; and

(13) the full name and address of each person from whom an amount described by Subdivision (9), (10), (11), or (12) is received, the date the amount is received, and the purpose for which the amount is received.

(a-1) A de minimis error in calculating or reporting a cash balance under Subsection (a)(8) is not a violation of this section.

(b) If no reportable activity occurs during a reporting period, the person required to file a report shall indicate that fact in the report.

## Sec. 254.0311. REPORT BY LEGISLATIVE CAUCUS.

(a) A legislative caucus shall file a report of contributions and expenditures as required by this section.

(b) A report filed under this section must include:

(1) the amount of contributions from each person, other than a caucus member, that in the aggregate exceed \$50 and that are accepted during the reporting period by the legislative caucus, the full name and address of the person making the contributions, and the dates of the contributions;

(2) the amount of loans that are made during the reporting period to the legislative caucus and that in the aggregate exceed \$50, the dates the loans are made, the interest rate, the maturity date, the type of collateral for the loans, if any, the full name and address of the person or financial institution making the loans, the full name and address, principal occupation, and name of the employer of each guarantor of the loans, the amount of the loans guaranteed by each guarantor, and the aggregate principal amount of all outstanding loans as of the last day of the reporting period;

(3) the amount of expenditures that in the aggregate exceed \$50 and that are made during the reporting period, the full name and address of the persons to whom the expenditures are made, and the dates and purposes of the expenditures;

(4) the total amount or a specific listing of contributions of \$50 or less accepted from persons other than caucus members and the total amount or a specific listing of expenditures of \$50 or less made during the reporting period; and

(5) the total amount of all contributions accepted, including total contributions from caucus members, and the total amount of all expenditures made during the reporting period.

(c) If no reportable activity occurs during a reporting period, the legislative caucus shall indicate that fact in the report.

(d) A legislative caucus shall file with the commission two reports for each year.

(e) The first report shall be filed not later than July 15. The report covers the period beginning January 1 or the day the legislative caucus is organized, as applicable, and continuing through June 30.

(f) The second report shall be filed not later than January 15. The report covers the period beginning July 1 or the day the legislative caucus is organized, as applicable, and continuing through December 31.

(g) A legislative caucus shall maintain a record of all reportable activity under this section and shall preserve the record for at least two years beginning on the filing deadline for the report containing the information in the record.

(h) In this section, "legislative caucus" has the meaning assigned by Section 253.0341.

### Sec. 254.0312. BEST EFFORTS.

(a) A person required to file a report under this chapter is considered to be in compliance with Section 254.0612, 254.0912, or 254.1212 only if the person or the person's campaign treasurer shows that the person has used best efforts to obtain, maintain, and report the information required by those sections. A person is considered to have used best efforts to obtain, maintain, and report that information if the person or the person's campaign treasurer complies with this section.

(b) Each written solicitation for political contributions from an individual must include:

(1) a clear request for the individual's full name and address, the individual's principal occupation or job title, and the full name of the individual's employer; and

(2) an accurate statement of state law regarding the collection and reporting of individual contributor information, such as:

(A) "State law requires (certain candidates, officeholders, or political committees, as applicable) to use best efforts to collect and report the full name and address, principal occupation or job title, and full name of employer of individuals whose contributions equal or exceed \$500 in a reporting period."; or

(B) "To comply with state law, (certain candidates, officeholders, or political committees, as applicable) must use best efforts to obtain, maintain, and report the full name and address, principal occupation or job title, and full name of employer of individuals whose contributions equal or exceed \$500 in a reporting period.".

(c) For each political contribution received from an individual that, when aggregated with all other political contributions received from the individual during the reporting period, equals or exceeds \$500 and for which the information required by Section 254.0612, 254.0912, or 254.1212 is not provided, the person must make at least one oral or written request for the missing information. A request under this subsection:

(1) must be made not later than the 30th day after the date the contribution is received;

(2) must include a clear and conspicuous statement that complies with Subsection (b);

(3) if made orally, must be documented in writing; and

(4) may not be made in conjunction with a solicitation for an additional political contribution.

(d) A person must report any information required by Section 254.0612, 254.0912, or 254.1212 that is not provided by the individual making the political contribution and that the person has in the person's records of political contributions or previous reports under this chapter.

(e) A person who receives information required by Section 254.0612, 254.0912, or 254.1212 after the filing deadline for the report on which the contribution is reported must include the missing information on the next report the person is required to file under this chapter.

#### Sec. 254.0313. OMISSION OF ADDRESS FOR JUDGE AND FAMILY MEMBER.

(a) In this section, "family member" has the meaning assigned by Section 31.006, Finance Code.

(b) On receiving notice from the Office of Court Administration of the Texas Judicial System of a judge's qualification for office or on receipt of a written request from a federal judge, including a federal bankruptcy judge, a state judge, or a family member of a federal judge, including a federal bankruptcy judge, or a state judge, the commission shall remove or redact the residence address of a federal judge, including a federal bankruptcy judge, a state judge, or the family member of a federal judge, including a federal bankruptcy judge, or a state judge, from any report filed by the judge in the judge's capacity or made available on the Internet under this chapter.

**Sec. 254.032. NONREPORTABLE PERSONAL TRAVEL EXPENSE**. A political contribution consisting of personal travel expense incurred by an individual is not required to be reported under this chapter if the individual receives no reimbursement for the expense.

Sec. 254.033. NONREPORTABLE PERSONAL SERVICE. A political contribution consisting of an individual's personal service is not required to be reported under this chapter if the individual receives no compensation for the service.

#### Sec. 254.034. TIME OF ACCEPTING CONTRIBUTION.

(a) A determination to accept or refuse a political contribution that is received by a candidate, officeholder, or political committee shall be made not later than the end of the reporting period during which the contribution is received.

(b) If the determination to accept or refuse a political contribution is not made before the time required by Subsection (a), for purposes of this chapter, the contribution is considered to have been accepted on the last day of that reporting period.

(c) A political contribution that is received but not accepted shall be returned to the contributor not later than the 30th day after the deadline for filing a report for the reporting period during which the contribution is received. A contribution not returned within that time is considered to be accepted.

(d) A candidate, officeholder, or political committee commits an offense if the person knowingly fails to return a political contribution as required by Subsection (c).

(e) An offense under this section is a Class A misdemeanor.

#### Sec. 254.035. TIME OF MAKING EXPENDITURE.

(a) For purposes of reporting under this chapter, a political expenditure is not considered to have been made until the amount is readily determinable by the person making the expenditure, except as provided by Subsection (b).

(b) If the character of an expenditure is such that under normal business practice the amount is not disclosed until receipt of a periodic bill, the expenditure is not considered made until the date the bill is received.

(c) The amount of a political expenditure made by credit card is readily determinable by the person making the expenditure on the date the person receives the credit card statement that includes the expenditure.

(d) Subsection (c) does not apply to a political expenditure made by credit card during the period covered by a report required to be filed under Section 254.064(b) or (c), 254.124(b) or (c), or 254.154(b) or (c).

### Sec. 254.036. FORM OF REPORT; AFFIDAVIT; MAILING OF FORMS.

(a) Each report filed under this chapter with an authority other than the commission must be in a format prescribed by the commission. A report filed with the commission that is not required to be filed by computer diskette, modem, or other means of electronic transfer must be on a form prescribed by the commission and written in black ink or typed with black typewriter ribbon or, if the report is a computer printout, the printout must conform to the same format and paper size as the form prescribed by the commission.

(b) Except as provided by Subsection (c) or (e), each report filed under this chapter with the commission must be filed by computer diskette, modem, or other means of electronic transfer, using computer software provided by the commission or computer software that meets commission specifications for a standard file format.

(c) A candidate, officeholder, or political committee that is required to file reports with the commission may file reports that comply with Subsection (a) if:

(1) the candidate, officeholder, or campaign treasurer of the committee files with the commission an affidavit stating that the candidate, officeholder, or committee, an agent of the candidate, officeholder, or committee, or a person with whom the candidate, officeholder, or committee contracts does not use computer equipment to keep the current records of political contributions, political expenditures, or persons making political contributions to the candidate, officeholder, or committee; and

(2) the candidate, officeholder, or committee does not, in a calendar year, accept political contributions that in the aggregate exceed \$20,000 or make political expenditures that in the aggregate exceed \$20,000.

(c-1) An affidavit under Subsection (c) must be filed with each report filed under Subsection (a). The affidavit must include a statement that the candidate, officeholder, or political committee understands that the candidate, officeholder, or committee shall file reports as required by Subsection (b) if:

(1) the candidate, officeholder, or committee, a consultant of the candidate, officeholder, or committee, or a person with whom the candidate, officeholder, or committee contracts uses computer equipment for a purpose described by Subsection (c); or

(2) the candidate, officeholder, or committee exceeds \$20,000 in political contributions or political expenditures in a calendar year.

(d) Repealed by Acts 2003, 78th Leg., ch. 249, Sec. 2.26.

(e) A candidate for an office described by Section 252.005(5) or a specific-purpose committee for supporting or opposing only candidates for an office described by Section 252.005(5) or a measure described by Section 252.007(5) may file reports that comply with Subsection (a).

(f) In prescribing the format of a report filed under this chapter with an authority other than the commission, the commission shall ensure that:

(1) a report may be filed:

(A) by first class United States mail or common or contract carrier;

(B) by personal delivery; or

(C) by electronic filing, if the authority with whom the report is required to be filed has adopted rules and procedures to provide for the electronic filing of the report and the report is filed in accordance with those rules and procedures; and

(2) an authority with whom a report is electronically filed issues an electronic receipt for the report to the person filing the report.

(g) In prescribing the format of a report filed under this chapter, including a report filed with an authority other than the commission under Subsection (a), the commission shall ensure that the report requires for political expenditures made with a credit card that:

(1) the expenditures be reported in a single itemized list; and

- (2) the list include, stated by credit card issuer:
  - (A) the name of the credit card issuer;
  - (B) the date and amount of each expenditure; and
  - (C) the date the credit card issuer was repaid for the expenditure.

(h) Each report filed under this chapter that is not filed by electronic transfer must be accompanied by an affidavit executed by the person required to file the report. The affidavit must contain the statement: "I swear, or affirm, under penalty of perjury, that the accompanying report is true and correct and includes all information required to be reported by me under Title 15, Election Code." Each report filed under this chapter by electronic transfer must be under oath by the person required to file the report and must contain, in compliance with commission specifications, the digitized signature of the person required to file the report. A report filed under this chapter is considered to be under oath by the person required to file the report, and the person is subject to prosecution under Chapter 37, Penal Code, regardless of the absence of or a defect in the affidavit.

(i) Each person required to file reports with the commission that comply with Subsection (b) shall file with the commission a written statement providing the manner of electronic transfer that the person will use to file the report. A statement under this subsection must be filed not later than the 30th day before the filing deadline for the first report a person is required to file under Subsection (b). A person who intends to change the manner of filing described by the person's most recent statement shall notify the commission of the change not later than the 30th day before the filing deadline for the report to which the change applies. If a person does not file a statement under this subsection, the commission may accept as authentic a report filed in any manner that complies with Subsection (b). If the commission receives a report that is not filed in the manner described by the person's most recent statement under this subsection, the commission shall promptly notify the person in writing that the commission has received a report filed in a different manner than expected.

(j) As part of the notification required by Section 251.033, the commission shall mail the appropriate forms to each person required to file a report with the commission during that reporting period.

(k) The commission shall prescribe forms for purposes of legislative caucus reports under Section 254.0311 that are separate and distinct from forms for other reports under this chapter.

(1) This section applies to a report that is filed electronically or otherwise.

## Sec. 254.0362. USE OF PUBLICLY ACCESSIBLE COMPUTER TERMINAL FOR PREPARATION OF REPORTS.

(a) Except as provided by Subsection (d), a person who is required to file reports under this chapter may use a publicly accessible computer terminal that has Internet access and web browser software to prepare the reports.

(b) A public entity may prescribe reasonable restrictions on the use of a publicly accessible computer terminal for preparation of reports under this chapter, except that a public entity may not prohibit a person from using a computer terminal for preparation of reports during the public entity's regular business hours if the person requests to use the computer terminal less than 48 hours before a reporting deadline to which the person is subject.

(c) This section does not require a public entity to provide a person with consumable materials, including paper and computer diskettes, in conjunction with the use of a publicly accessible computer terminal.

(d) An officeholder may not use a computer issued to the officeholder for official use to prepare a report under this title.

(e) In this section:

(1) "Public entity" means a state agency, city, county, or independent school district.

(2) "Publicly accessible computer terminal" means a computer terminal that is normally available for use by members of the public and that is owned by a state agency, an independent school district, or a public library operated by a city or county.

## Sec. 254.037. FILING DEADLINE.

(a) Except as provided by Subsection (b), the deadline for filing a report required by this chapter is 5 p.m. on the last day permitted under this chapter for filing the report.

(b) The deadline for filing a report electronically with the commission as required by this chapter is midnight on the last day for filing the report.

# Sec. 254.038. SPECIAL REPORT NEAR ELECTION BY CERTAIN CANDIDATES AND POLITICAL COMMITTEES.

(a) In addition to other reports required by this chapter, the following persons shall file additional reports during the period beginning the ninth day before election day and ending at 12 noon on the day before election day:

(1) a candidate for an office specified by Section 252.005(1) who accepts political contributions from a person that in the aggregate exceed \$1,000 during that reporting period; and

(2) a specific-purpose committee for supporting or opposing a candidate described by Subdivision (1) and that accepts political contributions from a person that in the aggregate exceed \$1,000 during that reporting period.

(b) Each report required by this section must include the amount of the contributions specified by Subsection (a), the full name and address of the person making the contributions, and the dates of the contributions.

(c) A report under this section shall be filed electronically, by [telegram or] telephonic facsimile machine, or by hand, in the form required by Section 254.036. The commission must receive a report under this section filed by [telegram,] telephonic facsimile machine or hand not later than 5 p.m. of the first business day after the date the contribution is accepted. The commission must receive a report under this section filed electronically not later than midnight of the first business day after the date the contribution is accepted. A report under this section is not required to be accompanied by the affidavit required under Section 254.036(h) or to be submitted on a form prescribed by the commission. A report under this section that complies with Section 254.036(a) must be accompanied by an affidavit under Section 254.036(c)(1) unless the candidate or committee has submitted an affidavit under Section 254.036(c)(1) with another report filed in connection with the election for which a report is required under this section.

(d) To the extent of a conflict between this section and Section 254.036, this section controls.

## Sec. 254.039. SPECIAL REPORT NEAR ELECTION BY CERTAIN GENERAL-PURPOSE COMMITTEES.

(a) In addition to other reports required by this chapter, a general-purpose committee shall file additional reports during the period beginning the ninth day before election day and ending at 12 noon on the day before election day if the committee:

(1) accepts political contributions from a person that in the aggregate exceed \$5,000 during that reporting period; or

(2) makes direct campaign expenditures supporting or opposing either a single candidate that in the aggregate exceed \$1,000 or a group of candidates that in the aggregate exceed \$15,000 during that reporting period.

(a-1) A report under this section shall be filed electronically, by [telegram or] telephonic facsimile machine, or by hand, in the form required by Section 254.036. The commission must receive a report under this section not later than 5 p.m. of the first business day after the date the contribution is accepted or the expenditure is made. A report under this section is not required to be accompanied by the affidavit required under Section 254.036(h) or to be submitted on a form prescribed by the commission. A report under this section that complies with Section 254.036(a) must be accompanied by an affidavit under Section 254.036(c)(1) unless the committee has submitted an affidavit under Section 254.036(c)(1) with another report filed in connection with the election for which a report is required under this section.

(a-2) Each report required by Subsection (a)(1) must include the amount of the contributions specified by that subsection, the full name and address of the person making the contributions, and the dates of the contributions.

(b) Each report required by Subsection (a)(2) must include the amount of the expenditures, the full name and address of the persons to whom the expenditures are made, and the dates and purposes of the expenditures.

(c) To the extent of a conflict between this section and Section 254.036, this section controls.

#### Sec. 254.0391. REPORT DURING SPECIAL LEGISLATIVE SESSION.

(a) A statewide officeholder, a member of the legislature, or a specific-purpose committee for supporting, opposing, or assisting a statewide officeholder or member of the legislature, or a candidate for statewide office or the legislature or a specific-purpose committee for supporting or opposing the candidate, that accepts a political contribution during the period beginning on the date the governor signs the proclamation calling a special legislative session and continuing through the date of final adjournment shall report the contribution to the commission not later than the 30th day after the date of final adjournment.

(b) A determination to accept or refuse the political contribution shall be made not later than the third day after the date the contribution is received.

(c) Each report required by this section must include the amount of the political contribution, the full name and address of the person making the contribution, and the date of the contribution.

(d) A report is not required under this section if a person covered by Subsection (a) is required to file another report under this chapter not later than the 10th day after the date a report required under this section would be due.

#### Sec. 254.040. PRESERVATION OF REPORTS; RECORD OF INSPECTION.

(a) Each report filed under this chapter shall be preserved by the authority with whom it is filed for at least two years after the date it is filed.

(b) Each time a person requests to inspect a report, the commission shall place in the file a statement of the person's name and address, whom the person represents, and the date of the request. The commission shall retain that statement in the file for one year after the date the requested report is filed. This subsection does not apply to a request to inspect a report by:

(1) a member or employee of the commission acting on official business; or

(2) an individual acting on the individual's own behalf.

#### Sec. 254.0401. AVAILABILITY OF REPORTS ON INTERNET.

(a) The commission shall make each report filed with the commission under Section 254.036(b) available to the public on the Internet not later than the second business day after the date the report is filed.

(a-1) *Repealed*.

(b) The clerk or secretary of a political subdivision's governing body or, if the governing body does not have a clerk or secretary, the governing body's presiding officer shall make a report filed with the political subdivision by a candidate, officeholder, or specific-purpose committee under this subchapter available to the public on the political subdivision's Internet website not later than the 10th business day after the date the report is received.

(c) *Repealed*.

(d) The access allowed by this section to reports is in addition to the public's access to the information through other electronic or print distribution of the information.

(e) Before making a report filed under Section 254.036(b) available on the Internet, the commission shall remove each portion, other than city, state, and zip code, of the address of a person listed as having made a political contribution to the person filing the report. The address information removed must remain available on the report maintained in the commission's office but may not be available electronically at that office.

(e-1) Before making a report available on the Internet as required by Subsection (b), the authority with whom the report is filed may remove each portion, other than city, state, and zip

code, of the address of a person listed as having made a political contribution to the person filing the report. The address information removed must remain available on the report maintained in the authority's office.

(f) The commission shall clearly state on the Internet website on which reports are provided that reports filed by an independent candidate, a third-party candidate, or a specific-purpose committee for supporting or opposing an independent or third-party candidate will not be available if the candidate or committee has not yet filed a report.

(g) Electronic report data saved in a temporary storage location of the authority with whom the report is filed for later retrieval and editing before the report is filed is confidential and may not be disclosed. After the report is filed with the authority, the information disclosed in the filed report is public information to the extent provided by this title.

(h) A report made available on an Internet website under this section must be accessible on that website until the fifth anniversary of the date the report is first made available.

#### Sec. 254.04011. Repealed.

#### Sec. 254.0402. PUBLIC INSPECTION OF REPORTS.

(a) Notwithstanding Section 552.222(a), Government Code, the authority with whom a report is filed under this chapter may not require a person examining the report to provide any information or identification.

(b) The commission shall make information from reports filed with the commission under Section 254.036(b) available by electronic means, including:

(1) providing access to computer terminals at the commission's office;

(2) providing information on computer diskette for purchase at a reasonable

cost; and

(3) providing modem or other electronic access to the information.

## Sec. 254.0405. AMENDMENT OF FILED REPORT.

(a) A person who files a semiannual report under this chapter may amend the report.

(b) A semiannual report that is amended before the eighth day after the date the original report was filed is considered to have been filed on the date on which the original report was filed.

(c) A semiannual report that is amended on or after the eighth day after the original report was filed is considered to have been filed on the date on which the original report was filed if:

(1) the amendment is made before any complaint is filed with regard to the subject of the amendment; and

(2) the original report was made in good faith and without an intent to mislead or to misrepresent the information contained in the report.

## Sec. 254.041. CRIMINAL PENALTY FOR UNTIMELY OR INCOMPLETE REPORT.

(a) A person who is required by this chapter to file a report commits an offense if the person knowingly fails:

(1) to file the report on time;

(2) to file a report by computer diskette, modem, or other means of electronic transfer, if the person is required to file reports that comply with Section 254.036(b); or

(3) to include in the report information that is required by this title to be included.

(b) Except as provided by Subsection (c), an offense under this section is a Class C misdemeanor.

(c) A violation of Subsection (a)(3) by a candidate or officeholder is a Class A misdemeanor if the report fails to include information required by Section 254.061(3) or Section 254.091(2), as applicable.

(d) It is an exception to the application of Subsection (a)(3) that:

(1) the information was required to be included in a semiannual report; and

(2) the person amended the report within the time prescribed by Section 254.0405(b) or under the circumstances described by Section 254.0405(c).

### Sec. 254.042. CIVIL PENALTY FOR LATE REPORT.

(a) The commission shall determine from any available evidence whether a report required to be filed with the commission under this chapter is late. On making that determination, the commission shall immediately mail a notice of the determination to the person required to file the report.

(b) If a report other than a report under Section 254.064(c), 254.124(c), or 254.154(c) or the first report under Section 254.063 or 254.123 that is required to be filed following the primary or general election is determined to be late, the person required to file the report is liable to the state for a civil penalty of \$500. If a report under Section 254.064(c), 254.124(c), or 254.154(c) or the first report under Section 254.063 or 254.153 that is required to be filed following the primary or general election is determined to be late, the person required to be filed following the primary or general election is determined to be late, the person required to file the report is liable to the state for a civil penalty of \$500 for the first day the report is late and \$100 for each day thereafter that the report is late. If a report is more than 30 days late, the commission shall issue a warning of liability by registered mail to the person required to file the report. If the penalty is not paid before the 10th day after the date on which the warning is received, the person is liable for a civil penalty in an amount determined by commission rule, but not to exceed \$10,000.

(c) A penalty paid voluntarily under this section shall be deposited in the State Treasury to the credit of the General Revenue Fund.

(d) Repealed by Acts 1991, 72nd Leg., ch. 304, Sec. 5.20, eff. Jan. 1, 1992.

## Sec. 254.043. ACTION TO REQUIRE COMPLIANCE.

(a) This section applies only to:

(1) a person required to file reports under this chapter with the commission; or

(2) a person required to file reports under this chapter with an authority other than the commission in connection with an office of a political subdivision in a county with a population of at least 500,000.

(b) A resident of the territory served by an office may bring an action for injunctive relief against a candidate for or holder of that office or a specific-purpose committee for supporting or opposing such a candidate or assisting such an officeholder to require the person to file a report under this chapter that the person has failed to timely file.

(c) An action under this section may be brought against a person required to file reports under this chapter only if:

(1) the report is not filed before the 60th day after the date on which the report was required to be filed;

(2) not earlier than the 60th day after the date on which the report was required to be filed, the person bringing the action delivers written notice by certified mail to the person required to file the report, stating:

(A) the person's intention to bring an action under this section if the report is not filed; and

(B) that an action to require the filing of the report may be filed if the report is not filed before the 30th day after the date on which the person required to file the report receives the notice; and

(3) the report is not filed before the 30th day after the date on which the person required to file the report receives the notice required by Subdivision (2).

(d) The court shall award a plaintiff who prevails in an action under this section reasonable attorney's fees and court costs.

## <u>Sec. 254.044. REPORTING OF POLITICAL CONTRIBUTIONS AND</u> EXPENDITURES MADE USING CREDIT CARD.

(a) A candidate or officeholder who accepts a political contribution made using a credit card shall:

(1) for a political contribution for which a processing fee is deducted by the credit card issuer from the political contribution amount:

(A) report as a political contribution the full amount, including the deducted amount; and

(B) report as a political expenditure the deducted amount; and

(2) for a political contribution for which a processing fee is paid by the person making the political contribution in excess of the political contribution amount, report only as a political contribution the full amount the candidate or officeholder accepts, not including the amount paid in excess of the political contribution amount.

(b) A candidate or officeholder who accepts a political contribution described by Subsection (a)(2) is not required to report the excess amount paid as a processing fee by the person making the political contribution.

## SUBCHAPTER C. REPORTING BY CANDIDATE

**Sec. 254.061. ADDITIONAL CONTENTS OF REPORTS**. In addition to the contents required by Section 254.031, each report by a candidate must include:

(1) the candidate's full name and address, the office sought, and the identity and date of the election for which the report is filed;

(2) the campaign treasurer's name, residence or business street address, and telephone number;

(3) for each political committee from which the candidate received notice under Section 254.128 or 254.161:

(A) the committee's full name and address;

(B) an indication of whether the committee is a general-purpose committee or a specific-purpose committee; and

(C) the full name and address of the committee's campaign treasurer; and

(4) on a separate page or pages of the report, the identification of any payment from political contributions made to a business in which the candidate has a participating interest

of more than 10 percent, holds a position on the governing body of the business, or serves as an officer of the business.

## Sec. 254.0611. ADDITIONAL CONTENTS OF REPORTS BY CERTAIN JUDICIAL CANDIDATES.

(a) In addition to the contents required by Sections 254.031 and 254.061, each report by a candidate for a judicial office covered by Subchapter F, Chapter 253, must include:

(1) the total amount of political contributions, including interest or other income, maintained in one or more accounts in which political contributions are deposited as of the last day of the reporting period;

(2) for each individual from whom the person filing the report has accepted political contributions that in the aggregate exceed \$50 and that are accepted during the reporting period:

(A) the principal occupation and job title of the individual and the full name of the employer of the individual or of the law firm of which the individual or the individual's spouse is a member, if any; or

(B) if the individual is a child, the full name of the law firm of which either of the individual's parents is a member, if any;

(3) a specific listing of each asset valued at \$500 or more that was purchased with political contributions and on hand as of the last day of the reporting period;

(4) for each political contribution accepted by the person filing the report but not received as of the last day of the reporting period:

(A) the full name and address of the person making the contribution;

(B) the amount of the contribution; and

(C) the date of the contribution; and

(5) for each outstanding loan to the person filing the report as of the last day of the reporting period:

(A) the full name and address of the person or financial institution making the loan; and

(B) the full name and address of each guarantor of the loan other than

the candidate.

(b) In this section:

(1) "Child" and "law firm" have the meanings assigned by Section 253.152.

(2) "Member" has the meaning assigned to "member of a law firm" by Section

253.152.

Sec. 254.0612. ADDITIONAL CONTENTS OF REPORTS BY CANDIDATE FOR STATEWIDE EXECUTIVE OFFICE OR LEGISLATIVE OFFICE. In addition to the contents required by Sections 254.031 and 254.061, each report by a candidate for a statewide office in the executive branch or a legislative office must include, for each individual from whom the person filing the report has accepted political contributions that in the aggregate equal or exceed \$500 and that are accepted during the reporting period:

- (1) the individual's principal occupation or job title; and
- (2) the full name of the individual's employer.

Sec. 254.062. CERTAIN OFFICEHOLDER ACTIVITY INCLUDED. If an officeholder who becomes a candidate has reportable activity that is not reported under Subchapter D before the end of the period covered by the first report the candidate is required to file under this subchapter, the reportable activity shall be included in the first report filed under this subchapter instead of in a report filed under Subchapter D.

### Sec. 254.063. SEMIANNUAL REPORTING SCHEDULE FOR CANDIDATE.

(a) A candidate shall file two reports for each year as provided by this section.

(b) The first report shall be filed not later than July 15. The report covers the period beginning January 1, the day the candidate's campaign treasurer appointment is filed, or the first day after the period covered by the last report required to be filed under this subchapter, as applicable, and continuing through June 30.

(c) The second report shall be filed not later than January 15. The report covers the period beginning July 1, the day the candidate's campaign treasurer appointment is filed, or the first day after the period covered by the last report required to be filed under this subchapter, as applicable, and continuing through December 31.

#### Sec. 254.064. ADDITIONAL REPORTS OF OPPOSED CANDIDATE.

(a) In addition to other required reports, for each election in which a person is a candidate and has an opponent whose name is to appear on the ballot, the person shall file two reports.

(b) The first report must be received by the authority with whom the report is required to be filed not later than the 30th day before election day. The report covers the period beginning the day the candidate's campaign treasurer appointment is filed or the first day after the period covered by the last report required to be filed under this chapter, as applicable, and continuing through the 40th day before election day.

(c) The second report must be received by the authority with whom the report is required to be filed not later than the eighth day before election day. The report covers the period beginning the 39th day before election day and continuing through the 10th day before election day.

(d) If a person becomes an opposed candidate after a reporting period prescribed by Subsection (b) or (c), the person's first report must be received by the authority with whom the report is required to be filed not later than the regular deadline for the report covering the period during which the person becomes an opposed candidate. The period covered by the first report begins the day the candidate's campaign treasurer appointment is filed.

(e) In addition to other required reports, an opposed candidate in a runoff election shall file one report for that election. The runoff election report must be received by the authority with whom the report is required to be filed not later than the eighth day before runoff election day. The report covers the period beginning the ninth day before the date of the main election and continuing through the 10th day before runoff election day.

#### Sec. 254.065. FINAL REPORT.

(a) If a candidate expects no reportable activity in connection with the candidacy to occur after the period covered by a report filed under this subchapter, the candidate may designate the report as a "final" report.

(b) The designation of a report as a final report:

(1) relieves the candidate of the duty to file additional reports under this subchapter, except as provided by Subsection (c); and

(2) terminates the candidate's campaign treasurer appointment.

(c) If, after a candidate's final report is filed, reportable activity with respect to the candidacy occurs, the candidate shall file the appropriate reports under this subchapter and is otherwise subject to the provisions of this title applicable to candidates. A report filed under this subsection may be designated as a final report.

Sec. 254.066. AUTHORITY WITH WHOM REPORTS FILED. Reports under this subchapter shall be filed with the authority with whom the candidate's campaign treasurer appointment is required to be filed.

### SUBCHAPTER D. REPORTING BY OFFICEHOLDER

Sec. 254.091. ADDITIONAL CONTENTS OF REPORTS. In addition to the contents required by Section 254.031, each report by an officeholder must include:

(1) the officeholder's full name and address and the office held;

(2) for each political committee from which the officeholder received notice under Section 254.128 or 254.161:

(A) the committee's full name and address;

(B) an indication of whether the committee is a general-purpose committee or a specific-purpose committee; and

(C) the full name and address of the committee's campaign treasurer;

(3) on a separate page or pages of the report, the identification of any payment from political contributions made to a business in which the officeholder has a participating interest of more than 10 percent, holds a position on the governing body of the business, or serves as an officer of the business.

Sec. 254.0911. ADDITIONAL CONTENTS OF REPORTS BY CERTAIN JUDICIAL OFFICEHOLDERS. In addition to the contents required by Sections 254.031 and 254.091, each report by a holder of a judicial office covered by Subchapter F, Chapter 253, must include the contents prescribed by Section 254.0611.

Sec. 254.0912. ADDITIONAL CONTENTS OF REPORTS BY STATEWIDE EXECUTIVE OFFICEHOLDERS AND LEGISLATIVE OFFICEHOLDERS. In addition to the contents required by Sections 254.031 and 254.091, each report by a holder of a statewide office in the executive branch or a legislative office must include the contents prescribed by Section 254.0612.

**Sec. 254.092. CERTAIN OFFICEHOLDER EXPENDITURES EXCLUDED**. An officeholder is not required to report officeholder expenditures made from the officeholder's personal funds, except as provided by Section 253.035(h).

# Sec. 254.093. SEMIANNUAL REPORTING SCHEDULE FOR OFFICEHOLDER.

(a) An officeholder shall file two reports for each year as provided by this section.

and

(b) The first report shall be filed not later than July 15. The report covers the period beginning January 1, the day the officeholder takes office, or the first day after the period covered by the last report required to be filed under this chapter, as applicable, and continuing through June 30.

(c) The second report shall be filed not later than January 15. The report covers the period beginning July 1, the day the officeholder takes office, or the first day after the period covered by the last report required to be filed under this chapter, as applicable, and continuing through December 31.

## Sec. 254.094. REPORT FOLLOWING APPOINTMENT OF CAMPAIGN TREASURER.

(a) An officeholder who appoints a campaign treasurer shall file a report as provided by this section.

(b) The report covers the period beginning the first day after the period covered by the last report required to be filed under this chapter or the day the officeholder takes office, as applicable, and continuing through the day before the date the officeholder's campaign treasurer is appointed.

(c) The report shall be filed not later than the 15th day after the date the officeholder's campaign treasurer is appointed.

Sec. 254.095. REPORT NOT REQUIRED. If at the end of any reporting period prescribed by this subchapter an officeholder who is required to file a report with an authority other than the commission has not accepted political contributions that in the aggregate exceed \$500 or made political expenditures that in the aggregate exceed \$500, the officeholder is not required to file a report covering that period.

Sec. 254.096. OFFICEHOLDER WHO BECOMES CANDIDATE. An officeholder who becomes a candidate is subject to Subchapter C during each period covered by a report required to be filed under Subchapter C.

Sec. 254.097. AUTHORITY WITH WHOM REPORTS FILED. Reports under this subchapter shall be filed with the authority with whom a campaign treasurer appointment by a candidate for the office held by the officeholder is required to be filed.

#### SUBCHAPTER E. REPORTING BY SPECIFIC-PURPOSE COMMITTEE

**Sec. 254.121. ADDITIONAL CONTENTS OF REPORTS**. In addition to the contents required by Section 254.031, each report by a campaign treasurer of a specific-purpose committee must include:

(1) the committee's full name and address;

(2) the full name, residence or business street address, and telephone number of the committee's campaign treasurer;

(3) the identity and date of the election for which the report is filed, if applicable;

(4) the name of each candidate and each measure supported or opposed by the committee, indicating for each whether the committee supports or opposes;

(5) the name of each officeholder assisted by the committee;

(6) the amount of each political expenditure in the form of a political contribution that is made to a candidate, officeholder, or another political committee and that is returned to the committee during the reporting period, the name of the person to whom the expenditure was originally made, and the date it is returned;

(7) on a separate page or pages of the report, the identification of any payment from political contributions made to a business in which the candidate or officeholder has a participating interest of more than 10 percent, holds a position on the governing body of the business, or serves as an officer of the business; and

(8) on a separate page or pages of the report, the identification of any contribution from a corporation or labor organization made and accepted under Subchapter D, Chapter 253.

Sec. 254.1211. ADDITIONAL CONTENTS OF REPORTS OF CERTAIN COMMITTEES. In addition to the contents required by Sections 254.031 and 254.121, each report by a specific-purpose committee for supporting or opposing a candidate for or assisting a holder of a judicial office covered by Subchapter F, Chapter 253, must include the contents prescribed by Section 254.0611.

Sec. 254.1212. ADDITIONAL CONTENTS OF REPORTS OF COMMITTEE SUPPORTING OR OPPOSING CANDIDATE FOR STATEWIDE EXECUTIVE **OFFICEHOLDERS** LEGISLATIVE OR **OFFICEHOLDERS** OR ASSISTING EXECUTIVE STATEWIDE **OFFICEHOLDERS** OR LEGISLATIVE OFFICEHOLDERS. In addition to the contents required by Sections 254.031 and 254.121, each report by a specific-purpose committee for supporting or opposing a candidate for or assisting a holder of a statewide office in the executive branch or a legislative office must include the contents prescribed by Section 254.0612.

Sec. 254.122. INVOLVEMENT IN MORE THAN ONE ELECTION BY CERTAIN COMMITTEES. If a specific-purpose committee for supporting or opposing more than one candidate becomes involved in more than one election for which the reporting periods prescribed by Section 254.124 overlap, the reportable activity that occurs during the overlapping period is not required to be included in a report filed after the first report in which the activity is required to be reported.

## Sec. 254.123. SEMIANNUAL REPORTING SCHEDULE FOR COMMITTEE.

(a) The campaign treasurer of a specific-purpose committee shall file two reports for each year as provided by this section.

(b) The first report shall be filed not later than July 15. The report covers the period beginning January 1, the day the committee's campaign treasurer appointment is filed, or the first day after the period covered by the last report required to be filed under this subchapter, as applicable, and continuing through June 30.

(c) The second report shall be filed not later than January 15. The report covers the period beginning July 1, the day the committee's campaign treasurer appointment is filed, or the first day after the period covered by the last report required to be filed under this subchapter, as applicable, and continuing through December 31.

## Sec. 254.124. ADDITIONAL REPORTS OF COMMITTEE FOR SUPPORTING OR OPPOSING CANDIDATE OR MEASURE.

(a) In addition to other required reports, for each election in which a specific-purpose committee supports or opposes a candidate or measure, the committee's campaign treasurer shall file two reports.

(b) The first report must be received by the authority with whom the report is required to be filed not later than the 30th day before election day. The report covers the period beginning the day the committee's campaign treasurer appointment is filed or the first day after the period covered by the committee's last required report, as applicable, and continuing through the 40th day before election day.

(c) The second report must be received by the authority with whom the report is required to be filed not later than the eighth day before election day. The report covers the period beginning the 39th day before election day and continuing through the 10th day before election day.

(d) If a specific-purpose committee supports or opposes a candidate or measure in an election after a reporting period prescribed by Subsection (b) or (c), the first report must be received by the authority with whom the report is required to be filed not later than the regular deadline for the report covering the period during which the committee becomes involved in the election. The period covered by the first report begins the day the committee's campaign treasurer appointment is filed or the first day after the period covered by the committee's last required report, as applicable.

(e) In addition to other required reports, the campaign treasurer of a specific-purpose committee that supports or opposes a candidate in a runoff election shall file one report for the runoff election. The runoff election report must be received by the authority with whom the report is required to be filed not later than the eighth day before runoff election day. The report covers the period beginning the ninth day before the date of the main election and continuing through the 10th day before runoff election day.

(f) This section does not apply to a specific-purpose committee supporting only candidates who do not have opponents whose names are to appear on the ballot.

# Sec. 254.125. FINAL REPORT OF COMMITTEE FOR SUPPORTING OR OPPOSING CANDIDATE OR MEASURE.

(a) If a specific-purpose committee for supporting or opposing a candidate or measure expects no reportable activity in connection with the election to occur after the period covered by a report filed under this subchapter, the committee's campaign treasurer may designate the report as a "final" report.

(b) The designation of a report as a final report:

(1) relieves the campaign treasurer of the duty to file additional reports under this subchapter, except as provided by Subsection (c); and

(2) terminates the committee's campaign treasurer appointment.

(c) If, after a committee's final report is filed, reportable activity with respect to the election occurs, the committee must file the appropriate reports under this subchapter and is otherwise subject to the provisions of this title applicable to political committees. A report filed under this subsection may be designated as a final report.

## Sec. 254.126. DISSOLUTION REPORT OF COMMITTEE FOR ASSISTING OFFICEHOLDER.

(a) If a specific-purpose committee for assisting an officeholder expects no reportable activity to occur after the period covered by a report filed under this subchapter, the committee's campaign treasurer may designate the report as a "dissolution" report.

(b) The filing of a report designated as a dissolution report:

(1) relieves the campaign treasurer of the duty to file additional reports under this subchapter; and

(2) terminates the committee's campaign treasurer appointment.

(c) A dissolution report must contain an affidavit, executed by the committee's campaign treasurer, that states that all the committee's reportable activity has been reported.

### Sec. 254.127. TERMINATION REPORT.

(a) If the campaign treasurer appointment of a specific-purpose committee is terminated, the terminated campaign treasurer shall file a termination report.

(b) A termination report is not required if the termination occurs on the last day of a reporting period under this subchapter and a report for that period is filed as provided by this subchapter.

(c) The report covers the period beginning the day after the period covered by the last report required to be filed under this subchapter and continuing through the day the campaign treasurer appointment is terminated.

(d) The report shall be filed not later than the 10th day after the date the campaign treasurer appointment is terminated.

(e) Reportable activity contained in a termination report is not required to be included in any subsequent report of the committee that is filed under this subchapter. The period covered by the committee's first report filed under this subchapter after a termination report begins the day after the date the campaign treasurer appointment is terminated.

# Sec. 254.128. NOTICE TO CANDIDATE AND OFFICEHOLDER OF CONTRIBUTIONS AND EXPENDITURES.

(a) If a specific-purpose committee accepts political contributions or makes political expenditures for a candidate or officeholder, the committee's campaign treasurer shall deliver written notice of that fact to the affected candidate or officeholder not later than the end of the period covered by the report in which the reportable activity occurs.

(b) The notice must include the full name and address of the political committee and its campaign treasurer and an indication that the committee is a specific-purpose committee.

(c) A campaign treasurer commits an offense if the campaign treasurer fails to comply with this section. An offense under this section is a Class A misdemeanor.

## Sec. 254.129. NOTICE OF CHANGE IN COMMITTEE STATUS.

(a) If a specific-purpose committee changes its operation and becomes a generalpurpose committee, the committee's campaign treasurer shall deliver written notice of the change in status to the authority with whom the specific-purpose committee's reports under this chapter are required to be filed.

(b) The notice shall be delivered not later than the next deadline for filing a report under this subchapter that:

(1) occurs after the change in status; and

(2) would be applicable to the political committee if the committee had not changed its status.

(c) The notice must indicate the filing authority with whom future filings are expected to be made.

(d) A campaign treasurer commits an offense if the campaign treasurer fails to comply with this section. An offense under this section is a Class B misdemeanor.

#### Sec. 254.130. AUTHORITY WITH WHOM REPORTS FILED.

(a) Except as provided by subsection (b), reports filed under this subchapter shall be filed with the authority with whom the political committee's campaign treasurer appointment is required to be filed.

(b) A specific-purpose committee created to support or oppose a measure on the issuance of bonds by a school district shall file reports under this subchapter with the commission.

#### SUBCHAPTER F. REPORTING BY GENERAL-PURPOSE COMMITTEE

**Sec. 254.151. ADDITIONAL CONTENTS OF REPORTS**. In addition to the contents required by Section 254.031, each report by a campaign treasurer of a general-purpose committee must include:

(1) the committee's full name and address;

(2) the full name, residence or business street address, and telephone number of the committee's campaign treasurer;

(3) the identity and date of the election for which the report is filed, if applicable;

(4) the name of each identified candidate or measure or classification by party of candidates supported or opposed by the committee, indicating whether the committee supports or opposes each listed candidate, measure, or classification by party of candidates;

(5) the name of each identified officeholder or classification by party of officeholders assisted by the committee;

(6) the principal occupation of each person from whom political contributions that in the aggregate exceed \$50 are accepted during the reporting period;

(7) the amount of each political expenditure in the form of a political contribution made to a candidate, officeholder, or another political committee that is returned to the committee during the reporting period, the name of the person to whom the expenditure was originally made, and the date it is returned;

(8) on a separate page or pages of the report, the identification of any contribution from a corporation or labor organization made and accepted under Subchapter D, Chapter 253; and

(9) on a separate page or pages of the report, the identification of the name of the donor, the amount, and the date of any expenditure made by a corporation or labor organization to:

(A) establish or administer the political committee; or

(B) finance the solicitation of political contributions to the committee under Section 253.100.

Sec. 254.152. TIME FOR REPORTING CERTAIN EXPENDITURES. If a general-purpose committee makes a political expenditure in the form of a political contribution to another general-purpose committee or to an out-of-state political committee and the contributing committee does not intend that the contribution be used in connection with a particular election, the contributing committee shall include the expenditure in the first report required to be filed under this subchapter after the expenditure is made.

# Sec. 254.153. SEMIANNUAL REPORTING SCHEDULE FOR COMMITTEE.

(a) The campaign treasurer of a general-purpose committee shall file two reports for each year as provided by this section.

(b) The first report shall be filed not later than July 15. The report covers the period beginning January 1, the day the committee's campaign treasurer appointment is filed, or the first day after the period covered by the last report required to be filed under this subchapter, as applicable, and continuing through June 30.

(c) The second report shall be filed not later than January 15. The report covers the period beginning July 1, the day the committee's campaign treasurer appointment is filed, or the first day after the period covered by the last report required to be filed under this subchapter, as applicable, and continuing through December 31.

# Sec. 254.154. ADDITIONAL REPORTS OF COMMITTEE INVOLVED IN ELECTION.

(a) In addition to other required reports, for each election in which a general-purpose committee is involved, the committee's campaign treasurer shall file two reports.

(b) The first report must be received by the authority with whom the report is required to be filed not later than the 30th day before election day. The report covers the period beginning the day the committee's campaign treasurer appointment is filed or the first day after the period covered by the committee's last required report, as applicable, and continuing through the 40th day before election day.

(c) The second report must be received by the authority with whom the report is required to be filed not later than the eighth day before election day. The report covers the period beginning the 39th day before election day and continuing through the 10th day before election day.

(d) If a general-purpose committee becomes involved in an election after a reporting period prescribed by Subsection (b) or (c), the first report must be received by the authority with whom the report is required to be filed not later than the regular deadline for the report covering the period during which the committee becomes involved in the election. The period covered by the first report begins the day the committee's campaign treasurer appointment is filed or the first day after the period covered by the committee's last required report, as applicable.

(e) In addition to other required reports, the campaign treasurer of a general-purpose committee involved in a runoff election shall file one report for the runoff election. The runoff election report must be received by the authority with whom the report is required to be filed not earlier than the 10th day or later than the eighth day before runoff election day. The report covers the period beginning the ninth day before the date of the main election and continuing through the 10th day before runoff election day.

# Sec. 254.1541. ALTERNATE REPORTING REQUIREMENTS FOR CERTAIN COMMITTEES.

(a) This section applies only to a general-purpose committee with less than \$20,000 in one or more accounts maintained by the committee in which political contributions are deposited, as of the last day of the preceding reporting period for which the committee was required to file a report.

(b) A report by a campaign treasurer of a general-purpose committee to which this section applies may include, instead of the information required under Sections 254.031(a)(1) and (5) and Section 254.151(6):

(1) the amount of political contributions from each person that in the aggregate exceed \$100 and that are accepted during the reporting period by the committee, the full name and address of the person making the contributions, the person's principal occupation, and the dates of the contributions; and

(2) the total amount or a specific listing of the political contributions of \$100 or less accepted and the total amount or a specific listing of the political expenditures of \$100 or less made during the reporting period.

# Sec. 254.155. OPTION TO FILE MONTHLY; NOTICE.

(a) As an alternative to filing reports under Sections 254.153 and 254.154, a generalpurpose committee may file monthly reports.

(b) To be entitled to file monthly reports, the committee must deliver written notice of the committee's intent to file monthly to the commission not earlier than January 1 or later than January 15 of the year in which the committee intends to file monthly. The notice for a committee formed after January 15 must be delivered at the time the committee's campaign treasurer appointment is filed.

(c) A committee that files monthly reports may revert to the regular filing schedule prescribed by Sections 254.153 and 254.154 by delivering written notice of the committee's intent not earlier than January 1 or later than January 15 of the year in which the committee intends to revert to the regular reporting schedule. The notice must include a report of all political contributions accepted and all political expenditures made that were not previously reported.

**Sec. 254.156. CONTENTS OF MONTHLY REPORTS**. Each monthly report filed under this subchapter must comply with Sections 254.031 and 254.151 except that the maximum amount of a political contribution, expenditure, or loan that is not required to be individually reported is:

(1) \$10 in the aggregate; or

(2) \$20 in the aggregate for a contribution accepted by a general-purpose committee to which Section 254.1541 applies.

# Sec. 254.157. MONTHLY REPORTING SCHEDULE.

(a) The campaign treasurer of a general-purpose committee filing monthly reports shall file a report not later than the fifth day of the month following the period covered by the report. A report covering the month preceding an election in which the committee is involved must be received by the authority with whom the report is required to be filed not later than the fifth day of the month following the period covered by the report.

(b) A monthly report covers the period beginning the 26th day of each month and continuing through the 25th day of the following month, except that the period covered by the first report begins January 1 and continues through January 25.

Sec. 254.158. EXCEPTION TO MONTHLY REPORTING SCHEDULE. If the campaign treasurer appointment of a general-purpose committee filing monthly reports is filed after January 1 of the year in which monthly reports are filed, the period covered by the first monthly report begins the day the appointment is filed and continues through the 25th day of the month in which the appointment is filed unless the appointment is filed the 25th or a succeeding day of the month. In that case, the period continues through the 25th day of the month following the month in which the appointment is filed.

**Sec. 254.1581. REPORTING BY OUT-OF-STATE POLITICAL COMMITTEE.** For each reporting period under this subchapter in which an out-of-state political committee accepts political contributions or makes political expenditures, the committee shall file with the commission a copy of one or more reports filed with the Federal Election Commission or with the proper filing authority of at least one other state that shows the political contributions accepted, political expenditures made, and other expenditures made by the committee. A report must be filed within the same period in which it is required to be filed under federal law or the law of the other state.

**Sec. 254.159. DISSOLUTION REPORT**. If a general-purpose committee expects no reportable activity to occur after the period covered by a report filed under this subchapter, the report may be designated as a "dissolution" report as provided by Section 254.126 for a specific-purpose committee and has the same effect.

**Sec. 254.160. TERMINATION REPORT**. If the campaign treasurer appointment of a general-purpose committee is terminated, the campaign treasurer shall file a termination report as prescribed by Section 254.127 for a specific-purpose committee.

Sec. 254.161. NOTICE TO CANDIDATE AND OFFICEHOLDER OF CONTRIBUTIONS AND EXPENDITURES. If a general-purpose committee other than the principal political committee of a political party or a political committee established by a political party's county executive committee accepts political contributions or makes political expenditures for a candidate or officeholder, notice of that fact shall be given to the affected candidate or officeholder as provided by Section 254.128 for a specific-purpose committee.

Sec. 254.162. NOTICE OF CHANGE IN COMMITTEE STATUS. If a generalpurpose committee changes its operation and becomes a specific-purpose committee, notice of the change in status shall be given to the commission as provided by Section 254.129 for a specific-purpose committee.

Sec. 254.163. AUTHORITY WITH WHOM REPORTS FILED. Reports filed under this subchapter shall be filed with the commission.

**Sec. 254.164. CERTAIN COMMITTEES EXEMPT FROM CIVIL PENALTIES.** The commission may not impose a civil penalty on a general-purpose committee for a violation of this chapter if the report filed by the committee that is the subject of the violation discloses that the committee did not accept political contributions totaling \$3,000 or more, accept political contributions from a single person totaling \$1,000 or more, or make or authorize political expenditures totaling \$3,000 or more during:

(1) the reporting period covered by the report that is the subject of the violation;

or

(2) either of the two reporting periods preceding the reporting period described by Subdivision (1).

# SUBCHAPTER G. MODIFIED REPORTING PROCEDURES; \$500 MAXIMUM IN CONTRIBUTIONS OR EXPENDITURES

# Sec. 254.181. MODIFIED REPORTING AUTHORIZED.

(a) An opposed candidate or specific-purpose committee required to file reports under Subchapter C or E may file a report under this subchapter instead if the candidate or committee does not intend to accept political contributions that in the aggregate exceed \$500 or to make political expenditures that in the aggregate exceed \$500 in connection with the election.

(b) The amount of a filing fee paid by a candidate is excluded from the \$500 maximum expenditure permitted under this section.

# Sec. 254.182. DECLARATION OF INTENT REQUIRED.

(a) To be entitled to file reports under this subchapter, an opposed candidate or specificpurpose committee must file with the campaign treasurer appointment a written declaration of intent not to exceed \$500 in political contributions or political expenditures in the election.

(b) The declaration of intent must contain a statement that the candidate or committee understands that if the \$500 maximum for contributions and expenditures is exceeded, the candidate or committee is required to file reports under Subchapter C or E, as applicable.

### Sec. 254.183. MAXIMUM EXCEEDED.

(a) An opposed candidate or specific-purpose committee that exceeds \$500 in political contributions or political expenditures in the election shall file reports as required by Subchapter C or E, as applicable.

(b) If a candidate or committee exceeds the \$500 maximum after the filing deadline prescribed by Subchapter C or E for the first report required to be filed under the appropriate subchapter, the candidate or committee shall file a report not later than 48 hours after the maximum is exceeded.

(c) A report filed under Subsection (b) covers the period beginning the day the campaign treasurer appointment is filed and continuing through the day the maximum is exceeded.

(d) The reporting period for the next report filed by the candidate or committee begins on the day after the last day of the period covered by the report filed under Subsection (b).

# Sec. 254.184. APPLICABILITY OF REGULAR REPORTING REQUIREMENTS.

(a) Subchapter C or E, as applicable, applies to an opposed candidate or specificpurpose committee filing under this subchapter to the extent that the appropriate subchapter does not conflict with this subchapter. (b) A candidate or committee filing under this subchapter is not required to file any reports of political contributions and political expenditures other than the semiannual reports required to be filed not later than July 15 and January 15.

# SUBCHAPTER H. UNEXPENDED CONTRIBUTIONS

# Sec. 254.201. ANNUAL REPORT OF UNEXPENDED CONTRIBUTIONS.

(a) This section applies to:

(1) a former officeholder who has unexpended political contributions after filing the last report required to be filed by Subchapter D; or

(2) a person who was an unsuccessful candidate who has unexpended political contributions after filing the last report required to be filed by Subchapter C.

(b) A person covered by this section shall file an annual report for each year in which the person retains unexpended contributions.

### Sec. 254.202. FILING OF REPORT; CONTENTS.

(a) A person shall file the report required by Section 254.201 not earlier than January 1 or later than January 15 of each year following the year in which the person files a final report under this chapter.

(b) The report shall be filed with the authority with whom the person's campaign treasurer appointment was required to be filed.

(c) The report must include:

(1) the person's full name and address;

(2) the full name and address of each person to whom a payment from unexpended political contributions was made during the previous year;

(3) the date, amount, and purpose of each payment made under Subdivision (2);

(4) the total amount of unexpended political contributions as of December 31 of the previous year; and

(5) the total amount of interest and other income earned on unexpended political contributions during the previous year.

# Sec. 254.203. RETENTION OF CONTRIBUTIONS.

(a) A person may not retain political contributions covered by this title, assets purchased with the contributions, or interest and other income earned on the contributions for more than six years after the date the person either ceases to be an officeholder or candidate or files a final report under this chapter, whichever is later.

(b) If the person becomes an officeholder or candidate within the six-year period, the prohibition in Subsection (a) does not apply until the person again ceases to be an officeholder or candidate.

(c) A person who violates Subsection (a) commits an offense. An offense under this section is a Class A misdemeanor.

# Sec. 254.204. DISPOSITION OF UNEXPENDED CONTRIBUTIONS.

(a) At the end of the six-year period prescribed by Section 254.203, the former officeholder or candidate shall remit any unexpended political contributions to one or more of the following:

(1) the political party with which the person was affiliated when the person's name last appeared on a ballot;

(2) a candidate or political committee;

(3) the comptroller for deposit in the state treasury;

(4) one or more persons from whom political contributions were received, in accordance with Subsection (d);

(5) a recognized charitable organization formed for educational, religious, or scientific purposes that is exempt from taxation under Section 501(c)(3), Internal Revenue Code of 1986, and its subsequent amendments; or

(6) a public or private postsecondary educational institution or an institution of higher education as defined by Section 61.003(8), Education Code, solely for the purpose of assisting or creating a scholarship program.

(b) A person who disposes of unexpended political contributions under Subsection (a)(2) shall report each contribution as if the person were a campaign treasurer of a specific-purpose committee.

(c) Political contributions disposed of under Subsection (a)(3) may be appropriated only for financing primary elections.

(d) The amount of political contributions disposed of under Subsection (a)(4) to one person may not exceed the aggregate amount accepted from that person during the last two years that the candidate or officeholder accepted contributions under this title.

Sec. 254.205. REPORT OF DISPOSITION OF UNEXPENDED CONTRIBUTIONS.

(a) Not later than the 30th day after the date the six-year period prescribed by Section 254.203 ends, the person required to dispose of unexpended political contributions shall file a report of the disposition.

(b) The report shall be filed with the authority with whom the person's campaign treasurer appointment was required to be filed.

(c) The report must include:

(1) the person's full name and address;

(2) the full name and address of each person to whom a payment from unexpended political contributions is made; and

(3) the date and amount of each payment reported under Subdivision (2).

### SUBCHAPTER I. CIVIL LIABILITY

#### Sec. 254.231. LIABILITY TO CANDIDATES.

(a) A candidate or campaign treasurer or assistant campaign treasurer of a political committee who fails to report in whole or in part a campaign contribution or campaign expenditure as required by this chapter is liable for damages as provided by this section.

(b) Each opposing candidate whose name appears on the ballot is entitled to recover damages under this section.

(c) In this section, "damages" means:

- (1) twice the amount not reported that is required to be reported; and
- (2) reasonable attorney's fees incurred in the suit.

(d) Reasonable attorney's fees incurred in the suit may be awarded to the defendant if judgment is rendered in the defendant's favor.

Sec. 254.232. LIABILITY TO STATE. A candidate, officeholder, or campaign treasurer or assistant campaign treasurer of a political committee who fails to report in whole or in part a political contribution or political expenditure as required by this chapter is liable in damages to the state in the amount of triple the amount not reported that is required to be reported.

# SUBCHAPTER J. REPORTING BY CERTAIN PERSONS MAKING DIRECT CAMPAIGN EXPENDITURES

# Sec. 254.261. DIRECT CAMPAIGN EXPENDITURE EXCEEDING \$100.

(a) A person not acting in concert with another person who makes one or more direct campaign expenditures in an election from the person's own property shall comply with this chapter as if the person were the campaign treasurer of a general-purpose committee that does not file monthly reports under Section 254.155.

(b) A person is not required to file a report under this section if the person is required to disclose the expenditure in another report required under this title within the time applicable under this section for reporting the expenditure.

(c) This section does not require a general-purpose committee that files under the monthly reporting schedule to file reports under Section 254.154.

(d) A person is not required to file a campaign treasurer appointment for making expenditures for which reporting is required under this section, unless the person is otherwise required to file a campaign treasurer appointment under this title.

**Sec. 254.262. TRAVEL EXPENSE**. A direct campaign expenditure consisting of personal travel expenses incurred by a person may be made without complying with Section 254.261.

# CHAPTER 255. REGULATING POLITICAL ADVERTISING AND CAMPAIGN COMMUNICATIONS

# Sec. 255.001. REQUIRED DISCLOSURE ON POLITICAL ADVERTISING.

(a) A person may not knowingly cause to be published, distributed, or broadcast political advertising containing express advocacy that does not indicate in the advertising:

(1) that it is political advertising; and

(2) the full name of:

- (A) the person who paid for the political advertising;
- (B) the political committee authorizing the political advertising; or

(C) the candidate or specific-purpose committee supporting the candidate, if the political advertising is authorized by the candidate.

(b) Political advertising that is authorized by a candidate, an agent of a candidate, or a political committee filing reports under this title shall be deemed to contain express advocacy.

(c) A person may not knowingly use, cause or permit to be used, or continue to use any published, distributed, or broadcast political advertising containing express advocacy that the person knows does not include the disclosure required by Subsection (a). A person is presumed to know that the use of political advertising is prohibited by this subsection if the commission notifies the person in writing that the use is prohibited. A person who learns that political advertising signs, as defined by Section 259.001, that have been distributed do not include the disclosure required by Subsection (a) or include a disclosure that does not comply with Subsection (a) does not commit a continuing violation of this subsection if the person makes a good faith attempt to remove or correct those signs. A person who learns that printed political advertising other than a political advertising sign that has been distributed does not include the disclosure required by Subsection (a) or includes a disclosure that does not comply with Subsection (a) is not required to attempt to recover the political advertising and does not commit a continuing violation of this subsection (a) is not required to attempt to recover the political advertising and does not commit a continuing violation of this subsection as to any previously distributed political advertising.

(d) This section does not apply to:

(1) tickets or invitations to political fund-raising events;

(2) campaign buttons, pins, hats, or similar campaign materials; or

(3) circulars or flyers that cost in the aggregate less than \$500 to publish and

(e) A person who violates this section is liable to the state for a civil penalty in an amount determined by the commission not to exceed \$4,000.

### Sec. 255.002. RATES FOR POLITICAL ADVERTISING.

(a) The rate charged for political advertising by a radio or television station may not exceed:

(1) during the 45 days preceding a general or runoff primary election and during the 60 days preceding a general or special election, the broadcaster's lowest unit charge for advertising of the same class, for the same time, and for the same period; or

(2) at any time other than that specified by Subdivision (1), the amount charged other users for comparable use of the station.

(b) The rate charged for political advertising that is printed or published may not exceed the lowest charge made for comparable use of the space for any other purposes.

distribute.

(c) In determining amounts charged for comparable use, the amount and kind of space or time used, number of times used, frequency of use, type of advertising copy submitted, and any other relevant factors shall be considered.

(d) Discounts offered by a newspaper or magazine to its commercial advertisers shall be offered on equal terms to purchasers of political advertising from the newspaper or magazine.

(e) A person commits an offense if the person knowingly demands or receives or knowingly pays or offers to pay for political advertising more consideration than permitted by this section.

(f) An offense under this section is a Class C misdemeanor.

# Sec. 255.003. UNLAWFUL USE OF PUBLIC FUNDS FOR POLITICAL ADVERTISING.

(a) An officer or employee of a political subdivision may not knowingly spend or authorize the spending of public funds for political advertising.

(b) Subsection (a) does not apply to a communication that factually describes the purposes of a measure if the communication does not advocate passage or defeat of the measure.

(b-1) An officer or employee of a political subdivision may not spend or authorize the spending of public funds for a communication describing a measure if the communication contains information that:

(1) the officer or employee knows is false; and

(2) is sufficiently substantial and important as to be reasonably likely to influence a voter to vote for or against the measure.

(c) A person who violates Subsection (a) or (b-1) commits an offense. An offense under this section is a Class A misdemeanor.

(d) It is an affirmative defense to prosecution for an offense under this section or the imposition of a civil penalty for conduct under this section that an officer or employee of a political subdivision reasonably relied on a court order or an interpretation of this section in a written opinion issued by:

(1) a court of record;

- (2) the attorney general; or
- (3) the commission.

(e) On written request of the governing body of a political subdivision that has ordered an election on a measure, the commission shall prepare an advance written advisory opinion as to whether a particular communication relating to the measure does or does not comply with this section.

(f) Subsections (d) and (e) do not apply to a port authority or navigation district.

# Sec. 255.0031. UNLAWFUL USE OF INTERNAL MAIL SYSTEM FOR POLITICAL ADVERTISING.

(a) An officer or employee of a state agency or political subdivision may not knowingly use or authorize the use of an internal mail system for the distribution of political advertising.

(b) Subsection (a) does not apply to:

(1) the use of an internal mail system to distribute political advertising that is delivered to the premises of a state agency or political subdivision through the United States Postal Service; or

(2) the use of an internal mail system by a state agency or municipality to distribute political advertising that is the subject of or related to an investigation, hearing, or other official proceeding of the agency or municipality.

(c) A person who violates this section commits an offense. An offense under this section is a Class A misdemeanor.

(d) In this section:

(1) "Internal mail system" means a system operated by a state agency or political subdivision to deliver written documents to officers or employees of the agency or subdivision.

(2) "State agency" means:

(A) a department, commission, board, office, or other agency that is in the legislative, executive, or judicial branch of state government;

(B) a university system or an institution of higher education as defined by Section 61.003, Education Code; or

(C) a river authority created under the constitution or a statute of this state.

### Sec. 255.004. TRUE SOURCE OF COMMUNICATION.

(a) A person commits an offense if, with intent to injure a candidate or influence the result of an election, the person enters into a contract or other agreement to print, publish, or broadcast political advertising that purports to emanate from a source other than its true source.

(b) A person commits an offense if, with intent to injure a candidate or influence the result of an election, the person knowingly represents in a campaign communication that the communication emanates from a source other than its true source.

(c) An offense under this section is a Class A misdemeanor.

(d) A person commits an offense if the person, with intent to injure a candidate or influence the result of an election:

(1) creates a deep fake video; and

(2) causes the deep fake video to be published or distributed within 30 days of an election.

(e) In this section, "deep fake video" means a video, created with the intent to deceive, that appears to depict a real person performing an action that did not occur in reality.

### Sec. 255.005. MISREPRESENTATION OF IDENTITY.

(a) A person commits an offense if, with intent to injure a candidate or influence the result of an election, the person misrepresents the person's identity or, if acting or purporting to act as an agent, misrepresents the identity of the agent's principal, in political advertising or a campaign communication.

(b) An offense under this section is a Class A misdemeanor.

### Sec. 255.006. MISLEADING USE OF OFFICE TITLE.

(a) A person commits an offense if the person knowingly enters into a contract or other agreement to print, publish, or broadcast political advertising with the intent to represent to an ordinary and prudent person that a candidate holds a public office that the candidate does not hold at the time the agreement is made.

(b) A person commits an offense if the person knowingly represents in a campaign communication that a candidate holds a public office that the candidate does not hold at the time the representation is made.

(c) For purposes of this section, a person represents that a candidate holds a public office that the candidate does not hold if:

(1) the candidate does not hold the office that the candidate seeks; and

(2) the political advertising or campaign communication states the public office sought but does not include the word "for" in a type size that is at least one-half the type size used for the name of the office to clarify that the candidate does not hold that office.

(d) A person other than an officeholder commits an offense if the person knowingly uses a representation of the state seal in political advertising.

(e) An offense under this section is a Class A misdemeanor.

# Sec. 255.008. DISCLOSURE ON POLITICAL ADVERTISING FOR JUDICIAL OFFICE.

(a) This section applies only to a candidate or political committee covered by Subchapter F, Chapter 253.

(b) Political advertising by a candidate who files a declaration of intent to comply with the limits on expenditures under Subchapter F, Chapter 253, or a specific-purpose committee for supporting such a candidate may include the following statement: "Political advertising paid for by (name of candidate or committee) in compliance with the voluntary limits of the Judicial Campaign Fairness Act."

(c) Political advertising by a candidate who files a declaration of intent to comply with the limits on expenditures under Subchapter F, Chapter 253, or a specific-purpose committee for supporting such a candidate that does not contain the statement prescribed by Subsection (b) must comply with Section 255.001.

(d) Political advertising by a candidate who files a declaration of intent to exceed the limits on expenditures under Subchapter F, Chapter 253, or a specific-purpose committee for supporting such a candidate must include the following statement: "Political advertising paid for by (name of candidate or committee), (who or which) has rejected the voluntary limits of the Judicial Campaign Fairness Act."

(e) The commission shall adopt rules providing for:

(1) the minimum size of the disclosure required by this section in political advertising that appears on television or in writing; and

(2) the minimum duration of the disclosure required by this section in political advertising that appears on television or radio.

(f) A person who violates this section or a rule adopted under this section is liable for a civil penalty not to exceed:

(1) \$15,000, for a candidate for a statewide judicial office or a specific-purpose committee for supporting such a candidate;

(2) \$10,000, for a candidate for chief justice or justice, court of appeals, or a specific-purpose committee for supporting such a candidate; or

(3) \$5,000, for a candidate for any other judicial office covered by Subchapter F, Chapter 253, or a specific-purpose committee for supporting such a candidate.

(g) Section 253.176 applies to the imposition and disposition of a civil penalty under this section.

# CHAPTER 257. POLITICAL PARTIES

# Sec. 257.001. PRINCIPAL POLITICAL COMMITTEE OF POLITICAL PARTY.

The state or county executive committee of a political party may designate a general-purpose committee as the principal political committee for that party in the state or county, as applicable.

# Sec. 257.002. REQUIREMENTS RELATING TO CORPORATE OR LABOR UNION CONTRIBUTIONS.

(a) A political party that accepts a contribution authorized by Section 253.104 may use the contribution only to:

(1) defray normal overhead and administrative or operating costs incurred by the party; or

(2) administer a primary election or convention held by the party.

(b) A political party that accepts contributions authorized by Section 253.104 shall maintain the contributions in a separate account.

### Sec. 257.003. REPORT REQUIRED.

(a) A political party that accepts contributions authorized by Section 253.104 shall report all contributions and expenditures made to and from the account required by Section 257.002.

(b) The report must be filed with the commission and must include the information required under Section 254.031 as if the contributions or expenditures were political contributions or political expenditures.

(c) Sections 254.001 and 254.032-254.037 apply to a report required by this section as if the party chair were a campaign treasurer of a political committee and as if the contributions or expenditures were political contributions or political expenditures.

(d) The commission shall prescribe by rule reporting schedules for each primary election held by the political party and for the general election for state and county officers.

# Sec. 257.004. RESTRICTIONS ON CONTRIBUTIONS BEFORE GENERAL ELECTION.

(a) Beginning on the 60th day before the date of the general election for state and county officers and continuing through the day of the election, a political party may not knowingly accept a contribution authorized by Section 253.104 or make an expenditure from the account required by Section 257.002.

(b) A person who violates this section commits an offense. An offense under this section is a felony of the third degree.

# Sec. 257.005. CANDIDATE FOR STATE OR COUNTY CHAIR OF POLITICAL PARTY.

(a) Except as provided by this section, the following are subject to the requirements of this title that apply to a candidate for public office:

(1) a candidate for state chair of a political party with a nominee on the ballot in the most recent gubernatorial general election; and

(2) a candidate for election to the office of county chair of a political party with a nominee on the ballot in the most recent gubernatorial general election if the county has a population of 350,000 or more.

(b) A political committee that supports or opposes a candidate covered by Subsection (a) is subject to the provisions of this title that apply to any other committee that supports or opposes candidates for public office, except as provided by this section.

(c) The reporting schedules for a candidate covered by Subsection (a) or a political committee supporting or opposing the candidate shall be prescribed by commission rule.

(d) Except as provided by this section, each contribution to and expenditure by a candidate covered by Subsection (a) is subject to the same requirements of this title as a political contribution to or a political expenditure by a candidate for public office. Each contribution to and expenditure by a political committee supporting or opposing a candidate covered by Subsection (a) is subject to the same requirements of this title as a political contribution to or political expenditure by any other specific-purpose committee.

(e) Section 251.001(1) does not apply to this section.

### Sec. 257.006. CRIMINAL PENALTY FOR FAILURE TO COMPLY.

(a) Except as provided by Section 257.004, a person who knowingly uses a contribution in violation of Section 257.002 or who knowingly fails to otherwise comply with this chapter commits an offense.

(b) An offense under this section is a Class A misdemeanor.

Sec. 257.007. RULES. The commission shall adopt rules to implement this chapter.

# CHAPTER 258. FAIR CAMPAIGN PRACTICES

Sec. 258.001. SHORT TITLE. This chapter may be cited as the Fair Campaign Practices Act.

### Sec. 258.002. PURPOSE.

(a) The purpose of this chapter is to encourage every candidate and political committee to subscribe to the Code of Fair Campaign Practices.

(b) It is the intent of the legislature that every candidate and political committee that subscribes to the Code of Fair Campaign Practices will follow the basic principles of decency, honesty, and fair play to encourage healthy competition and open discussion of issues and candidate qualifications and to discourage practices that cloud the issues or unfairly attack opponents.

#### Sec. 258.003. DELIVERY OF COPY OF CODE.

(a) When a candidate or political committee files its campaign treasurer appointment, the authority with whom the appointment is filed shall give the candidate or political committee a blank form of the Code of Fair Campaign Practices and a copy of this chapter.

(b) The authority shall inform each candidate or political committee that the candidate or committee may subscribe to and file the code with the authority and that subscription to the code is voluntary.

Sec. 258.004. TEXT OF CODE. The Code of Fair Campaign Practices reads as follows:

# CODE OF FAIR CAMPAIGN PRACTICES

There are basic principles of decency, honesty, and fair play that every candidate and political committee in this state has a moral obligation to observe and uphold, in order that, after vigorously contested but fairly conducted campaigns, our citizens may exercise their constitutional rights to a free and untrammeled choice and the will of the people may be fully and clearly expressed on the issues.

THEREFORE:

(1) I will conduct the campaign openly and publicly and limit attacks on my opponent to legitimate challenges to my opponent's record and stated positions on issues.

(2) I will not use or permit the use of character defamation, whispering campaigns, libel, slander, or scurrilous attacks on any candidate or the candidate's personal or family life.

(3) I will not use or permit any appeal to negative prejudice based on race, sex, religion, or national origin.

(4) I will not use campaign material of any sort that misrepresents, distorts, or otherwise falsifies the facts, nor will I use malicious or unfounded accusations that aim at creating or exploiting doubts, without justification, as to the personal integrity or patriotism of my opponent.

(5) I will not undertake or condone any dishonest or unethical practice that tends to corrupt or undermine our system of free elections or that hampers or prevents the full and free expression of the will of the voters, including any activity aimed at intimidating voters or discouraging them from voting.

(6) I will defend and uphold the right of every qualified voter to full and equal participation in the electoral process, and will not engage in any activity aimed at intimidating voters or discouraging them from voting.

(7) I will immediately and publicly repudiate methods and tactics that may come from others that I have pledged not to use or condone. I shall take firm action against any subordinate who violates any provision of this code or the laws governing elections.

I, the undersigned, candidate for election to public office in the State of Texas or campaign treasurer of a political committee, hereby voluntarily endorse, subscribe to, and solemnly pledge myself to conduct the campaign in accordance with the above principles and practices.

# $VOID - COPY ONLY - VOID^1$

Date

Signature

**Sec. 258.005. FORMS**. The commission shall print copies of the Code of Fair Campaign Practices and shall supply the forms to the authorities with whom copies of the code may be filed in quantities and at times requested by the authorities.

### Sec. 258.006. ACCEPTANCE AND PRESERVATION OF COPIES.

(a) An authority with whom a campaign treasurer appointment is filed shall accept each completed copy of the code submitted to the authority that is properly subscribed to by a candidate or the campaign treasurer of a political committee.

(b) Each copy of the code accepted under this section shall be preserved by the authority with whom it is filed for the period prescribed for the filer's campaign treasurer appointment.

**Sec. 258.007. SUBSCRIPTION TO CODE VOLUNTARY**. The subscription to the Code of Fair Campaign Practices by a candidate or a political committee is voluntary.

**Sec. 258.008. INDICATION ON POLITICAL ADVERTISING**. A candidate or a political committee that has filed a copy of the Code of Fair Campaign Practices may so indicate on political advertising in a form to be determined by the commission.

Sec. 258.009. CIVIL CAUSE OF ACTION. This chapter does not create a civil cause of action for recovery of damages or for enforcement of this chapter.

<sup>&</sup>lt;sup>1</sup> This document is a copy of chapter 258, Election Code. To subscribe to the Code of Fair Campaign Practices, a candidate or campaign treasurer of a political committee must submit Texas Ethics Commission FORM CFCP, not a signed copy of this document.

# **CHAPTER 259. POLITICAL SIGNS**

[Section 259.001, Election Code, was moved from Section 255.007, Election Code, with amendments indicated.]

Sec. 259.001. NOTICE REQUIREMENT ON POLITICAL ADVERTISING SIGNS. (a) The following notice must be written on each political advertising sign:

"NOTICE: IT IS A VIOLATION OF STATE LAW (CHAPTERS 392 AND 393, TRANSPORTATION CODE), TO PLACE THIS SIGN IN THE RIGHT-OF-WAY OF A HIGHWAY."

(b) A person commits an offense if the person:

(1) knowingly enters into a contract to print or make a political advertising sign that does not contain the notice required by Subsection (a); or

(2) instructs another person to place a political advertising sign that does not contain the notice required by Subsection (a).

(c) An offense under this section is a Class C misdemeanor.

(d) It is an exception to the application of Subsection (b) that the political advertising sign was printed or made before September 1, 1997, and complied with Subsection (a) as it existed immediately before that date.

(e) In this section, "political advertising sign" means a written form of political advertising designed to be seen from a road but does not include a bumper sticker.

[Section 259.002, Election Code, was moved from Section 202.009, Property Code, with amendments indicated.]

# Sec. 259.002. REGULATION OF DISPLAY OF POLITICAL SIGNS BY PROPERTY OWNERS' ASSOCIATIONS.

(a) In this section, "property owners' association" has the meaning assigned by Section 202.001, Property Code.

(b) Except as otherwise provided by this section, a property owners' association may not enforce or adopt a restrictive covenant that prohibits a property owner from displaying on the owner's property one or more signs advertising a candidate or measure for an election:

(1) on or after the 90th day before the date of the election to which the sign relates; or

(2) before the 10th day after that election date.

(c) This section does not prohibit the enforcement or adoption of a covenant that:

(1) requires a sign to be ground-mounted; or

(2) limits a property owner to displaying only one sign for each candidate or

(d) This section does not prohibit the enforcement or adoption of a covenant that prohibits a sign that:

(1) contains roofing material, siding, paving materials, flora, one or more balloons or lights, or any other similar building, landscaping, or nonstandard decorative component;

(2) is attached in any way to plant material, a traffic control device, a light, a trailer, a vehicle, or any other existing structure or object;

(3) includes the painting of architectural surfaces;

measure.

- (4) threatens the public health or safety;
- (5) is larger than four feet by six feet;
- (6) violates a law;

(7) contains language, graphics, or any display that would be offensive to the ordinary person; or

(8) is accompanied by music or other sounds or by streamers or is otherwise distracting to motorists.

(e) A property owners' association may remove a sign displayed in violation of a restrictive covenant permitted by this section.

[Section 259.003, Election Code, was moved from Section 216.903, Local Government Code, with amendments indicated.]

#### Sec. 259.003. REGULATION OF POLITICAL SIGNS BY MUNICIPALITY.

(a) In this section, "private real property" does not include real property subject to an easement or other encumbrance that allows a municipality to use the property for a public purpose.

(b) A municipal charter provision or ordinance that regulates signs may not, for a sign that contains primarily a political message and that is located on private real property with the consent of the property owner:

- (1) prohibit the sign from being placed;
- (2) require a permit or approval of the municipality or impose a fee for the sign

to be placed;

(3) restrict the size of the sign; or

(4) provide for a charge for the removal of a political sign that is greater than the charge for removal of other signs regulated by ordinance.

(c) Subsection (b) does not apply to a sign, including a billboard, that contains primarily a political message on a temporary basis and that is generally available for rent or purchase to carry commercial advertising or other messages that are not primarily political.

(d) Subsection (b) does not apply to a sign that:

- (1) has an effective area greater than 36 feet;
- (2) is more than eight feet high;
- (3) is illuminated; or
- (4) has any moving elements.

#### OF NALOO OF

# 10

# CHAPTER 302 – SPEAKER OF THE HOUSE OF REPRESENTATIVES

# TEXAS ETHICS COMMISSION CHAPTER 302, GOVERNMENT CODE SPEAKER OF THE HOUSE OF REPRESENTATIVES



Effective September 1, 2019 (Revised 8/1/2023)

Texas Ethics Commission, P.O. Box 12070, Austin, Texas 78711 www.ethics.state.tx.us (512) 463-5800 • TDD (800) 735-2989 Promoting Public Confidence in Government.

# **CHAPTER 302, GOVERNMENT CODE**

# SPEAKER OF THE HOUSE OF REPRESENTATIVES

# TABLE OF CONTENTS

| FABLE OF CONTENTS  | . 2                        |
|--|----------------------------|
| CHAPTER 302. SPEAKER OF THE HOUSE OF REPRESENTATIVES   | . 3                        |
| SUBCHAPTER A. ELECTION OF SPEAKER  | . 3                        |
| Sec. 302.001. ELECTION   |                            |
| SUBCHAPTER B. CANDIDATE FOR SPEAKER: CAMPAIGN FINANCE  | . 3                        |
| Sec. 302.011. DEFINITIONS  | . 3                        |
| Sec. 302.012. RECORDS  |                            |
| Sec. 302.0121. DECLARATION OF SPEAKER CANDIDACY  | . 3                        |
| Sec. 302.013. FILING OF STATEMENT OF CONTRIBUTIONS, LOANS, AND   |                            |
| EXPENDITURES   |                            |
| Sec. 302.014. CONTENTS OF STATEMENT  |                            |
| Sec. 302.015. REQUISITES OF FILING   | . 5                        |
| Sec. 302.016. PUBLIC RECORDS   | . 5                        |
| Sec. 302.018. CONTRIBUTIONS FROM EXECUTIVE OR JUDICIAL OFFICERS OR   |                            |
|  |                            |
| EMPLOYEES  | . 5                        |
| EMPLOYEES<br>Sec. 302.0191. CONTRIBUTIONS AND EXPENDITURES FROM POLITICAL  | . 5                        |
| Sec. 302.0191. CONTRIBUTIONS AND EXPENDITURES FROM POLITICAL CONTRIBUTIONS   | . 5                        |
| Sec. 302.0191. CONTRIBUTIONS AND EXPENDITURES FROM POLITICAL   | . 5                        |
| Sec. 302.0191. CONTRIBUTIONS AND EXPENDITURES FROM POLITICAL CONTRIBUTIONS   | . 5<br>. 5                 |
| Sec. 302.0191. CONTRIBUTIONS AND EXPENDITURES FROM POLITICAL<br>CONTRIBUTIONS  | .5<br>.5<br>.6             |
| Sec. 302.0191. CONTRIBUTIONS AND EXPENDITURES FROM POLITICAL<br>CONTRIBUTIONS<br>Sec. 302.020. PERMITTED EXPENDITURES<br>Sec. 302.0201. DISPOSITION OF UNEXPENDED FUNDS; REPORT<br>Sec. 302.021. OFFENSES AND PENALTIES<br>Sec. 302.022. PROSECUTION BY INDICTMENT   | .5<br>.5<br>.6<br>.6       |
| Sec. 302.0191. CONTRIBUTIONS AND EXPENDITURES FROM POLITICAL<br>CONTRIBUTIONS  | .5<br>.5<br>.6<br>.7<br>.7 |
| Sec. 302.0191. CONTRIBUTIONS AND EXPENDITURES FROM POLITICAL<br>CONTRIBUTIONS<br>Sec. 302.020. PERMITTED EXPENDITURES<br>Sec. 302.0201. DISPOSITION OF UNEXPENDED FUNDS; REPORT<br>Sec. 302.021. OFFENSES AND PENALTIES<br>Sec. 302.022. PROSECUTION BY INDICTMENT   | .5<br>.5<br>.6<br>.7<br>.7 |
| Sec. 302.0191. CONTRIBUTIONS AND EXPENDITURES FROM POLITICAL<br>CONTRIBUTIONS  | .5<br>.5<br>.6<br>.7<br>.7 |
| Sec. 302.0191. CONTRIBUTIONS AND EXPENDITURES FROM POLITICAL<br>CONTRIBUTIONS<br>Sec. 302.020. PERMITTED EXPENDITURES<br>Sec. 302.0201. DISPOSITION OF UNEXPENDED FUNDS; REPORT<br>Sec. 302.021. OFFENSES AND PENALTIES<br>Sec. 302.022. PROSECUTION BY INDICTMENT<br>SUBCHAPTER C. LEGISLATIVE BRIBERY<br>Sec. 302.031. DEFINITION<br>Sec. 302.032. LEGISLATIVE BRIBERY: PROMISES OR THREATS<br>Sec. 302.033. LEGISLATIVE BRIBERY: ACCEPTING BENEFITS | .5<br>.6<br>.7<br>.7<br>.7 |
| Sec. 302.0191. CONTRIBUTIONS AND EXPENDITURES FROM POLITICAL<br>CONTRIBUTIONS  | .5<br>.6<br>.7<br>.7<br>.7 |

# **GOVERNMENT CODE**

# TITLE 3. LEGISLATIVE BRANCH

# SUBTITLE A. LEGISLATURE

# CHAPTER 302. SPEAKER OF THE HOUSE OF REPRESENTATIVES

# SUBCHAPTER A. ELECTION OF SPEAKER

**Sec. 302.001. ELECTION**. When the house of representatives first convenes in regular session and a quorum is present and has been qualified, the house shall elect a speaker unless a majority of the members present decides to defer the election.

# SUBCHAPTER B. CANDIDATE FOR SPEAKER: CAMPAIGN FINANCE

# Sec. 302.011. DEFINITIONS. In this subchapter:

(1) "Speaker candidate" means a member of or candidate for the house of representatives who has announced his candidacy for or who by his actions, words, or deeds seeks election to the office of speaker of the house of representatives.

(2) "Campaign expenditure" means the expenditure of money or the use of services or any other thing of value to aid or defeat the election of a speaker candidate.

(3) "Campaign funds" means the speaker candidate's personal funds that are devoted to the campaign for speaker and any money, services, or other things of value that are contributed or loaned to the speaker candidate for use in the candidate's campaign for speaker.

# Sec. 302.012. RECORDS.

(a) Each speaker candidate shall keep records of all information required to be filed under this subchapter.

(b) The records must be kept separate from the records required under the Texas Election Code for the speaker candidate's campaign for any other public office.

# Sec. 302.0121. DECLARATION OF SPEAKER CANDIDACY.

(a) Each speaker candidate shall file a declaration of candidacy with the Texas Ethics Commission as provided by this section.

(b) A declaration of speaker candidacy must:

- (1) be in writing;
- (2) identify the legislative session as to which the candidacy relates; and
- (3) include:
  - (A) the speaker candidate's name;
  - (B) the speaker candidate's residence or business street address; and
  - (C) the speaker candidate's telephone number.

(c) Except as provided by Subsection (e), a speaker candidate may not knowingly accept a contribution, loan, or promise of a contribution or loan in connection with the speaker candidacy or make or authorize a campaign expenditure at a time when a declaration of candidacy for the candidate is not in effect.

(d) A declaration of speaker candidacy terminates on the earlier of:

(1) the date the speaker candidate files a written statement with the Texas Ethics Commission stating that the candidate has terminated the candidacy; or

(2) the date a speaker is elected for the legislative session as to which the speaker candidate filed the statement.

(e) A former speaker candidate whose declaration of speaker candidacy is terminated under Subsection (d) may make a campaign expenditure in connection with a debt incurred during the period the former speaker candidate's declaration of candidacy was in effect.

# Sec. 302.013. FILING OF STATEMENT OF CONTRIBUTIONS, LOANS, AND EXPENDITURES.

(a) Each speaker candidate shall file a sworn statement with the Texas Ethics Commission listing the information required by Section 302.014.

(b) Each speaker candidate shall file the statement on:

(1) the first filing date after the date on which the speaker candidate files the declaration of candidacy required by Section 302.0121;

(2) each filing date during the candidacy; and

(3) each filing date until all campaign loans have been repaid.

(c) The filing dates are the first day of January, March, May, July, September, and November and the day before each regular or called session of the legislature convenes.

(d) Each speaker candidate shall file the statement by computer diskette, modem, or other means of electronic transfer, using computer software provided by the Texas Ethics Commission or computer software that meets commission specifications for a standard file format.

(e) Expired.

**Sec. 302.014. CONTENTS OF STATEMENT**. Each statement must list the following information for the period since the last filing date:

(1) each contribution of money the speaker candidate or the speaker candidate's agent, servant, staff member, or employee received for the campaign, the complete name and address of the contributor, and the date and amount of the contribution;

(2) each contribution of services and other things of value other than money that the speaker candidate or the speaker candidate's agent, servant, staff member, or employee received for the campaign, the nature of the contribution, the complete name and address of the contributor, and the date and value of the contribution;

(3) each loan made to the speaker candidate or to the speaker candidate's agent, servant, staff member, or employee for the campaign, including all loans listed in previous filings that are as yet unpaid or that were paid during the period covered by the present filing, the complete name and address of the lender and each person other than the speaker candidate who is responsible on the note, the date and amount of the note, the intended source of funds to repay the note, and any payments already made on the note and the source of the payments; and

(4) each expenditure of campaign funds that the speaker candidate or the speaker candidate's agent, servant, staff member, or employee made for the campaign, the complete name and address of each person to whom a payment of more than \$10 was made, and the purpose of each expenditure.

### Sec. 302.015. REQUISITES OF FILING.

(a) Except as provided by Subsection (b), a statement is considered to be filed in compliance with this subchapter if the postmark shows that it was sent to the Texas Ethics Commission at its official post office address by registered or certified mail from any point in this state before the filing deadline.

(b) A statement required to be filed on the day before a regular or called session convenes must actually be delivered and in the possession of the Texas Ethics Commission not later than midnight of that day.

### Sec. 302.016. PUBLIC RECORDS.

(a) Each statement filed under this subchapter is public information and shall be preserved for two years after the election for which it was filed.

(b) Unless a court of competent jurisdiction orders further preservation, a statement may be destroyed after the two-year period prescribed by Subsection (a).

### Sec. 302.018. CONTRIBUTIONS FROM EXECUTIVE OR JUDICIAL

**OFFICERS OR EMPLOYEES**. An elected officer or employee of the executive or judicial branch of state government may not contribute personal services, money, or goods of value to a speaker candidate's campaign.

### Sec. 302.0191. CONTRIBUTIONS AND EXPENDITURES FROM POLITICAL CONTRIBUTIONS. A person, including a speaker candidate, may not make a contribution to a speaker candidate's campaign or an expenditure to aid or defeat a speaker candidate from:

(1) political contributions accepted under Title 15, Election Code;

(2) interest earned on political contributions accepted under Title 15, Election

Code; or

(3) an asset purchased with political contributions accepted under Title 15,

Election Code.

**Sec. 302.020. PERMITTED EXPENDITURES**. A speaker candidate may expend campaign funds for:

(1) travel for the speaker candidate and the speaker candidate's immediate family and campaign staff;

- (2) the employment of clerks and stenographers;
- (3) clerical and stenographic supplies;
- (4) printing and stationery;
- (5) office rent;
- (6) telephone, telegraph, postage, freight, and express expenses;
- (7) advertising and publicity;
- (8) the expenses of holding political and other meetings designed to promote the

candidacy;

- (9) the employment of legal counsel; and
- (10) the retirement of campaign loans.

# Sec. 302.0201. DISPOSITION OF UNEXPENDED FUNDS; REPORT.

(a) A former speaker candidate may:

(1) use unexpended campaign funds to retire debt incurred in connection with the speaker candidacy; or

(2) remit unexpended campaign funds to one or more of the following:

(A) one or more persons from whom campaign funds were received, in accordance with Subsection (c); or

(B) a recognized charitable organization formed for educational, religious, or scientific purposes that is exempt from taxation under Section 501(c)(3), Internal Revenue Code of 1986, and its subsequent amendments.

(b) A former speaker candidate may not retain contributions covered by this subchapter, assets purchased with the contributions, or interest and other income earned on the contributions for more than six years after the date the person ceases to be a speaker candidate or hold the office of speaker.

(c) The amount of campaign funds disposed of under Subsection (a)(2)(A) to one person may not exceed the aggregate amount accepted from that person in connection with the former speaker candidate's most recent campaign for election to the office of speaker.

(d) Not later than January 15 of each year, a former speaker candidate who retains unexpended campaign funds shall file a sworn report with the Texas Ethics Commission that includes:

(1) the full name and address of each person to whom a payment from unexpended campaign funds is made;

(2) the date and amount of each payment reported under Subdivision (1); and

(3) the information required by Section 302.014 as to any contribution, loan, or expenditure not previously reported on a statement filed under Section 302.013.

(e) A report filed under this section covers, as applicable:

(1) the period:

(A) beginning on the date after the last day of the period covered by the most recent statement filed by the former speaker candidate under Section 302.013; and

(B) ending on December 31 of the preceding year; or

(2) the preceding calendar year.

(f) A former speaker candidate shall file the report on an official form designed by the Texas Ethics Commission. Sections 302.015 and 302.016 apply to a report filed under this section.

(g) For purposes of this section, a speaker candidate elected as speaker of the house of representatives is considered to be a former speaker candidate.

# Sec. 302.021. OFFENSES AND PENALTIES.

(a) A speaker candidate or former speaker candidate commits an offense if the person:

(1) knowingly fails to file the declaration of candidacy required by Section

302.0121;

(2) knowingly fails to file the statement required by Section 302.013;

(3) knowingly accepts a contribution, loan, or promise of a contribution or loan in violation of Section 302.0121(c);

(4) knowingly accepts a contribution from a person who uses political contributions, interest earned on political contributions, or an asset purchased with political contributions to make the contribution in violation of Section 302.0191;

(5) expends campaign funds for any purpose other than those enumerated in Section 302.020;

(6) knowingly retains contributions, assets purchased with contributions, or interest or other income earned on contributions in violation of Section 302.0201(b); or

(7) knowingly fails to file the report of unexpended campaign funds as required by Section 302.0201(d).

(b) An agent, officer, or director of a corporation, partnership, association, firm, union, foundation, committee, club, or other organization or group of persons commits an offense if the agent, officer, or director consents to a contribution, loan, or promise of a contribution or loan prohibited by this subchapter.

(c) A person commits an offense if the person conspires with another person to circumvent any provision of this subchapter.

(d) An individual other than the speaker candidate commits an offense if the individual, either acting alone or with another individual, expends or authorizes the expenditure of more than \$100 for correspondence to aid or defeat the election of a speaker candidate or expends funds for any purpose other than for personal services and traveling expenses to aid or defeat the election of a speaker candidate.

(e) A person commits an offense if the person contributes personal services, money, or goods in violation of Section 302.018.

(e-1) A person commits an offense if the person knowingly makes a contribution to a speaker candidate's campaign or an expenditure to aid or defeat a speaker candidate from political contributions, interest earned on political contributions, or an asset purchased with political contributions in violation of Section 302.0191.

(f) An offense under this section is a Class A misdemeanor.

**Sec. 302.022. PROSECUTION BY INDICTMENT**. Each prosecution under this subchapter must be brought by indictment rather than by complaint and information.

# SUBCHAPTER C. LEGISLATIVE BRIBERY

**Sec. 302.031. DEFINITION**. In this subchapter, "economic benefit" means anything reasonably regarded as economic gain or advantage, including a campaign contribution.

Sec. 302.032. LEGISLATIVE BRIBERY: PROMISES OR THREATS. A person commits an offense if, with the intent to influence a member of or candidate for the house of representatives in casting a vote for speaker of the house of representatives, the person: (1) promises or agrees to cause:

(A) the appointment of a person to a chairmanship or vice-chairmanship of a house committee or subcommittee;

(B) the appointment of a person to a particular house committee or subcommittee, the Legislative Budget Board, the Texas Legislative Council, the Legislative Library Board, the Legislative Audit Committee, or any other position the speaker appoints;

- (C) preferential treatment on any legislation or appropriation;
- (D) the employment of a person; or
- (E) economic benefit to a person; or
- (2) threatens to cause:

(A) the failure to appoint a person to a chairmanship or vicechairmanship of a house committee or subcommittee;

(B) the failure to appoint a person to a particular house committee or subcommittee, the Legislative Budget Board, the Texas Legislative Council, the Legislative Library Board, the Legislative Audit Committee, or any other position the speaker appoints;

- (C) unfavorable treatment on any legislation or appropriation;
- (D) the refusal of or removal from employment of a person; or
- (E) the withholding of economic benefit from a person.

**Sec. 302.033. LEGISLATIVE BRIBERY: ACCEPTING BENEFITS**. A member of or candidate for the house of representatives commits an offense if, on the representation or understanding that the member or candidate will cast a vote for a particular person for speaker of the house of representatives, the member or candidate solicits, accepts, or agrees to accept:

(1) the appointment of or refusal to appoint a person to a chairmanship or vicechairmanship of a house committee or subcommittee;

(2) the appointment of or refusal to appoint a person to a particular house committee or subcommittee, the Legislative Budget Board, the Texas Legislative Council, the Legislative Library Board, the Legislative Audit Committee, or any other position the speaker appoints;

(3) preferential or unfavorable treatment on any legislation or appropriation;

(4) the employment of, refusal of employment of, or removal from employment of a person; or

(5) economic benefit to or withholding of economic benefit from a person.

Sec. 302.034. PENALTY. An offense under this subchapter is a felony punishable by imprisonment for not less than two years nor more than five years.

# Sec. 302.035. PERMITTED COMMUNICATIONS, DISCUSSIONS, AND

ADVOCACY. This subchapter does not prohibit:

(1) a person from contacting or communicating with a member of or candidate for the house of representatives about a legislative matter; or

(2) a member of or candidate for the house from discussing, taking a position on, or advocating any action on a substantive issue in a speaker's race or any other legislative matter.

# PERSONAL FINANCIAL DISCLOSURE, STANDARDS OF CONDUCT AND CONFLICT OF INTEREST

11



# **TEXAS ETHICS COMMISSION**

# **CHAPTER 572, GOVERNMENT CODE**

# PERSONAL FINANCIAL DISCLOSURE, STANDARDS OF CONDUCT, AND CONFLICT OF INTEREST



Effective September 1, 2021

(Revised 8/1/2023)

Texas Ethics Commission, P.O. Box 12070, Austin, Texas 78711 www.ethics.state.tx.us (512) 463-5800 • TDD (800) 735-2989 Promoting Public Confidence in Government

# **CHAPTER 572, GOVERNMENT CODE**

# PERSONAL FINANCIAL DISCLOSURE, STANDARDS OF CONDUCT, AND CONFLICT OF INTEREST

# TABLE OF CONTENTS

| CHAPTER 572. PERSONAL FINANCIAL DISCLOSURE, STANDARDS OF          |     |
|---|-----|
| CONDUCT, AND CONFLICT OF INTEREST                                 |     |
| SUBCHAPTER A. GENERAL PROVISIONS                                  | 1   |
| Sec. 572.001. POLICY; LEGISLATIVE INTENT                          | 1   |
| Sec. 572.002. GENERAL DEFINITIONS                                 |     |
| Sec. 572.003. DEFINITION: APPOINTED OFFICER OF MAJOR STATE AGENC  | Y 3 |
| Sec. 572.004. DEFINITION: REGULATION                              | 4   |
| Sec. 572.005. DETERMINATION OF SUBSTANTIAL INTEREST               | 4   |
| Sec. 572.006. DETERMINATION OF DEPENDENT CHILD                    | 5   |
| Sec. 572.007. PENALTIES IMPOSED BY COMMISSION                     | 5   |
| Sec. 572.008. VENUE   |     |
| SUBCHAPTER B. PERSONAL FINANCIAL STATEMENT                        | 5   |
| Sec. 572.021. FINANCIAL STATEMENT REQUIRED                        | 5   |
| Sec. 572.0211. FILING BY HOLDOVER OFFICER NOT REQUIRED            |     |
| Sec. 572.022. REPORTING CATEGORIES; REQUIRED DESCRIPTIONS         |     |
| Sec. 572.023. CONTENTS OF FINANCIAL STATEMENT IN GENERAL          |     |
| Sec. 572.024. INFORMATION ABOUT SERVICES FOR LOBBYISTS OR LOBBY   |     |
| EMPLOYERS   |     |
| Sec. 572.025. INFORMATION ABOUT LEGISLATORS' REPRESENTATION BEF   |     |
| EXECUTIVE STATE AGENCIES  |     |
| Sec. 572.0251. INFORMATION ABOUT LEGISLATIVE CONTINUANCES         |     |
| Sec. 572.0252. INFORMATION ABOUT REFERRALS                        |     |
| Sec. 572.026. FILING DATES FOR STATE OFFICERS AND STATE PARTY CHA |     |
|   |     |
| Sec. 572.027. FILING DATES FOR CANDIDATES                         |     |
| Sec. 572.028. DUPLICATE STATEMENTS                                |     |
| Sec. 572.029. TIMELINESS OF FILING                                |     |
| Sec. 572.0291. ELECTRONIC FILING REQUIRED                         | 11  |
| Sec.572.0295. AMENDMENT OF FINANCIAL STATEMENT                    |     |
| Sec. 572.030. PREPARATION AND MAILING OF FORMS                    |     |
| Sec. 572.031. DETERMINATION OF COMPLIANCE WITH SUBCHAPTER         |     |
| Sec. 572.032. PUBLIC ACCESS TO STATEMENTS                         |     |
| Sec. 572.033. CIVIL PENALTY                                       |     |
| Sec. 572.034. CRIMINAL PENALTY                                    |     |
| Sec. 572.035. REMOVAL OF PERSONAL INFORMATION FOR FEDERAL JUDG    |     |
| STATE JUDGES, AND FAMILY MEMBERS.                                 |     |
| SUBCHAPTER C. STANDARDS OF CONDUCT                                |     |
| AND CONFLICT OF INTEREST PROVISIONS                               | 14  |

| Sec. 572.051. STANDARDS OF CONDUCT; STATE AGENCY ETHICS POLICY 14    | 4 |
|--|---|
| Sec. 572.052. REPRESENTATION BY LEGISLATORS BEFORE STATE AGENCIES;   |   |
| CRIMINAL OFFENSE   | 5 |
| Sec. 572.053. VOTING BY LEGISLATORS ON CERTAIN MEASURES OR BILLS;    |   |
| CRIMINAL OFFENSE   | 5 |
| Sec. 572.0531. NOTICE REQUIRED FOR INTRODUCTION OR SPONSORSHIP OF OF | R |
| VOTING ON CERTAIN MEASURES OR BILLS BY LEGISLATORS 10                | 6 |
| Sec. 572.054. REPRESENTATION BY FORMER OFFICER OR EMPLOYEE OF        |   |
| REGULATORY AGENCY RESTRICTED; CRIMINAL OFFENSE 10                    | 6 |
| Sec. 572.055. CERTAIN SOLICITATIONS OF REGULATED BUSINESS ENTITIES   |   |
| PROHIBITED; CRIMINAL OFFENSE 1                                       | 7 |
| Sec. 572.056. CONTRACTS BY STATE OFFICERS WITH GOVERNMENTAL          |   |
| ENTITIES; CRIMINAL OFFENSE   | 8 |
| Sec. 572.057. CERTAIN LEASES PROHIBITED                              | 8 |
| Sec. 572.058. PRIVATE INTEREST IN MEASURE OR DECISION; DISCLOSURE;   |   |
| REMOVAL FROM OFFICE FOR VIOLATION                                    | 8 |
| Sec. 572.059. INDEPENDENCE OF STATE AND LOCAL OFFICERS ACTING IN     |   |
| LEGISLATIVE CAPACITY   | 9 |
| Sec. 572.060. SOLICITATION OF OR RECOMMENDATIONS REGARDING           |   |
| CONTRIBUTIONS TO CHARITABLE ORGANIZATIONS AND GOVERNMENTAL           |   |
| ENTITIES   | 9 |
| Sec. 572.061. CERTAIN GRATUITIES AUTHORIZED                          | 0 |
| Sec. 572.069. CERTAIN EMPLOYMENT FOR FORMER STATE OFFICER OR         |   |
| EMPLOYEE RESTRICTED  | 0 |

# **GOVERNMENT CODE**

# TITLE 5. OPEN GOVERNMENT; ETHICS

# **SUBTITLE B. ETHICS**

# CHAPTER 572. PERSONAL FINANCIAL DISCLOSURE, STANDARDS OF CONDUCT, AND CONFLICT OF INTEREST

### SUBCHAPTER A. GENERAL PROVISIONS

#### Sec. 572.001. POLICY; LEGISLATIVE INTENT.

(a) It is the policy of this state that a state officer or state employee may not have a direct or indirect interest, including financial and other interests, or engage in a business transaction or professional activity, or incur any obligation of any nature that is in substantial conflict with the proper discharge of the officer's or employee's duties in the public interest.

(b) To implement this policy and to strengthen the faith and confidence of the people of this state in state government, this chapter provides standards of conduct and disclosure requirements to be observed by persons owing a responsibility to the people and government of this state in the performance of their official duties.

(c) It is the intent of the legislature that this chapter serve not only as a guide for official conduct of those persons but also as a basis for discipline of those who refuse to abide by its terms.

### Sec. 572.002. GENERAL DEFINITIONS. In this chapter:

(1) "Appointed officer" means:

(A) the secretary of state;

(B) an individual appointed with the advice and consent of the senate to the governing board of a state-supported institution of higher education;

(C) an officer of a state agency who is appointed for a term of office specified by the Texas Constitution or a statute of this state, excluding an appointee to a vacated elective office; or

(D) an individual who is a member of the governing board or commission of a state agency, who is not appointed, and who is not otherwise:

- (i) an elected officer;
- (ii) an officer described by Paragraphs (A) through (C); or
- (iii) an executive head of a state agency.

(2) "Business entity" means any entity recognized by law through which business for profit is conducted, including a sole proprietorship, partnership, firm, corporation, holding company, joint stock company, receivership, or trust.

- (3) "Commission" means the Texas Ethics Commission.
- (4) "Elected officer" means:
  - (A) a member of the legislature;
  - (B) an executive or judicial officer elected in a statewide election;
  - (C) a judge of a court of appeals or of a district court;
  - (D) a member of the State Board of Education;
  - (E) a district attorney or criminal district attorney; or

(F) an individual appointed to fill a vacancy in an office or appointed to a newly created office who, if elected to the office instead of appointed, would be an elected officer under this subdivision.

(5) "Executive head of a state agency" means the director, executive director, commissioner, administrator, chief clerk, or other individual who is appointed by the governing body or highest officer of the state agency to act as the chief executive or administrative officer of the agency and who is not an appointed officer. The term includes the chancellor or highest executive officer of a university system and the president of a public senior college or university as defined by Section 61.003, Education Code.

(6) "State party chair" means the state chair of any political party receiving more than two percent of the vote for governor in the most recent general election.

(7) "Person" means an individual or a business entity.

(8) "Regulatory agency" means any department, commission, board, or other agency, except the secretary of state and the comptroller, that:

- (A) is in the executive branch of state government;
- (B) has authority that is not limited to a geographical portion of the state;
- (C) was created by the Texas Constitution or a statute of this state; and
- (D) has constitutional or statutory authority to engage in regulation.

(9) "Salaried appointed officer" means an appointed officer who receives or is authorized to receive a salary for state service but not a per diem or other form of compensation.

(10) "State agency" means:

(A) a department, commission, board, office, or other agency that:

(i) is in the executive branch of state government;

(ii) has authority that is not limited to a geographical portion of

the state; and

(iii) was created by the Texas Constitution or a statute of this

state;

(B) a university system or an institution of higher education as defined by Section 61.003, Education Code, other than a public junior college; or

(C) a river authority created under the Texas Constitution or a statute of

this state.

(11) "State employee" means an individual, other than a state officer, who is employed by:

(A) a state agency;

(B) the Supreme Court of Texas, the Court of Criminal Appeals of Texas, a court of appeals, or the Texas Judicial Council; or

(C) either house of the legislature or a legislative agency, council, or committee, including the Legislative Budget Board, the Texas Legislative Council, the State Auditor's Office, and the Legislative Reference Library.

(11-a) "State judge" means:

(A) a judge, former judge, or retired judge of an appellate court, a district court, a constitutional county court, a county court at law, or a statutory probate court of this state;

(B) an associate judge appointed under Chapter 201, Family Code, or a retired associate judge or former associate judge appointed under that chapter;

(C) a magistrate or associate judge appointed under Chapter 54 or 54A;

(D) a justice of the peace; or

(E) a municipal court judge.

(12) "State officer" means an elected officer, an appointed officer, a salaried appointed officer, an appointed officer of a major state agency, or the executive head of a state agency.

# Sec. 572.003. DEFINITION: APPOINTED OFFICER OF MAJOR STATE AGENCY.

(a) In this chapter, "appointed officer of a major state agency" means an individual listed in Subsection (b) or (c).

(b) The term means:

(1) the Banking Commissioner of The Banking Department of Texas;

(2) the administrative director of the Office of Court Administration of the Texas Judicial System;

(3) the chief executive of the Office of Public Utility Counsel;

(4) the executive director of the State Bar of Texas;

(5) the director of the lottery division of the Texas Lottery Commission;

(6) the deputy in charge of the department of security in the lottery division of the Texas Lottery Commission;

- (7) the director of the bingo division of the Texas Lottery Commission; or
- (8) the secretary of state.

(c) The term means a member of:

- (1) the Public Utility Commission of Texas;
- (2) the Texas Commission on Environmental Quality;
- (3) the Texas Alcoholic Beverage Commission;
- (4) the Finance Commission of Texas;
- (5) the Texas Facilities Commission;
- (6) the Texas Board of Criminal Justice;
- (7) the board of trustees of the Employees Retirement System of Texas;
- (8) the Texas Transportation Commission;
- (9) the Texas Department of Insurance;
- (10) the Parks and Wildlife Commission;
- (11) the Public Safety Commission;
- (12) the Texas Ethics Commission;
- (13) the State Securities Board;
- (14) the Texas Water Development Board;

(15) the governing board of a public senior college or university as defined by Section 61.003, Education Code, or of The University of Texas Southwestern Medical Center, The University of Texas Medical Branch at Galveston, The University of Texas Health Science Center at Houston, The University of Texas Health Science Center at San Antonio, The University of Texas M. D. Anderson Cancer Center, The University of Texas Health Science Center at Tyler, University of North Texas Health Science Center at Fort Worth, Texas Tech University Health Sciences Center, Texas State Technical College--Harlingen, Texas State Technical College--Marshall, Texas State Technical College--Sweetwater, or Texas State Technical College--Waco;

- (16) the Texas Higher Education Coordinating Board;
- (17) the Texas Workforce Commission;

- (18) the board of trustees of the Teacher Retirement System of Texas;
- (19) the Credit Union Commission;
- (20) the School Land Board;
- (21) the board of the Texas Department of Housing and Community Affairs;
- (22) the Texas Racing Commission;
- (23) the State Board of Dental Examiners;
- (24) the Texas Medical Board;
- (25) the Board of Pardons and Paroles;
- (26) the Texas State Board of Pharmacy;
- (27) the Department of Information Resources governing board;
- (28) the board of the Texas Department of Motor Vehicles-Board;
- (29) the Texas Real Estate Commission;
- (30) the board of directors of the State Bar of Texas;
- (31) the Bond Review Board;
- (32) the Health and Human Services Commission;
- (33) the Texas Funeral Service Commission;

(34) the board of directors of a river authority created under the Texas Constitution or a statute of this state;

(35) the Texas Lottery Commission; or

(36) the Cancer Prevention and Research Institute of Texas.

(d) The term includes the successor in function as provided by law to an office listed in Subsection (b) or (c) if that office is abolished.

**Sec. 572.004. DEFINITION: REGULATION**. In this chapter, "regulation" means rulemaking, adjudication, or licensing. In this definition:

(1) "Adjudication" means the process of an agency for formulating an order.

(2) "License" includes all or part of an agency permit, certificate, approval, registration, charter, membership, statutory exemption, or other form of permission.

(3) "Licensing" includes the process of an agency concerning the grant, renewal, denial, revocation, suspension, annulment, withdrawal, limitation, amendment, modification, or conditioning of a license.

(4) "Order" means all or part of a final disposition, whether affirmative, negative, injunctive, or declaratory in form, of an agency in a matter other than rulemaking but including licensing.

(5) "Rule" means all or part of an agency statement of general or particular applicability and future effect designed to implement, interpret, or prescribe law or policy or to describe the organization, procedure, or practice requirements of an agency.

**Sec. 572.005. DETERMINATION OF SUBSTANTIAL INTEREST**. An individual has a substantial interest in a business entity if the individual:

- (1) has a controlling interest in the business entity;
- (2) owns more than 10 percent of the voting interest in the business entity;
- (3) owns more than \$25,000 of the fair market value of the business entity;
- (4) has a direct or indirect participating interest by shares, stock, or otherwise, regardless of whether voting rights are included, in more than 10 percent of the profits, proceeds, or capital gains of the business entity;

(5) is a member of the board of directors or other governing board of the business entity;

- (6) serves as an elected officer of the business entity; or
- (7) is an employee of the business entity.

Sec. 572.006. DETERMINATION OF DEPENDENT CHILD. An individual's child, including an adopted child or stepchild, is the individual's dependent during a calendar year in which the individual provides more than 50 percent of the child's support.

Sec. 572.007. PENALTIES IMPOSED BY COMMISSION. This chapter does not prohibit the imposition of civil penalties by the commission in addition to criminal penalties or other sanctions imposed by law.

**Sec. 572.008. VENUE.** An offense under this chapter, including perjury, may be prosecuted in Travis County or in any other county in which it may be prosecuted under the Code of Criminal Procedure.

# SUBCHAPTER B. PERSONAL FINANCIAL STATEMENT

Sec. 572.021. FINANCIAL STATEMENT REQUIRED. Except as provided by Section 572.0211, a state officer, a partisan or independent candidate for an office as an elected officer, and a state party chair shall file with the commission a verified financial statement complying with Sections 572.022 through 572.0252.

### Sec. 572.0211. FILING BY HOLDOVER OFFICER NOT REQUIRED.

(a) An appointed officer who resigns from office and who ceases to participate in the state agency's functions is not required to file a financial statement that is due because of service in that office after the effective date of the resignation.

(b) An appointed officer whose term of office expires and who ceases to participate in the functions of the state agency is not required to file a financial statement that is due because of service in that office after the date the term of office expires.

(c) An appointed officer of a state agency that is abolished or whose functions are transferred to another state agency is not required to file a financial statement that is due because of service after the date that the agency is abolished or the functions of the agency are transferred.

(d) An appointed officer who resigns or whose term of office expires who does not intend to participate in the functions of the state agency shall deliver written notice of the officer's intention to the governor and the commission.

# Sec. 572.022. REPORTING CATEGORIES; REQUIRED DESCRIPTIONS.

(a) If an amount in a financial statement is required to be reported by category, the individual filing the statement shall report whether the amount is:

- (1) less than \$5,000;
- (2) at least \$5,000 but less than \$10,000;
- (3) at least \$10,000 but less than \$25,000; or
- (4) \$25,000 or more.

(b) The individual filing the statement shall report an amount of stock by category of number of shares instead of by category of dollar value and shall report whether the amount is:

- (1) less than 100 shares;
- (2) at least 100 but less than 500 shares;
- (3) at least 500 but less than 1,000 shares;
- (4) at least 1,000 but less than 5,000 shares;
- (5) at least 5,000 but less than 10,000 shares; or
- (6) 10,000 shares or more.

(c) The individual filing the statement shall report a description of real property by reporting:

(1) the street address, if available, or the number of lots or number of acres, as applicable, in each county, and the name of the county, if the street address is not available; and

(2) the names of all persons retaining an interest in the property, excluding an interest that is a severed mineral interest.

(d) For a gift of cash or a cash equivalent such as a negotiable instrument or gift certificate that is reported in accordance with Section 572.023(b)(7), the individual filing the statement shall include in the description of the gift a statement of the value of the gift.

# Sec. 572.023. CONTENTS OF FINANCIAL STATEMENT IN GENERAL.

(a) A financial statement must include an account of the financial activity of the individual required by this subchapter to file a financial statement and an account of the financial activity of the individual's spouse and dependent children if the individual had actual control over that activity for the preceding calendar year.

(b) The account of financial activity consists of:

(1) a list of all sources of occupational income, identified by employer, or if selfemployed, by the nature of the occupation, including identification of a person or other organization from which the individual or a business in which the individual has a substantial interest received a fee as a retainer for a claim on future services in case of need, as distinguished from a fee for services on a matter specified at the time of contracting for or receiving the fee, if professional or occupational services are not actually performed during the reporting period equal to or in excess of the amount of the retainer, and the category of the amount of the fee;

(2) identification by name and the category of the number of shares of stock of any business entity held or acquired, and if sold, the category of the amount of net gain or loss realized from the sale;

(3) a list of all bonds, notes, and other commercial paper held or acquired, and if sold, the category of the amount of net gain or loss realized from the sale;

(4) identification of each source and the category of the amount of income in excess of \$500 derived from each source from interest, dividends, royalties, and rents;

(5) identification of each guarantor of a loan and identification of each person or financial institution to whom a personal note or notes or lease agreement for a total financial liability in excess of \$1,000 existed at any time during the year and the category of the amount of the liability;

(6) identification by description of all beneficial interests in real property and business entities held or acquired, and if sold, the category of the amount of the net gain or loss realized from the sale;

(7) identification of a person or other organization from which the individual or the individual's spouse or dependent children received a gift of anything of value in excess of \$250 and a description of each gift, except:

(A) a gift received from an individual related to the individual at any time within the second degree by consanguinity or affinity, as determined under Subchapter B, Chapter 573;

(B) a political contribution that was reported as required by Chapter 254, Election Code; and

(C) an expenditure required to be reported by a person required to be registered under Chapter 305;

(8) identification of the source and the category of the amount of all income received as beneficiary of a trust, other than a blind trust that complies with Subsection (c), and identification of each trust asset, if known to the beneficiary, from which income was received by the beneficiary in excess of \$500;

(9) identification:

(A) by description of a corporation, firm, partnership, limited partnership, limited liability partnership, professional corporation, professional association, joint venture, or other business association in which five percent or more of the outstanding ownership was held, acquired, or sold; and

(B) by description and the category of the amount of all assets and liabilities of a corporation, firm, partnership, limited partnership, limited liability partnership, professional corporation, professional association, joint venture, or other business association in which 50 percent or more of the outstanding ownership was held, acquired, or sold;

(10) a list of all boards of directors of which the individual is a member and executive positions that the individual holds in corporations, firms, partnerships, limited partnerships, limited liability partnerships, professional corporations, professional associations, joint ventures, or other business associations or proprietorships, stating the name of each corporation, firm, partnership, limited partnership, limited liability partnership, professional corporation, professional association, joint venture, or other business association or proprietorship and the position held;

(11) identification of any person providing transportation, meals, or lodging expenses permitted under Section 36.07(b), Penal Code, and the amount of those expenses, other than expenditures required to be reported under Chapter 305;

(12) any corporation, firm, partnership, limited partnership, limited liability partnership, professional corporation, professional association, joint venture, or other business association, excluding a publicly held corporation, in which both the individual and a person registered under Chapter 305 have an interest;

(13) identification by name and the category of the number of shares of any mutual fund held or acquired, and if sold, the category of the amount of net gain or loss realized from the sale;

(14) identification of each blind trust that complies with Subsection (c), including:

- (A) the category of the fair market value of the trust;
- (B) the date the trust was created;
- (C) the name and address of the trustee; and

(D) a statement signed by the trustee, under penalty of perjury, stating

that:

(i) the trustee has not revealed any information to the individual, except information that may be disclosed under Subdivision (8); and

(ii) to the best of the trustee's knowledge, the trust complies with this section;

(15) if the aggregate cost of goods or services sold under one or more written contracts described by this subdivision exceeds \$10,000 in the year covered by the report, identification of each written contract, including the name of each party to the contract:

(A) for the sale of goods or services in the amount of \$2,500 or more;

(B) to which the individual, the individual's spouse, the individual's dependent child, or any business entity of which the individual, the individual's spouse, or the individual's dependent child, independently or in conjunction with one or more persons described by this subsection, has at least a 50 percent ownership interest is a party; and

(C) with:

(i) a governmental entity; or

(ii) a person who contracts with a governmental entity, if the individual or entity described by Paragraph (B) performs work arising out of the contract, subcontract, or agreement between the person and the governmental entity for a fee; and

(16) if the individual is a member of the legislature and provides bond counsel services to an issuer, as defined by Section 1201.002(1), identification of the following for each issuance for which the individual served as bond counsel:

(A) the amount of the issuance;

(B) the name of the issuer;

(C) the date of the issuance;

(D) the amount of fees paid to the individual, and whether the amount is:

(i) less than \$5,000;

(ii) at least \$5,000 but less than \$10,000;

(iii) at least \$10,000 but less than \$25,000; or

(iv) \$25,000 or more; and

(E) the amount of fees paid to the individual's firm, if applicable, and whether the amount is:

(i) less than \$5,000;

(ii) at least \$5,000 but less than \$10,000;

(iii) at least \$10,000 but less than \$25,000; or

(iv) \$25,000 or more.

(c) For purposes of Subsections (b)(8) and (14), a blind trust is a trust as to which:

(1) the trustee:

(A) is a disinterested party;

(B) is not the individual;

(C) is not required to register as a lobbyist under Chapter 305;

(D) is not a public officer or public employee; and

(E) was not appointed to public office by the individual or by a public officer or public employee the individual supervises; and

(2) the trustee has complete discretion to manage the trust, including the power to dispose of and acquire trust assets without consulting or notifying the individual.

(d) If a blind trust under Subsection (c) is revoked while the individual is subject to this subchapter, the individual must file an amendment to the individual's most recent financial statement, disclosing the date of revocation and the previously unreported value by category of each asset and the income derived from each asset.

(e) In this section, "governmental entity" means this state, a political subdivision of the state, or an agency or department of the state or a political subdivision of the state.

(f) Subsection (b)(15) does not require the disclosure of an employment contract between a school district or open-enrollment charter school and an employee of the district or school.

(g) An individual who complies with any applicable requirements of Sections 51.954 and 51.955, Education Code, and Section 2252.908 of this code, in an individual capacity or as a member or employee of an entity to which those sections apply, is not required to include in the account of financial activity the information described by Subsection (b)(15) unless specifically requested by the commission to include the information.

Sec. 572.024. INFORMATION ABOUT SERVICES FOR LOBBYISTS OR LOBBYIST EMPLOYERS. A state officer who receives a fee for services rendered by the officer to or on behalf of a person required to be registered under Chapter 305, or to or on behalf of a person or entity that the officer actually knows directly compensates or reimburses a person required to be registered under Chapter 305, shall report on the financial statement the name of each person or entity for which the services were rendered and the category of the amount of each fee.

Sec. 572.025. INFORMATION ABOUT LEGISLATORS' REPRESENTATION BEFORE EXECUTIVE STATE AGENCIES. A member of the legislature who represents another person for compensation before an executive state agency shall report on the financial statement:

- (1) the name of the agency;
- (2) the person represented by the member; and

(3) the category of the amount of compensation received by the member for that representation.

Sec. 572.0251. INFORMATION ABOUT LEGISLATIVE CONTINUANCES. A

member or member-elect of the legislature licensed to practice law in this state who represents a party to a civil or criminal case for compensation and on that party's behalf applies for or obtains a legislative continuance under Section 30.003, Civil Practice and Remedies Code, or under another law or rule that requires or permits a court to grant a continuance on the grounds that an attorney for a party is a member or member-elect of the legislature shall report on the financial statement:

- (1) the name of the party represented;
- (2) the date on which the member or member-elect was retained to represent the

party;

(3) the style and cause number of the action in which the continuance was sought and the court and jurisdiction in which the action was pending when the continuance was sought;

(4) the date on which the member or member-elect applied for a continuance;

and

(5) whether the continuance was granted.

Sec. 572.0252. INFORMATION ABOUT REFERRALS. A state officer who is an attorney shall report on the financial statement:

(1) making or receiving any referral for compensation for legal services; and

(2) the category of the amount of any fee accepted for making a referral for legal services.

## Sec. 572.026. FILING DATES FOR STATE OFFICERS AND STATE PARTY CHAIRS.

(a) Not later than April 30 each year, a state officer or a state party chair shall file the financial statement as required by this subchapter.

(b) An individual who is appointed to serve as a salaried appointed officer or an appointed officer of a major state agency or who is appointed to fill a vacancy in an elective office shall file a financial statement not later than the 30th day after the date of appointment or the date of qualification for the office, or if confirmation by the senate is required, before the first committee hearing on the confirmation, whichever date is earlier.

(c) An individual who is appointed or employed as the executive head of a state agency shall file a financial statement not later than the 45th day after the date on which the individual assumes the duties of the position. A state agency shall immediately notify the commission of the appointment or employment of an executive head of the agency.

(d) An individual required to file a financial statement under Subsection (a) may request the commission to grant an extension of not more than 60 days for filing the statement. The commission shall grant the request if it is received before the filing deadline or if a timely filing or request for extension is prevented because of physical or mental incapacity. The commission may not grant more than one extension to an individual in one year except for good cause shown.

#### Sec. 572.027. FILING DATES FOR CANDIDATES.

(a) An individual who is a partisan or independent candidate for an office as an elected officer shall file the financial statement required by this subchapter not later than the later of:

(1) the 60th day after the date of the regular filing deadline for an application for a place on the ballot in the general primary election; or

(2) February 12.

(b) If the deadline under which a candidate files an application for a place on the ballot, other than the regular filing deadline for an independent candidate, or files a declaration of writein candidacy falls after the date of the regular filing deadline for candidates in the general primary election, the candidate shall file the financial statement not later than the 30th day after that later deadline. However, if that deadline falls after the 35th day before the date of the election in which the candidate is running, the candidate shall file the statement not later than the fifth day before the date of that election.

(c) An individual who is a candidate in a special election for an office as an elected officer shall file the financial statement not later than the fifth day before the date of that election.

(d) An individual nominated to fill a vacancy in a nomination as a candidate for a position as an elected officer under Chapter 145, Election Code, shall file the financial statement not later than the 15th day after the date the certificate of nomination required by Section 145.037 or 145.038, Election Code, is filed.

Sec. 572.028. DUPLICATE STATEMENTS. If an individual has filed a financial statement under one provision of this subchapter covering the preceding calendar year, the individual is not required to file a financial statement required under another provision of this subchapter to cover that same year if, before the deadline for filing the statement under the other provision, the individual notifies the commission in writing that the individual has already filed a financial statement under the provision specified.

#### Sec. 572.029. TIMELINESS OF FILING.

(a) The deadline for filing a financial statement required by this subchapter is 5 p.m. of the last day designated in the applicable provision for filing the statement.

(b) If the last day for filing the financial statement is a Saturday, Sunday, or holiday included under Subchapter B, Chapter 662, the statement is timely if filed on the next day that is not a Saturday, Sunday, or listed holiday.

(c) A financial statement is timely filed if it is properly addressed and placed in the United States Post Office or in the hands of a common or contract carrier not later than the last day for filing the financial statement. The post office cancellation mark or the receipt mark of a common or contract carrier is prima facie evidence of the date the statement was deposited with the post office or carrier. The individual filing the statement may show by competent evidence that the actual date of posting was different from that shown by the marks.

#### Sec. 572.0291. ELECTRONIC FILING REQUIRED.

(a) Except as provided by Subsection (b), a financial statement filed with the commission must be filed by computer diskette, modem, or other means of electronic transfer, using computer software provided by the commission or computer software that meets commission specifications for a standard file format.

(b) An individual who was appointed to office and who is required to file a financial statement with the commission under this subchapter may file the financial statement by certified mail. The filing by mail must be in compliance with Section 572.029.

#### Sec.572.0295. AMENDMENT OF FINANCIAL STATEMENT.

(a) A person who files a financial statement under this chapter may amend the person's statement.

(b) A financial statement that is amended is considered to have been filed on the date on which the original statement was filed if:

(1) the amendment is made on or before the 14th day after the date the person filing the statement learns of an error or omission in the original statement;

(2) the original financial statement was made in good faith and without an intent to mislead or to misrepresent the information contained in the statement; and

(3) the person filing the amendment accompanies the amendment with a declaration that:

(A) the person became aware of the error or omission in the original statement during the preceding 14 days; and

(B) the original statement was made in good faith and without intent to mislead or to misrepresent the information contained in the statement.

#### Sec. 572.030. PREPARATION AND MAILING OF FORMS.

(a) The commission shall design forms that may be used for filing the financial statement under this subchapter.

(b) The commission shall mail to each individual required to file under this subchapter a notice that:

(1) states that the individual is required to file a financial statement under this subchapter;

(2) identifies the filing dates for the financial statement as provided by Sections 572.026 and 572.027;

(3) describes the manner in which the individual may obtain the financial statement forms and instructions from the commission's Internet website;

(4) states that on request of the individual, the commission will mail to the individual a copy of the financial statement forms and instructions; and

(5) states, if applicable, the fee for mailing the forms and instructions and the manner in which the individual may pay the fee.

(c) The notice required by Subsection (b) must be mailed:

(1) before the 30th day before the deadline for filing the financial statement under Section 572.026(a) or (c), except as otherwise provided by this subsection;

(2) not later than the 15th day after the applicable deadline for filing an application for a place on the ballot or a declaration of write-in candidacy for candidates required to file under Section 572.027(a), (b), or (c);

(3) not later than the seventh day after the date of appointment for individuals required to file under Section 572.026(b), or if the legislature is in session, sooner if possible; and

(4) not later than the fifth day after the date the certificate of nomination is filed for candidates required to file under Section 574.027(d).

(d) The commission shall mail a copy of the financial statement forms and instructions to an individual not later than the third business day after the date the commission receives the individual's request for the forms and instructions.

(e) The commission may charge a fee for mailing the financial statement forms and instructions to an individual. The amount of the fee may not exceed the reasonable cost of producing and mailing the forms and instructions.

#### Sec. 572.031. DETERMINATION OF COMPLIANCE WITH SUBCHAPTER.

(a) The commission shall conduct a continuing survey to determine whether all individuals required to file financial statements under this subchapter have filed statements in compliance with this subchapter.

(b) If the commission determines that an individual has failed to file the statement in compliance with this subchapter, the commission shall send a written statement of the determination to the appropriate prosecuting attorneys of the state.

#### Sec. 572.032. PUBLIC ACCESS TO STATEMENTS.

(a) Financial statements filed under this subchapter are public records. The commission shall maintain the statements in separate alphabetical files and in a manner that is accessible to the public during regular office hours.

Text of subdivision as added by Acts 2017, 85th R.S., ch. 983 (H.B. 776)

(a-1) The commission shall remove the home address, the telephone number, and the names of the dependent children of an individual from a financial statement filed by the individual under this subchapter before:

(1) permitting a member of the public to view the statement;

(2) providing a copy of the statement to a member of the public; or

(3) making the statement available to the public on the commission's Internet website, if the commission makes statements filed under this subchapter available on its website.

Text of subdivision as added by Acts 2017, 85th R.S., ch. 34 (S.B. 1576)

(a-1) Before permitting a member of the public to view a financial statement filed under this subchapter or providing a copy of the statement to a member of the public, the commission shall remove from the statement, if applicable, the home address of:

(1) a judge or justice; or

(2) a member of the governing board or executive head of the Texas Civil Commitment Office.

(b) During the one-year period following the filing of a financial statement, each time a person requests to see the financial statement, excluding the commission or a commission employee acting on official business, the commission shall place in the file a statement of the person's name and address, whom the person represents, and the date of the request. The commission shall retain that statement in the file for one year after the date the requested financial statement is filed.

(c) After the second anniversary of the date the individual ceases to be a state officer, the commission may and on notification from the former state officer shall destroy each financial statement filed by the state officer.

#### Sec. 572.033. CIVIL PENALTY.

(a) The commission shall determine from any available evidence whether a statement required to be filed under this subchapter is late. On making a determination that the statement is late, the commission shall immediately mail a notice of the determination to the individual responsible for filing the statement and to the appropriate attorney for the state.

(b) If a statement is determined to be late, the individual responsible for filing the statement is liable to the state for a civil penalty of \$500. If a statement is more than 30 days late, the commission shall issue a warning of liability by registered mail to the individual responsible for the filing. If the penalty is not paid before the 10th day after the date on which the warning is received, the individual is liable for a civil penalty in an amount determined by commission rule, but not to exceed \$10,000.

(c) This section is cumulative of any other available sanction for a late filing of a sworn statement.

#### Sec. 572.034. CRIMINAL PENALTY.

(a) An individual commits an offense if the individual is a state officer or candidate or state party chair and knowingly and wilfully fails to file a financial statement as required by this subchapter.

(b) An offense under this section is a Class B misdemeanor.

(c) In a prosecution for failure to file a financial statement under this section, it is a defense that the individual did not receive copies of the financial statement form required by this subchapter to be mailed to the individual.

## Sec. 572.035. REMOVAL OF PERSONAL INFORMATION FOR FEDERAL JUDGES, STATE JUDGES, AND FAMILY MEMBERS.

(a) On receiving notice from the Office of Court Administration of the Texas Judicial System of the judge's qualification for the judge's office, the commission shall remove or redact from any financial statement, or information derived from a financial statement, that is available to the public the residence address of a federal judge, including a federal bankruptcy judge, a state judge, or the family member of a federal judge, including a federal bankruptcy judge or a state judge.

(b) In this section, "family member" has the meaning assigned by Section 31.006, Family Code.

#### SUBCHAPTER C. STANDARDS OF CONDUCT AND CONFLICT OF INTEREST PROVISIONS

## Sec. 572.051. STANDARDS OF CONDUCT; STATE AGENCY ETHICS POLICY.

(a) A state officer or employee should not:

(1) accept or solicit any gift, favor, or service that might reasonably tend to influence the officer or employee in the discharge of official duties or that the officer or employee knows or should know is being offered with the intent to influence the officer's or employee's official conduct;

(2) accept other employment or engage in a business or professional activity that the officer or employee might reasonably expect would require or induce the officer or employee to disclose confidential information acquired by reason of the official position;

(3) accept other employment or compensation that could reasonably be expected to impair the officer's or employee's independence of judgment in the performance of the officer's or employee's official duties;

(4) make personal investments that could reasonably be expected to create a substantial conflict between the officer's or employee's private interest and the public interest; or

(5) intentionally or knowingly solicit, accept, or agree to accept any benefit for having exercised the officer's or employee's official powers or performed the officer's or employee's official duties in favor of another.

(b) A state employee who violates Subsection (a) or an ethics policy adopted under Subsection (c) is subject to termination of the employee's state employment or another employment-related sanction. Notwithstanding this subsection, a state officer or employee who violates Subsection (a) is subject to any applicable civil or criminal penalty if the violation also constitutes a violation of another statute or rule. (c) Each state agency shall:

(1) adopt a written ethics policy for the agency's employees consistent with the standards prescribed by Subsection (a) and other provisions of this subchapter; and

(2) distribute a copy of the ethics policy and this subchapter to:

(A) each new employee not later than the third business day after the date the person begins employment with the agency; and

(B) each new officer not later than the third business day after the date the person qualifies for office.

(d) The office of the attorney general shall develop, in coordination with the commission, and distribute a model policy that state agencies may use in adopting an agency ethics policy under Subsection (c). A state agency is not required to adopt the model policy developed under this subsection.

(e) Subchapters E and F, Chapter 571, do not apply to a violation of this section.

(f) Notwithstanding Subsection (e), if a person with knowledge of a violation of an agency ethics policy adopted under Subsection (c) that also constitutes a criminal offense under another law of this state reports the violation to an appropriate prosecuting attorney, then, not later than the 60th day after the date a person notifies the prosecuting attorney under this subsection, the prosecuting attorney shall notify the commission of the status of the prosecuting attorney's investigation of the alleged violation. The commission shall, on the request of the prosecuting attorney, assist the prosecuting attorney in investigating the alleged violation. This subsection does not apply to an alleged violation by a member or employee of the commission.

## Sec. 572.052. REPRESENTATION BY LEGISLATORS BEFORE STATE AGENCIES; CRIMINAL OFFENSE.

(a) A member of the legislature may not, for compensation, represent another person before a state agency in the executive branch of state government unless the representation:

(1) is pursuant to an attorney-client relationship in a criminal law matter; or

(2) involves the filing of documents that involve only ministerial acts on the part of the commission, agency, board, department, or officer.

(b) A member of the legislature commits an offense if the member violates this section. An offense under this subsection is a Class A misdemeanor.

## Sec. 572.053. VOTING BY LEGISLATORS ON CERTAIN MEASURES OR BILLS; CRIMINAL OFFENSE.

(a) A member of the legislature may not vote on a measure or a bill, other than a measure that will affect an entire class of business entities, that will directly benefit a specific business transaction of a business entity in which the member has a controlling interest.

(b) In this section, "controlling interest" includes:

(1) an ownership interest or participating interest by virtue of shares, stock, or otherwise that exceeds 10 percent;

(2) membership on the board of directors or other governing body of the business entity; or

(3) service as an officer of the business entity.

(c) A member of the legislature commits an offense if the member violates this section. An offense under this subsection is a Class A misdemeanor.

#### Sec. 572.0531. NOTICE REQUIRED FOR INTRODUCTION OR SPONSORSHIP OF OR VOTING ON CERTAIN MEASURES OR BILLS BY LEGISLATORS.

(a) A member shall file a notice as required by Subsection (b) before introducing, sponsoring, or voting on a measure or bill if the member's spouse or a person related to the member within the first degree by consanguinity, as determined under Subchapter B, Chapter 573, is registered as a lobbyist under Chapter 305 with respect to the subject matter of the measure or bill.

(b) A member of the house of representatives to whom Subsection (a) applies shall file a written notice of that fact with the chief clerk of the house of representatives. A senator to whom Subsection (a) applies shall file a written notice of that fact with the secretary of the senate. The member shall also file a notice with the commission. A notice filed under this subsection must:

(1) identify:

(A) the member;

(B) the measure, bill, or class of measures or bills with respect to which the notice is required under this section; and

(C) the person registered as a lobbyist; and

(2) be included in the journal of the house to which the member belongs.

(c) A person related to the member to whom Subsection (a) applies shall file a notice with the commission identifying:

(1) the person;

(2) the member; and

(3) the class of measures or bills with respect to which notice is required under this section.

(d) A person related to the member to whom Subsection (a) applies shall file the notice required by Subsection (c) not later than:

(1) the beginning of a regular or special legislative session as to which the person is registered as a lobbyist under Chapter 305 and will communicate directly with a member of the legislative branch with respect to the measure, bill, or class of measures or bills; or

(2) the seventh business day after the day the person agrees to accept reimbursement or compensation to communicate directly with a member of the legislative branch with respect to the measure, bill, or class of measures or bills, if the person agrees to accept the reimbursement or compensation after the beginning of a legislative session.

(e) A member of the legislature who violates this section is subject to discipline by the house to which the member belongs, as provided by Section 11, Article III, Texas Constitution.

(f) In this section, "communicates directly with" and "member of the legislative branch" have the meanings assigned by Section 305.002.

## Sec. 572.054. REPRESENTATION BY FORMER OFFICER OR EMPLOYEE OF REGULATORY AGENCY RESTRICTED; CRIMINAL OFFENSE.

(a) A former member of the governing body or a former executive head of a regulatory agency may not make any communication to or appearance before an officer or employee of the agency in which the member or executive head served before the second anniversary of the date the member or executive head ceased to be a member of the governing body or the executive head of the agency if the communication or appearance is made:

(1) with the intent to influence; and

(2) on behalf of any person in connection with any matter on which the person seeks official action.

(b) A former state officer or employee of a regulatory agency who ceases service or employment with that agency on or after January 1, 1992, may not represent any person or receive compensation for services rendered on behalf of any person regarding a particular matter in which the former officer or employee participated during the period of state service or employment, either through personal involvement or because the case or proceeding was a matter within the officer's or employee's official responsibility.

(c) Subsection (b) applies only to:

(1) a state officer of a regulatory agency; or

(2) a state employee of a regulatory agency who is compensated, as of the last date of state employment, at or above the amount prescribed by the General Appropriations Act for step 1, salary group 17, of the position classification salary schedule, including an employee who is exempt from the state's position classification plan.

(d) Subsection (b) does not apply to a rulemaking proceeding that was concluded before the officer's or employee's service or employment ceased.

(e) Other law that restricts the representation of a person before a particular state agency by a former state officer or employee of that agency prevails over this section.

(f) An individual commits an offense if the individual violates this section. An offense under this subsection is a Class A misdemeanor.

(g) In this section, the comptroller and the secretary of state are not excluded from the definition of "regulatory agency."

(g-1) For purposes of this section, the Department of Information Resources is a regulatory agency.

(h) In this section:

(1) "Participated" means to have taken action as an officer or employee through decision, approval, disapproval, recommendation, giving advice, investigation, or similar action.

(2) "Particular matter" means a specific investigation, application, request for a ruling or determination, rulemaking proceeding, contract, claim, charge, accusation, arrest, or judicial or other proceeding.

## Sec. 572.055. CERTAIN SOLICITATIONS OF REGULATED BUSINESS ENTITIES PROHIBITED; CRIMINAL OFFENSE.

(a) An association or organization of employees of a regulatory agency may not solicit, accept, or agree to accept anything of value from a business entity regulated by that agency and from which the business entity must obtain a permit to operate that business in this state or from an individual directly or indirectly connected with that business entity.

(b) A business entity regulated by a regulatory agency and from which the business entity must obtain a permit to operate that business in this state or an individual directly or indirectly connected with that business entity may not offer, confer, or agree to confer on an association or organization of employees of that agency anything of value.

(c) This section does not apply to an agency regulating the operation or inspection of motor vehicles or an agency charged with enforcing the parks and wildlife laws of this state.

(d) A person commits an offense if the person intentionally or knowingly violates this section. An offense under this subsection is a Class A misdemeanor.

## Sec. 572.056. CONTRACTS BY STATE OFFICERS WITH GOVERNMENTAL ENTITIES; CRIMINAL OFFENSE.

(a) A state officer may not solicit or accept from a governmental entity a commission, fee, bonus, retainer, or rebate that is compensation for the officer's personal solicitation for the award of a contract for services or sale of goods to a governmental entity.

(b) This section does not apply to:

(1) a contract that is awarded by competitive bid as provided by law and that is not otherwise prohibited by law; or

(2) a court appointment.

(c) In this section, "governmental entity" means the state, a political subdivision of the state, or a governmental entity created under the Texas Constitution or a statute of this state.

(d) A state officer who violates this section commits an offense. An offense under this subsection is a Class A misdemeanor.

#### Sec. 572.057. CERTAIN LEASES PROHIBITED.

(a) Except as provided by Subsection (d), a member of the legislature, an executive or judicial officer elected in a statewide election, or a business entity in which the legislator or officer has a substantial interest may not lease any office space or other real property to the state, a state agency, the legislature or a legislative agency, the Supreme Court of Texas, the Court of Criminal Appeals, or a state judicial agency.

(b) A lease made in violation of Subsection (a) is void.

(c) This section does not apply to an individual who is an elected officer on June 16, 1989, for as long as the officer holds that office.

(d) A member of the legislature or a business entity in which the legislator has a substantial interest may donate the use of office space that the member or entity owns and that is located in the member's district to the house of the legislature in which the member serves to be used for the member's official business. Office space donated under this subsection is not a contribution for purposes of Title 15, Election Code. Acceptance of a donation of office space under this subsection is not subject to Section 301.032.

## Sec. 572.058. PRIVATE INTEREST IN MEASURE OR DECISION; DISCLOSURE; REMOVAL FROM OFFICE FOR VIOLATION.

(a) An elected or appointed officer, other than an officer subject to impeachment under Article XV, Section 2, of the Texas Constitution, who is a member of a board or commission having policy direction over a state agency and who has a personal or private interest in a measure, proposal, or decision pending before the board or commission shall publicly disclose the fact to the board or commission in a meeting called and held in compliance with Chapter 551. The officer may not vote or otherwise participate in the decision. The disclosure shall be entered in the minutes of the meeting.

(b) An individual who violates this section is subject to removal from office on the petition of the attorney general on the attorney general's own initiative or on the relation of a resident or of any other member of the board or commission. The suit must be brought in a district court of Travis County or of the county where the violation is alleged to have been committed.

(c) If the court or jury finds from a preponderance of the evidence that the defendant violated this section and that an ordinary prudent person would have known the individual's

conduct to be a violation of this section, the court shall enter judgment removing the defendant from office.

(d) A suit under this section must be brought before the second anniversary of the date the violation is alleged to have been committed, or the suit is barred.

(e) The remedy provided by this section is cumulative of other methods of removal from office provided by the Texas Constitution or a statute of this state.

(f) In this section, "personal or private interest" has the same meaning as is given to it under Article III, Section 22, of the Texas Constitution, governing the conduct of members of the legislature. For purposes of this section, an individual does not have a "personal or private interest" in a measure, proposal, or decision if the individual is engaged in a profession, trade, or occupation and the individual's interest is the same as all others similarly engaged in the profession, trade, or occupation.

### Sec. 572.059. INDEPENDENCE OF STATE AND LOCAL OFFICERS ACTING IN LEGISLATIVE CAPACITY.

(a) In this section, "legislative measure" includes:

(1) a bill, resolution, order, or other proposal to adopt, enact, amend, or repeal a statute, ordinance, rule, or policy of general application;

(2) a proposal to adopt, enact, amend, or repeal, or to grant a variance or other exception to, a zoning ordinance; or

(3) a proposed constitutional amendment or charter amendment subject to a vote of the electorate.

(b) For purposes of Subsection (a), a measure that is applicable to a class or subset of persons or matters that is defined in general terms without naming the particular persons or matters is a measure of general application.

(c) To protect the independence of state and local officers acting in a legislative capacity, a state or local officer, whether elected or appointed, including a member of the governing body of a school district or other political subdivision of this state, may not be subject to disciplinary action or a sanction, penalty, disability, or liability for:

(1) an action permitted by law that the officer takes in the officer's official capacity regarding a legislative measure;

(2) proposing, endorsing, or expressing support for or opposition to a legislative measure or taking any action permitted by law to support or oppose a legislative measure;

(3) the effect of a legislative measure or of a change in law proposed by a legislative measure on any person; or

(4) a breach of duty, in connection with the member's practice of or employment in a licensed or regulated profession or occupation, to disclose to any person information, or to obtain a waiver or consent from any person, regarding:

(A) the officer's actions relating to a legislative measure; or

(B) the substance, effects, or potential effects of a legislative measure.

#### Sec. 572.060. SOLICITATION OF OR RECOMMENDATIONS REGARDING CONTRIBUTIONS TO CHARITABLE ORGANIZATIONS AND GOVERNMENTAL ENTITIES.

(a) Unless otherwise prohibited by the Code of Judicial Conduct, a state officer or state employee may:

(1) solicit from any person a contribution to:

(A) an organization that:

(i) is exempt from income taxation under Section 501(a), Internal Revenue Code of 1986, by being listed under Section 501(c)(3) of that code;

(ii) does not attempt to influence legislation as a substantial part of the organization's activities; and

(iii) has not elected under Section 501(h), Internal Revenue Code of 1986, to have that subsection apply to the organization; or

(B) a governmental entity; or

(2) recommend to any person that the person make a contribution to an organization or entity described by Subdivision (1).

(b) A monetary contribution solicited or recommended as provided by Subsection (a) must:

(1) be paid or made directly to the charitable organization or governmental entity by the person making the contribution;

(2) be in the form of a check, money order, or similar instrument payable to the charitable organization or governmental entity; or

(3) be in the form of a deduction from a state employee's salary or wage payment under the state employee charitable campaign under Subchapter I, Chapter 659.

(c) A contribution solicited or recommended as provided by Subsection (a) that is not a monetary contribution must be delivered directly to the charitable organization or governmental entity by the person making the contribution.

(d) A contribution paid as provided by Subsection (b) or delivered as provided by Subsection (c) is not:

(1) a political contribution to, or political expenditure on behalf of, the state officer or state employee for purposes of Title 15, Election Code;

(2) an expenditure for purposes of Chapter 305; or

(3) a benefit to the state officer or state employee for purposes of Sections 36.08 and 36.09, Penal Code.

**Sec. 572.061. CERTAIN GRATUITIES AUTHORIZED**. This subchapter does not prohibit the acceptance of a gratuity that is accepted and reported in accordance with Section 11.0262, Parks and Wildlife Code.

Sec. 572.069. CERTAIN EMPLOYMENT FOR FORMER STATE OFFICER OR EMPLOYEE RESTRICTED. A former state officer or employee of a state agency who during the period of state service or employment participated on behalf of a state agency in a procurement or contract negotiation involving a person may not accept employment from that person before the second anniversary of the date the contract is signed or the procurement is terminated or withdrawn.



## 12

# FILING SCHEDULE FOR REPORTS

FORMS



#### TEXAS ETHICS COMMISSION 2024 FILING SCHEDULE FOR REPORTS DUE IN CONNECTION WITH ELECTIONS HELD ON UNIFORM ELECTION DATES

This is a filing schedule for reports to be filed in connection with elections held on uniform election dates in May and November. Examples of elections held on uniform election dates are elections for school board positions and city offices. The uniform election dates in 2024 are May 4 and November 5.

Candidates and officeholders must file semiannual reports (due on January 16, 2024, and July 15, 2024). In addition, a candidate who has an opponent on the ballot in an election held on a uniform election date must file two pre-election reports (unless the candidate has elected modified reporting).

The campaign treasurer of a political committee that is involved in an election held on a uniform election date must also file pre-election reports (unless the committee is a general-purpose political committee that files monthly or a specific-purpose political committee that files on the modified reporting schedule). This schedule sets out the due dates for pre-election reports in connection with elections on uniform election dates. Please consult the 2024 REGULAR FILING SCHEDULE FOR GENERAL-PURPOSE POLITICAL COMMITTEES (GPAC), COUNTY EXECUTIVE COMMITTEES (CEC), AND SPECIFIC-PURPOSE POLITICAL COMMITTEES (SPAC) for a complete listing of political committee deadlines.

Candidates for and officeholders in local offices regularly filled at the general election for state and county officers (the November election in even-numbered years) should use the 2024 FILING SCHEDULE FOR CANDIDATES AND OFFICEHOLDERS FILING WITH THE COUNTY CLERK OR ELECTIONS ADMINISTRATOR.

#### EXPLANATION OF THE FILING SCHEDULE CHART

<u>COLUMN I: REPORT DUE DATE</u> - This is the date by which the report must be filed. If the due date for a report falls on a Saturday, Sunday, or legal holiday, the report is due on the next regular business day. This schedule shows the extended deadline where applicable. A report transmitted to the Texas Ethics Commission over the Internet is considered timely filed if it is transmitted **by midnight**, **Central Time Zone**, **on the night of the filing deadline**. For most filing deadlines, a report filed on paper is considered timely filed if it is deposited with the U.S. Post Office or a common or contract carrier properly addressed with postage and handling charges prepaid, or hand-delivered to the filing authority by the filing deadline. **Pre-Election Reports:** A report due 30 days before an election and a report due 8 days before an election must be *received* by the appropriate filing authority no later than the report due date to be considered timely filed.

<u>COLUMN II: TYPE OF REPORT (WHO FILES)</u> - This column gives the report type and explains which reporting form to use and which filers are required to file the report.

<u>COLUMN III: BEGINNING DATE OF PERIOD COVERED</u> - This column sets out the beginning date of the time period covered by the report. Use the latest one of the applicable dates. The "date of campaign treasurer appointment" is the beginning date only for the *first* report filed after filing a campaign treasurer appointment. For officeholders recently appointed to an elective office, the beginning date for the first report will be the date the officeholder took office, provided that he or she was not already filing as an officeholder or candidate at the time of the appointment. (*NOTE:* If you are ever confused about the beginning date for a required report, remember this rule: **There should never be gaps between reporting periods and, generally, there should not be overlaps.**)

<u>COLUMN IV: ENDING DATE OF PERIOD COVERED</u> - This column sets out the ending date of the time period covered by the report. The report must include reportable activity occurring on the ending date.

Please consult the CAMPAIGN FINANCE GUIDE FOR CANDIDATES AND OFFICEHOLDERS WHO FILE WITH LOCAL FILING AUTHORITIES or the CAMPAIGN FINANCE GUIDE FOR POLITICAL COMMITTEES for further information.

| <u>COLUMN I</u><br>DUE DATE   | <u>COLUMN II</u><br>TYPE OF REPORT<br>(WHO FILES)  | <u>COLUMN III</u><br>BEGINNING DATE OF<br>PERIOD COVERED  | <u>COLUMN IV</u><br>ENDING DATE<br>OF PERIOD<br>COVERED |
|---|--|---|---|
| Tuesday,<br>January 16, 2024<br>Deadline is extended<br>because of holiday. | January semiannual<br>[FORM C/OH] (all local<br>candidates and officeholders,<br>except for officeholders who do<br>not have a campaign treasurer<br>appointment on file and who do<br>not exceed \$1,010 in contributions<br>or expenditures for the reporting<br>period)<br>[FORM GPAC] (all GPACs)<br>[FORM SPAC] (all SPACs) | July 1, 2023, <u>or</u><br>the date of campaign<br>treasurer appointment, <u>or</u><br>the day after the date the last<br>report ended. | December 31, 2023                                       |
| Tuesday,<br>January 16, 2024<br>Deadline is extended<br>because of holiday. | Annual report of unexpended<br>contributions<br>[FORM C/OH-UC]<br>(former candidates and former<br>officeholders who have filed a<br>final report and who retained<br>unexpended contributions or<br>assets purchased with<br>contributions)   | January 1, 2023, <u>or</u><br>the day after the date the<br>final report was filed.   | December 31, 2023                                       |

#### **REPORTS DUE BEFORE THE MAY 4, 2024, UNIFORM ELECTION**

| Thursday,                  | 30th day before the May 4, 2024,               | January 1, 2024, <u>or</u>          | March 25, 2024          |
|----------------------------|--|-------------------------------------|-------------------------|
| April 4, 2024              | uniform election                               |                                     |                         |
|                            |  | the date of campaign treasurer      |                         |
| NOTE: This report          | [FORM C/OH] (all local candidates              | appointment, <u>or</u>              |                         |
| must be <u>received</u> by | who have an opponent on the ballot             |                                     |                         |
| the appropriate filing     | in the May 4 election and who do               | the day after the date the last     |                         |
| authority no later         | not file on the modified reporting             | report ended.                       |                         |
| than April 4, 2024.        | schedule)                                      |                                     |                         |
|                            |  |                                     |                         |
|                            | [FORM GPAC] (all GPACs that                    |                                     |                         |
|                            | are involved in the May 4 election)            |                                     |                         |
|                            |  |                                     |                         |
|                            | [FORM SPAC] (all SPACs that do                 |                                     |                         |
|                            | not file on the modified reporting             |                                     |                         |
|                            | schedule and that supported or                 |                                     |                         |
|                            | opposed an opposed candidate or a              |                                     |                         |
|                            | measure in the May 4 election)                 |                                     |                         |
|                            |  |                                     |                         |
|                            |  |                                     |                         |
|                            |  |                                     |                         |
|                            |  |                                     |                         |
| -                          | ittee must file pre-election reports if the co |                                     | <b>U</b>                |
| reporting period. A politi | cal committee must file an 8-day pre-ele       | ection report if the committee file | d a 30-day pre-election |

reporting period. A political committee must file an 8-day pre-election report if the committee filed a 30-day pre-election report, even if there is no activity to report during the 8-day reporting period. The campaign treasurer of a political committee may be required to file 30-day and 8-day pre-election reports in connection with elections not listed on this schedule.

| <u>COLUMN I</u><br>DUE DATE   | <u>COLUMN II</u><br>TYPE OF REPORT<br>(WHO FILES)  | <u>COLUMN III</u><br>BEGINNING DATE OF<br>PERIOD COVERED   | <u>COLUMN IV</u><br>ENDING DATE<br>OF PERIOD<br>COVERED  |
|---|--|--|--|
| Friday,<br>April 26, 2024<br>NOTE: This report<br>must be <u>received</u> by<br>the appropriate<br>filing authority no<br>later than April 26,<br>2024. | <ul> <li>8th day before May 4, 2024, uniform election</li> <li>[FORM C/OH] (all local candidates who have an opponent on the ballot in the May 4 election and who do not file on the modified reporting schedule)</li> <li>[FORM GPAC] (all GPACs that filed a "30th Day Before Election Report" or that are involved in the May 4 election)</li> <li>[FORM SPAC] (all SPACs that do not file on the modified reporting schedule and that filed a "30th Day Before Election Report" or that supported or opposed an opposed candidate or a measure in the May 4 election)</li> </ul> | March 26, 2024, <u>or</u><br>the date of campaign<br>treasurer appointment, <u>or</u><br>the day after the date the last<br>report ended.  | April 24, 2024<br><b>NOTE:</b> Daily pre-<br>election reports of<br>contributions<br>accepted and direct<br>campaign<br>expenditures made<br>after April 24,<br>2024, may be<br>required. Please<br>consult the<br>Campaign Finance<br>Guide for further<br>information. |
| Monday,<br>July 15, 2024  | July semiannual<br>[FORM C/OH] (all local<br>candidates and officeholders,<br>except for officeholders who do<br>not have a campaign treasurer<br>appointment on file and who do<br>not exceed \$1,080 in contributions<br>or expenditures for the reporting<br>period)<br>[FORM GPAC] (all GPACs)<br>[FORM SPAC] (all SPACs)  | January 1, 2024, <u>or</u><br>the date of campaign<br>treasurer appointment, <u>or</u><br>the day after the date the last<br>report ended. | June 30, 2024  |

**NOTE**: A political committee must file pre-election reports if the committee is involved in the election during each preelection reporting period. A **political committee must file an 8-day pre-election report if the committee filed a 30-day pre-election report, even if there is no activity to report during the 8-day reporting period.** The campaign treasurer of a political committee may be required to file 30-day and 8-day pre-election reports in connection with elections not listed on this schedule. COLUMN I

#### **DUE DATE**

#### <u>COLUMN II</u> TYPE OF REPORT (WHO FILES)

#### COLUMN III

BEGINNING DATE OF PERIOD COVERED

#### COLUMN IV ENDING DATE OF PERIOD COVERED

#### **REPORTS DUE BEFORE THE NOVEMBER 5, 2024, UNIFORM ELECTION**

| Monday,<br>October 7, 2024   | 30th day before the November 5, 2024, uniform election   | July 1, 2024, <u>or</u>   | September 26, 2024   |
|--|--|---|--|
| Deadline is extended<br>because of weekend.<br>NOTE: This report<br>must be <u>received</u> by<br>the appropriate<br>filing authority no<br>later than October<br>7, 2024. | <ul> <li>[FORM C/OH] (all local candidates who have an opponent on the ballot in the November 5 election and who do not file on the modified reporting schedule)</li> <li>[FORM GPAC] (all GPACs that are involved in the November 5 election)</li> <li>[FORM SPAC] (all SPACs that do not file on the modified reporting schedule and that supported or opposed an opposed candidate or a measure in the November 5 election)</li> </ul>  | the date of campaign<br>treasurer appointment, <u>or</u><br>the day after the date the last<br>report ended.                                  |  |
| Monday,<br>October 28, 2024<br>NOTE: This report<br>must be <u>received</u> by<br>the appropriate<br>filing authority no<br>later than October<br>28, 2024.                | <ul> <li>8th day before the November 5, 2024, uniform election</li> <li>[FORM C/OH] (all local candidates who have an opponent on the ballot in the November 5 election and who do not file on the modified reporting schedule)</li> <li>[FORM GPAC] (all GPACs that filed a "30th Day Before Election Report" or that are involved in the November 5 election)</li> <li>[FORM SPAC] (all SPACs that do not file on the modified reporting schedule and that filed a "30th Day Before Election Report" or that supported or opposed an opposed candidate or a measure in the November 5 election)</li> </ul> | September 27, 2024, <u>or</u><br>the date of campaign<br>treasurer appointment, <u>or</u><br>the day after the date the last<br>report ended. | October 26, 2024<br><b>NOTE:</b> Daily pre-<br>election reports of<br>contributions<br>accepted and direct<br>campaign<br>expenditures made<br>after October 26,<br>2024, may be<br>required. Please<br>consult the<br>Campaign Finance<br>Guide for further<br>information. |
| election reporting period <b>pre-election report, eve</b>  | mittee must file pre-election reports if the <b>A political committee must file an 8-</b><br><b>n if there is no activity to report during</b><br>be required to file 30-day and 8-day pre-  | day pre-election report if the co<br>g the 8-day reporting period. The  | <b>mmittee filed a 30-day</b><br>campaign treasurer of a   |

this schedule.

| <u>COLUMN I</u><br>DUE DATE    | <u>COLUMN II</u><br>TYPE OF REPORT<br>(WHO FILES)  | <u>COLUMN III</u><br>BEGINNING DATE OF<br>PERIOD COVERED  | COLUMN IV<br>ENDING DATE<br>OF PERIOD<br>COVERED |
|--------------------------------|--|---|--|
| Wednesday,<br>January 15, 2025 | January semiannual<br>[FORM C/OH] (all local<br>candidates and officeholders,<br>except for officeholders who do<br>not have a campaign treasurer<br>appointment on file and who do<br>not exceed \$1,080 in contributions<br>or expenditures for the reporting<br>period)<br>[FORM GPAC] (all GPACs)<br>[FORM SPAC] (all SPACs) | July 1, 2024, <u>or</u><br>the date of campaign<br>treasurer appointment, <u>or</u><br>the day after the date the last<br>report ended. | December 31, 2024                                |
| Wednesday,<br>January 15, 2025 | Annual report of unexpended<br>contributions<br>[FORM C/OH-UC]<br>(former candidates and former<br>officeholders who have filed a<br>final report and who retained<br>unexpended contributions or<br>assets purchased with<br>contributions)   | January 1, 2024, <u>or</u><br>the day after the date the<br>final report was filed.   | December 31, 2024                                |

#### APPOINTMENT OF A CAMPAIGN TREASURER BY A CANDIDATE

1 Total pages filed: See CTA Instruction Guide for detailed instructions. MS / MRS / MR FIRST MI CANDIDATE **OFFICE USE ONLY** 2 NAME Filer ID # NICKNAME LAST SUFFIX Date Received ADDRESS / PO BOX; APT / SUITE #; CITY; STATE; ZIP CODE CANDIDATE 3 MAILING ADDRESS Date Hand-delivered or Postmarked AREA CODE PHONE NUMBER EXTENSION Receipt# Amount \$ CANDIDATE 4 PHONE ) ( Date Processed 5 OFFICE Date Imaged HELD (if any) OFFICE 6 SOUGHT (if known) MS/MRS/MR FIRST МІ NICKNAME LAST SUFFIX 7 CAMPAIGN TREASURER NAME STREET ADDRESS: APT / SUITE #; CITY; STATE: ZIP CODE 8 CAMPAIGN TREASURER STREET ADDRESS (residence or business) AREA CODE PHONE NUMBER EXTENSION 9 CAMPAIGN TREASURER PHONE ) **10** CANDIDATE SIGNATURE I am aware of the Nepotism Law, Chapter 573 of the Texas Government Code. I am aware of my responsibility to file timely reports as required by title 15 of the Election Code. I am aware of the restrictions in title 15 of the Election Code on contributions from corporations and labor organizations. Date Signed Signature of Candidate

GO TO PAGE 2

FORM CTA

#### CANDIDATE MODIFIED REPORTING DECLARATION

| 11 | CANDIDATE<br>NAME   |  |  |  |  |  |
|----|---|--|--|--|--|--|
| 12 | MODIFIED<br>REPORTING<br>DECLARATION  | COMPLETE THIS SECTION ONLY IF YOU ARE<br>CHOOSING MODIFIED REPORTING   |  |  |  |  |
|    |   | •• This declaration must be filed no later than the 30th day before the first election to which the declaration applies. ••  |  |  |  |  |
|    |   | •• The modified reporting option is valid for one election cycle only. ••<br>(An election cycle includes a primary election, a general election, and any related runoffs.)   |  |  |  |  |
|    |   | •• Candidates for the office of state chair of a political party may NOT choose modified reporting. ••   |  |  |  |  |
|    |   | I do not intend to accept more than \$1,080 in political contributions or<br>make more than \$1,080 in political expenditures (excluding filing<br>fees) in connection with any future election within the election<br>cycle. I understand that if either one of those limits is exceeded, I<br>will be required to file pre-election reports and, if necessary, a<br>runoff report. |  |  |  |  |
|    |   | Year of election(s) or election cycle to Signature of Candidate which declaration applies  |  |  |  |  |
|    |   | This appointment is effective on the date it is filed with the appropriate filing authority.   |  |  |  |  |
|    | TEC   | Filers may send this form to the TEC electronically at <u>treasappoint@ethics.state.tx.us</u><br>or mail to<br>Texas Ethics Commission<br>P.O. Box 12070<br>Austin TX 78711 2070   |  |  |  |  |
|    | Austin, TX 78711-2070<br>Non-TEC Filers must file this form with the local filing authority<br>DO NOT SEND TO TEC |  |  |  |  |  |
|    |   | For more information about where to file go to:<br>https://www.ethics.state.tx.us/filinginfo/QuickFileAReport.php  |  |  |  |  |

#### **AMENDMENT:** APPOINTMENT OF A CAMPAIGN TREASURER BY A CANDIDATE

| 1 | CANDIDATE |
|---|-----------|
|   | NAME      |

2 FILER ID#

**3** Total pages filed:

FORM ACTA

PG 1

| Use this form  | n for cha           |  | A Instruction Guing information o |                                 |                | ns.<br>nation previously                            | disclosed.       |
|--|---------------------|--|-----------------------------------|---------------------------------|----------------|---|------------------|
| 4 CANDIDATE<br>NAME  | NEW                 | MS / MRS / MR                                | FIRST                             |                                 | MI             | OFFIC   | E USE ONLY       |
|  |                     | NICKNAME                                     | LAST                              |                                 | SUFFIX         | Date Received                                       |                  |
| 5 CANDIDATE<br>MAILING<br>ADDRESS  | NEW                 | ADDRESS / PO BO                              | X; APT / SUITE #;                 | CITY;                           | STATE; ZIP COD | Date Hand-deliver                                   | ed or Postmarked |
| 6 CANDIDATE<br>PHONE   | NEW                 | AREA CODE                                    | PHONE NUMBER                      |                                 | EXTENSION      | Date Processed                                      |                  |
| THOME  |                     | ( )  |                                   |                                 |                | , , , , , , , , , , , , , , , , , , ,               |                  |
| 7 OFFICE HELD<br>(if any)  | NEW                 |  |                                   |                                 |                |   |                  |
| 8 OFFICE<br>SOUGHT<br>(if known)   | NEW                 |  |                                   |                                 |                |   |                  |
| 9 CAMPAIGN<br>TREASURER<br>NAME  | NEW                 | MS / MRS / MR                                | FIRST                             | MI                              | NICKNAME       | LAST  | SUFFIX           |
| 10 CAMPAIGN<br>TREASURER<br>STREET<br>ADDRESS<br>(residence or business) | NEW                 | STREET ADDRESS                               | (NO PO BOX PLEASE);               | APT / SUITE #;                  | СІТҮ;          | STA   | TE; ZIP CODE     |
| 11 CAMPAIGN<br>TREASURER<br>PHONE  | NEW                 | AREA CODE                                    | PHONE NUMBER                      |                                 | EXTENSION      |   |                  |
| 12 CANDIDATE<br>SIGNATURE  | l ai<br>the<br>I ai | m aware of r<br>Election Co<br>m aware of tl | my responsibil<br>de.             | ity to file ti<br>in title 15 c | mely reports   | e Texas Govern<br>s as required b<br>on Code on con | y title 15 of    |
|  | -                   | Sigr   | nature of Candidat                | e                               |                | Date Sigr   | ned              |
|  | •                   |  | GO TO                             | PAGE 2                          |                |   |                  |

#### **AMENDMENT:** CANDIDATE MODIFIED REPORTING DECLARATION

| 13 CANDIDATE             |   |
|--------------------------|---|
| NAME                     |   |
|                          |   |
| 14 MODIFIED              | NEW   |
| REPORTING<br>DECLARATION | COMPLETE THIS SECTION ONLY IF YOU ARE   |
| DECENTATION              | CHOOSING MODIFIED REPORTING   |
|                          |   |
|                          |   |
|                          | •• This declaration must be filed no later than the 30th day before   |
|                          | the first election to which the declaration applies. ••   |
|                          |   |
|                          |   |
|                          | •• The modified reporting option is valid for one election cycle only. ••   |
|                          | (An election cycle includes a primary election, a general election, and any related runoffs.)   |
|                          |   |
|                          |   |
|                          | <ul> <li>Candidates for the office of state chair of a political party<br/>may NOT choose modified reporting.</li> </ul>              |
|                          | may NOT choose mounted reporting.   |
|                          |   |
|                          | I do not intend to accept more than \$1,080 in political contributions  |
|                          | I do not intend to accept more than \$1,080 in political contributions or make more than \$1,080 in political expenditures (excluding |
|                          | filing fees) in connection with any future election within the election   |
|                          | cycle. I understand that if either one of those limits is exceeded, I   |
|                          | will be required to file pre-election reports and, if necessary, a  |
|                          | runoff report.  |
|                          |   |
|                          |   |
|                          |   |
|                          |   |
|                          |   |
|                          | Year of election(s) or election cycle to Signature of Candidate   |
|                          | Year of election(s) or election cycle to Signature of Candidate which declaration applies   |
|                          |   |
|                          |   |
|                          |   |
| This appoi               | ntment is effective on the date it is filed with the appropriate filing authority.  |
|                          |   |
| TEC F                    | lers may send this form to the TEC electronically at treasappoint@ethics.state.tx.us  |
|                          | or mail to  |
|                          | Texas Ethics Commission   |
|                          | P.O. Box 12070  |
|                          | Austin, TX 78711-2070   |
|                          |   |
|                          | Non-TEC Filers must file this form with the local filing authority  |
|                          | DO NOT SEND TO TEC  |
|                          | For more information about where to file go to:   |
|                          | https://www.ethics.state.tx.us/filinginfo/QuickFileAReport.php  |
|                          |   |

#### CANDIDATE / OFFICEHOLDER CAMPAIGN FINANCE REPORT

#### FORM C/OH COVER SHEET PG 1

| The C/OH Instruction G  | uide explains how   | to complete this form.                                     | 1 Filer ID (Ethics Commission Filers)       | 2 Total pages filed:   |  |  |
|---|---|--|---|--|--|--|
| 3 CANDIDATE /<br>OFFICEHOLDER                                 | MS / MRS / MR   | FIRST  | MI  | OFFICE USE ONLY  |  |  |
| NAME  | NICKNAME  | LAST   | SUFFIX                                      | Date Received  |  |  |
| 4 CANDIDATE /<br>OFFICEHOLDER<br>MAILING<br>ADDRESS           | ADDRESS / PO BOX;   | APT / SUITE #; (   | CITY; STATE; ZIP CODE                       |  |  |  |
| Change of Address   |   |  |   |  |  |  |
| 5 CANDIDATE/<br>OFFICEHOLDER<br>PHONE                         | AREA CODE   | PHONE NUMBER   | EXTENSION                                   | Date Hand-delivered or Date Postmarked Receipt # Amount \$   |  |  |
| 6 CAMPAIGN<br>TREASURER                                       | MS / MRS / MR   | FIRST  | MI  |  |  |  |
| NAME  | NICKNAME  | LAST   | SUFFIX                                      | Date Processed   |  |  |
|   |   |  |   | Date Imaged  |  |  |
| 7 CAMPAIGN<br>TREASURER<br>ADDRESS<br>(Residence or Business) | STREET ADDRESS (  | NO PO BOX PLEASE); APT / S                                 | SUITE #; CITY;                              | STATE; ZIP CODE  |  |  |
|   | AREA CODE   |  |   |  |  |  |
| 8 CAMPAIGN<br>TREASURER<br>PHONE                              | ( )   | PHONE NUMBER   | EXTENSION                                   |  |  |  |
| 9 REPORT TYPE   | January 15  | 30th day before e  | election Runoff                             | 15th day after campaign<br>treasurer appointment<br>(Officeholder Only)  |  |  |
|   | July 15   | 8th day before ele   | ection Exceeded Modified<br>Reporting Limit | Final Report (Attach C/OH - FR)  |  |  |
| 10 PERIOD<br>COVERED  | Month /   | Day Year   | Month                                       | Day Year   |  |  |
| 11 ELECTION   | ELECTION DA   | TE Primary   | ELECTION TYPE                               |  |  |  |
| 12 OFFICE   | OFFICE HELD (if any)  | '  | 13 OFFICE SOUGHT (if known                  | )  |  |  |
| 14 NOTICE FROM<br>POLITICAL<br>COMMITTEE(S)                   | THE CANDIDATE / OFFIC<br>CONSENT. CANDIDATES                        | CEHOLDER. THESE EXPENDITURE<br>AND OFFICEHOLDERS ARE REQUI | S MAY HAVE BEEN MADE WITHOUT THE CAN        | MADE BY POLITICAL COMMITTEES TO SUPPORT<br>IDIDATE'S OR OFFICEHOLDER'S KNOWLEDGE OR<br>THEY RECEIVE NOTICE OF SUCH EXPENDITURES. |  |  |
|   | COMMITTEE TYPE  | COMMITTEE NAME   |   |  |  |  |
| Additional Pages  | GENERAL   | COMMITTEE ADDRESS  |   |  |  |  |
|   | COMMITTEE CAMPAIGN TREASURER NAME COMMITTEE CAMPAIGN TREASURER NAME |  |   |  |  |  |
|   |   | COMMITTEE CAMPAIGN TR                                      | EASURER ADDRESS                             |  |  |  |
| GO TO PAGE 2  |   |  |   |  |  |  |

#### CANDIDATE / OFFICEHOLDER CAMPAIGN FINANCE REPORT

#### FORM C/OH COVER SHEET PG 2

| 15 C/OH NAME                   |  | <b>16</b> Filer ID (Ethics Commission Filers) |  |  |  |  |
|--------------------------------|--|---|--|--|--|--|
| 17 CONTRIBUTION<br>TOTALS      | 1. TOTAL UNITEMIZED POLITICAL CONTRIBUTIONS (OTHER THAN<br>PLEDGES, LOANS, OR GUARANTEES OF LOANS, OR<br>CONTRIBUTIONS MADE ELECTRONICALLY)    | \$  |  |  |  |  |
|                                | 2. TOTAL POLITICAL CONTRIBUTIONS<br>(OTHER THAN PLEDGES, LOANS, OR GUARANTEES OF LOANS)  | \$  |  |  |  |  |
| EXPENDITURE<br>TOTALS          | 3. TOTAL UNITEMIZED POLITICAL EXPENDITURE.   | \$  |  |  |  |  |
|                                | 4. TOTAL POLITICAL EXPENDITURES  | \$  |  |  |  |  |
| CONTRIBUTION<br>BALANCE        | 5. TOTAL POLITICAL CONTRIBUTIONS MAINTAINED AS OF THE LAS<br>OF REPORTING PERIOD   | T DAY \$                                      |  |  |  |  |
| OUTSTANDING<br>LOAN TOTALS     | 6. TOTAL PRINCIPAL AMOUNT OF ALL OUTSTANDING LOANS AS OF<br>LAST DAY OF THE REPORTING PERIOD   | THE \$  |  |  |  |  |
|                                | swear, or affirm, under penalty of perjury, that the accompanying report is true<br>quired to be reported by me under Title 15, Election Code. | and correct and includes all information      |  |  |  |  |
|                                | Signature of Candidate or Officeholder   |   |  |  |  |  |
|                                | Please complete either option below  | <i>r</i> :                                    |  |  |  |  |
| (1) Affidavit                  |  |   |  |  |  |  |
| NOTARY STAMP/SEA               | L  |   |  |  |  |  |
| Sworn to and subscribed        | before me by this the  | day of,                                       |  |  |  |  |
| 20, to certify                 | which, witness my hand and seal of office.   |   |  |  |  |  |
| Signature of officer administe | Printed name of officer administering oath   | Title of officer administering oath           |  |  |  |  |
|                                | OR   |   |  |  |  |  |
| (2) Unsworn Declarati          | on   |   |  |  |  |  |
| My name is                     | , and my date of birth is  |   |  |  |  |  |
|                                |  | ,,,,  |  |  |  |  |
|                                |  | tate) (zip code) (country)                    |  |  |  |  |
| Executed in                    | County, State of, on the day of(month  | ), 20 <u>(year)</u> .                         |  |  |  |  |
|                                | Signature of Candio  | ate/Officeholder (Declarant)                  |  |  |  |  |

#### SUBTOTALS - C/OH

#### FORM C/OH COVER SHEET PG 3

| 19  | FILER NAME   | mmission Filers) |    |  |  |  |
|-----|--|------------------|----|--|--|--|
| 21  | 21 SCHEDULE SUBTOTALS<br>NAME OF SCHEDULE                                |                  |    |  |  |  |
| 1.  | SCHEDULE A1: MONETARY POLITICAL CONTRIBUTIONS                            |                  | \$ |  |  |  |
| 2.  | SCHEDULE A2: NON-MONETARY (IN-KIND) POLITICAL CONTRIBUTIONS              |                  | \$ |  |  |  |
| 3.  | SCHEDULE B: PLEDGED CONTRIBUTIONS  |                  | \$ |  |  |  |
| 4.  | SCHEDULE E: LOANS  |                  | \$ |  |  |  |
| 5.  | SCHEDULE F1: POLITICAL EXPENDITURES MADE FROM POLITICAL CC               | NTRIBUTIONS      | \$ |  |  |  |
| 6.  | SCHEDULE F2: UNPAID INCURRED OBLIGATIONS                                 |                  | \$ |  |  |  |
| 7.  | SCHEDULE F3: PURCHASE OF INVESTMENTS MADE FROM POLITICAL                 | CONTRIBUTIONS    | \$ |  |  |  |
| 8.  | SCHEDULE F4: EXPENDITURES MADE BY CREDIT CARD                            |                  | \$ |  |  |  |
| 9.  | SCHEDULE G: POLITICAL EXPENDITURES MADE FROM PERSONAL FU                 | NDS              | \$ |  |  |  |
| 10. | SCHEDULE H: PAYMENT MADE FROM POLITICAL CONTRIBUTIONS TO A               | BUSINESS OF C/OH | \$ |  |  |  |
| 11. | SCHEDULE I: NON-POLITICAL EXPENDITURES MADE FROM POLITICAL CO            | NTRIBUTIONS      | \$ |  |  |  |
| 12. | SCHEDULE K: INTEREST, CREDITS, GAINS, REFUNDS, AND CONTRIBUT<br>TO FILER | TIONS RETURNED   | \$ |  |  |  |
|     |  |                  |    |  |  |  |
|     |  |                  |    |  |  |  |

#### MONETARY POLITICAL CONTRIBUTIONS

SCHEDULE A1

If the requested information is not applicable, DO NOT include this page in the report.

|   | The             | Instruction Guide explains how to complete this                                    | form.                   | 1 Total pages Schedule A1:                   |
|---|-----------------|--|-------------------------|--|
| 2 | FILER NAME      |  |                         | <b>3</b> Filer ID (Ethics Commission Filers) |
| 4 | Date            | 5 Full name of contributor Out-of-state PAC  | C (ID#:)                | 7 Amount of contribution (\$)                |
|   |                 | <b>6</b> Contributor address; City;  | State; Zip Code         |  |
| 8 | Principal occu  | oation / Job title (See Instructions)  | 9 Employer (See Instruc | tions)                                       |
|   | Date            | Full name of contributor   Out-of-state PAC  | (ID#:)                  | Amount of contribution (\$)                  |
|   |                 | Contributor address; City;   | State; Zip Code         |  |
|   | Principal occup | ation / Job title (See Instructions)   | Employer (See Instruc   | tions)                                       |
|   | Date            | Full name of contributor   Out-of-state PAC  | C (ID#:)                | Amount of contribution (\$)                  |
|   |                 | Contributor address; City;   | State; Zip Code         |  |
|   | Principal occup | ation / Job title (See Instructions)   | Employer (See Instruc   | tions)                                       |
|   | Date            | Full name of contributor out-of-state PAC  | (ID#:)                  | Amount of contribution (\$)                  |
|   |                 | Contributor address; City;   | State; Zip Code         |  |
|   | Principal occup | ation / Job title (See Instructions)   | Employer (See Instruc   | tions)                                       |
|   |                 |  |                         |  |
|   |                 | ATTACH ADDITIONAL COPIES (<br>If contributor is out-of-state PAC, please see Instr |                         |  |

## NON-MONETARY (IN-KIND) POLITICAL CONTRIBUTIONS

SCHEDULE A2

| If the requested in | formation is not | applicable, DC | O NOT include the | nis page in the report. |
|---------------------|------------------|----------------|-------------------|-------------------------|
|                     |                  |                |                   |                         |

| т                      | he Instruction Guide explains how to complete this forr                                   | n.   | 1 Total pages Schedu                  | ule A2:                             |  |
|------------------------|---|--|---------------------------------------|-------------------------------------|--|
| 2 FILER NAM            | 1E  |  | 3 Filer ID (Ethics Commission Filers) |                                     |  |
| 4 TOTAL C              | OF UNITEMIZED IN-KIND POLITICAL CONTRIE   | BUTIONS  | \$                                    |                                     |  |
| 5 Date                 | 6 Full name of contributor out-of-state PAC (ID#:   | )  | 8 Amount of<br>Contribution \$        | 9 In-kind contribution description  |  |
|                        | <b>7</b> Contributor address; City; State;  | Zip Code   | Check if travel outsid                | de of Texas. Complete Schedule T.   |  |
| <b>10</b> Principal oc | cupation / Job title (FOR NON-JUDICIAL)(See Instructions)                                 | 11 Employe   | er (FOR NON-JUDICIA                   | AL)(See Instructions)               |  |
| 12 Contributor's       | s principal occupation (FOR JUDICIAL)   | 13 Contribu  | utor's job title (FOR JU              | DICIAL) (See Instructions)          |  |
| 14 Contributor's       | s employer/law firm (FOR JUDICIAL)  | 15 Law firm  | n of contributor's spous              | se (if any) (FOR JUDICIAL)          |  |
| 16 If contributo       | r is a child, law firm of parent(s) (if any) (FOR JUDICIAL)                               |  |                                       |                                     |  |
| Date                   | Full name of contributor out-of-state PAC (ID#:   | )  | Amount of<br>Contribution \$          | In-kind contribution<br>description |  |
|                        | Contributor address; City; State;   | Zip Code   | Check if travel outsic                | de of Texas. Complete Schedule T.   |  |
| Principal oc           | cupation / Job title (FOR NON-JUDICIAL) (See Instructions)                                | Employe  | er (FOR NON-JUDICIA                   | AL)(See Instructions)               |  |
| Contributor'           | s principal occupation (FOR JUDICIAL)   | Contribu   | utor's job title (FOR JU              | DICIAL) (See Instructions)          |  |
| Contributor'           | s employer/law firm (FOR JUDICIAL)  | Law firm of contributor's spouse (if any) (FOR JUDICIAL) |                                       |                                     |  |
| If contributo          | r is a child, law firm of parent(s) (if any) (FOR JUDICIAL)                               |  |                                       |                                     |  |
|                        |   |  |                                       |                                     |  |
|                        |   |  |                                       |                                     |  |
|                        |   |  |                                       |                                     |  |
|                        |   |  |                                       |                                     |  |
|                        |   |  |                                       |                                     |  |
|                        | ATTACH ADDITIONAL COPIES OF T<br>If contributor is out-of-state PAC, please see Instructi |  | -                                     | g requirements.                     |  |

#### PLEDGED CONTRIBUTIONS

#### SCHEDULE **B**

If the requested information is not applicable, DO NOT include this page in the report.

|          | The              | Instruction Guide explains how to complete thi                              | s form.          | 1 Total pages Sched      | ule B:   |
|----------|------------------|---|------------------|--------------------------|--|
| 2        | FILER NAME       |   |                  | 3 Filer ID (Ethics C     | commission Filers)                                   |
| 4        | TOTAL OF         | UNITEMIZED PLEDGES  |                  | \$                       |  |
| 5        | Date             | 6 Full name of pledgor out-of-state PAC (ID#:                               |                  | 8 Amount<br>of Pledge \$ | 9 In-kind contribution<br>description                |
|          |                  | 7 Pledgor address; City; S  | ·<br>            | <br> <br>                |  |
|          |                  |   |                  | Check if travel outs     | ide of Texas. Complete Schedule T.                   |
| 10       | ) Principal occu | pation / Job title (See Instructions)                                       | 11 Employer (See | Instructions)            |  |
|          | Date             | Full name of pledgor out-of-state PAC (ID#:                                 | )                | Amount<br>of Pledge \$   | In-kind contribution<br>description                  |
|          |                  | Pledgor address; City; S  | tate; Zip Code   |                          | <br>   |
|          |                  |   |                  | Check if travel outsi    | I.<br>ide of Texas. Complete Schedule T.             |
|          | Principal occup  | ation / Job title (See Instructions)  | Employer (See    | Instructions)            |  |
|          | Date             | Full name of pledgor out-of-state PAC (ID#:                                 |                  | Amount of<br>Pledge \$   | In-kind contribution<br>description                  |
|          |                  | Pledgor address; City; S  | tate; Zip Code   |                          |  |
|          |                  |   |                  | Check if travel outsi    | ide of Texas. Complete Schedule T.                   |
|          | Principal occu   | pation / Job title (See Instructions)                                       | Employer (See    | Instructions)            |  |
|          | Date             | Full name of pledgor out-of-state PAC (ID#:                                 | )                | Amount of<br>Pledge \$   | In-kind contribution description                     |
|          |                  | Pledgor address; City; Stat   | e; Zip Code      | •                        | <br> <br>  |
|          |                  |   |                  |                          | l <sub>.</sub><br>ide of Texas. Complete Schedule T. |
|          | Principal occur  | pation / Job title (See Instructions)                                       | Employer (See    |                          |  |
|          |                  |   |                  | ,                        |  |
|          |                  |   |                  |                          |  |
|          |                  |   |                  |                          |  |
| <u> </u> |                  |   |                  |                          |  |
|          | lf               | ATTACH ADDITIONAL COPIES<br>contributor is out-of-state PAC, please see Ins |                  | -                        | requirements.  |

| 5 |
|---|
|   |

#### SCHEDULE E

If the requested information is not applicable, DO NOT include this page in the report.

| The  | Instruction Guide explains how to compl                            | lete this form.                                     | 1 Total pages Schedule E:                  |
|--|--|---|--|
| 2 FILER NAME                               |  |   | 3 Filer ID (Ethics Commission Filers)      |
| 4 TOTAL OF UN                              | NITEMIZED LOANS  |   | \$   |
| 5 Date of loan                             | 7 Name of lender 🗌 out-of-state F                                  | PAC (ID#: )   | 9 Loan Amount (\$)                         |
| 6 Is lender<br>a financial<br>Institution? | 8 Lender address; City;  | State; Zip Code                                     | 10 Interest rate                           |
| Y N  |  |   | <b>11</b> Maturity date                    |
| 12 Principal occupati                      | on / Job title (See Instructions)                                  | 13 Employer (See Instructions)                      |  |
| <b>14</b> Description of Col               | lateral  | 15<br>Check if personal fun<br>account (See Instruc | ds were deposited into political<br>tions) |
| 16 GUARANTOR<br>INFORMATION                | <b>17</b> Name of guarantor  |   | <b>19</b> Amount Guaranteed (\$)           |
|  | <b>18</b> Guarantor address; City;                                 | State; Zip Code                                     |  |
| not applicable                             |  |   |  |
| 20 Principal Occupa                        | tion (See Instructions)  | 21 Employer (See Instructions)                      |  |
| Date of loan                               | Name of lender Out-of-state  | PAC (ID#:)  | Loan Amount (\$)                           |
| ls lender<br>a financial<br>Institution?   | Lender address; City;  | State; Zip Code                                     | Interest rate                              |
| Y N  |  |   | Maturity date                              |
| Principal occupati                         | on / Job title (See Instructions)                                  | Employer (See Instructions)                         | 1  |
| Description of Col                         | lateral  |   | ds were deposited into political           |
| none                                       | 1  | account (See Instruc                                |  |
| GUARANTOR<br>INFORMATION                   | Name of guarantor  |   | Amount Guaranteed (\$)                     |
|  | Guarantor address; City;   | State; Zip Code                                     |  |
| not applicable                             |  |   |  |
| Principal Occupat                          | ion (See Instructions)   | Employer (See Instructions)                         | ·  |
| If I                                       | ATTACH ADDITIONAL COP<br>ender is out-of-state PAC, please see Ins | IES OF THIS SCHEDULE AS NE                          |  |

#### POLITICAL EXPENDITURES MADE FROM POLITICAL CONTRIBUTIONS

#### SCHEDULE F1

If the requested information is not applicable, **DO NOT include this page in the report.** 

#### **EXPENDITURE CATEGORIES FOR BOX 8(a)**

| Event Ex |
|----------|
| Fees     |
| Food/Be  |
| Gift/Awa |
| Legal Se |
| The l    |
|          |

rent Expense ies od/Beverage Expense ft/Awards/Memorials Expense igal Services Loan Repayment/Reimbursement Office Overhead/Rental Expense Polling Expense Printing Expense Salaries/Wages/Contract Labor Solicitation/Fundraising Expense Transportation Equipment & Related Expense Travel In District Travel Out Of District Other (enter a category not listed above)

| The Instruction Guide explains how to complete this for |
|---|
|---|

| <b>1</b> Total pages Schedule F1:                               | 2 FILER NAME   |                 | <b>3</b> Filer ID (Ethics Commission Filers) |
|---|--|-----------------|--|
| 4 Date  | 5 Payee name   |                 | I  |
| <b>6</b> Amount (\$)  | <b>7</b> Payee address;  | City;           | State; Zip Code                              |
| 8<br>PURPOSE<br>OF<br>EXPENDITURE                               | (a) Category (See Categories listed at the top of this schedule) | (b) Description |  |
|   | (C) Check if travel outside of Texas. Complete Schedule T.       | Check if Austi  | in, TX, officeholder living expense          |
| 9 Complete <u>ONLY</u> if direct<br>expenditure to benefit C/OI | Candidate / Officeholder name                                    | Office sought   | Office held                                  |
| Date  | Payee name   |                 |  |
| Amount (\$)   | Payee address;   | City;           | State; Zip Code                              |
| PURPOSE<br>OF<br>EXPENDITURE                                    | Category (See Categories listed at the top of this schedule)     | Description     |  |
|   | Check if travel outside of Texas. Complete Schedule T.           | Check if Austi  | in, TX, officeholder living expense          |
| Complete <u>ONLY</u> if direct<br>expenditure to benefit C/OF   | Candidate / Officeholder name                                    | Office sought   | Office held                                  |
| Date  | Payee name   |                 |  |
| Amount (\$)   | Payee address;   | City;           | State; Zip Code                              |
| PURPOSE<br>OF<br>EXPENDITURE                                    | Category (See Categories listed at the top of this schedule)     | Description     |  |
|   | Check if travel outside of Texas. Complete Schedule T.           | Check if Austi  | n, TX, officeholder living expense           |
| Complete <u>ONLY</u> if direct<br>expenditure to benefit C/OF   |  | Office sought   | Office held                                  |
|   | ATTACH ADDITIONAL COPIES OF THIS                                 | SCHEDULE AS NEE | EDED   |

#### **UNPAID INCURRED OBLIGATIONS**

#### SCHEDULE F2

If the requested information is not applicable, DO NOT include this page in the report.

| EXPENDITURE CATEGORIES FOR BOX 10(a)   |                |                            |   |              |   |   |                  |                    |
|--|----------------|----------------------------|---|--------------|---|---|------------------|--------------------|
| Advertising Expense<br>Accounting/Banking<br>Consulting Expense<br>Contributions/Donations Made B<br>Candidate/Officeholder/Politica |                |                            | ees     Office Overhead/Rental Expense       ood/Beverage Expense     Polling Expense       ift/Awards/Memorials Expense     Printing Expense |              |   | Solicitation/Fundraising Expense<br>Transportation Equipment & Related Expense<br>Travel In District<br>Travel Out Of District<br>Other (enter a category not listed above) |                  |                    |
|  |                | The Instruction            | Guide explain   | s how to c   | omplete this form.  |   |                  |                    |
| <b>1</b> Total pages Schedule F2:  | 2 FILER        | NAME                       |   |              |   | 3 Filer ID  | ) (Ethics Co     | mmission Filers)   |
| 4 TOTAL OF UNITER  | MIZED UN       | IPAID INCURF               |   | GATION       | S   | \$  |                  |                    |
| 5 Date   | 6 Payee        | name                       |   |              |   |   |                  |                    |
| <b>7</b> Amount (\$)   | <b>8</b> Payee | address;                   |   |              | City;   |   | State;           | Zip Code           |
| 9 TYPE OF<br>EXPENDITURE   |                | Political                  |   | Non-Po       | itical  |   |                  |                    |
| 10<br>PURPOSE<br>OF<br>EXPENDITURE   | (a) Catego     | ry (See Categories liste   | d at the top of this  | schedule)    | (b) Description   |   |                  |                    |
|  | (C)            | Check if travel outside of | Texas. Complete S   | chedule T.   | Check if Aus  | stin, TX, officel   | nolder living ex | pense              |
| <b>11</b> Complete <u>ONLY</u> if direct<br>expenditure to benefit C/O   |                | didate / Officeholo        | ler name  | O            | ffice sought  |   | Office held      | 3                  |
| Date   | Payee          | name                       |   |              |   |   |                  |                    |
| Amount (\$)  | Payee          | address;                   |   |              | City;   |   | State;           | Zip Code           |
| TYPE OF<br>EXPENDITURE   |                | Political                  |   | Non-Po       | litical   |   |                  |                    |
| PURPOSE<br>OF<br>EXPENDITURE   | Catego         | ry (See Categories liste   | d at the top of this  | schedule)    | Description   |   |                  |                    |
|  |                | Check if travel outside o  | f Texas. Complete   | Schedule T.  | Check if Au   | ustin, TX, office   | eholder living e | expense            |
| Complete <u>ONLY</u> if direct Candidate / Officeholder name Office sought Office held expenditure to benefit C/OH                   |                |                            |   |              |   |   |                  |                    |
|  |                |                            |   |              |   |   |                  |                    |
|  |                |                            |   |              | CHEDULE AS NE   | EDED  |                  |                    |
| Forms provided by Texas Ethi   | cs Commissi    | on                         | www.ethics  | .state.tx.us | i de la companya de l |   |                  | Revised 11/15/2022 |

#### PURCHASE OF INVESTMENTS MADE FROM POLITICAL CONTRIBUTIONS

SCHEDULE F3

If the requested information is not applicable, **DO NOT include this page in the report.** 

| Tr           | ne Instruction Guide explains how to complete this form.   | <b>1</b> Total pages Schedule F3:     |  |  |  |
|--------------|--|---------------------------------------|--|--|--|
| 2 FILER NAME |  | 3 Filer ID (Ethics Commission Filers) |  |  |  |
| 4 Date       | <b>5</b> Name of person from whom investment is purchased  |                                       |  |  |  |
|              | 6 Address of person from whom investment is purchased; Cit | y; State; Zip Code                    |  |  |  |
|              | 7 Description of investment                                |                                       |  |  |  |
|              | 8 Amount of investment (\$)                                |                                       |  |  |  |
| Date         | Name of person from whom investment is purchased           |                                       |  |  |  |
|              | Address of person from whom investment is purchased; City  | y; State; Zip Code                    |  |  |  |
|              | Description of investment                                  |                                       |  |  |  |
|              | Amount of investment (\$)                                  |                                       |  |  |  |
|              |  |                                       |  |  |  |
|              | ATTACH ADDITIONAL COPIES OF THIS SCHEDULE                  | ASNEEDED                              |  |  |  |

Forms provided by Texas Ethics Commission

| EXPENDITU   | RES M        | ADE BY CRED  | DIT CAF   | RD                           | SCHE   | DULE <b>F4</b>        |  |  |  |
|---|--------------|--|---|------------------------------|--|-----------------------|--|--|--|
| If the requested inform   | mation is no | ot applicable, <b>DO NOT</b> i   | include this  | page in the rep              | port.  |                       |  |  |  |
| EXPENDITURE CATEGORIES FOR BOX 10(a)  |              |  |   |                              |  |                       |  |  |  |
| Advertising Expense<br>Accounting/Banking<br>Consulting Expense<br>Contributions/Donations Made E<br>Candidate/Officeholder/Politic |              | Event Expense<br>Fees<br>Food/Beverage Expense<br>Gift/Awards/Memorials Expense<br>Legal Services<br>The Instruction Guide exp | Office Over<br>Polling Exp<br>Printing Exp<br>Salaries/Wa | oense<br>ages/Contract Labor | Solicitation/Fundraisi<br>Transportation Equipr<br>Travel In District<br>Travel Out Of District<br>Other (enter a catego | nent & Related Expens |  |  |  |
| <b>1</b> Total pages Schedule F4:   | 2 FILER      | NAME   |   |                              | 3 Filer ID (Ethics 0   | Commission Filers)    |  |  |  |
| 4 TOTAL OF UNITEM   | IIZED EXP    | ENDITURES CHARGE   | EDTOACR   | EDIT CARD                    | \$   |                       |  |  |  |
| 5 Date  | 6 Payee      | name   |   |                              |  |                       |  |  |  |
| 7 Amount (\$)   | 8 Payee      | address;   |   | City;                        | State;   | Zip Code              |  |  |  |
| 9 TYPE OF<br>EXPENDITURE  | F            | Political  | Non-Pol   | itical                       |  |                       |  |  |  |
| 10<br>PURPOSE<br>OF<br>EXPENDITURE  | (a) Categor  | / (See Categories listed at the top of   | this schedule)  | (b) Description              |  |                       |  |  |  |
|   | (c)          | Check if travel outside of Texas. Comp   | lete Schedule T.  | Check if Au                  | ustin, TX, officeholder living   | expense               |  |  |  |
| <b>11</b><br>Complete <u>ONLY</u> if direct<br>expenditure to benefit C/OH  | Can          | didate / Officeholder name   | Of  | fice sought                  | Office h   | əld                   |  |  |  |
| Date  | Payee        | name   |   |                              |  |                       |  |  |  |
| Amount (\$)   | Payee        | address;   |   | City;                        | State;   | Zip Code              |  |  |  |
| TYPE OF<br>EXPENDITURE  | F            | Political  | Non-Po  | litical                      |  |                       |  |  |  |
| PURPOSE<br>OF<br>EXPENDITURE  | Categor      | y (See Categories listed at the top of   | f this schedule)  | Description                  |  |                       |  |  |  |
|   |              | Check if travel outside of Texas. Comp   | blete Schedule T.   | Check if A                   | ustin, TX, officeholder living   | g expense             |  |  |  |
| Complete <u>ONLY</u> if direct<br>expenditure to benefit C/OH   | Can          | didate / Officeholder name   | O1  | ffice sought                 | Office h   | eld                   |  |  |  |
|   |              |  |   |                              |  |                       |  |  |  |
|   | ΑΤΤΑΟ        | H ADDITIONAL COPIES  | S OF THIS S   | CHEDULE AS NE                | EDED   |                       |  |  |  |

#### POLITICAL EXPENDITURES MADE FROM PERSONAL FUNDS

#### SCHEDULE ${f G}$

If the requested information is not applicable, **DO NOT include this page in the report.** 

| EXPENDITURE CATEGORIES FOR BOX 8(a)   |   |   |          |                                    |   |                    |
|---|---|---|----------|------------------------------------|---|--------------------|
| Advertising Expense<br>Accounting/Banking<br>Consulting Expense<br>Contributions/Donations Made By<br>Candidate/Officeholder/Political Committee<br>Credit Card Payment |   | Fees     Office Overhead       Food/Beverage Expense     Polling Expense       Gift/Awards/Memorials Expense     Printing Expense |          | kpense<br>/ages/Contract Labor     | Solicitation/Fundraising Expense<br>Transportation Equipment & Related Expense<br>Travel In District<br>Travel Out Of District<br>Other (enter a category not listed above) |                    |
| <b>1</b> Total pages Schedule G:  | 2 FILER NAME  |   |          |                                    | 3 Filer ID (Ethics  | Commission Filers) |
| 4 Date  | 5 Payee name  |   |          |                                    |   |                    |
| 6 Amount (\$)<br>Reimbursement from<br>political contributions<br>intended  | 7 Payee address; City; State  |   |          |                                    |   | Zip Code           |
| 8<br>PURPOSE<br>OF<br>EXPENDITURE   | (a) Category (See Categories listed at the top of this schedule) (b) Description                        |   |          |                                    |   |                    |
|   | (c)   | Check if travel outside of Texas. Complete Sch  | edule T. | Check if Austir                    | n, TX, officeholder living ex   | pense              |
| 9<br>Complete <u>ONLY</u> if direct<br>expenditure to benefit C/OH  | Candid  | ate / Officeholder name   |          | Office sought                      |   | Office held        |
| Date  | Payee nar   | ne  |          |                                    |   |                    |
| Amount (\$)   | Payee ad  | dress;  |          | City;                              | State;  | Zip Code           |
| Reimbursement from<br>political contributions<br>intended   |   |   |          |                                    |   |                    |
| PURPOSE<br>OF<br>EXPENDITURE  | Category  | (See Categories listed at the top of this sc  | hedule)  | Description                        |   |                    |
|   | Check if travel outside of Texas. Complete Schedule T. Check if Austir                                  |   |          | n, TX, officeholder living expense |   |                    |
| Complete <u>ONLY</u> if direct expenditure to benefit C/  |   | ate / Officeholder name   |          | Office sought                      |   | Office held        |
| Date  | Payee nar   | ne  |          |                                    |   |                    |
| Amount (\$)<br>Reimbursement from political contributions intended  | Payee ad  | dress;  |          | City;                              | State;  | Zip Code           |
| PURPOSE<br>OF<br>EXPENDITURE  | Category (See Categories listed at the top of this schedule)  |   |          | Description                        |   |                    |
|   | Check if travel outside of Texas. Complete Schedule T. Check if Austin, TX, officeholder living expense |   |          |                                    |   | pense              |
| Complete ONLY if direct<br>expenditure to benefit C/OH  |   |   |          |                                    |   |                    |
| ATTACH ADDITIONAL COPIES OF THIS SCHEDULE AS NEEDED   |   |   |          |                                    |   |                    |

## PAYMENT MADE FROM POLITICAL CONTRIBUTIONS TO A BUSINESS OF C/OH

### SCHEDULE H

If the requested information is not applicable, **DO NOT include this page in the report.** 

| EXPENDITURE CATEGORIES FOR BOX 8(a)   |              |   |                                   |  |   |                         |  |
|---|--------------|---|-----------------------------------|--|---|-------------------------|--|
| Advertising Expense<br>Accounting/Banking<br>Consulting Expense<br>Contributions/Donations Made<br>Candidate/Officeholder/Polit |              | Event Expense<br>Fees<br>Food/Beverage Expense<br>Gift/Awards/Memorials Expense<br>Legal Services | Office O<br>Polling E<br>Printing | payment/Reimbursement<br>verhead/Rental Expense<br>Expense<br>Expense<br>/Wages/Contract Labor | Solicitation/Fundrais<br>Transportation Equip<br>Travel In District<br>Travel Out Of Distric<br>Other (enter a catego | oment & Related Expense |  |
| Credit Card Payment   |              | The Instruction Guide explai  | ns how to                         | complete this form.  |   |                         |  |
| <b>1</b> Total pages Schedule H:  | 2 FILER N    | AME   |                                   |  | 3 Filer ID (Ethic   | s Commission Filers)    |  |
| 4 Date  | 5 Business   | name  |                                   |  |   |                         |  |
| <b>6</b> Amount (\$)  | 7 Business   | address;  |                                   | City;  | State;  | Zip Code                |  |
| 8<br>PURPOSE<br>OF<br>EXPENDITURE   | (a) Category | (See Categories listed at the top of this s   | schedule)                         | (b) Description  |   |                         |  |
|   | (c)          | Check if travel outside of Texas. Complete So   | chedule T.                        | Check if Austin  | , TX, officeholder living e   | expense                 |  |
| 9 Complete <u>ONLY</u> if direct expenditure to benefit C/C   |              | ate / Officeholder name   |                                   | Office sought  |   | Office held             |  |
| Date  | Business     | name  |                                   |  |   |                         |  |
| Amount (\$)   | Business     | address;  |                                   | City;  | State;  | Zip Code                |  |
| PURPOSE<br>OF<br>EXPENDITURE  | Category     | (See Categories listed at the top of this s   | chedule)                          | Description  |   |                         |  |
|   |              | Check if travel outside of Texas. Complete Sc   | hedule T.                         | Check if Austin,   | , TX, officeholder living e   | expense                 |  |
| Complete <u>ONLY</u> if direct expenditure to benefit C/C   |              | ate / Officeholder name   |                                   | Office sought  |   | Office held             |  |
| Date  | Business     | name  |                                   |  |   |                         |  |
| Amount (\$)   | Business     | address;  |                                   | City;  | State;  | Zip Code                |  |
| PURPOSE<br>OF<br>EXPENDITURE  | Category     | (See Categories listed at the top of this s   | schedule)                         | Description  |   |                         |  |
|   |              | Check if travel outside of Texas. Complete So   | chedule T.                        | Check if Austin  | , TX, officeholder living e   | expense                 |  |
| Complete <u>ONLY</u> if direct expenditure to benefit C/C   |              | ate / Officeholder name   |                                   | Office sought  |   | Office held             |  |
|   | ATT          | ACH ADDITIONAL COPIES   | OF THIS                           | SCHEDULE AS NEE  | DED   |                         |  |

## NON-POLITICAL EXPENDITURES MADE FROM POLITICAL CONTRIBUTIONS

SCHEDULE |

If the requested information is not applicable, DO NOT include this page in the report.

| The Instruction Guide explains how to complete this form. |   |                                    |                     |               |                  |  |
|---|---|------------------------------------|---------------------|---------------|------------------|--|
| <b>1</b> Total pages Schedule I:                          | 2 FILER NAME  |                                    | 3 Filer ID          | (Ethics Co    | mmission Filers) |  |
| 4 Date  | 5 Payee name  |                                    | I                   |               |                  |  |
| <b>6</b> Amount (\$)                                      | 7 Payee address;  | City                               |                     | State         | Zip Code         |  |
| 8<br>PURPOSE<br>OF<br>EXPENDITURE                         | (a)Category (See instructions for examples of acceptable categories.)     | (b) Description (See<br>required.) | e instructions rega | rding type of | information      |  |
| Date  | Payee name  |                                    |                     |               |                  |  |
| Amount (\$)   | Payee address;  | City                               |                     | State         | Zip Code         |  |
| PURPOSE<br>OF<br>EXPENDITURE                              | <b>Category</b> (See instructions for examples of acceptable categories.) | Description (See<br>required.)     | e instructions rega | rding type of | information      |  |
| Date  | Payee name  |                                    |                     |               |                  |  |
| Amount (\$)   | Payee address;  | City                               |                     | State         | Zip Code         |  |
| PURPOSE<br>OF<br>EXPENDITURE                              | <b>Category</b> (See instructions for examples of acceptable categories.) | Description (See<br>required.)     | e instructions rega | rding type of | information      |  |
| Date  | Payee name  |                                    |                     |               |                  |  |
| Amount (\$)   | Payee address;  | City                               |                     | State         | Zip Code         |  |
| PURPOSE<br>OF<br>EXPENDITURE                              | Category (See instructions for examples of acceptable categories.)        | Description (See<br>required.)     | e instructions rega | rding type of | information      |  |
| ATTACH ADDITIONAL COPIES OF THIS SCHEDULE AS NEEDED       |   |                                    |                     |               |                  |  |

Forms provided by Texas Ethics Commission

# INTEREST, CREDITS, GAINS, REFUNDS, AND CONTRIBUTIONS RETURNED TO FILER

SCHEDULE K

If the requested information is not applicable, DO NOT include this page in the report.

| The Instruction Guide explains how to complete this form.            | 1 Total pages Schedule K:                |
|--|--|
| 2 FILER NAME   | 3 Filer ID (Ethics Commission Filers)    |
| 4 Date 5 Name of person from whom amount is received                 | <b>8</b> Amount (\$)                     |
| <b>6</b> Address of person from whom amount is received; City; State | e; Zip Code                              |
| 7 Purpose for which amount is received Check if p                    | political contribution returned to filer |
| Date Name of person from whom amount is received                     | Amount (\$)                              |
| Address of person from whom amount is received; City; Stat           | te; Zip Code                             |
| Purpose for which amount is received Check if p                      | political contribution returned to filer |
| Date Name of person from whom amount is received                     | Amount (\$)                              |
| Address of person from whom amount is received; City; State          | e; Zip Code                              |
| Purpose for which amount is received Check if p                      | political contribution returned to filer |
| Date Name of person from whom amount is received                     | Amount (\$)                              |
| Address of person from whom amount is received; City; Stat           | te; Zip Code                             |
| Purpose for which amount is received Check if p                      | political contribution returned to filer |
| ATTACH ADDITIONAL COPIES OF THIS SCHEDULE                            | AS NEEDED                                |

#### IN-KIND CONTRIBUTIONS OR POLITICAL EXPENDITURES FOR TRAVEL OUTSIDE OF TEXAS

SCHEDULE T

If the requested information is not applicable, DO NOT include this page in the report.

| The Instruction Guide explains how to complete this form.   |   |              |                         |                             | 1 Total pages Schedule T:     |  |
|---|---|--------------|-------------------------|-----------------------------|-------------------------------|--|
| 2 FILER NAME 3 Filer ID (Ethics Commission Filers)  |   |              |                         |                             |                               |  |
| 4 Name of Contributor /   | <sup>/</sup> Corporation  | or Labor C   | Organization / Pledgo   | r / Payee                   |                               |  |
| Schedule F2 Schedule F4 Schedule G Schedule H Schedule COH-UC Schedule B-                         |   |              |                         |                             |                               |  |
| 6 Dates of travel   | 7 Name of   |              |                         |                             |                               |  |
|   | 8 Departu   | re city or n | ame of departure loc    | ation                       |                               |  |
|   | 9 Destinat  | ion city or  | name of destination     | location                    |                               |  |
| 10 Means of transportation  | ion   | 11 Purpo     | ose of travel (includin | g name of conference        | e, seminar, or other event)   |  |
| Name of Contributor   | / Corporation   | or Labor C   | Organization / Pledgo   | r / Payee                   |                               |  |
| Contribution / Expend   | liture reported   | l on:        |                         |                             |                               |  |
| Schedule A2   | Sche  | edule B      | Schedule B(J)           | Schedule C2                 | Schedule D Schedule F1        |  |
| Schedule F2   | School  | edule F4     | Schedule G              | Schedule H                  | Schedule COH-UC Schedule B-SS |  |
| Dates of travel   | Dates of travel         Name of person(s) traveling   |              |                         |                             |                               |  |
|   | Departure city or name of departure location  |              |                         |                             |                               |  |
|   | Destinat  | ion city or  | name of destination     | location                    |                               |  |
| Means of transportat  | Means of transportation Purpose of travel (including name of conference, seminar, or other event) |              |                         |                             |                               |  |
| Name of Contributor   | / Corporation   | or Labor C   | Organization / Pledgo   | r / Payee                   |                               |  |
| Contribution / Expend   | liture reported   | l on:        |                         |                             |                               |  |
| Schedule A2   | Schedu  | ile B [      | Schedule B(J)           | Schedule C2                 | Schedule D Schedule F1        |  |
| Schedule F2   | Schedu  | ıle F4 [     | Schedule G              | Schedule H                  | Schedule COH-UC Schedule B-SS |  |
| Dates of travel   | Name of person(s) traveling   |              |                         |                             |                               |  |
|   | Departure city or name of departure location  |              |                         |                             |                               |  |
|   | Destination city or name of destination location  |              |                         |                             |                               |  |
| Means of transportation Purpose of travel (including name of conference, seminar, or other event) |   |              |                         | e, seminar, or other event) |                               |  |
| ATTACH ADDITIONAL COPIES OF THIS SCHEDULE AS NEEDED   |   |              |                         |                             |                               |  |

## CANDIDATE / OFFICEHOLDER REPORT: DESIGNATION OF FINAL REPORT

## FORM C/OH - FR

|   | The Instruction Guide explains how to complete this form.   |   |  |  |  |  |  |
|---|---|---|--|--|--|--|--|
|   | •• Complete only if "Report Type" on page 1 is marked "Final Report" ••   |   |  |  |  |  |  |
| 1 | C/OH N  | NAME 2  | 2 Filer ID (Ethics Commission Filers)    |  |  |  |  |
| 3 | SIGNA   | ATURE   |  |  |  |  |  |
|   |   |   |  |  |  |  |  |
|   | I do not expect any further political contributions or political expenditures in connection with my candidacy. I understand that designating a report as a final report terminates my campaign treasurer appointment. I also understand that I may not accept any campaign contributions or make any campaign expenditures without a campaign treasurer appointment on file.  |   |  |  |  |  |  |
|   |   | Signature   | of Candidate / Officeholder              |  |  |  |  |
| 4 |   | WHO IS NOT AN OFFICEHOLDER<br>nplete A & B below <i>only</i> if you are not an officeholder. ••   |  |  |  |  |  |
|   | Α.  | CAMPAIGN FUNDS  |  |  |  |  |  |
|   | Chec  | k only one:   |  |  |  |  |  |
|   | I do not have unexpended contributions or unexpended interest or income earned from political contributions.  |   |  |  |  |  |  |
|   | I have unexpended contributions or unexpended interest or income earned from political contributions. I understand that I may not convert unexpended political contributions or unexpended interest or income earned on political contributions to personal use. I also understand that I must file an annual report of unexpended contributions and that I may not retain unexpended contributions or unexpended interest or income earned on political contributions after filing this final report. Further, I understand that I must dispose of unexpended political contributions and unexpended interest or income earned on political contributions and unexpended interest or income earned on political contributions and unexpended interest or income earned on political contributions and unexpended interest or income earned on political contributions and unexpended interest or income earned on political contributions and unexpended interest or income earned on political contributions and unexpended interest or income earned on political contributions and unexpended interest or income earned on political contributions and unexpended interest or income earned on political contributions and unexpended interest or income earned on political contributions and unexpended interest or income earned on political contributions in accordance with the requirements of Election Code, § 254.204. |   |  |  |  |  |  |
|   | B. ASSETS   |   |  |  |  |  |  |
|   | Chec  | k only one:   |  |  |  |  |  |
|   |   | I do not retain assets purchased with political contributions or interest or other income   | from political contributions.            |  |  |  |  |
|   | I do retain assets purchased with political contributions or interest or other income from political contributions. I understand that I may not convert assets purchased with political contributions or interest or other income from political contributions to personal use. I also understand that I must dispose of assets purchased with political contributions in accordance with the requirements of Election Code, § 254.204.   |   |  |  |  |  |  |
|   |   | Sig   | gnature of Candidate                     |  |  |  |  |
| 5 |   | EHOLDER<br>nplete this section <i>only</i> if you are an officeholder ••  |  |  |  |  |  |
|   |   | I am aware that I remain subject to filing requirements applicable to an officeholder who doe<br>file. I am also aware that I will be required to file reports of unexpended contributions if, a<br>an officeholder, I retain political contributions, interest or other income from political contri-<br>political contributions or interest or other income from political contributions. | after filing the last required report as |  |  |  |  |
|   |   | Sigr  | nature of Officeholder                   |  |  |  |  |

## CODE OF FAIR CAMPAIGN PRACTICES

| OFFICE USE ONLY   |   |                                    |       |  |                |  |  |
|---|---|------------------------------------|-------|--|----------------|--|--|
| Pursuant to chapter 258 of<br>political committee is enco<br>Campaign Practices. The C<br>authority upon submission<br>form. Candidates or poli<br>current campaign treasurer<br>1997, may subscribe to the<br><i>Subscription to the Code of</i> | Date Received Date Hand-delivered or R Date Processed Date Imaged | Postmarked                         |       |  |                |  |  |
| 1 ACCOUNT NUMBER  | 2 TYPE OF FIL   | ER                                 |       | _  |                |  |  |
| (Ethics Commission Filers)  | CANDIDATE   |                                    | POI   | LITICAL COMM                                     | ITTEE          |  |  |
|   | If filing as a cand<br>then read and sig                          | idate, complete boxes<br>n page 2. |       | ing for a political cor<br>es 7 and 8, then read | · · ·          |  |  |
| 3 NAME OF CANDIDATE   | TITLE (Dr., Mr., Ms., etc.)                                       | FIRST                              |       | МІ   |                |  |  |
| (PLEASE TYPE OR PRINT)  |   |                                    |       |  |                |  |  |
|   | NICKNAME  | LAST                               |       | SUFFIX (SR., J                                   | R., III, etc.) |  |  |
| <b>4</b> TELEPHONE NUMBER<br>OF CANDIDATE   | AREA CODE   | PHONE NU                           | MBER  | EXTENSION  |                |  |  |
| (PLEASE TYPE OR PRINT)  | ( )   |                                    |       |  |                |  |  |
| 5 ADDRESS OF CANDIDATE  | STREET / PO BOX;  | APT / SUITE #;                     | CITY; | STATE;   | ZIP CODE       |  |  |
| (PLEASE TYPE OR PRINT)  |   |                                    |       |  |                |  |  |
| 6 OFFICE SOUGHT<br>BY CANDIDATE   |   |                                    |       |  |                |  |  |
| (PLEASE TYPE OR PRINT)  |   |                                    |       |  |                |  |  |
| 7 NAME OF COMMITTEE   |   |                                    |       |  |                |  |  |
| (PLEASE TYPE OR PRINT)  |   |                                    |       |  |                |  |  |
| 8 NAME OF CAMPAIGN<br>TREASURER   | TITLE (Dr., Mr., Ms., etc.)                                       | FIRST                              |       | MI   |                |  |  |
| (PLEASE TYPE OR PRINT)  |   |                                    |       |  |                |  |  |
|   | NICKNAME  | LAST                               |       | SUFFIX (SR., J                                   | к., III, ӨІС.) |  |  |
|   | GO GO   | TO PAGE 2                          |       |  |                |  |  |

## **CODE OF FAIR CAMPAIGN PRACTICES**

There are basic principles of decency, honesty, and fair play that every candidate and political committee in this state has a moral obligation to observe and uphold, in order that, after vigorously contested but fairly conducted campaigns, our citizens may exercise their constitutional rights to a free and untrammeled choice and the will of the people may be fully and clearly expressed on the issues.

#### THEREFORE:

- (1) I will conduct the campaign openly and publicly and limit attacks on my opponent to legitimate challenges to my opponent's record and stated positions on issues.
- (2) I will not use or permit the use of character defamation, whispering campaigns, libel, slander, or scurrilous attacks on any candidate or the candidate's personal or family life.
- (3) I will not use or permit any appeal to negative prejudice based on race, sex, religion, or national origin.
- (4) I will not use campaign material of any sort that misrepresents, distorts, or otherwise falsifies the facts, nor will I use malicious or unfounded accusations that aim at creating or exploiting doubts, without justification, as to the personal integrity or patriotism of my opponent.
- (5) I will not undertake or condone any dishonest or unethical practice that tends to corrupt or undermine our system of free elections or that hampers or prevents the full and free expression of the will of the voters, including any activity aimed at intimidating voters or discouraging them from voting.
- (6) I will defend and uphold the right of every qualified voter to full and equal participation in the electoral process, and will not engage in any activity aimed at intimidating voters or discouraging them from voting.
- (7) I will immediately and publicly repudiate methods and tactics that may come from others that I have pledged not to use or condone. I shall take firm action against any subordinate who violates any provision of this code or the laws governing elections.

I, the undersigned, candidate for election to public office in the State of Texas or campaign treasurer of a political committee, hereby voluntarily endorse, subscribe to, and solemnly pledge myself to conduct the campaign in accordance with the above principles and practices.

Signature

Date

|  | <b>VERNMENT OFFICER</b><br><b>RE STATEMENT</b><br>(Instructions for completing and filing |                                   | FORM CIS                            |  |  |  |
|--|---|-----------------------------------|-------------------------------------|--|--|--|
| This questionnaire re  | eflects changes made to the law by H.B  | . 23, 84th Leg., Regular Session. | OFFICE USE ONLY                     |  |  |  |
| This is the notice to government officer lin accordance with (   | Date Received   |                                   |                                     |  |  |  |
| <sup>1</sup> Name of Local G   | overnment Officer   |                                   |                                     |  |  |  |
| 2 Office Held  |   |                                   |                                     |  |  |  |
| 3 Name of vendor d<br>Code   | described by Sections 176.001(7) and  | d 176.003(a), Local Government    |                                     |  |  |  |
| 4 Description of th<br>with vendor nam   | ne nature and extent of each employn<br>ned in item 3.                                    | nent or other business relationsh | p and each family relationship      |  |  |  |
|  | ed by the local government officer a<br>ned in item 3 exceeds \$100 during tl             |                                   |                                     |  |  |  |
| Date Gift Accept   | ted Description of G  | lift                              |                                     |  |  |  |
| Date Gift Accept   | ed Description of G   | ift                               |                                     |  |  |  |
| Date Gift Accepte  | ed Description of Gift  | t                                 |                                     |  |  |  |
|  | (attach addition  | nal forms as necessary)           |                                     |  |  |  |
| 6 SIGNATURE I swear under penalty of perjury that the above statement is true and correct. I acknowledge that the disclosure applies to each family member (as defined by Section 176.001(2), Local Government Code) of this local government officer. I also acknowledge that this statement covers the 12-month period described by Section 176.003(a)(2)(B), Local Government Code. |   |                                   |                                     |  |  |  |
| Signature of Local Government Officer  |   |                                   |                                     |  |  |  |
|  | Please com  | plete either option below:        |                                     |  |  |  |
| (1) Affidavit  |   |                                   |                                     |  |  |  |
| NOTARY STAMP/SE  | EAL   |                                   |                                     |  |  |  |
| Sworn to and subscribe   | ed before me by   | this the                          | day of,                             |  |  |  |
| 20, to cert  | tify which, witness my hand and seal of office.   |                                   |                                     |  |  |  |
| Signature of officer adminis   | stering oath Printed name of o  | fficer administering oath         | Title of officer administering oath |  |  |  |
| (2) Unsworn Declara  | ation   | OR                                |                                     |  |  |  |
| My name is   |   | , and my date of birth is         |                                     |  |  |  |
| My address is  |   |                                   | ,,                                  |  |  |  |
|  | (street)  |                                   | e) (zip code) (country)             |  |  |  |
| Executed in  | County, State of  | , on theday of<br>(month)         | , 20<br>(year)                      |  |  |  |
|  |   | Signature of Local Gover          | mment Officer (Declarant)           |  |  |  |

### LOCAL GOVERNMENT OFFICER CONFLICTS DISCLOSURE STATEMENT

Section 176.003 of the Local Government Code requires certain local government officers to file this form. A "local government officer" is defined as a member of the governing body of a local governmental entity; a director, superintendent, administrator, president, or other person designated as the executive officer of a local governmental entity; or an agent of a local governmental entity who exercises discretion in the planning, recommending, selecting, or contracting of a vendor. This form is required to be filed with the records administrator of the local governmental entity not later than 5 p.m. on the seventh business day after the date on which the officer becomes aware of the facts that require the filing of this statement.

A local government officer commits an offense if the officer knowingly violates Section 176.003, Local Government Code. An offense under this section is a misdemeanor.

Refer to chapter 176 of the Local Government Code for detailed information regarding the requirement to file this form.

#### INSTRUCTIONS FOR COMPLETING THIS FORM

The following numbers correspond to the numbered boxes on the other side.

1. Name of Local Government Officer. Enter the name of the local government officer filing this statement.

2. Office Held. Enter the name of the office held by the local government officer filing this statement.

**3.** Name of vendor described by Sections 176.001(7) and 176.003(a), Local Government Code. Enter the name of the vendor described by Section 176.001(7), Local Government Code, if the vendor: a) has an employment or other business relationship with the local government officer or a family member of the officer as described by Section 176.003(a)(2)(A), Local Government Code; b) has given to the local government officer or a family member of the officer one or more gifts as described by Section 176.003(a)(2)(B), Local Government Code; or c) has a family relationship with the local government officer as defined by Section 176.001(2-a), Local Government Code.

**4.** Description of the nature and extent of each employment or other business relationship and each family relationship with vendor named in item 3. Describe the nature and extent of the employment or other business relationship the vendor has with the local government officer or a family member of the officer as described by Section 176.003(a)(2)(A), Local Government Code, and each family relationship the vendor has with the local government officer as defined by Section 176.001(2-a), Local Government Code.

**5.** List gifts accepted, if the aggregate value of the gifts accepted from vendor named in item 3 exceeds \$100. List gifts accepted during the 12-month period (described by Section 176.003(a)(2)(B), Local Government Code) by the local government officer or family member of the officer from the vendor named in item 3 that in the aggregate exceed \$100 in value.

**6. Signature**. Signature of local government officer. Complete this section after you finish the rest of this report. You have the option to either: (1) take the completed form to a notary public where you will sign above the first line that says "Signature of Local Government Officer" (an electronic signature is not acceptable) and your signature will be notarized, or (2) sign above both lines that say "Signature of Local Government Officer" (an electronic signature of Local Government)" (an electronic signature is not acceptable), and fill out the unsworn declaration section.

**Local Government Code § 176.001(2-a)**: "Family relationship" means a relationship between a person and another person within the third degree by consanguinity or the second degree by affinity, as those terms are defined by Subchapter B, Chapter 573, Government Code.

#### Local Government Code § 176.003(a)(2)(A):

(a) A local government officer shall file a conflicts disclosure statement with respect to a vendor if:

(2) the vendor:

(A) has an employment or other business relationship with the local government officer or a family member of the officer that results in the officer or family member receiving taxable income, other than investment income, that exceeds \$2,500 during the 12-month period preceding the date that the officer becomes aware that:

(i) a contract between the local governmental entity and vendor has been executed; or(ii) the local governmental entity is considering entering into a contract with the vendor.

| CONFLICT OF INTEREST QUESTIONNAIRE<br>For vendor doing business with local governmental entity   | FORM CIQ   |  |  |  |
|--|--|--|--|--|
| This questionnaire reflects changes made to the law by H.B. 23, 84th Leg., Regular Session.  | OFFICE USE ONLY  |  |  |  |
| This questionnaire is being filed in accordance with Chapter 176, Local Government Code, by a vendor who has a business relationship as defined by Section 176.001(1-a) with a local governmental entity and the vendor meets requirements under Section 176.006(a).   | Date Received  |  |  |  |
| By law this questionnaire must be filed with the records administrator of the local governmental entity not later than the 7th business day after the date the vendor becomes aware of facts that require the statement to be filed. <i>See</i> Section 176.006(a-1), Local Government Code.   |  |  |  |  |
| A vendor commits an offense if the vendor knowingly violates Section 176.006, Local Government Code. An offense under this section is a misdemeanor.   |  |  |  |  |
| 1 Name of vendor who has a business relationship with local governmental entity.   |  |  |  |  |
| 2 Check this box if you are filing an update to a previously filed questionnaire. (The law re completed questionnaire with the appropriate filing authority not later than the 7th busines you became aware that the originally filed questionnaire was incomplete or inaccurate.)   | s day after the date on which  |  |  |  |
| 3 Name of local government officer about whom the information is being disclosed.  |  |  |  |  |
| Name of Officer  |  |  |  |  |
| <ul> <li>4 Describe each employment or other business relationship with the local government offi officer, as described by Section 176.003(a)(2)(A). Also describe any family relationship wit Complete subparts A and B for each employment or business relationship described. Attac CIQ as necessary.</li> <li>A. Is the local government officer or a family member of the officer receiving or I</li> </ul> | th the local government officer.<br>In additional pages to this Form |  |  |  |
| other than investment income, from the vendor?   |  |  |  |  |
| Yes No   |  |  |  |  |
| B. Is the vendor receiving or likely to receive taxable income, other than investment<br>of the local government officer or a family member of the officer AND the taxable<br>local governmental entity?   |  |  |  |  |
| Yes No   |  |  |  |  |
| 5 Describe each employment or business relationship that the vendor named in Section 1 maintains with a corporation or<br>other business entity with respect to which the local government officer serves as an officer or director, or holds an<br>ownership interest of one percent or more.   |  |  |  |  |
| 6 Check this box if the vendor has given the local government officer or a family member as described in Section 176.003(a)(2)(B), excluding gifts described in Section 176.0  |  |  |  |  |
|  |  |  |  |  |
| Signature of vendor doing business with the governmental entity  | Date   |  |  |  |

## CONFLICT OF INTEREST QUESTIONNAIRE For vendor doing business with local governmental entity

A complete copy of Chapter 176 of the Local Government Code may be found at http://www.statutes.legis.state.tx.us/ Docs/LG/htm/LG.176.htm. For easy reference, below are some of the sections cited on this form.

Local Government Code § 176.001(1-a): "Business relationship" means a connection between two or more parties based on commercial activity of one of the parties. The term does not include a connection based on:

(A) a transaction that is subject to rate or fee regulation by a federal, state, or local governmental entity or an agency of a federal, state, or local governmental entity;

(B) a transaction conducted at a price and subject to terms available to the public; or

(C) a purchase or lease of goods or services from a person that is chartered by a state or federal agency and that is subject to regular examination by, and reporting to, that agency.

#### Local Government Code § 176.003(a)(2)(A) and (B):

(a) A local government officer shall file a conflicts disclosure statement with respect to a vendor if:

(2) the vendor:

(A) has an employment or other business relationship with the local government officer or a family member of the officer that results in the officer or family member receiving taxable income, other than investment income, that exceeds \$2,500 during the 12-month period preceding the date that the officer becomes aware that

 $(\bar{\textbf{i}})$  a contract between the local governmental entity and vendor has been executed; or

(ii) the local governmental entity is considering entering into a contract with the vendor;

(B) has given to the local government officer or a family member of the officer one or more gifts that have an aggregate value of more than \$100 in the 12-month period preceding the date the officer becomes aware that:

- (i) a contract between the local governmental entity and vendor has been executed; or
- (ii) the local governmental entity is considering entering into a contract with the vendor.

#### Local Government Code § 176.006(a) and (a-1)

(a) A vendor shall file a completed conflict of interest questionnaire if the vendor has a business relationship with a local governmental entity and:

(1) has an employment or other business relationship with a local government officer of that local governmental entity, or a family member of the officer, described by Section 176.003(a)(2)(A);

(2) has given a local government officer of that local governmental entity, or a family member of the officer, one or more gifts with the aggregate value specified by Section 176.003(a)(2)(B), excluding any gift described by Section 176.003(a-1); or

(3) has a family relationship with a local government officer of that local governmental entity.

(a-1) The completed conflict of interest questionnaire must be filed with the appropriate records administrator not later than the seventh business day after the later of:

(1) the date that the vendor:

(A) begins discussions or negotiations to enter into a contract with the local governmental entity; or

(B) submits to the local governmental entity an application, response to a request for proposals or bids, correspondence, or another writing related to a potential contract with the local governmental entity; or

(2) the date the vendor becomes aware:

(A) of an employment or other business relationship with a local government officer, or a family member of the officer, described by Subsection (a);

(B) that the vendor has given one or more gifts described by Subsection (a); or

(C) of a family relationship with a local government officer.

All information is required to be provided unless indicated as optional.

#### **DECLARATION OF WRITE-IN CANDIDACY FOR PRESIDENT**

To: Secretary of State

TO BE COMPLETED BY FILING OFFICER:

| I declare that I am a write-in candidate for  | or the office o | f President.       |   |                    |                    |                 |
|---|-----------------|--------------------|---|--------------------|--------------------|-----------------|
| FULL NAME (First, Middle, Last)   |                 |                    | PRINT NAME AS YOU WANT IT TO APPEAR ON THE LIST OF DECLARED WRITE-IN CANDIDATES |                    |                    |                 |
|   |                 |                    | DECLARED WRIT   | IE-IN CANDIDATES   |                    |                 |
| PERMANENT RESIDENCE ADDRESS (Do not include a P.O. Box or Rural   |                 |                    |   | G ADDRESS (Addres  | s for which you re | eceive campaign |
| Route. If you do not have a residence address, describe location of residence.)   |                 |                    | related correspo  |                    |                    |                 |
| СІТҮ  | STATE           | ZIP                | CITY  |                    | STATE              | ZIP             |
| PUBLIC EMAIL ADDRESS (If available)     DATE OF BIRTH       /     /   |                 | /                  | ARE YOU A NATUR   | RAL BORN CITIZEN   | ?                  |                 |
| TELEPHONE CONTACT INFORMATION (C  | ptional)        |                    | LEN   | GTH OF RESIDENCE   | IN THE UNITED ST   | ATES            |
| Home:   |                 |                    |   |                    | year(s)            |                 |
| Work:   |                 |                    |   |                    | month(s)           |                 |
| DECLA   | RATION OI       | WRITE-IN CA        | NDIDACY FOF   | R VICE PRESIDEN    | ΙТ                 |                 |
| To: Secretary of State  |                 |                    |   |                    |                    |                 |
| I consent to be a write-in candidate for t  | he office of Vi | ce President for _ |   |                    |                    |                 |
| FULL NAME (First, Middle, Last)   |                 |                    | PRINT NAME AS YOU WANT IT TO APPEAR ON THE LIST OF                              |                    |                    |                 |
|   |                 |                    | DECLARED WRITE-IN CANDIDATES  |                    |                    |                 |
|   |                 |                    |   |                    |                    |                 |
| PERMANENT RESIDENCE ADDRESS (Do r   | ot include a F  | P.O. Box or Rural  | PUBLIC MAILING  | G ADDRESS (Address | for which you rec  | eive campaign   |
| Route. If you do not have a residence ad  | dress, descrit  | e location of      | related correspondence.)  |                    |                    |                 |
| residence.)   |                 |                    |   |                    |                    |                 |
| CITY  | STATE           | ZIP                | CITY  |                    | STATE              | ZIP             |
| PUBLIC EMAIL ADDRESS (If available)   |                 | DATE OF BIRTH      |   | ARE YOU A NATU     | RAL BORN CITIZEN   | 2               |
|   |                 | ,                  |   |                    |                    |                 |
|   |                 |                    | /   |                    |                    |                 |
| TELEPHONE CONTACT INFORMATION (C  | ptional)        |                    | LEN   | GTH OF RESIDENCE   | IN THE UNITED ST   | ATES            |
| Home:   |                 |                    | year(s)   |                    |                    |                 |
| Work:   |                 |                    |   |                    |                    |                 |
|   | ,               |                    |   |                    | month(s)           | 0 11 1 1        |
| I hereby give my written consent to be a vice presidential candidate for<br>the office of President of the United States. |                 |                    |   | а                  | Declared Write-In  | Candidate for   |
|   |                 |                    | X   |                    |                    |                 |
|   |                 | F VICE PRESIDENTIA | L CANDIDATE   |                    |                    |                 |
| I hereby swear that the foregoing statem  |                 |                    | -   |                    |                    |                 |
|   | X               |                    |   |                    |                    |                 |
|   |                 | 4                  |   | F CANDIDATE        |                    | <del></del>     |
| Sworn to and subscribed before me at  |                 | , this the _       |   |                    |                    |                 |
| _   |                 | _                  |   |                    |                    | EAL             |
|   |                 |                    |   |                    |                    |                 |
| Signature of Officer Administering Oath <sup>1</sup> Title of   |                 |                    |   |                    |                    |                 |

Date Received

Signature of Filing Officer

#### INSTRUCTIONS FOR DECLARED WRITE-IN CANDIDATE FOR PRESIDENT

The declaration must be accompanied by written statements signed by each presidential elector giving their written consent to be an elector for the presidential candidate (total of 38 electors).

The declaration, together with the written statements, must be filed with the Secretary of State no later than 5:00 p.m. of the 78th day before election day.

#### FOOTNOTE

<sup>1</sup>All oaths, affidavits, or affirmations made within this State may be administered and a certificate of the fact given by a judge, clerk, or commissioner of any court of record, a notary public, a justice of the peace, city secretary, and the Secretary of State of Texas. Se requiere toda la información, a menos que haya alguna indicación que no es obligatoria.

#### DECLARACIÓN DE SER UN CANDIDATO POR INSERCIÓN PARA EL PUESTO OFICIAL DE PRESIDENTE

A: Secretario del Estado de Texas

| Por la presente declaro que soy un candidato por inserción para el puesto oficial de Presidente de los Estados Unidos. |                        |   |  |  |  |
|--|------------------------|---|--|--|--|
| NOMBRE COMPLETO (Nombre de Pila, Segundo Nombre,   |                        | ESCRIBA SU NOMBRE COMO DESEA QUE APAREZCA SOBRE LA LISTA DE     |  |  |  |
|  |                        | CANDIDATOS DECLARADOS PARA INSERCIÓN                            |  |  |  |
| CORREO ELECTRÓNICO (Optativo)  |                        |   |  |  |  |
|  |                        |   |  |  |  |
| DIRECCION DE RESIDENCIA PERMANENTE: Calle  |                        | DIRECCION POSTAL (Si es diferente a su dirección de residencia) |  |  |  |
| departamento. Si no tiene, describa la localidad de su   | i residencia. No       |   |  |  |  |
| incluya su caja postal ni ruta rural.  |                        |   |  |  |  |
|  |                        |   |  |  |  |
| CIUDAD ZONA POSTAL   |                        | CIUDAD ZONA POSTAL  |  |  |  |
| NÚMERO DE TELÉFONO – Incluya el código de área<br>(Optativo)   | FECHA DE<br>NACIMIENTO | TIEMPO QUE HA RESIDIDO EN LOS ESTADOS UNIDOS                    |  |  |  |
| (optatio)  |                        |   |  |  |  |
| DE SU OFICINA:   |                        | ¿ES USTED CIUDADANO POR NACIMIENTO DEL LOS ESTADOS UNIDOS?      |  |  |  |
| DE SU DOMICILIO:   |                        |   |  |  |  |
|  | ANDIDATO INSERCI       | ÓN PARA EL PUESTO OFICIAL DE VICEPRESIDENTE                     |  |  |  |
| A: Secretario del Estado de Texas  |                        |   |  |  |  |
| Doy consentimiento de ser candidato por inserción para   | el puesto oficial d    | e vicepresidente de los Estados Unidos para .)                  |  |  |  |
| NOMBRE COMPLETO (Nombre de Pila, Segundo Nombre  |                        | ESCRIBA SU NOMBRE COMO DESEA QUE APAREZCA SOBRE LA LISTA DE     |  |  |  |
|  |                        | CANDIDATOS DECLARADOS PARA INSERCION                            |  |  |  |
| CORREO ELECTRÓNICO (Optativo)  |                        |   |  |  |  |
| CORRECTED ELECTRONICO (Optativo)   |                        |   |  |  |  |
|  |                        |   |  |  |  |
| DIRECCIÓN DE RESIDENCIA PERMANENTE: Calle  | •                      | DIRECCIÓN POSTAL (Si es diferente a su dirección de residencia) |  |  |  |
| departamento. Si no tiene, describa la localidad de su<br>incluya su caja postal ni ruta rural.                        | i residencia. No       |   |  |  |  |
| niciuya su caja postal ni ruta rurai.  |                        |   |  |  |  |
| CIUDAD ZONA POSTA  | I                      | CIUDAD ZONA POSTAL  |  |  |  |
| NUMERO DE TELÉFONO – Incluya el código de área   | FECHA DE               | TIEMPO QUE HA RESIDIDO EN LOS ESTADOS UNIDOS                    |  |  |  |
| (Optativo)   | NACIMIENTO             |   |  |  |  |
|  |                        | ¿ES USTED CIUDADANO POR NACIMIENTO DEL LOS ESTADOS UNIDOS?      |  |  |  |
| DE SU OFICINA:   |                        |   |  |  |  |
| DE SU DOMICILIO:   |                        |   |  |  |  |
| Juro que las anteriores declaraciones incluidas en mi soli   | icitud son verdade     | ras y están correctas en todo sentido.                          |  |  |  |
|  |                        |   |  |  |  |
|  |                        | X   |  |  |  |
|  |                        | FIRMA DEL CANDIDATO   |  |  |  |
| Jurado y suscrito ante mi en   | . este día             | de .20  |  |  |  |
|  | ) cote una             | ,,,,  |  |  |  |
|  |                        |   |  |  |  |
| Firma del oficial quien administra el juramento  |                        | Titulo del oficial quien administra el juramento                |  |  |  |
| Por le presente doy mi consentimiento por escrito de se<br>un candidato por inserción declarado para el puesto ofic    |                        |   |  |  |  |
|  |                        |   |  |  |  |
| X  |                        |   |  |  |  |
|  | Firn                   | na del Candidato para Vicepresidente                            |  |  |  |
| TO BE COMPLETED BY FILING OFFICER:   |                        |   |  |  |  |
|  |                        |   |  |  |  |
|  |                        | Date received Signature of Filing Officer                       |  |  |  |

#### INSTRUCCIONES PARA UN CANDIDATO POR INSERCIÓN DECLARADO PARA EL PUESTO OFICIAL DE PRESIDENTE

La declaración deberá de estar acompañada por las declaraciones por escrito que fueron firmadas por cada elector presidencial en las cuales dan su consentimiento por escrito de ser elector para el candidato para presidente (38 electores en total).

La declaración, junto con estos documentos, deberá estar archivada con el Secretario de Estado a más tardar para las 5 de la tarde 78 días antes de la elección general.

#### NOTA

<sup>1</sup> Todo juramento, testimonio o afirmación hecho dentro de este Estado se podrá administrar y se podrá dar un certificado del hecho por un juez, escribano, o comisionado de alguna corte de registro, un notario público, un juez de paz, secretario de la ciudad, y el Secretario del Estado de Texas.